## 15.2 DECREASE IN HEAT REMOVAL BY THE SECONDARY SYSTEM

A number of transients and accidents have been postulated which could result in a reduction of the capacity of the secondary system to remove heat generated in the reactor coolant system. Detailed analyses are presented in this section for the following events which have been identified as more limiting than the others:

- Steam pressure regulator malfunction or failure that results in decreasing steamflow.
- Loss of external electrical load.
- Turbine trip.
- Inadvertent closure of main steam isolation valves.
- Loss of condenser vacuum and other events resulting in turbine trip.
- Loss of nonemergency ac power to the station auxiliaries.
- Loss of normal feedwater flow.
- Feedwater system pipe break.

All of the accidents in this section have been analyzed. It has been determined that the most severe radiological consequences will result from the loss of ac power accident discussed in subsection 15.2.6. Therefore, the radiological consequences are only reported for that limiting case.

#### 15.2.1 STEAM PRESSURE REGULATOR MALFUNCTION OR FAILURE THAT RESULTS IN DECREASING STEAMFLOW

There are no steam pressure regulators in the VEGP units whose failure or malfunction could cause a steamflow transient.

# 15.2.2 LOSS OF EXTERNAL ELECTRICAL LOAD

#### 15.2.2.1 Identification of Causes and Accident Description

A major load loss on the plant can result from loss of external electrical load due to some electrical system disturbance. Offsite ac power remains available to operate plant components, such as the reactor coolant pumps; as a result, the onsite emergency diesel generators are not required to function for this event. Following the loss of generator load, an immediate fast closure of the turbine control valves will occur. The automatic turbine bypass system would accommodate the excess steam generation. Reactor coolant temperatures and pressure do not significantly increase if the turbine bypass system and pressurizer pressure control system are functioning properly. If the condenser is not available, the excess steam generation is relieved

to the atmosphere. Additionally, main feedwater flow is lost if the condenser is not available. For this situation, feedwater flow is maintained by the auxiliary feedwater system.

For a loss of external electrical load without subsequent turbine trip, no direct reactor trip signal would be generated, and the plant would be expected to trip from the reactor protection system if a safety limit were approached. A continued steam load of approximately 5 percent would exist after total loss of external electrical load, because of the steam demand of plant auxiliaries.

In the event that a safety limit is approached, protection would be provided by high pressurizer pressure, high pressurizer water level, low-low steam generator water level, and overtemperature  $\Delta T$  trip. Voltage and frequency relays associated with the reactor coolant pump provide no additional safety function for this event. Following a complete loss of external load, the maximum turbine overspeed would be approximately 8 to 9 percent, resulting in an overfrequency of less than 6 Hz. This resulting overfrequency is not expected to damage the voltage and frequency sensors in any way. Any degradation in their performance could be ascertained at that time. Any increased frequency to the reactor coolant pump motors will result in slightly increased flowrate and subsequent additional margin to safety limits. For postulated loss of load and subsequent turbine-generator overspeed, any overfrequency condition is not seen by other safety-related pump motors, reactor protection system equipment, or other safetyrelated loads. Safety-related loads are alternately supplied from offsite power or from emergency diesels. Reactor protection system equipment is supplied from the 120-V-ac instrument power supply system, which, in turn, is supplied from the inverters; the inverters are supplied from a dc bus energized from batteries or by a rectified ac voltage from safety-related buses.

In the event that the steam dump valves fail to open following a large loss of load, the steam generator safety valves may lift, and the reactor may be tripped by the high pressurizer pressure signal, the high pressurizer water level signal, or the overtemperature  $\Delta T$  signal. The steam generator shell side pressure and reactor coolant temperature will increase rapidly. The pressurizer safety valves and steam generator safety valves are, however, sized to protect the reactor coolant system (RCS) and steam generator against overpressure for all load losses, without assuming the operation of the turbine bypass system, pressurizer spray, pressurizer power-operated relief valves, automatic rod cluster control assembly control, or direct reactor trip on turbine trip.

The steam generator safety valve capacity is sized to remove the steam flow at the engineered safety features rating (105 percent of steam flow at rated power) from the steam generator, without exceeding 110 percent of the steam system design pressure. The pressurizer safety valve capacity is sized to accommodate a complete loss of heat sink with the plant initially operating at the maximum calculated turbine load along with operation of the steam generator safety valves. The pressurizer safety valves are then able to relieve sufficient steam to maintain the RCS pressure within 110 percent of the RCS design pressure.

A more complete discussion of overpressure protection can be found in reference 1.

A loss of external load is classified as an American Nuclear Society Condition II event, fault of moderate frequency.

A loss-of-external-load event results in a nuclear steam supply system transient that is bounded by the turbine trip event analyzed in subsection 15.2.3. Therefore, a detailed transient analysis is not presented for the loss-of-external-load event.

The primary side transient is caused by a decrease in heat transfer capability from primary to secondary due to a rapid termination of steam flow to the turbine, accompanied by an automatic reduction of feedwater flow. (Should feedwater flow not be reduced, a larger heat sink would be available and the transient would be less severe.) Termination of steam flow to the turbine

following a loss of external load occurs due to automatic fast closure of the turbine control valves in approximately 0.3 s. Following a turbine trip event, termination of steam flow occurs via turbine stop valve closure, which occurs in approximately 0.1 s. Therefore, the transient in primary pressure, temperature, and water volume will be less severe for the loss of external load than for the turbine trip due to a slightly slower loss of heat transfer capability.

The protection available to mitigate the consequences of a loss of external load is the same as that for a turbine trip, as listed in table 15.0.8-1.

# 15.2.2.2 Analysis of Effects and Consequences

Refer to paragraph 15.2.3.2 for the method used to analyze the limiting transient (turbine trip) in this grouping of events. The results of the turbine trip event analysis bound those expected for the loss of external load, as discussed in paragraph 15.2.2.1.

Plant systems and equipment which may be required to function to mitigate the effects of a complete loss of load are discussed in subsection 15.0.8 and listed in table 15.0.8-1.

The reactor protection system may be required to function to terminate core heat input and to prevent departure from nucleate boiling. Depending on the magnitude of the load loss, pressurizer safety valves and/or steam generator safety valves may be required to open to maintain system pressures below allowable limits. No single active failure will prevent operation of any system required to function. Normal reactor control systems and engineered safety systems are not required to function. The auxiliary feedwater system may, however, be automatically actuated following a loss of main feedwater; this will further mitigate the effects of the transient.

# 15.2.2.3 <u>Conclusions</u>

Based on results obtained for the turbine trip event and considerations described in paragraph 15.2.2.1, the applicable acceptance criteria for a loss-of-external-load event are met. (See subsection 15.2.3.)

#### 15.2.2.4 <u>Reference</u>

 Cooper, L., Miselis, V., and Starek, R. M., "Overpressure Protection for Westinghouse Pressurized Water Reactors," <u>WCAP-7769</u>, Revision 1, June, 1972. (Also letter NS-CE-622, C. Eicheldinger (Westinghouse) to D. B. Vassallo (NRC), additional information on <u>WCAP-7769</u>, Revision 1, April 16, 1975.)

#### 15.2.3 TURBINE TRIP

#### 15.2.3.1 Identification of Causes and Accident Description

For a turbine trip event, the reactor trips directly (unless below approximately 40-percent power) from a signal derived from the turbine stop emergency trip fluid pressure and turbine stop valves. The turbine stop valves close rapidly (typically in 0.1 s) on loss of trip fluid pressure actuated by one of several possible turbine trip signals. Turbine trip initiation signals include:

- Generator trip.
- Low condenser vacuum.
- Loss of lubricating oil.
- Turbine thrust bearing failure.
- Turbine overspeed.
- Manual trip.

Upon initiation of stop valve closure, steam flow to the turbine stops abruptly. Sensors on the stop valve detect the turbine trip and initiate turbine bypass through steam dump valves and, if above 40-percent power, a reactor trip. The loss of steam flow results in an almost immediate rise in secondary system temperature and pressure with a resultant increase in primary system temperature and pressure transient than the loss of electrical load event occurs for the turbine trip event due to a more rapid loss of steam flow caused by the more rapid valve closure.

The automatic turbine bypass system would accommodate up to 40 percent of rated steam flow. Reactor coolant temperatures and pressure do not increase significantly if the turbine bypass system and pressurizer pressure control system are functioning properly. If the condenser was not available, the excess steam generation would be relieved to the atmosphere, and main feedwater flow would be lost. For this situation, feedwater flow would be maintained by the auxiliary feedwater system to ensure adequate residual and decay heat removal capability. Should the turbine bypass system fail to operate, the steam generator safety valves may lift to provide pressure control. See paragraph 15.2.2.1 for a further discussion of the transient.

A turbine trip is classified as an American Nuclear Society Condition II event, fault of moderate frequency.

A turbine trip is more limiting than loss of external load, loss of condenser vacuum, and other events which result in a turbine trip. As such, this event has been analyzed in detail. Results and discussion of the analysis are presented in paragraph 15.2.3.2.

#### 15.2.3.2 Analysis of Effects and Consequences

#### 15.2.3.2.1 Method of Analysis

In this analysis, evaluation of the behavior of the units is for a complete loss of steam load from nominal full power, with a turbine trip not causing a direct reactor trip. This demonstrates the adequacy of the pressure-relieving devices and the core protection margins. This assumption delays reactor trip until conditions in the RCS result in a trip due to other signals. Thus, the analysis models a worst-case transient. In addition, no credit is taken for the turbine bypass system. Main feedwater flow terminates at the time of turbine trip with no credit taken for auxiliary feedwater (except for long-term recovery) to mitigate the consequences of the transient.

The analysis of the turbine trip transients employs the detailed digital computer program LOFTRAN (reference 1). The program simulates the neutron kinetics, RCS, pressurizer, pressurizer relief and safety valves, pressurizer spray, steam generator, and steam generator

safety valves. LOFTRAN computes pertinent plant variables, including temperatures, pressures, and power level.

The analysis of this accident uses RTDP methodology. Plant characteristics and initial conditions are shown in tables 15.0.3-2 and 15.0.3-3.

The following summarizes the major assumptions used in the analysis:

A. Initial Operating Conditions

The analysis assumes nominal values of core power, reactor coolant average temperature, and nominal reactor coolant average pressure. The limit-DNBR includes uncertainties in initial conditions as described in chapter 4. Previous studies have shown that the peak pressurizer pressure reached for the turbine trip event is insensitive to the initial conditions of temperature and pressure, and the peak pressurizer pressure is only slightly sensitive to the initial power condition. Therefore, the use of these initial conditions is appropriate for this event.

B. Moderator and Doppler Coefficients of Reactivity

The analysis of the turbine trip is with both maximum and minimum reactivity feedback. With maximum feedback, the analysis assumes a large negative moderator temperature coefficient and the most-negative Doppler-only power coefficient. With minimum feedback, the analysis assumes the most positive moderator temperature coefficient and the least-negative Doppler-only power coefficient.

C. Rod Control

It is conservative to assume that the reactor is in manual rod control with respect to the maximum pressures attained. If the reactor were in automatic rod control, the control rod banks would move before the trip and reduce the severity of the transient.

D. Steam Release

No credit is taken for the operation of the steam dump system or steam generator power-operated relief valves. The steam generator safety valves are operable. The steam generator pressure rises to the safety valve setpoint where steam release through the safety valves limits secondary steam pressure at the setpoint value.

E. Pressurizer Spray and Power-Operated Relief Valves

The following analyses are the two cases for both the minimum and maximum reactivity feedback cases examined:

- 1. The loss of load event is analyzed with full credit taken for the effect of pressurizer spray and power-operated relief valves in reducing or limiting the coolant pressure. Safety valves are also operable. This case minimizes the increase in primary pressure which is conservative for the DNBR transient.
- 2. The loss of load event is analyzed with no credit taken for the effect of pressurizer spray and power-operated relief valves in reducing or limiting the coolant pressure. Safety valves are operable. This case maximizes the pressure increase which is conservative for the RCS overpressurization transient. In this case, the pressurizer safety valves

begin to open at 2550 psia. The effects of the pressurizer safety valve loop seals are accounted for by modeling a 1-percent set pressure shift and assuming no steam flow until the valve loop seals are purged.

F. Feedwater Flow

The analysis assumes main feedwater flow to the steam generators to be lost at the time of turbine trip. No credit is taken for auxiliary feedwater flow since the plant will reach a stabilized condition before auxiliary feedwater initiation is normally assumed to occur; however, the auxiliary feedwater pumps would start on a trip of the main feedwater pumps. The auxiliary feedwater flow would remove core decay heat following plant stabilization.

G. Reactor Trip

Reactor trip actuates by the first reactor protection system trip setpoint reached, with no credit taken for direct reactor trip on turbine trip. Trip signals are expected due to high pressurizer pressure,  $OT\Delta T$ , high pressurizer water level, and low-low steam generator water level.

Plant systems and equipment which may be required to function to mitigate the effects of a turbine trip event are discussed in subsection 15.0.8 and listed in table 15.0.8-1.

The reactor protection system may be required to function following a turbine trip. Pressurizer safety valves and/or steam generator safety valves may be required to open to maintain system pressures below allowable limits. No single active failure will prevent operation of any system required to function. Normal RCS and engineered safety systems are not required to function. However, cases are analyzed both with and without the operation of pressurizer spray and power-operated relief valves to ensure that the worst case is presented.

#### 15.2.3.2.2 Results

The transient responses for a turbine trip from nominal full power operation are shown for the following four cases: two cases with minimum reactivity feedback and two cases with maximum reactivity feedback (figures 15.2.3-1 through 15.2.3-8).

Figures 15.2.3-1 and 15.2.3-2 show the transient responses for the turbine trip event with minimum reactivity feedback, assuming full credit for the pressurizer spray and pressurizer power-operated relief valves. No credit is taken for the turbine bypass through the steam dumps. The reactor trips on overtemperature  $\Delta T$ . The minimum DNBR remains well above the limit value. The pressurizer safety valves and steam generator safety valves prevent overpressurization in the primary and secondary systems, respectively.

Figures 15.2.3-3 and 15.2.3-4 show the responses for the turbine trip event with maximum reactivity feedback. All other plant parameters are the same as the above. The DNBR increases throughout the transient and never drops below its initial value. The pressurizer power-operated relief valves and steam generator safety valves prevent overpressurization in the primary and secondary systems, respectively. The reactor trips on low-low steam generator water level. The pressurizer safety valves do not actuate for this case.

The turbine trip accident was also studied assuming the plant to be initially operating at nominal full power with no credit taken for the pressurizer spray, pressurizer power-operated relief valves, or turbine bypass system. The reactor trips on the high pressurizer pressure signal. Figures 15.2.3-5 and 15.2.3-6 show the transient responses with minimum reactivity feedback. The neutron flux remains constant at nominal full power until the reactor trips. The DNBR never

goes below its initial value throughout the transient. In this case the pressurizer safety valves and steam generator safety valves actuate to maintain the RCS and main steam system pressure below 110 percent of their respective design values.

Figures 15.2.3-7 and 15.2.3-8 show the transients with maximum reactivity feedback, with the other assumptions being the same as in the preceding case. The reactor trips on the high pressurizer pressure signal and the DNBR increases throughout the transient. The pressurizer safety valves and steam generator safety valves actuate to limit primary and secondary system pressures, respectively.

The calculated sequence of events for the turbine trip event is shown in table 15.2.3-1.

Reference 2 presents additional results of analysis for a complete loss of heat sink, including loss of main feedwater. This analysis shows the overpressure protection that is afforded by the pressurizer and steam generator safety valves.

# 15.2.3.3 <u>Conclusions</u>

Results of the analyses, including those in reference 2, show that the plant design is such that a turbine trip without a direct reactor trip presents no hazard to the integrity of the RCS or the main steam system. Pressure-relieving devices incorporated in the two systems are adequate to limit the maximum pressures to within the design limits.

The analyses show that the DNBR will not decrease below the limit value at any time during the transient. Thus, the DNB design basis, as described in section 4.4, is met.

#### 15.2.3.4 References

- 1. Burnett, T. W. T., <u>et al.</u>, "LOFTRAN Code Description," <u>WCAP-7907-P-A</u> (Proprietary), <u>WCAP-7907-A</u> (Non-Proprietary), April 1984.
- Cooper, L., Miselis, V., and Starek, R. M., "Overpressure Protection for Westinghouse Pressurized Water Reactors," <u>WCAP-7769</u>, Revision 1, June 1972. (Also letter NS-CE-622, C. Eicheldinger (Westinghouse) to D. B. Vassallo (NRC), additional information on <u>WCAP-7769</u>, Revision 1, April 16, 1975.)

# 15.2.4 INADVERTENT CLOSURE OF MAIN STEAM ISOLATION VALVES

Inadvertent closure of the main steam isolation valves would result in a turbine trip with no credit taken for the turbine bypass system. Turbine trips are discussed in subsection 15.2.3.

# 15.2.5 LOSS OF CONDENSER VACUUM AND OTHER EVENTS RESULTING IN TURBINE TRIP

Loss of condenser vacuum is one of the events that can cause a turbine trip. Turbine trip-initiating events are described in subsection 15.2.3. A loss of condenser vacuum would preclude the use of steam dump to the condenser; however, since steam dump is assumed to be unavailable in the turbine trip analysis, no additional adverse effects would result if the turbine trip were caused by loss of condenser vacuum. Therefore, the analysis results and conclusions contained in subsection 15.2.3 apply to the loss of the condenser vacuum. In addition, analyses for the other possible causes of a turbine trip, as listed in paragraph 15.2.3.1,

are covered by subsection 15.2.3. Possible overfrequency effects due to a turbine overspeed condition are discussed in paragraph 15.2.2.1 and are not a concern for this type of event.

# 15.2.6 LOSS OF NONEMERGENCY AC POWER TO THE PLANT AUXILIARIES

#### 15.2.6.1 Identification of Causes and Accident Description

A complete loss of nonemergency ac power may result in the loss of all power to the plant auxiliaries; i.e., the reactor coolant pumps, condensate pumps, etc. The loss of power may be caused by a complete loss of the offsite grid accompanied by a turbine-generator trip at the plant or by a loss of the onsite ac distribution system.

This transient is more severe than the turbine trip event analyzed in subsection 15.2.3 because for this case the decrease in heat removal by the secondary system is accompanied by a flow coastdown which further reduces the capacity of the primary coolant to remove heat from the core. The reactor will trip:

- Due to turbine trip.
- Upon reaching one of the trip setpoints in the primary and secondary systems as a result of the flow coastdown and decrease in secondary heat removal.
- Due to the loss of power to the control rod drive mechanisms as a result of the loss of power to the plant.

Following a loss of ac power with turbine and reactor trips, the sequence described below will occur:

- A. Plant vital instruments are supplied from emergency dc power sources.
- B. As the steam system pressure rises following the trip, the steam generator power-operated relief valves may be automatically opened to the atmosphere. The condenser is assumed not to be available for turbine bypass. If the steam flowrate through the power-operated relief valves is not available, the steam generator safety valves may lift to dissipate the sensible heat of the fuel and coolant plus the residual decay heat produced in the reactor.
- C. As the no-load temperature is approached, the steam generator power-operated relief valves (or safety valves, if the power-operated relief valves are not available) are used to dissipate the residual decay heat and to maintain the plant at the hot shutdown condition.
- D. The standby diesel generators, started on loss of voltage on the plant emergency buses, begin to supply plant vital loads.

The auxiliary feedwater system is started automatically, as follows:

- Two motor-driven auxiliary feedwater pumps are started on any of the following:
  - Low-low level in any steam generator.
  - Any safety injection signal.
  - Loss of offsite power.
  - Manual actuation.

- The turbine-driven auxiliary feedwater pump is started on any of the following:
  - Low-low level in any two steam generators.
  - Loss of offsite power.
  - Manual actuation.

The motor-driven auxiliary feedwater pumps are supplied power by the diesels, and the turbinedriven pump utilizes steam from the secondary system. Both types of pumps are designed to supply rated flow within 1 min of the initiating signal, even if a loss of all nonemergency ac power occurs simultaneously with loss of normal feedwater. The auxiliary feedwater turbine exhausts the secondary steam to the atmosphere. The auxiliary feedwater pumps take suction from the two condensate storage tanks for delivery to the steam generators.

Upon the loss of power to the reactor coolant pumps, coolant flow necessary for core cooling and the removal of residual heat is maintained by natural circulation in the reactor coolant loops.

A loss of nonemergency ac power to the plant auxiliaries is classified as an American Nuclear Society Condition II event, fault of moderate frequency. This event is more limiting with respect to RCS heatup than the turbine-trip-initiated decrease in secondary heat removal without loss of ac power, which was analyzed in subsection 15.2.3. However, a loss of ac power to the plant auxiliaries, as postulated above, could also result in a loss of normal feedwater if the condensate pumps lose their power supply.

Following the reactor coolant pump coastdown caused by the loss of ac power, the natural circulation capability of the reactor coolant system (RCS) will remove residual and decay heat from the core, aided by auxiliary feedwater in the secondary system. An analysis is presented here to show that the natural circulation flow in the RCS following a loss of ac power event is sufficient to remove residual heat from the core.

The plant systems and equipment available to mitigate the consequences of a loss of ac power event are discussed in subsection 15.0.8 and listed in table 15.0.8-1.

# 15.2.6.2 <u>Analysis of Effects and Consequences</u>

#### 15.2.6.2.1 Method of Analysis

A detailed analysis using the LOFTRAN code<sup>(1)</sup> is performed to simulate the system transient following a plant blackout. The simulation describes the plant thermal kinetics and RCS, including the natural circulation, pressurizer, steam generators, and feedwater system. The digital program computes pertinent variables, including the steam generator level, pressurizer water level, and reactor coolant average temperature.

The assumptions used in the analysis are as follows:

- A. The plant is initially operating at 100.3 percent of the uprated power, 3653 MWt (includes +0.3% power uncertainty and 17MWt for the reactor coolant pump heat), with initial reactor coolant temperature 6°F above the nominal value and the pressurizer pressure 50 psi above the nominal value.
- B. Core residual heat generation is based on the 1979 version of ANS 5.1<sup>(2)</sup>. ANSI/ANS-5.1-1979 is a conservative representation of the decay energy release rates.

- C. The auxiliary feedwater system is actuated by the low-low steam generator water level signal. The auxiliary feedwater system is assumed to supply a minimum of 510 gal/min from each motor-driven pump with each pump supplying flow to two steam generators. Therefore, a total minimum auxiliary feedwater flow rate of 1020 gal/min is supplied by both motor-driven pumps split equally to the four steam generators.
- D. The auxiliary feedwater system reliability concerns resulting from the Three Mile Island accident are addressed assuming one motor-driven pump is available, supplying a minimum of 510 gal/min split equally to two steam generators. Better-estimate conditions are modeled for this scenario, where plant operating parameters are set to their nominal values, with no consideration of uncertainties.
- E. Secondary system steam relief is achieved through the steam generator safety valves.
- F. The pressurizer relief valves, safety valves, sprays, and heaters are assumed to function to maximize the peak pressurizer water volume.
- G. The high-head charging pumps, which are initiated on a loss of offsite power signal, are not assumed to function for this event as their operation is a benefit with respect to long term core decay heat removal. Note, however, that charging pump actuation following a loss of offsite power will increase the reactor coolant inventory if the letdown isolation valve fails closed due to a subsequent loss of instrument air. This scenario was examined to determine if the operators have sufficient time to unblock the pressurizer power-operated relief valves to preclude water relief through the pressurizer safety valves.

The assumptions used in the analysis are similar to the loss of normal feedwater flow incident (subsection 15.2.7) except that power is assumed to be lost to the reactor coolant pumps at the time of reactor trip.

Plant characteristics and initial conditions are further discussed in subsection 15.0.3.

# 15.2.6.2.2 Results

The transient response of the RCS following a loss of ac power is shown in figures 15.2.6-1 and 15.2.6-2. The calculated sequence of events for this event is listed in table 15.2.3-1.

The LOFTRAN code results show that the natural circulation flow and the auxiliary feedwater flow rates are sufficient to provide adequate core decay heat removal following reactor trip and reactor coolant pump coastdown.

A separate case was run to address the auxiliary feedwater system reliability concerns resulting from the Three Mile Island accident. This case did not result in the filling of the pressurizer, which confirms the reliability of the auxiliary feedwater system.

A separate case was run with charging pumps initiated on a loss of offsite power signal. This case did not result in the filling of the pressurizer prior to 10 minutes following the initiation of the event. Thus, there is sufficient time available for the operator to unblock the pressurizer power-operated relief valves and thus preclude water relief through the pressurizer safety valves. This case is analyzed similar to the inadvertent emergency core cooling system (ECCS) at power event, discussed in subsection 15.6.1, where operator action is required to unblock the pressurizer safety valves.

#### 15.2.6.3 Radiological Consequences

The evaluation of the radiological consequences of a postulated loss of nonemergency ac power assumes that the reactor has been operating with a small percent of defective fuel and leaking steam generator tubes for sufficient time to establish equilibrium concentrations of radionuclides in the reactor coolant and in the secondary coolant.

Following the loss of nonemergency ac power, radionuclides are carried by the primary coolant to the steam generator via leaking tubes and are released to the environment via the steam line safety or power-operated relief valves.

# 15.2.6.3.1 Analytical Assumptions

The major assumptions and parameters used in the analysis are itemized in table 15.2.6-1. The following is a more detailed discussion of the source term.

15.2.6.3.1.1 <u>Source Term Calculation</u>. The concentration of nuclides in the primary and secondary system, prior to the accident, are determined as follows:

- A. The iodine concentrations in the reactor coolant will be based upon preaccident and accident initiated iodine spikes.
  - 1. Accident Initiated Spike

The reactor trip associated with the loss of nonemergency ac power creates an iodine spike in the primary system which increases the iodine release rate from the fuel to the primary coolant to a value 500 times greater than the release rate corresponding to the maximum equilibrium primary system iodine concentration of 1  $\mu$ Ci/g of dose equivalent (DE) I-131. The elevated appearance rates are assumed to return to normal once the reactor coolant iodine level increases to 100  $\mu$ Ci/g DE I-131.

2. Preaccident Spike

A reactor transient has occurred prior to the loss of nonemergency ac power and has raised the primary coolant iodine concentration to 60  $\mu$ Ci/g of DE I-131.

- B. The noble gas concentrations in the primary coolant are based on 1-percent defective fuel.
- C. The secondary coolant activity is based on DE of 0.1  $\mu$ Ci/g of I-131.

15.2.6.3.1.2 <u>Mathematical Models Used in the Analysis</u>. Mathematical models used in the analysis are described in the following sections:

- A. The mathematical models used to analyze the activity released during the course of the accident are described in appendix 15A.
- B. The atmospheric dispersion factors used in the analysis were calculated based on the onsite meteorological measurement programs described in subsection 2.3.3.

C. The thyroid inhalation and total-body gamma immersion doses to a receptor at the exclusion area boundary and outer boundary of the low population zone were analyzed using the models described in appendix 15A.

15.2.6.3.1.3 <u>Identification of Leakage Pathways and Resultant Leakage Activity</u>. Radionuclides carried from the primary coolant to the steam generators via leaking tubes are released to the environment via the steam line safety or power-operated relief valves. Iodines are assumed to mix with the secondary coolant and partition between the generator liquid and steam before release to the environment. Noble gases are assumed to be directly released.

All activity is released to the environment with no consideration given to radioactive decay or cloud depletion by ground deposition during transport to the exclusion area boundary and low population zone. Hence, the resultant radiological consequences represent the most conservative estimate of the potential integrated dose due to the postulated loss of nonemergency ac power.

#### 15.2.6.3.2 Identification of Uncertainties and Conservative Elements in the Analysis

- A. The initial reactor coolant iodine activity is based on the technical specification limit of 1.0  $\mu$ Ci/g of DE I-131 which is significantly greater than the activities associated with normal operating conditions.
- B. The preaccident iodine spike activity based on 60  $\mu$ Ci/g and the accident initiated spike iodine appearance rate multiplier of 500 are much greater than expected for typical plant operation.
- C. The noble gas activities are based on 1 percent defective fuel which cannot exist simultaneously with 1.0- $\mu$ Ci/g I-131. For iodines, 1 percent defects would be approximately three times the technical specification limit.
- D. A 1-gal/min steam generator primary-to-secondary leakage is assumed, which is significantly greater than that anticipated during normal operation.
- E. The meteorological conditions which may be present at the site during the course of the accident are uncertain. However, it is highly unlikely that the assumed meteorological conditions would be present during the course of the accident for any extended period of time. Therefore, the radiological consequences evaluated, based on the meteorological conditions assumed, are conservative.

# 15.2.6.3.3 Conclusions

15.2.6.3.3.1 <u>Filter Loadings</u>. No filter serves to limit the release of radioactivity in this accident. There is no significant activity buildup on any filters as a consequence of a loss of nonemergency ac power.

15.2.6.3.3.2Doses to Receptor at Exclusion Area Boundary and Low Population ZoneOuter Boundary.The potential radiological consequences resulting from the occurrence of a

postulated loss of nonemergency ac power have been conservatively analyzed using the assumptions and models described. The total-body gamma dose due to immersion from direct radiation and the thyroid dose due to inhalation have been analyzed for the 0- to 2-h dose at the exclusion area boundary and for the duration of the accident (0 to 20 h) at the low population zone outer boundary. The results are presented in table 15.2.6-2. The resultant doses are well within the guidelines of 10 CFR 100.

# 15.2.6.4 <u>Conclusions</u>

Results of the analysis show that for the loss of nonemergency ac power to plant auxiliaries event all safety criteria are met. Auxiliary feedwater capacity is sufficient to prevent water relief through the pressurizer relief and safety valves. The analysis demonstrates that sufficient long-term RCS heat removal capability exists via natural circulation and auxiliary feedwater following reactor coolant pump coastdown to prevent fuel or clad damage and assures that the RCS is not overpressurized.

#### 15.2.6.5 <u>References</u>

- 1. Burnett, T. W. T., <u>et al</u>., "LOFTRAN Code Description," <u>WCAP-7907-P-A</u> (proprietary), <u>WCAP-7907-A</u> (nonproprietary), April 1984.
- 2. "American National Standard for Decay Heat Power in Light Water Reactors," <u>ANSI/ANS-5.1-1979</u>, August 1979.

# 15.2.7 LOSS OF NORMAL FEEDWATER FLOW

#### 15.2.7.1 Identification of Causes and Accident Description

A loss of normal feedwater (from pump failures, valve malfunctions, or loss of offsite ac power) results in a reduction in the capability of the secondary system to remove the heat generated in the reactor core. If an alternative supply of feedwater were not supplied to the plant, core residual heat following reactor trip would heat the primary system water to the point where water relief from the pressurizer would occur, resulting in a substantial loss of water from the reactor coolant system (RCS). Since the plant is tripped well before the steam generator heat transfer capability is reduced, the primary system variables never approach a departure from nucleate boiling condition.

A small secondary system break could affect normal feedwater flow control causing low steam generator levels prior to protective actions for the break. This scenario is addressed by the assumptions made for the feedwater system pipe break (subsection 15.2.8).

The following occur upon loss of normal feedwater (assuming main feedwater pump failures or valve malfunctions):

A. As the steam system pressure rises following the trip, the steam generator power-operated relief valves are automatically opened to the atmosphere. The condenser is assumed to be unavailable for turbine bypass. If the steam flow path through the power-operated relief valves is not available, the steam generator safety valves may lift to dissipate the sensible heat of the fuel and coolant plus the residual decay heat produced in the reactor.

B. As the no-load temperature is approached, the steam generator power-operated relief valves (or safety valves, if the power-operated relief valves are not available) are used to dissipate the residual decay heat and to maintain the plant at the hot shutdown condition.

A loss of normal feedwater is classified as an American Nuclear Society Condition II event, fault of moderate frequency.

The reactor trip on low-low water level in any steam generator provides the necessary protection against a loss of normal feedwater.

The auxiliary feedwater system is started automatically, as discussed in paragraph 15.2.6.1. The turbine-driven auxiliary feedwater pump utilizes steam from the secondary system and exhausts it to the atmosphere. The motor-driven auxiliary feedwater pumps are supplied power by the diesel generators. The auxiliary feedwater pumps take suction directly from the condensate storage tank for delivery to the steam generators.

An analysis of the system transient is presented below to show that following a loss of normal feedwater the auxiliary feedwater system is capable of removing the stored and residual decay heat, thus preventing either overpressurization of the RCS or loss of water from the reactor coolant system, and returning the plant to a safe condition.

# 15.2.7.2 Analysis of Effects and Consequences

# 15.2.7.2.1 Method of Analysis

A detailed analysis using the LOFTRAN code<sup>(1)</sup> is performed to obtain the plant transient following a loss of normal feedwater. The simulation describes the plant thermal kinetics, RCS (including the natural circulation), pressurizer, steam generators, and feedwater system. The digital program computes pertinent variables, including the steam generator level, pressurizer water level, and reactor coolant average temperature.

The assumptions used in the analysis are as follows:

- A. The plant is initially operating at 100.3 percent of the uprated full power, design rating, 3653 MWt (includes +0.3% power uncertainty and 17 MWt for the reactor coolant pump heat).
- B. Core residual heat generation is based on the 1979 version of ANS 5.1<sup>(2)</sup>. <u>ANSI/ANS-5.1-1979</u> is a conservative representation of the decay energy release rates.
- C. Reactor trip occurs on steam generator low-low level.
- D. The worst single failure in the auxiliary feedwater system occurs. The auxiliary feedwater system is assumed to supply a total of 510 gal/min from each motordriven pump with each pump supplying flow to two steam generators. Therefore, a total minimum auxiliary feedwater flow rate of 1020 gal/min is supplied by both motor-driven pumps split equally to the four steam generators.
- E. The auxiliary feedwater system reliability concerns resulting from the Three Mile Island accident are addressed assuming one motor-driven pump is available,

supplying a minimum of 510 gal/min split equally to two steam generators. Better-estimate conditions are modeled for this scenario, where plant operating parameters are set to their nominal values, with no consideration of uncertainties.

- F. The auxiliary feedwater system is actuated by the low-low steam generator water level signal.
- G. Secondary system steam relief is achieved through the steam generator safety valves.
- H. The initial reactor coolant average temperature is 6°F higher than the nominal value, and initial pressurizer pressure is 50 psi higher than nominal.

The loss of normal feedwater analysis is performed to demonstrate the adequacy of the reactor protection and engineered safeguards systems (e.g., the auxiliary feedwater system) in removing long-term decay heat and preventing excessive heatup of the RCS with possible resultant RCS overpressurization or loss of RCS water.

As such, the assumptions used in this analysis minimize the energy removal capability of the system and maximize the possibility of water relief from the coolant system by maximizing the coolant system expansion.

For the loss of normal feedwater transient, the reactor coolant volumetric flow remains at its normal value, and the reactor trips via the low-low steam generator level trip. The reactor coolant pumps may be manually tripped at some later time to reduce heat addition to the RCS. However, the pumps are not assumed to trip in the analysis.

An additional assumption made for the loss of normal feedwater evaluation is that the pressurizer relief valves, safety valves, sprays, and heaters are assumed to function to maximize the peak pressurizer water volume. Operation of the valves, if required, maintains peak RCS pressure below 110 percent of RCS design pressure throughout the transient.

Plant characteristics and initial conditions are further discussed in subsection 15.0.3.

Plant systems and equipment which are necessary to mitigate the effects of a loss of normal feedwater accident are discussed in subsection 15.0.8 and listed in table 15.0.8-1. Normal reactor control systems are not required to function. The reactor protection system is required to function following a loss of normal feedwater, as analyzed here. The auxiliary feedwater system is required to deliver a minimum auxiliary feedwater flow rate. No single active failure will prevent operation of any system to perform its required function. A discussion of anticipated transients without scram considerations is presented in section 15.8.

# 15.2.7.2.2 Results

Figures 15.2.7-1 and 15.2.7-2 show the significant plant parameters following a loss of normal feedwater.

Following the reactor and turbine trip from full load, the water level in the steam generators will fall due to the reduction of steam generator void fraction, and because steam flow through the safety valves continues to dissipate the stored and generated heat. One minute following the initiation of the low-low level trip, both motor-driven auxiliary feedwater pumps are automatically started, reducing the rate of water inventory decrease in the four steam generators.

The capacity of the auxiliary feedwater pumps is such that the water level in the steam generators is sufficient to dissipate core residual heat without water relief from the RCS safety valves.

The calculated sequence of events for this accident is listed in table 15.2.3-1. As shown in figures 15.2.7-1 and 15.2.7-2, the plant slowly approaches a stabilized condition following reactor trip and auxiliary feedwater initiation. Plant procedures may be followed to further cool down the plant.

A separate case was run to address the auxiliary feedwater system reliability concerns resulting form the Three Mile Island accident. This case did not result in the filling of the pressurizer, which confirms the reliability of the auxiliary feedwater system.

# 15.2.7.3 <u>Conclusions</u>

Results of the analysis show that a loss of normal feedwater does not adversely affect the core, the RCS, or the steam system. The auxiliary feedwater capacity is such that reactor coolant water is not relieved from the pressurizer relief or safety valves.

#### 15.2.7.4 <u>References</u>

- 1. Burnett, T. W. T., <u>et al.</u>, "LOFTRAN Code Description," <u>WCAP-7907-P-A</u> (proprietary), <u>WCAP-7907-A</u> (nonproprietary), April 1984.
- 2. "American National Standard for Decay Heat Power in Light Water Reactors," ANSI/ANS-5.1-1979, August 1979.

# 15.2.8 FEEDWATER SYSTEM PIPE BREAK

#### 15.2.8.1 Identification of Causes and Accident Description

A major feedwater line rupture is defined as a break in a feedwater line large enough to prevent the addition of sufficient feedwater to the steam generators to maintain shell-side fluid inventory in the steam generators. If the break is postulated in a feedwater line between the check valve and the steam generator, fluid from the steam generator may also be discharged through the break. A break in this location could preclude the subsequent addition of auxiliary feedwater to the affected steam generator. Furthermore, due to the piping system design the feed break piping may cause a safety injection branch line to break which decreases the safety injection flow delivered to the core. (A break upstream of the feedwater line check valve would affect the plant only as a loss of feedwater. This case is covered by the evaluation in subsections 15.2.6 and 15.2.7.)

Depending upon the size of the break and the plant operating conditions at the time of the break, the break could cause either a reactor coolant system (RCS) cooldown (by excessive energy discharge through the break) or an RCS heatup. Potential RCS cooldown resulting from a secondary pipe rupture is evaluated in subsection 15.1.5. Therefore, only the RCS heatup effects are evaluated for a feedwater line rupture.

A feedwater line rupture reduces the ability to remove heat generated by the core from the RCS for the following reasons:

A. Feedwater flow to the steam generators is reduced. Since feedwater is subcooled, its loss may cause reactor coolant temperatures to increase prior to reactor trip.

- B. Fluid in the steam generator may be discharged through the break and would then not be available for decay heat removal after trip.
- C. The break may be large enough to prevent the addition of any main feedwater after trip.

An auxiliary feedwater system functions to ensure the availability of adequate feedwater so that:

- A. No substantial overpressurization of the RCS occurs (less than 110 percent of design pressures).
- B. Sufficient liquid in the RCS is maintained so that the core remains in place and geometrically intact with no loss of core cooling capability.

A major feedwater line rupture is classified as an American Nuclear Society Condition IV event.

The severity of the feedwater line rupture transient depends on a number of system parameters, including break size, initial reactor power, and the functioning of various control and safety systems. Sensitivity studies presented in reference 1 illustrate that the most limiting feedwater line rupture is a double-ended rupture of the largest feedwater line. The main feedwater control system is assumed to malfunction due to an adverse environment. The water levels in all steam generators are assumed to decrease equally until the low-low steam generator level reactor trip setpoint is reached. After reactor trip, a double-ended rupture of the largest feedwater line rupture that can occur. Analyses have been performed at full power, with and without loss of offsite power. For the case without offsite power available, the power is assumed to be lost at the time of reactor trip. This is more conservative than the case where power is lost at the initiation of the event. These cases are analyzed below.

The following provides the protection for a main feedwater line rupture:

- A. A reactor trip on any of the following conditions:
  - 1. High pressurizer pressure.
  - 2. Overtemperature  $\Delta T$ .
  - 3. Low-low steam generator water level in any steam generator.
  - 4. Low pressurizer pressure.
  - 5. High pressurizer level.
  - 6. Safety injection signals from either of the following:
    - Two out of three low steam line pressure in any steam generator.
    - Two out of three high containment pressure (high-1).

Refer to chapter 7 for a description of the actuation system.

B. The auxiliary feedwater system provides an assured source of feedwater to the steam generators for decay heat removal. Refer to subsection 10.4.9 for a description of the auxiliary feedwater system.

# 15.2.8.2 Analysis of Effects and Consequences

#### 15.2.8.2.1 Method of Analysis

A detailed analysis using the LOFTRAN code (reference 2) is performed in order to determine the plant transient following a feedwater line rupture. The code describes the plant thermal kinetics, RCS (including natural circulation), pressurizer, steam generators, and feedwater system and computes pertinent variables, including the pressurizer pressure, pressurizer water level, and reactor coolant average temperature.

The cases analyzed assume a double-ended rupture of the largest feedwater pipe at full power. Major assumptions used in the analysis are as follows:

- A. The plant is initially operating at 102 percent of the engineered safety features design rating.
- B. Initial reactor coolant average temperature is 6.0°F above the nominal value, and the initial pressurizer pressure is 50 psi below its nominal value.
- C. The pressurizer power-operated relief valves and the safety relief valves are assumed to be operable. No credit is taken for pressurizer spray.
- D. Initial pressurizer level is at the nominal programmed value plus 10 percent error; a conservative initial steam generator water level is assumed in all steam generators.
- E. No credit is taken for the high pressurizer pressure, low pressurizer pressure, high pressurizer level, and overtemperature  $\Delta T$  reactor trips.
- F. Main feedwater to all steam generators is assumed to stop at the time the break occurs. (All main feedwater spills out through the break.) As a result, the water level in all four steam generators begins to drop.
- G. A double-ended break area of 0.89 ft<sup>2</sup> is assumed at the time of reactor trip. This maximizes the blowdown discharge rate following the time of trip, which maximizes the resultant heatup of the reactor coolant.
- H. A conservative feedwater line break discharge quality is assumed at the time the reactor trip occurs. This minimizes the heat removal capability of the affected steam generator.
- I. Reactor trip is assumed to be initiated when the low-low level setpoint is reached on the ruptured steam generator.
- J. The auxiliary feedwater system is actuated by the low-low steam generator water level signal. The auxiliary feedwater system is assumed to supply a minimum of 510 gal/min to three unaffected steam generators, including allowance for possible spillage through the main feedwater line break. A 60-s delay was assumed following the low-low level signal to allow time for startup of the standby diesel generators and the auxiliary feedwater pumps. An additional 223 s were assumed before the feedwater lines were purged and the relatively cold auxiliary feedwater entered the unaffected steam generators.
- K. No credit is taken for heat energy deposited in RCS metal during the RCS heatup.
- L. No credit is taken for charging or letdown.

- M. Steam generator heat transfer area is assumed to decrease as the shell-side liquid inventory decreases.
- N. Conservative core residual heat generation is assumed based upon long-term operation at the initial power level preceding the trip (reference 3).
- O. No credit is taken for the following potential protection logic signals to mitigate the consequences of the accident:
  - 1. High pressurizer pressure.
  - 2. Overtemperature  $\Delta T$ .
  - 3. High pressurizer level.
  - 4. High containment pressure.

Receipt of a low-low steam generator water level signal in at least one steam generator starts the motor-driven auxiliary feedwater pumps, which in turn initiate auxiliary feedwater flow to the steam generators. The turbine-driven auxiliary feedwater pump is initiated if the low-low steam generator water level signal is reached in at least two steam generators. Similarly, receipt of a low steam line pressure signal in at least one steam line initiates a steam line isolation signal which closes all main steam line isolation valves. This signal also gives a safety injection (SI) signal which initiates flow of cold borated water into the RCS. The amount of SI flow is a function of RCS pressure.

Emergency operating procedures following a feedwater system pipe rupture require the following actions to be taken by the reactor operator:

- A. Isolate main and auxiliary feedwater flow spilling from the ruptured feedwater line.
- B. Stop all but one high-head SI charging pump if the SI termination criteria are met.

Isolating auxiliary feedwater flow spilling through the break allows additional auxiliary feedwater flow to be diverted to the intact steam generators; however, no credit for this operator action is explicitly modeled in the analysis.

Subsequent to recovery of level in the intact steam generators, the high-head SI pumps will be turned off and plant operating procedures will be followed in cooling the plant to hot shutdown conditions.

Plant characteristics and initial conditions are further discussed in subsection 15.0.3.

No reactor control systems are assumed to function other than a conservative assumption that the pressurizer PORVs are operable. The reactor protection system is required to function following a feedwater line rupture as analyzed here. No single active failure will prevent operation of this system.

The engineered safety systems assumed to function are the auxiliary feedwater system and the SI system. For the auxiliary feedwater system, the worst case configuration has been used; i.e., only three intact steam generators receive auxiliary feedwater following the break. Most of the flow from the motor-driven auxiliary feedwater pump feeding the affected steam generator was assumed to spill through the break. The turbine-driven auxiliary feedwater pump has been assumed to fail. The total flow from the two motor-driven auxiliary feedwater pumps is assumed to be a minimum of 510 gal/min divided among the three intact steam generators.

For the case without offsite power, there will be a flow coastdown until flow in the loops reaches the natural circulation value. The natural circulation capability of the RCS has been shown (in

subsection 15.2.6) to be sufficient to remove core decay heat following reactor trip for the loss of ac power transient. Pump coastdown characteristics are demonstrated in subsections 15.3.1 and 15.3.2 for single and multiple reactor coolant pump trips, respectively.

A detailed description and analysis of the SI system is provided in section 6.3. The auxiliary feedwater system is described in subsection 10.4.9.

# 15.2.8.2.2 Results

Calculated plant parameters following a major feedwater line rupture are shown in figures 15.2.8-1 through 15.2.8-8. Results for the case with offsite power available are presented in figures 15.2.8-1 through 15.2.8-4. Results for the case where offsite power is lost are presented in figures 15.2.8-5 through 15.2.8-8. The calculated sequence of events for both cases analyzed is listed in table 15.2.3-1.

The system response following the feedwater line rupture is similar for both cases analyzed. Results presented in figures 15.2.8-2 and 15.2.8-4 (with offsite power available) and figures 15.2.8-6 and 15.2.8-8 (without offsite power) show that pressures in the RCS and main steam system remain below 110 percent of the respective design pressures. Pressurizer pressure decreases after reactor trip of low-low steam generator level (66 s). Pressurizer pressure decreases due to the loss of heat input, until the SI system is actuated on low steam line pressure in the ruptured loop. Coolant expansion occurs due to reduced heat transfer capability in the steam generators; the pressurizer PORVs are conservatively assumed to open, which prevents pressurizer pressure from increasing beyond the PORV setpoint. The addition of the SI flow aids in cooling down the primary and helps to ensure that sufficient fluid exists to keep the core covered with water.

Figure 15.2.8-2 shows that the minimum volume in the pressurizer is reached at approximately 150 s; however, the results of LOFTRAN show that the core remains covered at all times and that no boiling occurs in the reactor coolant loops.

The major difference between the two cases analyzed can be seen in the plots of hot and cold leg temperatures, figure 15.2.8-3 (with offsite power available) and figure 15.2.8-7 (without offsite power). It is apparent that for the initial transient (150 s) the case without offsite power results in higher temperatures in the hot leg. For longer times, however, the case with offsite power results in a more severe rise in temperature. The pressurizer fills more rapidly for the case with power due to the increased coolant expansion resulting from the pump heat addition; however, water is not relieved for the case without offsite power. As previously stated, the core cooling capability is maintained.

# 15.2.8.3 <u>Conclusions</u>

Results of the analyses show that for the postulated feedwater line rupture, the assumed auxiliary feedwater system capacity is adequate to remove decay heat, to prevent overpressurizing the RCS, and to maintain the core cooling capability. Radioactivity doses from the postulated feedwater lines rupture are less than those previously presented for the postulated steam line break. All applicable acceptance criteria are therefore met.

#### 15.2.8.4 <u>References</u>

- 1. Lang, G. E., and Cunningham, J. P., "Report on the Consequences of a Postulated Main Feedline Rupture," <u>WCAP-9230</u> (proprietary) and <u>WCAP-9231</u> (nonproprietary), January 1978.
- 2. Burnett, T. W. T., <u>et al</u>., "LOFTRAN Code Description," <u>WCAP 7907-P-A</u> (proprietary), <u>WCAP-7907-A</u> (nonproprietary), April 1984.
- 3. "American National Standard for Decay Heat Power in Light Water Reactor," <u>ANSI/ANS-</u> <u>5.1-1979</u>, August 1979.

# TABLE 15.2.3-1 (SHEET 1 OF 5)

#### TIME SEQUENCE OF EVENTS FOR INCIDENTS WHICH RESULT IN A DECREASE IN HEAT REMOVAL BY THE SECONDARY SYSTEM

I.

Ac	<u>cident</u>	Event	<u>Time (s)</u>
Tu	rbine Trip		
A.	With pressurizer control (minimum reactivity feedback)	Turbine trip; loss of main feedwater flow	0.0
		Initiation of steam release from steam generator safety valves	7.1
		Peak pressurizer pressure occurs	8.4
		Overtemperature $\Delta T$ reactor trip point reached	8.9
		Rods begin to drop	10.9
		Minimum DNBR occurs	12.1
B.	With pressurizer control (maximum reactivity feedback)	Turbine trip; loss of main feedwater flow	0.0
		Initiation of steam release from steam generator safety valves	7.1
		Peak pressurizer pressure occurs	7.1
		Low-low steam generator water level reactor trip setpoint reached	40.9
		Rods begin to drop	42.9
		Minimum DNBR occurs	(a)

# TABLE 15.2.3-1 (SHEET 2 OF 5)

	<u>Accident</u>		Event	<u>Time (s)</u>
I.	I. Turbine Trip			
	C.	Without pressurizer control (minimum reactivity feedback)	Turbine trip; loss of main feedwater flow	0.0
			High pressurizer pressure reactor trip point reached	4.5
			Rods begin to drop	6.5
			Initiation of steam release from steam generator safety valves	7.0
			Peak pressurizer pressure occurs	8.6
			Minimum DNBR occurs	(a)
	D.	Without pressurizer control (maximum reactivity feedback)	Turbine trip; loss of main feedwater flow	0.0
			High pressurizer pressure reactor trip point reached	4.5
			Rods begin to drop	6.5
			Initiation of steam release from steam generator safety valves	7.0
			Peak pressurizer pressure occurs	7.4
II. Loss of nonem power to the s auxiliaries	s of nonemergency ac /er to the station iliaries	Minimum DNBR occurs Main feedwater flow stops	(a) 10.0	
			Low-low steam generator water level trip setpoint reached	56.5

# TABLE 15.2.3-1 (SHEET 3 OF 5)

	<u>Accident</u>	Event	<u> Time (s)</u>
II.	Loss of nonemergency ac power to the station auxiliaries (continued)	Rods begin to drop	58.5
		Reactor coolant pumps begin coasting down	60.5
		Peak pressurizer volume occurs	62.4
		Auxiliary feedwater initiated	131.5
		Core decay heat decreases to the auxiliary feedwater heat removal capacity	≃450.0
111.	Loss of normal feedwater flow	Main feedwater flow stops	10.0
		Low-low steam generator water level reactor trip setpoint reached	56.5
		Rods begin to drop	58.5
		Peak pressurizer volume occurs	62.0
		Auxiliary feedwater initiated	116.5
		Core decay heat plus pump decreases to auxiliary feedwater heat removal capacity	≃400

# TABLE 15.2.3-1 (SHEET 4 OF 5)

	<u>Accident</u>		<u>Event</u>	<u> Time (s)</u>	
IV.	Feedwater system pipe break				
	A.	With offsite power available	Feedwater control system fails	10.0	
			Pressurizer relief valve setpoint reached	25.5	
			Low-low steam generator level reactor trip setpoint reached in all steam generator	66.5	
			Rods begin to drop and feedline rupture occurs	68.5	
			Steam generator safety valve setpoint reached in intact steam generators	70.0	
			Low steam line pressure setpoint reached in affected steam generator	110.0	
			All main steam line isolation valves close	120.0	
			Auxiliary feedwater is delivered to intact steam generators	126.5	
			Core decay heat decreases to auxiliary feedwater heat removal capacity	<b>≃2500</b>	
	В.	Without offsite power	Feedwater control system	10.0	
			Pressurizer relief valve setpoint reached	25.5	

# TABLE 15.2.3-1 (SHEET 5 OF 5)

<u>Event</u>	<u>Time (s)</u>
Low-low steam generator level reactor trip setpoint reached in affected steam generator	66.5
Rods begin to drop; power lost to the reactor coolant pumps, feedline rupture occurs	68.5
Steam generator safety valve setpoint reached in intact steam generators	70.0
Low steam line pressure setpoint reached in affected steam generator	101.0
All main steam line isolation valves close	111.0
Auxiliary feedwater is delivered to intact steam generators	126.5
Core decay heat decreases to auxiliary feedwater heat removal capacity	≃ <b>1000</b>
	Event Low-low steam generator level reactor trip setpoint reached in affected steam generator Rods begin to drop; power lost to the reactor coolant pumps, feedline rupture occurs Steam generator safety valve setpoint reached in intact steam generators Low steam line pressure setpoint reached in affected steam generator All main steam line isolation valves close Auxiliary feedwater is delivered to intact steam generators Core decay heat decreases to auxiliary feedwater heat removal capacity

a. DNBR does not decrease below its initial value.

# TABLE 15.2.6-1 (SHEET 1 OF 2)

#### PARAMETERS USED IN EVALUATING RADIOLOGICAL CONSEQUENCES OF LOSS OF NONEMERGENCY ac POWER<sup>(a)</sup>

#### I. Source Data

Α.	Core power level (MWt)	3636
	Total steam generator tube leakage (gal/min)	1
	Reactor coolant iodine activity	
	1. Accident initiated spike	Initial activity equal to the DE of $1.0 \ \mu$ Ci/g of I-131 with an iodine spike that increases the rate of iodine release into the reactor coolant by a factor of 500. See table 15A-7.
	2. Preaccident spike	An assumed preaccident iodine spike which has resulted in the DE of 60 $\mu$ Ci/g of I-131 in the reactor coolant. See table 15A-6.
D.	Gap activity released to reactor coolant from failed fuel	None
E.	Reactor coolant noble gas activity	Based on 1 percent defective fuel. See table 15A-4.
F.	Secondary system initial activity	DE of 0.1 μCi/g of I-131.
G.	Secondary coolant mass, four generators (g)	1.9 x 108
H.	Reactor coolant mass (g)	2.3 x 10 <sup>8</sup>
	(0)	
I.	Offsite power	Lost after trip
I. J.	Offsite power Primary-to-secondary leakage duration (h)	Lost after trip 20

# TABLE 15.2.6-1 (SHEET 2 OF 2)

II.	Atmo	ospheric Dispersion Factors	See table 15A-2.
III.	Activity Release Data		
	Α.	Primary-to-secondary leak rate(gal/min) <sup>(b)</sup>	1.0
	B.	Steam released (lb) 0 to 2 h 2 to 8 h 8 to 20 h <sup>(c)</sup>	555,000 1,365,000 2,730,000
	C.	lodine partition factor	0.01

c. The evaluation has included the impact of a longer time required to cool the plant resulting from the deletion of the RHR suction valve thermal relief (as specified by PS-06-1981). The 20-hour assumption is consistent with the evaluation provided by Westinghouse Electric Corporation when the RHR change was first evaluated.

a. Power is lost at 10 s.

b. Based on water at 62.4 lb/ft<sup>3</sup>.

# TABLE 15.2.6-2

#### RADIOLOGICAL CONSEQUENCES OF LOSS OF NONEMERGENCY ac POWER

	Doses (rem)
Case 1 - Accident Initiated Iodine Spike Exclusion area boundary (0 to 2 h)	
Thyroid	< 0.1
Whole Body	< 0.2
Low population zone outer boundary (0 to 20 h)	
Thyroid	< 0.1
Whole Body	0.1
Case 2 - Preaccident lodine Spike Exclusion area boundary (0 to 2 h)	
Thyroid	< 0.1
Whole Body	< 0.1
Low population zone outer boundary (0 to 20 h)	
Thyroid	0.1
Whole Body	< 0.1
Both Cases - Whole Body Gamma	
Exclusive area bounding (0 to 2 h)	< 0.1
Low population zone outer boundary (8 h)	< 0.1








































#### 15.3 DECREASE IN REACTOR COOLANT SYSTEM FLOWRATE

A number of faults which could result in a decrease in the reactor coolant system flowrate are postulated. These events are discussed in this section. Detailed analyses are presented for the most limiting of the following flow decrease events:

- A. Partial loss of forced reactor coolant flow.
- B. Complete loss of forced reactor coolant flow.
- C. Reactor coolant pump shaft seizure (locked rotor).
- D. Reactor coolant pump shaft break.

All of the accidents in this section have been analyzed. It has been determined that the most severe radiological consequences will result from the reactor coolant pump shaft seizure accident discussed in subsection 15.3.3. Therefore, doses are reported only for that limiting case.

# 15.3.1 PARTIAL LOSS OF FORCED REACTOR COOLANT FLOW

#### 15.3.1.1 Identification of Causes and Accident Description

A partial loss-of-forced-reactor-coolant flow accident can result from a mechanical or electrical failure in an RCP or from a fault in the power supply to the pump or pumps supplied by an RCP bus. If the reactor is at power at the time of the accident, the immediate effect of the loss-of-forced-reactor-coolant flow is a rapid increase in the coolant temperature. This increase could result in DNB with subsequent fuel damage if the reactor does not trip promptly.

Two buses connected to the generators supply power to the pumps. When a generator trip occurs, the buses are automatically transferred to a transformer supplied from external power lines, and the pumps continue to operate. Following any turbine trip where there are no electrical faults which require tripping the generator from the network, the generator remains connected to the network for approximately 30 seconds. The RCPs remain connected to the generator, thus ensuring full flow for approximately 30 seconds after the reactor trip before any transfer is made.

The low primary coolant flow reactor trip signal, which actuates in any reactor coolant loop by two out of three low-flow signals, provides the necessary protection against this event. Above permissive P-8, low flow in any loop will actuate a reactor trip. Between approximately 10-percent power (permissive P-7) and the power level corresponding to permissive P-8, low flow in any two loops will actuate a reactor trip. Above permissive P-7, two or more RCP circuit breakers from the same bus will open which will actuate the corresponding undervoltage relays. This results in a reactor trip which serves as backup to the flow trip.

This is an ANS Condition II incident.

# 15.3.1.2 Analysis of Effects and Consequences

#### 15.3.1.2.1 Method of Analysis

This analysis examines partial loss-of-forced-reactor-coolant flow involving loss of two pumps with four loops in operation.

This analysis uses three digital computer codes. First the LOFTRAN code (reference 1) calculates the loop and core flow during the transient, the time of reactor trip based on the calculated flows, the nuclear power transient, and the primary system pressure and temperature transients. The FACTRAN code (reference 2) then calculates the heat flux transient based on the nuclear power and flow from LOFTRAN. Finally, the VIPRE-01 code (section 4.4) calculates the DNBR during the transient based on the heat flux from FACTRAN and flow from LOFTRAN. The DNBR transients presented represent the minimum of the typical or thimble fuel assembly cell.

This analysis employs RTDP methodology; therefore, the initial conditions assume nominal values of power, reactor coolant average temperature, and RCS average pressure. (See tables 15.0.3-2 and 15.0.3-3.) The limit DNBR includes uncertainties in the initial conditions.

This analysis assumes a conservatively large absolute value of the Doppler-only power coefficient. (See figure 15.0.4-1.) This is equivalent to a total integrated Doppler reactivity from 0 to 100 percent power of 0.016  $\Delta k$ .

The analysis assumes the most positive moderator temperature coefficient (minimum moderator density coefficient) since this results in the maximum core power during the initial part of the transient when the transient reaches minimum DNBR. (See figure 15.0.4-2.)

These analyses use the curve of trip reactivity insertion versus time (figure 15.0.5-3).

The basis for the flow coastdown analysis is a momentum balance around each reactor coolant loop and across the reactor core. This momentum balance is combined with the continuity equation, a pump momentum balance, and the pump characteristics and is based on high estimates of system pressure losses.

Plant systems and equipment which are necessary to mitigate the effects of the accident are discussed in subsection 15.0.8 and listed in table 15.0.8-1. No single active failure in any of these systems or equipment will adversely affect the consequences of the accident.

# 15.3.1.2.2 Results

Figures 15.3.1-1 through 15.3.1-4 show the transient response for the loss of power to two RCPs with four loops in operation. The reactor trips on the low-flow signal. Figure 15.3.1-4 shows the DNBR to be always greater than the safety analysis limit value for the most limiting fuel assembly cell.

Since DNB does not occur, the ability of the primary coolant to remove heat from the fuel rod is not significantly reduced. Thus, the average fuel and clad temperature do not increase significantly above their respective initial values.

The time sequence of events is shown in table 15.3.1-1 for the partial loss of flow event.

The affected reactor coolant pumps will continue to coast down, and the core flow will reach a new equilibrium value. With the reactor tripped, a stable plant condition will eventually be attained. Normal plant shutdown may then proceed.

#### 15.3.1.3 <u>Conclusions</u>

The analysis shows that the minimum DNBR always remains above the limit value during the transient. Thus, all applicable acceptance criteria are met.

#### 15.3.1.4 <u>References</u>

- 1. Burnett, T. W. T., <u>et al.</u>, "LOFTRAN Code Description," <u>WCAP-7907-P-A (Proprietary)</u>, <u>WCAP-7907-A (Nonproprietary)</u>, April 1984.
- 2. Hargrove, H. G., "FACTRAN A FORTRAN-IV Code for Thermal Transients in a UO<sub>2</sub> Fuel Rod," <u>WCAP-7908-A</u>, December 1989.

#### 15.3.2 COMPLETE LOSS OF FORCED REACTOR COOLANT FLOW

#### 15.3.2.1 Identification of Causes and Accident Description

A loss-of-forced-reactor-coolant flow may result from a simultaneous loss of electrical power to all RCPs. If the reactor is at power at the time of the accident, the immediate effect of a loss-of-forced-coolant flow is a rapid increase in the coolant temperature. This increase could result in DNB with subsequent adverse effects to the fuel if the reactor does not trip promptly. The reactor trip together with flow sustained by the inertia of the pump impeller will be sufficient to prevent RCS overpressurization and the DNBR from exceeding the limit values.

Two buses connected to the generators supply power to the pumps. When a generator trip occurs, the buses are automatically transferred to a transformer supplied from external power lines, and the pumps continue to operate. Following any turbine trip where there are no electrical faults which require tripping the generator from the network, the generator remains connected to the network for approximately 30 seconds. The RCPs remain connected to the generator, thus ensuring full flow for approximately 30 seconds after the reactor trip before any transfer is made.

The trip systems available to mitigate the consequences of this accident are the following:

- Reactor coolant pump power supply bus undervoltage or underfrequency.
- Low reactor coolant loop flow.

The reactor trip on reactor coolant pump undervoltage is provided to protect against conditions which can cause a loss of voltage to all reactor coolant pumps; i.e., station blackout. This function is blocked below approximately 10-percent power (permissive P7).

The reactor trip on reactor coolant pump underfrequency is provided to trip the reactor for an underfrequency condition, resulting from frequency disturbances on the power grid. Reference 1 provides analyses of grid frequency disturbances and the resulting nuclear steam supply system protection requirements which are applicable to the VEGP units.

The reactor trip on low primary coolant loop flow is provided to protect against loss-of-flow conditions which affect only one reactor coolant loop. This function is generated by two out of three low flow signals per reactor coolant loop. Above permissive P8, low flow in any loop will actuate a reactor trip. Between approximately 10-percent power (permissive P7) and the power level corresponding to permissive P8, low flow in any two loops will actuate a reactor trip. If the

maximum grid frequency decay rate is less than approximately 2.5 Hz/s, this trip function will protect the core from underfrequency events. This effect is fully described in reference 1.

This is an ANS Condition III incident.

#### 15.3.2.2 <u>Analysis of Effects and Consequences</u>

#### 15.3.2.2.1 Method of Analysis

The method of analysis and the assumptions made regarding initial operating conditions and reactivity coefficients are identical to those discussed in subsection 15.3.1, except that following the loss of power supply to all pumps at power, a reactor trip actuates by either RCP power supply undervoltage or underfrequency.

#### 15.3.2.2.2 Results

Figures 15.3.2-1 through 15.3.2-4 show the transient response for the loss of power to all RCPs with four loops in operation. The reactor trips on the undervoltage signal. Figure 15.3.2-4 shows the DNBR to be always greater than the safety analysis limit value for the most limiting fuel assembly cell.

Since DNB does not occur, the ability of the primary coolant to remove heat from the fuel rod is not significantly reduced. Thus, the average fuel and clad temperature do not increase significantly above their respective initial values.

The RCPs will continue to coast down, and natural circulation flow will eventually be established as demonstrated in subsection 15.2.6. With the reactor tripped, a stable plant condition will be attained. Normal plant shutdown may then proceed.

Besides the complete loss-of-forced-reactor-coolant flow (loss of power to four pumps), an underfrequency event with a frequency decay rate of 5 Hz/sec was also analyzed. For this event, the reactor trip occurs on an underfrequency signal. The DNBR analysis of the underfrequency event verified that the DNBR remains above the safety analysis limit value.

The time sequence of events in shown in table 15.3.1-1 for the complete loss-of-forced-reactor-coolant flow.

#### 15.3.2.3 Conclusions

The analysis shows that the minimum DNBR always remains above the limit value during the transient. Thus, the analysis does not predict any adverse fuel effects or clad rupture and all applicable acceptance criteria are met. The design basis for the DNBR is described in section 4.4.

#### 15.3.2.4 <u>Reference</u>

 Baldwin, M. S., <u>et al.</u>, "An Evaluation of Loss of Flow Accidents Caused by Power System Frequency Transients in Westinghouse PWRs," <u>WCAP-8424</u>, Revision 1, May 975.

# 15.3.3 REACTOR COOLANT PUMP SHAFT SEIZURE (LOCKED ROTOR)

#### 15.3.3.1 Identification of Causes and Accident Description

For the instantaneous seizure of an RCP rotor, flow through the affected reactor coolant loop is rapidly reduced, leading to a reactor trip on a low flow signal. Following the trip, heat stored in the fuel rods continues to be transferred into the core coolant, causing the coolant to expand. At the same time, heat transfer to the shell side of the steam generator reduces, first because the reduced flow results in a decreased tube side film coefficient and then because the reactor coolant in the tubes cools down while the shell side temperature increases (turbine steam flow reduces to zero upon plant trip). The rapid expansion of the coolant in the reactor core, combined with the reduced heat transfer in the steam generator, causes an insurge into the pressurizer and a pressure increase throughout the RCS. The insurge into the pressurizer causes a pressure increase, which in turn actuates the automatic spray system, opens the power-operated relief valves are safety grade and would be expected to function properly during an accident; however, for conservatism, the analysis does not use the pressure-reducing effect of the spray.

The analysis of the locked rotor event demonstrates that overpressurization of the RCS does not occur and that the core remains in a coolable geometry.

This is an ANS Condition IV incident.

#### 15.3.3.2 Analysis of Effects and Consequences

#### 15.3.3.2.1 Method of Analysis

The analysis of this transient uses two digital computer codes. The LOFTRAN code (reference 1) calculates: 1) the resulting loop and core flow transients following the pump seizure; 2) the time of reactor trip based on the loop flow transients; 3) the nuclear power following reactor trip; and 4) the peak RCS pressure. The thermal behavior of the fuel located at the core hot spot is investigated using the FACTRAN code (reference 2) based on the core flow and the nuclear power calculated by LOFTRAN.

At the beginning of the postulated locked rotor accident (at the time the shaft in one of the RCPs is assumed to seize), the plant is assumed to be in operation under the most adverse steadystate operating conditions; i.e., maximum guaranteed steady-state thermal power, maximum steady-state pressure, and maximum steady-state coolant average temperature.

The plant characteristics and the initial conditions are shown in table 15.0.3-2 and table 15.0.3-3. The analysis evaluates the transient with and without offsite power available.

For the case without offsite power available, power is lost to the unaffected pumps 2 s after reactor trip. (Note: Grid stability analyses show that the grid will remain stable and that offsite power will not be lost because of a unit trip from 100-percent power. The 2-s delay is a conservative assumption based on grid stability analyses.)

For the peak pressure evaluation, the initial pressure is conservatively estimated as 50 psi above nominal pressure (2250 psia) to allow for errors in the pressurizer pressure measurement and control channels. This is done to obtain the highest possible rise in the coolant pressure

during the transient. To obtain the maximum pressure in the primary side, conservatively high loop pressure drops are added to the calculated pressurizer pressure.

Plant systems and equipment which are available to mitigate the effects of the accident are discussed in subsection 15.0.8 and listed in table 15.0.8-1. No single active failure in any of these systems or equipment will adversely affect the consequences of the accident.

# 15.3.3.2.2 Evaluation of the Pressure Transient

After pump seizure, the neutron flux is rapidly reduced by control rod insertion due to reactor trip on low coolant flow in the affected loop. Rod motion begins 1 second after the flow in the affected loop reaches 87 percent of nominal flow. No credit is taken for the pressure reducing effect of the pressurizer relief valves, pressurizer spray, steam dump, or controlled feedwater flow after plant trip. Although these components will operate and will result in a lower peak RCS pressure, ignoring their effect provides an additional degree of conservatism.

The analysis conservatively bounds the pressurizer safety valves opening at 2500 psia and achieving rated flow at 2575 psia.

# 15.3.3.2.3 Evaluation of Departure from Nucleate Boiling (DNB) in the Core During the Accident

Because DNB occurs in the core for this accident, there is an evaluation of the consequences with respect to fuel rod thermal transients. Results obtained from analyses of this "hot spot" condition represent the upper limit with respect to clad temperature and zirconium-water reaction.

In the evaluation, the rod power at the hot spot is conservatively assumed to be 2.55 times the average rod power (i.e.,  $F_Q = 2.55$ ) at the initial core power level.

A second evaluation is performed for this transient to determine what percentage, if any, of the fuel rods are expected to experience DNB during the transient. For this evaluation, core conditions are generated with the LOFTRAN and FACTRAN computer codes and a detailed DNB analysis is performed with the VIPRE-01 computer code. Results from the VIPRE-01 calculation are then used to determine the percentage of fuel rods which experience DNB. Table 15.0.3-2 presents the initial conditions assumed for the rods-in-DNB evaluation.

# 15.3.3.2.4Film Boiling Coefficient

To model the effect of DNB occurring, the FACTRAN code calculates the film boiling coefficient using the Bishop-Sandberg-Tong film boiling correlation. Fluid properties are evaluated at film temperature (average between wall and bulk temperatures). The program calculates the film coefficient at every time step based upon the actual heat transfer conditions at the time. The neutron flux, system pressure, bulk density, and mass flowrate as a function of time are program inputs.

This analysis uses the initial values of the pressure and the bulk density throughout the transient since they are the most conservative with respect to clad temperature response. For conservatism, the analysis assumes DNB to start at the beginning of the accident to maximize the fuel rod thermal transient.

# 15.3.3.2.5 Fuel Clad Gap Coefficient

The magnitude and time dependence of the heat transfer coefficient between fuel and clad (gap coefficient) have a pronounced influence on the thermal results. The larger the value of the gap coefficient, the more heat transferred between pellet and clad. Based on investigations of the effect of the gap coefficient upon the maximum clad temperature during the transient, the analysis assumes the gap coefficient to increase from a steady-state value consistent with initial fuel temperature to 10,000 Btu/h-ft<sup>2</sup>-°F at the initiation of the transient. Thus, the large amount of energy stored in the fuel because of the small initial value releases to the clad at the initiation of the transient.

# 15.3.3.2.6 Zirconium-Steam Reaction

The zirconium-steam reaction can become significant above 1800°F (clad temperature). In order to take this phenomenon into account, the models (reference 4) introduced the following correlation which defines the rate of the zirconium-steam reaction.

$$\frac{d(w^2)}{dt} = 33.3 \times 10^6 \times e^{-[(45,000.)/(1.986 \text{ T})]}$$

where:

w = amount reacted,  $mg/cm^2$ .

t = time, s.

T = temperature, °F.

The reaction heat is 1510 cal/g.

The effect of zirconium-steam reaction is included in the calculation of the hot spot clad temperature transient.

# 15.3.3.2.7 Results

The transient results for the locked rotor accident are shown in figures 15.3.3-1 through 15.3.3-4. Table 15.3.3-1 also summarizes the results of the locked rotor calculations. The peak RCS pressure reached during the transient is less than that which would cause stresses to exceed the faulted condition stress limits of the American Society of Mechanical Engineers Code, Section III. Also, the peak clad temperature is less than 2700°F (ZIRLO<sup>®</sup>) and the more restrictive limit of 2375°F (associated with Optimized ZIRLO<sup>™</sup> cladding). Note that the clad temperature was conservatively calculated assuming DNB occurs at the initiation of the transient. These results represent the most limiting conditions of the locked rotor event or RCP shaft break.

The calculated sequence of events for the locked rotor event is shown in table 15.3.1-1. Figure 15.3.3-1 shows that the core flow rapidly reaches a new equilibrium value (for the case with offsite power available). With the reactor tripped, a stable plant condition will eventually be attained. Normal plant shutdown may then proceed.

#### 15.3.3.3 Radiological Consequences

The evaluation of the radiological consequences of a postulated seizure of a reactor coolant pump rotor; i.e., locked rotor accident (LRA), assumes that the reactor has been operating with a small percent of defective fuel and leaking steam generator tubes for sufficient time to establish equilibrium concentrations of radionuclides in the reactor coolant and in the secondary coolant.

As a result of the accident, a fraction of the fuel rods will undergo DNB and will release gap inventory to the reactor coolant. Radionuclides carried by the primary coolant to the steam generator via leaking tubes are released to the environment via the steam line safety or power-operated relief valves.

#### 15.3.3.3.1 Analytical Assumptions

The major assumptions and parameters used in the analysis are itemized in table 15.3.3-2. The following is a more detailed discussion of the source term.

15.3.3.3.1.1 <u>Source Term Calculations</u>. The concentration of nuclides in the primary and secondary system prior to and following the LRA are determined as follows:

- A. The iodine activity in the reactor coolant prior to the accident is based upon an iodine spike which has raised the reactor coolant concentration to 60 μCi/g of dose equivalent (DE) I-131.
- B. The noble gas concentrations in the reactor coolant are based upon 1-percent defective fuel.
- C. Following the LRA, 5 percent of the fuel rods in the core undergo DNB. Hence, 5 percent of the core iodine and noble gas gap inventory is released to the reactor coolant.
- D. The secondary coolant iodine activity is based on the DE of 0.1  $\mu$ Ci/g of I-131.

15.3.3.3.1.2 <u>Mathematical Models Used in the Analysis</u>. Mathematical models used in the analysis are described in the following sections:

- A. The mathematical models used to analyze the activity released during the course of the accident are described in appendix 15A.
- B. The atmospheric dispersion factors used in the analysis were calculated based on the onsite meteorological measurement programs described in subsection .3.3.
- C. The thyroid inhalation dose and total-body gamma immersion doses to a receptor at the exclusion area boundary and outer boundary of the low population zone were analyzed using the models described in appendix 15A.

15.3.3.1.3 <u>Identification of Leakage Pathways and Resultant Leakage Activity</u>. Radionuclides carried from the primary coolant to the steam generators via leaking tubes are released to the environment via the steam line safety or power-operated relief valves. Iodines are assumed to mix with the secondary coolant and partition between the generator liquid and steam before release to the environment. Noble gases are assumed to be directly released.

All activity is released to the environment with no consideration given to radioactive decay or to cloud depletion by ground deposition during transport to the exclusion area boundary and low population zone. Hence, the resultant radiological consequences represent the most conservative estimate of the potential intergrated dose due to the postulated LRA.

# 15.3.3.2 Identification of Uncertainties and Conservative Elements in the Analysis

- A. The initial reactor coolant iodine activity is based on the technical specification limit of 1.0  $\mu$ Ci/g of DE I-I3I which is further increased by a large preaccident iodine spike to 60  $\mu$ Ci/g resulting in equivalent concentrations many times greater than the reactor coolant activities based on 0.12-percent defective fuel and expected iodine spiking values associated with normal operating conditions.
- B. The noble gas activities are based on 1-percent defective fuel which cannot exist simultaneously with 1.0- $\mu$ Ci/g I-131. For iodines, 1-percent defects would be approximately three times the technical specification limit.
- C. The fraction of failed fuel is assumed to be equal to the fraction of fuel rods experiencing DNB without consideration given to the extent of the zirc-water reaction. Based on experimental data<sup>(3)</sup> no oxidation related fuel rod clad failure is predicted.
- D. A 1-gal/min steam generator primary-to-secondary leakage is assumed, which is significantly greater than that anticipated during normal operation.
- E. The meteorological conditions which may be present at the site during the course of the accident are uncertain. However, it is highly unlikely that the assumed meteorological conditions would be present during the course of the accident for any extended period of time. Therefore, the radiological consequences evaluated, based on the meteorological conditions assumed, are conservative.

# 15.3.3.3 Conclusions

15.3.3.3.1 <u>Filter Loadings</u>. The only engineered safety feature filtration system considered in the analysis which limits the consequences of the LRA is the control room filtration system.

Integrated activity on the control room filters have been evaluated for the more limiting loss-ofcoolant accident (LOCA) analysis, as discussed in paragraph 15.6.5.4.6. Since the control room filters are capable of accommodating the potential design basis LOCA fission product iodine loadings, there will be sufficient capacity to accommodate any fission product loading due to a postulated LRA.

15.3.3.3.2Doses to Receptor at the Exclusion Area Boundary and Low Population ZoneOuter Boundary.The potential radiological consequences resulting from the occurrence of apostulated LRA have been conservatively analyzed using assumptions and models described.

The total-body gamma dose due to immersion from direct radiation and the thyroid dose due to inhalation have been analyzed for the 0- to 2-h dose at the exclusion area boundary and for the duration of the accident (0 to 20 h) at the low population zone outer boundary. The results are listed in table 15.3.3-3. The resultant doses are well within the guideline values of 10 CFR 100.

#### 15.3.3.4 <u>References</u>

- 1. Burnett, T. W. T., <u>et al.</u>, "LOFTRAN Code Description," <u>WCAP-7907-P-A</u> (Proprietary), <u>WCAP-7907-A</u> (Nonproprietary), April 1984.
- 2. Hargrove, H. G., "FACTRAN A FORTRAN-IV Code for Thermal Transients in a UO<sub>2</sub> Fuel Rod," <u>WCAP-7908-A</u>, December 1989.
- 3. Van Houten, R., "Fuel Rod Failure as a Consequence of Departure from Nucleate Boiling or Dryout," <u>NUREG-0562</u>, June 1979.
- 4. Baker, L. and Just, L., "Studies of Metal Water Reactions of High Temperatures, III Experimental and Theoretical Studies of the Zirconium-Water Reacton, <u>ANL-6548</u>, Argonne National Laboratory, May 1962.

# 15.3.4 REACTOR COOLANT PUMP SHAFT BREAK

#### 15.3.4.1 Identification of Causes and Accident Description

The accident is postulated as an instantaneous failure of a reactor coolant pump shaft, as discussed in section 5.4. Flow through the affected reactor coolant loop is rapidly reduced, though the initial rate of reduction of coolant flow is greater for the reactor coolant pump rotor seizure event. Reactor trip is initiated on a low-flow signal in the affected loop.

Following initiation of the reactor trip, heat stored in the fuel rods continues to be transferred to the coolant, causing the coolant to expand. At the same time, heat transfer to the shell side of the steam generators is reduced--first, because the reduced flow results in a decreased tubeside film coefficient; second, because the reactor coolant in the tubes cools down while the shell-side temperature increases. (Turbine steam flow is reduced to zero upon plant trip.) The rapid expansion of the coolant in the reactor core, combined with reduced heat transfer in the steam generators, causes an insurge into the pressurizer and a pressure increase throughout the reactor coolant system. The insurge into the pressurizer compresses the steam volume, actuates the automatic spray system, opens the power-operated relief valves, and opens the pressurizer safety valves, in that sequence. The two power-operated relief valves are designed for reliable operation and would be expected to function properly during the accident. However, for conservatism, their pressure-reducing effect, as well as the pressure-reducing effect of the spray, is not included in the analysis.

This is an ANS Condition IV incident.

# 15.3.4.2 <u>Conclusion</u>

The consequences of a locked rotor (subsection 15.3.3) represent the most limiting event with respect to the locked rotor or the pump shaft break. With a failed shaft, the impeller could conceivably be free to spin in a reverse direction as opposed to being fixed in position as

assumed in the locked-rotor analysis. However, the net effect on core flow is negligible, resulting in only a slight decrease in the end point (steady-state) core flow. For both the shaft break and locked-rotor incidents, reactor trip occurs very early in the transient. In addition, the locked-rotor analysis conservatively assumes that departure from nucleate boiling occurs at the beginning of the transient.

# TABLE 15.3.1-1 (SHEET 1 OF 2)

# TIME SEQUENCE OF EVENTS FOR INCIDENTS WHICH RESULT IN A DECREASE IN REACTOR COOLANT SYSTEM FLOWRATE

Accident	Event	<u>Time (s)</u>
Partial loss of forced reactor coolant flow		
Loss of two pumps with four loops in operation	Coastdown begins	0.0
	Low-flow reactor trip	1.4
	Rods begin to drop	2.4
	Minimum DNBR occurs	3.6
Complete loss of forced reactor coolant flow		
Loss of four pumps with four loops in operation	All operating pumps lose power and begin coasting down	0.0
	Reactor coolant pump undervoltage trip point reached	0.0
	Rods begin to drop	1.5
	Minimum DNBR occurs	3.2
Reactor coolant pump shaft seizure (locked rotor)		
One locked rotor with four loops in operation with offsite power available	Rotor on one pump locks	0.0
	Low-flow trip point reached	0.0
	Rods begin to drop	1.0

# TABLE 15.3.1-1 (SHEET 2 OF 2)

Accident	<u>Event</u>	<u>Time (s)</u>
	Maximum reactor coolant system pressure occurs	3.3
	Maximum clad average temperature occurs	3.5
One locked rotor with four loops in operation without offsite power available	Rotor on one pump locks	0.0
	Low-flow trip point reached	0.0
	Rods begin to drop	1.0
	Maximum reactor coolant system pressure occurs	3.4
	Maximum clad average temperature occurs	3.6

#### TABLE 15.3.3-1

#### SUMMARY OF RESULTS FOR LOCKED ROTOR TRANSIENTS (FOUR LOOPS OPERATING INITIALLY)

	With Offsite Power Available	Without Offsite Power Available
Maximum RCS pressure (psia)	2669	2669
Maximum clad average temperature, core hot spot (°F)	2048	2054
Zr- $H_2O$ reaction, core hot spot (percent by weight)	0.6	0.7

# TABLE 15.3.3-2 (SHEET 1 OF 2)

#### PARAMETERS USED IN EVALUATING THE RADIOLOGICAL CONSEQUENCES OF A LOCKED ROTOR ACCIDENT

#### I. Source Data

II.

Α.	Core power level (MWt)	3636
В.	Total steam generator tube leakage (gal/min)	1
C.	Reactor coolant iodine activity prior to accident	An assumed preaccident iodine spike, which has resulted in the DE of 60 $\mu$ Ci/g of I-131 in the reactor coolant. See table 15A-6.
D.	Gap activity released to reactor coolant from failed fuel	5 percent. See table 15A-3.
E.	Reactor coolant noble gas activity	Based on 1-percent defective fuel. See table 15A-4.
F.	Secondary system initial activity	DE of 0.1 $\mu\text{Ci/g}$ of I-131.
G.	Reactor coolant mass (g)	2.3 x 10 <sup>8</sup>
Н.	Secondary coolant mass, 4 generators (g)	1.9 x 10 <sup>8</sup>
I.	Offsite power	Lost after trip
J.	Primary-to-secondary leakage duration (h)	20
K.	Species of iodine	100-percent elemental
	Atmospheric Dispersion Factors	See table 15A-2.

# TABLE 15.3.3-2 (SHEET 2 OF 2)

III. Activity Release Data

A.	Primary-to-secondary leak-rate (gal/min) <sup>(a</sup>	1.0	
B.	Steam Released (lb) 0 to 2 h 2 to 8 h 8 to 20 h		555,000 1,365,000 2,730,000

C. Iodine partition factor 100

1

a. Based on water at 62.4 lb/ft<sup>3</sup>.

# TABLE 15.3.3-3

#### RADIOLOGICAL CONSEQUENCES OF A LOCKED ROTOR ACCIDENT

Exclusion Area Boundary (0 to 2 h)	<u>Doses (rem)</u>
Thyroid	0.6
Whole-body gamma	0.1
Lower Population Zone Outer Boundary (20 h )	
Thyroid	4.2
Whole-body gamma	0.1
































### 15.4 REACTIVITY AND POWER DISTRIBUTION ANOMALIES

Several postulated faults can result in reactivity and power distribution anomalies. Control rod motion, control rod ejection, boron concentration changes, or addition of cold water to the RCS results in reactivity changes. Control rod motion, control rod misalignment, control rod ejection, or fuel assembly mislocation results in power distribution changes. This section discusses these events. Detailed analyses are presented for the most limiting of these events.

This section presents the following incidents:

- Uncontrolled rod cluster control assembly (RCCA) bank withdrawal from a subcritical or low-power startup condition.
- Uncontrolled RCCA bank withdrawal at power.
- RCCA misalignment.
- Startup of an inactive reactor coolant pump at an incorrect temperature.
- A malfunction or failure of the flow controller in a boiling water reactor recirculation loop that results in an increased reactor coolant flowrate (not applicable to VEGP).
- Chemical and volume control system malfunction that results in a decrease in the boron concentration in the reactor coolant.
- Inadvertent loading and operation of a fuel assembly in an improper position.
- Spectrum of RCCA ejection accidents.
- Steamline break with coincidental RCCA bank withdrawal at power.

All of the accidents in this section have been analyzed. The most severe radiological consequences result from the complete rupture of a control rod drive mechanism housing provided in subsection 15.4.8; therefore, radiological consequences are reported only for that limiting case.

#### 15.4.1 UNCONTROLLED ROD CLUSTER CONTROL ASSEMBLY BANK WITHDRAWAL FROM A SUBCRITICAL OR LOW-POWER STARTUP CONDITION

#### 15.4.1.1 Identification of Causes and Accident Description

An RCCA withdrawal incident is an uncontrolled addition of reactivity to the reactor core caused by withdrawal of RCCA banks resulting in a power excursion. While the occurrence of a transient of this type is highly unlikely, a malfunction of the control rod drive system can cause such a transient. This could occur with the reactor either subcritical, low power startup, or at power. Subsection 15.4.2 discusses the "at power" case. RCCA bank withdrawal adds reactivity at a prescribed and controlled rate to bring the reactor from a subcritical condition to a low power level during startup. Although the initial startup procedure uses the method of boron dilution, the normal startup is with RCCA bank withdrawal. RCCA bank motion can cause much faster changes in reactivity than can be made by changing boron concentration.

The control rod drive mechanisms wire into preselected banks which remain unchanged during the core life. The circuit design is such that RCCAs cannot be withdrawn in other than their proper withdrawal sequence. Control of the power supplied to the rod banks is such that no more than two banks can be withdrawn at any time. The RCCA drive mechanism is the magnetic latch type, and the coil actuation sequencing provides variable speed travel. The analysis of the maximum reactivity insertion rate includes the assumption of the simultaneous withdrawal of the two sequential banks having the maximum combined worth at maximum speed.

The neutron flux response to a continuous reactivity insertion is characterized by a fast rise terminated by the reactivity feedback effect of the negative Doppler coefficient. This self-limitation of the power excursion is of primary importance since it limits the power to a tolerable level during the delay time for protective action. Should a continuous control rod assembly withdrawal initiate, the transient will terminate by the following reactor trip functions:

- Source range high neutron flux reactor trip is actuated when either of two independent source range channels indicates a flux level above a preselected, manually adjustable setpoint. This trip function may be manually bypassed when the intermediate range flux channel indicates a flux level above the source range cutoff level. It is automatically reinstated when both intermediate range channels indicate a flux level below a specified setpoint.
- Intermediate range high neutron reactor flux trip is actuated when either of two independent intermediate range channels indicates a flux level above a preselected, manually adjustable setpoint. This trip function may be manually bypassed when two of the four power range channels are reading above approximately 10 percent of full power/flux level and is automatically reinstated when three of the four power range channels indicate a power/flux level below this setpoint.
- Power range high neutron flux reactor trip (low setting) is actuated when two out of the four power range channels indicate a flux level above approximately 25 percent of full power/flux level. This trip function may be manually bypassed when two of the four power range channels indicate a flux level above approximately 10 ercent of full power/flux level and is automatically reinstated when three of the four channels indicate a power/flux level below this setpoint.
- Power range high neutron flux reactor trip (high setting) is actuated when two out of the four power range channels indicate a flux level above a preset setpoint. This trip function is always active.
- High neutron flux rate reactor trip is actuated when the positive rate of change of neutron flux on two out of four nuclear power range channels indicates a rate above the preset setpoint. It is always active.

In addition, control rod stops on high intermediate range flux (one out of two) and high power range flux (one out of four) serve to cease rod withdrawal and prevent the need to actuate the intermediate range flux trip and the power range flux trip, respectively.

This is an ANS Condition II incident.

## 15.4.1.2 Analysis of Effects and Consequences

#### 15.4.1.2.1 Method of Analysis

The following three stages comprise the analysis of the uncontrolled RCCA bank withdrawal from subcritical accident: first, an average core nuclear power transient calculation; then, an average core heat transfer calculation; and finally, the DNBR calculation. The spatial neutron kinetics computer code TWINKLE (reference 1) performs the average core calculation to determine the average power generation with time including the various total core feedback effects; i.e., Doppler reactivity and moderator reactivity. FACTRAN (reference 2) performs a fuel rod transient heat transfer calculation to determine the average heat flux and temperature transients. The average heat flux is next used in VIPRE-01 (section 4.4) for transient DNBR calculations.

In order to give conservative results for a startup incident, the following additional assumptions are made concerning the initial reactor conditions:

- A. Since the magnitude of the neutron flux peak reached during the initial part of the transient for any given rate of reactivity insertion is strongly dependent on the Doppler power reactivity coefficient, the analysis employs a conservatively low value for Doppler power defect (-998 pcm).
- B. The contribution of the moderator reactivity coefficient is negligible during the initial part of the transient because the heat transfer time constant between the fuel and the moderator is much longer than the neutron flux response time constant; however, after the initial neutron flux peak, the moderator temperature reactivity coefficient affects the succeeding rate of power increase. The analysis assumes a moderator temperature coefficient which is +7 pcm/°F at the zero power nominal temperature.
- C. The analysis assumes the reactor to be at hot zero power (557°F). This assumption is more conservative than that of a lower initial system temperature. The higher initial system temperature yields a larger fuel-to-water heat transfer coefficient, a larger fuel-specific heat, and a less-negative (smaller absolute magnitude) Doppler coefficient; these reduce the Doppler feedback effect, thereby increasing the neutron flux peak. The high neutron flux peak combined with a high fuel specific eat and larger heat transfer coefficient yields a larger peak heat flux. The analysis assumes the initial effective multiplication factor ( $k_{eff}$ ) to be 1.0 since this results in the maximum neutron flux peak.
- D. The most adverse combination of instrumentation error, setpoint error, delay for trip signal actuation, and delay for control rod assembly release is taken into account. The analysis assumes a 10-percent increase in the power range flux trip setpoint, raising it from the nominal value of 25 percent to a value of 35 percent and not taking any credit for the source and intermediate range protection. Figure 15.4.1-1 shows that the rise in nuclear flux is so rapid that the effect of error in the trip setpoint on the actual time at which the rods release is negligible. Besides the above, the assumption that the highest worth control rod assembly is stuck in its fully withdrawn position is the basis of the rate of negative reactivity insertion corresponding to the reactor trip.

- E. The maximum positive reactivity insertion rate assumed is greater than that for the simultaneous withdrawal of the two sequential control banks having the greatest combined worth at maximum speed (45 in./min).
- F. The DNB analysis assumes the most limiting axial and radial power shapes associated with having the two highest combined worth banks in their high-worth position.
- G. The analysis assumes the initial power level to be below the power level expected for any shutdown condition (10<sup>-9</sup> fraction of nominal power). The combination of highest reactivity insertion rate and low initial power produces the highest peak heat flux.
- H. The analysis assumes two RCPs to be in operation (Mode 3 Technical Specification allowed operation). This is conservative with respect to the DNB transient.

The accident analysis employs the STDP with the initial conditions shown in tables 15.0.3-2 and 5.0.3-3.

Plant systems and equipment which are available to mitigate the effects of the accident are discussed in subsection 15.0.8 and listed in table 15.0.8-1. No single active failure in any of these systems or components will adversely affect the consequences of the accident.

## 15.4.1.2.2 Results

Figures 15.4.1-1 through 15.4.1-3 show the transient behavior for the uncontrolled RCCA bank withdrawal incident, with the accident terminated by reactor trip at 35 percent of nominal power. The reactivity insertion rate used is greater than that calculated for the two highest worth sequential control banks, both assumed to be in their highest incremental worth region.

Figure 15.4.1-1 shows the average neutron flux transient.

The energy release and the fuel temperature increases are relatively small. The thermal flux response, of interest for DNB considerations, is shown on figure 15.4.1-2. The beneficial effect of the inherent thermal lag in the fuel is evidenced by a peak heat flux much less than the full-power nominal value. There is margin to DNB during the transient. Figure 15.4.1-3 shows the response of the hot spot average fuel and clad inner temperatures. The minimum DNBR at all times remains above the limiting value.

The calculated sequence of events for this accident is shown on table 15.4.1-1. With the reactor tripped, the plant returns to a stable condition. The plant may subsequently be cooled down further by following normal plant shutdown procedures.

## 15.4.1.3 <u>Conclusions</u>

In the event of an RCCA withdrawal accident from the subcritical condition, the core and the RCS are not adversely affected, since the combination of thermal power and the coolant temperature results in a DNBR greater than the limiting value. The DNBR design basis is described in section 4.4; applicable acceptance criteria have been met.

#### 15.4.1.4 <u>References</u>

- 1. Risher, D. H., Jr., and Barry, R. F., "TWINKLE--A Multi-Dimensional Neutron Kinetics Computer Code," <u>WCAP-7979-P-A</u> (Proprietary) and <u>WCAP-8028-A</u> (Nonproprietary), January 1975.
- 2. Hargrove, H. G., "FACTRAN--A FORTRAN-IV Code for Thermal Transients in a UO<sub>2</sub> Fuel Rod," <u>WCAP-7908-A</u>, December 1989.

#### 15.4.2 UNCONTROLLED ROD CLUSTER CONTROL ASSEMBLY BANK WITHDRAWAL AT POWER

#### 15.4.2.1 Identification of Causes and Accident Description

An uncontrolled RCCA withdrawal at power results in an increase in core heat flux. Since the heat extraction from the steam generator lags behind the core power generation until the steam generator pressure reaches the relief or safety valve setpoint, there is a net increase in the reactor coolant temperature. Unless terminated by manual or automatic action, the power mismatch and resultant coolant temperature rise could eventually result in DNB; therefore, to avert damage to the fuel clad the reactor protection system is designed to terminate any such transient before DNBR falls below the safety analysis limit.

The automatic features of the reactor protection system which prevent core damage in an RCCA bank withdrawal incident at power include the following:

- Power range neutron flux instrumentation actuates a reactor trip on high neutron flux if two out of four channels exceed an overpower setpoint.
- Reactor trip actuates if any two out of four ∆T channels exceed an OT∆T setpoint. This setpoint is automatically varied with axial power distribution, coolant average temperature, and coolant average pressure to protect against DNB.
- Reactor trip actuates if any two out of four ∆T channels exceed an OP∆T setpoint. This setpoint is automatically varied with coolant average temperature so that the allowable heat generation rate (kW/ft) is not exceeded.
- A high pressurizer pressure reactor trip, actuated from any two out of four pressure channels, is set at a fixed point. This set pressure is less than the set pressure for the pressurizer safety valves.
- Any two out of three level channels when the reactor power is above approximately 10 percent (permissive P-7) will actuate a high pressurizer water level reactor trip.
- Power range neutron flux instrumentation actuates a reactor trip if two out of four channels exceed a specified positive flux rate. (This trip is credited for the RCS overpressure limit. It is not credited in the reactor core protection analyses.)

Besides the above listed reactor trips, there are the following RCCA withdrawal blocks:

• High neutron flux (one out of four).

- OP $\Delta$ T (two out of four).
- OT $\Delta$ T (two out of four).

The manner in which the combination of  $OP\Delta T$  and  $OT\Delta T$  trips provide protection over the full range of RCS conditions is described in chapter 7. Figure 15.0.6-1 presents allowable reactor coolant loop average temperature and  $\Delta T$  for the design power distribution and flow as a function of primary coolant pressure. The boundaries of operation defined by the  $OP\Delta T$  and  $OT\Delta T$  trip are represented as "protection lines" on this diagram. The protection lines are drawn to include all adverse instrumentation and setpoint errors so that under nominal conditions trip would occur well within the area bounded by these lines. The utility of this diagram is in the fact that the limit imposed by any given DNBR can be represented as a line. The DNB lines represent the locus of conditions for which the DNBR equals the safety analysis limit. All points below and to the left of a DNB line for a given pressure have a DNBR greater than the safety analysis limit. The diagram shows that DNB is prevented for all cases if the area enclosed with the maximum protection lines is not traversed by the applicable DNBR line at any point.

The area of permissible operation (power, pressure, and temperature) is bounded by the combination of reactor trips:

- High neutron flux (fixed setpoint).
- High pressure (fixed setpoint).
- Low pressure (fixed setpoint).
- OP $\Delta$ T and OT $\Delta$ T (variable setpoints).

The purpose of this analysis is to demonstrate the manner in which the above protective systems function for various reactivity insertion rates from different initial conditions to prevent fuel damage. Reactivity insertion rates and initial conditions influence which protection function actuates first.

Reference 3 documents that a conservative analysis has been performed, assuming the reactor trip listed in table 15.0.6-1, that ensures that the RCS overpressure limit will not be exceeded for an uncontrolled rod withdrawal during power operation.

This is an ANS Condition II incident.

## 15.4.2.2 Analysis of Effects and Consequences

## 15.4.2.2.1 Method of Analysis

This transient is analyzed by the LOFTRAN code (reference 1). This code simulates the neutron kinetics, RCS, pressurizer, pressurizer relief and safety valves, pressurizer spray, steam generator, and steam generator safety valves. The code computes pertinent plant variables including temperatures, pressures, and power level. The core limits as illustrated in figure 15.0.6-1 are used as input to LOFTRAN to determine the minimum DNBR during the transient.

The analysis of this accident uses the RTDP described in reference 2. Subsection 15.0.3 discusses the plant characteristics and initial conditions. For an uncontrolled RCCA bank withdrawal at power accident, the analysis assumes the following conservative assumptions:

- A. Nominal values form the basis of the initial reactor power, pressure, and RCS temperature assumption. The limit DNBR includes uncertainties in initial conditions as described in reference 2.
- B. Reactivity coefficients -- two cases analyzed:
  - 1. A +7 pcm/°F moderator temperature coefficient of reactivity and a leastnegative Doppler-only power coefficient form the basis of the beginningof-life minimum reactivity feedback assumption.
  - 2. A conservatively large positive moderator density coefficient (corresponding to a large negative moderator temperature coefficient) and a most-negative Doppler-only power coefficient form the basis of the end-of-life maximum reactivity feedback assumption.
- C. A conservative value of 118 percent of nominal full core power actuates the reactor trip on high neutron flux. The  $\Delta T$  trips include all adverse instrumentation and setpoint errors while maximum values form the basis of the delays for the trip signal actuation assumption.
- D. The assumption that the highest worth assembly is stuck in its fully withdrawn position forms the basis of the RCCA trip insertion characteristic.
- E. The analysis examines a range of reactivity insertion rates. The maximum positive reactivity insertion rate is greater than that for the simultaneous withdrawal of the two control banks having the maximum combined worth at maximum speed assuming normal overlap.

The effect of RCCA movement on the axial core power distribution is accounted for by causing a decrease in  $OT_{\Delta}T$  trip setpoint proportional to a decrease in margin to DNB.

Plant systems and equipment which are available to mitigate the effects of the accident are discussed in subsection 15.0.8 and listed in table 15.0.8-1. No single active failure in any of these systems or equipment will adversely offset the consequences of the accident. A discussion of anticipated transients without trip considerations is presented in section 15.8.

## 15.4.2.2.2 Results

Figures 15.4.2-1 through 15.4.2-3 show the transient response for a rapid RCCA bank withdrawal incident starting from full power with minimum feedback. Reactor trip on high neutron flux occurs shortly after the start of the accident. Because of the rapid reactor trip with respect to the thermal time constants of the plant, small changes in  $T_{avg}$  and pressure result, and the margin to DNB is maintained.

The transient response for a slow RCCA withdrawal from full power with minimum feedback is shown in figures 15.4.2-4 through 15.4.2-6. Reactor trip on  $OT\Delta T$  occurs after a longer period and the rise in temperature and pressure is consequently larger than for rapid RCCA bank withdrawal. Again, the minimum DNBR is greater than the safety analysis limit.

Figure 15.4.2-7 shows the minimum DNBR as a function of reactivity insertion rate from initial full-power operation for both minimum and maximum reactivity feedback. It can be seen that the two reactor trip functions (high neutron flux and  $OT\Delta T$  functions) provide DNB protection

over the whole range of reactivity insertion rates. The minimum DNBR is always greater than the safety analysis limit.

Figures 15.4.2-8 and 15.4.2-9 show the minimum DNBR as a function of reactivity insertion rate for RCCA withdrawal incidents starting at 60- and 10-percent power, respectively. The results are similar to the 100-percent power case; however, as the initial power decreases, the range over which the  $OT\Delta T$  trip is effective is increased. In neither case does the DNBR fall below the safety analysis limit.

The shape of the curves of minimum DNBR versus reactivity insertion rate in the referenced figures is due both to reactor core and coolant system transient response and to protection system action initiating a reactor trip.

Referring to figure 15.4.2-8, for example, it is noted that:

- A. For high reactivity insertion rates; i.e., between approximately  $1.5 \times 10^{-4} \Delta k/s$  and  $1.0 \times 10^{-3} \Delta k/s$ , reactor trip is initiated by the high neutron flux trip for the minimum reactivity feedback cases. The neutron flux level in the core rises rapidly for these insertion rates while core heat flux and coolant system temperature lag behind due to the thermal capacity of the fuel and coolant system fluid. Thus, the reactor is tripped prior to significant increase in heat flux or water temperature with resultant high minimum DNBRs during the transient. As reactivity insertion rate decreases, core heat flux and coolant temperatures can remain more nearly in equilibrium with the neutron flux; minimum DNBR during the transient thus decreases with decreasing insertion rate.
- B. The OT∆T reactor trip circuit initiates a reactor trip when measured coolant loop ∆T exceeds a setpoint based on measured RCS average temperature and pressure. This trip circuit is described in detail in chapter 7; however, it is important in this context to note that the average temperature contribution to the circuit is lead-lag compensated to decrease the effect of the thermal capacity of the RCS in response to power increases.
- C. With further decrease in reactivity insertion rate, the OT $\Delta$ T and high neutron flux trips become equally effective in terminating the transient; e.g., at approximately  $1.5 \times 10^{-4} \Delta k/s$  reactivity insertion rate. For reactivity insertion rates between approximately  $1 \times 10^{-4} \Delta k/s$  and approximately  $2 \times 10^{-5} \Delta k/s$ , the effectiveness of the OT $\Delta$ T trip increases (in terms of increased minimum DNBR) due to the fact that with lower insertion rates the power increase rate is slower, the rate of rise of average coolant temperature is slower, and the system lags and delays become less significant.
- D. Referring to figure 15.4.2-9, it is shown that for reactivity insertion rates less than approximately 5 x  $10^{-5} \Delta k/s$ , the rise in the reactor coolant temperature is sufficiently high so that the steam generator safety valve setpoint is reached prior to trip in the minimum feedback case. Opening of these valves, which act as an additional heat load of the RCS, sharply decreases the rate of increase of RCS average temperature. This decrease in rate of increase of the average coolant system temperature during the transient is accentuated by the lead-lag compensation, causing the OT $\Delta$ T trip setpoint to be reached later, with resulting lower minimum DNBRs.

For transients initiated from higher power levels (for example, see figure 15.4.2-7) this effect, described in item D above, which results in the sharp peak in minimum DNBR at approximately

3 x 10<sup>-5</sup>  $\Delta$ k/s, does not occur since the steam generator safety valves are never actuated prior to trip.

Figures 15.4.2-7, 15.4.2-8, and 15.4.2-9 illustrate minimum DNBR calculated for minimum and maximum reactivity feedback.

Since the RCCA withdrawal at power incident is an overpower transient, the fuel temperatures rise during the transient until after reactor trip occurs. For high reactivity insertion rates, the overpower transient is fast with respect to the fuel rod thermal time constant and the core heat flux lags behind the neutron flux response. Due to this lag, the peak core heat flux does not exceed 118 percent of its nominal value; i.e., the high neutron flux trip setpoint assumed in the analysis. Taking into account the effect of the RCCA withdrawal on the axial core power distribution, the peak fuel temperature will still remain below the fuel melting temperature.

For slow reactivity insertion rates, the core heat flux remains more nearly in equilibrium with the neutron flux. The overpower transient is terminated by the  $OT\Delta T$  reactor trip before a DNB condition is reached. The peak heat flux again is maintained below 118 percent of its nominal value. Taking into account the effect of the RCCA withdrawal on the axial core power distribution, the peak fuel centerline temperature will remain below the fuel melting temperature.

The reactor is tripped sufficiently fast during the RCCA bank withdrawal at power transient to ensure that the ability of the primary coolant to remove heat from the fuel rods is not reduced. Thus, the fuel cladding temperature does not rise significantly above its initial value during the transient.

The calculated sequence of events for this accident is shown on table 15.4.1-1. With the reactor tripped, the plant eventually returns to a stable condition. The plant may subsequently be cooled down further by following normal plant shutdown procedures.

## 15.4.2.3 <u>Conclusions</u>

The high neutron flux and  $OT\Delta T$  trip channels provide adequate protection over the entire range of possible reactivity insertion rates; i.e., the minimum value of DNBR is always larger than the limiting value.

#### 15.4.2.4 <u>References</u>

- 1. Burnett, T. W. T., <u>et al.</u>, "LOFTRAN Code Description," <u>WCAP-7907-P-A</u> (Proprietary), <u>WCAP-7907-A</u> (Nonproprietary), April 1984.
- 2. Friedland, A. J. and Ray, S., "Revised Thermal Design Procedure," <u>WCAP-11397-P-A</u> (Proprietary), April 1989.
- 3. Westinghouse letter (GP-18572 dated October 14, 2009, "Transmittal of Results and Recommendations Regarding Rod Withdrawal at Power RCS Overpressurization."

## 15.4.3 ROD CLUSTER CONTROL ASSEMBLY MISALIGNMENT (SYSTEM MALFUNCTION OR OPERATOR ERROR)

#### 15.4.3.1 Identification of Causes and Accident Description

RCCA misoperation accidents include the following:

- One or more dropped RCCAs within the same group.
- A dropped RCCA bank.
- Statically misaligned RCCA.
- Withdrawal of a single RCCA.

Each RCCA has a position indicator channel which displays the position of the assembly in a display grouping that is convenient to the operator. Fully inserted assemblies are also indicated by a rod at bottom signal which actuates a local alarm and a control room annunciator. Group demand position is also indicated.

RCCAs move in preselected banks, and the banks move in the same preselected sequence. Each bank of RCCAs consists of two groups. The rods comprising a group operate in parallel through multiplexing thyristors. The two groups in a bank move sequentially such that the first group is always within one step of the second group in the bank. A definite schedule of actuation (or deactuation of the stationary gripper, movable gripper, and lift coils of a mechanism) withdraws the RCCA attached to the mechanism. Mechanical failures are in the direction of insertion or immobility.

The dropped RCCAs, dropped RCCA bank, and statically misaligned RCCA events are classified as ANS Condition II incidents (incidents of moderate frequency) as defined in subsection 15.0.1. The single RCCA withdrawal incident is classified as an ANS Condition III event, as discussed below.

No single electrical or mechanical failure in the rod control system could cause the accidental withdrawal of a single RCCA from the inserted bank at full-power operation. The operator could withdraw a single RCCA in the control bank, since this feature is necessary in order to retrieve an assembly should one be accidentally dropped. The event analyzed must result from multiple wiring failures (probability for single random failure is on the order of 10<sup>-6</sup>/h (paragraph 7.7.2.2)) or multiple significant operator errors and subsequent and repeated operator disregard of event indication. The probability of such a combination of conditions is considered low such that the limiting consequences may include slight fuel damage.

Thus, consistent with the philosophy and format of American National Standards Institute N18.2, the event is classified as a Condition III event. By definition, "Condition III occurrences include incidents, any one of which may occur during the lifetime of a particular plant," and "shall not cause more than a small fraction of fuel elements in the reactor to be damaged . . . ."

This selection of criteria is in accordance with General Design Criterion (GDC) 25, which states, "The protection system shall be designed to assure that specified acceptable fuel design limits are not exceeded for any <u>single</u> malfunction of the reactivity control systems, such as accidental withdrawal (not ejection or dropout) of control <u>rods</u>." (Emphases have been added.) It has been shown that single failures resulting in RCCA bank withdrawals do not violate specified fuel design limits. Moreover, no single malfunction can result in the withdrawal of a single RCCA.

Thus, it is concluded that criteria established for the single rod withdrawal at power are appropriate and in accordance with GDC 25.

The following indicators detect one or more dropped RCCAs, RCCA group, or RCCA bank:

- Sudden drop in the core power level as seen by the nuclear instrumentation system.
- Asymmetric power distribution as seen on out-of-core neutron detectors or core exit thermocouples.
- Rod at bottom signal.
- Rod deviation alarm.
- Rod position indication.

The following indicators detect misaligned RCCAs:

- Asymmetric power distribution as seen on out-of-core neutron detectors or core exit thermocouples.
- Rod deviation alarm.
- Rod position indicators.

The resolution of the rod position indicator channel is  $\pm 5$  percent of span ( $\pm 7.5$  in.). Deviation of any RCCA from its group by twice this distance (10 percent of span or 15 in.) will not cause power distributions worse than the design limits. The deviation alarm alerts the operator to rod deviation with respect to the group position in excess of 5 percent of span. If the rod deviation alarm is not operable, the Technical Specifications require the operator to take action.

If one or more rod position indicator channel is out of service, the operator must follow detailed operating instructions to ensure the alignment of the nonindicated RCCAs. The operator is also required to take action, as required by the Technical Specifications.

In the unlikely event of simultaneous electrical failures which could result in single RCCA withdrawal, the plant annunciator will display both the rod deviation and rod control urgent failure; and the rod position indicators will indicate the relative positions of the RCCAs in the bank. The urgent failure alarm also inhibits automatic rod motion in the group in which it occurs. Withdrawal of a single RCCA by operator action, whether deliberate or by a combination of errors, would result in activation of the same alarm and the same visual indication. The OT $\Delta$ T reactor trip provides automatic protection for this event, although due to the increase in local power density, it is not possible to always provide assurance that the core safety limits will not be exceeded.

Plant systems and equipment which are available to mitigate the effects of the various control rod misoperations are discussed in subsection 15.0.8 and listed in table 15.0.8-1. No single active failure in any of these systems or equipment will adversely affect the consequences of the accident.

15.4.3.2 Analysis of Effects and Consequences

#### 15.4.3.2.1 Dropped RCCAs, Dropped RCCA Bank, and Statically Misaligned RCCA

#### 15.4.3.2.1.1 <u>Method of Analysis.</u>

#### A. One or More Dropped RCCAs From the Same Group

The LOFTRAN computer code (reference 2) calculates the transient system response for the evaluation of the dropped RCCA event. The code simulates the neutron kinetics, RCS, pressurizer, pressurizer relief and safety valves, pressurizer spray, steam generator, and steam generator safety valves. The code computes pertinent plant variables including temperatures, pressures, and power level.

Calculated statepoints and nuclear models form the basis used to obtain a hot channel factor consistent with the primary system conditions and reactor power. By incorporating the primary conditions from the transient and the hot channel factor from the nuclear analysis, the DNB design basis is shown to be met using the VIPRE-01 code. The transient response analysis, nuclear peaking factor analysis, and performance of the DNB design basis confirmation are in accordance with the methodology described in reference 1. Note that the analysis does not take credit for the negative flux rate reactor trip.

B. Dropped RCCA Bank

A dropped RCCA bank results in a symmetric power change in the core. As discussed in reference 1, assumptions made for the dropped RCCA(s) analysis provide a bounding analysis for the dropped RCCA bank.

C. Statically Misaligned RCCA

Table 4.1-2 described the computer codes used in the analysis of steady-state power distributions. The peaking factors are then used as input to the VIPRE-01 code to calculate the DNBR. The analysis examines the case of the worst rod withdrawn from bank D inserted at the insertion limit with the reactor initially at full power. The analysis assumes this incident to occur at beginning of life since this results in the minimum value of moderator temperature coefficient. This assumption maximizes the power rise and minimizes the tendency of increased moderator temperature to flatten the power distribution.

D. Single RCCA Withdrawal at Full Power

Table 4.1-2 describes the computer codes used in the calculation of power distributions within the core. The peaking factors are then used in the DNB evaluation for the event. The plant's analysis is for the case of the worst withdrawn rod from D bank inserted at the insertion limit, with the reactor initially at full power. The analysis assumes the transient to occur at beginning of life since this results in the minimum value of moderator temperature coefficient. This assumption maximizes the power rise and minimizes the tendency of increased moderator temperature to flatten the power distribution.

#### 15.4.3.2.1.2 <u>Results.</u>

#### A. One or More Dropped RCCAs

Single or multiple dropped RCCAs within the same group result in a negative reactivity insertion. The core is not adversely affected during this period, since power is decreasing rapidly. Power may be reestablished either by reactivity feedback or control bank withdrawal.

Following a dropped rod event in manual rod control, the plant will establish a new equilibrium condition. The equilibrium process without control system interaction is monotonic, thus removing power overshoot as a concern and establishing the automatic rod control mode of operation as the limiting case.

For a dropped RCCA event in the automatic rod control mode (insertion and withdrawal), the rod control system detects the drop in power and initiates control bank withdrawal. Power overshoot may occur due to this action by the automatic rod controller after which the control system will insert the control bank to restore nominal power. Figures 15.4.3-1 and 15.4.3-2 show a typical transient response to a dropped RCCA (or RCCAs) in automatic control. Uncertainties in the initial condition are included in the DNB evaluation as described in reference 1. In all cases, the minimum DNBR remains above the limit value.

Following plant stabilization, the operator may manually retrieve the RCCA by following approved operating procedures.

B. Dropped RCCA Bank

A dropped RCCA bank results in a negative reactivity insertion greater than 500 pcm. The core is not adversely affected during the insertion period, since power is decreasing rapidly. The transient will proceed as described in part A; however, the return to power will be less due to the greater worth of the entire bank. The power transient for a dropped RCCA bank is symmetric. Following plant stabilization, normal procedures are followed.

#### C. Statically Misaligned RCCA

The most severe misalignment situations with respect to DNBR at significant power levels arise from cases in which one RCCA is fully inserted or where bank D is fully inserted with one RCCA fully withdrawn. Multiple independent alarms, including a bank insertion limit alarm, alert the operator well before the transient approaches the postulated conditions. The bank can be inserted to its insertion limit with any one assembly fully withdrawn without the DNBR falling below the limit value.

The insertion limits in the Core Operating Limits Report (COLR) may vary from time to time depending on several limiting criteria. The full-power insertion limits on control bank D must be chosen to be above that position which meets the minimum DNBR and peaking factors. The full power insertion limit is usually dictated by other criteria. Detailed results will vary from cycle to cycle depending on fuel arrangements.

For this RCCA misalignment, with bank D inserted to its full-power insertion limit and one RCCA fully withdrawn, DNBR does not fall below the limit value. The analysis of this case assumes that the initial reactor power, pressure, and RCS temperature are at their nominal values, with the increased radial peaking factor associated with the misaligned RCCA. For RCCA misalignments with one RCCA fully inserted, the DNBR does not fall below the limit value. The analysis of this case assumes that initial reactor power, pressure, and RCS temperatures are at their nominal values, with the increased radial peaking factor associated with the misaligned RCCA.

DNB does not occur for the RCCA misalignment incident, thus there is no reduction in the ability of the primary coolant to remove heat from the fuel rod. The peak fuel temperature corresponds to a linear heat generation rate based on the radial peaking factor penalty associated with the misaligned RCCA and the design axial power distribution. The resulting linear heat generation rate is well below that which would cause fuel melting.

After identifying an RCCA group misalignment condition, the operator must take action as required by the plant Technical Specifications and operating instructions.

- D. The analysis of the single rod withdrawal event considers the following two events:
  - 1. If the reactor is in the manual rod control mode, continuous withdrawal of a single RCCA results in both an increase in core power and coolant temperature and an increase in the local hot channel factor in the area of the withdrawing RCCA. Depending on initial bank insertion and location of the withdrawn RCCA, automatic reactor trip may not occur quickly enough to prevent the minimum DNBR from falling below the limit value. Evaluation of this case at the power and coolant conditions at which the OT∆T trip would trip the plant shows that an upper limit for the number of rods with a DNBR less than the limit value is 5 percent.
  - 2. If the reactor is in the automatic rod control mode, the multiple failures that result in the withdrawal of a single RCCA cause immobility of the other RCCAs in the controlling bank. The transient will then proceed in the same manner as case 1 described above.

For such cases as above, a reactor trip will ultimately ensue, although not quickly enough in all cases to prevent a minimum DNBR in the core of less than the limit value. Following reactor trip, normal shutdown procedures are followed.

## 15.4.3.3 <u>Conclusions</u>

For cases of dropped RCCAs or dropped banks, the DNBR remains greater than the limit value; therefore, the DNB design criterion is met.

For all cases of any RCCA fully inserted, or bank D inserted to its rod insertion limits with any single RCCA in that bank fully withdrawn (static misalignment), the DNBR remains greater than the limit value.

For the case of the accidental withdrawal of a single RCCA, with the reactor in the automatic or manual control mode and initially operating at full power with bank D at the insertion limit, an upper bound of the number of fuel rods experiencing DNBR is 5 percent of the total fuel rods in the core.

#### 15.4.3.4 <u>References</u>

- 1. Haessler, R. L., <u>et al.</u>, "Methodology For the Analysis of the Dropped Rod Event," <u>WCAP-11394-P-A</u> (Proprietary) and <u>WCAP-11395-A</u> (Nonproprietary), January 1990.
- 2. Burnett, T. W. T., <u>et al.</u>, "LOFTRAN Code Description," <u>WCAP-7907-P-A</u> (Proprietary) and <u>WCAP-7907-A</u> (Nonproprietary), April 1984.

## 15.4.4 STARTUP OF AN INACTIVE REACTOR COOLANT PUMP AT AN INCORRECT TEMPERATURE

#### 15.4.4.1 Identification of Causes and Accident Description

Technical Specification 3.4.4 does not permit VEGP Units 1 and 2 operation in Modes 1 and 2 with less than four loops operating; however, this analysis assumes approximately 75-percent power in Mode 1 in order to bound Mode 3 operation where the Technical Specifications permit operation with less than four loops.

If the plant operates with one reactor coolant pump (RCP) out of service, there is reverse flow through the inactive loop due to the pressure difference across the reactor vessel. The cold leg temperature of the inactive loop is identical to the cold leg temperature of the active loops. If the reactor is operated at power, and assuming there is no isolation of the secondary side of the steam generator in the inactive loop, there is a temperature drop across the steam generator in the inactive loop, there is a temperature of the inactive loop is lower than the reactor core inlet temperature.

Administrative procedures require that the unit be brought to a load of less than 25 percent of full power prior to starting the pump in an inactive loop in order to bring the inactive loop hot leg temperature closer to the core inlet temperature. Starting an idle RCP without bringing the inactive loop hot leg temperature close to the core inlet temperature would result in the injection of cold water into the core, which would cause a reactivity insertion and subsequent power increase.

If the startup of an inactive RCP accident occurs, the transient terminates automatically by a reactor trip on low coolant loop flow when the power range neutron flux (two out of four channels) exceeds the P-8 setpoint, which has been previously reset for three-loop operation.

This is an ANS Condition II incident.

#### 15.4.4.2 Analysis of Effects and Consequences

#### 15.4.4.2.1 Method of Analysis

The analysis of this transient uses three digital computer codes. The LOFTRAN code (reference 1) calculates the loop and core flow, nuclear power, and core pressure and temperature transients following the startup of an idle pump. FACTRAN (reference 2) calculates the core heat flux transient based on core flow and nuclear power from LOFTRAN. The THINC code is then used to calculate the DNBR during the transient based on system conditions calculated by LOFTRAN and heat fluxes calculated by FACTRAN.

Subsection 15.0.3 discusses plant characteristics and initial conditions. In order to obtain conservative results for the startup of an inactive pump accident, the following assumptions are made (this analysis employed STDP):

- A. Initial conditions of maximum core power and reactor coolant average temperatures and minimum reactor coolant pressure resulting in minimum initial margin to DNB. These values are consistent with maximum steady-state power level that would be permitted with three loops in operation. The high initial power gives the greatest temperature difference between the core inlet temperature and the inactive loop hot leg temperature.
- B. Following initiation of startup of the idle pump, the inactive loop flow reverses and accelerates to its nominal full-flow value in approximately 9 s.
- C. The analysis assumes a conservatively large negative moderator temperature coefficient.
- D. The analysis assumes a least-negative Doppler-only power coefficient.
- E. The initial reactor coolant loop flows are at the appropriate values for one pump out of service.
- F. The reactor trip occurs on low coolant flow when the power range neutron flux exceeds the P-8 setpoint. The P-8 setpoint is conservatively assumed to be 84 percent of rated power, which corresponds to the nominal setpoint plus 9 percent for nuclear instrumentation errors.

Plant systems and equipment which are available to mitigate the effects of the accident are discussed in subsection 15.0.8 and listed in table 15.0.8-1. No single active failure in any of these systems or equipment will adversely affect the consequences of the accident.

## 15.4.4.2.2 Results

The results following the startup of an idle pump with the above listed assumptions are shown in figures 15.4.4-1 through 15.4.4-5. These curves show that during the first part of the transient, the increase in core flow with cooler water results in an increase in nuclear power and a decrease in the core average temperature. The minimum DNBR during the transient is greater than the safety analysis limit values.

Reactivity addition for the inactive loop startup accident is due to the decrease in core water temperature. During the transient, this decrease is due both to the increase in reactor coolant flow and, as the inactive loop flow reverses, to the colder water entering the core from the hot leg side of the steam generator in the inactive loop. Thus, the reactivity insertion rate for this transient changes with time. The resultant core nuclear power transient, computed with consideration of both moderator and Doppler reactivity feedback effects, is shown in figure 15.4.4-1. The calculated sequence of events for this accident is shown in table 15.4.1-1. The transient results illustrated in figures 15.4.4-1 through 15.4.4-5 indicate that a stabilized plant condition, with the reactor tripped, is rapidly approached. By following normal shutdown procedures, the plant can subsequently achieve cooldown.

#### 15.4.4.3 <u>Conclusions</u>

The transient results show that the core is not adversely affected. There is considerable margin to the safety analysis limit DNBRs; thus, the DNB design basis as described in section 4.4 is met.

#### 15.4.4.4 <u>References</u>

- 1. Burnett, T. W. T., <u>et al.</u>, "LOFTRAN Code Description," <u>WCAP-7907-P-A</u>, (Proprietary) and <u>WCAP-7907-A</u> (Nonproprietary), April 1984.
- 2. Hargrove, H. G., "FACTRAN--A FORTRAN-IV Code for Thermal Transients in UO<sub>2</sub> Fuel Rod," <u>WCAP-7908-A</u>, December 1989.

#### 15.4.5 A MALFUNCTION OR FAILURE OF THE FLOW CONTROLLER IN A BOILING WATER REACTOR LOOP THAT RESULTS IN AN INCREASED REACTOR COOLANT FLOWRATE

This subsection is not applicable to the VEGP.

#### 15.4.6 CHEMICAL AND VOLUME CONTROL SYSTEM MALFUNCTION THAT RESULTS IN A DECREASE IN THE BORON CONCENTRATION IN THE REACTOR COOLANT

#### 15.4.6.1 Identification of Causes and Accident Description

Feeding primary grade water into the RCS via the reactor makeup portion of the chemical and volume control (CVCS) adds reactivity to the core. Boron dilution is a manual operation under strict administrative controls with procedures calling for a limit on the rate and duration of dilution. A boric acid blend system permits the operator to match the boron concentration of reactor coolant makeup water during normal charging to that in the RCS. Even under various postulated failure modes, the design of the CVCS limits the potential rate of dilution to a value which gives the operator sufficient time to correct the situation in a safe and orderly manner.

The opening of the primary water makeup control valve supplies water to the RCS which can dilute the reactor coolant. Inadvertent dilution can be readily terminated by closing one of the valves in the makeup pathway. In order to add makeup water to the RCS at pressure, at least one charging pump in addition to the primary makeup water pumps must be running. Normally, only one primary water supply pump is operating while the other is on standby.

The boric acid from the boric acid tank blends with primary grade water at the mixing tee, and the preset flowrates of boric acid and primary grade water on the control board determine the composition.

Information on the status of reactor coolant makeup is continuously available to the operator. Lights on the control board indicate the operating condition of pumps in the CVCS. Alarms actuate to warn the operator if boric acid or demineralized water flowrates deviate from preset values as a result of system malfunction.

This is an ANS Condition II incident.

#### 15.4.6.2 Analysis of Effects and Consequences

#### 15.4.6.2.1 Method of Analysis

To cover all phases of the plant operation, this analysis considers boron dilution during refueling, cold shutdown, hot shutdown, hot standby, startup, and power operation. The analysis assumes conservative values for the critical parameters; i.e., high RCS critical boron concentrations, most negative boron worths, minimum shutdown margins, and small RCS volumes. These result in conservative calculations of the time available for the operator to determine the cause of the addition and take corrective action before shutdown margin is lost.

A. Dilution During Refueling

This analysis evaluates boron dilution events during refueling (Mode 6). During refueling, a very small amount of unborated chemical solution is allowed to enter the RCS for water chemistry quality control. The dilution flow path is provided by opening CVCS valves 176 and 177. The maximum flowrate possible through this flow path is less than 3.5 gal/min which is approximately 3.0 percent of the limiting flowrate considered in the analysis for Modes 3, 4, and 5a. At all other times during Mode 6, valves 176 and 177 will be locked closed. Any other chemical makeup solution which is required during refueling will be borated water supplied from the refueling water storage tank by the RHR pumps.

Valves 175 and 183 in the CVCS will be locked closed or isolated by removal of control air or electrical supply during refueling operations (Mode 6). These valves will block additional flow paths which could allow unborated chemical makeup water in excess of 3.5 gal/min to reach the RCS.

B. Dilution During Cold Shutdown, Hot Shutdown, and Hot Standby

This analysis evaluates boron dilution events during cold shutdown with the RCS in the "loop filled" condition (Mode 5a), cold shutdown with the RCS in the "loop not filled" condition (midloop operation, Mode 5b), hot shutdown (Mode 4), and hot standby (Mode 3). Failure modes and effects analysis, human error analysis, and event tree analysis were used to identify credible boron dilution initiators and to evaluate the plant response to these events. For the initiators identified, time intervals from alarm to loss of shutdown margin were calculated to determine the length of time available for operator response. These calculations depended on dilution flowrates, boron concentrations, and reactor coolant system volumes specific to the event and mode of operation. The technique modeled realistic plant conditions and responses, including both mechanical failure and human errors.

The analysis identified four events considered to be the most likely initiators:

- 1. Demineralizer outlet isolation valve open during resin flushing.
- 2. Valve 226 open following BTRS demineralizer flushing operation.
- 3. Failure to secure chemical addition.
- 4. Boric acid flow control valve (FV-110A) fails closed during makeup.

Initiator 4 was found to be the most limiting event for Modes 3, 4, and 5a. For Mode 5b, initiator 3 was considered to allow the addition of small amounts of unborated chemical solution into the RCS for water chemistry control. The maximum flowrate possible through this flow path is

approximately 3 percent of that associated with the limiting flow path for Modes 3, 4, and 5a. The parameters used in the calculation of time available for operator response are listed in table 15.4.6-1. Conservative values of boron worth (pcm/ppm), as a function of RCS boron concentration, were assumed in the analysis.

Since the active volumes considered are so small in Mode 5b, it was determined that the same valves locked closed in refueling (valves 175 and 183, and, except when required for small chemical additions, valves 176 and 177) would need to be locked closed in Mode 5b. (See paragraph A.)

C. Dilution During Full Power Operation, Including Startup. For the dilution during startup (Mode 2), the analysis assumes an initial maximum critical boron concentration of 2100 ppm based on the rods being inserted to the insertion limits. The analysis assumes the minimum change in the boron concentration from this initial condition to a hot zero power critical condition to be 300 ppm. The analysis also assumes full rod insertion to occur due to reactor trip, minus the most reactive stuck rod. The analysis assumes the dilution flow to be the combined capacity of the two primary water makeup pumps (approximately 242 gal/min) and a minimum RCS water volume of 9583 ft<sup>3</sup>. This volume corresponds to the active volume of the RCS minus the pressurizer and accounts for 10-percent steam generator tube plugging.

During power operation (mode 1), the plant operates under either manual or automatic rod control. While the plant is in manual control, the analysis assumes the dilution flow to be a maximum of 242 gal/min, which is the combined capacity of the two primary water makeup pumps. While in automatic control, the maximum letdown flow (approximately 130 gal/min) limits the dilution flow. The analysis assumes an initial maximum critical boron concentration, corresponding to the rods inserted to the insertion limits at hot full power, of 2100 ppm. The analysis also assumes the minimum change in the boron concentration from this initial condition to a hot zero power critical condition to be 300 ppm. The analysis assumes full rod insertion to occur due to reactor trip, minus the most reactive stuck rod. The analysis uses a minimum water volume of 9583 ft<sup>3</sup> in the RCS, corresponding to the active volume of the RCS minus the pressurizer volume and accounts for 10-percent tube plugging.

No single active failure in any plant systems or equipment will adversely affect the consequences of the accident.

## 15.4.6.2.2 Results

The calculated sequence of events is shown in table 15.4.1-1.

A. Dilution During Refueling

Since the maximum flowrate associated with the available dilution flow paths in mode 6 is very small, the total time from initiation of event to the eventual complete loss of shutdown margin is significantly large compared to the minimum required operator action time. Therefore, a considerable amount of time is available for the operator to initiate and terminate procedures for RCS water chemistry adjustments before potential loss of shutdown becomes a concern. Additionally, assuming the availability of one high flux at shutdown (HFAS) alarm set at not more than 2.3 times background, it is shown that the technical specification shutdown margin requirement for mode 6 is sufficient to ensure that the operator has 30 minutes from the time of alarm to terminate the dilution before shutdown margin is lost.

B. Dilution During Cold Shutdown, Hot Shutdown, and Hot Standby

For dilution during cold shutdown, hot shutdown, and hot standby, the Core Operating Limits Report (COLR) provides the required shutdown margin as a function of RCS boron concentration. The specified shutdown margin ensures that the operator has 15 min from the time of the HFAS alarm to the total loss of shutdown margin due to initiator 4, which is the limiting case for Modes 3, 4, and 5a. Since the maximum flowrate associated with the available dilution flow paths in Mode 5b is very small, the total time from initiation of event to the eventual complete loss of shutdown margin is significantly large compared to the minimum required operator action time of 15 minutes.

C. Dilution During Full Power Operation, Including Startup

In the event of an unplanned approach to criticality or dilution during power escalation while in the startup mode, the operator is alerted to an unplanned dilution by a reactor trip at the power range neutron flux high, low setpoint. After reactor trip there are at least 15 minutes for operator action prior to loss of shutdown margin.

During full power operation with the reactor in manual control, the operator is alerted to an uncontrolled dilution by an  $OT\Delta T$  reactor trip. At least 15 minutes are available after the trip for operator action prior to loss of shutdown margin.

During full power operation with the reactor in automatic control, the operator is alerted to an uncontrolled reactivity insertion by the rod insertion limit alarms. At least 15 minutes are available for operator action from the low-low rod insertion limit alarm until a loss of shutdown margin occurs.

## 15.4.6.3 <u>Conclusions</u>

The results presented above show that adequate time is available for the operator to manually terminate the source of dilution flow. Following termination of the dilution flow, the operator can initiate boration to establish adequate shutdown margin.

# 15.4.7 INADVERTENT LOADING AND OPERATION OF A FUEL ASSEMBLY IN AN IMPROPER POSITION

#### 15.4.7.1 Identification of Causes and Accident Description

Fuel- and core-loading errors such as can arise from the inadvertent loading of one or more fuel assemblies into improper positions, loading a fuel rod during manufacture with one or more pellets of the wrong enrichment, or loading a full fuel assembly during manufacture with pellets of the wrong enrichment, will lead to increased heat fluxes if the error results in placing fuel in core positions calling for fuel of lesser enrichment. Also included among possible core-loading errors is the inadvertent loading of one or more fuel assemblies requiring burnable poison rods into a new core without burnable poison rods.

Any error in enrichment, beyond the normal manufacturing tolerances, can cause power shapes which are more peaked than those calculated with the correct enrichments. There is a 5-percent uncertainty margin included in the design value of power peaking factor assumed in the analysis of Condition I and Condition II transients. The in-core system of movable flux detectors

which is used to verify power shapes at the start of life is capable of revealing any assembly enrichment error or loading error which causes power shapes to be peaked in excess of the design value.

To reduce the probability of core loading errors, each fuel assembly is marked with an identification number and loaded in accordance with a core-loading diagram. During core loading, the identification number will be checked before each assembly is moved into the core. Serial numbers read during fuel movement are subsequently recorded on the loading diagram as a further check on proper placing after the loading is completed.

The power distortion due to any combination of misplaced fuel assemblies would significantly raise peaking factors and would be readily observable with in-core flux monitors. In addition to the flux monitors, thermocouples are located at the outlet of about one-third of the fuel assemblies in the core. There is a high probability that these thermocouples would also indicate any abnormally high coolant temperature rise. In-core flux measurements are taken during the startup subsequent to every refueling operation.

This event is classified as an American Nuclear Society Condition III incident (an infrequent fault) as defined in subsection 15.0.1.

## 15.4.7.2 Analysis of Effects and Consequences

### 15.4.7.2.1 Method of Analysis

Steady-state power distributions in the x-y plane of the core are calculated using the TURTLE code<sup>(1)</sup> based on macroscopic cross section calculated by the LEOPARD code.<sup>(2)</sup> A discrete representation is used wherein each individual fuel rod is described by a mesh interval. Representative power distributions in the x-y plane for a correctly loaded core assembly are also given in chapter 4.

For each core loading error case analyzed, the percent deviations from detector readings for a normally loaded core are shown in all in-core detector locations. (See figures 15.4.7-1 through 15.4.7-5.)

#### 15.4.7.2.2 Results

The following core loading error cases have been analyzed:

Case A:

Case in which a region 1 assembly is interchanged with a region 3 assembly. The particular case considered was the interchange to two adjacent assemblies near the periphery of the core. (See figure 15.4.7-1.)

Case B:

Case in which a region 1 assembly is interchanged with a neighboring region 2 fuel assembly. Two analyses have been performed for this case. (See figures 15.4.7-2 and 15.4.7-3.)

In Case B-1, the interchange is assumed to take place with the burnable poison rods transferred with the region 2 assembly mistakenly loaded into region 1.

In Case B-2, the interchange is assumed to take place closer to core center and with burnable poison rods located in the correct region 2 position but in a region 1 assembly mistakenly loaded in the region 2 position.

Case C:

Enrichment error: Case in which a region 2 fuel assembly is loaded in the core central position. (See figure 15.4.7-4.)

Case D:

Case in which a region 2 fuel assembly instead of a region 1 assembly is loaded near the core periphery. (See figure 15.4.7-5.)

## 15.4.7.3 <u>Conclusions</u>

Fuel assembly enrichment errors would be prevented by administrative procedures implemented in fabrication.

In the event that a single pin or pellet has a higher enrichment than the nominal value, the consequences in terms of reduced departure from nucleate boiling ratio and increased fuel and clad temperatures will be limited to the incorrectly loaded pin or pins and perhaps the immediately adjacent pins.

Fuel assembly loading errors are prevented by administrative procedures implemented during core loading. In the unlikely event that a loading error occurs, analyses in this section confirm that resulting power distribution effects will either be readily detected by the in-core movable detector system or will cause a sufficiently small perturbation to be acceptable within the uncertainties allowed between nominal and design power shapes.

#### 15.4.7.4 <u>References</u>

- 1. Barry, R. F., and Altomore, A., "The TURTLE 24.0 Diffusion Depletion Code," <u>WCAP-7213-P-A</u> (Proprietary) and <u>WCAP-7758-A</u> (Nonproprietary), February 1975.
- 2. Barry, R. F., "LEOPARD A Spectrum Dependent Non-Spacial Depletion Code for the IBM-7094," <u>WCAP-3269-26</u>, September 1963.

# 15.4.8 SPECTRUM OF ROD CLUSTER CONTROL ASSEMBLY EJECTION ACCIDENTS

#### 15.4.8.1 Identification of Causes and Accident Description

This accident is defined as the mechanical failure of a control rod mechanism pressure housing, resulting in the ejection of a rod cluster control assembly (RCCA) and drive shaft. The consequence of this mechanical failure is a rapid positive reactivity insertion together with an adverse core power distribution, possibly leading to localized fuel rod damage.

## 15.4.8.1.1 Design Precautions and Protection

Certain features in the VEGP reactors are intended to preclude the possibility of a rod ejection accident or to limit the consequences if the accident occurs. These include a sound, conservative mechanical design of the rod housings, together with a thorough quality control (testing) program during assembly, and a nuclear design which lessens the potential ejection worth of RCCAs and minimizes the number of assemblies inserted at high-power levels.

15.4.8.1.1.1 <u>Mechanical Design</u>. The mechanical design is discussed in section 4.6. Mechanical design and quality control procedures intended to preclude the possibility of an RCCA drive mechanism housing failure are listed below:

- A. Each control rod drive mechanism housing is completely assembled and shop tested at 4100 psi.
- B. The mechanism housings are individually hydrotested after they are attached to the head adapters in the reactor vessel head and checked during the hydrotest of the completed reactor coolant system (RCS).
- C. Stress levels in the mechanism are not affected by anticipated system transients at power or by the thermal movement of the coolant loops. Moments induced by the design earthquake can be accepted within the allowable primary working stress range specified by the American Society of Mechanical Engineers Code, Section III, for Class 1 components.
- D. The latch mechanism housing and rod travel housing are each a single length of forged type 304 stainless steel. This material exhibits excellent notch toughness at all temperatures which will be encountered.

A significant margin of strength in the elastic range together with the large energy absorption capability in the plastic range gives additional assurance that gross failure of the housing will not occur. The joints between the latch mechanism housing and head adapter, and between the latch mechanism housing and rod travel housing, are threaded joints reinforced by canopy-type rod welds which are subject to periodic inspections.

15.4.8.1.1.2 <u>Nuclear Design</u>. Even if a rupture of an RCCA drive mechanism housing is postulated, the operation utilizing chemical shim is such that the severity of an ejected RCCA is inherently limited. In general, the reactor is operated with the RCCA inserted only far enough to permit load follow. Reactivity changes caused by core depletion and xenon transients are compensated for by boron changes. Further, the location and grouping of control RCCA banks are selected during the nuclear design to lessen the severity of an RCCA ejection accident. Therefore, should an RCCA be ejected from its normal position during full-power operation, only a minor reactivity excursion, at worst, could be expected to occur.

However, it may be occasionally desirable to operate with larger than normal insertions. For this reason, a rod insertion limit is defined as a function of power level. Operation with the RCCAs above this limit guarantees adequate shutdown capability and acceptable power distribution. The position of all RCCAs is continuously indicated in the control room. An alarm will occur if a bank of RCCAs approaches its insertion limit or if one RCCA deviates from its bank. Procedures require action per the Technical Specifications if shutdown or control RCCA banks are below insertion limits.

15.4.8.1.1.3 <u>Reactor Protection</u>. The reactor protection in the event of a rod ejection accident has been described in reference 1. The protection for this accident is provided by high neutron flux trip (high and low setting) and high rate of neutron flux increase trip. These protection functions are described in detail in section 7.2.

15.4.8.1.1.4 <u>Effects on Adjacent Housings</u>. Disregarding the remote possibility of the occurrence of an RCCA mechanism housing failure, investigations have shown that failure of a housing due to either longitudinal or circumferential cracking would not cause damage to adjacent housings. The control rod drive mechanism is described in paragraph 3.9.4.1.1.

15.4.8.1.1.5 <u>Effects of Rod Travel Housing Longitudinal Failures</u>. If a longitudinal failure of the rod travel housing should occur, the region of the position indicator assembly opposite the break would be stressed by the reactor coolant pressure of 2250 psia. The most probable leakage path would be provided by the radial deformation of the position indicator coil assembly, resulting in the growth of axial flow passages between the rod travel housing and the hollow tube along which the coil assemblies are mounted.

If failure of the position indicator coil assembly should occur, the resulting free radial jet from the failed housing could cause it to bend and contact adjacent rod housings. If the adjacent housings were on the periphery, they might bend outward from their bases. The housing material is quite ductile; plastic hinging without cracking would be expected. Housings adjacent to a failed housing, in locations other than the periphery, would not be bent because of the rigidity of multiple adjacent housings.

15.4.8.1.1.6 Effect of Rod Travel Housing Circumferential Failures. If circumferential failure of a rod travel housing should occur, the broken-off section of the housing would be ejected vertically because the driving force is vertical and the position indicator coil assembly and the drive shaft would tend to guide the broken-off piece upwards during its travel. Travel is limited by the missile shield, thereby limiting the projectile acceleration. When the projectile reached the missile shield, it would partially penetrate the shield and dissipate its kinetic energy. The water jet from the break would continue to push the broken-off piece against the missile shield.

If the broken-off piece of the rod travel housing were short enough to clear the break when fully ejected, it would rebound after impact with the missile shield. The top end plates of the position indicator coil assemblies would prevent the broken piece from directly hitting the rod travel housing of a second drive mechanism. Even if a direct hit by the rebounding piece were to occur, the low kinetic energy of the rebounding projectile would not be expected to cause significant damage (sufficient to cause failure of an adjacent housing).

15.4.8.1.1.7 <u>Possible Consequences</u>. From the above discussion, the probability of damage to an adjacent housing must be considered remote. However, even if damage is postulated, it would not be expected to lead to a more severe transient since RCCAs are inserted in the core in symmetric patterns, and control rods immediately adjacent to worst ejected rods are not in the core when the reactor is critical. Damage to an adjacent housing could, at worst, cause that RCCA not to fall on receiving a trip signal; however, this is already taken into account in the analysis by assuming a stuck rod adjacent to the ejected rod.

15.4.8.1.1.8 <u>Summary</u>. The considerations given above lead to the conclusion that failure of a control rod housing, due either to longitudinal or circumferential cracking, would not cause damage to adjacent housings that would increase severity of the initial accident.

## 15.4.8.1.2 Limiting Criteria

This event is classified as an American Nuclear Society (ANS) Condition IV incident. See subsection 15.0.1 for a discussion of ANS classifications. Due to the extremely low probability of an RCCA ejection accident, some fuel damage would be considered an acceptable consequence.

Comprehensive studies of the threshold of fuel failure and of the threshold of significant conversion of the fuel thermal energy to mechanical energy have been carried out as part of the SPERT project by the Idaho Nuclear Corporation.<sup>(2)</sup> Extensive tests of UO<sub>2</sub> zirconium-clad fuel rods representative of those in pressurized water reactor-type cores such as VEGP have demonstrated failure thresholds in the range of 240 to 257 cal/g. However, other rods of a slightly different design have exhibited failure as low as 225 cal/g. These results differ significantly from the TREAT<sup>(3)</sup> results, which indicated a failure threshold of 280 cal/g. Limited results have indicated that this threshold decreases by about 10 percent with fuel burnup. The clad failure mechanism appears to be melting for zero burnup rods and brittle fracture for irradiated rods. Also important is the conversion ratio of thermal to mechanical energy. This ratio becomes marginally detectable above 300 cal/g for unirradiated rods and 200 cal/g for irradiated rods; catastrophic failure (large fuel dispersal, large pressure rise), even for irradiated rods, did not occur below 300 cal/g.

In view of the above experimental results and conformance with Regulatory Guide 1.77 (subsection 1.9.77), criteria are applied to ensure that there is little or no possibility of fuel dispersal in the coolant, gross lattice distortion, or severe shock waves. These limiting criteria are as follows (reference 9):

- A. Average fuel pellet enthalpy at the hot spot will be below 200 cal/g for unirradiated fuel and irradiated fuel.
- B. Peak reactor coolant pressure will be less than that which could cause stresses to exceed the faulted condition stress limits.
- C. Fuel melting will be limited to less than 10 percent of the fuel volume at the hot spot even if the average fuel pellet enthalpy is below the limits of criterion A, above.

## 15.4.8.2 Analysis of Effects and Consequences

A. Method of Analysis

The calculation of the RCCA ejection transient is performed in two stages, first an average core channel calculation and then, a hot region calculation. The average core calculation is performed using spatial neutron kinetics methods to determine the average power generation with time including the various total core feedback effects; i.e., Doppler reactivity and moderator reactivity. Enthalpy and temperature transients at the hot spot are then determined by multiplying the average core energy generation by the hot channel factor and performing a fuel rod transient heat transfer calculation. The power distribution calculated without feedback is conservatively assumed to persist throughout the transient.

A detailed discussion of the method of analysis can be found in reference 1.

B. Average Core Analysis

The spatial kinetics computer code, TWINKLE,<sup>(4)</sup> is used for the average core transient analysis. This code uses cross sections generated by LEOPARD<sup>(5)</sup> to solve the two-group neutron diffusion theory kinetic equation in one, two, or three spatial dimensions (rectangular coordinates) for six delayed neutron groups and up to 2000 spatial points. The computer code includes a detailed multiregion, transient fuel clad coolant heat transfer model for calculation of pointwise Doppler and moderator feedback effects. In this analysis, the code is used as a one-dimensional axial kinetics code, since it allows a more realistic representation of the spatial effects of axial moderator feedback and RCCA movement. However, since the radial dimension is missing, it is still necessary to employ conservative methods (described below) of calculating the ejected rod worth and hot channel factor. Further description of TWINKLE appears in subsection 15.0.11.

C. Hot Spot Analysis

In the hot spot analysis, the initial heat flux is equal to the nominal volume multiplied by the design hot channel factor. During the transient, the heat flux hot channel factor is linearly increased to the transient value in 0.1 s, the time for full ejection of the rod. Therefore, the assumption is made that the hot spots before and after ejection are coincident. This is conservative, since the peak after ejection will occur in or adjacent to the assembly with the ejected rod, and prior to ejection, the power in this region will be depressed due to the inserted rod.

The hot spot analysis is performed using the detailed fuel and clad transient heat transfer computer code, FACTRAN.<sup>(6)</sup> This computer code calculates the transient temperature distribution in a cross section of a metal clad UO<sub>2</sub> fuel rod and the heat flux at the surface of the rod, using as input the nuclear power versus time and the local coolant conditions. The zirconium-water reaction is explicitly represented, and all material properties are represented as functions of temperature. A conservative radial power distribution is used within the fuel rod.

FACTRAN uses the Dittus-Boelter or Jens-Lottes correlation to determine the film heat transfer before departure from nucleate boiling (DNB) and the Bishop-Sandburg-Tong correlation<sup>(7)</sup> to determine the film boiling coefficient after DNB. The Bishop-Sandburg-Tong correlation is conservatively used, assuming zero-bulk fluid quality. The departure from nucleate boiling ratio is not calculated; instead, the code is forced into DNB by specifying a conservative DNB heat flux. The gap heat transfer coefficient can be calculated by the code; however, it is adjusted in order to force the full-power, steady-state temperature distribution to agree with the fuel heat transfer design codes. Further description of FACTRAN appears in subsection 15.0.11.

D. System Overpressure Analysis

Because safety limits for fuel damage specified earlier are not exceeded, there is little likelihood of fuel dispersal into the coolant. The pressure surge may therefore be calculated on the basis of conventional heat transfer from the fuel and prompt heat absorption by the coolant.

The pressure surge is calculated by first performing the fuel heat transfer calculation to determine the average and hot spot heat flux versus time. Using
this heat flux data, a core thermal-hydraulic calculation is conducted to determine the volume surge. Finally, the volume surge is simulated using the LOFTRAN computer code. This code calculates the pressure transient taking into account fluid transport in the RCS and heat transfer to the steam generators. No credit is taken for the possible pressure reduction caused by the assumed failure of the control rod pressure housing.

### 15.4.8.2.1 Calculation of Basic Parameters

Input parameters for the analysis are conservatively selected on the basis of values calculated for this type of core. The more important parameters are discussed below. Table 15.4.8-1 presents the parameters used in this analysis.

15.4.8.2.1.1 Ejected Rod Worths and Hot Channel Factors. The values for ejected rod worths and hot channel factors are calculated using either three-dimensional static methods or by a synthesis method employing one-dimensional and two-dimensional calculations. Standard nuclear design codes are used in the analysis. No credit is taken for the flux flattening effects of reactivity feedback. The calculation is performed for the maximum allowed bank insertion at a given power level, as determined by the rod insertion limits. Adverse xenon distributions are considered in the calculation.

Appropriate margins are added to the ejected rod worth and hot channel factors to account for any calculational uncertainties, including an allowance for nuclear peaking due to densification.

Power distributions before and after ejection for a worst case can be found in reference 1. During initial plant startup physics testing, ejected rod worths and power distributions are measured in the zero-power and part power configurations and compared to values used in the analysis. Experience shows that the ejected rod worth and power peaking factors are consistently overpredicted in the analysis.

Reactivity Feedback Weighting Factors. The largest temperature rises and 15.4.8.2.1.2 hence the largest reactivity feedbacks occur in channels where the power is higher than average. Since the weight of a region is dependent on flux, these regions have high weights. This means that the reactivity feedback is larger than that indicated by a simple channel analysis. Physics calculations have been performed for temperature changes with a flat temperature distribution and with a large number of axial and radial temperature distributions. Reactivity changes were compared and effective weighting factors determined. These weighting factors take the form of multipliers which, when applied to single-channel feedbacks, correct them to effective whole-core feedbacks for the appropriate flux shape. In this analysis, since a one-dimensional (axial) spatial kinetics method is employed, axial weighting is not necessary if the initial condition is made to match the ejected rod configuration. In addition, no weighting is applied to the moderator feedback. A conservative radial weighting factor is applied to the transient fuel temperature to obtain an effective fuel temperature as a function of time accounting for the missing spatial dimension. These weighting factors have also been shown to be conservative compared to three-dimensional analysis.<sup>(1)</sup>

15.4.8.2.1.3 <u>Moderator and Doppler Coefficient</u>. The critical boron concentrations at the beginning of life and end of life are adjusted in the nuclear code in order to obtain moderator

density coefficient curves which are conservative compared to actual design conditions for the plant. As discussed above, no weighting factor is applied to these results.

The Doppler reactivity defect is determined as a function of power level using a onedimensional, steady-state computer code with a Doppler weighting factor of 1.0. The Doppler defect used is given in subsection 15.0.4. The Doppler weighting factor will increase under accident conditions, as discussed above.

15.4.8.2.1.4 Delayed Neutron Fraction,  $\beta_{eff}$ . Calculations of the effective delayed neutron fraction ( $\beta_{eff}$ ) typically yield values no less than 0.70 percent at beginning of life and 0.50 percent at end of life for the first cycle. The accident is sensitive to  $\beta_{eff}$  if the ejected rod worth is equal to or greater than  $\beta_{eff}$  as in zero-power transients. To allow for future cycles, the analysis used conservative  $\beta_{eff}$  estimates of 0.54 percent at beginning-of-life hot zero power, 0.57 percent at beginning-of-life hot full power, and 0.46 percent for both end-of-life cases.

15.4.8.2.1.5 <u>Trip Reactivity Insertion</u>. The trip reactivity insertion assumed is given in table 15.4.8-1 and includes the effect of one stuck RCCA adjacent to the ejected rod. These values are reduced by the ejected rod reactivity. The shutdown reactivity was simulated by dropping a rod of the required worth into the core. The start of rod motion occurred 0.5 s after the high neutron flux trip point was reached. This delay is assumed to consist of 0.2 s for the instrument channel to produce a signal, 0.15 s for the trip breaker to open, and 0.15 s for the coil to release the rods. A curve of trip rod insertion versus time was used which assumed that insertion to the dashpot does not occur until 3.3 s after the start of fall. The choice of such a conservative insertion rate means that there is over 1 s after the trip point is reached before significant shutdown reactivity is inserted into the core. This is particularly important conservatism for hot full-power (HFP) accidents.

The minimum design shutdown margin available for this plant at hot zero power (HZP) may be reached only at end of life in the equilibrium cycle. This value includes an allowance for the worst stuck rod, adverse xenon distribution, conservative Doppler and moderator defects, and an allowance for calculational uncertainties. Physics calculations for this plant have shown that the effect of two stuck RCCAs (one of which is the worst ejected rod) is to reduce the shutdown by about an additional 1-percent k/k. Therefore, following a reactor trip resulting from an RCCA ejection accident, the reactor will be subcritical when the core returns to HZP.

Depressurization calculations have been performed for a typical four-loop plant, assuming the maximum possible size break (2.75-in. diameter) located in the reactor pressure vessel head. The results show a rapid pressure drop and a decrease in system water mass due to the break. The safety injection system is actuated on low pressurizer pressure within 1 min after the break. The RCS pressure continues to drop and reaches saturation (1200 psi) in about 2 to 3 min. Due to the large thermal inertia of primary and secondary systems, there has been no significant decrease in the RCS temperature below no-load by this time, and the depressurization itself has caused an increase in shutdown margin by about 0.2-percent  $\Delta k$  due to the pressure coefficient. The cooldown transient could not absorb the available shutdown margin until more than 10 min after the break. The addition of borated (2400 ppm) safety injection flow starting 1 min after the break is much more than sufficient to ensure that the core remains subcritical during the cooldown.

15.4.8.2.1.6 <u>Reactor Protection</u>. As discussed in paragraph 15.4.8.1.1.3, reactor protection for a rod ejection is provided by high neutron flux trip (high and low setting) and high positive rate of neutron flux increase trip; however, the analysis only models the high neutron flux trip. These protection functions are part of the reactor trip system. No single failure of the reactor trip system will negate the protection functions required for the rod ejection accident or adversely affect the consequences of the accident.

No single active failure in any plant systems or equipment will adversely affect the consequences of the accident.

15.4.8.2.1.7 <u>Results</u>. Cases are presented for both beginning and end of life at zero and full power.

A. Beginning of Cycle, Full Power

Control bank D was assumed to be inserted to its insertion limit. The worst ejected rod worth and hot channel factor were conservatively calculated to be 0.24-percent  $\Delta k/k$  and 5.5, respectively. The peak hot spot fuel centerline temperature reached melting at 4900°F. However, melting was restricted to less than 10 percent of the pellet volume at the hot spot.

B. Beginning of Cycle, Zero Power

For this condition, control bank D was assumed to be fully inserted and banks B and C were at their insertion limits. The worst ejected rod is located in control bank D and has a worth of 0.75-percent  $\Delta k/k$  and a hot channel factor of 11.0. The fuel center temperature was 3985°F.

C. End of Cycle, Full Power

Control bank D was assumed to be inserted to its insertion limit. The ejected rod worth and hot channel factors were conservatively calculated to be 0.25-percent  $\Delta k/k$  and 6.0, respectively.

The peak hot spot fuel centerline temperature reached melting at 4800°F. However, melting was restricted to less than 10 percent of the pellet volume at the hot spot.

D. End of Cycle, Zero Power

The ejected rod worth and hot channel factor for this case were obtained assuming control bank D to be fully inserted with banks C and B at their insertion limits. The results were 0.84-percent  $\Delta k/k$  and 26.0, respectively. The fuel centerline temperature was 3891°F. The Doppler weighting factor for this case is significantly higher than for the other cases due to the very large transient hot channel factor.

For all four cases analyzed, average fuel pellet enthalpy at the hot spot remained below 200 cal/g.

A summary of the cases presented above is given in table 15.4.8-1. The nuclear power and hot spot fuel and clad temperature transients for the worst cases are presented in figures 15.4.8-1 through 15.4.8-4.

The calculated sequence of events for the worst case rod ejection accidents, as shown in figures 15.4.8-1 through 15.4.8-4, is presented in table 15.4.8-1. For all cases, reactor trip

occurs early in the transient after the nuclear power excursion is terminated by Doppler feedback. As discussed previously in paragraph 15.4.8.2.1, the reactor remains subcritical following reactor trip.

The ejection of an RCCA constitutes a break in the RCS, located in the reactor pressure vessel head. The effects and consequences of loss-of-coolant accidents (LOCAs) are discussed in subsection 15.6.5. Following the RCCA ejection, the operator would follow the same emergency instructions as for any other LOCA to recover from the event.

15.4.8.2.1.8 <u>Fission Product Release</u>. It is assumed that fission products are released from the gaps of all rods entering DNB. In all cases considered, less than 10 percent of the rods entered DNB based on a detailed three-dimensional THINC analysis.<sup>(1)</sup> Although limited fuel melting at the hot spot was predicted for the full-power cases, in practice, melting is not expected since the analysis conservatively assumed that the hot spots before and after ejection were coincident.

15.4.8.2.1.9 <u>Pressure Surge</u>. A detailed calculation of the pressure surge for an ejection worth of one dollar at beginning of life, hot full power, indicates that the peak pressure does not exceed that which would cause stress to exceed the faulted condition stress limits.<sup>(2)</sup> Since the severity of the present analysis does not exceed the worst-case analysis, the accident for this plant will not result in an excessive pressure rise or further damage to the RCS.

Lattice Deformations. A large temperature gradient will exist in the region of 15.4.8.2.1.10 the hot spot. Since the fuel rods are free to move in the vertical direction, differential expansion between separate rods cannot produce distortion. However, the temperature gradients across individual rods may produce a differential expansion tending to bow the midpoint of the rods toward the hotter side of the rod. Calculations have indicated that this bowing would result in a negative reactivity effect at the hot spot since Westinghouse cores are undermoderated, and bowing will tend to increase the undermoderation at the hot spot. In practice, no significant bowing is anticipated, since the structural rigidity of the core is more than sufficient to withstand the forces produced. Boiling in the hot spot region would produce a net flow away from that region. However, the heat from the fuel is released to the water relatively slowly, and it is considered inconceivable that crossflow will be sufficient to produce sufficient lattice forces. Even if massive and rapid boiling, sufficient to distort the lattice, is hypothetically postulated, the large void fraction in the hot spot region would produce a reduction in the total core moderator to fuel ratio and a large reduction in this ratio at the hot spot. The net effect would therefore be a negative feedback. It can be concluded that no conceivable mechanism exists for a net positive feedback resulting from lattice deformation. In fact, a small negative feedback may result. The effect is conservatively ignored in the analysis.

### 15.4.8.3 Radiological Consequences

The evaluation of the radiological consequences of a postulated control rod ejection accident assumes that the reactor has been operating with a small percent of defective fuel and leaking generator tubes for sufficient time to establish equilibrium concentrations of radionuclides in the reactor coolant and in the secondary coolant.

As a result of the accident, a fraction of the fuel rods will undergo DNB and will release gap inventory to the reactor coolant. Additionally, a small fraction of fuel is assumed to melt and release core inventory to the reactor coolant. Radionuclides carried by the primary coolant to the steam generators via leaking tubes are released to the environment via the steam line safety or power-operated relief valves. Radionuclides released to the containment via the spill from the reactor vessel head are released to the environment via containment leakage.

### 15.4.8.3.1 Analytical Assumptions

The major assumptions and parameters used in the analysis are itemized in table 15.4.8-2. The following is a more detailed discussion of the source term.

15.4.8.3.1.1 <u>Source Term Calculations</u>. The concentration of nuclides in the primary and secondary system prior to and following the rod ejection accident are determined as follows:

- A. The iodine activity in the reactor coolant prior to the accident is based upon an iodine spike which has raised the reactor coolant concentration to 60  $\mu$ Ci/g of dose equivalent (DE) I-131.
- B. The noble gas concentrations in the reactor coolant are based upon 1-percent defective fuel.
- C. Following the rod ejection accident, 10 percent of the fuel rods in the core undergo DNB. Hence, 10 percent of the core iodine and noble gas gap inventory is released to the reactor coolant. In addition, 0.25 percent of the fuel in the core is assumed to melt and release 0.00125 of the core iodines and 0.0025 of the core noble gases to the reactor coolant.
- D. The secondary coolant iodine activity is based on the DE of 0.1  $\mu$ Ci/g of I-131.

15.4.8.3.1.2 <u>Mathematical Models Used in the Analysis</u>. Mathematical models used in the analysis are described in the following sections:

- A. The mathematical models used to analyze the activity released during the course of the accident are described in appendix 15A.
- B. The atmospheric dispersion factors used in the analysis were calculated based on the onsite meteorological measurement programs described in subsection 2.3.3.
- C. The thyroid inhalation dose and total-body gamma immersion doses to a receptor at the exclusion area boundary and outer boundary of the low population zone were analyzed using the models described in appendix 15A.

15.4.8.3.1.3 <u>Identification of Leakage Pathways and Resultant Leakage Activity</u>. Radionuclides carried from the primary coolant to the steam generators via leaking tubes are released to the environment via the steam line safety or power-operated relief valves. Iodines are assumed to mix with the secondary coolant and partition between the generator liquid and steam before release to the environment. Noble gases are assumed to be directly released. Forty-five percent of the iodines and one hundred percent of the noble gases carried by the primary coolant spill are released to the containment vapor space and are leaked to the environment at the containment design leak rate. For the iodine release, 39 percent of the break flow is assumed to initially flash to vapor and 10 percent of the nonflashed portion is assumed to become airborne; i.e., 0.39 + 10 percent of 0.61 for a total of 0.45.

All activity is released to the environment with no consideration given to radioactive decay or to cloud depletion by ground deposition during transport to the exclusion area boundary and low population zone. Hence, the resultant radiological consequences represent the most conservative estimate of the potential integrated dose due to the postulated rod ejection accident.

### 15.4.8.3.2 Identification of Uncertainties and Conservative Elements in the Analysis

- A. The initial reactor coolant iodine activity is based on the technical specification limit of 1.0  $\mu$ Ci/g of DE I-131 which is further increased by a large preaccident iodine spike to 60  $\mu$ Ci/g, resulting in equivalent concentrations many times greater than the reactor coolant activities based on 0.12-percent defective fuel and expected iodine spiking values associated with normal operating conditions.
- B. The noble gas activities are based on 1-percent defective fuel which cannot exist simultaneously with 1.0- $\mu$ Ci/g I-131. For iodines, 1-percent defects would be approximately three times the technical specification limit.
- C. The fraction of failed fuel is assumed to be equal to the fraction of fuel rods experiencing DNB, without consideration given to the extent of the zirc-water reaction. Based on experimental data<sup>(8)</sup> no oxidation related fuel rod clad failure is predicted. Likewise, the small amount of melted fuel assumed (0.25 percent) is not predicted.
- D. A 1-gal/min steam generator primary-to-secondary leakage is assumed, which is significantly greater than that anticipated during normal operation.
- E. The meteorological conditions which may be present at the site during the course of the accident are uncertain. However, it is highly unlikely that the assumed meteorological conditions would be present during the course of the accident for any extended period of time. Therefore, the radiological consequences evaluated, based on the meteorological conditions assumed, are conservative.

### 15.4.8.3.3 Conclusions

15.4.8.3.3.1 <u>Filter Loadings</u>. The only engineered safety feature filtration system considered in the analysis which limits the consequences of the rod ejection accident is the control room filtration system.

Integrated activity on the control room filters have been evaluated for the more limiting LOCA analysis as discussed in paragraph 15.6.5.4.6. Since the control room filters are capable of accommodating the potential design basis LOCA fission product iodine loadings, there will be sufficient capacity to accommodate any fission product loading due to a postulated rod ejection accident.

15.4.8.3.3.2 <u>Dose to Receptor at the Exclusion Area Boundary and Low Population Zone</u> <u>Outer Boundary</u>. The potential radiological consequences resulting from the occurrence of a postulated rod ejection accident have been conservatively analyzed using assumptions and models described. The total-body gamma dose due to immersion from direct radiation and the thyroid dose due to inhalation have been analyzed for the 0- to 2-h dose at the exclusion area boundary and for the duration of the accident (0 to 30 days) at the low population zone outer boundary. The results are listed in table 15.4.8-3. The resultant doses are well within the guideline values of 10 CFR 100.

### 15.4.8.4 <u>References</u>

- Risher, D. H., Jr., "An Evaluation of the Rod Ejection Accident in Westinghouse Pressurized Water Reactors Using Spatial Kinetics Methods," <u>WCAP-7588</u>, Revision 1A, January 1975.
- 2. Taxelius, T. G., ed, "Annual Report-Spert Project, October 1968, September 1969," Idaho Nuclear Corporation, <u>IN-1370</u>, June 1970.
- 3. Liimataninen, R. C., and Testa, F. J., "Studies in TREAT of Zircaloy-2-Clad, UO<sub>2</sub>-Core Simulated Fuel Elements," <u>ANL-7225</u>, January-June 1966, p 177, November 1966.
- 4. Risher, D. H., Jr., and Barry, R. F., "TWINKLE--A Multi- Dimensional Neutron Kinetics Computer Code," <u>WCAP-7979-P-A (Proprietary)</u> and <u>WCAP-8028-A (Nonproprietary)</u>, January 1975.
- 5. Barry, R. F., "LEOPARD--A Spectrum Dependent or Non-Spacial Depletion Code for the IBM-7904," <u>WCAP-3269-26</u>, September 1963.
- 6. Hargrove, H. G., "FACTRAN-A FORTRAN-IV Code for Thermal Transients in a UO<sub>2</sub> Fuel Rod," <u>WCAP-7908-A</u>, December 1989.
- 7. Bishop, A. A., Sandburg, R. O., and Tong, L. S., "Forced Convection Heat Transfer at High Pressure After the Critical Heat Flux," <u>ASME 65-HT-31</u>, August 1965.
- 8. Van Houten, R., "Fuel Rod Failure as a Consequence of Departure from Nucleate Boiling or Dryout," <u>NUREG-0562</u>, June 1979.
- 9. Johnson, V. J., "Use of 2700°F PCT Acceptance Limit in Non-LOCA Accidents," <u>NS-NRC-89-3466</u>, October 23, 1989.

### 15.4.9 STEAMLINE BREAK WITH COINCIDENTAL ROD CLUSTER CONTROL ASSEMBLY WITHDRAWAL AT POWER

The automatic rod withdrawal capability of the rod control system is disabled for Vogtle Unit 1 and Unit 2. Physically disabling the automatic rod withdrawal capability eliminates the possibility that a steam line break event will result in a consequential and coincidental rod withdrawal. The analysis presented in this section is retained for historical purposes.

### 15.4.9.1 Introduction

The coincidental and consequential occurrence of an uncontrolled RCCA bank withdrawal at power following steamline break event is one of four potential interaction scenarios resulting from adverse environmental conditions (either inside or outside of containment) following a high

energy line break; these scenarios are identified in "IE Information Notice 79-22." The premise of this concern is that during a high energy line break (such as steamline rupture), certain sensors used in the control systems could be exposed to an adverse environment. If the equipment is not qualified for the adverse environment, a control system malfunction might occur.

The automatic rod control system is one of the control systems that could malfunction. The rod control system relies on the measurement of  $T_{avg}$ , nuclear power, and turbine impulse pressure to determine if control rod motion is required. A small steamline rupture may occur outside containment near the turbine impulse pressure transmitters or inside containment in the vicinity of the excore detectors, thus exposing equipment used in the rod control system to an adverse environment. If this equipment is not properly qualified for these conditions, a consequential RCCA withdrawal following a steamline rupture may occur.

The steamline break affects the rod control system (via either an inside containment break near the excore detectors or an outside containment break near the turbine impulse transmitters) and causes the control rods to withdraw following the initiation of the transient. This causes an increase in reactor power and core heat flux to the point at which an OP∆T trip setpoint is reached. This trip terminates the most adverse part of the transient. The steamline break causes increased heat removal and subsequent decrease in primary pressure simultaneous with the increase in reactor power. Secondary pressure also decreases until the low steamline pressure setpoint is reached, initiating steamline isolation and safety injection actuation.

Because of the lower RCS pressure coincident with the increase in reactor power, the consequences at the point of peak heat flux may be more adverse than the RCCA bank withdrawal at power transient analyzed in the FSAR.

The most limiting part of this transient pertinent to this study is immediately before reactor trip (i.e., rod motion). The most limiting case is that for the largest steamline break that trips on  $OP\Delta T$  prior to reaching a reactor trip on a safety injection signal (e.g., low steamline pressure). Therefore, the analysis assumes the largest steamline break size for which a low steamline pressure signal will not occur prior to the  $OP\Delta T$  reactor trip, and the analysis terminates 5 seconds after reactor trip. "Steam System Piping Failure" presented in subsection 15.1.5 bounds the return to power following reactor trip and steamline isolation. If the low steamline pressure setpoint is reached, a reactor trip on safety injection actuation would result and terminate the event. Therefore, like the analysis performed for "Uncontrolled Rod Cluster Control Assembly Bank Withdrawal at Power" (subsection 15.4.2), to demonstrate protection by the  $\Delta T$  trips, only the applicable range of these trips needs to be considered. Also note that no credit is taken in the steamline break with coincident rod withdrawal at power analysis for a reactor trip via the high neutron flux overpower protection signal, since this trip function may be inoperable due to adverse environmental conditions associated with a steamline break inside containment.

The performance of the analysis for a steamline break with coincident withdrawal of the control rods due to an adverse environment demonstrates that the corresponding minimum DNBR does not decrease below the appropriate safety analysis limit DNBR value, and no fuel or clad damage occurs. Additionally, no system overpressurization is expected since the steamline break results in an RCS depressurization as described above.

This is an ANS Condition III/IV incident.

### 15.4.9.2 Analysis of Effects and Consequences

#### 15.4.9.2.1 Method of Analysis

The analysis of this transient uses the LOFTRAN computer code (reference 1). The following assumptions were made for this transient:

- A. The analysis employs RTDP methodology in determining initial conditions of maximum core power, reactor coolant average temperature, and minimum reactor coolant pressure.
- B. For end of life shutdown margin and equilibrium xenon conditions, the analysis assumes the most reactive RCCA stuck in its fully withdrawn position for conditions following reactor trip.
- C. The analysis uses a negative moderator coefficient corresponding to the end of life unrodded core. This maximizes the reactivity insertion caused by the cooldown during the steamline break.
- D. The analysis assumes the reactor trip setpoint on OP $\Delta$ T at a conservative value. The  $\Delta$ T trip includes all adverse instrumentation and setpoint errors; the delays for trip actuation are at the maximum values.
- E. The analysis bases the RCCA trip insertion characteristic on the assumption that the highest worth assembly is stuck in its fully withdrawn position.
- F. A spectrum of break sizes are analyzed. The limiting break is the largest break size for which a low steam line pressure signal will not occur and a reactor trip occurs on OPΔT.
- G. The calculation of the steam flow during a steamline break uses the Moody Curve for f L/D = 0.
- H. A conservatively large reactivity insertion rate is used.

No single active failure in any plant systems or equipment will adversely affect the consequences of the accident.

### 15.4.9.2.2 Results

The minimum DNBR occurred with beginning of life reactivity coefficients and a 0.7 ft break area. The calculated sequence of events for the limiting case is shown in table 15.4.1-1.

Figures 15.4.9-1, 15.4.9-2 and 15.4.9-3 show transient conditions following the steam line rupture with coincident RCCA bank withdrawal.

The steamline break affects the turbine impulse transmitters and causes the control rods to withdraw at the initiation of the transient. This causes an increase in reactor power and core heat flux to the point at which the  $OP\Delta T$  trip setpoint is reached. The reactor trip terminates the most adverse part of the transient. The steamline break causes increased heat removal and subsequent decrease in primary pressure simultaneous with the increase in reactor power. If the transient extends beyond post-reactor trip, secondary pressure will decrease until the low steamline pressure setpoint is reached, initiating steamline isolation and safety injection actuation.

15.4-35

The analysis of the steamline break with coincident RCCA bank withdrawal demonstrates that the DNBR limit is met. The most limiting part of this transient pertinent to this study was immediately before reactor trip (i.e., rod motion). The transient for the steamline break presented in subsection 15.1.5 bounds the return to power following reactor trip and steamline isolation. The other FSAR steamline break analysis assumed a larger break size and initial conditions corresponding to no-load temperatures (i.e., less stored energy in the RCS and reactor fuel). The DNBR is always greater than the limit value. Figure 15.4.9-3 shows the DNBR as a function of time for this transient.

# 15.4.9.3 <u>Conclusions</u>

The analysis demonstrates that the DNBR does not decrease below the limit value and no fuel or clad damage occurs. Additionally, no system overpressurization will occur; thus, all applicable safety criteria are met. As stated in the results, the large steamline break analysis presented in subsection 15.1.5 bounds the return to power following a reactor trip and steamline isolation; therefore, there is adequate protection to ensure plant safety for this transient.

### 15.4.9.4 <u>Reference</u>

1. Burnett, T. W. T., <u>et al.</u>, "LOFTRAN Code Description," <u>WCAP-7907-P-A</u> (proprietary), and <u>WCAP-7907-A</u> (nonproprietary), April 1984.

# TABLE 15.4.1-1 (SHEET 1 OF 4)

# TIME SEQUENCE OF EVENTS FOR INCIDENTS WHICH RESULT IN REACTIVITY AND POWER DISTRIBUTION ANOMALIES

Accident	Event	Time Delays (s)
Uncontrolled RCCA bank withdrawal from a subcritical or		
low-power startup condition	Initiation of uncontrolled rod withdrawal from 10 <sup>-9</sup> fraction of nominal power	0.0
	Power range high neutron flux low setpoint reached	12.5
	Peak nuclear power occurs	12.7
	Rods begin to fall into core	13.0
	Minimum DNBR occurs	14.9
	Peak heat flux occurs	14.9
	Peak average clad temperature occurs	15.1
	Peak average fuel temperature occurs	15.4
Uncontrolled RCCA bank withdrawal at power (full power with minimum feedback)		
1. Case A	Initiation of uncontrolled RCCA withdrawal at a high-reactivity insertion rate (80 pcm/s)	0
	Power range high neutron flux high setpoint reached	1.4
	Rods begin to fall into core	1.9
	Minimum DNBR occurs	1.95

# TABLE 15.4.1-1 (SHEET 2 OF 4)

Accide	ent	<u>Event</u>	Time Delays (s)
2.	Case B	Initiation of uncontrolled RCCA withdrawal at a small-reactivity insertion rate (2 pcm/s)	0
		OT∆T setpoint reached	34.0
		Rods begin to fall into core	36.0
		Minimum DNBR occurs	37.1
Startu coolar	p of an inactive reactor nt loop at an incorrect		
tempe		Initiation of pump startup	0.0
		Power reaches P-8 trip setpoint	3.5
		Rods begin to drop	4.0
		Minimum DNBR occurs	5.0
CVCS a deci conce coolar	a malfunction that results in rease in the boron Intration in the reactor Int		
1.	Dilution during startup	Power range - low setpoint reactor trip due to dilution	0
		Shutdown margin lost (if dilution continues after trip)	2010

# TABLE 15.4.1-1 (SHEET 3 OF 4)

Accio	<u>dent</u>		Event	Time Delays (s)
2.	Dilution during full-power operation			
	а.	Automatic reactor control	Operation receives low-low rod insertion limit alarm due to dilution	0
			Shutdown margin lost	3560
	b.	Manual reactor control	Reactor trip on $OT\Delta T$ due to dilution	0
			Shutdown margin is lost (if dilution continues after trip)	1860
RCC	A ejecti	on accident		
1.	End o	of life, zero power	Initiation of rod ejection	0.0
			Power range high neutron flux low setpoint reached	0.22
			Peak nuclear power occurs	0.24
			Rods begin to fall into core	0.72
			Peak clad average temperature occurs	1.79
			Peak heat flux occurs	1.79
			Peak fuel average temperature occurs	1.98

l

# TABLE 15.4.1-1 (SHEET 4 OF 4)

Accident	Event	Time Delays (s)
2. Beginning of life, full power	Initiation of rod injection Power range high neutron flux high setpoint reached	0.0 0.05
	Peak nuclear power occurs	0.13
	Rods begin to fall into core	0.55
	Peak fuel average temperature occurs	2.44
	Peak clad average temperature occurs	2.52
	Peak heat flux occurs	2.53
Steam line break with coincident	Steam line ruptures, RCCA bank	0.0
rod withdrawar at power	$OP\Delta T$ reactor trip setpoint reached	11.8
	Rods begin to fall	13.8
	Minimum DNBR occurs	14.6

# TABLE 15.4.6-1

# PARAMETERS

Dilution Flowrates:

Initiator	<u>Flowrate (gal/min)</u>
1	63
2	100
3	3.5
4	110

### Volumes:

Mode	Volume (ft <sup>3</sup> )	<u>Volume (gal)</u>
3, 4	9583	71,681
5a (loops filled)	4120	30,818
5b (loops not filled)	3460 <sup>(a)</sup>	25,880
6 (loops not filled)	3460 <sup>(a)</sup>	25,880

a. This volume corresponds with the reactor vessel coolant level at the mid-plane of the nozzles.

### TABLE 15.4.8-1

# PARAMETERS USED IN THE ANALYSIS OF THE ROD CLUSTER CONTROL ASSEMBLY EJECTION ACCIDENT

Time in Life	HZP <u>Beginning</u>	HZP <u>Beginning</u>	HZP End	HZP End
Power level (%)	0	102	0	102
Ejected rod worth (% $\Delta k$ )	0.75	0.24	0.84	0.25
Delayed neutron fraction (%)	0.54	0.57	0.46	0.46
Doppler feedback reactivity weighting	1.744	1.30	3.55	1.30
Trip reactivity (%∆k)	2.0	4.0	2.0	4.0
F <sub>Q</sub> before rod ejection		2.55		2.55
$F_Q$ after rod ejection	11.0	5.5	26.0	6.0
Number of operational pumps	2	4	2	4
Maximum fuel pellet average temperature at the hot spot (°F)	3425	4091	3412	3970
Maximum fuel center temperature at the	3985	>4900	3891	>4800
Maximum fuel stored energy at the hot spot (cal/g)	144.9	179.2	144.2	172.7
Percent of fuel melted at the hot spot	0	<10	0	<10

# TABLE 15.4.8-2 (SHEET 1 OF 2)

# PARAMETERS USED IN EVALUATING THE RADIOLOGICAL CONSEQUENCES OF A CONTROL ROD EJECTION ACCIDENT

### I. Source Data

II.

Α.	Core power level (MWt)	3636
B.	Total steam generator tube leakage (gal/min)	1
C.	Reactor coolant iodine activity prior to accident	An assumed preaccident iodine spike, which has resulted in the DE of 60 $\mu$ Ci/g of I-131 in the reactor coolant. See table 15A-6.
D.	Gap activity released to reactor coolant from failed fuel	10 percent See table 15A-3.
E.	Melted fuel	0.25 percent of core (0.00125 of core iodines, 0.0025 of core noble gases)
F.	Reactor coolant noble gas activity	Based on 1 percent defective fuel. See table 15A-4.
G.	Secondary system initial activity	DE of 0.1 $\mu \text{Ci/g}$ of I-131.
Н.	Reactor coolant mass (g)	2.3 x 10 <sup>8</sup>
I.	Secondary coolant mass, 4 generators (g)	1.9 x 10 <sup>8</sup>
J.	Offsite power	Lost after trip
Atm Fact	ospheric Dispersion ors	See table 15A-2.

# TABLE 15.4.8-2 (SHEET 2 OF 2)

# III. Activity Release Data

Α.	Containment		
	1.	Leak rate (percent/day)	0.2
	2.	Mass of primary coolant discharged to containment (lb)	
		0 to 1600 s 1600 to 4700 s 4700 to 10000 s	9.3 x 10 <sup>4</sup> 3.4 x 10 <sup>5</sup> 6.9 x 10 <sup>5</sup>
	3.	Fraction of activity carried by reactor coolant spill that is assumed to be airborne	
		lodines Noble gases	0.45 1.0
В.	Stea	am generators	
	1.	Primary-to-secondary leak rate (gal/min) <sup>(a)</sup>	1.0
	2.	Mass of steam released (lb)	
		0 - 214 s	4.9 x 10 <sup>4</sup>
	3.	lodine partition factor	100

a. Based on water at 62.4 lb/ft<sup>3</sup>.

# TABLE 15.4.8-3

# RADIOLOGICAL CONSEQUENCES OF A CONTROL ROD EJECTION ACCIDENT

Exclusion Area Boundary (0 to 2 h)	<u>Doses (rem)</u>
Thyroid	9.6
Whole-body gamma	0.2
Low Population Zone Outer Boundary (30 days)	
Thyroid	23
Whole-body gamma	0.1






































R Ρ Ν MLKJH GFED С В Α -8.9 -7.4 1 -5.6 -9.1 -8.5 2 -8.2 -4.1 0.2 -6.8 3 -7.9 -8.2 -7.7 4 -8.4 -6.0 -3.8 -1.8 5 -8.5 -8.4 -7.4 -5.5 -0.3 6 -7.7 -5.0 -1.2 -1.0 7 -7.7 -7.3 -5.9 -3.2 1.5 3.2 3.4 3.6 8 -6.9 2.7 5.9 6.0 9 -3.4 0.7 10.6 10 -5.3 -1.8 5.9 17.1 11.4 11 1.3 12.3 24.6 12 0.1 0.7 7.7 23.6 13 2.5 4.7 11.1 17.6 14 2.1 6.5 15 CASE A REV 14 10/07 REPRESENTATIVE % CHANGE IN LOCAL ASSY. AVG. POWER FOR INTERCHANGE BETWEEN VOGTLE SOUTHERN **REGION 1 AND REGION 3 ASSY.** ELECTRIC GENERATING PLANT COMPANY UNIT 1 AND UNIT 2 Energy to Serve Your World  $^{\otimes}$ FIGURE 15.4.7–1

R	Ρ	N	M	L	к	J	н	G	F	E	D	С	В	A				
			1			-4.8			-4.5						1			
		-0.4			-1.8		-1.8						1		2			
		-0.4			-4.0		-4.0								2			
							-4.8		-4.4		-2.6		1.4		3			
<b></b>	-2.8	-3.1					-4.5								4			
				-3.9				-4.3		-4.6		-1.5			5			
-3.8		-3.8			-3.6		-2.9						0.5		6			
			-3.6			-2.0			-2.3	$\times$		2.2			7			
-3.5		-3.4		-2.6		-0.7			11.4	$\aleph$	11.3	5.8	4.4		8			
	-3.2							5.2		16.7				5.4	9			
				-1.7		0.5					8.8				10			
-2.2				-1.0			2.2			6.9				6.6	11			
					0.0			2.9			6.5				12			
		1.2		0.0			1.6						10.3		13			
		3.2			0.8				3.2		6.0			•	14			
				0.3			1.5						•		15			
						CA	ASE E	3-1			•							
							RI	EV 14	10/07									
SOUTH	SOUTHERN A VOGTLE SOUTHERN A V											L ASSY. WEEN P RODS						
Energy to Serv	<b>COMPANY</b> Energy to Serve Your World® ELECTRIC GENERATING PLANT UNIT 1 AND UNIT 2									FIGURE 15.4.7–2								

R	Ρ	N	Μ	L	к	J	н	G	F	E	D	С	В	А	
						1.0			1.1		1				1
		5.1			1.0		1.0								2
							1.1		1.1		1.9		4.9		3
	1.7	1.7					1.4								4
				1.1				1.8		1.1		0.7			5
0.0		0.2			1.8		3.9						4.0		6
			0.0			5.2			2.2			-0.3			7
-0.7		-0.6		0.3		5.1	$\geq$		1.5		-0.3	-0.6	-0.7		8
	-1.0						$\mathbb{R}$	-1.1		-0.8				-0.9	9
				-1.4		-3.1					-1.3				10
-0.9				-1.7						-1.7				-0.9	11
					-2.5			-2.9			-1.1				12
		0.7		-1.9			-2.9						-2.5		13
		2.3				-2.8			-2.4		-0.8				14
				-2.1			-2.8								15
						CA	ASE E	3-2							
							_			_					
							F	REPF	10/0 RESEN	07 NTATI	VE %	CHAN			CAL ASSY
SOUTI	SOUTHERN A VOGTLE COMPANY ELECTRIC GENERATING PLANT							AVG REG RC	6. POV ION 1 DDS T	VER F AND RANF	FOR IN REGI FERRE	NTER ON 2 ED TC	CHAN ASSY REG	GE BI	ETWEEN H THE BP ASSY.
Energy to Set	Energy to Serve Your World® UNIT 1 AND UNIT 2 FIGURE 15.4.7–3														

	R	Ρ	N	Μ	L	к	J	н	G	F	Ε	D	С	в	А	
				1			-2.2			-2.1						1
			2.0			-2.0		-2.1								2
								-1.5		-1.6		-1.0		2.0		3
		-0.9	-1.0					-0.4								4
ſ					-0.4				1.2		-0.5		-1.4			5
ſ	-2.1		-1.6			2.3		5.7						-2.0		6
				-3.2			9.7			4.4			-1.7			7
	-2.3		-1.6		1.8		13.6	$\boxtimes$	]	5.6		-0.4	-1.6	-2.1		8
		-2.2							9.7		1.1				-2.2	9
					0.3		4.5					-0.9				10
	-1.9				-0.4			1.8			-0.5				-1.9	11
-						-0.9			-0.6			-1.1				12
			0.4		-1.4			-1.5						2.0		13
			2.0				-2.1			-2.0		-0.9			-	14
					-1.9			-2.2								15
							(	CASE	С							
								R	FV 14	10/07	,					
<b>SOU</b> Energy	OUTHERNAN COMPANY VOGTLE ELECTRIC GENERATING PLANT UNIT 1 AND UNIT 2 REPRESENTATIVE % CHANGE IN LOCAL ASSY. AVG. POWER FOR ENRICHMENT ERROR (REGION 2 ASSY. LOADED INTO CORE CENTRAL POSITION)															
											F	IGUR	E 15.4	1.7–4		

R	Ρ	N	М	L	к	J	н	G	F	E	D	С	в	A		
						-11			-14						1	
		0.4			-9.2		-12								2	
							-12		-14		-15		-13		3	
	3.2	1.2					-11								4	
				-1.5				-12		-15		-16			5	
9.8		7.1			-1.6		-8.0						-16		6	
			9.2			-2.3			-12			-14			7	
20.0		17.8		10.8		0.8			-10		-14	-15	-16		8	
	27.2							-5.5		-11				-15	9	
				20.7		5.8					-12				10	
42.0		$\ltimes$		23.6			1.9			-8.6				-13	11	
					14.0			-1.7			-8.9				12	
		38.6		20.4			2.8						-7.0		13	
		35.9				7.0			-3.3		-6.3				14	
				15.3			2.9								15	
	CASE D															
							REV 1	4 10/	07							
SOUTHER COM Energy to Serve Your	VOGTLE ELECTRIC GENERATING PLANT UNIT 1 AND UNIT 2							REPRESENTATIVE % CHANGE IN LOCAL ASSY. AVG. POWER FOR LOADING REGION 2 ASSY. INTO REGION 1 POSITION NEAR CORE PERIPHERY								
							FIGURE 15.4.7–5									















## 15.5 INCREASE IN REACTOR COOLANT INVENTORY

Discussion and analysis of the following events which cause an increase in reactor coolant inventory are presented in this section:

- A. Inadvertent operation of the emergency core cooling system (ECCS) during power operation.
- B. Chemical and volume control system (CVCS) malfunction that increases reactor coolant inventory.
- C. A number of boiling water reactor (BWR) transients (not applicable to VEGP).

### 15.5.1 INADVERTENT OPERATION OF THE EMERGENCY CORE COOLING SYSTEM DURING POWER OPERATION

### 15.5.1.1 Identification of Causes and Accident Description

Inadvertent operation of the Emergency Core Cooling System (ECCS) at power could be caused by operator error, test sequence error, or a false electrical actuation signal. A spurious signal initiated after the logic circuitry in one solid-state protection system train for any of the following engineered safety feature (ESF) functions could cause this incident by actuating the ESF equipment associated with the affected train.

- A. High containment pressure.
- B. Low pressurizer pressure.
- C. Low steam line pressure.

Following the actuation signal, the suction of the coolant charging pumps diverts from the volume control tank to the refueling water storage tank. Simultaneously, the valves isolating the injection header from the charging pumps open and the normal charging line isolation valves close. The charging pumps force the borated water from the RWST through the pump discharge header, the injection line, and into the cold leg of each loop. The safety injection (SI) pumps also start automatically but provide no flow when the reactor coolant system (RCS) is at normal pressure. The passive accumulator tank safety injection and low head system are available. However, they do not provide flow when the reactor coolant system (RCS) is at normal pressure.

A SI signal normally results in a direct reactor trip and a turbine trip. However, any single fault that actuates the ECCS will not necessarily produce a reactor trip. If an SI signal generates a reactor trip, the operator should determine if the signal is spurious. If the SI signal is determined to be spurious, the operator should terminate SI and maintain the plant in the hot standby condition as determined by appropriate recovery procedures. If repair of the ESF actuation system instrumentation is necessary, future plant operation will be in accordance with the Technical Specifications.

If the reactor protection system does not produce an immediate trip as a result of the spurious SI signal, the reactor experiences a negative reactivity addition due to the injected boron, which causes a decrease in reactor power. The power mismatch causes a drop in  $T_{avg}$  and consequent coolant shrinkage. The pressurizer pressure and water level decrease. Load decreases due to the effect of reduced steam pressure on load after the turbine throttle valve is

fully open. If automatic rod control is used, these effects will lessen until the rods have moved out of the core. The transient is eventually terminated by the reactor protection system low pressurizer pressure trip or by manual trip.

The time to trip is affected by initial operating conditions. These initial conditions include the core burnup history which affects initial boron concentration, rate of change of boron concentration, and Doppler and moderator coefficients.

# 15.5.1.2 Analysis of Effects and Consequences

## 15.5.1.2.1 Method of Analysis

Inadvertent operation of the ECCS is analyzed using the LOFTRAN<sup>(1)</sup> computer code. The code simulates the neutron kinetics, RCS, pressurizer, pressurizer relief and safety valves, pressurizer spray, feedwater system, steam generator, steam generator safety valves, and the effect of the SI system. The code computes pertinent plant variables including temperatures, pressures, and power level.

Inadvertent operation of the ECCS at power is classified as a Condition II event, a fault of moderate frequency. The criteria established for Condition II events include the following:

- A. Pressure in the reactor coolant and main steam systems should be maintained below 110 percent of the design values.
- B. Fuel cladding integrity shall be maintained by ensuring that the minimum departure from nucleate boiling ratio (DNBR) remains above the 95/95 DNBR limit for PWRs.
- C. An incident of moderate frequency should not generate a more serious plant condition without other faults occurring independently.

To address criterion C, the analysis historically used the more restrictive criterion that a water-solid pressurizer condition be precluded when the pressurizer is at or above the set pressure of the pressurizer safety valves (PSVs). This addressed any concerns regarding subcooled water relief through the plant PSVs. The current analysis conservatively predicts the minimum time to reaching a pressurizer water solid condition and the resulting water relief characteristics (if applicable). An evaluation of the PSV operability for temporary water relief under the specific conditions of this transient was performed. The evaluation demonstrated that a more serious plant condition will not result following an inadvertent ECCS actuation by confirming that the RCS pressure boundary remains intact for the post-transient plant shutdown.

The inadvertent ECCS actuation at power event is analyzed to determine both the minimum DNBR value and maximum pressurizer volume (or minimum time to a pressurizer water-solid condition and subsequent water relief). The most limiting case with respect to DNB is a minimum reactivity feedback condition with the plant assumed to be in manual rod control. Because of the power and temperature reduction during the transient, operating conditions do not approach the core limits.

For maximizing the potential for pressurizer filling, the most limiting case is a maximum reactivity feedback condition with an immediate reactor trip, and subsequent turbine trip, on the initiating SI signal. The transient results are presented for each case.

The analysis assumptions are as follows:

A. Initial Operating Conditions

The DNB case is analyzed with the Revised Thermal Design Procedure as described in WCAP-11397-P-A (reference 2). Initial reactor power, RCS pressure, and temperature are assumed to be at the nominal full power values. Uncertainties in initial conditions are included in the limit DNBR as described in reference 2. VEGP has a vessel average temperature window from 570°F to 588.4°F; therefore, cases at each end of this window are analyzed for pressurizer filling. The initial conditions for these cases assume maximum uncertainties on power (+2 percent), vessel average temperature (-5°F), and pressurizer pressure (-50 psia).

B. Moderator and Doppler Coefficients of Reactivity

The minimum feedback case (DNB) assumes a positive (+7 pcm/°F) moderator temperature coefficient and a low absolute value Doppler power coefficient at beginning of life (BOL). The maximum feedback case (pressurizer filling) assumes a large (absolute value) negative moderator temperature coefficient and a most-negative Doppler power coefficient.

C. Reactor Control

For the DNB case (without direct reactor trip on SI) the reactor is assumed to be in manual rod control.

D. Pressurizer Pressure Control

For the DNB case, the pressurizer heaters are inoperable. This assumption yields a higher rate of pressurizer decrease. The pressurizer spray portion of the pressurizer pressure control system is assumed available in order to minimize the RCS pressure. The PORVs are also assumed operable. PORVs reduce RCS pressure which is conservative for DNB analyses.

For the pressurizer filling cases, the following pressurizer pressure control system assumptions are made:

- Normal operation of the pressurizer spray is assumed. Spray actuates as a result of the pressure increase. Spray is assumed to be fully effective in condensing steam and thus maintaining a lower pressure until the water level increases to the point where the spray nozzle is submerged (i.e., when the pressurizer is nearly filled). This maximizes the ECCS injection flow.
- Operation of the pressurizer heaters minimizes the time to fill the pressurizer and subsequent water relief through the PSV(s). This is important since the maximum number of water relief cycles supported in the valve operability evaluation is four. Operation of the heaters, however, increases the temperature of the water that is relieved. Colder water relief temperatures are more limiting with respect to valve operability. Therefore, to ensure that both criteria are conservatively addressed, cases have been analyzed assuming both normal heater operation and no heater operation.
- PORVs are not assumed as an automatic pressure control system for the pressurizer filling case. Automatic actuation of the PORVs would directly mitigate the event consequences by preventing water relief through the PSVs. Operator action to make one PORV available following an acceptable

delay is credited in the analysis to mitigate the event. Should the PORVs fail due to water relief, the block valves would be available to isolate the RCS.

E. ECCS Injection

The inadvertent ECCS analysis models a maximum ECCS flow rate that bounds operation at the original design which included a positive displacement pump and the plant configuration with the replacement centrifugal pump. Safety injection (SI) is actuated at time zero, with flow injected to the RCS from two high-head centrifugal charging pumps plus the normal charging pump. Both high- head charging and SI pumps automatically start on an SI signal, and the associated alternate minimum flow protection lines receive a signal to open at high RCS back pressures. The failure of the normal charging pump to be stripped from the bus was taken as the single failure since it results in higher flow rates than the failure to open one safety-related miniflow path. The analysis also assumes zero injection line purge volume for calculation simplicity; thus, the boration transient begins immediately in the analysis.

F. Turbine Load

For the DNB case (without direct reactor trip/turbine trip on SI), the turbine load remains constant until the governor drives the throttle valve wide open. After the throttle valve is full open, turbine load decreases as steam pressure drops.

G. Reactor Trip

Reactor trip is initiated by a low pressurizer pressure signal at 1935 psia for the DNB case. The pressurizer filling case assumes reactor trip on the initiating SI signal.

H. Decay Heat

Core residual heat generation is based on the 1979 version of ANS 5.1 (reference 3). ANSI/ANS-5.1-1979 is a conservative representation of the decay energy release rates. Long-term operation at the initial power level preceding the trip is assumed.

I. Pressurizer Safety Valves

The safety valves open at a pressure of 2425 psia which corresponds to a tolerance of -2 percent relative to the set pressure of 2475 psia. The valves are assumed to close at a pressure of 2300 psia, which corresponds to a blowdown of 5 percent below the opening pressure of 2425 psia.

J. Operator Actions

An operator action, to make one PORV available for water relief, was assumed at 590 seconds in the pressurizer fill case initialized from the low end of the vessel average window (570.7°F), and 625 seconds was assumed for the pressurizer fill case initialized from the high end of the vessel average window (588.4°F).

#### 15.5.1.2.2 Results

The transient responses for the DNB and limiting pressurizer filling cases are shown in figure 15.5.1-1. Table 15.5.1-1 shows the calculated sequence of events.

For the DNB case, nuclear power starts decreasing immediately due to boron injection, but steam flow does not decrease until later in the transient when the turbine throttle valve is wide open. The mismatch between load and nuclear power causes  $T_{avg}$ , pressurizer water level, and pressurizer pressure to drop. The reactor trips and control rods start moving into the core when the pressurizer pressure reaches the pressurizer low pressure trip setpoint. The DNBR increases throughout the transient.

For the pressurizer filling case, reactor trip occurs at event initiation followed by a rapid initial cooldown of the RCS. Coolant contraction results in a short-term reduction in pressurizer pressure and water level. The combination of the RCS heatup, due to residual RCS heat generation, and ECCS injected flow causes the pressure and level transients to rapidly turn around. Pressurizer water level then increases throughout the transient. For the cases with operation of the heaters, the number of PSV water relief cycles is four; whereas, for the cases without operation of the heaters show a slightly lower minimum final water relief temperature which is a factor considered in determining the limiting PSV operability condition. However, although the minimum final water relief temperature of the cases with operation of the heaters is slightly higher, these cases are considered to be more limiting since the number of PSC water relief cycles is more.

In the case initialized from the low end of the vessel average window (570.7°F), the analysis assumes that at 590 seconds the operator manually opens a PORV. For the case initialized from the high end of the vessel average window (588.4°F), this operator action is assumed at 625 seconds. In each of these cases, once the PORV is fully open, the RCS begins to depressurize to below the pressure where the safety valves reseat.

## 15.5.1.3 <u>Conclusions</u>

Results of the analysis show that spurious ECCS operation without immediate reactor trip does not present any hazard to the integrity of the RCS with respect to DNBR. The minimum DNBR is never less than the initial value. Thus, there will be no cladding damage and no release of fission products to the RCS. If the reactor does not trip immediately, the low pressurizer pressure reactor trip will provide protection. This trips the turbine and prevents excess cooldown, which expedites recovery from the incident.

With respect to pressurizer filling, the pressurizer may reach a water-solid condition. However, the resulting potential water relief will not impair PSV operability. The RCS pressure boundary will therefore remain intact and the event will not generate a more serious plant condition.

## 15.5.1.4 <u>Reference</u>

- 1. Burnett, T.W.T., <u>et al.</u>, "LOFTRAN Code Description," <u>WCAP-7907-P-A</u>, (proprietary), <u>WCAP-7907-A</u> (nonproprietary), April 1984.
- 2. Friedland, A.J. and Ray, S., "Revised Thermal Design Procedure," <u>WCAP-11397-P-A</u>, April 1989.
- 3. ANSI/ANS-5.1-1979, "Decay Heat Power in Light Water Reactors," August 29, 1979.

### 15.5.2 CHEMICAL AND VOLUME CONTROL SYSTEM MALFUNCTION THAT INCREASES REACTOR COOLANT INVENTORY

An increase in reactor coolant inventory which results from the addition of cold, unborated water to the reactor coolant system (RCS) is analyzed in subsection 15.4.6, Chemical and Volume Control System Malfunction That Results in a Decrease in Boron Concentration in the Reactor Coolant. An increase in reactor coolant inventory which results from the injection of highly borated water into the RCS is analyzed in subsection 15.5.1, Inadvertent Operation of the Emergency Core Cooling System During Power Operation.

## 15.5.3 A NUMBER OF BOILING WATER REACTOR TRANSIENTS

This subsection is not applicable to the VEGP.

#### VEGP-FSAR-15

# TABLE 15.5.1-1 (SHEET 1 OF 2)

## TIME SEQUENCE OF EVENTS FOR INCIDENTS WHICH RESULT IN AN INCREASE IN REACTOR COOLANT INVENTORY (Accident: Inadvertent operation of ECCS during power operation)

<u>Case</u>	<u>Event</u>	<u>Tim</u>	<u>e(s)</u>				
DNBR case:	SI pumps begin injecting borated water	0.0					
	Low pressurizer pressure reactor trip setpoint reached	45	45.5				
	Rods begin to drop	47	7.5				
	Minimum DNBR occurs	(;	(a)				
Pressurizer filling case:		Pressu	Pressurizer Heaters				
		<u>On</u>	Off				
A nominal T <sub>AVG</sub> = 570.7°F	SI actuation, reactor trip	0.0	0.0				
Operator action to manually	Pressurizer fills with water	451.0	470.5				
open PORV at 590 S	PSV opens – cycle #1	482.0	504.5				
	PSV closes – cycle #1	486.0	509.0				
	PSV opens – cycle #2	516.0	537.5				
	PSV closes – cycle #2	520.0	541.5				
	PSV opens – cycle #3	547.0	569.0				
	PSV closes – cycle #3	551.5	573.0				
	PSV opens – cycle #3	579.5	N/A				
	PSV closes – cycle #4	584.0	N/A				
	Time of last PSV cycle (minimum final water relief temperature)	584.0 (631.5°F)	573.0 (628.4°F)				
	Operator action to open PORV	590.0	590.0				

# VEGP-FSAR-15

TABLE 15.5.1-1 (SHEET 2 OF 2)

<u>Accident</u>	<u>Event</u>	<u>Time(s)</u>				
		Pressurize	<u>er Heaters</u>			
		<u>On</u>	Off			
A nominal T <sub>AVG</sub> = 588.4°F	SI actuation, reactor trip	0.0	0.0			
Operator action to manually open PORV at 625 s	Pressurizer fills with water	484.5	505.5			
	PSV opens – cycle #1	520.5	543.0			
	PSV closes – cycle #1	525.0	547.0			
	PSV opens – cycle #2	555.0	578.5			
	PSV closes – cycle #2	559.0	583.0			
	PSV opens – cycle #3	589.5	611.5			
	PSV closes – cycle #3	593.5	615.5			
	PSV opens – cycle #4	620.5	N/A			
	PSV closes – cycle #4	625.0	N/A			
	Time of last PSV cycle (minimum final water relief temperature)	625.0 (634.5°F)	615.5 (631.7°F)			
	Operator action to open PORV	625.0	625.0			

a. DNBR does not decrease below its initial value.



















## 15.6 DECREASE IN REACTOR COOLANT INVENTORY

Events which result in a decrease in reactor coolant inventory, as discussed in this section, are as follows:

- A. Inadvertent opening of a pressurizer safety or relief valve.
- B. Break in instrument line or other lines from reactor coolant pressure boundary that penetrate the containment.
- C. Steam generator tube failure.
- D. Spectrum of boiling water reactor (BWR) steam system piping failures outside of containment (not applicable to the VEGP).
- E. Loss-of-coolant accident (LOCA) resulting from a spectrum of postulated piping breaks within the reactor coolant pressure boundary.
- F. A number of BWR transients (not applicable to VEGP).

All of the applicable accidents in this category have been analyzed. It has been determined that the most severe radiological consequences will result from the major LOCA of subsection 15.6.5. Therefore, the LOCA is the design basis accident. The LOCA chemical and volume control system (CVCS) letdown line break outside the containment and the steam generator tube rupture accident have been analyzed radiologically. All other accidents in this section are bounded by these accidents.

## 15.6.1 INADVERTENT OPENING OF A PRESSURIZER SAFETY OR RELIEF VALVE

### 15.6.1.1 Identification of Causes and Accident Description

An accidental depressurization of the reactor coolant system (RCS) could occur as a result of an inadvertent opening of a pressurizer relief or safety valve. Since a pressurizer safety valve is sized to relieve approximately twice the steam flowrate of a relief valve and will therefore allow a much more rapid depressurization upon opening, the most severe core conditions resulting from an accidental depressurization of the RCS are associated with an inadvertent opening of a pressurizer safety valve. Initially the event results in a rapidly decreasing RCS pressure. The effect of the pressure decrease is to increase power via the moderator density feedback. The average coolant temperature remains approximately the same, but the pressurizer level increases until reactor trip.

The reactor may be tripped by the following reactor protection system signals:

- Overtemperature  $\Delta T$ .
- Pressurizer low pressure.

This is an ANS Condition II incident.

## 15.6.1.2 <u>Analysis of Effects and Consequences</u>

#### 15.6.1.2.1 Method of Analysis

The accidental depressurization transient is analyzed by employing the detailed digital computer code LOFTRAN.<sup>(1)</sup> The code simulates the neutron kinetics, RCS, pressurizer, pressurizer relief and safety valves, pressurizer spray, steam generator, and steam generator safety valves. The code computes pertinent plant variables including temperatures, pressures, and power level.

In order to produce conservative results in calculating the DNBR during the transient, the following assumptions are made:

- A. Nominal values are assumed for the initial reactor power, pressure, and RCS temperatures. Uncertainties in initial conditions are included in the limit DNBR as described in reference 7. (See tables 15.0.3-2 and 15.0.3-3.)
- B. A most positive moderator temperature coefficient is assumed. The spatial effect of voids resulting from local or subcooled boiling is not considered in the analysis with respect to reactivity feedback or core power shape.
- C. A small (absolute value) Doppler coefficient of reactivity such that the resultant amount of negative feedback is conservatively low. This tends to maximize any power increase resulting from moderator reactivity feedback.

Plant systems and equipment which are necessary to mitigate the effects of RCS depressurization caused by an inadvertent safety valve opening are discussed in subsection 15.0.8 and listed in table 15.0.8-1.

Normal automatic rod control systems are not required to function. The reactor protection system functions to trip the reactor on the appropriate signal. No single active failure will prevent the reactor protection system from functioning properly.

### 15.6.1.2.2 Results

The system response to an inadvertent opening of a pressurizer safety valve is shown in figures 15.6.1-1 and 15.6.1-2. Figure 15.6.1-1 illustrates the nuclear power transient following the depressurization. Nuclear power increases slowly until reactor trip occurs on overtemperature delta T. The pressure decay transient and average temperature transient following the accident are given in figure 15.6.1-2. Pressure drops more rapidly while core heat generation is reduced via the trip and then slows once saturation temperature is reached in the hot leg. The DNBR decreases initially but increases rapidly following the trip, as shown in figure 15.6.1-1. The DNBR remains above its limit throughout the transient. The DNBR design basis is described in section 4.4.

The calculated sequence of events for the inadvertent opening of a pressurizer safety valve incident is shown in table 15.6.1-1.

#### 15.6.1.3 <u>Conclusion</u>

The results of the analysis show that the pressurizer low pressure and the overtemperature WT reactor protection system signals provide adequate protection against the RCS depressurization event.

## 15.6.1.4 <u>Reference</u>

1. Burnett, T. W. T., <u>et al</u>., "LOFTRAN Code Description," <u>WCAP-7907-P-A</u> (proprietary), <u>WCAP-7907-A</u> (nonproprietary), April 1984.

### 15.6.2 BREAK IN INSTRUMENT LINE OR OTHER LINES FROM REACTOR COOLANT PRESSURE BOUNDARY THAT PENETRATE CONTAINMENT

The reactor coolant system (RCS) wide-range pressure instrument lines penetrate the containment. In addition, there are the sample lines from the hot legs of reactor coolant loops 1 and 3, from the steam and liquid space of the pressurizer, and from the 3-in. chemical and volume control system (CVCS) letdown line penetrating the containment. The sample lines are provided with isolation valves on both sides of the containment wall and are designed in accordance with the requirements of General Design Criterion (GDC) 55.

A break in one of the RCS wide-range pressure instrument lines at a point outside the containment but before the hydraulic isolator combined with the failure of the instrument bellows would result in a nonisolable release of reactor coolant outside the containment. While the release would continue until the plant is shut down and the RCS is depressurized, the release rate never exceeds 0.106 lb/s.

The most severe pipe rupture outside containment, with regard to radioactivity release during normal plant operation, occurs in the CVCS. This would be a complete severance of the 3-in. letdown line just outside containment but between the outboard letdown isolation valve and letdown heat exchanger at rated power condition. (See drawings 1X4DB114, 1X4DB115, 1X4DB116-1, 1X4DB116-2, 1X4DB117, and 1X4DB118.) The occurrence of a complete severance of the letdown line would result in a loss of reactor coolant at the rate of about 21.3 lb/s. Since the release rate is within the capability of the reactor makeup system, it would not result in engineered safety features system actuation. Frequent operation of the automatic reactor makeup system in addition to the low pressurizer pressure deviation alarm will provide the operator some indication of the loss of reactor coolant. After the 30 min, the operator is assumed to isolate the letdown line.

### 15.6.2.1 <u>Assumptions - Instrument Line Break</u>

The major assumptions and parameters used in the analysis are provided in table 15.6.2-1 and summarized below:

- A. The reactor coolant iodine activity is based on a pre-existing iodine spike of 60  $\mu$ Ci/g dose equivalent I-131. (See table 15A-6.)
- B. The noble gas activity in the reactor coolant is based on 1-percent fuel defects. (See table 15A-4.)
- C. Reactor shutdown is assumed to be initiated 8 h after the break occurs.
- D. One residual heat removal system train is assumed to be available. Time to cool the RCS to < 212°F is 24 h.
- E. A total of 12,200 lb of reactor coolant is spilled (based on a release for 32 h).
- F. All of the noble gases in the spilled reactor coolant are released to the environment.

- G. The fraction of the spill assumed to flash is 58 percent. All of the iodine activity in the flashed fraction of the spill is assumed to be released. Ten percent of the iodine in the unflashed fraction is assumed to be released.
- H. No credit is taken for mixing and holdup of the releases within the auxiliary building nor are the auxiliary building normal exhaust filters credited with reducing the release. That is, the release is modeled as being direct to the environment.
- I. No credit is taken for ground deposition or decay in transit to the exclusion area boundary or outer boundary of the low population zone.

## 15.6.2.2 Assumptions - Letdown Line Break

The major assumptions and parameters used in the analysis are provided in table 15.6.2-2 and summarized below:

- A. The reactor coolant iodine activity is based on a pre-existing iodine spike of  $60-\mu$ Ci/g dose equivalent I-131. (See table 15A-6.)
- B. The noble gas activity in the reactor coolant is based on 1-percent fuel defects. (See table 15A-4.).
- C. A total of 38,400 lb of reactor coolant is spilled (based on a release for 30 min).
- D. All of the noble gases in the spilled reactor coolant are released to the environment.
- E. The fraction of the spill assumed to flash is 8.6 percent. All of the iodine activity in the flashed fraction of the spill is assumed to be released. Ten percent of the iodine in the unflashed fraction is assumed to be released.
- F. No credit is taken for mixing and holdup of the releases within the auxiliary building nor are the auxiliary building normal exhaust filters credited with reducing the release. That is, the release is modeled as being direct to the environment.
- G. No credit is taken for ground deposition or decay in transit to the exclusion area boundary or outer boundary of the low population zone.

## 15.6.2.3 <u>Mathematical Models Used in the Analysis</u>

Mathematical models used in the analysis are described in the following sections:

- A. The mathematical models used to analyze the activity released during the course of the accident are described in appendix 15A.
- B. The atmospheric dispersion factors used in the analysis were calculated based on the onsite meteorological measurement program described in section 2.3 and are provided in table 15A-2.
- C. The thyroid inhalation and total-body immersion doses to a receptor at the exclusion area boundary or outer boundary of the low population zone were analyzed using the models described in appendix 15A.

# 15.6.2.4 Identification of Leakage Pathways

The reactor coolant spilled in the auxiliary building will collect in the floor drain sumps. From there, it will be pumped to the radwaste treatment system. Therefore, the only release paths that present a radiological hazard involve the volatile fraction of spilled coolant.

Normally, gases released in the auxiliary building mix with the building atmosphere and are gradually exhausted through the filtered building ventilation system. The charcoal filters normally remove a very large fraction of the airborne iodine in the building atmosphere. However, the ventilation system is not designed to mitigate the consequences of an accident (e.g., it might not survive an earthquake) nor can the possibility of unplanned leakage from the auxiliary building be eliminated; hence, no credit is taken for these effects reducing the released activity.

## 15.6.2.5 Identification of Uncertainties and Conservatisms in the Analysis

The uncertainties and conservatisms in the assumptions used to evaluate the radiological consequences result principally from assumptions made involving the amount of the gaseous fission products available for release to the environment and the meteorology present at the site during the course of the accident. The most significant of these assumptions are:

- A. The meteorological conditions assumed to be present at the site during the course of the accident are based on  $\chi/Q$  values which are worse than those which will exist at the site 95 percent of the time. This condition results in the poorest values of atmospheric dispersion calculated for the exclusion area boundary and the low population zone outer boundary. Furthermore, no credit has been taken for the transit time required for activity to travel from the point of release to the exclusion area boundary and to the low population zone outer boundary. Hence, the radiological consequences evaluated under these conditions are conservative.
- B. The concentrations of noble gases and iodines assumed in the reactor coolant is significantly greater than those expected during normal plant operation.
- C. No credit is taken for deposition of airborne iodine.

## 15.6.2.6 <u>Conclusions</u>

The radiological consequences resulting from the occurrence of postulated line ruptures have been conservatively analyzed, using assumptions and models described in previous sections.

The thyroid inhalation and total-body immersion doses have been analyzed for the 0- to 2-h dose at the exclusion area boundary and for the duration of the accident at the low population zone outer boundary. The results are listed in table 15.6.2-3 for the instrument line break and in table 15.6.2-4 for the letdown line break. These doses do not exceed 10 percent of the guidelines values of 10 CFR 100.
## 15.6.3 STEAM GENERATOR TUBE FAILURE

#### 15.6.3.1 Identification of Causes and Accident Description

The accident examined is the complete severance of a single steam generator tube. This event is considered an American Nuclear Society (ANS) Condition IV event, a limiting fault. (See subsection 15.0.1.) The accident is assumed to take place at full power with the reactor coolant contaminated with fission products corresponding to continuous operation with a limited number of defective fuel rods. The accident leads to an increase in contamination of the secondary system due to leakage of radioactive coolant from the reactor coolant system (RCS). In the event of a coincident loss of offsite power or failure of the condenser steam dump system, discharge of activity to the atmosphere takes place via the steam generator atmospheric relief valves (and safety valves if their setpoint is reached).

Complete severance of a steam generator tube is considered a somewhat conservative assumption since the Inconel-600 tube material is highly ductile. The more probable mode of tube failure would be one or more minor leaks of undetermined origin. Activity in the steam and power conversion system is subject to continual surveillance, and an accumulation of minor leaks which exceed the limits established in the Technical Specifications is not permitted during the unit operation.

The operator is expected to determine that a steam generator tube rupture has occurred, to identify and isolate the ruptured steam generator, and to complete the required recovery actions to stabilize the plant and terminate the primary to secondary break flow. These actions should be performed on a restricted time scale in order to minimize contamination of the secondary system and ensure termination of radioactive release to the atmosphere from the faulted unit. Consideration of the indications provided at the control board, together with the magnitude of the break flow, leads to the conclusion that the recovery procedure can be carried out on a time scale which ensures that break flow to the secondary system is terminated before water level in the affected steam generator rises into the main steam pipe. Sufficient indications and controls are provided to enable the operator to carry out these functions satisfactorily.

If normal operation of the various plant control system is assumed, the following sequence of events is initiated by a tube rupture.

- A. Pressurizer low-pressure and low-level alarms are actuated and charging pump flow increases in an attempt to maintain pressurizer level. On the secondary side, steam flow/feedwater flow mismatch occurs, since feedwater flow to the affected steam generator is reduced as a result of primary coolant break flow to that unit.
- B. The main steamline radiation monitors, the condenser air ejector radiation monitor, and/or the steam generator blowdown liquid monitor will alarm, indicating a sharp increase in radioactivity in the secondary system. The high radiation level alarm from the steam generator blowdown process monitor automatically isolates the system and terminates discharge. The high radiation level alarm from the air ejector monitor automatically diverts the air ejector and steam seal exhauster blower discharges through a filtration unit.
- C. The decrease in RCS pressure due to continued loss of reactor coolant inventory leads to a reactor trip signal on low pressurizer pressure or  $OT\Delta T$ . Resultant plant cooldown following reactor trip leads to a rapid decrease in RCS pressure and pressurizer level, and a safety injection signal initiated by low pressurizer

pressure follows soon after reactor trip. The safety injection signal automatically terminates normal feedwater supply and initiates auxiliary feedwater addition.

- D. The reactor trip automatically trips the turbine, and if offsite power is available, the steam dump valves open, permitting steam dump to the condenser. In the event of a coincident loss of offsite power, the steam dump valves automatically close to protect the condenser. The steam generator pressure rapidly increases, resulting in steam discharge to the atmosphere through the steam generator atmospheric relief valves (and safety valves if their setpoint is reached).
- E. Following reactor trip and safety injection actuation, the continued action of the auxiliary feedwater supply and borated safety injection flow (supplied from the refueling water storage tank) provides a heat sink which absorbs some of the decay heat. This reduces the amount of steam bypass to the condenser, or in the case of loss of offsite power, steam relief to the atmosphere.
- F. Safety injection flow results in increasing RCS pressure and pressurizer water level, and the RCS pressure trends toward the equilibrium value where the safety injection flow rate equals the break flow rate.

In the event of a steam generator tube rupture (SGTR), the plant operators must diagnose the SGTR and perform the required recovery actions to stabilize the plant and terminate the primary to secondary leakage. The operator actions for SGTR recovery are provided in the Emergency Operating Procedures. The major operator actions include identification and isolation of the ruptured steam generator, cooldown and depressurization of the RCS to restore inventory, and termination of SI to stop primary to secondary leakage. These operator actions are described below.

1. Identify the ruptured steam generator.

High secondary side activity, as indicated by the main steamline radiation monitors, the condenser air ejector radiation monitor, or steam generator blowdown radiation monitors, typically will provide the first indication of an SGTR event. The ruptured steam generator can be identified by an unexpected increase in steam generator narrow range level or a high radiation indication on the corresponding main steamline radiation monitor. For an SGTR that results in a reactor trip at high power, the steam generator water level will decrease due to void collapse but is expected to remain in the narrow range for all of the steam generators. The AFW flow will begin to refill the steam generators, distributing approximately equal flow to each of the steam generators. Since primary to secondary leakage adds additional liquid inventory to the ruptured steam generator, the water level will increase more rapidly in that steam generator. This response, as indicated by the steam generator water level instrumentation, provides confirmation of an SGTR event and also identifies the ruptured steam generator.

2. Isolate the ruptured steam generator from the intact steam generators and isolate feedwater to the ruptured steam generator.

Once a tube rupture has been identified, recovery actions begin by isolating steam flow from and stopping feedwater flow to the ruptured steam generator. In addition to minimizing radiological releases, this also reduces the possibility of overfilling the ruptured steam generator with water by 1) minimizing the accumulation of feedwater flow and 2) enabling the operator to establish a pressure differential between the ruptured and intact steam generators as a necessary step toward terminating primary to secondary leakage. 3. Cool down the RCS using the intact steam generators.

After isolation of the ruptured steam generator, the RCS is cooled as rapidly as possible to less than the saturation temperature corresponding to the ruptured steam generator pressure by dumping steam from only the intact steam generators. This ensures adequate subcooling in the RCS after depressurization to the ruptured steam generator pressure in subsequent actions. If offsite power is available, the normal steam dump system to the condenser can be used to perform this cooldown. However, if offsite power is lost, the RCS is cooled using the atmospheric relief valves (ARVs) on the intact steam generators.

4. Depressurize the RCS to restore reactor coolant inventory.

When the cooldown is completed, SI flow will increase RCS pressure until break flow matches SI flow. Consequently, SI flow must be terminated to stop primary to secondary leakage. However, adequate reactor coolant inventory must first be assured. This includes both sufficient reactor coolant subcooling and pressurizer inventory to maintain a reliable pressurizer level indication after SI flow is stopped. Since leakage from the primary side will continue after SI flow is stopped until RCS and ruptured steam generator pressures equalize, an "excess" amount of inventory is needed to ensure pressurizer level remains on span. The "excess" amount required depends on RCS pressure and reduces to zero when RCS pressure equals the pressure in the ruptured steam generator.

The RCS depressurization is performed using normal pressurizer spray if the rector coolant pumps (RCPs) are running. However, if offsite power is lost or the RCPs are not running for some other reason, normal pressurizer spray is not available. In this event, RCS depressurization can be performed using a pressurizer PORV or auxiliary pressurizer spray.

5. Terminate SI to stop primary to secondary leakage.

The previous actions will have established adequate RCS subcooling, a secondary side heat sink, and sufficient reactor coolant inventory to ensure that SI flow is no longer needed. When these actions have been completed, SI flow must be stopped to terminate primary to secondary leakage. Primary to secondary leakage will continue after SI flow is stopped until RCS and ruptured steam generator pressures equalize. Charging flow, letdown, and pressurizer heaters will then be controlled to prevent repressurization of the RCS and reinitiation of leakage into the ruptured steam generator.

Following SI termination, the plant conditions will be stabilized, the primary to secondary break flow will be terminated, and all immediate safety concerns will have been addressed. At this time, a series of operator actions are performed to prepare the plant for cool down to cold shutdown conditions. Subsequently, actions are performed to cool down and depressurize the RCS to cold shutdown conditions and to depressurize the ruptured steam generator.

#### 15.6.3.2 Analysis of Effects and Consequences

An SGTR results in the leakage of contaminated reactor coolant into the secondary system and subsequent release of a portion of the activity to the atmosphere. Therefore, an analysis must

be performed to assure that the offsite radiological consequences resulting from an SGTR are within the allowable guidelines. One of the major concerns for an SGTR is the possibility of steam generator overfill since this could potentially result in a significant increase in the offsite radiological consequences. Therefore, an analysis was performed to demonstrate margin to steam generator overfill (MTO) as documented in reference 11, assuming the limiting single failure relative to overfill. The analysis assumes that one of the steam generator atmospheric relief valves (ARVs) is out of service, as allowed by the Technical Specifications. The analysis was performed assuming one operable ARV was on the ruptured steam generator and the other two operable ARVs were on intact steam generators. The limiting single failure for the margin to overfill analysis results in the loss of control room (CR) control of the ARVs on the intact steam generators. Action outside the CR is required to open an intact steam generator ARV to perform the plant cooldown. The analysis assumed that a single intact steam generator ARV is used for the cooldown, and it is not opened until 12 minutes after the ruptured steam generator is isolated. The results of this analysis demonstrated that there is margin to steam generator overfill for VEGP. An analysis was also performed to determine the offsite radiological consequences as documented in reference 1, assuming the limiting single failure relative to offsite doses without steam generator overfill. Since steam generator overfill does not occur, the results of this analysis represent the limiting consequences for an SGTR for VEGP.

A thermal and hydraulic analysis was performed to determine the plant response for a design basis SGTR, and to determine the integrated primary to secondary break flow and the mass releases from the ruptured and intact steam generators to the condenser and to the atmosphere. This information was then used to calculate the quantity of radioactivity released to the environment and the resulting radiological consequences.

## 15.6.3.3 Thermal and Hydraulic Analysis

The plant response following an SGTR was analyzed with the LOFTTR2 (reference 1 and reference 11) program until the primary to secondary break flow is terminated. The reactor protection system and the automatic actuation of the engineered safeguards systems were modeled in the analysis. The major operator actions which are required to terminate the break flow for an SGTR were also simulated in the analysis.

## Analysis Assumptions

The accident modeled is a double-ended break of one steam generator tube located at the top of the tube sheet on the outlet (cold leg) side of the steam generator. It was assumed that the reactor is operating at full power at the time of the accident and the secondary level was assumed to correspond to operation at the nominal steam generator level minus an allowance for uncertainties. It was also assumed that a loss of offsite power occurs at the time of reactor trip and the highest worth control assembly was assumed to be stuck in its fully withdrawn position at reactor trip.

Other important analysis assumptions include:

- A. NSSS power = 3579 MWt\* 1.02 (uncertainty) = 3653 MWt.
- B. Average RCS temperature = 588.4°F for the dose consequence analysis. For the MTO analysis, the average RCS temperature = 570.7°F.
- C. RCS pressure = 2250 psia 50 psia (uncertainty) = 2200.
- D. Thermal design flow = 374400 gal/min.

- E. Pressurizer level = 60% (includes uncertainty) for dose consequence analysis and 40.3% for overall analysis.
- F. SG tube plugging = 0% for dose consequence analysis and 10% for overfill analysis.
- G. Auxiliary feed flow = 2110 gal/min for overfill analysis and 1600 gal/min for the dose analysis.
- H. Auxiliary feed flow delay time = 90 seconds for dose analysis, 30 seconds for overfill analysis.

The limiting single failure for offsite doses was assumed to be the failure of the ARV on the ruptured steam generator. Failure of this ARV in the open position will cause an uncontrolled depressurization of the ruptured steam generator which will increase primary to secondary leakage and the mass release to the atmosphere. It was assumed that the ruptured steam generator ARV fails open when the ruptured steam generator is isolated, and that the ARV is isolated by locally closing the associated block valve.

The offsite radiological analysis also assumed that one of the intact steam generator ARVs was inoperable. The analysis assumed that only two of the three intact steam generator ARVs were available for the cooldown.

The major operator actions required for the recovery from an SGTR are discussed in paragraph 15.6.3.1, and these operator actions were simulated in the analysis. The operator action times used for the analysis were established in reference 2 and are presented in table 15.6.3-1. It is noted that the ARV on the ruptured steam generator was assumed to fail open at the time the ruptured steam generator was isolated. Before proceeding with the recovery operations, the failed-open ARV on the ruptured steam generator was assumed to be isolated by locally closing the associated block valve. It was assumed that the ruptured steam generator ARV is isolated at 16 minutes after the valve was assumed to fail open. After the ruptured steam generator ARV was isolated, an additional delay time of 12 minutes (table 15.6.3-1) was assumed for the operator action time to initiate the RCS cooldown.

#### Transient Description

The LOFTTR2 analysis results are described below. The sequence of events for this transient is presented in table 15.6.3-2.

Following the tube rupture, reactor coolant flows from the primary into the secondary side of the ruptured steam generator since the primary pressure is greater than the steam generator pressure. In response to this loss of reactor coolant, pressurizer level decreases as shown in figure 15.6.3-1. The RCS pressure also decreases as shown in figure 15.6.3-2, as the steam bubble in the pressurizer expands. As the RCS pressure decreases due to the continued primary to secondary leakage, automatic reactor trip occurs on an OT∆T trip signal.

After reactor trip, core power rapidly decreases to decay heat levels. The turbine stop valves close, and steam flow to the turbine is terminated. The steam dump system is designed to actuate following reactor trip to limit the increase in secondary pressure, but the steam dump valves remain closed due to the loss of condenser vacuum resulting from the assumed loss of offsite power at the time of reactor trip. Thus, the energy transfer from the primary system causes the secondary side pressure to increase rapidly after reactor trip until the steam generator ARVs (and safety valves if their setpoints are reached) lift to dissipate the energy, as shown in figures 15.6.3-3. The main feedwater flow will be terminated and AFW flow will be automatically initiated following reactor trip and the loss of offsite power.

The RCS pressure decreases more rapidly after reactor trip as energy transfer to the secondary shrinks the reactor coolant and the tube rupture break flow continues to deplete primary inventory. Pressurizer level also decreases more rapidly following reactor trip. The decrease in RCS inventory results in a low pressurizer pressure SI signal. After SI actuation, the SI flowrate exceeds the tube rupture break flowrate, and the pressurizer level begins to increase. This also results in an increase in the RCS pressure which trends toward the equilibrium value where the SI flowrate equals the break flowrate.

Since offsite power is assumed lost at reactor trip, the RCPs trip, and a gradual transition to natural circulation flow occurs. Immediately following reactor trip, the temperature differential across the core decreases as core power decays (see figures 15.6.3-4 and 15.6.3-5); however, the temperature differential subsequently increases as natural circulation flow develops. The cold leg temperatures trend toward the steam generator temperature as the fluid residence time in the tube region increase. The intact steam generator loop temperatures continue to slowly decrease due to the continued AFW flow until operator actions are taken to control the AFW flow to maintain the specified level in the intact steam generators. The ruptured steam generator loop temperatures also continue to slowly decrease until the ruptured steam generator was isolated and the ARV was assumed to fail open.

#### **Major Operator Actions**

1. Identify and Isolate the Ruptured Steam Generator

The ruptured steam generator was assumed to be identified and the MSIV isolated at 20 minutes after the initiation of the SGTR or when the narrow range level recovers to 30 percent, whichever time is greater. Isolation of AFW flow to the ruptured steam generator takes place 4 minutes after event initiation. The ruptured steam generator ARV was assumed to fail open at the time of MSIV isolation, and the failure was simulated at 1202 seconds. The failure causes the ruptured steam generator to rapidly depressurize as shown in figure 15.6.3-3, which results in an increase in primary to secondary leakage. The depressurization of the ruptured steam generator increases the break flow and energy transfer from primary to secondary, which results in a decrease in the ruptured loop temperatures, as shown in figure 15.6.3-5. As noted previously, the intact steam generator loop temperatures also decrease, as shown in figure 15.6.3-4, until the AFW flow to the intact steam generators is throttled. After this time, the heat transfer to the intact steam generators decreases and the temperature differential across the intact steam generators decreases. The decrease in the RCS temperatures results in an initial decrease in the pressurizer level and RCS pressure. However, the increased SI flow subsequently causes the pressurizer level and RCS pressure to increase again, as shown in figures 15.6.3-1 and 15.6.3-2, respectively. It was assumed that the time required for the operator to identify that the ruptured steam generator ARV is open and to locally close the associated block valve is 16 minutes. Thus, at 2162 s the depressurization of the ruptured steam generator was terminated.

2. Cooldown the RCS to establish Subcooling Margin

After the ruptured steam generator ARV block valve was closed, a 12-minute operator action time was imposed prior to initiation of cooldown. The depressurization of the ruptured steam generator affects the RCS cooldown target temperature since the temperature is dependent upon the pressure in the ruptured steam generator. Since offsite power was lost, the RCS was cooled by dumping steam to the atmosphere using the two available, intact steam generator ARVs. The cooldown was continued until RCS cooldown target was met. Because of the lower

#### VEGP-FSAR-15

pressure in the ruptured steam generator, the associated temperature the RCS must be cooled to is also lower, which has the net effect of extending the time for cooldown. The cooldown was initiated at 2882 seconds and was completed at 3264 seconds.

The reduction in the intact steam generator pressures required to accomplish the cooldown is shown in figure 15.6.3-3, and the effect of the cooldown on the RCS temperature is shown in figure 15.6.3-4. The pressurizer level and RCS pressure also decrease during this cooldown process due to shrinkage of the reactor coolant, as shown in figures 15.6.3-1 and 15.6.3-2.

3. Depressurize to Restore Inventory

After the RCS cooldown, a 5-min and 45-s operator action time was assumed prior to depressurization. The RCS was depressurized at 3612 s to assure adequate coolant inventory prior to terminating SI flow. With the RCPS stopped, normal pressurizer spray is not available and, thus, the RCS was depressurized by opening a pressurizer PORV. The depressurization was continued until any of the following conditions were satisfied: RCS pressure is less than the ruptured steam generator pressure and pressurized level is greater than 5 percent or RCS pressure is within 300 psi of the ruptured steam generator pressure and pressurizer level is greater than 50 percent, or pressurizer level is greater than 75 percent, or RCS subcooling is less than the 22 °F allowance for subcooling uncertainty. The effect of the RCS depressurization on the RCS pressure and the differential pressure between the RCS and the ruptured steam generator is shown in figures 15.6.3-2 and 15.6.3-6. RCS depressurization reduces the break flow as shown in figure 15.6.3-1.

4. Terminate SI to Stop Primary to Secondary Leakage

The previous actions have established adequate RCS subcooling, verified a secondary side heat sink, and restored the reactor coolant inventory to ensure that SI flow is no longer needed. When these actions have been completed, the SI flow must be stopped to prevent repressurization of the RCS and to terminate primary to secondary leakage. The SI flow is terminated at this time if RCS subcooling is greater than the 22°F allowance for uncertainty, minimum AFW flow is available or at least one intact steam generator level is in the narrow range, the RCS pressure is increasing, and the pressurizer level is greater than the 5-percent allowance for uncertainty. To assure that the RCS pressure is increasing, SI was not terminated until the RCS pressure increases by at least 50 psi.

After depressurization was completed, an operator action time of 3 minutes was assumed prior to SI termination. Since the above requirements are satisfied, SI termination was performed at this time. After SI termination, the RCS pressure decreases as shown in figure 15.6.3-2. The differential pressure between the RCS and the ruptured steam generator also decreases as shown in figure 15.6.3-6. Figure 15.6.3-7 shows that the primary to secondary leakage continues after the SI flow is stopped until the RCS and ruptured steam generator pressures equalize.

The ruptured steam generator water volume is shown in figure 15.6.3-8. It is noted that the water volume in the ruptured steam generator is significantly less than the total steam generator volume of 5837 ft<sup>3</sup> when the break flow is terminated. The mass of water in the ruptured steam generator is also shown as a function of time in figure 15.6.3-9.

#### Mass Releases

The mass releases were determined for use in evaluating the exclusion area boundary and low population zone radiation exposure. The steam releases from the ruptured and intact steam generators, the feedwater flows to the ruptured and intact steam generators, and primary to secondary break flow into the ruptured steam generator were determined for the period from accident initiation until 2 hours after the accident and from 2 to 20 hours after the accident. The releases for 0 to 2 hours were used to calculate the radiation doses at the exclusion area boundary for a 2-hour exposure, and the releases for 0 to 20 hours were used to calculate the radiation doses at the low population zone for the duration of the accident.

The operator actions for the SGTR recovery up to the termination of primary to secondary leakage were simulated in the LOFTTR2 analysis. Thus, the steam releases from the ruptured and intact steam generators, the feedwater flows to the ruptured and intact steam generators, and the primary to secondary leakage into the ruptured steam generator were determined from the LOFTTR2 results for the period from the initiation of the accident until the leakage is terminated.

Following the termination of leakage, it was assumed that the actions are taken to cool down the plant to cold shutdown conditions. The ARVs for the intact steam generators were assumed to be used to cool down the RCS to the RHR system operating temperature of 350°F, at the maximum allowable cooldown rate of 100°F/h. The steam releases and the feedwater flows for the intact steam generators for the period from leakage termination until 2 hours were determined from a mass and energy balance using the calculated RCS and intact steam generator conditions at the time of leakage termination and at 2 hours. The RCS cooldown was assumed to be continued after 2 hours until the RHR system in-service temperature of 350°F is reached. Depressurization of the ruptured steam generator was then assumed to be performed to the RHR in-service pressure of 350 psia via steam release from the ruptured steam generator ARV. The RCS pressure was also assumed to be reduced concurrently as the ruptured steam generator is depressurized. It was assumed that the continuation of the RCS cooldown and depressurization to RHR operating conditions are completed within 20 hours after the accident since there is ample time to complete the operations during this time period. The steam releases and feedwater flows from 2 to 20 hours were determined for the intact and ruptured steam generators from a mass and energy balance using the conditions at 2 hours and at the RHR system in-service conditions.

After 20 hours, it was assumed that further plant cooldown to cold shutdown as well as longterm cooling is provided by the RHR system. Therefore, the steam releases to the atmosphere were terminated after RHR in-service conditions were assumed to be reached at 20 hours.

For the time period from initiation of the accident until leakage termination, the releases were determined from the LOFTTR2 results for the time prior to reactor trip and following reactor trip. Since the condenser is in service until reactor trip, any radioactivity released to the atmosphere prior to reactor trip would be through the condenser air ejector. After reactor trip, the releases to the atmosphere were assumed to be via the steam generator ARVs. The mass release rates to the atmosphere from the LOFTTR2 analysis are presented in figures 15.6.3-10 and 15.6.3-11 for the ruptured and intact steam generators, respectively, for the time period until leakage termination. The mass releases calculated from the time of leakage termination until 2 hours and from 2 to 20 hours are also assumed to be released to the atmosphere via the steam generator ARVs. The mass release for the SGTR event for the 0 to 2-hour and 2 to 20-hour time intervals are presented in table 15.6.3-3.

## 15.6.3.4 Offsite Radiation Dose Analysis

The evaluation of the radiological consequences of a steam generator tube rupture event assumes that the reactor has been operating at the maximum allowable Technical Specification limit for primary coolant activity and primary to secondary leakage for sufficient time to establish equilibrium concentrations of radionuclides in the reactor coolant and in the secondary coolant. Radionuclides from the primary coolant enter the steam generator via the ruptured tube and are released to the atmosphere through the steam generator PORVs (and safety valves) and via the condenser air ejector exhaust.

The quantity of radioactivity released to the environment, due to an SGTR, depends upon primary and secondary coolant activity, iodine spiking effects, primary to secondary break flow, break flow flashing fractions, attenuation of iodine carried by the flashed portion of the break flow, partitioning of iodine between the liquid and steam phases, the mass of fluid released from the generator, and liquid-vapor partitioning in the turbine condenser hot well. All of these parameters were conservatively evaluated in a manner consistent with the recommendations in Standard Review Plan 15.6.3.

1. Design Basis Analytical Assumptions

The major assumptions and parameters used in the analysis are itemized in table 15.6.3-4.

2. Source Term Calculations

The radionuclide concentrations in the primary and secondary system, prior to and following the SGTR, are determined as follows:

- a. The iodine concentrations in the reactor coolant will be based upon preaccident and accident-initiated iodine spikes.
  - i. Accident-Initiated Spike The initial primary coolant iodine concentration is 1  $\mu$ Ci/g of Dose Equivalent (D.E.) I-131. Following the primary system depressurization associated with the SGTR, an iodine spike is initiated in the primary system which increases the iodine release rate from the fuel to the coolant to a value 500 times greater than the release rate corresponding to the initial primary system iodine concentration.
  - ii. Preaccident Spike A reactor transient has occurred prior to the SGTR and has raised the primary coolant iodine concentration from 1 to 60  $\mu$ Ci/gram of D.E. I-131.
- b. The initial secondary coolant iodine concentration is 0.1  $\mu$ Ci/gram of D.E. I-131.
- c. The chemical form of iodine in the primary and secondary coolant is assumed to be elemental.
- d. The initial noble gas concentrations in the reactor coolant are based upon 1-percent fuel defects.
- 3. <u>Dose Calculations</u>

The iodine transport model utilized in this analysis was proposed by Postma and Tam (reference 3). The model considers break flow flashing, steaming, and partitioning. The model assumes that a fraction of the iodine carried by the break flow becomes airborne immediately due to flashing and atomization. The fraction of primary coolant iodine which is not assumed to become airborne immediately mixes

with the secondary water and is assumed to become airborne at a rate proportional to the steaming rate and the iodine partition coefficient. This analysis conservatively assumes an iodine partition coefficient of 100 between the steam generator liquid and steam phases. Droplet removal by the dryers is conservatively assumed to be negligible. The iodine transport model is illustrated in figure 15.6.3-12.

The offsite radiological analysis did not consider steam generator tube uncovery in the calculation, since the steam generator tube uncovery issue was investigated and closed by the Westinghouse Owners Group (WOG). Reference 7 documents the Westinghouse position on the issue that the affect of tube uncovery on the limiting SGTR transient is essentially negligible and need not be considered in the analysis. Reference 8 documents the NRC agreement on this issue.

The following assumptions and parameters were used to calculate the activity released to the atmosphere and the offsite doses following an SGTR.

- a. The mass of reactor coolant discharged into the secondary system through the rupture and the mass of steam released from the ruptured and intact steam generators to the atmosphere are presented in table 15.6.3-3.
- b. The time dependent fraction of rupture flow that flashes to steam and is immediately released to the environment is presented in figure 15.6.3-13. The break flow flashing fraction was conservatively calculated assuming that 100 percent of the break flow comes from the hot leg side of the steam generator, whereas the break flow actually comes from both the hot leg and the cold leg sides of the steam generator.
- c. The total primary to secondary leak rate is assumed to be 1.0 gal/min as allowed by the Technical Specifications. The leak rate is assumed to be 0.70 gal/min to the three intact steam generators and 0.30 gal/min to the ruptured steam generator.
- d. The iodine partition factor between the liquid and steam of the ruptured and intact steam generators is assumed to be 100.
- e. No credit was taken for radioactive decay during release and transport, or for cloud depletion by ground deposition during transport to the site boundary or outer boundary of the low population zone.
- f. Short-term atmospheric dispersion factors ( $\chi/Qs$ ) for accident analysis and breathing rates are provided in table 15.6.3-8 and table 15A-2. The breathing rates were obtained from NRC Regulatory Guide 1.4, (reference 4).

4. Offsite Thyroid Dose Calculation Model

Offsite thyroid doses are calculated using the equation:

$$D_{Th} = \sum_{i} \left[ DCF_{i} \left( \sum_{j} (IAR)_{IJ} (BR)_{j} \left( \frac{\chi}{Q} \right)_{j} \right) \right]$$

where:

- $(IAR)_{ij} =$ integrated activity of iodine nuclide i released during the time interval j in  $Ci^{(a)}$
- $(BR)_j$  = breathing rate during time interval j in m<sup>3</sup>/s (table 15.6.3-8)
- $(\chi/Q)_j$  = atmospheric dispersion factor during time interval j in s/m<sup>3</sup> (table 15A-2)

$$(DCF)_i =$$
thyroid dose conversion factor via inhalation for iodine nuclide i in rem/Ci (table 15A-5)

D<sub>Th</sub> = Thyroid dose via inhalation in rem

Offsite whole-body gamma doses are calculated using the equation:

$$D\gamma = \sum_{i} \left[ DCF_{\gamma i} \left( \sum_{j} (IAR)_{ij} \left( \frac{\chi}{Q} \right)_{j} \right) \right]$$

where:

 $(IAR)_{ij}$  = integrated activity of noble gas nuclide i released during time interval j in Ci<sup>(a)</sup>

 $(\chi/Q)_j$  = atmospheric dispersion factor during time interval j in seconds/m<sup>3</sup>

 $DCF_{\gamma i}$  = whole body dose conversion factor via submersion for noble gas nuclide i in rem-m<sup>3</sup>/Ci-sec (table 15.6.3-10)

 $D\gamma$  = whole body gamma dose due to immersion in rem

Offsite beta-skin doses are calculated using the equation:

$$\mathsf{D}\beta = \sum_{i} \left[ \mathsf{D}\mathsf{C}\mathsf{F}_{\beta i} \left( \sum_{j} (\mathsf{I}\mathsf{A}\mathsf{R})_{ij} \left( \overset{\chi}{\nearrow}_{\mathsf{Q}} \right)_{j} \right) \right]$$

where:

 $(IAR)_{ij} =$  integrated activity of noble gas nuclide i released during time interval j in Ci<sup>(a)</sup>

$$\begin{pmatrix} \chi / \\ Q \end{pmatrix}_j$$
 = atmospheric dispersion factor during time interval j in seconds/m<sup>3</sup>

 $DCF_{\beta i}$  = beta skin dose conversion factor via submersion for noble gas nuclide i in rem-m<sup>3</sup>/Ci-sec (table 15.6.3-10)

 $D\beta$  = beta-skin dose due to immersion in rem

5. <u>Results</u>

Thyroid, whole-body gamma, and beta-skin doses at the Exclusion Area Boundary and Low Population Zone are presented in table 15.6.3-11. All doses are well within the allowable guidelines as specified by Standard Review Plan 15.6.3 and 10 CFR 100.

<sup>&</sup>lt;sup>(a)</sup> No credit is taken for cloud depletion by ground deposition or by radioactive decay during transport to the exclusion area boundary or to the outer boundary of the low-population zone.

#### 15.6.3.5 <u>References</u>

- 1. Westinghouse Electric Company, "SGTR Reanalysis Results for Revised Operator Action Times," <u>SECL 98-124</u>, Rev. 1. Letter from J. L. Tain to J. B. Beasley dated July 20, 1999, GP-16989.
- 2. Southern Nuclear Operating Company, Inc., <u>ELV-06244-A</u>, memo from L. A. Ward (Southern Nuclear) to J. L. Tain (W) dated August 18, 1998.
- 3. Postma, A. K. and Tam, P. S., "Iodine Behavior in a PWR Cooling System Following a Postulated Steam Generator Tube Rupture," <u>NUREG-0409</u>.
- 4. <u>NRC Regulatory Guide 1.4</u>, Rev. 2, "Assumptions Used for Evaluating the Potential Radiological Consequences of a LOCA for Pressurized Water Reactors," June 1974.
- 5. International Commission on Radiological Protection, "Limits for Intakes of Radionuclides by Workers," <u>ICRP Publication 30</u>, Volume 3, No. 1-4, 1979.
- 6. Bell, M. J., "ORIGEN The ORNL Isotope Conversion and Depletion Code," <u>ORNL-</u> <u>4628</u>, 1973.
- 7. <u>WOG-92-25</u>, "Westinghouse Owners Group Steam Generator Tube Uncovery Issue," March 31, 1992.
- Attachment to <u>WOG-93-066</u>, letter to Lawrence A. Walsh (Chairman WOG) from Robert C. Jones (NRC), "Westinghouse Owners Group - Steam Generator Tube Uncovery Issue," March 10, 1993.
- Environmental Protection Agency, "Limiting Values of Radionuclide Intake and Air Concentration and Dose Conversion Factors for Inhalation, Submersion, and Ingestion," <u>Federal Guidance Report 11</u>, EPA 520/1-88-020, 1988.
- 10. Environmental Protection Agency, "External Exposure to Radionuclides in Air, Water, and Soil," <u>Federal Guidance Report 12</u>, EPA 402-R-93-081, 1993.
- 11. Westinghouse Electric Company, "Vogtle Units 1 and 2 SGTR Margin to Overfill Report to Address NSAL-07-11, "LTR-CRA-16-132, Rev. 1, December 21, 2016, GP-19499.
- 12. Westinghouse Electric Company, "Report for Vogtle Units 1 and 2 Steam Generator Tube Rupture Input to Dose and Offsite Dose Analyses," LTR-CRA-19-39, Rev. 0.
- 13. Southern Nuclear Operating Company, Inc., SA-19-001, memo from J. Douglass (Southern Nuclear) to K. W. Bonadio (W) dated February 28, 2019.

#### 15.6.4 SPECTRUM OF BOILING WATER REACTOR STEAM SYSTEM PIPING FAILURES OUTSIDE OF CONTAINMENT

This subsection is not applicable to VEGP.

## 15.6.5 LOSS-OF-COOLANT ACCIDENTS

#### 15.6.5.1 Identification of Causes and Frequency Classification

A loss-of-coolant accident (LOCA) is the result of a pipe rupture of the reactor coolant system (RCS) pressure boundary. For the analyses reported here, a major pipe break (large break) is

defined as a rupture with a total cross-sectional area equal to or greater than 1.0 ft<sup>2</sup>. This event is considered a limiting fault, an American Nuclear Society (ANS) Condition IV event, in that it is not expected to occur during the lifetime of the plant but is postulated as a conservative design basis.

For large-break LOCAs, the most limiting single failure is the loss of one train of emergency core cooling system (ECCS) injection. The large-break LOCA analyses assume both maximum containment safeguards (to analyze lowest containment pressure conditions) and minimum ECCS safeguards (to analyze the loss of one complete train of ECCS components), which results in the minimum delivered ECCS flow available to the RCS.

A minor pipe break (small break), as considered in this subsection, is defined as a rupture of the reactor coolant pressure boundary with a total cross-sectional area less than 1.0 ft<sup>2</sup>, in which the normally operating charging system flow is not sufficient to sustain pressurizer level and pressure. This is considered a Condition III event in that it is an infrequent fault that may occur during the life of the plant.

For small-break LOCAs, the most limiting single active failure is of an emergency power train which results in loss of one complete train of ECCS components. The minimum delivered ECCS flow available to the RCS is based on this single failure.

The acceptance criteria for the LOCA described in 10 CFR 50.46<sup>(1)</sup> are met as follows:

- A. The calculated maximum fuel element cladding temperature is below the requirement of 2200 °F.
- B. The local oxidation of the cladding does not exceed 0.17 times the thickness before oxidation.
- C. The calculated total amount of hydrogen generated from the chemical reaction of the cladding with water or steam does not exceed 0.01 times the hypothetical amount that would be generated if all the fuel cladding metal, excluding the cladding surrounding the plenum volume, were to react.
- D. Calculated changes in core geometry are such that the core remains amenable to cooling.
- E. The core temperature is reduced and decay heat is removed for an extended period of time, as required by the long-lived radioactivity remaining in the core.

These criteria were established to provide a significant margin in ECCS performance following a LOCA. Reference 2 presents a recent study in regards to the probability of occurrence of RCS pipe ruptures.

In all cases, small breaks (less than 1.0 ft<sup>2</sup>) yield results with more margin to the acceptance criteria limits than the limiting large break.

## 15.6.5.2 Sequence of Events and Systems Operations

Should a major break occur, depressurization of the RCS results in a pressure decrease in the pressurizer. Loss-of-offsite power (LOOP) is assumed coincident with the occurrence of the break. The reactor trip signal subsequently occurs when the pressurizer low-pressure trip setpoint is reached. A safety injection signal is generated when the appropriate setpoint is reached. These countermeasures will limit the consequences of the accident in two ways:

A. Reactor trip and borated water injection complement void formation in causing rapid reduction of power to a residual level corresponding to fission product

decay heat. However, no credit is taken in the LOCA ECCS thermal analysis for boron content of the injection water. In addition, the insertion of control rods to shut down the reactor is neglected in the large-break ECCS thermal analysis.

B. Injection of borated water provides for heat transfer from the core and prevents excessive cladding temperatures.

In the present Westinghouse design, the large-break single failure is the loss of an entire train of ECCS components (i.e., one high head charging pump, one safety injection pump, and one low head pump).

The small-break single failure is the loss of one ECCS train. This means that for a small break, credit could be taken for one high head charging pump, one safety injection pump, and one low head pump, though low head flow was neglected in the small-break analysis for VEGP, since the transient is terminated before the cut-in pressure is reached.

The current design for both small and large breaks assumes that at least one train is available for delivery of water to the RCS. This means that one pump in each subsystem delivers to the primary loop.

For the large-break analysis, one ECCS train starts and delivers flow through the injection lines (one for each loop) with one branch injection line spilling to the containment backpressure. To minimize delivery to the reactor, the branch line chosen to spill is selected as the one with the minimum resistance.

For the small-break analysis, one ECCS train starts and delivers flow through the injection lines (one for each loop) with one branch injection line spilling to the RCS backpressure. To minimize delivery to the reactor, the branch line chosen to spill is the one with the minimum resistance.

## 15.6.5.2.1 Description of Large-Break LOCA Transient

The sequence of events following a large-break LOCA are presented in figure 15.6.5-1.

Before the break occurs, the unit is in an equilibrium condition; i.e., the heat generated in the core is being removed via the secondary system. During blowdown, heat from fission product decay, hot internals, and the vessel continues to be transferred to the reactor coolant. At the beginning of the blowdown phase, the entire RCS contains subcooled liquid which transfers heat from the core by forced convection with some fully developed nucleate boiling. Thereafter, the core heat transfer is based on local conditions with transition boiling and forced convection to steam as the major heat transfer mechanisms.

The heat transfer between the RCS and the secondary system may be in either direction depending on the relative temperatures. In the case of continued heat addition to the secondary system, the secondary system pressure increases and the main steam safety valves may actuate to limit the pressure. Makeup water to the secondary side is automatically provided by the auxiliary feedwater system. The safety injection signal actuates a feedwater isolation signal which isolates main feedwater flow by closing the main feedwater isolation valves and also initiates auxiliary feedwater flow by starting the auxiliary feedwater pumps. The secondary flow aids in the reduction of RCS pressure.

When the RCS depressurizes to 611.3 psia, the accumulators begin to inject borated water into the reactor coolant loops. Since LOOP is assumed, the reactor coolant pumps are assumed to trip at the inception of the accident. The effects of pump coastdown are included in the blowdown analysis.

The blowdown phase of the transient ends when the RCS pressure (initially assumed at 2300 psia) falls to a value approaching that of the containment atmosphere. Prior to or at the end of the blowdown, the mechanisms that are responsible for the emergency core cooling water bypassing the core are calculated not to be effective. At this time (end of bypass) refill of the reactor vessel lower plenum begins. Refill is complete when emergency core cooling water has filled the lower plenum of the reactor vessel which is bounded by the bottom of the fuel rods (bottom of core recovery time).

The reflood phase of the transient is defined as the time period lasting from the bottom of recovery until the reactor vessel is filled with water to the extent that the core temperature rise is terminated. From the latter stage of blowdown to early reflood, the safety injection accumulator tanks rapidly discharge borated cooling water into the RCS, contributing to the filling of the reactor vessel downcomer. The downcomer water elevation head provides the driving force required for the reflooding of the reactor core. The RHR (low head), safety injection (intermediate head), and high head charging pumps also aid the filling of the downcomer and subsequently supply water to maintain a full downcomer and complete the reflooding process.

Continued operation of the ECCS pumps supplies water during long-term cooling. By this time, core temperatures are reduced to long-term, steady-state levels associated with the dissipation of residual heat generation. After the water level of the refueling water storage tank (RWST) reaches a minimum allowable value, coolant for long-term cooling of the core is obtained by switching to the cold leg recirculation mode of operation, in which spilled borated water is drawn from the containment emergency sumps and returned to the RCS cold legs. The containment spray pumps are manually aligned to the containment emergency sumps and continue to operate to further reduce containment pressure.

Approximately 7.5 hours after initiation of the LOCA, the ECCS is realigned to supply water to the RCS hot legs to control the boric acid concentration in the reactor vessel.

#### 15.6.5.2.2 Description of Small-Break LOCA Transient

As contrasted with the large break, the blowdown phase of the small break occurs over a longer time period. Thus, for the small-break LOCA there are only three characteristic stages; i.e., a gradual blowdown with a decrease in water level and a partial core uncovery, core recovery, and long-term recirculation.

Should a small break occur, depressurization of the RCS causes fluid to flow into the loops from the pressurizer resulting in a pressure and level decrease in the pressurizer. Reactor trip occurs when the low pressurizer pressure trip setpoint is reached. During the earlier part of the small-break transient, the effect of the break flow is not strong enough to overcome the flow maintained by the reactor coolant pumps through the core as they are coasting down following reactor trip. Due to the LOOP assumption, the reactor coolant pumps are assumed to be tripped coincident with reactor trip during the accident. Upward flow through the core is maintained. However, the core flow is not sufficient to prevent a partial core uncovery. The ECCS is actuated when the appropriate setpoint is reached and provides sufficient core flow to recover the core.

Before the break occurs the plant is in an equilibrium condition; i.e., the heat generated in the core is being removed via the secondary system. During blowdown, heat from fission product decay, hot internals, and the vessel continues to be transferred to the RCS. The heat transfer between the RCS and the secondary system may be in either direction depending on the relative temperatures. In the case of continued heat addition to the secondary, secondary system pressure increases and steam relief via the atmospheric relief and/or safety valves may

occur. The auxiliary feedwater pumps provide makeup to the secondary side. The reactor trip signal isolates normal feedwater flow by closing the main feedwater isolation valves and initiates auxiliary feedwater flow by starting the auxiliary feedwater pumps. The secondary flow aids in the reduction of RCS pressure.

When the RCS depressurizes to approximately 611.3 psia, the cold leg accumulators begin to inject borated water into the reactor coolant loops. For some breaks, the vessel mixture level starts to increase with ECCS pumped injection before the accumulators come on. For the breaks that do reach the accumulator injection setpoint, the accumulation injection provides enough water to bring the mixture level up to the upper plenum region where it is maintained.

#### 15.6.5.3 Core and System Performance

#### 15.6.5.3.1 Mathematical Model

The requirements of an acceptable ECCS evaluation model are presented in Appendix K of 10 CFR 50.(1)

15.6.5.3.1.1 <u>Large-Break LOCA Evaluation Model</u>. The large break analysis was performed with the 1981 version of the Westinghouse ECCS evaluation model using BASH (reference 3), including the changes in the methodology for execution of the model which are described in references 4 and 5. The BASH evaluation model for dry containment plants includes the following main computer codes:

- SATAN Blowdown thermal-hydraulics code;
- BASH Refill and reflood thermal-hydraulics code;
- COCO Containment backpressure code; and
- LOCBART Rod temperature and blockage code.

A brief summary of each of these codes is presented in the following paragraphs.

SATAN (reference 6) is a one-dimensional nodal network code which models the thermalhydraulic phenomena during the blowdown depressurization in the reactor core and RCS after a postulated large rupture of a primary coolant pipe. It was developed specifically as part of the evaluation model that meets 10 CFR 50 Appendix K requirements. The code provides blowdown thermal and hydraulic parameters that define the heat transfer boundary conditions in the LOCBART code, which is used to calculate the hot assembly and hot rod fuel cladding temperature transients during a LOCA. SATAN also provides mass and energy discharge rates from the RCS to containment for the COCO code, which is used for containment backpressure calculations.

Some specific features of the SATAN code include the use of a drift flux model and the use of a two-phase friction multiplier. In the core, a hot channel and an average channel flow calculation, effects of crossflow between channels, cladding swelling and rupture effects, and metal-water reaction effects are considered. In the rest of the primary loop, accumulator bypass effects and a two-phase pump model are included.

SATAN begins calculation of the transient at the time of the rupture. Calculations continue until the end of blowdown which is determined when the following two conditions are met:

- 1. Downflow of ECCS water into the reactor vessel lower plenum is established, and
- 2. The RCS pressure is equalized with the containment pressure.

When both of these conditions are met, SATAN will terminate and the refill phase calculations will begin.

The BASH code (reference 3) is used to calculate the refill and reflood portions of the large break LOCA transient. The REFILL module in the BASH code contains the thermal-hydraulic models that are used to describe the storage and transport of water from the ECCS injection points to the reactor vessel lower plenum. The only regions modeled in REFILL are the lumped intact loop cold legs, broken loop cold leg stub, reactor vessel downcomer, and lower plenum up to the bottom of the active fuel. The ECCS systems, including the accumulators, are also modeled. REFILL obtains the conditions from the SATAN tape at the end of blowdown which is used to initialize the refill portion of the transient. REFILL models the cold leg transit and fill, hot wall delay due to flashing of ECCS water in the downcomer, freefall of the water in the downcomer, and lower plenum fill. The refill stage of the LOCA ends when bottom-of-core recovery occurs. The time to bottom-of-core recovery is the total amount of time required for the ECCS to increase the water level in the reactor vessel lower plenum to the bottom of the active fuel.

When bottom-of-core recovery occurs, the reflood phase calculations in BASH begin. The BASH code consists of the BART code (reference 7) for core thermal-hydraulic and heat transfer calculations, and a modified version of the NOTRUMP code (reference 8) for the RCS transient response calculations. The BART computer code was developed primarily as a best-estimate design code for application to the reflood stage of LOCA analysis, and the basic features are described in reference 7. Some specific features of BART are as follows:

- 1. Conservation of mass and energy in liquid, vapor, and two-phase regions in the reactor core.
- 2. Radial conduction heat transfer within the fuel rod.
- 3. Heat exchange between rods and coolant in liquid, vapor, and two-phase regions in the core.
- 4. Quench-front propagation and heat release.
- 5. Thermal nonequilibrium and heat transfer between phases.

The loop models and equations are based on the equilibrium version of the NOTRUMP code, which is used for a variety of applications. The main code components are fluid nodes, metal nodes, flow links, and heat links. Physical problems are modeled by using the components to form a network of multiple fluid and metal nodes, appropriately interconnected by flow and heat links. The nodes provide for mass and energy storage; the links provide for mass, energy, and momentum transfer. Thermal-hydraulic effects in the RCS during core reflooding are modeled in the code. Flow correlations model the effects of pressure drop and phase separation. Heat transfer correlations represent all regimes from liquid convection, through nucleate and transition boiling, to stable film boiling or forced convection vaporization, and finally to steam forced convection.

BART, with numerical modifications and with changes to some of the physical models, was combined with the NOTRUMP code described above, creating BASH. BASH calculates the reflood rate which is input to LOCBART, which then calculates the hot assembly and hot rod thermal transient performance during reflood.

The BASH code is also used to provide the mass and energy discharge rates from the RCS to the containment during reflood, which are used in the COCO code for the containment backpressure calculations.

The COCO code (reference 9) models the containment behavior for dry containment plants during a large break LOCA transient. The code calculates the pressure and temperature transients inside the containment during the depressurization and post-blowdown phases following a LOCA.

A detailed examination is made of the nonlinear physical phenomena occurring within the containment during the transient. Transient conditions are determined for both the containment steam-air atmosphere and the containment sump water. Temperature gradients in and heat absorption by the containment structures are also considered. The code has the flexibility to analyze various safeguards systems, including internal and external sprays, containment venting and pump back, ventilation fan coolers, and a sump water recirculation system. In the current version of the BASH evaluation model, the COCO code is run interactively in the BASH code, to provide direct feedback between the containment and RCS during the refill and reflood phases of the transient.

The LOCBART code calculates fuel rod temperature profiles, cladding burst, and cladding oxidation during the accident sequence. LOCBART is a combination of the LOCTA-IV code (reference 10) and the BART code (reference 7). The heat transfer regimes which are analyzed by the LOCTA-IV code include single-phase convective cooling, nucleate boiling, transition boiling, stable film boiling, and heat transfer to steam using laminar or turbulent heat transfer film coefficients. The effects of fuel rod-to-coolant radiation and rod-to-rod radiation are considered within the program. Heat transfer coefficients are computed for each axial increment on the basis of local coolant flows, qualities, and temperatures.

After the blowdown phase of the accident, the ECCS delivers water to the RCS which ultimately fills the reactor vessel lower plenum. During this refill phase of the accident, only rod-to-rod radiation heat transfer is considered in LOCTA-IV. When the lower plenum is full, water begins to enter the core region, and the reflood phase of the accident begins. During the reflooding period, the core is cooled by a two-phase mixture that results from steam generation and droplets entrained leaving the flooded region of the core. In the two-phase period, the heat transfer coefficients are calculated by BART using rigorous mechanistic models. When the reflooding rate is less than 1 in./s, the heat transfer coefficient is calculated based on a steam cooling assumption.

During the blowdown and refill phases of a LOCA transient, the LOCTA-IV part of LOCBART is used to calculate the average fuel temperatures. The required mass flow, pressure, and enthalpy information to the fuel rod code during blowdown is provided by SATAN output. During refill, the rod-to-fluid heat transfer coefficient is conservatively assumed to be zero, which results in essentially adiabatic conditions. LOCBART also calculates the thermal-hydraulic conditions in the hot assembly during reflood using the flooding rate obtained from the BASH code. The complete LOCBART code is used to calculate the hot rod temperature during the blowdown, refill, and reflood phases.

15.6.5.3.1.2 <u>Small-Break LOCA Evaluation Model.</u> The small break analysis was performed with the Westinghouse ECCS evaluation model using NOTRUMP (references 8 and 16), including changes to the model and methodology as described in references 17 and 20. The NOTRUMP evaluation model includes the following computer codes:

• NOTRUMP - Thermal-hydraulic response of RCS during transient;

• SBLOCTA - Fuel rod/cladding heatup during transient.

A brief summary of each of these codes is presented in the following paragraphs.

The NOTRUMP computer code is a one-dimensional general thermal-hydraulic network code consisting of a number of features. Among these features are the calculation of thermal nonequilibrium in all fluid volumes, flow regime-dependent drift flux calculations with counter-current flooding limitations, mixture level tracking logic in multiple-stacked fluid nodes, and regime-dependent heat transfer correlations. Heat transfer in the core is calculated based on LOCTA-IV code (reference 18) which considers heat transfer regimes, including single-phase convection to subcooled liquid, nucleate boiling, transition boiling, film boiling, and convection to superheated vapor in both laminar and turbulent flows. In addition, the NOTRUMP small break loss-of-coolant accident emergency core cooling system evaluation model (NOTRUMP SBLOCA ECCS EM) was developed to determine the RCS response to design basis small break LOCAs and to address the NRC concerns expressed in NUREG-0611.

The NOTRUMP model is generally made up of nodes (control volumes) and links (mass and energy transport). This model determines the thermal-hydraulic response of the RCS during the SBLOCA transient based upon initial operating conditions, core nuclear design parameters, and steam generator and reactor vessel characteristics, as well as ECCS performance and other plant specific parameters. Select RCS response boundary conditions are extracted from the NOTRUMP calculations and are used in the SBLOCTA fuel rod heatup code.

SBLOCTA is a small-break specific version of the LOCTA-IV code (reference 18). Peak cladding temperature (PCT) calculations are performed with the LOCTA-IV code using the NOTRUMP calculated core pressure, fuel rod power history, uncovered core steam flow, and mixture levels as boundary conditions. LOCTA-IV models the hot rod and the average hot assembly rod. As stated above, LOCTA-IV contains many heat transfer models; however, due to the relatively low velocities experienced in the core during the SBLOCA transient, heat transfer is basically limited to forced convection to super-heated vapor and rod-to-rod radiation. In addition to PCT, SBLOCTA also calculates maximum local and hot rod axial average ZrO<sub>2</sub> reaction.

#### 15.6.5.3.2 Input Parameters and Initial Conditions

The large break LOCA analysis for Units 1 and 2 was performed for plant operation at 3565 MWt (plus 2% uncertainty, which bounds the current rated thermal power level of 3625.6 MWt) with 10% steam generator tube plugging. The analysis conditions are based on a total RCS thermal design flow of 93,600 gpm/loop which is consistent with 10% steam generator tube plugging. The analysis was performed with the upper head fluid temperature equal to the RCS cold leg fluid temperature, achieved by increasing the upper head cooling flow.

The effects of a combined LOCA/SSE event were explicitly accounted for in the large break LOCA analysis via an increase in the modeled steam generator tube plugging level. The steam generator tube plugging adder was conservatively assumed to be 2.5% (reference 15). The key parameters which were used in the large break LOCA analysis are summarized in table 15.6.5-1.

The small break LOCA analysis for Units 1 and 2 was also performed for plant operation at 3565 MWt (plus 2% uncertainty, which bounds the current rated thermal power level of 3625.6 MWt) with 10% steam generating tube plugging. The analysis conditions are based on a total RCS thermal design flow of 93,600 gpm/loop. A summary of the key parameters used in the small break LOCA analysis is presented in table 15.6.5-5.

The SBLOCA analysis imposed no restriction on the rate of return-to-power.

The integrated fuel burnable absorber (IFBA) fuel analysis was performed at 0 MWD/MTU, meaning the LBLOCA analysis imposes no restriction on the rate of return-to-power following an outage. Also, the analysis is considered to be insensitive to loop flow variations representative of normal plant operation.

The initial steady-state fuel pellet temperatures and the fuel rod internal pressures used in the LOCA analysis were generated with the PAD 4.0 Fuel Rod Design Code (reference 11), which was approved by the Nuclear Regulatory Commission.

Reference 23 concluded that the LOCA ZIRLO<sup>®</sup> models are acceptable for application to Optimized ZIRLO<sup>™</sup> cladding in the large break and small break analyses, and that no additional calculations are necessary for evaluating the use of Optimized ZIRLO cladding provided plant specific ZIRLO calculations were previously performed. An exception to this is identified in the Safety Evaluation Report (SER) of reference 23 for plants licensed with the Westinghouse Appendix K LBLOCA code LOCBART with a blowdown or early reflood PCT; however, an evaluation performed for the implementation of Optimized ZIRLO cladding concluded that ZIRLO models are acceptable for application to Optimized ZIRLO cladding for Vogtle 1 and 2 and no additional calculations are necessary. Further, for the LBLOCA cases analyzed, PCT does not occur during blowdown or early reflood.

The bases used to select the numerical values that are input parameters to the analysis were conservatively determined from extensive sensitivity studies.(12)(13)(14) In addition, the requirements of 10 CFR 50 Appendix K (reference 1) regarding specific model features were met by selecting models which provide a significant overall conservatism in the analysis. The assumptions made pertain to the conditions of the reactor and associated safety system equipment at the time that the LOCA occurs and include such items as the core peaking factors, the containment pressure, and the performance of the ECCS. Decay heat generated throughout the transient is also conservatively calculated.

#### 15.6.5.3.3 Results

15.6.5.3.3.1 <u>Large-Break Results</u>. Based on the results of the LOCA sensitivity studies (references 12, 13, and 14), the limiting large break was found to be the double-ended cold leg guillotine (DECLG). Therefore, only the DECLG break is considered in the large-break ECCS performance analysis. Calculations were performed for a range of Moody break discharge coefficients. The results of these calculations are summarized in table 15.6.5-2. (Note that the results in table 15.6.5-2 are obtained directly from the LOCBART output and are based on a T<sub>AVG</sub> window of 573.0 °F to 588.4 °F. An evaluation was performed to extend the low T<sub>AVG</sub> to 570.7 °F.) The time sequence of events for each case is summarized in table 15.6.5-3. Changes to the PCT subsequent to the ECCS model analysis are summarized in table 15.6.5-4.

The mass and energy release data for the break resulting in the highest calculated peak clad temperature are discussed in paragraph 6.2.1.5.

Figures 15.6.5-2 through 15.6.5-16 present the parameters of principal interest from the largebreak ECCS analyses. For all cases, analyzed transients of the following parameters are presented in the following figures:

Figure Parameter

#### VEGP-FSAR-15

<u>Figure</u>	Parameter
15.6.5-2	Cladding temperature at PCT and burst elevations
15.6.5-3	Core pressure during blowdown
15.6.5-4	Vessel liquid levels during reflood
15.6.5-5	Core inlet flooding rate during reflood
15.6.5-6	Normalized core power during blowdown
15.6.5-7	Containment pressure transient
15.6.5-8	Core inlet and outlet mass flow rate during blowdown
15.6.5-9	Cladding surface heat transfer coefficient at PCT and burst elevations
15.6.5-10	Vapor temperature at PCT and burst elevations
15.6.5-11	Break mass flow rate during blowdown
15.6.5-12	Break energy release rate during blowdown
15.6.5-13	Fluid quality at PCT and burst elevations
15.6.5-14	Fluid mass velocity at PCT and burst elevations
15.6.5-15	Intact loop accumulator mass flow rate during blowdown
15.6.5-16	Intact leg accumulator and SI mass flow rate during reflood

The peak cladding temperature calculated by LOCBART for a large break is 2061.6 °F for 156-IFBA, which is less than the acceptance limit of 2200 °F. The maximum local metal-water reaction is below the embrittlement limit of 17% as required by 10 CFR 50.46. The total core metal-water reaction is less than the 1% criterion of 10 CFR 50.46. The cladding temperature transient is terminated at a time when the core geometry is still amenable to cooling. As a result, the core temperature will continue to drop, and the ability to remove decay heat generated in the fuel for an extended period of time will be provided.

15.6.5.3.3.1.1 Additional Analyses and Safety Evaluations. An evaluation was performed to extend the large break LOCA analysis to support operation at PCWG  $T_{AVG}$  values between 570.7 °F and 573.0 °F with up to 10% tube plugging in any or all steam generators. This evaluation results in a PCT increase of about 0.5 °F, for an overall PCT value of 2062.1 °F with 156 IFBA. Note that the 156 IFBA PCT value is considered the "analysis-of-record" PCT value and will be reported as such on the 10 CFR 50.46 rack-up sheet.

The potential limited use of LOPAR fuel in future reloads will be evaluated on a cycle-specific basis.

15.6.5.3.3.2 <u>Small-Break Results</u>. Based on the results of the LOCA sensitivity studies (references 12 and 19), the limiting small break was found to be less than a 10-in. diameter rupture of the RCS cold leg. Therefore, a range of small-break is presented cases which established the limiting small break. The results of these analyses are summarized in tables 15.6.5-6 and 15.6.5-7. Changes to the PCT subsequent to the ECCS model analysis are summarized in table 15.6.5-8.

Figures 15.6.5-22 through 15.6.5-33 present the principal parameters of interest for the smallbreak ECCS analyses. The figures are as follows:

<u>Figure</u>	<u>Parameter</u>
15.6.5-22	2-in., Low $T_{AVG}$ Reactor Coolant System Pressurizer Pressure
15.6.5-23	2-in., Low T <sub>AVG</sub> Core Mixture Level 2 in.
15.6.5-24	3-in., Low $T_{AVG}$ Reactor Coolant System Pressurizer Pressure
15.6.5-25	3-in., Low T <sub>AVG</sub> Core Mixture Level
15.6.5-26	3-in., Low T <sub>AVG</sub> Core Steam Flow Rate
15.6.5-27	3-in., Low T <sub>AVG</sub> Pumped Safety Injection
15.6.5-28	3-in., Low T <sub>AVG</sub> Peak Clad Temperature at 11.25 ft
15.6.5-29	3-in., Low $T_{AVG}$ Hot Spot Fluid Temperature at 11.25 ft
15.6.5-30	3-in., Low $T_{AVG}$ Hot Rod Heat Transfer Coefficient at 11.25 ft
15.6.5-31	4-in., Low $T_{AVG}$ Reactor Coolant System Pressurizer Pressure
15.6.5-32	4-in., Low T <sub>AVG</sub> Core Mixture Level
15.6.5-33	4-in., Low T <sub>AVG</sub> Peak Clad Temperature at 10.75 ft

The PCT calculated for the small-break LOCA is 1138 °F. The maximum local metal-water reaction is below the acceptance criteria limit of 17%. The total core metal-water reaction is less than the 1 percent acceptance criteria. These results are below all acceptance criteria limits of 10 CFR 50.46.

15.6.5.3.3.2.1 Additional Analyses and Safety Evaluations. The potential use of LOPAR fuel in future reloads will be evaluated on a cycle-specific basis.

15.6.5.3.3.2.2 The impact on the SBLOCA analysis of installing a pressure breakdown orifice in the cold leg common header of the HHSI subsystem and one pressure breakdown orifice in each IHSI pump discharge line and replacing HIJSI branch line flow elements FE-924, FE-925, FE-926, and FE-927 with more restrictive flat plate orifices has been evaluated. The SBLOCA evaluation results arc bounded by the current SBLOCA AOR for Units 1 and 2.

## 15.6.5.4 Radiological Consequences

The results of the analyses presented in this section demonstrate that the radioactivity released to the environment by a LOCA does not result in doses exceeding the limits specified in 10 CFR 100. The dose calculations take into account radioactivity released to the environment by containment leakage of gases, by leakage of the recirculating sump solution, and by containment purge at the beginning of the accident. The results presented in this section are applicable to both initial and reload cycles and remain valid for both the VANTAGE 5 and LOPAR fuel assembly designs to 60,000 MWd/Mtu lead rod average burnups.

The major assumptions and parameters assumed in the analysis are itemized in tables 15.6.5-9 and 15A-1.

In the evaluation of a LOCA, the fission product release assumptions of Regulatory Guide 1.4 have been followed with some exceptions. Table 15.6.5-10 provides a comparison of the analysis to the recommendations of Regulatory Guide 1.4.

The mathematical models used to calculate the activity releases during the course of the accident and the resultant doses are described in appendix 15A.

## 15.6.5.4.1 Fission Product Release to the Containment

Following a postulated double-ended rupture of a reactor coolant pipe with subsequent blowdown, the ECCS limits the fuel clad temperature to well below the melting point and ensures that the reactor core remains intact and in a coolable geometry, thus minimizing the release of fission products to the containment. However, to demonstrate that the operation of a nuclear power plant does not represent an undue radiological hazard to the general public, a hypothetical accident involving a significant release of fission products to the containment is evaluated. It is assumed that, 20 s into the accident, 100 percent of the noble gases and 50 percent of the iodine equilibrium core fission product inventory are released to the containment atmosphere. The iodine and the noble gas activity are assumed to be immediately available for leakage from the containment.

## 15.6.5.4.2 Fission Product Release Due to Containment Leakage

Once the gaseous fission product activity is released to the containment atmosphere, it is subject to various mechanisms of removal which operate simultaneously to reduce the amount of activity in the containment atmosphere. The removal mechanisms include radioactive decay, containment sprays, deposition, and containment leakage. For the noble gas fission products, the only removal processes considered in the containment are radioactive decay and containment leakage. Credit for radioactive decay of fission products located within the containment is assumed throughout the course of the accident. Once the activity is released to the environment, no credit is taken for radioactive decay or deposition. The containment leakage to the environment is assumed to be direct and unfiltered.

The quantity of activity released through leakage from the containment was calculated with a two-volume model of the containment to represent sprayed and unsprayed regions of the containment. This model is discussed in appendix 15A.

Of the total free volume of the containment, part is covered by the containment spray, while some is not. The unsprayed fraction has been calculated as approximately 22 percent. The transfer rate between the sprayed and unsprayed regions is assumed to be limited to the forced convection induced by the fan cooler units. The number of units assumed in operation and the total mixing flow are presented in table 15.6.5-9. This assumed minimum flowrate conservatively neglects the effects of natural convection, spray-induced turbulence, steam condensation, and diffusion, although these effects are expected to enhance the mixing rate between the sprayed and unsprayed volumes.

For fission products other than iodine, the only removal processes considered are radioactive decay and leakage. Iodine is assumed to be removed not only by radioactive decay and leakage, but also by deposition and by the containment spray system. The effectiveness of the containment spray for the removal of the iodine in the containment atmosphere and the model used to determine the iodine removal efficiency are discussed in subsection 6.5.2. The iodine removal constants are given in table 15.6.5-9.

Credit for iodine removal by the containment spray system is taken, starting at the time spray is initiated and continuing until a decontamination factor of 21.4 for elemental iodine is achieved. Credit for particulate removal in the sprayed region is taken for the duration of the accident, with the removal rate reduced after a decontamination factor of 50 is achieved. Credit for deposition removal of elemental iodine is taken from accident initiation until a decontamination factor of 200 is achieved.

Release from the containment by containment leakage is assumed to be 0.2% per day for the first 24 h and 0.1% per day thereafter. The offsite doses at the site boundary and at the low population zone and the doses to control room personnel are given in table 15.6.5-11. The activity released due to containment leakage is given in table 15.6.5-12.

## 15.6.5.4.3 Fission Product Release Due to Containment Purge Operation

During normal power operation the containment purge system (described in section 6.2) is operating, venting the containment at 5000 ft<sup>3</sup>/min. In the event of a LOCA, the purge system supply and exhaust isolation valves are assumed to close within 5 s of receiving a containment isolation signal as designed.

The containment airborne fission product inventory available for release is based on 100% of the total primary coolant iodine inventory assuming a preexisting iodine spike level of  $60-\mu$ Ci/g dose equivalent I-131 and 100% of the primary coolant noble gas inventory assuming the reactor has been operating with 1% fuel defects (i.e., defects in the cladding of fuel rods generating 1% of the core rated power). No credit is taken for removal of iodine by the purge filter train.

The offsite doses at the site boundary and at the low population zone and the doses to control room personnel are given in table 15.6.5-11. The activity released due to containment purge operation is given in table 15.6.5-12.

# 15.6.5.4.4 Radioactive Releases Resulting from Leakage from ECCS and Containment Spray Recirculation Lines

Subsequent to the injection phase of engineered safety features (ESF) system operation, the water in the containment recirculation sumps is recirculated by the residual heat removal, centrifugal charging, safety injection, and containment spray pumps. The radiological consequences of leakage from these systems is considered because the recirculated sump water contains a large fraction of the core iodines released as a result of the LOCA.

Because noble gases are assumed to be available for leakage from the containment atmosphere and are not readily entrained in water, the noble gases are not assumed to be part of the source term for this contribution to the total LOCA dose.

Since much of the radioiodine released during the LOCA would be retained by the containment sump water, it is conservatively assumed that 50 percent of the core iodine inventory is contained in the sump water to be recirculated through the external piping systems. Radioactive decay of this iodine is taken into account.

The fraction of the leakage that flashes to steam is assumed to carry 100% of the associated iodines into the air.

The recirculation flowpaths outside the containment are entirely within building areas served by the ESF ventilation system (subsection 6.5.1), which recirculates the air through charcoal filters

to remove airborne iodine and maintains the areas at subatmospheric pressure to prevent the release of unfiltered air. A fraction of the recirculating air is discharged to the environment.

The offsite doses at the site boundary and at the low population zone and the doses to control room personnel are given in table 15.6.5-11. The activity released due to recirculation leakage is given in table 15.6.5-12.

#### 15.6.5.4.5 Identification of Uncertainties and Conservatisms in the Analysis

The uncertainties and conservatisms in the assumptions used to evaluate the radiological consequences of a LOCA result principally from assumptions made involving the amount of the gaseous fission products available for release to the environment and the meteorology present at the site during the course of the accident. The most significant of these assumptions are:

- A. The ECCS is designed to prevent fuel cladding damage that would release the fission products, contained in the fuel, to the reactor coolant. Severe degradation of the ECCS (simultaneous failure of redundant components) would be necessary to release the quantity of fission products assumed in the analysis.
- B. The release of core fission products to the containment is assumed to occur after only 20 s from accident initiation.
- C. The activity released to the containment atmosphere is assumed to leak to the environment at the containment leakage rate of 0.2 volume percent/day for the first 24 h and 0.1 volume percent/day thereafter. The initial containment leakage rate is based on the peak calculated internal containment pressure anticipated after a LOCA. The pressure within the containment actually decreases with time. Taking into account that the containment leak rate is a function of pressure, the calculated doses could be reduced significantly.
- D. The meteorological conditions assumed to be present at the site during the course of the accident are based on x/Q values which are worse than those which will exist at the site 95% of the time. This condition results in the poorest values of atmospheric dispersion calculated for the exclusion area boundary and the low population zone outer boundary. Furthermore, no credit has been taken for the transit time required for activity to travel from the point of release to the exclusion area boundary and to the low population zone outer boundary. Hence, the radiological consequences evaluated under these conditions are conservative.

#### 15.6.5.4.6 Conclusions

15.6.5.4.6.1 <u>Filter Loadings</u>. No recirculating or single-pass filters are used for fission product cleanup and control within the containment following a postulated LOCA. The only ESF filtration systems expected to be operating under post-LOCA conditions are the control room heating, ventilation, and air-conditioning (HVAC) system and the auxiliary building emergency exhaust filtration system.

Activity loadings on the control room charcoal adsorbers are based on the flowrate through the adsorber, the concentration of activity at the adsorber inlet, and the adsorber efficiency. Based on the radioactive iodine release assumptions previously described, the assumption that 50% of

the core inventory of isotopes I-127 and I-129 is available for release from the containment atmosphere, and the assumption that the charcoal adsorber is 100% efficient, the calculated filter loadings are in accordance with Regulatory Guide 1.52, which limits the maximum loading to 2.5 mg of iodine/g of activated charcoal.

15.6.5.4.6.2 <u>Doses to a Receptor at the Exclusion Area Boundary and Low Population</u> <u>Zone Outer Boundary</u>. The potential radiological consequences resulting from the occurrence of the postulated LOCA have been conservatively analyzed, using assumptions and models described in previous sections.

The total-body immersion dose and the thyroid inhalation dose have been analyzed for the 0- to 2-h dose at the exclusion area boundary and for the duration of the accident at the low population zone. The results are listed in table 15.6.5-6. The resultant doses are within the guideline values of 10 CFR 100.

15.6.5.4.6.3 <u>Doses to Control Room Personnel</u>. Radiation doses to control room personnel following a postulated LOCA are based on the ventilation, cavity dilution, and dose model discussed in section 15A.3.

Control room personnel are subject to a total-body immersion dose and a thyroid inhalation dose. These doses have been analyzed and are provided in table 15.6.5-6. The operator will take appropriate action to ensure that the resultant doses are within the limits established by General Design Criterion 19.

#### 15.6.5.5 <u>References</u>

- 1. "Acceptance Criteria for Emergency Core Cooling Systems for Light Water Nuclear Power Reactors," 10 CFR 50.46, and, "ECCS Evaluation Models," Appendix K of 10 CFR 50, <u>Federal Register 1974</u>, Volume 39, Number 3.
- U.S. Nuclear Regulatory Commission 1975, "Reactor Safety Study An Assessment of Accident Risks in U.S. Commercial Nuclear Power Plants," WASH-1400, <u>NUREG-75/014</u>.
- 3. "The 1981 Version of the Westinghouse ECCS Evaluation Model Using the BASH Code," <u>WCAP-10266-P-A</u>, Revision 2 (Proprietary) and <u>WCAP-11524-A</u>, Revision 2 (Non-Proprietary), March 1987.
- 4. Westinghouse letter NTD-NRC-94-4143 from N. J. Liparulo to W. T. Russell (USNRC), "Change in Methodology for Execution of BASH Evaluation Model," May 23, 1994.
- 5. Westinghouse letter NTD-NRC-95-4540 from N. J. Laparulo to W. T. Russell (USNRC), "Change in Methodology for Execution of BASH Evaluation Model," August 29, 1995.
- 6. "SATAN VI Program: Comprehensive Space-Time Dependent Analysis of Loss-of-Coolant," <u>WCAP-8302</u> (Proprietary) and <u>WCAP-8306</u> (Non-Proprietary), June 1974.
- 7. "BART-A1: A Computer Code for the Best Estimate Analysis of Reflood Transients," <u>WCAP-9561-P-A</u> (Proprietary) and <u>WCAP-9695-A</u> (Non-Proprietary), March 1984.
- 8. "NOTRUMP: A Nodal Transient Small Break and General Network Code," <u>WCAP-</u> <u>10079-P-A</u> (Proprietary) and <u>WCAP-10080-A</u> (Non-Proprietary), August 1985.

- 9. "Containment Pressure Analysis Code (COCO)," <u>WCAP-8327</u>, July 1974.
- 10. "LOCTA-IV Program: Loss-of-Coolant Transient Analysis," <u>WCAP-8301</u> (Proprietary) and <u>WCAP-8305</u> (Non-Proprietary), June 1974.
- 11. "Westinghouse Improved Performance Analysis and Design Model (PAD 4.0)," <u>WCAP-15063-P-A</u>, Revision 1 (Proprietary), and <u>WCAP-15064-NP-A</u> (Non-Proprietary), July 2000.
- 12. "Westinghouse Emergency Core Cooling System Evaluation Model-Sensitivity Studies," <u>WCAP-8341</u> (Proprietary) and <u>WCAP-8342</u> (Non-Proprietary), July 1974.
- 13. "Westinghouse Emergency Core Cooling System Plant Sensitivity Studies," <u>WCAP-8340</u> (Proprietary) and <u>WCAP-8356</u> (Non-Proprietary), July 1974.
- 14. "Westinghouse ECCS Four Loop Plant (17 x 17) Sensitivity Studies," <u>WCAP-8565-P-A</u> (Proprietary) and <u>WCAP-8566-A</u> (Non-Proprietary), July 1975.
- 15. "Steam Generator Tube Plugging Margin Analysis for the Alvin W. Vogtle Nos. 1 & 2 Nuclear Power Plants," <u>WCAP-10338</u>, August 1983.
- 16. "Westinghouse Small Break ECCS Evaluation Model Using the NOTRUMP Code," <u>WCAP-10054-P-A</u>, August 1985.
- 17. "Addendum to the Westinghouse Small Break ECCS Evaluation Model Using the NOTRUMP Code: Safety Injection into the Broken Loop and COSI Condensation Model," <u>WCAP-10054-P-A</u>, Addendum 2, Revision 1, July 1997.
- 18. "LOCTA-IV Program: Loss-of-Coolant Transient Analysis," <u>WCAP-8301</u>, June 1974.
- 19. "Westinghouse Small Break LOCA ECCS Evaluation Model Generic Study with the NOTRUMP Code," <u>WCAP-11145-P-A</u>, October 1986.
- 20. "Model Changes to the Westinghouse Appendix K Small Break LOCA NOTRUMP Evaluation Model: 1988-1997," <u>WCAP-15085</u>, July 1998.
- 21. Federal Guidance Report No. 11.
- 22. Federal Guidance Report No. 12.
- 23. Shah, H., "Optimized ZIRLO<sup>™</sup>," <u>WCAP-12610-P-A & CENPD-404-P-A Addendum 1-A</u>, July 2006.

#### VEGP-FSAR-15

#### TABLE 15.6.1-1

# TIME SEQUENCE OF EVENTS FOR INCIDENTS WHICH CAUSE A DECREASE IN REACTOR COOLANT INVENTORY

Accident	<u>Event</u>	Time (s)
Inadvertent opening of a pressurizer safety valve	Pressurizer safety valve opens fully	0.0
	Overtemperature $\Delta T$ reactor trip setpoint reached	24.5
	Rods begin to drop	26.5
	Minimum DNBR occurs	27.0

#### TABLE 15.6.2-1

## PARAMETERS USED TO EVALUATE THE CONSEQUENCES OF A BREAK IN AN RCS INSTRUMENT LINE OUTSIDE THE CONTAINMENT

Core power level (MWt)	3636
Fuel defect level (%)	1.0
Reactor coolant iodine spike ( $\mu$ Ci/g dose equivalent I-131)	60
Reactor coolant radionuclide concentration (μCi/g) lodines Noble gases	Table 15A-6 Table 15A-4
Break flowrate (lbm/s)	0.106
Fraction of spill flashing	0.58
Fraction of iodine in the spill becoming airborne Flashed portion Nonflashed portion	1.0 0.1
Time to recognize break and to initiate shutdown (h)	8
Time to cool reactor coolant to 212°F; one RHR train in operation (h)	24
Atmospheric dispersion factors	Table 15A-2
Dose conversion factors	Table 15A-5

#### TABLE 15.6.2-2

## PARAMETERS USED TO EVALUATE THE CONSEQUENCES OF A LETDOWN LINE BREAK OUTSIDE THE CONTAINMENT

Core power level (MWt)	3636	
Fuel defect level (%)	1.0	
Reactor coolant iodine spike ( $\mu$ Ci/g dose equivalent I-131)	60	
Reactor coolant radionuclide concentration (μCi/g) lodines Noble gases	Table 15A-6 Table 15A-4	
Break flowrate (lbm/s)	21.3	
Fraction of spill flashing	0.086	
Fraction of iodine in the spill becoming airborne Flashed portion Nonflashed portion	1.0 0.1	
Time to isolate letdown line (min)	30	
Atmospheric dispersion factors	Table 15A-2	
Dose conversion factors	Table 15A-5	

## VEGP-FSAR-15

## TABLE 15.6.2-3

## DOSES RESULTING FROM AN RCS INSTRUMENT LINE BREAK OUTSIDE CONTAINMENT

## <u>Doses</u>

Thyroid dose (rem)			
	Site boundary	0.9	
	Low population zone	1.4	
Whole-body dose (rem)			
	Site boundary	< 0.1	
	Low population zone	< 0.1	

## TABLE 15.6.2-4

## DOSES RESULTING FROM A LETDOWN LINE BREAK OUTSIDE CONTAINMENT

## <u>Doses</u>

Thyroid dose (rem)		
Site boundary	12.5	
Low population zone	5.0	
Whole-body dose (rem)		
Site boundary	< 0.1	
Low population zone	< 0.1	

I

#### TABLE 15.6.3-1

## OPERATOR ACTION TIMES FOR DESIGN BASIS SGTR ANALYSIS

Action	Dose Analysis <u>Time (min)</u>	Margin to Overfill Analysis (ref. 11) <u>Time (min)</u>
Identify and isolate AFW flow to ruptured steam generator	4 minutes or LOFTTR2 calculated time to reach 9-percent narrow range level in the ruptured SG, whichever is longer <sup>(a)</sup>	6
Isolate ruptured steam generator	20 min or LOFTTR2 calculated time to recover to 30-percent narrow range level in the ruptured SG, whichever is longer <sup>(a)</sup>	15 min or LOFTTR2 calculated time to recover to 30-percent narrow range level in the ruptured SG, whichever is longer <sup>(a)</sup>
Operator action time to initiate cooldown	12	12
Cooldown	Calculated by LOFTTR2	Calculated by LOFTTR2
Operator action time to initiate depressurization	5.75	5.75
Depressurization	Calculated by LOFTTR2	Calculated by LOFTTR2
Operator action time to initiate SI termination	3	2
SI termination and pressure equalization	Calculated time for SI termination and equalization of RCS and ruptured SG pressures	Calculated time for SI termination and equalization of RCS and ruptured SG pressures

a. Earlier AFW isolation time is conservative for the dose analysis. The AFW isolation time used in the dose analysis is not a Time Critical Operator Action.

## TABLE 15.6.3-2

## SEQUENCE OF EVENTS

<u>Event</u>	<u>Time (s)</u>
SG tube rupture	0
Reactor trip	49.7
SI actuated	113
AFW flow isolated to ruptured steam generator	240
Ruptured SG isolated	1200
Ruptured SG ARV fails open	1202
Ruptured SG ARV block valve closed	2162
RCS cooldown initiated	2882
RCS cooldown terminated	3264
RCS depressurization initiated	3612
RCS depressurization terminated	3666
SI terminated	3846
Break flow terminated	5502

## TABLE 15.6.3-3

# MASS RELEASES RESULTS TOTAL MASS FLOW (Ib)

	<u>0 - 2 h</u>	<u>2 - 20 h</u>	
Ruptured SG			
- Condenser	57,000	0	
- Atmosphere	157,000	37,100	
- Feedwater	70,300	0	
Intact SGs			
- Condenser	169,900	0	
- Atmosphere	526,600	2,495,600	
- Feedwater	738,200	2,579,100	
Break Flow	204,900	0	

## VEGP-FSAR-15

# TABLE 15.6.3-4 (SHEET 1 OF 2)

## PARAMETERS USED IN EVALUATING RADIOLOGICAL CONSEQUENCES

l.	Sou A.	rce D Cor	ata e power level (MWt)	3636
	В.	Total steam generator tube leakage, prior to accident (gal/min)		1.0
	C.	Rea	actor coolant iodine activity:	
		1.	Accident initiated spike	The initial RC iodine activities based on 1 $\mu$ Ci/gram of D.E. I-131 are presented in table 15A-6. The iodine appearance rates assumed for the accident initiated spike are presented in table 15A-7.
		2.	Preaccident spike	Primary coolant iodine activities based on 60 $\mu$ Ci/gram of D.E. I-131 are presented in table 15A-6.
		3.	Noble gas activity	The initial RC noble gas activities based on 1-percent fuel defects are presented in table 15A-4.
	D.	Sec	condary system initial activity	Dose equivalent of 0.1 $\mu$ Ci/g of I-131, presented in table 15.6.3-5
	E.	Rea	actor coolant mass, grams	2.53 x 10 <sup>8</sup>
# TABLE 15.6.3-4 (SHEET 2 OF 2)

F.	Initia grar	al steam generator mass (each), ns	4.2 x 10 <sup>7</sup>
G.	Offs	ite power	Lost at time of reactor trip
H.	Prin for ii	nary-to-secondary leakage duration ntact SG, h	20
I.	Spe	cies of iodine	100-percent elemental
Activ	vity R	elease Data	
Α.	Rup	tured steam generator	
	1.	Rupture flow	See table 15.6.3-3
	2.	Rupture flow flashing fraction	See figure 15.6.3-13
	3.	lodine scrubbing efficiency	See figure 15.6.3-15
	4.	Total steam release, lb	See table 15.6.3-3
	5.	lodine partition factor	100
В.	Inta	ct steam generators	
	1.	Total primary-to-secondary leakage, gal/min	0.7
	2.	Total steam release, lb	See table 15.6.3-3
	3.	lodine partition factor	100
C.	Con	denser	
	1.	lodine partition factor	100
D.	Atm	ospheric Dispersion Factors	See table 15A-2

II.

## TABLE 15.6.3-5

# IODINE SPECIFIC ACTIVITIES IN THE SECONDARY COOLANT BASED ON 0.1 $\mu \text{Ci/gram}$ OF D.E. I-131

	Specific Activity (µ Ci/gm)
Nuclide	<u>Secondary Coolant</u> <u>0.1 μCi/gm</u>
I-131	0.074
I-132	0.075
I-133	0.141
I-134	0.018
I-135	0.069

TABLE 15.6.3-6

AND

TABLE 15.6.3-7

#### DELETED

## **BREATHING RATES**

Time (hours)	Breathing Rate (m <sup>3</sup> /s) (ref. 4)
0-2	3.47 x 10 <sup>-4</sup>
2-8	3.47 x 10 <sup>-4</sup>
8-20	1.75 x 10 <sup>-4</sup>

TABLE 15.6.3-9

# DELETED

#### WHOLE BODY AND BETA SKIN DOSE CONVERSION FACTORS FOR NOBLE GASSES (rem-m<sup>3</sup>/Ci-sec) (ref. 10)

	Conversion F	actors
Nuclide	Whole Body	<u>Beta Skin</u>
Xe-131m	1.44 x 10 <sup>-3</sup>	1.78 x 10 <sup>-2</sup>
Xe-133m	5.07 x 10 <sup>-3</sup>	3.85 x 10 <sup>-2</sup>
Xe-133	5.77 x 10 <sup>-3</sup>	1.84 x 10 <sup>-2</sup>
Xe-135m	7.55 x 10 <sup>-2</sup>	1.10 x 10 <sup>-1</sup>
Xe-135	4.40 x 10 <sup>-2</sup>	1.15 x 10⁻¹
Xe-138	2.13 x 10 <sup>-1</sup>	3.96 x 10⁻¹
Kr-85m	2.77 x 10 <sup>-2</sup>	8.29 x 10 <sup>-2</sup>
Kr-85	4.40 x 10 <sup>-4</sup>	4.88 x 10 <sup>-2</sup>
Kr-87	1.52 x 10 <sup>-1</sup>	5.07 x 10 <sup>-1</sup>
Kr-88	3.77 x 10 <sup>-1</sup>	5.00 x 10⁻¹

## OFFSITE RADIATION DOSES

		<u>Doses (rem)</u>		
4	Assident Initiated Indina Spiles	Calculated <u>Value</u>	Allowable Guideline <u>Value</u>	
١.	Accident Initiated Iodine Spike			
	Exclusion Area Boundary (0 - 2 h) Thyroid Dose	16.1	30	
	Low Population Zone (0 – 20 h) Thyroid Dose	6.9	30	
2.	Preaccident Iodine Spike			
	Exclusion Area Boundary (0 - 2 h) Thyroid Dose	26.3	300	
	Low Population Zone (0 – 20 h) Thyroid Dose	10.8	300	
3.	Whole-Body Gamma and Beta-Skin Dose			
	Exclusion Area Boundary (0 - 2 h) Whole-Body Gamma Dose	0.2	2.5	
	Beta-Skin Dose	0.3	not specified	
	Low Population Zone (0 – 20 h) Whole-Body Gamma Dose Beta-Skin Dose	0.1 0.2	2.5 not specified	

### SUMMARY OF LARGE BREAK LOCA ANALYSIS ASSUMPTIONS

Core power	3565 MWt
Calorimetric uncertainty	2%
Fuel type	17 x 17
Total core peaking factor, $F_{Q}$	2.50
Hot channel enthalpy rise factor, $F_{\Delta H}$	1.65
K(Z) limit	1.0 from 0 to 6 ft; 1.0 to 0.925 from 6 to 12 ft
Thermal design flow	93,600 gpm/loop
Nominal vessel average temperature	570.7 °F/588.4 °F <sup>(a)</sup>
Vessel average temperature uncertainty	±6 °F
Pressurizer pressure	2250 psia
Pressurizer pressure Pressurizer pressure uncertainty	2250 psia ±50 psi
Pressurizer pressure Pressurizer pressure uncertainty Steam generator tube plugging	2250 psia ±50 psi 10%
Pressurizer pressure Pressurizer pressure uncertainty Steam generator tube plugging Accumulator water volume, nominal	2250 psia ±50 psi 10% 900 ft <sup>3</sup> /accumulator
Pressurizer pressure Pressurizer pressure uncertainty Steam generator tube plugging Accumulator water volume, nominal Accumulator gas pressure, minimum	2250 psia ±50 psi 10% 900 ft <sup>3</sup> /accumulator 611.3 psia
Pressurizer pressure Pressurizer pressure uncertainty Steam generator tube plugging Accumulator water volume, nominal Accumulator gas pressure, minimum Safety injection pumped flow	2250 psia ±50 psi 10% 900 ft <sup>3</sup> /accumulator 611.3 psia Figures 15.6.5-17 and 15.6.5-18

a. Note that the LOCBART calculations are based on a  $T_{AVG}$  window of 573.0°F to 588.4°F, and an evaluation was performed to support operation at  $T_{AVG}$  values between 570.7°F and 573.0°F with up to 10 percent steam generator tube plugging. See paragraph 15.6.5.3.3.1.1 for more information.

### TABLE 15.6.5-2

# LARGE BREAK LOCA RESULTS

	C <sub>D</sub> =0.4	C <sub>D</sub> =0.6	C <sub>D</sub> =0.8	C <sub>D</sub> =1.0	C <sub>D</sub> =0.6	C <sub>D</sub> =0.6	C <sub>D</sub> =0.6	C <sub>D</sub> =0.6	C <sub>D</sub> =0.6
	Low T <sub>AVG</sub>	Low T <sub>AVG</sub>	Low T <sub>AVG</sub>	Low T <sub>AVG</sub>	High T <sub>AVG</sub>	Low T <sub>AVG</sub>	Low T <sub>AVG</sub>	Low T <sub>AVG</sub>	Low T <sub>AVG</sub>
	MIN SI	MAX SI	MIN SI	MIN SI	MIN SI				
	Cosine	Cosine	Cosine	Cosine	Cosine	Cosine	8.5'	Cosine	Cosine
	Shape	Shape	Shape	Shape	Shape	Shape	Shape	Shape	Shape
	non-IFBA	non-IFBA	non-IFBA	non-IFBA	non-IFBA	non-IFBA	non-IFBA	128-IFBA	156-IFBA
Peak Cladding	1997 0	1026.2	1000.2	1979.6	1022.0	1969 0	1010 7	2040.0	2061.6
Temperature (°F)	1007.2	1930.5	1909.2	1070.0	1952.9	1000.9	1919.7	2040.0	2001.0
Peak Cladding									
Temperature	191.5	174.4	168.8	162.5	172.8	181.0	199.1	176.1	175.6
Time(s)									
Peak Cladding									
Temperature	7.25	7.25.	7.25	7.25	7.25	7.25	9.0	7.25	7.25
Location (ft)									
Maximum Local									
Zr/H <sub>2</sub> O	<17.0	<17.0	<17.0	<17.0	<17.0	<17.0	<17.0	<17.0	<17.0
Reaction (%)									
Maximum Local									
Zr/H₂O	7.25	7.25	7.25	7.25	5.25	7.25	9.0	7.25	7.25
Location (ft)									
Total Zr/H₂O	<10	<10	<10	<10	<10	<10	<10	<10	<10
Reaction (%)	\$1.0	\$1.0	<1.0	\$1.0	\$1.0	<1.0	\$1.0	\$1.0	<1.0
Hot Rod Burst	74.44	40.05	00.07	40.77	44.05	40.05	40.40	40.05	54.00
Time(s)	71.11	43.85	39.97	42.77	41.35	43.85	48.48	43.85	51.62
Hot Rod Burst									
	6.25	5.75	5.50	5.50	5.25	5.75	8.25	5.75	5.50

#### TABLE 15.6.5-3

# LARGE BREAK LOCA TIME SEQUENCE OF EVENTS

RESULTS (sec)	C <sub>D</sub> =0.4 Low T <sub>AVG</sub> MIN SI Cosine Shape non-IFBA	C <sub>D</sub> =0.6 Low T <sub>AVG</sub> MIN SI Cosine Shape non-IFBA	C <sub>D</sub> =0.8 Low T <sub>AVG</sub> MIN SI Cosine Shape non-IFBA	$C_D$ =1.0 Low T <sub>AVG</sub> MIN SI Cosine Shape non-IFBA	C <sub>D</sub> =0.6 High T <sub>AVG</sub> MIN SI Cosine Shape non-IFBA	C <sub>D</sub> =0.6 Low T <sub>AVG</sub> MAX SI Cosine Shape non-IFBA	$\begin{array}{c} C_{\text{D}} = 0.6 \\ \text{Low } T_{\text{AVG}} \\ \text{MIN SI} \\ 8.5' \\ \text{Shape} \\ \text{non-IFBA} \end{array}$	$C_D=0.6$ Low $T_{AVG}$ MIN SI Cosine Shape 128-IFBA	$C_D=0.6$ Low $T_{AVG}$ MIN SI Cosine Shape 156-IFBA
Start	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Reactor Trip Signal	0.44	0.43	0.42	0.42	0.49	0.43	0.43	0.43	0.43
Safety Injection Signal	2.6	2.0	1.8	1.6	2.0	2.0	2.0	2.0	2.0
Accumulator Injection	19.6	13.6	11.0	9.8	14.8	13.6	13.8	13.6	13.6
End of Blowdown	41.1	33.4	28.8	26.7	31.7	33.4	33.1	33.4	33.4
Start of Safety Injection	42.6	42.0	41.8	41.6	42.0	42.0	42.0	42.0	42.0
Bottom of Core Recovery	55.5	45.9	39.2	36.8	45.3	45.6	44.1	45.9	45.9
Accumulator Empty	61.7	54.7	51.0	49.0	54.4	55.4	54.6	54.7	54.7

#### PEAK CLAD TEMPERATURE CHANGES FOR LARGE BREAK LOCA ANALYSIS (156 IFBA)

	<u>Unit 1</u>	<u>Unit 2</u>
Calculated PCT from ECCS Model Analysis (Analysis of Record)	2062.1 °F <sup>(a)(b)(c)</sup>	2062.1 °F <sup>(a)(b)(c)</sup>
Rebaseline of Limiting LOCBART Calculation	-22 °F <sup>(d)</sup>	-22 °F <sup>(d)</sup>
LOCBART Pellet Volumetric Heat Generation Rate Error Correction	18 °F <sup>(d)</sup>	18 °F <sup>(d)</sup>
PWROG TCD Evaluation - Rebaseline of AOR	22 °F <sup>(e)</sup>	22 °F <sup>(e)</sup>
Changes to Grid Blockage Ratio and Porosity	29 °F <sup>(f)</sup>	29 °F <sup>(f)</sup>
Total Resultant PCT	2109.1 °F	2109.1 °F

#### Notes:

(a) Includes 0.5 °F to address a reduction in low  $T_{AVG}$  from 573.0 °F to 570.7 °F.

(b) Includes additional metal mass allowance. Refer to table 6.2.1-72.

(c) Changes in assumed accumulator line resistance can affect plant testing acceptance criteria. The current PCT and test criteria are based on Westinghouse SECL-98-104, Revision 1.

(d) Per Westinghouse letter LTR-LIS-07-308, "10 CFR 50.46 Reporting Text for LOCBART Version 37.0 issues and Revised PCT Rackup Sheets for Vogtle Units 1 and 2," May 2007.

(e) Per Westinghouse letter LTR-12-514, "Vogtle Units 1 and 2 10 CFR 50.46 Notification and Reporting for Fuel Pellet Thermal Conductivity Degradation and Peaking Factor Burndown," September 20, 2012.

(f) Per Westinghouse letter LTR-LIS-13-408, "10 CFR 50.46 Report for the A. W. Vogtle Units 1 and 2 Evaluation of Changes to Grid Parameters Modeled in the BASH-EM Analysis," October 2013.

# SUMMARY OF SMALL BREAK LOCA ANALYSIS ASSUMPTIONS

Core power	3565 MWt
Calorimetric uncertainty	2%
Fuel type	17 x 17
Total core peaking factor, F <sub>Q</sub>	2.58
Hot channel enthalpy rise factor, $F_{\Delta H}$	1.7
K(Z) limit	1.0 from 0 to 6 ft; 1.0 to 0.925 from 6 to 12 ft
Thermal design flow	93,600 gmp/loop
Nominal vessel average temperature	570.7 °F/588.4 °F
Vessel average temperature uncertainty	±6 °F
Pressurizer pressure	2250 psia
Pressurizer pressure uncertainty	±50 psi
Steam generator tube plugging	10%
Accumulator water volume, nominal	900 ft <sup>3</sup> /accumulator
Accumulator gas pressure, minimum	611.3 psia
Safety injection pumped flow	Figures 15.6.5-20 and 15.6.5-21
Power shape	Figure 15.6.5-22

1

#### SMALL BREAK LOCA LOW TAVG NOTRUMP TRANSIENT RESULTS

<u>Event Time (sec)</u>	<u>2 in.<sup>(a)</sup></u>	<u>3 in.</u>	<u>4 in.</u>
Break initiation	0	0	0
Reactor trip signal	35.5	14.9	8.58
S-signal	76.1	28.0	16.6
SI delivered	116.1	68.0	56.6
Loop seal clearing <sup>(b)</sup>	1090	473	280
Core uncovery	N/A	707	737
Accumulator injection	N/A	1888	945
RWST empty time	2537.7	2479.4	2441.8
PCT time	N/A	1665	973.4
Core recovery	N/A	2388	1567

Note that there was no core uncovery for the 2-in. break case. Loop seal clearing is defined as break vapor flow > 1 lb/s. a.

b.

# $\label{eq:smallbreak} \begin{array}{c} \text{SMALL BREAK LOCA} \\ \text{LOW } T_{\text{AVG}} \text{ BEGINNING OF LIFE (BOL) ROD HEATUP RESULTS} \end{array}$

	<u>2 in.<sup>(a)</sup></u>	<u>3 in.</u>	<u>4 in.</u>
PCT (°F)	N/A	1138	975
PCT time (s)	N/A	1665	973.4
PCT elevation (ft)	N/A	11.25	10.75
Burst time (s)	N/A	N/A	N/A
Burst elevation (ft)	N/A	N/A	N/A
Max. local ZrO <sub>2</sub> (%)	N/A	0.08	0.02
Max. local ZrO <sub>2</sub> elev (ft)	N/A	11.25	11.00
Core-Wide avg. ZrO <sub>2</sub> (%)	N/A	0.02	0.01

a. Note that there was no core uncovery for the 2-in. break case.

#### TABLE 15.6.5-8

#### PEAK CLAD TEMPERATURE CHANGES FOR SMALL BREAK LOCA ANALYSIS

	<u>Unit 1</u>	<u>Unit 2</u>
Calculated PCT from ECCS analysis (Analysis of Record)	1138.0 °F	1138.0 °F
Total Resultant PCT	1138.0 °F	1138.0 °F

# TABLE 15.6.5-9 (SHEET 1 OF 4)

### PARAMETERS USED IN EVALUATING THE RADIOLOGICAL CONSEQUENCES OF A LOSS-OF-COOLANT ACCIDENT

# Source Data

Core power level (MWt)	3636	
Core activity released in the containment atmosphere after 20 s into the accident (%)		
Noble gas Iodine	100 50	
Core inventories	Table 15A-3	
lodine distribution (%)		
Elemental Organic Particulate	91 4 5	
Atmospheric Dispersion Factors	Table 15A-2	
Control Room Parameters	Tables 15A-1 and 15A-2	
Containment Leakage of Activity		
Containment leak rate (volume %/day)		
0 to 24 h 1 to 30 days	0.2 0.1	
Unfiltered containment leakage (%)	100	
Deposition iodine removal constants elemental iodine only (h <sup>-1</sup> )	4.8 (DF ≤ 200)	

# TABLE 15.6.5-9 (SHEET 2 OF 4)

# Credit for containment sprays

Spray iodine removal constants (h <sup>-1</sup> )	
Elemental Organic Particulate	10 <sup>(1)</sup> (DF ≤ 21.4) 0.0 4.2 (DF ≤ 50) 0.42 (DF > 50)
Duration of sprays (h) Sprayed volume (%) Unsprayed volume (%) Number of fan coolers operating Sprayed-unsprayed mixing rate (ft <sup>3</sup> /min) Containment volume (ft <sup>3</sup> )	2 78 22 2 87,000 2.93 x 10 <sup>6</sup>
Activity released to containment atmosphere	
<u>Isotope</u>	Curies
I-131 I-132 I-133 I-134 I-135 Xe-131m Xe-133m Xe-133 Xe-135m Xe-135 Xe-135 Xe-138 Kr-85m Kr-85 Kr-87 Kr-88	$5.15 \times 10^7$ $7.50 \times 10^7$ $1.05 \times 10^8$ $1.13 \times 10^8$ $9.75 \times 10^7$ $7.13 \times 10^5$ $3.01 \times 10^7$ $2.12 \times 10^8$ $4.18 \times 10^7$ $1.69 \times 10^8$ $2.68 \times 10^7$ $1.04 \times 10^6$ $4.93 \times 10^7$ $7.02 \times 10^7$

# TABLE 15.6.5-9 (SHEET 3 OF 4)

# Containment Purge of Activity

Purge flowrate (ft <sup>3</sup> /min)	5000
Duration of purge, from accident initiation (s)	8.5
Reactor coolant iodine spike (μCi/g I-131 dose equivalent)	60
Reactor coolant activity airborne in the containment (%)	
Noble gas Iodine	100 100

Activity released to the containment atmosphere from the reactor coolant

Isotope	<u>Curies</u>
I-131	1.12 x 10 <sup>4</sup>
I-132	1.14 x 10 <sup>4</sup>
I-133	2.14 x 10 <sup>4</sup>
I-134	2.73 x 10 <sup>3</sup>
I-135	1.05 x 10 <sup>4</sup>
Xe-131m	5.11 x 10 <sup>2</sup>
Xe-133m	4.45 x 10 <sup>3</sup>
Xe-133	6.48 x 10 <sup>4</sup>
Xe-135m	1.42 x 10 <sup>2</sup>
Xe-135	2.10 x 10 <sup>3</sup>
Xe-138	1.87 x 10 <sup>2</sup>
Kr-85m	5.16 x 10 <sup>2</sup>
Kr-85	2.12 x 10 <sup>3</sup>
Kr-87	3.24 x 10 <sup>2</sup>
Kr-88	9.31 x 10 <sup>2</sup>

# TABLE 15.6.5-9 (SHEET 4 OF 4)

# Recirculation Leakage Outside Containment

Leak rate (gal/min, measured at 70 °F)	2
Temperature of recirculating fluid (°F)	
0 to 0.5 h 0.5 to 2.0 h 2.0 to 720 h	No recirculation 240 < 212
Mass of water in the containment sump (lb)	6.77 x 10 <sup>6</sup>
Activity in the sump solution at time = 0	
Isotope	<u>Curies</u>
I-131 I-132 I-133 I-134 I-135 Volume of building served by the auxiliary building emergency ventilation system (ft <sup>3</sup> ) Auxiliary building emergency ventilation	$5.15 \times 10^{7}$ 7.50 x 10 <sup>7</sup> 1.0 x 10 <sup>8</sup> 1.1 x 10 <sup>8</sup> 9.75 x 10 <sup>7</sup> 525,000
system parameters (for each of two trains) Recirculation flow (ft <sup>3</sup> /min) Discharge flow (ft <sup>3</sup> /min) Elemental iodine filter efficiency (%)	13,950 2970 90

Notes:

(1) Calculated value is 22.5  $h^{-1}$ .

#### TABLE 15.6.5-10 (SHEET 1 OF 7)

#### DESIGN COMPARISON TO THE REGULATORY POSITIONS OF REGULATORY GUIDE 1.4, ASSUMPTIONS USED FOR EVALUATING THE POTENTIAL RADIOLOGICAL CONSEQUENCES OF A LOSS-OF-COOLANT ACCIDENT FOR PRESSURIZED WATER REACTORS, REVISION 2, JUNE 1974

#### **Regulatory Guide 1.4 Position**

<u>Design</u>

 The assumptions related to the release of radioactive material from the fuel and containment are as follows:

a. 25% of the equilibrium radioactive iodine inventory developed from maximum full-power operation of the core should be assumed to be immediately available for leakage from the primary reactor containment. 91% of this 25% is to be assumed to be in the form of elemental iodine; 5% of this 25% in the form of particulate iodine; and 4% of this 25% in the form of organic iodides.

b. 100% of equilibrium radioactive noble gas inventory developed from maximum full-power operation of the core should be assumed to be immediately available for leakage from the reactor containment.

c. The effects of radiological decay during holdup in the containment or other buildings should be taken into account.

Fifty percent of core inventory of iodine is assumed to be immediately available for leakage from the containment. The iodine is assumed to be 91% elemental, 5% particulate, and 4% organic. These assumptions are in accordance with Section 6.5.2 of NUREG-0800.

Conforms.

Conforms. Credit for radioactive decay is taken until the activity is assumed to be released.

#### TABLE 15.6.5-10 (SHEET 2 OF 7)

#### Regulatory Guide 1.4 Position

#### <u>Design</u>

Conforms.

d. The reduction in the amount of radioactive material available for leakage to the environment by containment sprays, recirculating filter systems, or other engineered safety features may be taken into account, but the amount of reduction in concentration of radioactive materials should be evaluated on an individual case basis.

e. The primary reactor containment should be assumed to leak at the leak rate incorporated or to be incorporated as a technical specification requirement at peak accident pressure for the first 24 h and at 50% of this leak rate for the remaining duration of the accident. Peak accident pressure is the maximum pressure defined in the Technical Specifications for containment leak testing.

2. Acceptable assumptions for atmospheric diffusion and dose conversion are:

a. The 0- to 8-h ground level release concentrations may be reduced by a factor ranging from 1 to a maximum of 3 (see Figure 1) for additional dispersion produced by the turbulent wake of the reactor building in calculating potential exposures. The volumetric building wake correction, as defined in section 3.3.5.2 of Meteorology and Atomic Energy 1968, should be used only in the 0- to 8-h period; it is used with a shape factor of 1/2 and the minimum cross-sectional area of the reactor building only. Short-term accident atmospheric dispersion factors were calculated based on site meteorological measurement program described in section 2.3. These factors are for ground level releases and are based on Regulatory Guide 1.145 methodology and represent the worst of the 5% site meteorology and the 0.5% worst sector meteorology.

Conforms.

#### TABLE 15.6.5-10 (SHEET 3 OF 7)

#### **Regulatory Guide 1.4 Position**

#### <u>Design</u>

b. No correction should be made for depletion of the effluent plume of radioactive iodine resulting from deposition on the ground or for the radiological decay of iodine in transit.

c. For the first 8 h, the breathing rate of persons offsite should be assumed to be  $3.47 \times 10-4 \text{ m}^3/\text{s}$ . From 8 to 24 h following the accident, the breathing rate should be assumed to be  $1.75 \times 10-4 \text{ m}^3/\text{s}$ . After that, until the end of the accident, the breathing rate should be assumed to be  $2.32 \times 10-4 \text{ m}^3/\text{s}$ . (These values were developed from the average daily breathing rate (2 x 107 cm<sup>3</sup>/day) assumed in the report of ICRP, Committee II-1959.)

d. The iodine dose conversion factors are given in ICRP Publication 2, Report of Committee II, Permissible Dose for Internal Radiation, 1959.

e. External whole body doses should be calculated using "infinite cloud" assumptions; i.e., the dimensions of the cloud are assumed to be large compared to the distance that the gamma rays and beta particles travel. "Such as a cloud would be considered an infinite cloud for a receptor at the center because any additional (gamma and) beta emitting material beyond the cloud dimensions would not alter the flux of (gamma rays and) beta particles to the receptor" (Meteorology and Atomic Energy, Section 7.4.1.1; editorial additions made so that gamma and beta emitting material could be considered). Under these conditions the rate of energy absorption per unit volume is equal

Same as response to 2a above.

Conforms.

The dose conversion factors provided in Federal Guidance Report 11 are used.

The dose conversion factors provided in Federal Guidance Report 12 are used.

#### TABLE 15.6.5-10 (SHEET 4 OF 7)

#### Regulatory Guide 1.4 Position

#### <u>Design</u>

to the rate of energy released per unit volume. For an infinite uniform cloud containing x curies of beta radioactivity per cubic meter, the beta dose in air at the cloud center is:

$$_{\beta}D_{\infty}' + 0.457 \overline{E}_{\beta\chi}$$

The surface body dose rate from beta emitters in the infinite cloud can be approximated as being onehalf this amount (i .e.,

$$_{\beta}D_{\infty}^{'}$$
 + 0.23  $\overline{E}_{\beta\chi}$  )

For gamma emitting material, the dose rate in air at the cloud center is:

$$\gamma D'_{\infty} + 0.507 \overline{E}_{\gamma \chi}$$

From a semi-infinite cloud, the gamma dose rate in air is:

$$\gamma D'_{\infty} + 0.25 \overline{E}_{\gamma \chi}$$

where:

- $\beta D'_{\infty}$  = Beta dose rate from an infinite cloud (rad/s).
- $\gamma D'_{\infty}$  = Gamma dose rate from an infinite cloud (rad/s).
- $\overline{E}_{\beta}$  = Average gamma energy per disintegration (MeV/dis).
- $\overline{E}_{\gamma}$  = Average beta energy per disintegration (MeV/dis).
- $\chi$  = Concentration of beta of gamma emitting isotope in the cloud (Ci/m<sup>3</sup>).

### TABLE 15.6.5-10 (SHEET 5 OF 7)

#### **Regulatory Guide 1.4 Position**

#### Design

f. The following specific assumptions are acceptable with respect to the radioactive cloud dose calculations:

(1)The dose at any distance from the See response to 2e reactor should be calculated based on the above. maximum concentration in the plume at that distance, taking into account specific meteorological, topographical, and other characteristics which may affect the maximum plume concentration. These site-related characteristics must be evaluated on an individual case basis. In the case of beta radiation, the receptor is assumed to be exposed to an infinite cloud at the maximum ground level concentration at that distance from the reactor. In the case of gamma radiation, the receptor is assumed to be exposed to only one-half the cloud owing to the presence of the ground. The maximum cloud concentration always should be assumed to be at around level.

(2) The appropriate average beta and gamma energies emitted per disintegration, as given in the Table of Isotopes, Sixth Edition, by C.
M. Lederer, J. M. Hollander, and I. Perlman; University of California, Berkeley, Lawrence Radiation Laboratory, should be used.

g. The atmospheric diffusion model should be as follows:

(1) The basic equation for atmospheric diffusion from a ground level point source is:

Short-term accident atmospheric dispersion factors were calculated based on onsite meteorological measurement program described in section 2.3. These factors are for ground level releases and are based on Regulatory Guide1.145 methodology and represent the worst of the 5% site meteorology and the 0.5% worst sector meteorology.

See response to 2e

above.

$$\chi_{/Q} = \frac{1}{\pi \mu \sigma_y \sigma_z}$$

#### TABLE 15.6.5-10 (SHEET 6 OF 7)

#### Regulatory Guide 1.4 Position

#### <u>Design</u>

where:

χ	=	The short term average centerline
		value of ground level concentration
		(Ci/m³).

- Q = Amount of material released (Ci/s)
- u = Windspeed (m/s).
- σ<sub>y</sub> = The horizontal standard deviation of the plume (m). (See Figure V-1, page 48, <u>Nuclear Safety</u>, June 1961, Volume 2, Number 4, Use of Routine Meteorological Observations for Estimating Atmospheric Dispersion, F. A. Gifford, Jr.)
- σ<sub>z</sub> = The vertical standard deviation of the plume (m). (See Figure V-2, page 48, <u>Nuclear Safety</u>, June 1961, Volume 2, Number 4, Use of Routine Meteorological Observations for Estimating Atmospheric Dispersion, F. A. Gifford, Jr.)

(2) For time period of greater
than 8 h the plume should be assumed to
meander and spread uniformly over a 22.5° sector.
The resultant equation is:

$$\chi/_{Q} = \frac{2.032}{\sigma_{x}u_{\chi}}$$

where:

χ

Distance from point of release to the receptor; other variables are given in 2g (1).

(3) The atmospheric diffusion See response to 2g (1) model2 for ground level release is based on the above. information below:

# TABLE 15.6.5-10 (SHEET 7 OF 7)

Time Following <u>Accident</u>	Atmospheric Conditions			
0 to 8 h	Pasquill type F, windspeed 1 m/s, uniform direction			
8 to 24 h	Pasquill type F, windspeed 1 m/s, variable direction within a 22.5 $^{\circ}$ sector			
1 to 4 days	(a) 40% Pasquill type D, windspeed 3 m/s			
	(b) 60% Pasquill type F, windspeed 2 m/s			
	(c) wind direction variable within a 22.5° sector			
4 to 30 days	(a) 33.3% Pasquill type C, windspeed 3 m/s			
	(b) 33.3% Pasquill type D, windspeed 3 m/s			
	(c) 33.3% Pasquill type F, windspeed 2 m/s			
	(d) windspeed direction 33.3% frequency in a	22.5° sector		
(4) Figure	es 2A and 2B give the atmospheric diffusion	See response to 2g (1) above.		

ground level release atmospheric diffusion factor based on the parameters given in 2g(3).

# TABLE 15.6.5-11 (SHEET 1 OF 2)

# DOSES RESULTING FROM A LOSS-OF-COOLANT ACCIDENT

# Site Boundary Dose (0 to 2 h)

Containment leakage Thyroid (rem) Gamma body (rem) Beta skin (rem)	74.4 1.9 4.1
Containment purge Thyroid (rem) Gamma body (rem) Beta skin (rem)	0.3 < 0.1 < 0.1
Recirculation leakage Thyroid (rem) Gamma body (rem) Beta skin (rem)	10.0 0.1 0.1
Thyroid (rem) Gamma body (rem) Beta skin (rem)	84.6 2.0 4.2
Low Population Zone (0 to 30 days)	
Containment leakage Thyroid (rem) Gamma body (rem) Beta skin (rem)	88.1 1.3 3.1
Containment purge Thyroid (rem) Gamma body (rem) Beta skin (rem)	0.1 < 0.1 < 0.1
Recirculation leakage Thyroid (rem) Gamma body Beta skin	35.3 0.2 0.5
Thyroid (rem) Gamma body (rem) Beta skin (rem)	124 1.5 3.5

I

# TABLE 15.6.5-11 (SHEET 2 OF 2)

# Control Room (0 to 30 days)

Containment leakage Thyroid (rem) Gamma body (rem) Beta skin (rem)	20.6 0.6 13.6
Containment purge Thyroid (rem) Gamma body (rem) Beta skin (rem)	0.3 < 0.1 < 0.1
Recirculation leakage Thyroid (rem) Gamma body (rem) Beta skin (rem)	8.8 0.1 3.1
Thyroid (rem) Gamma body (rem) Beta skin (rem)	29.7 1.0 <sup>(a)</sup> 16.7

(a) Includes contributions from inside and outside the control room.

1

## TABLE 15.6.5-12

# ACTIVITY RELEASES TO THE ENVIRONMENT DUE TO A LOSS-OF-COOLANT ACCIDENT (Ci)

<u>0 to 2 h</u>

<u>Isotope</u>	Containment Leakage	Containment Purge	Recirculation Leakage	<u>Total</u>
I-131 I-132 I-133 I-134 I-135	$7.91 \times 10^2$ $9.90 \times 10^2$ $1.59 \times 10^3$ $1.23 \times 10^3$ $1.42 \times 10^3$	2.71 2.75 5.17 6.59 x 10 <sup>-1</sup> 2.54	$\begin{array}{c} 1.07 \times 10^2 \\ 1.08 \times 10^2 \\ 2.11 \times 10^2 \\ 9.30 \times 10^1 \\ 1.79 \times 10^2 \end{array}$	$9.00 \times 10^{2}$ $1.10 \times 10^{3}$ $1.80 \times 10^{3}$ $1.33 \times 10^{3}$ $1.60 \times 10^{3}$
Kr-85m Kr-85 Kr-87 Kr-88	3.83 x 10 <sup>3</sup> 1.73 x 10 <sup>2</sup> 5.00 x 10 <sup>3</sup> 9.26 x 10 <sup>3</sup>	1.25 x 10 <sup>-1</sup> 5.12 x 10 <sup>-1</sup> 7.82 x 10 <sup>-2</sup> 2.25 x 10 <sup>-1</sup>	NA NA NA NA	3.83 x 10 <sup>3</sup> 1.74 x 10 <sup>2</sup> 5.00 x 10 <sup>3</sup> 9.26 x 10 <sup>3</sup>
Xe-131m Xe-133m Xe-133 Xe-135m Xe-135 Xe-138	$\begin{array}{c} 1.19 \times 10^{2} \\ 4.96 \times 10^{3} \\ 3.51 \times 10^{4} \\ 3.18 \times 10^{3} \\ 8.38 \times 10^{3} \\ 4.79 \times 10^{3} \end{array}$	$1.24 \times 10^{-1}$ 1.08 $1.57 \times 10^{1}$ $3.54 \times 10^{-2}$ $5.08 \times 10^{-1}$ $4.50 \times 10^{-2}$	NA NA NA NA NA	$\begin{array}{c} 1.19 \times 10^{2} \\ 4.96 \times 10^{3} \\ 3.53 \times 10^{4} \\ 5.85 \times 10^{3} \\ 1.00 \times 10^{4} \\ 4.79 \times 10^{3} \end{array}$
<u>2 to 720 h</u>				
I-131 I-132 I-133 I-134 I-135	$2.62 \times 10^4$ $5.04 \times 10^2$ $8.35 \times 10^3$ $1.10 \times 10^2$ $2.67 \times 10^3$		$1.83 \times 10^{4}$ $1.88 \times 10^{4}$ $4.13 \times 10^{3}$ $4.08 \times 10^{1}$ $1.05 \times 10^{3}$	$\begin{array}{c} 4.45 \times 10^4 \\ 6.92 \times 10^2 \\ 1.25 \times 10^4 \\ 1.51 \times 10^2 \\ 3.72 \times 10^3 \end{array}$
Kr-85m Kr-85 Kr-87 Kr-88	$1.04 \times 10^{4}$ 3.15 x 10 <sup>4</sup> 2.53 x 10 <sup>3</sup> 1.48 x 10 <sup>4</sup>		NA NA NA NA	$1.04 \times 10^4$ 3.15 x 10 <sup>4</sup> 2.53 x 10 <sup>3</sup> 1.48 x 10 <sup>4</sup>
Xe-131m Xe-133m Xe-133 Xe-135m Xe-135 Xe-138	$1.46 \times 10^4$ $1.20 \times 10^5$ $1.95 \times 10^6$ $9.63 \times 10^3$ $1.02 \times 10^5$ $1.36 \times 10^1$		$1.33 \times 10^{4}$ $1.26 \times 10^{4}$ $4.21 \times 10^{5}$ $1.69 \times 10^{4}$ $1.28 \times 10^{5}$ NA	$\begin{array}{c} 2.78 \times 10^{4} \\ 1.32 \times 10^{5} \\ 2.37 \times 10^{6} \\ 2.66 \times 10^{4} \\ 2.30 \times 10^{5} \\ 1.36 \times 10^{1} \end{array}$
































•		BREAK OCCURS		
		REACTOR TRIP (COMPENSATED PRESSURIZER	PRESSURE)	
		PUMPED SAFETY INJECTION SIGNAL (HI-I CO	PUMPED SAFETY INJECTION SIGNAL (HI-I CONT. PRESS. OR LO PRESSURIZER PRESS.)	
В		SAFETY INJECTION BEGINS (ASSUMING OFFSITE POWER AVAILABLE)		
L O	¥	ACCUMULATOR INJECTION		
D		CONTAINMENT HEAT REMOVAL SYSTEM INIT	ATION (ASSUMING DEESITE DOWER AVAILADED)	
0 ₩ N			ATON (ASSOMING OFFSTE POWER AVAILABLE)	
		END OF BYPASS		
<u> </u>	R	END OF BLOWDOWN		
	F PUMPED SAFETY INJECTION BEGINS (ASSUMING LOSS OF OFFSITE POWER)		NG LOSS OF OFFSITE POWER)	
	L .	BOTTOM OF CORE RECOVERY		
	A	CONTAINMENT HEAT REMOVAL SYSTEM INITIATION (ASSUMING LOSS OF OFESITE POWER)		
R	1		ATTOR (ASSOMING LOSS OF OFFSITE POWER)	
E F		ACCUMULATORS EMPTY		
LO				
O D				
		CORE QUENCHED		
1				
L O N G SWITCH TO COLD LEG RECIRCULATION ON RWST LOW LEVEL AL				
		WITCH TO COLD LEG RECIRCULATION ON RWST LOW LEVEL ALARM (SEMI-AUTOMATIC)		
E				
M C O C SWITCH TO LONG-TERM RECIRCULATION (MANUAL ACTION)				
		SWITCH TO LONG-TERM RECIRCULATION (MA	NUAL ACTION)	
Ğ	G			
			<i>i</i>	
¥ I	· 1			
RFV 14 10/07				
			BREAK LOCA ANALVSIS	
SOL	JTHE CO	RN A VOGILL ELECTRIC GENERATING PLANT UNIT 1 AND UNIT 2	BILLAIL LOOK ANAL I SIS	
Energy	to Serve Yo		FIGURE 15.6.5–1	













































































































































































































































































































# 15.7 RADIOACTIVE RELEASE FROM A SUBSYSTEM OR COMPONENT

This class of accident can be caused by any of the following events:

- A. Radioactive gas waste system leak or failure--this is an American Nuclear Society (ANS) Condition III event.
- B. Radioactive liquid waste system leak or failure--this is an ANS Condition III event.
- C. Postulated radioactive release due to liquid tank failures--this is an ANS Condition III event.
- D. Fuel handling accident--this is an ANS Condition IV event.
- E. Spent fuel cask drop accidents--this is an ANS Condition III event.

All of the accidents in this section have been analyzed. It has been determined that the most severe radiological consequences will result from the fuel handling accident analyzed in subsection 15.7.4.

## 15.7.1 RADIOACTIVE WASTE GAS DECAY TANK FAILURE

### 15.7.1.1 Identification of Causes

This accident is an infrequent fault. Its consequences will be considered in this section. The accident is defined as an unexpected and uncontrolled release of radioactive iodine, xenon, and krypton fission product gases stored in a waste gas decay tank as a consequence of a failure of a single gas tank or associated piping.

### 15.7.1.2 <u>Sequence of Events and System Operations</u>

During a refueling shutdown, the radioactive gases are stripped from the primary coolant and are stored in the gas decay tanks. After the transfer has been completed, the tank is assumed to fail. This releases all of the contents of the tank to the auxiliary building. Also, since the tanks are isolated from each other, the only radioactivity released is from the failed tank. For conservatism, the tank is assumed to fail after 40 years<sup>a</sup>, releasing the peak inventory expected in the tank.

<sup>&</sup>lt;sup>a</sup> The renewed operating licenses authorized a 20-year period of extended operation for both VEGP units, resulting in a total plant operating life of 60 years. Since the inventory in the Waste Gas Decay Tanks (WGDTs) has been routinely released during the first 20 years of operation and is expected to continue to be routinely released during future operation, the inventory of the WGDTs accumulated during the first 20 years of operation will be released prior to entering the period of extended operation. Therefore, the stated design capacity of the GWPS remains sufficient, and the analysis of the maximum fission product inventory in the GWPS over a 40-year plant life remains bounding for a 60-year plant life.

## 15.7.1.3 Core and System Performance

This accident occurs when the reactor is in the shutdown condition. There is no impact on the core or its system performance.

### 15.7.1.4 Barrier Performance

The only barrier between the released activity and the environment is the auxiliary building. During the course of this accident, the auxiliary building is assumed to remain intact. This means that the only method of release is through the auxiliary building ventilation system.

### 15.7.1.5 Radiological Consequences

### 15.7.1.5.1 Method of Analysis

15.7.1.5.1.1 <u>Physical Model</u>. Radioactive waste gas decay tanks (WGDTs) are used in the design to permit the decay of radioactive gases as a means of reducing or preventing the release of radioactive materials to the atmosphere. To evaluate the radiological consequences of the gaseous waste processing system, it is postulated that there is an accidental release of the contents of one of the WGDTs resulting from a rupture of the tank or from another cause, such as operator error or valve malfunction. The gaseous waste processing system (GWPS) is so designed that the tanks are isolated from each other during use, limiting the quantity of gas released in the event of an accident by preventing the flow of radioactive gas between the tanks.

The principal radioactive nuclides of the WGDTs are the noble gases krypton and xenon, the particulate daughters of some of the krypton and xenon isotopes, and trace quantities of halogens. The maximum amount of waste gases stored in any one tank occurs during a refueling shutdown, at which time the WGDTs store the radioactive gases stripped from the reactor coolant.

The maximum content of a gas decay tank given in table 15.7.1-1 is based on conservative assumptions used for the purpose of computing the noble gas and iodine inventory available for release. Rupture of the WGDT is assumed to occur immediately upon completion of the waste gas transfer, releasing the entire contents of the tank to the auxiliary building. For the purposes of evaluating the accident, it is assumed that all the activity is released directly to the environment during the 2-h period immediately following the accident.

15.7.1.5.1.2 <u>Assumptions and Conditions</u>. The major assumptions and parameters assumed in the analysis are itemized in table 15.7.1-1.

In the evaluation of the WGDT rupture, the fission product accumulation and release assumptions of Regulatory Guide 1.24 have been used. Table 15.7.1-2 provides a comparison of the assumptions used in the analysis to those of Regulatory Guide 1.24. The assumptions related to the release of radioactive gases from the postulated rupture of a WGDT are:

- A. The iodine concentrations in the reactor coolant, prior to shutdown, are based on  $1 \ \mu$ Ci/g of dose equivalent I-131. Coincident with the shutdown, an iodine spike is created which increases the iodine release rate from the fuel to the primary coolant to a value 500 times greater than the release rate corresponding to the maximum equilibrium primary system iodine concentration of  $1 \ \mu$ Ci/g of dose equivalent I-131. The duration of the spike is assumed to be sufficient to raise the reactor coolant iodine concentration to 60  $\mu$ Ci/g of dose equivalent I-131 (approximately 2.5 h).
- B. The noble gas concentrations in the primary coolant are based on 1-percent defective fuel.
- C. All gaseous activity has been removed from the reactor coolant system (RCS) and transferred to the gas decay tank that is assumed to fail.
- D. The maximum content of the WGDT was conservatively assumed to be the isotopic activities given in table 15.7.1-1 for the accumulated radioactivity in the GWPS after 40 years' operation<sup>a</sup> and immediately following plant shutdown and degasification of the RCS.
- E. The failure is assumed to occur immediately upon completion of the waste gas transfer, releasing the entire contents of the tank to the auxiliary building.
- F. The dose is calculated as if the release were from the auxiliary building at ground level during the 2-h period immediately following the accident. No credit for radioactive decay is taken.

15.7.1.5.1.3 <u>Mathematical Models Used in the Analysis</u>. The mathematical models used in the analysis are described in the following sections:

- A. The mathematical models used to analyze the activity released during the course of the accident are described in appendix 15A.
- B. The atmospheric dispersion factors used in the analysis were calculated based on the onsite meteorological measurement programs described in section 2.3.
- C. The thyroid inhalation and total body gamma immersion doses to a receptor at the exclusion area boundary and outer boundary of the low population zone were analyzed, using the models described in appendix 15A, subsections 15A.2.4 and 15A.2.6, respectively.

15.7.1.5.1.4 <u>Identification of Leakage Pathways and Resultant Leakage Activity</u>. For the purpose of evaluating the radiological consequences due to the postulated WGDT rupture, the resultant activity is conservatively assumed to be released directly to the environment during the 2-h period immediately following the occurrence of the accident. This is a considerably higher

<sup>&</sup>lt;sup>a</sup> The renewed operating licenses authorized a 20-year period of extended operation for both VEGP units, resulting in a total plant operating life of 60 years. Since the inventory in the WGDTs has been routinely released during the first 20 years of operation and is expected to continue to be routinely released during future operation, the inventory of the WGDTs accumulated during the first 20 years of operation will be released prior to entering the period of extended operation. Therefore, the stated design capacity of the GWPS remains sufficient, and the analysis of the maximum fission product inventory in the GWPS over a 40-year plant life remains bounding for a 60-year plant life.

release rate than that based on the actual building exhaust ventilation rate. Therefore, the results of the analysis are based on the most conservative pathway available.

### 15.7.1.5.2 Identification of Uncertainties and Conservatisms in the Analysis.

The uncertainties and conservatisms in the assumptions used to evaluate the radiological consequences of a WGDT rupture result from assumptions made involving the release of the waste gas from the decay tank and the meteorology present at the site during the course of the accident.

- A. The iodine inventory in the GWPS is based on a reactor coolant concentration of  $1 \mu \text{Ci/g}$  of dose equivalent I-131 with extremely large iodine spike values, persisting for 2.5 h, and resulting in equivalent concentrations many times greater than the reactor coolant activities based on 0.12-percent defective fuel associated with normal operating conditions.
- B. The noble gas inventory in the GWPS is based on a reactor coolant concentration corresponding to 1-percent defective fuel and 40 years of operation<sup>a</sup> (to maximize the Kr-85 inventory; all other nuclides equilibrate in approximately 60 days or less). Furthermore, 1-percent defects cannot exist simultaneously with 1.0  $\mu$ Ci/g of dose equivalent I-131. For iodines, 1-percent defects would be approximately three times the Technical Specification limit.
- C. It is assumed that the WGDT fails immediately after the transfer of the noble gases and iodines from the reactor coolant to the WGDT is complete. These assumptions result in the greatest amount of gaseous activity available for release to the environment.
- D. The gaseous activity contained in the ruptured WGDT was assumed to be released over a 2-h period immediately following the accident. This is a conservative assumption. If the contents of the tank were assumed to mix uniformly with the volume of air within the auxiliary building where the decay tanks are located, then, using the actual building exhaust ventilation rate, a considerable amount of holdup time would be gained. This reduces, by natural decay, the amount of gaseous activity available for release to the environment. However, no credit for radioactive decay is taken.
- E. The meteorological conditions which may be present at the site during the course of the accident are uncertain. However, it is highly unlikely that meteorological conditions assumed will be present during the course of the accident for any extended period of time. Therefore, the radiological consequences evaluated, based on these meteorological conditions, will be conservative.

<sup>&</sup>lt;sup>a</sup> The renewed operating licenses authorized a 20-year period of extended operation for both VEGP units, resulting in a total plant operating life of 60 years. Since the inventory in the WGDTs has been routinely released during the first 20 years of operation and is expected to continue to be routinely released during future operation, the inventory of the WGDTs accumulated during the first 20 years of operation will be released prior to entering the period of extended operation. Therefore, the stated design capacity of the GWPS remains sufficient, and the analysis of the maximum fission product inventory in the GWPS over a 40-year plant life remains bounding for a 60-year plant life.

## 15.7.1.5.3 Conclusions

15.7.1.5.3.1 <u>Filter Loading</u>. Since the accumulated iodine activity in the WGDTs is negligible, filter loading due to WGDT rupture does not establish the necessary design margin for the auxiliary building exhaust or the control room intake filters. Hence, the respective filter loadings were not evaluated.

15.7.1.5.3.2 Dose to Receptor at the Exclusion Area Boundary and the Low Population Zone Outer Boundary. The radiological consequences resulting from the occurrence of a postulated WGDT rupture have been conservatively analyzed, using assumptions and models described in previous sections.

The total body gamma dose due to immersion and the thyroid dose due to inhalation have been analyzed for the 0- to 2-h dose at the exclusion area boundary and for the duration of the accident at the low population zone outer boundary. The results are listed in table 15.7.1-3. The resultant doses are well within the guideline values of 10 CFR 100.

## 15.7.2 RADIOACTIVE LIQUID WASTE SYSTEM LEAK OR FAILURE

#### 15.7.2.1 Identification of Causes

This is an infrequent fault, but the potential for release of significant amounts of radioactivity is present. The accident may be caused by an equipment malfunction or tank failure.

Liquid radwaste system leaks resulting in gaseous releases to the atmosphere and liquid releases to the ground water are bounded by the tank failure analyses presented in paragraphs 15.7.2.5 and 15.7.3.4, respectively. Accidental releases of radwaste processing facility liquid effluents to surface water are discussed in paragraph 2.4.13.2.

### 15.7.2.2 Sequence of Events and System Operation

The recycle holdup tank (RHT) is assumed to fail. This releases 100 percent of the tank capacity to the tank compartment.

#### 15.7.2.3 Core and System Performance

This accident does not affect the core or the core system performance.

#### 15.7.2.4 Barrier Performance

It is assumed that there are no barriers to the release of radioactivity from the auxiliary building.

# 15.7.2.5 Radiological Consequences

# 15.7.2.5.1 Method of Analysis

15.7.2.5.1.1 <u>Physical Model</u>. Reactor coolant shim bleed and some valve leakage is held in the RHT.

Table 15.7.3-1 provides an inventory and the concentrations of stored radioactivity in the tank. In the analyses, it is assumed that the liquid contents of the tank are released to the auxiliary building, and subsequently the airborne activity is released to the environment during the 2-h period immediately following the tank failure.

The RHT was selected because it contains the maximum total inventory.

15.7.2.5.1.2 <u>Assumptions and Conditions</u>. The major assumptions and parameters assumed in this analysis are listed below and in table 15.7.2-1.

- A. The nuclide inventory of the failed tank is taken from table 15.7.3-1 and is based on 1-percent defective fuel.
- B. The RHT failure is assumed to occur when the contents of the tank are at a maximum.
- C. The doses are calculated as if the release were from the auxiliary building at ground level during the 2-h period immediately following the accident. No credit is taken for radioactive decay during holdup in the tank or in transit to the site boundary.
- D. One-hundred percent of all noble gas activity in the tank is released while 1 percent of the iodine activity is released as airborne activity.
- E. Credit is not taken for iodine removal by the nonsafety-grade auxiliary building heating, ventilation, and air-conditioning (HVAC) charcoal adsorber.

### 15.7.2.5.2 Mathematical Models Used in the Analysis.

- A. The mathematical models used to analyze the activity released during the course of the accident are described in appendix 15A.
- B. The atmospheric dispersion factors used in the analysis were calculated based on the onsite meteorological measurement program described in section 2.3; they are provided in table 15A-2.
- C. The thyroid inhalation dose and total body immersion dose to a receptor at the exclusion area boundary or outer boundary of the low population zone were analyzed, using the models described in appendix 15A, sections 15A.2.4, and 15A.2.6, respectively.

15.7.2.5.2.1 <u>Identification of Leakage Pathways and Resultant Leakage Activity</u>. For the purpose of evaluating the radiological consequences due to the postulated RHT failure, the

resultant activity is conservatively assumed to be released directly to the environment during the 2-h period immediately following the occurrence of the accident. This is a considerably higher release rate than that based on the actual building exhaust ventilation rate. Therefore, the results of the analysis are based on the most conservative pathway available.

## 15.7.2.5.3 Identification of Uncertainties and Conservatisms in the Analysis

The uncertainties and conservatisms in the assumptions used to evaluate the radiological consequences of the RHT failure result from assumptions made involving the release of the radioactivity from the tank and the meteorology assumed for the site.

- A. It was assumed that the RHT fails when the inventory in the tank is a maximum. This assumption results in the greatest amount of activity available for release to the environment.
- B. The contents of the failed tank are assumed to be released over a 2-h period immediately following the accident. If the contents of the tank were assumed to mix uniformly with the volume of air within the auxiliary building where the tank is located, then, using the actual building exhaust ventilation rate, a considerable amount of holdup time would be gained. This reduces the amount of activity released to the environment due to the natural decay. Also, no credit is taken for iodine removal by the auxiliary building HVAC charcoal adsorbers.
- C. The meteorological conditions which may be present at the site during the course of the accident are uncertain. However, it is highly unlikely that meteorological conditions assumed will be present during the course of the accident for any extended period of time.
- D. The RHT is assumed to have collected liquid waste based on operation at 100-percent power with 1-percent defective fuel for an extended period of time.

## 15.7.2.5.4 Conclusions

15.7.2.5.4.1 <u>Filter Loadings</u>. The filter loading due to an RHT failure does not establish the necessary design margin for the control room intake filters. Thus, the filter loading was not evaluated.

15.7.2.5.4.2 <u>Doses to Receptor at the Exclusion Area Boundary and the Low Population</u> <u>Zone Outer Boundary</u>. The radiological consequences resulting from the occurrence of a postulated liquid radwaste tank failure have been conservatively analyzed, using assumptions and models described in previous sections.

The total body dose due to immersion and the thyroid dose due to inhalation have been analyzed for the 0- to 2-h dose at the exclusion area boundary and for the duration of the accident at the low population zone outer boundary. The results are listed in table 15.7.2-2. The resultant dose is well within the guideline values of 10 CFR 100.

# 15.7.3 POSTULATED RADIOACTIVE RELEASE DUE TO LIQUID TANK FAILURE (GROUND RELEASE)

## 15.7.3.1 Identification of Causes and Frequency Classification

This accident is defined as an unexpected and uncontrolled postulated rupture of the recycle holdup tank (RHT). This tank is located in the Seismic Category 1 auxiliary building at el 119 ft 3 in. Since plant grade is at el 220 ft, the only way any effluents from the postulated rupture can be released accidentally is through postulated cracks in the auxiliary building, which would allow the contents of the tank to enter ground water. This accident is postulated to occur with the frequency of a limiting fault; in all other respects it is to be considered an infrequent, Condition III, event. (See paragraph 15.7.2.1.)

## 15.7.3.2 <u>Sequence of Events and Systems Operation</u>

See subsection 2.4.13.

### 15.7.3.3 Modeling of Accident Sequence

### 15.7.3.3.1 Mathematical Model

Subsection 2.4.13 gives the dispersion, dilution, and travel times of accidental releases of liquid effluents in surface water.

### 15.7.3.3.2 Input Parameters and Initial Conditions

The tank failure is evaluated in accordance with the following sets of assumptions and conditions:

- A. One-hundred percent of the liquid volume of the RHT is released into the RHT cubicle.
- B. The liquid enters the ground water environment through postulated cracks in the auxiliary building. RHT data is provided in table 15.7.3-1.

### 15.7.3.4 Radiological Consequences

The radiological consequences of this accident are presented in subsection 2.4.13.

The concentrations of any postulated accidental release of radioactive effluents from the RHT would not exceed 10 CFR 20 limits at the nearest surface water intake.

### 15.7.4 FUEL HANDLING ACCIDENTS

The postulated fuel handling accident has been analyzed for three cases: case 1, a fuel handling accident outside the containment in the fuel handling building; case 2, a fuel handling

accident inside the reactor containment building with the containment airlocks and equipment hatch closed; and case 3, a fuel handling accident inside the containment with the personnel airlock doors and/or the equipment hatch open.

### 15.7.4.1 Identification of Causes and Accident Description

The accident is defined as dropping of a spent fuel assembly onto another fuel assembly in the fuel storage area or refueling pool, resulting in the rupture of the cladding of all the fuel rods in the dropped assembly plus additional rods in the struck assembly (for the accident inside containment), despite many administrative controls and physical limitations imposed on fuel handling operations. All refueling operations are conducted in accordance with prescribed procedures.

## 15.7.4.2 <u>Sequence of Events and Systems Operations</u>

The first step in fuel handling is the safe shutdown and cooldown of the reactor. After a radiation survey of the containment, the disassembly of the reactor vessel is started. After disassembly is complete, the first fuel handling is started. The first fuel transfer operation shall not begin until at least 90 h after shutdown.

The fuel handling accident is assumed to occur after a fuel assembly has been removed from the core but before it has been placed in its designated location in the spent fuel storage racks.

### 15.7.4.3 Core and System Performance

The fuel handling accident in the containment building or the fuel building does not impact the integrity of the core or its system performance.

### 15.7.4.4 Barrier Performance

The barriers between the released activity and the environment are the containment building or the fuel building. Since these buildings are designed Seismic Category 1, it is safe to assume that during the course of a fuel handling accident their integrity is maintained. Normally, release of radioactivity for a postulated accident in the fuel building is via the fuel building emergency filtration system. An open door in the fuel handling building pressure boundary could create another release path.

For a postulated accident in the containment building with the airlocks and equipment hatch closed, the release is limited to the minimal amount of radioactivity which could potentially be released prior to containment isolation. For a postulated accident in the containment building with the airlock and/or equipment hatch open, the limiting pathway for the release of activity is via the equipment building ventilation fan. During core alterations or movement of irradiated fuel assemblies within containment, the air lock door interlock mechanism may remain disabled, but the air lock must always be isolable by at lease one air lock door with a designated individual available to close the air lock door, or at least one air lock door must be closed. Similarly, the equipment hatch must be isolable and capable of being held in place by four bolts. The requirements for containment penetration closure are sufficient to ensure fission product radioactivity release from containment due to a fuel handling accident during refueling is

maintained to within the acceptance criteria of Standard Review Plan subsection 15.7.4 and General Design Criteria 19.

The equipment hatch and the emergency air lock are farther away from the control room air intake than the personnel airlock. Therefore, the release path from the personnel airlock remains bounding for control room dose. Similarly, potential release paths from the purge supply and exhaust ductwork are no closer than the personnel airlock release path. Offsite dose is not affected by the relative locations of the personnel and emergency airlocks, the containment purge supply and exhaust ventilation, or the equipment hatch.

The spent fuel pool and the refueling pool provide a minimum decontamination factor of 200 for elemental iodine.

## 15.7.4.5 <u>Radiological Consequences</u>

### 15.7.4.5.1 Method of Analysis

15.7.4.5.1.1 <u>Physical Model</u>. The possibility of a fuel handling accident is remote because of the many administrative controls and physical limitations imposed on the fuel handling operations. (Refer to subsection 9.1.4.) All refueling operations are conducted in accordance with prescribed procedures.

When transferring irradiated fuel from the core to the spent fuel pool for storage, the reactor cavity and refueling pool are filled with borated water at a boron concentration equal to or greater than that concentration specified for the spent fuel pool or that concentration specified for refueling, whichever is highest, which ensures subcritical conditions in the core even if all rod cluster control (RCC) assemblies are withdrawn. After the reactor head and RCC drive shafts are removed, fuel assemblies are lifted from the core, transferred vertically to the upender, lowered to a horizontal position on the transfer car and pulled through the transfer tube and canal, upended and transferred through the spent fuel pool transfer gate, then lowered into steel racks for storage in the spent fuel pool in a pattern which precludes any possibility of a criticality accident.

Fuel handling manipulators and hoists are designed so that the fuel cannot be raised above a position that provides an adequate water shield depth for radiation protection of operation personnel.

The containment, fuel building, refueling cavity, refueling pool, and spent fuel pool are designed to Seismic Category 1 requirements, which prevent the structures themselves from failing in the event of a safe shutdown earthquake. The spent fuel storage racks are also located to prevent any credible external missile from reaching the stored irradiated fuel. The fuel handling manipulators, cranes, trollies, bridges, and associated equipment above the water cavities through which the fuel assemblies move are designed to prevent this equipment from generating missiles and damaging the fuel. The construction of the fuel assemblies precludes damage to the fuel should portable or hand tools drop on an assembly.

A fuel handling accident could occur during the transfer of a fuel assembly from the core to its storage position in the spent fuel pool. The facility is designed so that heavy objects, such as a spent fuel cask, cannot be carried over or tipped over onto the irradiated fuel stored in the spent fuel pool. Only one fuel assembly can be handled at a time. Movement of equipment handling

the fuel is kept at low speeds, while exercising caution that the fuel assembly does not strike another object or structure during transfer from the core to its storage position. In the unlikely event that an assembly becomes stuck in the transfer tube, natural convection will maintain adequate cooling.

#### A. Containment Building Accident

During core alterations or movement of irradiated fuel assemblies inside containment, the containment may be either open or closed. If the containment is closed, the equipment hatch will be held in place with at least four bolts, the air locks will be isolated by at least one closed air lock door, and each penetration providing direct access from the containment atmosphere to the outside atmosphere (with the exception of the containment purge and exhaust penetrations) will be closed in accordance with the Technical Specifications. The containment purge and exhaust penetrations will be capable of being closed, by operator action, by at least two containment ventilation isolation valves. The containment radiation monitors (gaseous, particulate, iodine, and area low range) will be operable in accordance with the Technical Specifications. In addition, the Technical Requirements Manual requires that direct communications be maintained between the control room and personnel at the refueling station during core alterations. If a fuel handling accident were to occur inside containment with containment closed, the control room would be immediately aware of the event as a result of direct communication or a radiation alarm. Steps would be taken to isolate containment purge and exhaust, and personnel would be evacuated. With containment closed, the only potential release pathway would be via the containment purge and exhaust system. However, because of the gaseous, particulate, and iodine monitors in the exhaust portion of the system, the potential for an unmonitored release is minimized. In addition, the purge exhaust system is equipped with HEPA filters and charcoal adsorbers which will further minimize any radiological release. While no credit is taken for filtration by the purge exhaust system in the dose analyses, the availability of the filters will be maintained in accordance with NUMARC 93-01 guidance.

With containment open, the equipment hatch and/or the air lock doors may be open. The equipment hatch must be capable of being closed by at least four bolts and the air locks must be isolable by at least one door with a designated individual to close the open doors. A designated hatch closure crew and the necessary tools and equipment will be available to effect timely closure of the hatch. The equipment hatch will be capable of being cleared of obstructions so that closure can be achieved as soon as possible. The air lock doors will be closed within 15 minutes of a fuel handling accident. For the equipment hatch, the current commitment for closure time in response to a loss of decay removal capability during reduced inventory conditions is 25 minutes. This closure time is bounding for the case of a fuel handling accident inside containment with the equipment hatch open. Radiation monitor operability, requirements for other penetrations providing direct access from containment atmosphere to the outside atmosphere, communication requirements, and purge exhaust system availability is the same as for the case discussed above for containment closed. During core offload and reload with the equipment hatch open, the containment purge exhaust system will normally be operating providing an inward flow of air into containment. This is consistent with NUMARC 93-01, section 11.3.6.5, which states that the goal of maintaining ventilation system and radiation monitor availability is to reduce doses even further below that provided by natural decay

### VEGP-FSAR-15

and to avoid unmonitored releases. However, the purge exhaust system must be shut down during hatch closure activities, and it may be shut down during fuel handling while the hatch is open as required to maintain noise and comfort levels, etc. If for any reason operation of the purge exhaust system must be discontinued during core alterations/fuel movement with the hatch open, the opening will be monitored for radioactive releases via the health physics air monitoring station.

B. Fuel Building Accident

In the fuel building, a fuel assembly could be dropped in the transfer canal or in the spent fuel pool.

In addition to the area and effluent radiation monitors, portable radiation monitors capable of emitting audible alarms are located in this area during fuel handling operations. The doors in the fuel building are normally kept closed to ensure controlled leakage characteristics in the spent fuel pool region during operations involving irradiated fuel. Doors on the pressure boundary, except for the railroad bay door, may be held open during operations involving irradiated fuel. The fuel handling building normal and post-accident ventilation system function is not lost due to opening of personnel doors. Administrative controls are in place to close the doors after a fuel handling accident to minimize any potential release to the environment. Additionally, if none of the fuel handling building exhaust filter units are in service, one should be placed in service to ensure flow past the radiation monitors, or the doors should be closed. Should a fuel assembly be dropped in the canal or in the pool and release radioactivity above a prescribed level, the radiation monitors would sound an alarm. (See section 11.5 and subsection 12.3.4.) On alarm signal, the fuel building ventilation is switched to the emergency mode and exhausts through the engineered safety features (ESF) emergency filtration system charcoal and high-efficiency particulate air filters to remove most of the halogens and particulates prior to discharging to the atmosphere via the plant vent.

A radiation monitor located in the fuel handling building ventilation exhaust duct sounds an alarm if the radioactivity in the vent discharge exceeds the prescribed level.

The probability of a fuel handling accident is very low because of the safety features, administrative controls, and design characteristics of the facility, as previously mentioned.

15.7.4.5.1.2 <u>Assumptions and Conditions</u>. The major assumptions and parameters assumed in the analysis are itemized in tables 15.7.4-1 and 15A-1.

In the evaluation of the fuel handling accident, the fission product release assumptions of Regulatory Guide 1.25 are followed. Table 15.7.4-2 provides a comparison of the design to the requirements of Regulatory Guide 1.25. The following assumptions, related to the release of fission product gases from the damaged fuel assembly, are used in the analyses except as identified in table 15.7.4-2:

A. The dropped fuel assembly is assumed to be the assembly containing the peak fission product inventory. All the fuel rods contained in the dropped assembly are assumed to be damaged. In addition, for the analyses of the accident in the containment building the dropped assembly is assumed to damage 50 rods of an additional assembly.

- B. The assembly fission product inventories are based on a radial peaking factor of 1.70.
- C. The accident occurs 90 h after shutdown, which is the earliest time fuel handling operations can begin. Radioactive decay of the fission product inventories was taken into account during this time period.
- D. Only that fraction of the fission products which migrates from the fuel matrix to the gap and plenum regions during normal operation is assumed to be available for immediate release to the water following clad damage.
- E. The gap activity released to the fuel pool from the damaged fuel rods consists of 5% of the total noble gases and iodines, other than Kr-85 (10%) and I-131 which is 8% (Reg. Guide 1.195).
- F. The pool decontamination factor is 1.0 for noble gases and organic iodine.
- G. The effective pool decontamination factor is 200 for elemental iodine.
- H. The iodine released from the fuel assembly is assumed to be composed of 99.75-percent inorganic and 0.25-percent organic species.
- I. The activity which escapes from the pool is assumed to be available for release to the environment in a time period of 2 h.
- J. No credit for decay or depletion during transit to the exclusion area boundary or the outer boundary of the low population zone is assumed.
- K. No credit is taken for mixing or holdup in the fuel building atmosphere or equipment building.
- L. For the case inside the reactor containment building, conservative credit is taken for mixing of the radioactivity released from the refueling pool with a minimum of the containment building free volume.

The mixing volume of 25 percent is assumed and is based on the normal airflow rate of four fan coolers.

- M. The containment purge rate is 15,000 ft<sup>3</sup>/min for the case with the containment airlocks and equipment hatch closed.
- N. Automatic containment ventilation isolation capability is no longer required by the Technical Specifications. The limiting radiological consequences for a fuel handling accident inside containment as reported in table 15.7.4-4 are based on an open containment with no automatic isolation. However, the radiological consequences for a closed containment reported in table 15.7.4-4 are based on automatic containment ventilation isolation. If automatic containment ventilation isolation is available, it would be assumed to occur within 10 s from the time the containment isolation signal is generated with a 5-s signal generation time.
- O. The control room emergency filtration system (CREFS) is initiated by RE-12116 and/or RE-12117 during a fuel handling accident in the fuel handling building.
- P. For the case with the airlocks and/or equipment hatch open, the activity is assumed to be released from the containment to the outside atmosphere at a rate in which all the activity released from the damaged fuel assemblies would be released to the outside atmosphere in 2 hours if the containment airlocks and/or equipment hatch remain open. The radiological consequences of a fuel handling accident in containment have been evaluated assuming that the containment is

## VEGP-FSAR-15

open to the outside atmosphere. All airborne activity reaching the containment atmosphere is assumed to be exhausted to the environment within 2 hours of the accident. The calculated offsite and control room operator doses are within the acceptance criteria of Standard Review Plan 15.7.4 and General Design Criteria 19. Therefore, although the containment penetrations do not satisfy any of 10 CFR 50.36 (c)(2)(ii) criteria, Technical Specifications provide containment closure capability to minimize potential offsite doses. Procedures provide administrative controls to ensure that the designated person available to close the personnel and/or emergency air lock doors does not have other duties that would preclude the ability to operate the door in a timely manner. In addition, a designated hatch closure crew is available to effect a timely closure of the equipment hatch.

Q. The control room volume, normal and emergency mode flow rates, and emergency mode filter efficiencies for the control room ventilation system used to determine control room doses following a fuel handling accident with the containment airlocks and/or equipment hatch open are given in table 15A-1 of appendix 15A.

15.7.4.5.1.3 <u>Mathematical Models Used in the Analysis</u>. Mathematical models used in the analysis are described in the following sections:

- A. The mathematical models used to analyze the activity released during the course of the accident are described in appendix 15A, section 15A.2.
- B. The atmospheric dispersion factors are based on the onsite meteorological measurement described in section 2.3 and are provided in table 15A-2.
- C. The thyroid inhalation and total-body immersion doses to a receptor located at the exclusion area boundary and outer boundary of the low population zone are analyzed using the models described in appendix 15A, subsections 15A.2.4 and 15A.2.6, respectively.
- D. The thyroid inhalation, beta skin, and gamma body doses to personnel in the control room are analyzed using the models described in appendix 15A, and subsections 15A.3.3, 15A.3.4, and 15A.3.5.

15.7.4.5.1.4 <u>Identification of Leakage Pathways and Resultant Leakage Activity</u>. For evaluating the radiological consequences due to the postulated fuel handling accident in the fuel building, the resultant activity is conservatively assumed to be released to the environment during the 0- to 2-h period immediately following the occurrence of the accident. This is a considerably higher release rate than that based on the actual ventilation rate. Therefore, the results of the analysis are based on the most conservative pathway available. Only the limiting case of a fuel handling accident is shown in table 15.7.4-1.

# 15.7.4.5.2 Identification of Uncertainties and Conservatisms in Analysis

The uncertainties and conservatisms in the assumptions used to evaluate the radiological consequences of a fuel handling accident result from assumptions made involving the amount of fission product gases available for release to the environment and the meteorology present at the site during the course of the accident. The most significant of these assumptions are:

- A. It is assumed in the analysis that all the fuel rods in the dropped assembly are damaged. This is a highly conservative assumption, since in transferring fuel under strict fuel handling procedures, only under the worst possible circumstances could the dropping of a spent fuel assembly result in damage to all the fuel rods contained in the assembly.
- B. The fission product gap inventory in a fuel assembly is dependent on the power rating of the assembly and the temperature of the fuel. The gap fractions from Regulatory Guide 1.195 are conservatively assumed. Realistic calculations of gap fractions show less than 2% for all short lived isotopes.
- C. lodine removal from the released fission product gas takes place as the gas rises to the pool surface through the body of liquid in the spent fuel pool. The extent of elemental iodine removal is determined by mass transfer from the gas phase to the surrounding liquid and is controlled by the bubble diameter and contact time of the bubble in the solution. The values used in the analysis result in a release of elemental iodine approximately a factor of 3 greater than anticipated. The release of activity from the pool to the containment atmosphere is time dependent, and, consequently, there would be sufficient time for this activity to mix homogeneously in a significantly greater percent of the containment volume than assumed in the analysis.
- D. Fuel handling building emergency filtration system charcoal filters are provided; however, no credit has been taken for their capability. This means a reduction in the iodine concentrations and, thus, a reduction in the thyroid doses at the exclusion area boundary and the outer boundary of the low population zone.
- E. The containment purge exhaust system has charcoal adsorber units which filter any containment purge release. However, no credit has been taken for its capability (90% efficiency, minimum) since these units are not specifically designed to Seismic Category 1 criteria. It is expected that for any event which would produce a catastrophic failure of the charcoal adsorber unit to the extent that its filtering capability would be negated would also result in the purge exhaust fan becoming inoperable. Therefore, failure within the purge exhaust system would terminate any high-volume release from the containment. In fact, the purge exhaust fan is considerably more likely to be inoperable following any postulated event than the failure of a passive charcoal adsorber unit. Thus, although no credit in the analysis has been given for the normal purge exhaust filters, any release prior to containment isolation would be filtered, reducing the calculated releases by another factor of 10.
- F. There is also conservatism in the time to first fuel transfer. Despite the fact that fuel could be transferred at 90 h, it is probable that fuel handling will begin sometime later.
- G. The exhaust from the personnel airlock area is via the equipment building ventilation fan (1526-B7-00200). The distance from the equipment building ventilation exhaust to the control room (CR) intake is 190 ft. If both the intake and exhaust fans fail (loss of power to nonsafety-grade equipment), activity may exfiltrate through the equipment building intake. The distance from the equipment building ventilation intake to the CR intake is 90 ft. For the case of the fuel handling accident inside containment with the personnel airlock open, the activity is assumed to be released from the equipment building to the environment and from the environment to the CR intake. Furthermore, the CR dose analysis conservatively assumes that activity is released from the

equipment building intake, which is closer to the CR than the equipment building exhaust. This assumption is limiting with respect to releases from the equipment hatch or the emergency air lock.

- H. The distance from the CR intake to the nearest point on the containment is 70 ft. The atmospheric dispersion factors ( $\chi/Q$ ) at the CR intake used to determine the CR doses following a LOCA are based on this distance of 70 ft. The distance of 90 ft from the equipment building intake to the CR intake is comparable to the distance of 70 ft used in the LOCA CR dose analysis. Thus, the same  $\chi/Q$  as used in the LOCA CR dose analysis, which are listed in table 15A-2, are conservatively used for the CR dose analysis for a fuel handling accident inside containment with the personnel airlock open.
- I. The assumption that all radioactivity released due to the fuel handling accident is released from the containment and to the outside atmosphere in the initial 2 h following the accident if the containment remains open is conservative. There is no driving force to push this activity out of the containment. The bulk of the radioactivity would likely stay in the containment for much longer than 2 h, even if the containment remained open.
- J. The meteorological conditions which may be present at the site during the course of the accident are uncertain. However, it is highly unlikely that meteorological conditions assumed will be present during the course of the accident for any extended period of time. Therefore, the radiological consequences evaluated, based on the meterological conditions assumed, are conservative.

15.7.4.5.2.1 <u>Filter Loadings</u>. The filtration systems which function to limit the consequences of a fuel handling accident in the fuel building are the fuel building emergency filtration system and the control room filtration system.

The activity loadings on the control room charcoal adsorbers as a function of time have been evaluated for the loss-of-coolant accident (LOCA), as described in subsection 15.6.5. Since these filters are capable of accommodating the design basis LOCA fission product iodine loadings, more than adequate design margin is available with respect to postulated fuel handling accident releases.

The activity loadings on the ESF filtration system charcoal adsorbers have been evaluated in accordance with Regulatory Guide 1.52, which limits the maximum loading to 2.5 mg iodine/g activated charcoal.

15.7.4.5.2.2 Doses to Receptor at the Exclusion Area Boundary and Low Population Zone Outer Boundary. The potential radiological consequences resulting from the occurrence of a postulated fuel handling accident occurring in the fuel building and in the reactor building have been conservatively analyzed, using assumptions and models described in previous sections. The total-body dose due to immersion from direct radiation and the thyroid dose due to inhalation have been analyzed for the 0- to 2-h dose at the exclusion area boundary and for the duration of the accident (0 to 2 h) at the low population zone outer boundary. The results are listed in table 15.7.4-4. The resultant doses are well within the guideline values of 10 CFR 100.

## 15.7.5 SPENT FUEL CASK DROP ACCIDENT

The spent fuel cask will follow the path outlined on drawing AX4DE501. Cask handling over the spent fuel pool or the new fuel pit is prevented by interlocks.

A Type 1 single-failure-proof crane designed according to NUREG-0554 is used in handling the spent fuel cask. Therefore, no cask drop will occur, and thus no radioactivity will be released. Refer to subsection 9.1.5 for a description of the spent fuel cask handling equipment.

## VEGP-FSAR-15

### TABLE 15.7.1-1

#### PARAMETERS USED IN EVALUATING THE RADIOLOGICAL CONSEQUENCES OF A WASTE GAS DECAY TANK RUPTURE

## I. Source Data

II.

III.

I-134

I-135

A. Core power level (MWt)		3636
<ul> <li>B. Reactor coolant iodine activity</li> <li>C. Volume control tank purge (sf<sup>3</sup>/min)</li> </ul>		Initial activity equal to 1.0 $\mu$ Ci/g of dose equivalent I-131 with a shutdown iodine spike that increases the activity to 60 $\mu$ Ci/g of dose equivalent I-131. (See tables 15A-6 and 15A-7.) 0.7
D. Number of gas decay tanks per unit		7
E. Number of tanks shared by Units 1 and 2		2
F. Tank switching time for normal operation (days)		2
G. Shutdown degassing (h)		24
Atmospheric Dispersion Factors		See table 15A-2.
Activity Release Data (Maximum/Tank)		
<u>Nuclide</u>	<u>0 to 2 h (Ci)</u>	
Kr-85 Kr-85m Kr-87 Kr-88 Xe-131m Xe-133 Xe-133m Xe-135 Xe-135m Xe-138 I-131 I-132 I-133	6.42E+3 1.33E+1 6.30E-4 2.66E+0 1.65E+2 2.94E+4 2.80E+3 4.15E+2 Negligible Negligible 3.71E+0 7.80E-1 3.59E+0	

1.02E-1

1.16E+0
### TABLE 15.7.1-2 (SHEET 1 OF 5)

### DESIGN COMPARISON TO THE REGULATORY POSITIONS OF REGULATORY GUIDE 1.24. ASSUMPTIONS USED FOR EVALUATING THE POTENTIAL RADIOLOGICAL CONSEQUENCES OF A PRESSURIZED WATER REACTOR RADIOACTIVE GAS STORAGE TANK FAILURE, REVISION 0, MARCH 23, 1972

Regulatory Guide 1.24 Position

Design

- 1. The assumptions related to the release of radioactive gases from the postulated failure of a gaseous waste storage tank are:
  - The reactor has been operating at full Α. power with 1 percent defective fuel and a shutdown to cold condition has been conducted near the end of an equilibrium core cycle. As soon as possible after shutdown, all noble gases have been removed from the primary cooling system and transferred to the gas decay tank that is assumed to fail.
  - Β. The maximum content of the decay tank assumed to fail should be used for the purpose of computing the noble gas inventory in the tank. Radiological decay may be taken into account in the computation only for the minimum time period required to transfer the gases from the primary system to the decay tank.
  - C. The failure is assumed to occur immediately upon completion of the waste gas transfer, releasing the entire contents of the tank to the building. The assumption of the release of the noble gas inventory from only a single tank is based on the premise that all gas decay tanks will be isolated from each other whenever they are in use.

1.A conforms.

1.B conforms.

1.C conforms.

### TABLE 15.7.1-2 (SHEET 2 OF 5)

### Regulatory Guide 1.24 Position

### <u>Design</u>

1.D conforms.

- D. All of the noble gases are assumed to leak out of the building at ground level over a 2-h time period.
- 2. The atmospheric diffusion assumptions for ground level releases are:
  - A. The basic equation for atmospheric diffusion from a ground level point source is

$$\chi/Q = \frac{1}{\pi u \sigma_y \sigma_z}$$

where:

- χ = the short term average centerline value of the ground level concentration (Ci/s)
- Q = amount of material released (Ci/s)
- u = windspeed (m/s)
- $\sigma_y$  = the horizontal standard deviation of the plume (m). See figure V-1, page 48, F. A. Gifford, Jr., Use of Routine Meteorological Observation for Estimating Atmospheric Dispersion, <u>Nuclear Safety</u>, Vol. II, No. 4, June 1961.

- Short torm atmosphori
- 2. Short-term atmospheric dispersion factors corresponding to a ground level release and accident conditions were calculated based on onsite meteorological measurement programs described in Regulatory Guide 1.1.45 and represent the worst of the 5-percent overall site meteorology and the 0.5percent worst sector meteorology.

### TABLE 15.7.1-2 (SHEET 3 OF 5)

Regulatory Guide 1.24 Position

### <u>Design</u>

- σ<sub>z</sub> = the vertical standard deviation of the plume (m). See figure V-2, page 48 in Gifford, Use of Routine Meteorological Observations...
- For ground level releases, atmospheric diffusion factors<sup>(a)</sup> used in evaluating the radiological consequences of the accident addressed in this guide are based on the following assumptions:
  - 1. Windspeed of 1 m/s.
  - 2. Uniform wind direction.
  - 3. Pasquill diffusion category F.
- C. Figure 1 is a plot of atmospheric diffusion factors  $(\chi/Q)$  versus distance derived by use of the equation for a ground level release given in regulatory position 2.A above under the meteorological conditions given in regulatory position 2.B above.
- The following assumptions and equations may be used to obtain conservative approximations of external whole body dose from radioactive clouds:
  - A. External whole body doses are calculated using "infinite cloud" assumptions; i.e., the dimensions of the cloud are assumed to be large compared to the distances that the gamma rays and beta particles travel. The dose at any
- 3. The dose models are described in Appendix 15A.

### TABLE 15.7.1-2 (SHEET 4 OF 5)

### Regulatory Guide 1.24 Position

### <u>Design</u>

A. distance from the reactor is calculated based on the maximum ground level concentration at that distance.

For an infinite uniform cloud containing  $\chi$  Ci of beta radioactivity/per m<sup>3</sup>, the beta dose rate in air at the cloud center is:<sup>(b)</sup>

$$_{\beta}D'_{\infty} = 0.457 \overline{E}_{\beta\chi}$$

where:

- $D'_{\infty}$  = beta dose rate from an inifinite cloud (rad/s)
- $\overline{E}_{\beta}$  = average beta energy per disintegration (MeV/dis)
- χ = concentration of beta or gamma emitting isotope in the cloud (Ci/m<sup>3</sup>)

Because of the limited range of beta particles in tissue, the surface body dose rate from beta emitters in the infinite cloud can be approximated as being one-half this amount or:

$$_{\beta}D'_{\infty} = 0.23 \ \overline{E}_{\beta\chi}$$

For gamma-emitting material, the dose rate in air at the cloud center is :

$$_{\gamma}D_{\beta}^{\prime} = 0.507 \ \overline{E}_{\gamma\chi}$$

where:

### TABLE 15.7.1-2 (SHEET 5 OF 5)

Regulatory Guide 1.24
Position

### <u>Design</u>

- $_{\chi}D'_{\beta}$  = Gamma dose rate from an infinite cloud (rad/s)
- Ē<sub>γ</sub> = average gamma energy per disintegration (MeV/dis)

However, because of the presence of the ground, the receptor is assumed to be exposed to only one-half of the cloud (semi-infinite) and the equation becomes:

$$_{\gamma}D'_{\beta} = 0.25 \ \overline{E}_{\gamma\chi}$$

Thus, the total beta or gamma dose to an individual located at the center of the cloud path may be approximated as:

$$_{\beta}D_{\infty}$$
 = 0.23  $\overline{E}_{\beta\psi}$  or

 $\gamma D = 0.25 \ \overline{E}_{\gamma \psi}$ 

where  $\psi$  is the concentration time integral for the cloud (Ci/m<sup>3</sup>)

B. The beta a gamma energies emitted per disintegration, as given in <u>Table of Isotopes</u>, <sup>©</sup> are averaged and used according to the methods described in ICRP Publication 2.

a. These diffusion factors should be used until adequate site meteorological data are obtained. In some cases, available information on such site conditions as meteorology, topography and geographical location may dictate the use of more restrictive parameters to insure a conservative estimate of potential offsite exposures.

b. Meteorology and Atomic Energy, 1968, chapter 7.

c. C. M. Lederer, J. M. Hollander, and I. Perlman, <u>Table of Isotopes</u>, Sixth Edition, John Wiley and Sons, Inc., New York, 1967.

### TABLE 15.7.1-3

# RADIOLOGICAL CONSEQUENCES OF A WASTE GAS DECAY TANK RUPTURE

	Doses (rem)
Exclusion Area Boundary (0 to 2 h)	
Thyroid (rem) Whole body (rem)	0.3 < 0.1
Low Population Zone Outer Boundary (duration)	
Thyroid (rem) Whole body (rem)	0.1 < 0.1

### TABLE 15.7.2-1

# PARAMETERS USED IN EVALUATING THE RADIOLOGICAL CONSEQUENCES OF A LIQUID RADWASTE TANK FAILURE

Source dat	a		
Core pov	wer level (MWt)		3636
Defectiv	re fuel (%)		1
Atmospher	ic dispersion factors		See table 15A-2.
Activity rele	ease data		
Noble ga of tank c	as activity (percent contents)		100
lodine ga of tank o	as activity (percent contents)		1
Tank co	ntents		
N	uclide	<u>Ci</u>	
Kı Kı Kı	r-87 r-88 r-89	5.42E+2 1.64E+3 	
Xe Xe Xe	e-133 e-135 e-138	1.04E+5 3.57E+3 3.12E+2	
-*  -*  -*	131 132 133 135	1.17E+2 1.18E+2 2.23E+2 1.10E+2	

### TABLE 15.7.2-2

### RADIOLOGICAL CONSEQUENCES OF A LIQUID RADWASTE TANK FAILURE

	Doses (rem)
Exclusion area boundary (0 to 2 h)	
Thyroid (rem) Whole body gamma (rem)	0.1 0.3
Low population zone outer boundary (duration)	
Thyroid (rem) Whole body gamma (rem)	0.1 0.1

### TABLE 15.7.3-1

### RECYCLE HOLDUP TANK DATA FOR FAILURE ANALYSIS

Volume of tank (gal)	112,000
Weight of liquid contained (g)	4.22 x 10 <sup>8</sup>
Radioactive contents	
Nuclide	Activity (Ci)
Kr-87	5.49 x 10 <sup>2</sup>
Kr-88	1.56 x 10 <sup>3</sup>
Kr-89	4.64 x 10 <sup>1</sup>
Xe-133	1.14 x 10⁵
Xe-135	3.08 x 10 <sup>3</sup>
Xe-138	2.7 x 10 <sup>2</sup>
I-131	1.18 x 10 <sup>2</sup>
I-132	1.18 x 10 <sup>2</sup>
I-133	1.77 x 10 <sup>2</sup>
I-135	9.7 x 10 <sup>1</sup>
Rb-88	2.03 x 10 <sup>2</sup>
Cs-136	1.22 x 10 <sup>2</sup>
Cs-138	4.05 x 10 <sup>1</sup>

### TABLE 15.7.4-1 (SHEET 1 OF 2)

### PARAMETERS USED IN EVALUATING THE RADIOLOGICAL CONSEQUENCES OF A FUEL HANDLING ACCIDENT

	Containment Open or In Fuel Building	<u>Containment Closed</u> <u>(HISTORICAL)</u>
Source Data	-	
Core power level (MWt)	3636	3636
Radial peaking factor	1.70	1.70
Decay time (h)	90	100
Number of fuel rods affected	314	1.2 assemblies
Fraction of fission product gases contained in the gap region of the fuel assembly	(Reg. Guide 1.195) 5% of the total noble gases and iodines except I-131 (8%) and Kr-85 (10%)	RG 1.25 for all except I-131 a (fraction of 0.12)
Atmospheric Dispersion Factors	Table 15A-2	Table 15A-2
Activity Release Data		
Percent of affected fuel assemblies gap activity released	100	100
Pool decontamination factors		
Elemental iodine	400	200
Organic iodine	1	1
Noble gas	1	1
Filter efficiency (%)	No credit	0
Building mixing volumes assumed (% total volume)	0	25

## TABLE 15.7.4-1 (SHEET 2 OF 2)

	Containment Open or <u>In Fuel Building</u>	<u>Containment Closed</u> <u>(HISTORICAL)</u>
HVAC exhaust rate (ft <sup>3</sup> /min)	N/A	15,000
Building isolation time (s)	No isolation	10+5
Activity release period (h)	2	Release terminated 10 s after containment isolation signal with 5 s allowed for signal generation

#### TABLE 15.7.4-2 (SHEET 1 OF 9)

#### REGULATORY GUIDE 1.25, ASSUMPTIONS USED FOR EVALUATING THE POTENTIAL RADIOLOGICAL CONSEQUENCES OF A FUEL HANDLING ACCIDENT IN THE FUEL HANDLING AND STORAGE FACILITY FOR BOILING AND PRESSURIZED WATER REACTORS, REVISION 0, DATED MARCH 23, 1972

Regulatory Guide 1.25 <u>Position</u>	Case 1 (In Fuel Building or Open Containment)	Case 2 (In Closed Containment <u>Building) (HISTORICAL)</u>
The assumptions <sup>(a)</sup> related to the release of radioactive material from the fuel and fuel storage facility as a result of a fuel handling accident are:		
The accident occurs at a time after shutdown identified in the Technical Requirements Manual as the earliest time fuel handling operations may begin. Radioactive decay of the fission product inventory during the interval between shutdown and commencement of fuel handling operations is taken into consideration.	Conforms. Accident occurs 90 h after shutdown.	Conforms. Accident occurs 100 h after shutdown.
The maximum fuel rod pressurization <sup>(b)</sup> is 1200 psig.	Conforms.	Conforms.
The minimum water depth <sup>(b)</sup> between the top of the damaged fuel rods and the fuel pool surface is 23 ft.	Conforms. Water depth is greater than 23 ft.	Conforms. Water depth is greater than 23 ft.
All of the gap activity in the damaged rods is released and consists of 10% of the total noble gases other than Kr-85, 30% of the Kr-85, and 10% of the total radioactive iodine in the rods at the time of the accident. For the purpose of sizing filters for the fuel handling accident addressed in this guide, 30% of the I-127 and I-129 inventory is assumed to be released from the damaged rods.	Conforms, except for I-131 8%, Kr-85 10%, and 5% noble gases and other iodines; the gap is consistent with Reg. Guide 1.195 for lead rod average burnup to 62,000 MWd/Mtu.	Conforms, except for I-131 which assumes 12%; the gap is consistent with NUREG CR-5009 for lead rod average burnup to 60,000 MWd/Mtu.
The values assumed for individual fission product inventories are calculated assuming full-power operation at the end of core life immediately preceding shutdown, and such calculation should include an appropriate radial peaking factor. The minimum acceptable radial peaking factors are 1.5 for BWRs and	A peaking factor of 1.70 is used since this is the maximum projected radial peaking factor. A value of 1.65 to 1.70 may be used for a cycle-specific core reload evaluation.	A peaking factor of 1.70 is used since this is the maximum projected radial peaking factor. A value of 1.65 to 1.70 may be used for a cycle-specific core reload evaluation.

1.65 for PWRs.

#### TABLE 15.7.4-2 (SHEET 2 OF 9)

Regulatory Guide 1.25 Position	Case 1 (In Fuel Building or Open Containment)	Case 2 (In Closed Containment <u>Building) (HISTORICAL)</u>
The iodine gap inventory is composed of 99.75% inorganic species and 0.25% organic species.	Conforms.	Conforms.
The pool decontamination factors for the inorganic and organic species are 133 and 1, respectively, giving an overall effective decontamination factor of 100 (i.e., 99% of the total iodine released from the damaged rods is retained by the pool water). This difference in decontamination factors for inorganic and organic iodine above the fuel pool being composed of 75% inorganic and 25% organic species.	The pool decontamination factors for the inorganic and organic species are 400 and 1, respectively, giving an overall effective decontamination factor of 200 (i.e., 99.5% of the total iodine released from the damaged rods is retained by the pool). This difference in decontamination factors for inorganic and organic iodine above the fuel pool being composed of 50% inorganic and 50% organic species.	The pool decontamination factors for the inorganic and organic species are 200 and 1, respectively, giving an overall effective decontamination factor of 133 (i.e., 99.25% of the total iodine released from the damaged rods is retained by the pool). This difference in decontamination factors for inorganic and organic iodine above the fuel pool being composed of 67% inorganic and 33% organic species.
The retention of noble gases in the pool is negligible (i.e., decontamination factor of 1).	Conforms. A decontamination factor of 1 is used.	Conforms. A decontamination factor of 1 is used.
The radioactive material that escapes from the pool to the building is released from the building <sup>(c)</sup> over a 2-h time period.	Conforms. A 0- to 2-h release from the pool to the building to the environment is assumed.	The release from pool to the building is automatically isolated upon detection of the first trace of release. Thus, the release is contained in the containment building after isolation.
If it can be shown that the building atmosphere is exhausted through adsorbers designed to remove iodine, the removal efficiency is 90% for inorganic species and 70% for organic species.	No credit is taken for the FHB post-accident exhaust filters that conform to Regulatory Guide 1.52 as described in table 9.4.1-2.	No credit is taken for the normal purge filters.
The effluent from the filter system passes directly to the emergency exhaust system without mixing (e) in the surrounding building atmosphere and is then released (as an elevated plume for those facilities with stacks (f)).	Conforms.	Conforms.
The assumptions for atmospheric diffusion for:	Short-term atmospheric dispersion factors corresponding to ground level release and accident conditions were	
Ground level releases	based on the meteorological measurements program described in section 2.3. The dispersion factors are in	
The basic equation for atmospheric diffusion from a ground level point source is:	compliance with the methodology described in Regulatory Guide 1.145 and represent the worst of the 5% overall site meteorology and the 0.5% worst sector meteorology.	
$\chi_{Q} = \frac{1}{\pi \mu \sigma_{y} \sigma_{z}}$		

where:

#### TABLE 15.7.4-2 (SHEET 3 OF 9)

Regulatory Guide 1.25 Position Case 1 (In Fuel Building or Open Containment) Case 2 (In Closed Containment Building) (HISTORICAL)

- χ = the short-term average centerline value of the ground level concentration (Ci/m).
- Q = amount of material released (Ci/s).
- $\mu$  = windspeed (m/s).
- $\sigma_y = \text{the horizontal standard deviation of} \\ \text{the plume (m). See figure V-1, page} \\ \text{48, in F. A. Gifford, Jr., Use of} \\ \text{Routine Meteorological Observation} \\ \text{Estimating Atmospheric Dispersion,} \\ \underline{\text{Nuclear Safety}}, \text{Vol. II, No. 4, June} \\ 1961. \\ \end{array}$
- $\sigma_z = the vertical standard deviation of the plume (m). See figure V-2, page 48 in F. A. Gifford, Jr., Use of Routine Meteorological Observation for Estimating Atmospheric Dispersion, <u>Nuclear Safety</u>, Vol. II, No. 4, June 1961.$

For ground level releases, atmospheric diffusion factors<sup>(g)</sup> used in evaluating the radiological consequences of the accident addressed in this guide are based on the following assumptions: windspeed of 1 m/s, uniform wind direction, and Pasquill diffusion category F.

Figure 1 is a plot of atmospheric diffusion factor

 $(\chi/Q)$  versus distance derived by use of the equation for a ground level release given in regulatory position 2.a.(1) and under the meteorological conditions given in regulatory position 2.a.(2).

Atmospheric diffusion factors for ground level releases may be reduced by a factor ranging from 1 to a maximum of 3 (see figure 2) for additional dispersion produced by the turbulent wake of the reactor building. The volumetric building wake correction as defined in subsection 3-3.5.2 of

#### TABLE 15.7.4-2 (SHEET 4 OF 9)

Regulatory Guide 1.25 Position Case 1 (In Fuel Building or Open Containment) Case 2 (In Closed Containment Building) (HISTORICAL)

<u>Meteorology and Atomic Energy-1968</u> is used with a shape factor of ½ and the minimum crosssectional area of the reactor building only.

Elevated releases

The basic equation for atmospheric diffusion from an elevated release is:

Not applicable; ground level releases were assumed.

Not applicable; ground level releases were assumed.

$$\chi_{Q} = \frac{e^{-h^{2/z}}\sigma_{z}}{\pi\mu\sigma_{y}\sigma_{z}}$$

where:

- $\chi$  = the short-term average centerline value of the ground level concentration (Ci/m<sup>3</sup>).
- Q = amount of material released (Ci/s).
- $\mu$  = windspeed (m/s).
- σ<sub>y</sub> = the horizontal standard deviation of the plume (m). See figure V-1, page 48 in, F. A. Gifford, Jr., Use of Routine Meteorological Observations for Estimating Atmospheric Dispersion, <u>Nuclear Safety</u>, Vol. II. No. 4, June 1961.
- $\sigma_z = the vertical standard deviation of the plume (m). See figure V-2, page 48 in, F. A. Gifford, Jr., Use of Routine Meteorological Observations for Estimating Atmospheric Dispersion, <u>Nuclear Safety</u>, Vol. II, No. 4, June 1961.$
- h = effective height of release (m).

For elevated releases, atmospheric diffusion factors<sup>(h)</sup> used in evaluating the radiological consequences of the accident addressed in this

#### TABLE 15.7.4-2 (SHEET 5 OF 9)

Regulatory Guide 1.25	
Position	

Case 1 (In Fuel Building or Open Containment) Case 2 (In Closed Containment Building) (HISTORICAL)

guide are based on the following assumptions: windspeed of 1 m/s, uniform wind direction, envelope of Pasquill diffusion categories for various release heights, and a fumigation condition existing at the time of the accident.<sup>(h)</sup>

Figure 3 is a plot of atmospheric diffusion factors versus distance for an elevated release assuming no fumigation, and figure 4 is for an elevated release with fumigation.

Elevated releases are considered to be at a height equal to no more than the actual stack height. Certain site conditions may exist, such as surrounding elevated topography or nearby structures, which will have the effect of reducing the effective stack height. The degree of stack height reduction will be evaluated on an individual case basis.

The following assumptions and equations may be used to obtain conservative approximations of thyroid dose from the inhalation of radioiodine and external whole-body dose from radioactive clouds:

The assumptions relative to inhalation thyroid dose approximations are:

The receptor is located at a point on or beyond the site boundary where the maximum ground level concentration is expected to occur.

No correction is made for depletion of the effluent plume of radioiodine due to deposition on the ground or for the radiological decay of radioiodine in transit.

Inhalation thyroid doses may be approximated by use of the following equation:

 $\mathsf{D} = \frac{\mathsf{F}_{\mathsf{g}}\mathsf{IFPBR}(\mathsf{f})}{\mathsf{F}_{\mathsf{g}}}$ (DF<sub>P</sub>)(D

where:

Conforms. See appendix15A, subsection 15A.2.4.

Conforms. See appendix15A, subsection 15A.2.4.

#### TABLE 15.7.4-2 (SHEET 6 OF 9)

Regulatory Guide 1.25 Position Case 1 (In Fuel Building or Open Containment) Case 2 (In Closed Containment Building) (HISTORICAL)

- D = thyroid dose (rd).
- F<sub>g</sub> = fraction of fuel rod iodine inventory in fuel rod void space (0.1).
- I = core iodine inventory at time of accident (Ci).
- F = fraction of core damaged so as to release void space iodine.
- P = fuel peaking factor.
- B = breathing rate = 3.47 x 10<sup>-4</sup> m<sup>3</sup>/s (i.e., 10 m<sup>3</sup>/8-h workday as recommended by the ICRP).
- DF<sub>p</sub> = effective iodine decontamination factor for pool water.
- DF<sub>f</sub> = effective iodine decontamination factor for filters (if present).
- $\chi/Q$  = atmospheric diffusion factor at receptor location (s/m<sup>3</sup>).
- R = adult thyroid dose conversion factor for the iodine isotope of interest (rd/Ci). Dose conversion factors for I-131 through I-135 are listed in the table below.(i) These values were derived from "standard man" parameters recommended in ICRP Publication 2.<sup>(0)</sup>

#### TABLE 15.7.4-2 (SHEET 7 OF 9)

Regulatory Guide 1.25 Position		Case (In Fuel Bu	e 1 iilding or Open Contair	iment)	Case 2 (In Closed Containment <u>Building) (HISTORICAL)</u>	
Adul <u>Dose</u>	It Inha	alation Thyroid version Factors	See table	15A-5 for dose convers	sion factors.	See table 15A-5 for dose conversion factors.
lodine Isotope		Conversion Factor (R) (rd/Ci inhaled)				
131 132 133 134 135		$1.48 \times 10^{\circ}$ $5.35 \times 10^{4}$ $4.0 \times 10^{5}$ $2.5 \times 10^{4}$ $1.24 \times 10^{5}$				
The assumption of the assumpti	ptions ximat	relative to external whole-body ions are:	Conforms.	See appendix 15A, su	ubsection 15A.2.6	Conforms. See appendix 15A, subsection 15A.2.6
The receptor is located at a point on or beyond the site boundary where the maximum ground level concentration is expected to occur.		s located at a point on or beyond ary where the maximum ground ation is expected to occur.				
External whole-body doses are calculated using "infinite cloud" assumptions; i.e., the dimensions of the cloud are assumed to be large compared to the distances that the gamma rays and beta particles travel. The dose at any distance from the reactor is calculated based on the maximum ground level concentration at that distance.		See table	15A-5 for dose convers	sion factors.	See table 15A-5 for dose conversion factors.	
For an infinite uniform cloud containing $\chi$ Ci of beta radioactivity per m <sup>3</sup> , the beta dose rate in air at the cloud center is:						
$\beta_{\infty}^{D'} = 0.4$	57 Ē	βχ				
where:						
$\beta_\infty^{D'}$	=	beta dose rate from an infinite cloud (rd/s).				
$\overline{\mathrm{E}}_{\beta}$	=	average beta energy per disintegration (MeV/dis).				
χ	=	concentration of beta or gamma emitting isotope in the cloud (Ci/m <sup>3</sup> ).				

#### TABLE 15.7.4-2 (SHEET 8 OF 9)

Regulatory Guide 1.25 Position Case 1 (In Fuel Building or Open Containment) Case 2 (In Closed Containment Building) (HISTORICAL)

Because of the limited range of beta particles in tissue, the surface-body dose rate from beta emitters in the infinite cloud can be approximated as being one-half this amount or:

$$\beta_{\infty}^{D'} = 0.23 \overline{E}_{\beta} \chi$$

For gamma-emitting material the dose rate in tissue at the cloud center is:

$$\gamma^{D'}_{\infty} \ = \ 0.507 \, \overline{E}_{\gamma} \, \chi$$

where:

 $\gamma_{\infty}^{D'}$  = gamma dose rate from an infinite cloud (rd/s).

E<sub>γχ</sub> = average gamma energy per disintegration (MeV/disintegration).

However, because of the presence of the ground, the receptor is assumed to be exposed to only one-half of the cloud (semi-infinite) and the equation becomes:

$$\gamma^{D'} = 0.25 \overline{E}_{\gamma} \chi$$

Thus, the total beta or gamma dose to an individual located at the center of the cloud path may be approximated as:

$$\beta^{D}_{\infty} = 0.23 \ \overline{E}_{\beta} \psi \text{ or }$$

$$\gamma^D = 0.25 \overline{E}_{\gamma} \psi$$

where  $\psi$  = the concentration time integral for the cloud (Ci s/m<sup>3</sup>).

The beta and gamma energies emitted per disintegration, as given in <u>Table of Isotopes</u>,<sup>(i)</sup> are averaged and used according to the methods described in ICRP Publication 2.

#### TABLE 15.7.4-2 (SHEET 9 OF 9)

a. The assumptions given are valid only for oxide fuels of the types currently in use and in cases where the following conditions are not exceeded:

- 1. Peak linear power density of 20.5 kW/ft for the highest power assembly discharged.
- 2. Maximum centerline operating fuel temperature less than 4500°F for this assembly.
- 3. Average burnup for the peak assembly of 25,000 MWd/t or less (this corresponds to a peak local burnup of about 45,000 MWd/t).

b. For release pressures greater than 1200 psig and water depths less than 23 ft, the iodine decontamination factors will be less than those assumed in this guide and must be calculated on an individual-case basis using assumptions comparable in conservatism to those of this guide.

c. The effectiveness of features provided to reduce the amount of radioactive material available for release to the environment will be evaluated on an individual-case basis.

d. These efficiencies are based upon a 2-in. charcoal bed depth with 1/4-s residence time. Efficiencies may be different for other systems and must be calculated on an individualcase basis.

e. Credit for mixing will be allowed in some cases; the amount of credit will be evaluated on an individual-case basis.

f. Credit for an elevated release will be given only if the point of release is more than 2 1/2 times the height of any structure close enough to affect the dispersion of the plume or located far enough from any structure which could affect the dispersion of the plume. For those plants without stacks the atmospheric diffusions factors assuming ground level release given in regulatory position 2.b should be used.

g. These diffusion factors should be used until adequate site meteorological data are obtained. In some cases, available information on such site conditions as meteorology, topography, and geographical location may dictate the use of more restrictive parameters to ensure a conservative estimate of potential offsite exposures.

h. For sites located more than 2 miles from large bodies of water such as oceans or one of the Great Lakes, a fumigation condition is assumed to exist at the time of the accident and continue for 1/2 h. For sites located less than 2 miles from large bodies of water a fumigation condition is assumed to exist at the time of the accident and continue for the duration of the release (2 h).

i. Dose conversion factors taken from F. D. Anderson, R. E. Baker, J. J. DiNunno, "Calculation of Distance Factors for Power and Test Reactor Site," TID-14844, 1962.

j. Recommendations of the International Commission on Radiological Protection, "Report of Committee II on Permissible Dose for Internal Radiation (1959)," ICRP Publication 2, Permagon Press, New York, 1960.

k. Meteorology and Atomic Energy-1968, chapter 7.

I. C. M. Lederer, J. M. Hollander, and I. Perlman, Table of Isotopes, sixth edition, John Wiley and Sons, Inc., New York, 1967.

TABLE 15.7.4-3

DELETED

I

### TABLE 15.7.4-4

### RADIOLOGICAL CONSEQUENCES OF A FUEL HANDLING ACCIDENT

Fuel Building or Open Containment	<u>Doses (rem)</u>
Exclusion area boundary (0 to 2 h)	
Thyroid Whole body	23.4 1.1
Low population zone outer boundary (0 to 2 h)	
Thyroid Whole body	9.4 0.4
Control Room <sup>(a)</sup>	
Thyroid Whole body Beta Skin	13.7 0.6 5.6
Containment Closed (HISTORICAL)	
Exclusion area boundary (0 to 2 h)	
Thyroid Whole body	0.3 < 0.1
Low population zone outer boundary (0 to 2 h)	
Thyroid Whole body	0.1 < 0.1

a. Doses from MURPU and Control Room Habitability (TSTF-448) implementation.

### 15.8 ANTICIPATED TRANSIENTS WITHOUT TRIP

The worst common mode failure which is postulated to occur is the failure to scram the reactor after an anticipated transient has occurred. A series of generic studies <sup>(1,2)</sup> on anticipated transients without scram (ATWS) showed acceptable consequences would result provided that the turbine trips and auxiliary feedwater flow is initiated in a timely manner. The effects of ATWS events are not considered as part of the design basis for transients analyzed in Chapter 15. The final NRC ATWS rule<sup>(3)</sup> requires that Westinghouse-designed plants install ATWS mitigation system circuitry (AMSAC) to initiate a turbine trip and actuate auxiliary feedwater flow independent of the reactor protection system. The Vogtle AMSAC design is described in section 7.7.

In support of the Measurement Uncertainty Recapture Power Uprate, a plant-specific evaluation was performed to demonstrate continued conformance with the analyses that formed the basis for the ATWS rule (Reference 4 and 5), at the uprated conditions.

### 15.8.1 REFERENCES

- 1. "Westinghouse Anticipated Transients Without Trip Analysis," <u>WCAP-8330</u>, August 1974.
- 2. Anderson, T. M., "ATWS Submittal," Westinghouse Letter NS-TMA-2182 to S. H. Hanauer of the NRC, December 1979.
- 3. ATWS Final Rule Code of Federal Regulations 10 CFR 50.62 and Supplementary Information Package, "Reduction of Risk from Anticipated Transients Without Scram (ATWS) Events for Light-Water-Cooled Nuclear Power Plants."
- 4. NL-07-1010, "Vogtle Electric Generating Plant Request to Change Licensed Maximum Power Level," L. M. Stinson to USNRC, August 28, 2007.
- 5. NL-08-0307, "Vogtle Electric Generating Plant, Units 1 and 2, Issuance of Amendments Regarding Measurement Uncertainty Recapture Power Uprate (TAC Nos. MD6625 and MD6626)," USNRC to T. E. Tynan, February 27, 2008.

### **APPENDIX 15A**

### ACCIDENT ANALYSIS RADIOLOGICAL CONSEQUENCES EVALUATION MODELS AND PARAMETERS

### 15A.1 GENERAL ACCIDENT PARAMETERS

This appendix contains the parameters used in analyzing the radiological consequences of postulated accidents. Table 15A-1 contains the general parameters used in all the accident analyses. For parameters specific only to particular accidents, refer to that accident parameter section. The site specific ground-level release short-term dispersion factors<sup>(a)</sup> are based on Regulatory Guide 1.145<sup>(1)</sup> methodology and represent the 0.5-percent worst sector meteorology; these factors are given in table 15A-2. (See subsection 2.3.4 for additional details on meteorology.) The core and gap inventories are given in table 15A-3. The thyroid (via inhalation pathway), beta skin, and gamma body (via submersion pathway) dose factors based on reference 2 are given in table 15A-5.

Reactor coolant iodine concentrations for the technical specification limit of 1  $\mu$ Ci/g of dose equivalent I-131 and for the assumed preaccident iodine spike concentration of 60  $\mu$ Ci/g of dose equivalent I-131 are presented in table 15A-6. Iodine appearance rates in the reactor coolant for normal steady-state operation at 1  $\mu$ Ci/g of dose equivalent I-131 and for an assumed accident-initiated iodine spike are presented in table 15A-7. Reactor coolant noble gas concentrations based on 1-percent fuel defects are presented in table 15A-4.

### 15A.2 OFFSITE RADIOLOGICAL CONSEQUENCES CALCULATIONAL MODELS

This section presents the models and equations used for calculating the integrated activity released to the environment, the accident flowpaths, and the equations for dose calculations. Two major release models are considered:

- A single holdup system with no internal cleanup.
- A holdup system wherein a two-region spray model is used for internal cleanup.

### 15A.2.1 ACCIDENT RELEASE PATHWAYS

The release pathways for the major accidents are given in figure 15A-1. The accidents and their pathways are as follows:

A. Loss-of-Coolant Accident (LOCA)

Immediately following a postulated LOCA, the release of radioactivity from the containment is to the environment with the containment spray and engineered safety features (ESF) systems in full operation. The release in this case is calculated using equations 6a and 6b, which take into account a two-region spray model within the containment. The release of radioactivity to the environment

<sup>&</sup>lt;sup>a</sup> For accidents, ground-level releases are assumed.

due to assumed ESF system leakages in the auxiliary building will be via ESF filters and is calculated using equation 5.

B. Waste Gas Decay Tank Rupture (WGDTR)

The activity release to the environment due to WGDTR will be direct and unfiltered, with no holdup. The release pathway is A'-D. The total activity release in this case is therefore assumed to be the initial source activity itself.

C. Fuel Handling Accident

The release to the environment due to a fuel handling accident in the fuel building is via filters. The release pathway is B-C-D. Since the release is calculated without any credit for holdup in the fuel building, the total release will be the product of the initial activity and the filter nonremoval efficiency fraction. (For noble gases, the nonremoval efficiency fraction is 1.) The release of radioactivity to the environment due to a fuel handling accident inside the containment is direct and unfiltered via the A-D pathway and occurs only until the containment is isolated. (Actually, the release is via the nonsafety-grade filters.) The release is calculated using equation 5. No mixing in the containment volume is assumed.

D. Control Assembly Ejection

Radioactivity release to the environment due to the control assembly ejection accident is direct and unfiltered. The releases from the primary system are calculated using equation 5, which considers holdup in the single-region primary system. (The spray removal is not assumed.) The secondary (steam) releases via the relief valves are calculated without any holdup. The pathways for these releases are A-D and A'-D.

E. Main Steam Line Break (MSLB) or Steam Generator Tube Rupture (SGTR)

Radioactivity releases to the environment due to MSLB or SGTR accidents are direct and unfiltered with no holdup via the A'-D pathway. The activity release calculations for these accidents are complex, involving spiking effects, time-dependent flashing fractions, and scrubbing of flashed activities; the release calculations are described in the sections that address these accidents.

### 15A.2.2 SINGLE-REGION RELEASE MODEL

It is assumed that any activity released to the holdup system instantaneously diffuses to uniformly occupy the system volume.

The following equations are used to calculate the integrated activity released from postulated accidents.

$$A_1(t) = A(0)e^{-\lambda}1^t$$
 (1)

where:

 $A_1(t)$  = source activity at time t, (Ci).

- $A_1(0)$  = initial source activity at time  $t_o$ , (Ci).
  - $\lambda_1$  = total removal constant from primary holdup system (s<sup>-1</sup>).

$$\lambda_1 = \lambda_d + \lambda_{1\ell} + \lambda_r \tag{2}$$

 $\lambda_d$  = decay removal constant (s<sup>-1</sup>).

 $\lambda_{1\ell}$  = primary holdup leak or release rate (s<sup>-1</sup>).

$$\lambda_r$$
 = internal removal constant (i.e., sprays, plateout, etc. (s<sup>-1</sup>).

thus, the direct release rate to the atmosphere from the primary holdup system

$$\mathsf{R}_{\mathsf{u}}(\mathsf{t}) = \lambda_{1\ell}[\mathsf{A}_1(\mathsf{t})] \tag{3}$$

where:

 $R_u(t)$  = unfiltered release rate (Ci/s).

The integrated activity release is the integral of the above equation.

IAR (t) = 
$$\int_{0}^{t} R_{u}(t) dt = \int_{0}^{t} \lambda_{1\ell} A_{1}(0) e^{-\lambda} I^{t} dt$$
 (4)

This yields:

IAR (t) = 
$$(\lambda_{1\ell} A_1(0)/\lambda_1) (1 - e^{-\lambda} I^t)$$
 (5)

### 15A.2.3 TWO-REGION SPRAY MODEL IN CONTAINMENT - LOCA

A two-region spray model is used to calculate the integrated activity released to the environment. The model consists of sprayed and unsprayed regions in containment and a constant mixing rate between them.

As it is assumed that there are no sources after initial release of the fission products, the remaining processes are removal and transfer so that the multivolume containment is described by a system of coupled first-order differential equations.

For a two-region model, the equations are:

$$\frac{dA_1}{dt} = -\sum_{j=1}^{K_1} \lambda_{1j}A_1 - Q_{12} \frac{A_1}{V_1} + Q_{21}\frac{A_2}{V_2}$$
(6a)

$$\frac{dA_2}{dt} = -\sum_{j=1}^{K_2} \lambda_{2j}A_2 - Q_{21}\frac{A_2}{V_2} + Q_{12}\frac{A_1}{V_1}$$
(6b)

where:

 $A_i$  = fission product activity in volume i (Ci).

 $Q_{i\ell}$  = transfer rate from volume i to volume  $\ell$  (cm<sup>3</sup>/s).

 $V_i$  = volume of the ith compartment (cm<sup>3</sup>).

 $\lambda_{ij}$  = removal rate of the jth removal process internal to volume i (s<sup>-1</sup>).

K<sub>i</sub> = total number of removal processes in volume i.

To calculate the integrated activity released to the atmosphere, the release rate of activity is first calculated. This is found from:

R (t) = 
$$\sum_{i=1}^{2} \lambda_{1\ell} A_i(t)$$
 (7)

The integrated activity released from time  $t_o - t_1$  is then

IAR = 
$$\int_{t_0}^{t_1} R(t) dt$$

### 15A.2.4 OFFSITE THYROID DOSE CALCULATION MODEL

Offsite thyroid doses are calculated using the equation:

$$D_{TH} = \sum_{i} DCF_{THi} \sum_{j} (IAR)_{ij} (BR)_{j} (\chi Q)_{j}$$
(8)

where:

(IAR) <sub>ij</sub>	=	integrated activity of isotope i released <sup>(b)</sup> during the time interval j (Ci).	
(BR) <sub>j</sub>	=	breathing rate during time interval j (m <sup>3</sup> /s).	
(χ/Q) <sub>j</sub>	=	offsite atmospheric dispersion factor during time interval j (s/m <sup>3</sup> ).	
$DCF_{THi}$	=	thyroid dose conversion factor via inhalation for isotope i (rem/Ci).	
D <sub>TH</sub>	=	thyroid dose via inhalation (rems).	

### 15A.2.5 OFFSITE BETA SKIN DOSE CALCULATIONAL MODEL

Assuming a semi-infinite cloud of gamma emitters, offsite beta skin doses are calculated using the equation:

$$D_{\beta S} = \sum_{i} DCF_{\beta i} \sum_{j} (IAR)_{ij} (\chi/Q)_{j}$$

where:

$D_{\betaS}$	=	beta skin dose (rem).
$DCF_{\beta i}$	=	beta skin dose conversion factor for the I isotope (rem-m <sup>3</sup> /Ci-s).
(IAR) <sub>ij</sub>	=	integrated activity of isotope released during the time interval j (Ci).
(χ/Q) <sub>j</sub>	=	offsite atmospheric dispersion factor during time interval j (s/m <sup>3</sup> ).

### 15A.2.6 OFFSITE GAMMA BODY DOSE CALCULATIONAL MODEL

Assuming a semi-infinite cloud of gamma emitters, offsite gamma body doses are calculated using the equation:

<sup>&</sup>lt;sup>b</sup> No credit is taken for cloud depletion by ground deposition and radioactive decay during transport to the exclusion area boundary or the outer boundary of the low population zone.

$$D_{\gamma\beta} = \sum_{i} DCF_{\lambda i} \sum_{j} (IAR)_{ij} (\chi/Q)_{j}$$

$D_{\gamma\beta}$	=	gamma body dose (rem).
$DCF_{\gamma i}$	=	gamma body dose conversion factor for the ith isotope (rem-m <sup>3</sup> /Ci-s).
(IAR) <sub>ij</sub>	=	integrated activity of isotope is released during the time interval j (Ci).
(χ/Q) <sub>j</sub>	=	offsite atmospheric dispersion factor during time interval j (Ci).

### 15A.3 CONTROL ROOM RADIOLOGICAL CONSEQUENCES CALCULATIONAL MODELS

Radiation doses to a control room operator as a result of a postulated LOCA are presented in this chapter. (A study of the radiological consequences in the control room due to various postulated accidents indicate that the LOCA is the limiting case.)

### 15A.3.1 INTEGRATED ACTIVITY IN CONTROL ROOM

The integrated activity in the control room during each time interval is found by multiplying the release by the appropriate  $\chi/Q$  to give the concentration at the control room intake. This activity is brought into the control room through the filtered intake valves and by unfiltered inleakage and is subjected to the control room ventilation system of recirculation through charcoal filters and exhaust to the atmosphere.



From this the total integrated activity in the control room during any time interval can be calculated.

The activity in the control room can be calculated by the same method used to calculate activity in the containment.

### 15A.3.2 INTEGRATED ACTIVITY CONCENTRATION IN CONTROL ROOM FROM SINGLE-REGION SYSTEM

To calculate the integrated activity concentration in the control room, the activity in the control room at any time, t, is calculated and then integrated again to find the integrated activity.

$$\frac{dA_{CR}(t)}{dt} = \left[F_2 R_{FIN} + R_{UIN}\right] \frac{\chi}{Q} R(t) - \lambda_3 A_{CR}(t)$$

where:

A <sub>CR</sub> (t)	=	activity in the control room at any time t (Ci).
$F_2$	=	filter nonremoval fraction on intake.
$R_{FIN}$	=	filtered intake rate (m <sup>3</sup> /s).
$R_{UIN}$	=	unfiltered intake rate (m <sup>3</sup> /s).
R(t)	=	activity of release rate in Ci/s given in equation 3 of subsection 15A.2.2.
$\lambda_3$	=	$\lambda_{3\ell} + \lambda_d + \lambda_{r.}$

$\lambda_3$	=	total removal rate from control room (s <sup>-1</sup> ).
$\lambda_{3\ell}$	=	exhaust rate from control room (s <sup>-1</sup> ).
$\lambda_{d}$	=	isotopic decay constant (s <sup>-1</sup> ).
$\lambda_r$	=	recirculation removal rate (s <sup>-1</sup> ).

The integrated activity concentration in the control room  $(IA_{CR})$  is determined by the expression

$$IA_{CR}(t) = \frac{1}{V_{CR}} \int_{0}^{t} A_{CR}(t) dt$$

where:

 $V_{CR}$  = control room volume.

This  $IA_{CR}(t)$  is used to calculate the doses to the operator in the control room. This activity is multiplied by an occupancy factor which accounts for the time fraction the operator is in the control room.

### 15A.3.3 CONTROL ROOM THYROID DOSE CALCULATIONAL MODEL

Control room thyroid doses via inhalation pathway are calculated using the following equation:

$$\mathsf{D}_{\mathsf{TH-CR}} \quad = \quad \mathsf{BR} \; \sum_{i} \; \; \mathsf{DCF}_{\mathsf{THi}} \; \sum_{j} \; \; (\mathsf{IA}_{\mathsf{CRij}}) \; \times \mathsf{o}_{j}$$

where:

$D_{TH-CR}$	=	control room thyroid dose (rem).
BR	=	breathing rate assumed to be always $3.47 \times 10^{-4} \text{ m}^3/\text{s}$ .
$DCF_{THi}$	=	thyroid dose conversion factor for adult via inhalation for isotope i (rem/Ci).
IA <sub>CRij</sub>	=	integrated activity concentration in control room, $\underline{\text{Ci-s}}$ for isotope i during time interval $\text{jm}^3$
Oi	=	control room occupancy fraction during time interval j.

### 15A.3.4 CONTROL ROOM BETA SKIN DOSE CALCULATIONAL MODEL

The beta skin doses to a control room operator are calculated using the following equation:

$D_{\beta\text{-}CR}$	=	beta skin dose in the control room (rem).
$DCF_{\beta i}$	=	beta skin dose conversion factor for isotope i (rem-m <sub>3</sub> /Ci-s).
IA <sub>CRij</sub>	=	integrated activity concentration in the control room, $\underline{\text{Ci-s}}$ for the isotope i during time interval $\text{jm}^3$

o<sub>j</sub> = control room occupancy fraction during time interval j.

### 15A.3.5 CONTROL ROOM GAMMA BODY DOSE CALCULATION

Due to the finite size of the control room, the gamma body doses to a control room operator will be substantially less than what they would be due to immersion in an infinite cloud of gamma emitters. The finite cloud gamma doses are calculated using Murphy's method which models the control room as a hemisphere.<sup>(3)</sup> The following equation is used:

$$D_{\gamma\beta\text{-CR}} = \frac{1}{GF} \sum_{i} DCF_{\lambda i} \sum_{j} (IA_{CRij}) \times o_{j}$$

where:

$D_{\gamma\beta}$ -CR	=	gamma body dose in the control room (rem).
GF	=	dose reduction due to control room geometry factor.
GF	=	1173/ $V_1^{0.338}$ (dimensionless).
V <sub>1</sub>	=	volume of the control room (ft <sup>3</sup> ).
DCF <sub>γi</sub>	=	gamma body dose conversion factor for isotope i (rem-m <sup>3</sup> /Ci-s).
IA <sub>CRij</sub>	=	integrated activity concentration in control room, $\underline{\text{Ci-s}}$ for isotope i during time interval $\text{jm}^3$
0 <sub>i</sub>	=	control room occupancy fraction during time interval j.

### 15A.3.5.1 <u>Model for Radiological Consequences Due to Radioactive Cloud</u> External to the Control Room

This dose is calculated based on the semi-infinite cloud model which is modified using the protection factors described in subsection 7.5.4 of reference 4 to account for the control room walls.

### 15A.3.6 REFERENCES

- "Atmospheric Dispersion Models for Potential Accident Consequence Assessments at Nuclear Power Plants," <u>United States Nuclear Regulatory Commission (USNRC)</u> <u>Regulatory Guide 1.145</u>, August 1979.
- 2. "Calculation of Annual Doses to Man from Routine Releases of Reactor Effluents for the Purpose of Evaluating Compliance with 10 CFR Part 50 Appendix I," <u>USNRC Regulatory</u> <u>Guide 1.109</u>, Rev 1, October 1977.

- 3. Murphy, K. G., and Campe, K. M., "Nuclear Power Plant Control Room Ventilation System Design for Meeting General Criterion 19," Paper presented at the 13th AEC Air Cleaning Conference.
- 4. "Meteorology and Atomic Energy 1968," D. H. Slade (ed.), USAEC Report, <u>TID-24190</u>, 1968.

## TABLE 15A-1 (SHEET 1 OF 2)

### PARAMETERS USED IN ACCIDENT ANALYSIS

### General

	Core power level (MWt) Number of fuel assemblies in the core Maximum radial peaking factor Steam generator tube leak (gal/min)	3636 193 1.70 1.0
Sourc	es	
	Core inventories (Ci) Gap inventories (Ci) Primary coolant specific activities for 1% fuel defects (µCi/g)	Table 15A-3 Table 15A-3 Table 15A-4
	Primary coolant activity, Technical Specification limit for iodines - I-131 dose equivalent ( $\mu$ Ci/g)	1.0 See table 15A-6.
	Secondary coolant activity Technical Specification limit for iodines - I-131 dose equivalent ( $\mu$ Ci/g)	0.1
Activit	y Release Parameters	
	Free volume of containment (ft <sup>3</sup> )	2.95 x 10 <sup>6</sup>
	0 to 24 h (percent per day) After 24 h (percent per day)	0.2 0.1
	Free volume (ft <sup>3</sup> ) Normal ventilation rate, unfiltered (ft <sup>3</sup> /min)	1.72 x 10⁵ 3000
	Time to isolate normal ventilation (s)	11.3
	Time to establish emergency ventilation one unit operating (s)	99.3
	Time to establish emergency ventilation, three units operating (s)	108
	Emergency ventilation intake rate - one unit operating (ft <sup>3</sup> /min)	1500
	Emergency ventilation intake rate - three units operating (ft <sup>3</sup> /min)	3870

I

### TABLE 15A-1 (SHEET 2 OF 2)

Emergency ventilation rate, - one unit operating (ft <sup>3</sup> /min)	17,100 <sup>(a)</sup>
Emergency ventilation rate, - three units operating (ft <sup>3</sup> /min)	47,500 <sup>(a)</sup>
Unfiltered infiltration rate (ft <sup>3</sup> /min) unpressurized control room pressurized control room	825 + 10 <sup>(b)</sup> 120 + 10 <sup>(c)</sup>
lodine removal efficiency for recirculation filters (all forms of iodine) (percent)	99
lodine removal efficiency for intake filters (all forms of iodine)	99
High-efficiency particulate air filter efficiency (percent)	99
Miscellaneous	
Atmospheric dispersion factors $(\chi/Q)(s/m^3)$	Table 15A-2
Gamma body and beta skin (rem-m <sup>3</sup> /Ci-s) Thyroid (rem/Ci)	Table 15A-5 Table 15A-5

a. The value is for combined intake and recirculation air flow. The value also reflects the Technical Specification acceptance criterion of  $\pm$  10% of the nominal flow for a single train.

b. 825 cfm unfiltered inleakage for inleakage testing. 10 cfm is for ingress/egress.

c. 120 cfm unfiltered inleakage for inleakage testing. 10 cfm is for ingress/egress.

### TABLE 15A-2

### LIMITING SHORT-TERM ATMOSPHERIC DISPERSION FACTORS FOR ACCIDENT ANALYSIS FOR VEGP (s/m<sup>3</sup>)

Location Type/ <u>Time Interval</u>	<u>(χ/Q)</u>
Site boundary	
0 to 2 h	1.8E-4
Low population zone	
0 to 2 h	7.2E-5
2 to 8 h	3.3E-5
8 to 24 h	2.2E-5
24 to 96 h	9.2E-6
96 to 720 h	2.7E-6
Control room	
0 to 2 h	1.0E-3
2 to 8 h	7.1E-4
8 to 24 h	3.1E-4
24 to 96 h	2.7E-4
96 to 720 h	2.1E-4

### TABLE 15A-3

### CORE FISSION PRODUCT INVENTORY<sup>(a)</sup>

Nuclide	Total Core Inventory (Ci)	Fuel Rod Gap Inventory (Ci) <sup>(b)(c)</sup>
I-131	1.03E+08	1.03E+07
I-132	1.50E+08	1.50E+07
I-133	2.10E+08	2.10E+07
I-134	2.26E+08	2.26E+07
I-135	1.95E+08	1.95E+07
Kr-85m	2.68E+07	2.68E+06
Kr-85	1.04E+06	3.12E+05
Kr-87	4.93E+07	4.93E+06
Kr-88	7.02E+07	7.02E+06
Xe-131m	7.13E+05	7.13E+04
Xe-133m	3.01E+07	3.01E+06
Xe-133	2.12E+08	2.12E+07
Xe-135m	4.18E+07	4.18E+06
Xe-135	4.65E+07	4.65E+06
Xe-138	1.69E+08	1.69E+07
I-127	4.45 kg	1.34 kg
I-129	18.3 kg	5.49 kg

a. Source term at end of fuel cycle with zero decay.

b. The gap fractions are assumed to be 10% of the core activity for all isotopes except for Kr-85, I-127, and I-129 for which the gap fraction is assumed to be 30%. An exception to this is taken for the fuel handling accident which assumes a gap fraction of 12% for I-131, following the recommendation in NUREG/CR-5009.

c. The gap fractions assumed for the fuel handling accident analyses in subsection 15.7.4 are based on Regulatory Guide 1.195.

### TABLE 15A-4

### PRIMARY COOLANT NOBLE GAS CONCENTRATIONS<sup>(a)</sup>

Nuclide	Concentration (µCi/g)
Kr-85m	2.04
Kr-85	8.37
Kr-87	1.28
Kr-88	3.68
Xe-131m	2.02
Xe-133m	17.6
Xe-133	256
Xe-135m	0.56
Xe-135	8.30
Xe-138	0.74

a. Based on operation with 1.0% of power produced by fuel rods with cladding defects and with no purge of noble gas activity from the volume control tank to the gaseous waste processing system.
# TABLE 15A-5

Nuclide	Whole Body Dose Conversion Factor (rem-m <sup>3</sup> /Ci-s)	Beta-Skin Dose Conversion Factor (rem-m <sup>3</sup> /Ci-s)	Thyroid Dose Conversion Factor (rem/Ci)
I-131	6.73E-02	3.20E-02	1.08E+06
I-132	4.14E-01	1.12E-01	6.44E+03
I-133	1.09E-01	9.04E-02	1.80E+05
I-134	4.81E-01	1.43E-01	1.07E+03
I-135	2.95E-01	7.99E-02	3.13E+04
Kr-85m	2.77E-02	5.11E-02	NA
Kr-85	4.40E-04	4.98E-02	NA
Kr-87	1.52E-01	3.36E-01	NA
Kr-88	3.77E-01	7.90E-02	NA
Xe-131m	1.44E-03	1.51E-02	NA
Xe-133m	5.07E-03	3.13E-02	NA
Xe-133	5.77E-03	1.06E-02	NA
Xe-135m	7.55E-02	2.17E-02	NA
Xe-135	4.40E-02	6.47E-02	NA
Xe-138	2.13E-01	1.48E-01	NA

#### DOSE CONVERSION FACTORS USED IN ACCIDENT ANALYSIS<sup>(a)</sup>

 a. Whole body dose conversion factors are from Table III.1 of EPA Federal Guidance Report No. 12 (EPA 402-R-93-081, September 1993). Beta-skin dose conversion factors are from DOE/EH-0070, "External Dose-Rate Conversion Factors for Calculation of Dose to the Public," July 1988. Thyroid dose conversion factors are from Table 2.1 of the EPA Federal Guidance Report 11 (EPA-520/1-88-020, September 1988).

#### VEGP-FSAR-15A

#### TABLE 15A-6

# REACTOR COOLANT IODINE CONCENTRATIONS FOR 1 $\mu Ci/g$ AND 60 $\mu Ci/g$ OF DOSE EQUIVALENT I-131 $^{(a)}$

# Reactor Coolant Concentration (µCi/g)

Nuclide	1 μCi/g Dose Equivalent I-131	60 μCi/g Dose Equivalentl-131
I-131 I-132	0.74 0.75	44.4 45.0
I-133	1.41	84.6
I-134	0.18	10.8
I-135	0.69	41.4

a. Values are based on the thyroid dose conversion factors in table 15A-5.

# TABLE 15A-7

# IODINE APPEARANCE RATES IN THE REACTOR COOLANT (Ci/s)

	Equilibrium Appearance Rates due to Fuel Defects <sup>(a)</sup>	Appearance Rates Due to an Accident-Initiated Iodine Spike <sup>(b)</sup>
I-131	7.3 x 10 <sup>-3</sup>	3.6
I-132	2.3 x 10 <sup>-2</sup>	11.5
I-133	1.7 x 10 <sup>-2</sup>	8.4
I-134	1.2 x 10 <sup>-2</sup>	5.9
I-135	1.2 x 10 <sup>-2</sup>	5.9

a. Based on RCS concentration of 1  $\mu$ Ci/g of dose equivalent I-131, an RCS leakage rate of 12 gpm and letdown flow of 140 gpm (130 gpm design letdown flow plus 10 gpm for instrument uncertainty) and dose conversion factors from table 15A-5.

b. 500 x equilibrium appearance rate.



# **APPENDIX 15B**

(This appendix has been deleted)

# **TECHNICAL SPECIFICATIONS**

#### 16.1 PRELIMINARY TECHNICAL SPECIFICATIONS

The preliminary Technical Specifications were provided in the VEGP Preliminary Safety Analysis Report as part of an application for a construction permit. The construction permit for VEGP was issued on June 28, 1974. Therefore, this section is not applicable to the VEGP Final Safety Analysis Report.

# 16.2 PROPOSED FINAL TECHNICAL SPECIFICATIONS

#### 16.2.1 FOREWORD

The following paragraphs briefly describe the applicability, format, and schedule for the development of the VEGP Technical Specifications.

#### 16.2.2 APPLICABILITY

The Nuclear Regulatory Commission's (NRC's) Standard Technical Specifications for Westinghouse Pressurized Water Reactors (NUREG-0452) and Standard Radiological Effluent Technical Specifications for Pressurized Water Reactors (NUREG-0472) will be adapted to reflect the VEGP design.

#### 16.2.3 FORMAT

The format of the Technical Specifications will address the categories required by 10 CFR 50 and will consist of six sections covering definitions, safety limits and limiting safety systems settings, limiting conditions for operation (LCOs), surveillance requirements, design features, and administrative controls. The LCOs and surveillance requirements (sections 3 and 4) will be presented in a combined format, with each LCO appearing first, followed immediately by the applicable surveillance requirements. The combined section 3/4 will be subdivided into 12 subsections covering the following areas:

- Reactivity control systems.
- Power distribution limits.
- Instrumentation.
- Reactor coolant system.
- Emergency core cooling systems.
- Containment systems.
- Plant systems.
- Electrical power systems.
- Refueling operations.
- Special test exceptions.
- Radioactive effluents.
- Radiological environmental monitoring.

# 16.2.4 SCHEDULE

The Technical Specifications for VEGP Units 1 and 2 will be submitted to the NRC approximately 15 months before the scheduled fuel load date of each unit. This submittal will be separate from the VEGP Final Safety Analysis Report but will be incorporated into the VEGP docket by confirmatory letter. The Technical Specifications for the VEGP will be based on the current version of the NRC's Standard Technical Specifications for Westinghouse Pressurized Water Reactors and Standard Radiological Effluent Technical Specifications for Pressurized Water Reactors.

### 16.3 TECHNICAL SPECIFICATION IMPROVEMENT PROGRAM

The Technical Specification Improvement Program for VEGP resulted in the inclusion of certain technical requirements into the FSAR. These improvements are provided below. Changes to these requirements shall be reviewed and approved in accordance with VEGP administrative procedures.

#### 16.3.1 REQUIREMENT 1 - REACTOR TRIP SYSTEM RESPONSE TIMES

The reactor trip system response times are addressed in FSAR paragraph 7.2.1.2.6, Minimum Performance Requirements.

#### 16.3.2 REQUIREMENT 2 - ENGINEERED SAFETY FEATURE ACTUATION SYSTEM RESPONSE TIMES

The engineered safety feature actuation system response times are addressed in FSAR paragraph 7.3.1.1.2.6, Minimum Performance Requirements.

# 16.3.3 REQUIREMENT 3 - LOOSE PART DETECTION SYSTEM

The loose part detection system requirements are addressed in the Technical Requirements Manual by TR 13.3.8, Loose Part Detection System.

# 16.3.4 REQUIREMENT 4 - REACTOR VESSEL MATERIAL IRRADIATION SPECIMENS

The reactor vessel material irradiation specimen requirements are addressed in FSAR paragraph 5.3.1.6, Material Surveillance.

#### 16.3.5 REQUIREMENT 5 - CONTAINMENT ISOLATION VALVES

The containment isolation valve isolation time requirements of Technical Specification 3.6.3, Containment Isolation Valves, are addressed in FSAR paragraph 6.2.4.2.1, General Description.

#### 16.3.6 REQUIREMENT 6 - CONTAINMENT PENETRATION CONDUCTOR OVERCURRENT PROTECTION

The containment penetration conductor overcurrent protection requirements are addressed in Technical Requirements Manual TR 13.8.1, Containment Penetration Conductor Overcurrent Protective Devices.

# 16.3.7 REQUIREMENT 7 - AREA TEMPERATURE MONITORING

The area temperature monitoring requirements are addressed in Technical Requirements Manual TR 13.7.5, Area Temperature Monitoring.

# 16.3.8 REQUIREMENT 8 - TURBINE OVERSPEED PROTECTION

The turbine overspeed protection requirements are addressed in the Technical Requirements Manual by TR 13.3.5, Turbine Overspeed Protection.

#### 16.3.9 REQUIREMENT 9 - TECHNICAL REQUIREMENTS MANUAL

Technical requirements that are licensing commitments, but which may be controlled by the licensee in accordance with the process for changes, tests, and experiments as provided in 10 CFR 50.59, can be maintained in the Technical Requirements Manual (TRM).

The TRM contains selected requirements that apply to the operation of VEGP with the intent being to provide a single, prominent, and easily accessible document for operating staff to reference and which will support the operating staff's compliance with these requirements with a minimum of effort. These requirements are conditions for operation, associated action requirements, and surveillance requirements with the format for presentation of the requirements being the same as used in the VEGP NUREG-1431 based Technical Specifications.

The administrative controls for the TRM are the same as used for the control of the FSAR. These administrative controls ensure proposed TRM changes do not require prior NRC approval, or if prior approval is required, the controls ensure NRC review and approval are obtained, prior to implementation of the change. Additionally, other federal regulations may apply to the control of certain technical requirements and may be so stated at the appropriate location in the body of the TRM.

# 16.3.10 REQUIREMENT 10 - REACTOR COOLANT SYSTEM PRESSURE ISOLATION VALVES AND LEAKAGE LIMITS

The reactor coolant system pressure isolation valves and leakage limit requirements of Technical Specification 3.4.14, RCS Pressure Isolation Valve (PIV) Leakage, are addressed in FSAR paragraph 5.4.12.4, Tests and Inspections.

# 16.4 RISK MANAGED TECHNICAL SPECIFICATIONS

The Risk Managed Technical Specifications (RMTS) will allow the calculation of risk informed completion times (RICT) for certain LCO Conditions. Those are identified in the Completion Time column of the particular LCO Condition, with the phrase, "…or in accordance with the Risk Informed Completion Time Program". Without this phrase in the LCO Condition Completion Time column, the particular Condition is not in the program and a RICT cannot be calculated.

The program is consistent with the guidelines of NEI 06-09, Rev. 0, "Risk Informed Technical Specifications Initiative 4b, Risk Managed Technical Specifications (RMTS) Guidelines".

# **16.4.1 PRA FUNCTIONALITY**

The RICT calculation performed depends on the configuration of the plant at the time of the inoperability of the component. Additionally, the inoperable component may be declared "PRA Functional" if the component's inoperability does not affect the PRA success criteria. For example, if an ECCS pump becomes inoperable, its PRA Functionality status would depend on the ability of the pump to meet the PRA flow rate success criteria. This is per the guidelines of NEI 06-09, section 11, which lists further requirements and restrictions on declaring a component PRA Functional.

PRA Functionality for a case of all trains and equipment required by the Technical Specifications inoperable has additional restrictions identified in Section 5.5.22 of the Technical Specifications.

#### 16.4.2 PRA TECHNICAL ADEQUACY

The on-record PRA model that forms the basis for the VEGP Configuration Risk Management (CRM) Tool has been developed to the requirements of Reg. Guide 1.200, "An approach for determining the technical adequacy of probabilistic risk assessment results for risk-informed activities" and is subjected to peer reviews per the requirements of NRC endorsed applicable PRA standards and SNC procedures. These peer reviews are formally documented along with the findings and observations of the review, and their corresponding resolutions.

# 17.2 OPERATIONS QUALITY ASSURANCE PROGRAM

The operations phase quality assurance program for Vogtle Electric Generating Plant (VEGP) is designed to assure the plant's safe and reliable operation and to satisfy the quality assurance (QA) requirements of Appendix B to 10 CFR Part 50. The quality assurance program applicable to operation phase activities for VEGP is described in the Southern Nuclear Operating Company (SNC) Quality Assurance Topical Report (QATR). Quality assurance program requirements formerly contained in VEGP FSAR Section 17.2 are superseded by those contained in the SNC QATR.

FIGURE 17.2.1-1 DELETED

# 18.0 HUMAN FACTORS ENGINEERING

# 18.1 <u>CONTROL ROOM DESIGN</u>

# 18.1.1 CONTROL ROOM DESCRIPTION

#### 18.1.1.1 Introduction

In accordance with the TMI Task Action Plan I.D.1 of NUREG-0660, NUREG-0694, and NUREG-0737 (references 1, 2, and 3, respectively) a control room design review was performed. The purpose of this review was the identification and analysis of human engineering deficiencies (HEDs) so they could be corrected in the final design or accommodated in the administrative and training procedures. This chapter presents the current control room design, the review procedures, and their results.

The preliminary control room design review included an assessment of the following:

- Control room layout.
- The adequacy of the information provided.
- The arrangement and identification of important controls and instrumentation displays.
- The usefulness of the audio and visual alarm systems.
- The information recording and recall capability.
- Other considerations of human factors that have an impact on operating effectiveness.

The final control room design review was conducted prior to licensing, as required by Supplement 1 to NUREG-0737, dated December 17, 1982. This review commenced approximately 1 year prior to Unit 1 fuel loading.

#### 18.1.1.2 Description

The control room plan is shown in figure 18.1-1. The layouts of the Unit 1 main control board (QMCB); electrical auxiliary control board (QEAB); heating, ventilation, and air-conditioning systems panel (QHVC), and miscellaneous systems and equipment panel (QPCP) are shown in drawings 1X3AE01-20, 1X3AE01-21, 1X3AE01-24, 1X5AB01-66, 1X5AB01-67, 1X5AB01-557, 1X5AB01-560, 1X5AB01-563, 1X6AV02-46, 1X6AV02-47, 1X6AV02-48, 1X6AV02-49, 1X6AV02-50, 1X6AV02-60, and AX3D-CA-L50A.

A. QMCB

The QMCB is divided into groups of controls by black demarcation lines. These groups may be entire systems or could be divided into subsystems depending on

the complexity of the system involved. The following are demarcation groups contained on each section of the QMCB:

- 1. QMCB Section A1
  - a. Circulating water.
  - b. Turbine plant cooling water.
  - c. Turbine plant closed-loop cooling water.
  - d. Instrument and service air.
  - e. Nuclear service cooling water.
  - f. Reactor makeup.
  - g. Component cooling water.
  - h. Auxiliary component cooling water.
  - i. Containment spray system.
  - j. Safety injection (SI) pumps.
- 2. QMCB Section A2
  - a. Residual heat removal (RHR).
  - b. Chemical and volume control system (CVCS) charging pumps.
  - c. Waste processing system.
  - d. CVCS letdown.
  - e. CVCS seal injection.
  - f. CVCS boron injection.
  - g. CVCS boron meter.
  - h. CVCS thermal regeneration.
- 3. QMCB Section B1
  - a. Main steam isolation valves.
  - b. Main steam, steam dump, and steam generator blowdown.
  - c. Feedwater, steam generator feed pump turbine A.
  - d. Feedwater, steam generator feed pump turbine B.
  - e. Auxiliary feedwater.

- f. Feedwater and condensate system.
- g. Main steam and main steam relief drains.
- 4. QMCB Section B2
  - a. Main steam and main steam relief drains.
  - b. Extraction steam.
  - c. Turbine.
  - d. Generator.
- 5. QMCB Section C
  - a. Reactor control.
  - b. Reactor control and nuclear instrumentation.
  - c. Reactor coolant system (RCS) pressurizer pressure.
  - d. RCS pressurizer level.
  - e. RCS reactor coolant pumps.
- 6. QMCB Section D
  - a. Head vents.
  - b. Post-accident monitoring system (PAMS) plasma displays.
  - c. Main steam isolation valve bypass valves.

#### B. QEAB

1. QEAB Section A - Plant Auxiliaries

QEAB section 1A covers electrical mimic boards for Unit 1 systems (common for both units on 1A and 1B). QEAB section 2A covers electrical mimic boards for Unit 2 systems.

2. QEAB Section B

QEAB section 1B covers electrical mimic boards for Unit 1 system switchyards. QEAB section 2B covers electrical mimic boards for Unit 2 switchyard.

#### C. QHVC

- 1. QHVC Section 1
  - a. Control building and equipment room, level 4.
  - b. Control building auxiliary relay room and computer room exhaust.
  - c. Control building control room (normal), level 1.
  - d. Control building control room engineered safety features (ESF), level 1.
  - e. Equipment room ESF, level B.
  - f. Control building electrical penetration room, level B.

- g. Electrical tunnel ventilation.
- 2. QHVC Section 2
  - a. ESF coolers.
  - b. Piping penetration area and recycle holdup tanks.
  - c. Containment heat removal ESF.
  - d. Containment heat removal (normal).
  - e. Containment air purification.
  - f. Containment air purification and post-accident cavity.
  - g. Containment control rod drive mechanism cavity and reactor support.
  - h. Diesel generator building.
  - i. Control building levels A and B (normal) and lighting switchgear (normal).
- 3. QHVC Section 3
  - a. Auxiliary building (normal).
  - b. Auxiliary feedwater pumphouse and control building control room chiller room.
  - c. Piping penetration ventilation and turbine building exhaust fans.
  - d. Control building cable spreading rooms, levels A2 and A3.
  - e. Control building levels 1 and 2 smoke exhaust.
  - f. Fuel handling building normal exhaust/air-conditioning units and fuel pool area ventilation.
  - g. Fuel handling building system isolation and post-accident airconditioning filter units.
  - h. Fuel handling building normal air-conditioning units supply header isolation.

#### D. QPCP

- 1. Section 1
  - a. Plant demineralized water (Unit 1 only).
  - b. Nuclear service cooling tower makeup water.
  - c. River makeup water (Unit 1 only).

- d. Waste water.
- e. Turbine plant cooling water.
- f. Core monitor panel.
- g. Main steam atmospheric relief valve.
- h. Cooling water system temperature, train A.
- i. Cooling water system temperature, train B.
- 2. QPCP Section 2
  - a. SI test line valves.
  - b. Main steam.
  - c. Steam generator feed pump turbine drains.
  - d. Reheater drains.
  - e. Auxiliary feedwater.
  - f. Feedwater and condensate chemical injection.
  - g. Containment building and auxiliary building drains.
  - h. Fire protection.
  - i. Refueling water storage tank sludge mixing.
- 3. QPCP Section 3
  - a. Train A condensate and feedwater.
  - b. Train A main steam.
  - c. Train A steam generator blowdown sampling.
  - d. Train A fire protection and containment service air.
  - e. Train A line break monitors and CVCS letdown header.
  - f. Train A containment hydrogen monitoring.
  - g. Train A containment air monitoring.
  - h. Train A pressurizer liquid.
  - i. Train A SI sample.
  - j. Train B condensate and feedwater.

- k. Train B main steam.
- I. Train B steam generator blowdown sampling.
- m. Train B fire protection and containment service air.
- n. Train B line break monitors and CVCS letdown header.
- o. Train B containment hydrogen monitoring.
- p. Train B containment air monitoring.
- q. Train B pressurizer liquid.
- r. Train B SI sample.

These control panels and their components provide display, control, alarm, and computer functions to assist in remote control, monitoring, and recording of plant operations during startup, normal, shutdown, and emergency operations. The control room and its boards are designed and configured to enable the operator to control the plant as required in an efficient manner without undue risk to public health and safety or potential problems to plant equipment. Should the need exist, the operator can take appropriate measures or can initiate protective action using the control room controls.

If the control room becomes uninhabitable, the necessary controls and displays for shutdown of the plant are provided at two secure, remotely located shutdown panels. (See section 7.4.)

#### 18.1.2 DESIGN BASES

The following design bases constitute the design requirements for the control room evaluation.

#### 18.1.2.1 <u>Anthropometrics</u>

This is defined as statistical data of physical dimensions for the control room personnel.

- A. The QMCB is a combination bench-vertical type.
- B. The remainder of the control boards are the vertical type.

#### 18.1.2.2 Auditory Communications

This is defined as indications or displays for equipment or subsystem conditions as interpreted by the control room personnel in auditory modes.

A. The annunciator is designed to provide five mutually discriminable audible signals to allow the operator to be immediately cognizant of an alarm and to be able to distinguish the relative board location of that alarm based on sound characteristics. To ensure immediate notice by the operator, each audible alarm is designed to have a minimum sound level differential of 10 dBA above the ambient level.

- 1. The QMCB annunciator panels have a medium-pitched, medium-warble horn sound.
- 2. The QEAB has a low-pitched, low-warble horn sound.
- 3. The QHVC is designed to have its own horn with a low-pitched, high-warble sound.
- 4. The QPCP is designed to have a separate horn with a high-pitched, high warble sound.
- 5. The ring-back (reset of alarm) is common to all control boards, using a bell tone.
- 6. First-out panel on the QMCB has a high-pitch, high-warble sound (notably different than the QPCP).

#### 18.1.2.3 Plant Communications

This is defined as the means provided for intercommunication between the control room, technical support center, emergency operations facility, and personnel on duty in various parts of the plant, as described below.

- A. Communication to/from the control room is provided using sound-powered telephone systems. The equipment is available for maintenance (six jacks/box), refueling (two jacks/box), and shutdown purposes (two jacks/box).
- B. The communication system within the plant is a public address system (one page line and party line common to all zones and four party lines). This system is for roving operator communications and is provided with the capability for the merger of four zones: Unit 1, Unit 2, administration building, and outside areas. This system provides remote merging in:
  - 1. Unit 1 control room.
  - 2. Unit 2 control room.
  - 3. Central alarm station.
  - 4. Secondary alarm station.
  - 5. Captain's office in the PESB.

This also functions as an emergency alarm (supplied with emergency power from an uninterruptible power supply system). This system is actuated by a multi-tone generator applicable for discriminable tones. Capability for initiation and selection of the tone for the emergency alarm is provided in the control room.

- C. Plant offsite communication system uses the public telephone systems.
- D. Other communication facilities are:
  - 1. Microwave communications at 57 MHz.
  - 2. Walkie-talkie with external antenna for radio contact between the control room, technical support center, emergency operations facility, and field personnel.
  - 3. Security intercom linking control room, security subsystem, and security monitoring station.

4. Federal Telephone System (FTS) for the emergency notification system (ENS) and the health physics network.

Communications are further discussed in subsection 9.5.2.

# 18.1.2.4 <u>Controls</u>

Controls are defined as devices actuated by the control room personnel (manual mode) or by the systems (automatic mode) to influence system functioning.

- A. The control devices used on the instrument panels are designed to minimize the size of the panels, while still allowing convenient and reliable operation. Insofar as is practicable, uniform types of control devices are used throughout the control room.
- B. Control devices of redundant trains are designed for separation by physical barriers where there is an operational requirement for redundant trains to be close to each other.
- C. Control devices for safety systems are designed to satisfy Seismic Category 1 requirements.
- D. Control switches selected for the auxiliary board and panels are similar to those selected for the QMCBs (Electroswitch series 20 or equivalent).
- E. Control switch development is designed to comply with the following:

1.	Left-right (right throw) or bottom-top (up-throw)	Start, close (circuit breaker), open (valve), on, raise, and other positive or increasing functions and associated indicating lights.
2.	Right-left (left throw) or top-bottom (down-throw)	Stop, trip (circuit breaker), close (valve), off, lower, and other negative or decreasing functions and associated indicating lights.

- F. Where possible, automatic controls for parallel valves are designed to be kept in parallel (left to right). The series valves are designed to be kept in series (top to bottom) to make it compatible with the flow direction on the control board layout. Flow paths for system controls are designed for layout on the QMCB starting with suction, pump, and discharge going from top to bottom.
- G. The analog control stations are Westinghouse 7300 series operator interface modules, which use Westinghouse-qualified designs for split architecture. Controls on the operator interface modules are marked to indicate in which direction to operate the controller to increase or decrease the variable to avoid operator errors.
- H. Controls are designed for functional arrangement on control boards so that systems or equipment supporting or interacting with the desired function are located in the same general area on the control board for ease of operation.
- I. The controls associated with core reactivity are located in the central area of the QMCB (section C) along with the other principal reactor controls.

- J. Speed and flow controls for the charging pumps are located so that the flow control and pressurizer control are in the reactor control section of the QMCB (section C).
- K. Controls for heating, ventilation, and air-conditioning (HVAC) and control rod drive motor fans are on the QHVC panel.
- L. Control switches for rod motion are designed to have a clear line of vision to the rod step counters and count rate recorder so that the operator can continuously observe the effects of the rod motion.
- M. Control switches for reactor trip and manual SI are located away from the other control switches to avoid inadvertent actuation.
- N. Functional groups of controls are located in accordance with operational sequence having left-to-right order and/or top-to-bottom order of use based on the accepted operator compatibility principles, namely, spatial, movement, and conceptual (e.g., suction isolation valve, pump, and discharge isolation valve, in that order).

# 18.1.2.5 <u>Control and Display Integration</u>

This is defined as the relative locations of related control and display devices.

- A. Controls and displays (including annunciators and recorders) are arranged in functional groups and in a functional sequence with the displays above their corresponding controls to prevent the arm of the operator from obscuring the indicated response to control adjustments.
- B. Main feedwater and bypass feedwater valve controls and displays are located close to each other to facilitate transferring modes of operation as power level is changed.
- C. Frequently used control and display systems are located on or near the central portion of the QMCBs to minimize operator movement.

#### 18.1.2.6 Design for Maintainability

This is defined as the degree to which panels are designed to accommodate preventive maintenance and corrective repair or replacement functions as quickly as possible with minimum disturbance to operator functions.

- A. Control boards in the control room are designed with split architecture controllers (controllers separate from process module) to facilitate maintenance and to reduce congestion in the operator's area of the control room.
- B. Control panel components are primarily modular equipment to simplify maintenance as well as to reduce the maintenance time.
- C. Maintenance accessibility is designed for minimum operator maintenance and technician interference by maintaining sufficient aisle clearance and by positioning panels so that there is adequate aisle space for panel back access.

#### 18.1.2.7 Design for Personnel Requirements

This is defined as the norms in the power industry for control room personnel physical and mental capabilities.

- A. A shift supervisor and operators are available in the control room for each shift as specified in Technical Specification 5.2.2, 10 CFR 50.54, and the Technical Requirements Manual.
- B. A shift technical advisor or similarly qualified senior reactor operator is available to assist in the interpretation of information and control actions.
- C. The qualifications for the operators are as specified in chapter 13.

#### 18.1.2.8 Environment

This is defined as the external physical elements affecting the control room personnel in their performance of their functions. The environment for the main control room is designed for the following conditions:

- A. Temperature of the control room is normally  $75^{\circ}$ +5°F with maximum allowable temperature of 85°F.
- B. Pressure of the control room is slightly above atmospheric.
- C. Relative humidity is maintained between 10 and 60 percent.
- D. The fluorescent lighting in the control room is designed to provide a uniform horizontal intensity of 50 to 100 fc adjustable in the control room general area, 50 to 100 fc adjustable at the console surface, and approximately 30 fc during an emergency resulting from loss of normal 120-V ac lighting. The control room lighting system is further described in subsection 9.5.3.
- E. Acoustical provisions are included in the design of the primary operating area of the control room to minimize the ambient noise to a sound level of 50 to 65 dBA.

#### 18.1.2.9 <u>Hazards/Safety</u>

These are defined as the physical, chemical, and other hazards to control room personnel.

- A. The HVAC system is designed to provide automatic protection against radiation and smoke. Manual operation provides protection against toxic chemicals.
- B. The control room design includes protection against dangerous voltage and ground faults by appropriate use of fuses, breakers, and grounding techniques.
- C. The control room is designed for operator warning should a hazardous situation arise.
- D. The control room is provided, where required, with special ladders for bulb replacement.
- E. Switch handles and other devices are designed so as not to protrude from boards and panels in a manner that could produce accidental actuation. A guard rail around the QMCB also prevents accidental activation.

- F. The control room is provided with fire, smoke, and radiation detection. The fire hazard and protection analysis of the control room is provided in appendix 9A.
- G. The control room is designed for automatic warnings against unsafe or emergency conditions in the plant by alarms.
- H. Automatic actuation for control room isolation is also provided with operator actuation as a backup.
- I. Accessibility of emergency and protective equipment to the control room operators is discussed in section 6.4.
- J. Control room suspended ceiling is designed and constructed to ensure that the ceiling will not fall or compromise the functioning of safety-related equipment during or after a safe shutdown earthquake.

# 18.1.2.10 Information Utilization

This is defined as the information processing facility, such as entry, access, and retrieval, provided to improve control room personnel performance. The control room and boards are designed to have the following features:

- A. Automated control of safety feature equipment if action is required within 10 min from the time indication is available for showing that action is required.
- B. Operator involvement in selection of automatic versus manual control of loops.
- C. Selection of Monitors and other computer display functions.
- D. Operator involvement in:
  - 1. Startup.
  - 2. Shutdown.
  - 3. Load change.
  - 4. Base load.
  - 5. Trips.
  - 6. Accidents.
  - 7. Emergencies (including fire and security).
- E. Selection of the particular train to use when a system has two trains.
- F. Display, diagram, or information selection (including details of annunciated items if required).
- G. Data handling and display (calculation, evaluation, and screening information before display).
- H. Annunciator interpretation, alarm handling, and display.
- I. Manual overrides for essential system.
- J. Backup facility for highly significant parameter control.
- K. Display of important parameters with the ability to call up backup information as needed by the operator to take any required action.

# 18.1.2.11 Labeling (Grouping, Marking)

This is defined as alphanumeric, color, and other visual methods used for controls and displays to improve the performance of control room personnel.

A color coding scheme for the switchplates on the QMCB, QEAB, QHVC, QPCP, PSDA, and PSDB is implemented in a plant procedure.

Additionally:

- A. Controls located on the bench board section of the QMCB are grouped by subsystems divided by the demarcation lines (paragraph 18.1.1.2.A) and enveloped, where room permits, with hierarchical labels for subsystems at the top center of the envelopes to improve recognition of functional grouping.
- B. Labels are located to minimize interference with operator view and to avoid interference with other control functions.
- C. Labels are designed for legibility and visibility based on the contrast between the lettering and its background.
- D. Labels are designed to have white letters on black tags. Black on white is sometimes used to highlight a display.
- E. Labels are designed with 3/16-in. capital letters. Annunciator window engravings are designed with 1/4-in. or 3/16-in. letters. The 1/4-in. letters are used to improve readability when message length permits.
- F. Controls and displays are labeled with service description and tag number engravings. In addition, the control switch modules shall contain engravings for switching development functions, equipment-actuated functions, and train, if applicable.
- G. Labels are placed above the instruments and on the escutcheon plates for control switches to allow adequate viewing from a distance of about 3 ft.
- H. Labels for similar devices throughout a board are designed to be uniform in style, size, lettering, and use of abbreviations, with the exception of integral panels supplied by the vendor and inserted into the boards or panels as a unit.
- I. Labels are designed and mounted so that they cannot easily be damaged or removed.
- J. Labels and tag numbers are designed for accessibility and visibility during maintenance.
- K. Labels are concise with minimum repetitive information and are directly usable with minimum decoding and interpretation of the service descriptions and abbreviations. A hierarchical label is used to highlight functional grouping.
- L. Labels are not designed to describe engineering characteristics, name of manufacturer, trademarks, or nonfunction-related nomenclatures of the equipment.
- M. Shades of colors used for mimics on the QEABs are designed to have maximum contrast between the mimic bus and the boards.
- N. Safety-related post-accident monitoring instrumentation is identified by a dark red line on the black bezel base of each instrument.

#### 18.1.2.12 Visual Display (Meters, Recorders, Lights, Monitors, and Annunciators)

This is defined as indication and display of equipment or subsystem conditions in visual modes.

#### 18.1.2.12.1 General

- A. Where possible, displays are arranged in the sequence in which they are used.
- B. Unusual aids, such as stools and ladders, extra lighting, etc., are not required to read or gain access to a display.
- C. Information for different types of activation are not combined unless activation requires the same information.
- D. In a standing position, the most frequently used displays are, insofar as possible, located at the eye level of the operator.
- E. Displays frequently used in conjunction are grouped together.
- F. Displays are located where they can be read with accuracy and minimum parallax.
- G. Scale face and graduation markings are designed to have a high degree of contrast.
- H. Displays are designed so that glare does not interfere with readability when viewed from a reasonable location.
- I. Displays which cannot or may not be watched continuously, but which need vigilance in monitoring, are designed to have a suitable auditory or visual alarm.
- J. Multiple displays grouped together are designed to have uniform brightness across the full range of the display faces.

#### 18.1.2.12.2 Meters

- A. Meters, dials, and instruments are designed for size and location so they can be read from normal operating positions.
- B. Information presentations are designed in such a form that minimum interpretation or decoding is necessary for operator implementation.
- C. The meter pointer is designed to extend to, but not obscure, the graduation marks.
- D. The meter pointer is designed for mounting as close as possible to the dial face to reduce or eliminate parallax and shadows.
- E. Meter scales generally have graduations of 1, 2, 5, 10, or any of their multiples.
- F. The display indicator does not move after the control movement stops.
- G. System indicators in a flow path are arranged in accordance with flow path from left to right; i.e., inlet temperature, flow, outlet temperature.
- H. Meters have nonglare covers where needed.
- I. The meter scales of selected instruments are color coded to indicate the normal operating region.

J.  $T_{avg}/T_{ref}$  indicators are located next to each other for easy comparisons.

#### 18.1.2.12.3 Recorders

- A. Whenever possible, small continuous multipen recorders, having dimensions of approximately 6 in. by 6 in. or 3 in. by 6 in. are used on the QMCB.
- B. Scanning-type indicators are of the low-profile type to reduce panel dimensional requirements.

#### 18.1.2.12.4 Lights

- A. Indicating lights are designed to show information regarding equipment status.
- B. Indicating lights are designed to show status derived from equipment response and not just to indicate control positions. They are designed to show information needed for effective system operation.
- C. Both the red and the green lights are designed to be lit for motor-operated valves and other devices with considerable travel time while the device is in an intermediate position.
- D. The indicating lights for different operational conditions are designed as follows:
  - 1. Red Equipment or process operating, flowing, or in an increasing condition, breaker closed, valve open.
  - 2. Green Equipment or process not operating, not flowing, or in a decreasing condition, breaker open, valve closed.
  - 3. Amber Trip (automatic protection and not by operator action or process; exceptions exist for a few valve indications where it is necessary to heighten operator awareness of off-normal valve configurations).
  - 4. White Electrical potential available or other special uses.

Use of blue indicating lights is limited because of the low level of brightness obtained with this color.

- E. Indicating lights are designed for showing the status of equipment interlocked with the controls of associated equipment.
- F. Indicating lights installed on the control boards are designed for uniform size and use standard abbreviations for their engravings.

#### 18.1.2.12.5 System Status Monitoring Panel

The system status monitoring panel (QBPS), shown in figure 18.1-1, is described in detail in subsection 7.5.5. All lights on the QBPS are white.

#### 18.1.2.12.6 Annunciators

- A. The annunciator displays alarms that denote an abnormal equipment status or plant condition. Nuisance alarms are kept to a minimum by suitable permissive interlocking contacts.
- B. Annunciators are designed for high reliability, using solid state equipment.
- C. The annunciator system is designed to provide reasonable assurance that failure of one alarm circuit does not disable any other alarm circuit. Preoperational testing verified that failure of one alarm card did not disable the adjacent alarm circuit. This was demonstrated by removing an alarm card and testing the adjacent circuit.
- D. Annunciator and computer alarms are designed not to be duplicated, except for the QBPS and selected computer sequential alarms.
- E. Each section of the QMCB or QEAB has a three-pushbutton station to acknowledge, test, and reset the annunciator windows located in that section of the board.
- F. The HVAC panel and miscellaneous system/equipment panel have the capability to acknowledge, test, and reset the annunciator only by the pushbutton station located on their respective panels.
- G. All annunciators are designed as non-Class 1E. The annunciator contacts accept signals from Class 1E circuits which are isolated from the Class 1E system.
- H. The power supply for annunciators is obtained from a non-Class 1E battery.
- I. Control room annunciator windows are designed for white opaque; colors are obtained by installing color lenses.
- J. The alarm sequences for the main control room annunciator are designed as indicated in tables 18.1-1, 18.1-2, and 18.1-3.
- K. The three-pushbutton stations have the following functions:
  - 1. Acknowledge To silence the horn and put the alarm light on steady.
  - 2. Test To sound the horn and put all alarm lights on a panel or control board section on fast (or gallop for first-out sequence alarms) flash.
  - 3. Reset To silence the horn and extinguish the lamp.
- L. Annunciator nominal window dimensions are 1.4 in. by 3 in.; letter size is 3/16 in. or 1/4-in.
- M. Dual alarms (such as high-low, high-high, etc.) are only allocated a single alarm window, provided a second point of information is available to distinguish between the two conditions (such as an indicator, monitor lights, or recorder).
- N. Each local annunciator panel is designed to have a common retransmitting contact to operate a single window in the control room annunciator (exceptions are the emergency diesel generator local annunciator panels which are duplicated on the QEABs).
- O. Reflash (occurrence of another alarm on the same window after acknowledge) capability is provided in special cases for annunciator windows with multiple contacts if required for safe and reliable operation of the plant.

#### 18.1.2.12.7 Monitors

These are defined as display devices with television-type screens to indicate a comprehensive group of information and with flexibility to call up any stored information as required.

- A. The control board is designed to locate a Monitor on the display section of the QMCB board section C for displaying the annunciated parameters from the plant computer.
- B. The control board is designed to locate a Monitor at the normal eye level on the display section of the QMCB section B2 for displaying parameters from the plant computer.
- C. The control room is designed to mount one Monitor on the operator desks to provide graphic displays.
- D. The control room is designed to mount two free-standing safety parameter display system Monitor consoles located near the QMCB, as shown in figure 18.1-1.
- E. The vertical section of control board QMCB Section B2 is designed to locate three monitors along with keyboards and touch pads for providing operator interfaces to control the main turbine, feed pump turbines, and the generator excitation system.

# 18.1.2.12.8 Plasma Display Modules

Two redundant dot matrix-type graphic/alphanumeric displays shall be provided in section D of the QMCB. These modules shall display the process variables classified in Regulatory Guide 1.97, Revision 2, as Category 1, as well as other selected variables. Plasma displays are a part of the PAMS described in section 7.5. They interface with the redundant display processing units and meet the single failure requirements.

#### 18.1.2.13 Workspace

This is defined as the space surrounding the area designated "at the controls."

- A. The control room area is designed to include dual control rooms (one per unit) with a supervisor's area positioned to observe both central control areas.
  - 1. Arrangements of the most frequently used boards and displays are designed for location readily visible to the operator in the central control area. Control boards are designed to be arranged to provide the best accessibility; i.e., shortest walking distance.
  - 2. Layout designs of the QMCB, QEAB, desks, and other panels are arranged so that physical interference among operators working in the central control area is minimized.
  - 3. The design for the central control area space and aisles allows efficient movement of operators between areas during normal, emergency, startup, and shutdown operations.
  - 4. The lines of sight to a display are designed so as not to be obstructed by poor arrangement of equipment.

5. A restricted area between the operators' desks and the QMCB and QEAB is designed for the use of licensed operators only.

# 18.1.2.14 <u>References</u>

- 1. U.S. Nuclear Regulatory Commission, "NRC Action Plan Developed as a Result of the TM1-2 Accident," <u>NUREG-0660</u>, May 1980.
- 2. U.S. Nuclear Regulatory Commission, "TMI Related Requirements for New Operating Licenses," <u>NUREG-0694</u>, June 1980.
- 3. U.S. Nuclear Regulatory Commission, "Clarification of TMI Action Plan Requirements," <u>NUREG-0737</u>, November 1980.

#### TABLE 18.1-1

# MAIN CONTROL ROOM ANNUNCIATOR ALARM SEQUENCE FIRST-OUT SEQUENCE OF OPERATION (MODIFIED)<sup>(a)</sup>

<u>Condition</u>		Field <u>Contact</u>	Lamp	<u>Horn 1</u>	Horn 2
Normal		Normal	Off	Off	Off
Alarm	First	Abnormal	Gallop flash	On	Off
	Subsequent	Abnormal	Fast flash	On	Off
Return to normal before acknowledge	First	Normal	Gallop flash	On	Off
	Subsequent	Normal	Fast flash	On	Off
Acknowledge	First	Abnormal	Slow flash	Off	Off
	Subsequent	Abnormal	Steady	Off	Off
Return to normal	First	Normal	Slow flash	Off	On
	Subsequent	Normal	Slow flash	Off	On
Reset	First	Normal	Off	Off	Off
	Subsequent	Normal	Off	Off	Off
Test	First	Normal	Gallop flash	On	Off

a. All windows will be on a first-out condition during the test sequence.

#### TABLE 18.1-2

# MAIN CONTROL ROOM ANNUNCIATOR ALARM SEQUENCE RING-BACK SEQUENCE OF OPERATION

Condition	Field <u>Contact</u>	<u>Lamp</u>	<u>Horn 1</u>	Horn 2
Normal	Normal	Off	Off	Off
Alarm	Abnormal	Fast flash	On	Off
Return to normal before acknowledge	Normal	Fast flash	On	Off
Acknowledge	Abnormal	Steady	Off	Off
Return to normal	Normal	Slow flash	Off	On
Reset	Normal	Off	Off	Off
Test	Normal	Fast flash	On	Off

# TABLE 18.1-3 (SHEET 1 OF 2)

# MAIN CONTROL ROOM ANNUNCIATOR ALARM SEQUENCE

#### Reflash Sequence of Operation - One Incoming Alarm Which Returns to Normal Before Being Acknowledged (Common Window)

Condition	Field	Lamp	Horn 1	Horn 2
	Contact	Lamp		<u>HUIII 2</u>
Normal	Normal	Off	Off	Off
Alarm <sup>(a)</sup>	Abnormal	Fast flash	On	Off
Return to norma before acknowledge	Normal	Fast flash	On	Off
Acknowledge	Normal	Slow flash	Off	On
Reset	Normal	Off	Off	Off
Refl	ash Sequence o Incoming Alarn	of Operation - Tv ns (Common Wir	vo or More ndow)	
Condition	Field <u>Contact</u>	Lamp	Horn 1	<u>Horn 2</u>
Normal	Normal	Off	Off	Off
Alarm 1 <sup>(a)</sup>	Abnormal	Fast flash	On	Off
Alarm 2 <sup>(a)</sup>	Abnormal	Fast flash	On	Off
Acknowledge	Abnormal	Steady	Off	Off
Alarm 3 <sup>(a)</sup>	Abnormal	Fast flash	On	Off
Alarm 1 returns to normal	Abnormal	Fast flash	On	Off
Acknowledge	Abnormal	Steady	Off	Off
Alarm 2 returns Abnormal to normal	Steady	Off	Off	

# TABLE 18.1-3 (SHEET 2 OF 2)

Condition	Field <u>Contact</u>	Lamp	<u>Horn 1</u>	Horn 2
Alarm 3 returns to normal	Normal	Slow flash	Off	On
Reset	Normal	Off	Off	Off

a. Incoming alarm shall always have precedence over return to normal.



## 18.2 PRELIMINARY REVIEW OF THE CONTROL ROOM AND CONTROL BOARD DESIGN

# 18.2.1 REVIEW PROCEDURES

#### 18.2.1.1 Introduction

The preliminary review of the control room and control board design was conducted by the General Physics Corporation, with the primary objective of providing an adequate evaluation of the human factors.

The review included primarily those panels which comprised the inner ring of panels in the control room.

Certain additional panels which are not in the inner ring were also reviewed. A listing of the panels reviewed is given below:

Inner Ring Panels	Outer Ring Panels
QMCB A1	HVC
QMCB A2	QPCP
QMCB C	
QMCB B1	
QMCB B2	
QMCB D	
QEAB	
	and a factor the state of the second and the state

Figure 18.1-1 illustrates the layout of the panels in the main control room.

The review procedures and results are presented in detail in reference 1 and summarized in this section. The review was performed in accordance with the guidelines of NUREG/CR-1580.<sup>(2)</sup>

The use of engineering checklists provided standards of assessment of various properties of the control room including, among other aspects:

- Functional grouping.
- Anthropometrics.
- Readability of labels.
- Controls discrimination.

These checklists were developed based upon the latest established and recommended human factors engineering criteria.

Several techniques were employed in the review of the VEGP control room. Operators were used to determine how the VEGP control board stood in relationship to similar vintage control boards. Human factors specialists judged the control board against applicable guidelines. Engineering checklists were employed to evaluate individual controls and control/display layout. Scenario evaluations were performed to assess the operability of the board. Using the above methods, the human factors engineering group was able to make assessments of:
- Anthropometrics and panel elements.
- System layout.
- Component usage.
- Readability of displays and labels.
- Coding/discrimination methods.
- Environmental conditions.
- Operability of control board.

Below is a detailed description of methodologies used in anthropometrics and panel elements analysis, system evaluation, component evaluation, and scenario evaluation. The information gathered by use of these human factors methodologies served as a basis for the development of preliminary recommendations. The evaluation of these recommendations and their resolutions are provided in subsection 18.2.2.

#### 18.2.1.2 <u>Anthropometrics</u>

Anthropometric evaluation was performed by comparing the control room design to established human factors guidelines. The main references used were Bechtel drawings, NUREG-1580,<sup>(2)</sup> and MIL-STD-1472B.<sup>(3)</sup>

The guidelines were applied three ways: by scale drawings, by use of a 1/4-in. scale panel silhouette and mannequins scaled for the 95th and 5th percentile male, and by actual measurements of reach and heights taken from the mockup.

Items examined were:

- Panel height.
- Distance of reach.
- Viewing distance.
- Viewing angle.
- Placement of controls and displays in relationship to anthropometric value.

#### 18.2.1.3 <u>System Evaluation</u>

System evaluation was performed after developing engineering checklists from the current applicable human factors engineering guidelines, specifically NUREG/CR-1580.<sup>(2)</sup>

A separate set of questions was asked about each control and each display. A list of these questions is provided below. The questions were asked about each individual component with answers recorded as a yes or no.

A. Control Checklist

- 1. Are controls placed such that they may be easily operated?
- 2. Is control located to prevent inadvertent operation?
- 3. Does control give the information needed for operation of the equipment/system?
- 4. Is the control clearly marked as to what it does?
- 5. Do controls move in the culturally normal direction (clockwise for on)?
- 6. Is this control the best type of control for the function needed?
- 7. Are controls grouped in a consistent left-to-right, top-to-bottom order?
- 8. Is control not prone to be misread or its position misinterpreted?
- B. Display Checklist
  - 1. Is display prone to be misread or misinterpreted?
  - 2. Is display the best type for indicating system/equipment function?
  - 3. Does display give the type of information needed?
  - 4. Does display movement correspond to the control movement?
  - 5. Is display in reasonable proximity to the control?

# 18.2.1.4 Component Evaluation

Component evaluation was performed by reviewing all pertinent human factors guidelines, then developing a list of these guidelines applicable to components in the control room. Additions to this list draw heavily on Nuclear Regulatory Commission (NRC) Safety Evaluation Reports.<sup>(4)(5)(6)</sup>

These documents were reviewed:

- Electrical one-line diagrams.
- Piping and instrumentation diagrams.
- Control logic diagrams.
- Instrument specification sheets.
- Instrument index.
- Digital rod position indicator specification.
- Specification for the electrical auxiliary board and the miscellaneous control board.
- Various vendor documents were utilized.

Where possible, a hands-on evaluation of controls was performed. This entailed the use of testing devices such as a snap-on torque meter or an Amtex torsional testing machine. In addition, observation of equipment equivalent to that found in the VEGP control rooms was made at various operating simulators. In particular, the Sequoyah and McGuire simulators were used.

Safety evaluation reports were reviewed to determine how the NRC implemented the various guidelines.

### 18.2.1.5 <u>Scenario Evaluation</u>

Scenario walkthroughs were performed at the control room mockup for Unit 1 to gain an operational perspective on the proposed layout and to evaluate it from the operator's point of view. The primary purpose of this methodology was to assess the logical sequencing of all actions and the flow of required motions to perform the required actions in a timely manner.

This evaluation was performed in three phases. The first phase was the creation of operating procedures from Westinghouse generic procedures. The second phase was the actual walkthrough. The third phase was the scenario evaluation.

Nine scenarios were selected from Westinghouse generic procedures. Four nonprocedural scenarios were added during the review. The nine selected were chosen because they represent a wide variety of plant activities and contain the basic parameters for many more operating tasks. The four shorter scenarios were added to provide a review of observational tasks. The nine scenarios were:

- Immediate action and diagnostics.
- Steam generator tube rupture.
- Operation with natural circulation.
- Plant shutdown from minimum load to cold shutdown.
- Reactor coolant system leak.
- Station blackout.
- Plant startup from cold shutdown to minimum load.
- Loss of reactor coolant.
- Reactor trip.

The four additional scenarios performed were:

- Loss of secondary coolant.
- Starting the diesel generators.
- Controlling steam generator level.
- Changeover from recirculating mode to injection mode.

The information collected by use of these human factors methodologies served as a basis for the development of preliminary recommendations.

The evaluation of these recommendations is provided in subsection 18.2.2.

#### 18.2.1.6 <u>Reviewer Experience</u>

General Physics Corporation, the prime control board reviewer, has had the following experience in design review:

- North Anna, Unit 2 Control Room Review.
- EPRI-RP769, Performance Measurement System for Training Simulators.
- Babcock & Wilcox and U.S. Department of Energy, Disturbance Analysis and Surveillance System.
- Clinch River Breeder Reactor Control Room Review.
- Edison Electric Institute, Operator Selection Study.
- Operability Assessment of Prototype Large Breeder Reactor Designs.
- Electric Power Research Institute, Survey and Analysis of Communication Problems in Nuclear Power Plants.

#### 18.2.2 CONTROL ROOM AND CONTROL BOARD EVALUATION RESULTS

#### 18.2.2.1 <u>Human Factors Deficiencies Categories</u>

The human factor deficiencies were categorized into five general groupings:

- A. Administrative, where no hardware solutions were available.
- B. Components, where the type of display, means and methods of manipulation, and elimination of the chances for inadvertant or accidental misoperation were considered.
- C. Labeling, where the identifications of systems, trains, and devices were reviewed so that quick and easy identification by the operators would be enhanced.
- D. Rearrangement of components, where component functional relationships were considered and their actual arrangement on the board was then critiqued.
- E. Workshop/environment, where noise, temperature, lighting, and workspace definition were considered which would lead to increased operation effectiveness and reduced distraction; other deficiencies which would negatively affect operator performance.

#### 18.2.2.2 Findings and the Resolutions Thereof

The preliminary control room design review identified human engineering deficiencies at an early stage in the design in order that they could be incorporated in design improvements, operator training, and plant operating and administrative procedures. When final NRC guidance was issued, the review process was repeated in the detailed control room design review, detailed in section 18.3.

#### 18.2.2.3 <u>References</u>

- 1. General Physics Corporation, "Report on Human Factors Evaluation of Alvin W. Vogtle Nuclear Power Plant Control Room," GP-R-23003, Columbia, Maryland.
- 2. U.S. Nuclear Regulatory Commission, "Human Engineering Guide to Control Room Evaluation - Draft Report," NUREG/CR-1580, July 1980.
- 3. U.S. Department of Defense, "Human Engineering Design Criteria for Military Systems, Equipment, and Facilities," MIL-STD-1472B, Washington, D.C., May 1978.
- 4. U.S. Nuclear Regulatory Commission, "Safety Evaluation Report, Supplement No. 1, Docket Nos. 50-327 and 50-328, Sequoyah Nuclear Plant Units 1 and 2," NUREG-0011, February 1980.
- 5. U.S. Nuclear Regulatory Commission, "Safety Evaluation Report, Supplement No. 10, Docket No. 50-339, North Anna Power Station Unit 2," NUREG-0053, July 1980.
- U.S. Nuclear Regulatory Commission, "Safety Evaluation Report, Supplement No. 4, Docket No. 50-311, Salem Nuclear Generating Station Unit 2, NUREG-0512, January 1979.

# 18.3 DETAILED CONTROL ROOM DESIGN REVIEW

# 18.3.1 BACKGROUND

Guidance for the control room design review (CRDR) was provided by various NUREGs and Regulatory Guides. A Nuclear Utility Task Action Committee (NUTAC) with staff support from the Institute of Nuclear Power Operations (INPO) developed a generic control room design review implementation plan from these guidelines. NRC guidance and NUTAC guidelines were used by Georgia Power Company (GPC) for the development of the detailed CRDR process.

The detailed control room design review stands alone in documenting compliance with NRC requirements. Many design changes recommended in the preliminary review were already incorporated, resulting in fewer design deficiencies when reviewed by NUREG 0700 guidance. However, some items identified in the preliminary review had not been incorporated into the design. To assure problems were not overlooked, those items were incorporated in the review as human engineering discrepancies (HEDs) for review and resolution.

### 18.3.2 OBJECTIVES OF THE CRDR

The objective of the CRDR was to determine the extent to which the VEGP control room provides the operators with sufficient information to complete their required functions and task responsibilities efficiently under emergency conditions. The review also determined the human engineering suitability of the designs of the instrumentation and equipment in the VEGP control room.

To ensure that the CRDR fulfilled its stated purpose, several specific objectives were identified and met during the review. The following specific objectives were defined for the CRDR:

- To perform a control room survey that compares the existing control room with accepted human engineering criteria.
- To review relevant plant operational experience using appropriate documentation and operator questionnaires.
- To determine the information and control requirements of control room operator tasks during emergency conditions.
- To identify human engineering discrepancies.
- To evaluate the extent and importance of identified discrepancies.
- To formulate and implement solutions for significant discrepancies.
- To ensure that the proposed solutions do, in fact, eliminate or mitigate the discrepancies for which they are formulated without creating new discrepancies.
- To verify that implemented solutions eliminate or mitigate identified discrepancies.

# 18.3.3 CONTROL ROOM DESIGN REVIEW PROCESS

This section describes the process that was used to accomplish the objectives of the CRDR.

#### 18.3.3.1 Preliminary CRDR Status Evaluation

Recognizing that the 1982 control room design review would provide valuable input to the current control room design review, each of the identified deficiencies/discrepancies was reviewed by the use of the following checklist:

- Action reflects the recommended intent and resolves the HED.
- Action reflects the recommended intent but does not resolve the HED.
- Action does not reflect the recommended intent.
- Actions taken create other HED(s).
- No action taken, deficiency remains.
- No action taken, deficiency resolved by resolution of HED.
- New HED(s) generated.
- Other (explain).

# 18.3.3.2 Operating Experience Review

The Vogtle Electric Generating Plant was under construction with no operating history, and the onsite experience of operational personnel and data from plant operating documents provided little information for the CRDR. Accordingly, the operating experience review focused primarily on industry experience at similar plants and considered the experience gained from the VEGP plant-specific simulator.

Two separate steps were involved in reviewing operating experience. The first was to review available and applicable historical documentation pertaining to plant-specific and generic occurrences. The second step was to survey operating personnel. Operating personnel surveys (operator questionnaire) identified specific problem areas in the VEGP control room that may occur during normal and emergency operation. The operator questionnaire was extracted from the Control Room Design Review Survey Development Guideline (INPO 83-014).

#### 18.3.3.3 Problem Reports

An open ended control room design problem form was sent to all control room and simulator personnel. Copies were made available in the control room and the simulator. This allowed any problem noted to be promptly reported. These problem reports were evaluated for possible HEDs by the review team leader. In many cases, the problem reports identified items already documented by surveys or prior problem reports.

## 18.3.3.4 <u>Control Room Surveys</u>

Surveys of the existing VEGP control room were conducted during the CRDR. The purpose of the surveys was to compare the design features of the existing control room with applicable human engineering design guidelines. The surveys were conducted by the CRDR review team. The survey team used questionnaires, checklists, and surveys to compile information regarding the as-built characteristics of the VEGP control room.

Eleven separate surveys were completed during the CRDR survey activity. Some of the surveys consisted simply of recording (or determining) control room conventions, such as color usage and instrument arrangement. In general, HEDs were not written during the convention surveys. Instead, the information obtained was used in other CRDR activities to determine where particular instruments or systems departed from the overall convention.

Other surveys measured certain physical quantities, such as illumination and sound level, and compared these measurements to acceptable, or preferred, human engineering standards for such quantities. HEDs were written for characteristics that fell outside the acceptable band.

The individual surveys were:

- General design convention survey (NUTAC 83-042).
- Design convention survey for repetitive groupings (NUTAC 83-042).
- Lighting survey (NUREG 0700 Section 6.1.5).
- Noise survey (NUREG 0700 Appendix E).
- Anthropometric survey (NUTAC 83-042 and NUREG 0700 Section 6.1.2).
- Annunciator survey (NUREG 0700 Section 6.3).
- Communication survey (NUREG 0700 Section 6.2).
- Abbreviation and acronym survey (NUTAC 83-042).
- Color coding survey (NUTAC 83-042).
- Control room computer survey (NUREG 0700 Section 6.7).
- NUTAC 83-042 appendices B-H survey (using applicable NUREG 0700 guidelines).

#### 18.3.3.5 <u>Task Analysis</u>

The operating experience review and the control room survey identified as HEDs those control room characteristics that had caused, or nearly caused, problems during normal operation and simulator exercises and those characteristics that did not conform to certain human engineering design criteria. The task analysis identified the tasks that operators must perform during emergency operation and determined whether the instrumentation, controls, and equipment were available and suitable to perform those tasks. In addition to determining the availability of suitable instrumentation, controls, and equipment, the task analysis validated that the

emergency tasks identified could be performed in real time under simulated emergency conditions in the VEGP control room.

The task analysis used as its basis the Emergency Response Guidelines (ERGs) developed by the Westinghouse Owners Group (WOG).

#### 18.3.3.6 Instrumentation and Control Characteristics Review

In response to the NRC in-progress audit concerns that the task analysis did not identify instrument and control (I&C) needs as compared to the I&C control room inventory, an independent review of the VEGP control room inventory, instrumentation and control characteristics was conducted. The instrumentation and control characteristics review program identified the instrumentation, controls, and characteristics necessary for proper operator response to emergency transients. This review program first identified generic characteristics based on the Westinghouse Owners Group high-pressure reference plant design, followed by the identification of plant-specific deviations. The characteristics of the installed control room instrumentation were justified with the development of or reference to appropriate generic or plant-specific basis documentation.

### 18.3.3.6.1 Identification of Required Instrumentation and Controls

The Emergency Response Guidelines and the ERG background documents were reviewed to identify:

- All operator tasks necessary to support the operator functions.
- Operator information and control needs necessary to support the operator functions and major actions.
- Plant systems necessary to provide information and control needs.
- Plant instrumentation and controls necessary to provide information and control needs.

For the required plant instrumentation and controls identified above, characteristics were determined based on the information and control needs. The characteristics for the instrumentation included the following:

- Setpoints.
- Units.
- Range.
- Resolution.
- Type of display.

Characteristics for controls included the following:

• Positions.

• Type of control (e.g., variable).

From the information gathered in the characteristics review, a required characteristics justification table was developed for required instrumentation. This table identified operator action categories and associated operator information needs, criteria, and characteristics. The basis for each action category or information need was described or a reference to other documentation was provided.

Following identification of the required characteristics, the VEGP specifics were identified. The VEGP-specific characteristics consisted of applicable generic characteristics and plant-specific deviations. To identify VEGP-specific instrumentation and controls and their associated required characteristics, the Vogtle Emergency Operating Procedures were reviewed to identify deviations from the Emergency Response Guidelines. These VEGP-specific characteristics were then entered in the required characteristics justification tables.

Generic and VEGP-specific required characteristics were reviewed and the limiting required characteristics were summarized in characteristics requirements tables.

# 18.3.3.6.2 Verification

Verification of the installed control board instrumentation with respect to the above defined required instrumentation and control characteristics was performed. The CRDR team developed an inventory of VEGP control board instrumentation and controls. The CRDR team then compared the inventory to the instrumentation and control characteristics requirements tables for consistency. No HEDs (not already identified in the original task analysis) were identified from this review. This verification provided assurance that the operator did in fact have the required instrumentation and controls assumed in the Westinghouse Owner's Group transient analysis for response to emergency transients.

# 18.3.4 EVALUATION OF HUMAN ENGINEERING DISCREPANCIES

# 18.3.4.1 Objectives of Evaluation Process

The objectives of this phase of the CRDR were:

- Evaluate the significance of the HEDs identified in the previous phases of the CRDR.
- Where HEDs were found to be of minor significance, describe the technical and operational basis for such a finding.
- Where the HEDs were found to be significant, formulate changes to the control room design, procedures, operator training, or any combination thereof to mitigate those HEDs.

# 18.3.4.2 <u>Evaluation Criteria</u>

Human engineering discrepancies found during the control room surveys, the operating experience review, and the task analysis were evaluated by the review team for their potential to adversely affect normal and emergency operation. A categorization scheme was used that

required each HED to be assessed by the review team and prioritized for resolution. The following five categories were designed to be unique so a consensus could be obtained from the review team as to which category each HED should be assigned.

- Category 1 (Safety Significant) HEDs that have caused errors in or are judged likely to adversely affect the management of emergency conditions by control room operators.
- Category 2 HEDs that are known to have caused problems during normal operation.
- Category 3 HEDs that can be fixed with simple and inexpensive enhancements, so called "paint, tape, and label" (PTL) fixes. This included HEDs that were easy to fix but difficult to assess as to the effect on emergency operation.
- Category 4 These HEDs were judged by the review team as unlikely to affect emergency operation, not documented as causing problems during normal operation, and not easily corrected. However, corrective action was recommended.
- Category 4a A minor deviation from standards not expected to cause a problem. No corrective action was recommended.

HEDs were initially categorized using a subjective approach that reflected the consensus judgment of the multidisciplinary CRDR team. This HED category review employed a systematic approach based on the assessment process identified in NUREG 0800 and NUREG 0801 (Draft).

HEDs were evaluated with respect to the following attributes and subjected to the algorithm presented in figure 18.3-1 to determine their significance and effects on plant safety.

- Item 1 HEDs experienced (EOP validation) or assessed (surveys or checklists) as having a high probability of contributing to operator error.
- Item 2 HEDs associated with engineered safety features (ESF) systems.
- Item 3 HEDs that could result in unsafe operation or violation of the Technical Specifications.
- Item 4 HEDs identified through the operating experience review or actual problems identified in the operator questionnaire.
- Item 5 HEDs determined to be easily correctable with paint, tape, labels, engraving changes, or work space environment improvements.
- Item 6 HEDs determined to contribute to the operator mental work load (cumulative effects) resulting in fatigue, confusion, or discomfort.

#### 18.3.4.3 <u>Resolution of Human Engineering Discrepancies</u>

#### 18.3.4.3.1 Approach to Correction

The correction of human engineering discrepancies was generally based on the following preferred order of approaches:

- A. Guideline compliance The first preference for correction of HEDs was to modify the control room to comply with the guidelines of NUREG 0700. This approach was used for labeling, procedural, and support equipment HEDs. Control board arrangement HEDs used this approach when consistent with regulatory (train separation) and physical (panel space) limitations.
- B. Compensatory measures When conflicting regulatory or physical constraints prevented a straightforward change to achieve guideline compliance, changes were used which eliminated or reduced the impact of an HED on the operators.

### 18.3.4.3.2 Engineering Consensus

The development and selection of corrections to HEDs was accomplished by group meetings of the detailed CRDR team. Alternate approaches, benefits, and costs were discussed, and an engineering consensus was arrived at on the recommended approach to correct each HED. The diverse backgrounds of team members were intended to provide input from all disciplines on the resolution of HEDs. The procedures in use called for final recording of the board recommendations. Brief summaries of the resolutions and supporting comments were reported in the Detailed CRDR Report of June 10, 1986.

### 18.3.4.3.3 Cost-Benefit Analysis

The engineering consensus approach considered costs vs. benefits but did not perform a detailed, documented cost-benefit analysis. The emphasis was on correction of the HEDs to enhance operator performance. The correction of safety significant HEDs was a commitment without regard to cost. In all cases the team members sought to develop the most practical solution balancing constructability, schedule, and cost to achieve the objective of enhanced operator performance.

Some example cost considerations were:

- A. Labeling and procedure changes were always implemented to achieve compliance with NUREG 0700. These were typically less than \$10,000-projects.
- B. Panel rearrangements generally less than \$100,000 were implemented.
- C. A major control room layout modification of \$300,000 was implemented.
- D. Major control panel replacements which would cost several million dollars were not recommended. HEDs involving such changes for exact compliance with NUREG 0700 guidelines were addressed with alternate solutions.

#### 18.3.5 COORDINATION WITH OTHER ACTIVITIES

The CRDR process was coordinated with other post-TMI activities in several ways. These activities included the following:

18.3-7

- Upgrading Emergency Operating Procedures.
- Detailed control room design review.
- Post-accident monitoring system (section 7.5).

- Safety parameter display system (subsection 9.5.10).
- Upgrading emergency response facilities.

The mechanism for coordination of control room improvements with other programs was the Emergency Operating Procedures validation program.

The coordination was achieved by: (1) having several members of the detailed CRDR team serve as members of the EOP observation teams during the EOP validation exercises, (2) the task analysis was a common basis for both the detailed CRDR and EOP programs, and (3) the findings or discrepancies identified during the validation exercises related to man and machine interface were processed via the detailed CRDR HED assessment process.

In addition, control room modifications that resulted from the Emergency Operating Procedures validation program were addressed in the detailed CRDR verification plan to ensure design improvements provided the necessary correction.

### 18.3.5.1 Emergency Operating Procedures

The task analysis portion of the detailed CRDR used the Westinghouse ERGs as plant-specific EOPs as its starting point. Thus, the task of upgrading emergency procedures is inherently integrated into the detailed CRDR. The simulator validation of the task analysis was combined with the VEGP EOP validation program.

### 18.3.5.2 Safety Parameter Display System (SPDS)

GPC used the SPDS during the EOP validation exercise, and evaluated it to NUREG 0700 Section 6.7 (process computer) guidelines. Some HEDs identified during the EOP validation exercise or the computer survey and judged to be significant by the review team were resolved by incorporating certain features into the SPDS and associated displays. In addition, the detailed CRDR team leader was chairman of a control room computer task force and participated in developing displays and board arrangements. This served to incorporate human engineering requirements into the design of the SPDS and further integrate them into the detailed CRDR process. (See the Emergency Plan, section H.4.5.)

# 18.3.5.3 Regulatory Guide 1.97

The design of Regulatory Guide 1.97 instrumentation was essentially complete at the time of this review. This instrumentation was evaluated in the detailed CRDR survey and task analysis.



# **Algorithm Attributes**

# ITEM 1

HEDs experienced or assessed as having a high probability of contributing to operator error.

# ITEM 3

HEDs that could result in unsafe operation or violation of the Technical Specifications.

# ITEM 5

HEDs determined to be easily correctable with paint, tape, labels, engraving changes, or work space environment improvements.

# ITEM 2

HEDs associated with engineered safety features systems.

# **ITEM 4**

HEDs identified through the operation experience review or actual problems identified in the operator questionnaire.

# ITEM 6

HEDs determined to contribute to operator mental work load (cumulative effects) resulting in fatigue, confustion, or discomfort.

# REV 14 10/07



VOGTLE ELECTRIC GENERATING PLANT UNIT 1 AND UNIT 2 HUMAN ENGINEERING DISCREPANCY EVALUATION ALGORITHM

FIGURE 18.3-1 (SHEET 2 OF 2)

### 19.0 LICENSE RENEWAL – AGING MANAGEMENT PROGRAMS AND ACTIVITIES

# 19.1 INTRODUCTION

#### 19.1.1 BACKGROUND

Renewed operating licenses for Vogtle Electric Generating Plant (VEGP) Units 1 and 2 were issued on June 3, 2009, extending the original licensed operating term by 20 years. Units 1 and 2 will enter the period of extended operation on January 17, 2027 and February 10, 2029 for Units 1 and 2, respectively.

#### 19.1.1.1 License Renewal Rule and Process

10 CFR Part 54, the license renewal rule, establishes the procedures, criteria, and standards governing nuclear plant license renewal.

Plant systems, structures, and components (SSCs) within the scope of license renewal are defined in 10 CFR 54.4(a) as:

- Safety-related SSCs (i.e., perform a safety-related function as defined in 10 CFR 54.4(a) (1)).
- Nonsafety-related SSCs whose failure could prevent satisfactory accomplishment of safety-related functions.
- All SSCs relied on in safety analyses or plant evaluations to perform a function that demonstrates compliance with the Commission's regulations for fire protection (10 CFR 50.48), environmental qualification (10 CFR 50.49), pressurized thermal shock (10 CFR 50.61), anticipated transients without scram (10 CFR 50.62), and station blackout (10 CFR 50.63).

The license renewal rule focuses on managing the effects of aging on the passive intended functions of long-lived structures and components, and on evaluation of time-limited aging analyses (TLAA), as defined in 10 CFR 54.21. (See paragraph 19.1.1.3 for a discussion of the definition of a TLAA.)

The license renewal rule generically excludes structures and components associated only with active functions from an aging management review. Functional degradation resulting from the effects of aging on active functions is more readily determinable and detectable, and existing programs and regulatory requirements are expected to directly detect the effects of aging. The license renewal rule credits the continued applicability of existing programs and regulatory requirements, and the maintenance rule requirements (10 CFR 50.65), to monitor the performance and condition of SSCs that perform active functions.

The license renewal process includes the identification of SSCs within the scope of the license renewal rule, determining the in-scope structures and components subject to aging management review (i.e., are passive and long-lived), and assuring the effects of aging on the intended functions are adequately managed through the identification and/or development of

various aging management programs and activities. The process also includes the identification and evaluation of TLAAs, including any exemptions containing TLAAs.

The license renewal rule and the renewed operating licenses require that a summary description of the aging management programs and activities and the TLAA evaluations become part of the FSAR. To meet this requirement, sections 19.2 through 19.4 are incorporated into the FSAR. After issuance of the renewed license, 10 CFR 54.37(b) requires that, for newly identified SSCs that would have been subject to aging management review or evaluation of TLAAs in accordance with 10 CFR 54.21, the FSAR be updated to describe how the effects of aging will be managed such that the intended functions(s) in 10 CFR 54.4(b) will be effectively maintained during the period of extended operation.

# 19.1.1.2 Aging Management Programs

The NRC, in the Standard Review Plan for License Renewal (NUREG-1800), Appendix A.1, "Aging Management Review – Generic (Branch Technical Position RLSB-1)," describes the elements of an acceptable aging management program to the NRC staff. Additionally, NUREG-1801, "Generic Aging Lessons Learned Report," describes aging management programs that have been found acceptable to the NRC Staff to manage the aging effects of SSCs for license renewal.

In many cases, programs and activities existing at the time of the license renewal application were found adequate for managing aging for the period of extended operation. In some cases, the existing programs or activities required some degree of enhancement. Also, some new programs and activities were identified. It is important to note that only a portion of certain programs or activities may be relied upon for managing the effects of aging under the license renewal rule.

More than one program or activity may be credited to manage aging in a single system, structure, or component. Conversely, in other cases, one program or activity may manage the effects of aging in multiple systems.

# 19.1.1.3 <u>Time-Limited Aging Analyses</u>

The license renewal rule requires that TLAA be evaluated to capture certain plant-specific aging analyses explicitly based on the original 40-year operating life of the plant. In addition, the Rule requires that any exemptions based on TLAAs be identified and analyzed to justify extension of those exemptions through the renewal term.

TLAA evaluations are defined by the license renewal rule in 10 CFR 54.3 as those calculations and analyses that meet all of the following six criteria:

- Involve SSCs within the scope of license renewal.
- Consider the effects of aging.
- Involve time-limited assumptions defined by the operating term, e.g., 40 years.
- Were determined to be relevant in making a safety determination.
- Involve conclusions or provide the bases for conclusions related to the capability of the SSC to perform its intended functions, as delineated in the Rule.

• Are contained or incorporated by reference in the current licensing basis.

Once a TLAA has been identified, the Rule in 10 CFR 54.21 (c) requires it to be dispositioned by one of the following three specific criteria:

- The analyses remain valid for the period of extended operation.
- The analyses have been acceptably projected to the end of the period of extended operation.
- The effects of aging on the intended functions(s) will be adequately managed (e.g., programs or activities are in place) for the period of extended operation.

After the renewed license has been issued, 10 CFR 54.37 (b) requires that any newly identified calculations or analyses that would have been a TLAA be evaluated and a summary description placed in the FSAR.

# **19.1.2 AGING MANAGEMENT PROGRAMS**

The following programs are credited to manage the effects of aging during the period of extended operation for license renewal and are described in section 19.2 as listed below:

- ACCW System Carbon Steel Components Program (19.2.1).
- Bolting Integrity Program (19.2.2).
- Boric Acid Corrosion Control Program (19.2.3).
- Buried Piping and Tanks Inspection Program (19.2.4).
- CASS RCS Fitting Evaluation Program (19.2.5).
- Closed Cooling Water Program (19.2.6).
- Diesel Fuel Oil Program (19.2.7).
- External Surfaces Monitoring Program (19.2.8).
- Fire Protection Program (19.2.9).
- Flow-Accelerated Corrosion Program (19.2.10).
- Flux Thimble Tube Inspection Program (19.2.11).
- Generic Letter 89-13 Program (19.2.12).
- Inservice Inspection Program (19.2.13).
- Nickel Alloy Management Program for Non-Reactor Vessel Closure Head Penetration Locations (19.2.14).

- Nickel Alloy Management Program for Reactor Vessel Closure Head Penetrations (19.2.15).
- Oil Analysis Program (19.2.16).
- One-Time Inspection Program (19.2.17).
- One-Time Inspection Program for ASME Class 1 Small Bore Piping (19.2.18).
- One-Time Inspection Program for Selective Leaching (19.2.19).
- Overhead and Refueling Crane Inspection Program (19.2.20).
- Periodic Surveillance and Preventive Maintenance Activities (19.2.21).
- Piping and Duct Internal Inspection Program (19.2.22).
- Reactor Vessel Closure Head Stud Program (19.2.23).
- Reactor Vessel Internals Program (19.2.24).
- Reactor Vessel Surveillance Program (19.2.25).
- Steam Generator Tubing Integrity Program (19.2.26).
- Steam Generator Program for Upper Internals (19.2.27).
- Water Chemistry Control Program (19.2.28).
- 10 CFR 50 Appendix J Program (19.2.29).
- Inservice Inspection Program IWE (19.2.30).
- Inservice Inspection Program IWL (19.2.31).
- Structural Monitoring Program (19.2.32).
- Structural Monitoring Program Masonry Walls (19.2.33).
- Non-EQ Cables and Connections Program (19.2.34).
- Non-EQ Inaccessible Medium-Voltage Cables Program (19.2.35).
- Non-EQ Electrical Cable Connections One-Time Inspection Program (19.2.36).

#### 19.1.3 AGING MANAGEMENT PROGRAMS – TIME LIMITED AGING ANALYSES (TLAA)

The aging management programs credited for managing the associated TLAAs during the period of extended operation are described in section 19.3 as listed below:

- Environmental Qualification Program (19.3.1).
- Fatigue Monitoring Program (19.3.2).

# 19.1.4TLAA EVALUATIONS

The evaluation of TLAAs for the period of extended operation is provided in section 19.4. The TLAAs evaluated for the period of extended operation are listed below:

- Reactor Vessel Neutron Embrittlement Analyses (19.4.1).
- Metal Fatigue Analysis (19.4.2).
- Environmental Qualification Calculations (19.4.3).
- Containment Tendon Pre-Stress Analysis (19.4.4).
- Penetration Load Cycles (19.4.5).
- Other Plant Specific Analysis (19.4.6).

## 19.1.5 REFERENCES

- NUREG-1800, Standard Review Plan for Review of License Renewal Applications for Nuclear Power Plants, U.S. Nuclear Regulatory Commission (Rev. 1), September 2005.
- 2. NUREG-1801, Generic Aging Lessons Learned (GALL) Report, U.S. Nuclear Regulatory Commission, (Rev. 1), September 2005.
- 3. Vogtle Electric Generating Plant Technical Specifications, Units 1 and 2.

# 19.2 AGING MANAGEMENT PROGRAM DESCRIPTIONS

The VEGP integrated plant assessment for license renewal identified the aging management programs credited to provide reasonable assurance that structures and components requiring an aging management review will continue to perform their intended functions consistent with the current licensing basis through the period of extended operation. This section describes the aging management programs and activities required to manage the effects of aging during the period of extended operation.

The aging management programs and activities in this section rely on the operations quality assurance program (OQAP) for VEGP and SNC for the elements of corrective action, confirmation process, and administrative controls. The VEGP quality assurance (QA) procedures, review and approval processes, and administrative controls are implemented in accordance with the requirements of 10 CFR 50, Appendix B. Corrective actions and administrative (document) control for both safety-related and nonsafety-related structures and components are accomplished per the existing corrective action program and document control program and are applicable to all aging management programs and activities that will be required during the period of extended operation. The confirmation process is part of the corrective action program and includes reviews to assure that corrective actions are adequate, tracking and reporting of corrective actions, and reviews of corrective action effectiveness. Any followup inspection required by the confirmation process is documented in accordance with the corrective action program. The corrective action, confirmation process, and administrative controls of the OQAP are applicable to all aging management programs and activities required during the period of extended operation.

# 19.2.1 ACCW SYSTEM CARBON STEEL COMPONENTS PROGRAM

The Auxiliary Component Cooling Water (ACCW) System Carbon Steel Components Program is a plant-specific program that manages cracking of carbon steel components exposed to ACCW through a combination of leakage monitoring and routine and periodic inspections. This includes the Units 1 and 2 ACCW systems, as well as carbon steel components serviced by these ACCW systems. The program is in response to operating experience related to nitrite-induced stress corrosion cracking (SCC) and subsequent component leakage in ACCW system components.

The program relies upon leakage detection monitoring, routine walkdowns, and periodic visual examinations. The program also includes preventive measures applicable to repairs and modifications intended to minimize crack initiation sites, lower stresses, and improve inspectability.

The ACCW System Carbon Steel Components Program will be implemented prior to the period of extended operation.

# 19.2.2 BOLTING INTEGRITY PROGRAM

The Bolting Integrity Program is a plant-specific program that manages cracking, loss of material, and loss of preload in mechanical bolted closures. The Bolting Integrity Program applies to safety-related and nonsafety-related bolting for pressure-retaining components within the scope of license renewal, with the exception of the reactor vessel head studs which are addressed by the Reactor Vessel Head Closure Stud Program.

Preventive aspects of the program include use of appropriate bolting and torquing practices, including control of thread lubricants. Periodic replacement of steam generator manway and handhole bolting is also included in the scope of the program as a preventive measure for managing cumulative fatigue damage for these fasteners. The program's bolting and torquing practices are based on industry guidelines, vendor recommendations, and VEGP operating experience, as appropriate for VEGP applications. Consistent with NUREG-1339 recommendations, the use of lubricants containing molybdenum disulfide, which has been specifically implicated in SCC of bolting, is prohibited by the program.

The program also includes periodic inspection of closure bolting assemblies to detect signs of leakage that may be indicative of loss of preload, loss of material, or cracking. Periodic inspection of bolted closures in conjunction with the Inservice Inspection Program and External Surfaces Monitoring Program will detect the effects of aging and joint leakage. Operator rounds and system walkdowns also identify joint leakage.

# 19.2.3 BORIC ACID CORROSION CONTROL PROGRAM

The Boric Acid Corrosion Control Program monitors the condition of components on which borated water may leak to ensure that borated water leakage and associated boric acid residue are identified, evaluated, and removed before any loss of intended function of affected components. The program detects boric acid leakage by periodic visual inspection of systems containing borated water for evidence of leakage and by inspection of adjacent structures and components for evidence of leakage. The program was developed in response to the recommendations of Generic Letter 88-05 and addresses operating experience contained in recent NRC generic communications.

Prior to the period of extended operation, VEGP will enhance the Boric Acid Corrosion Control Program to address the effects of borated water leakage on materials other than steels, including electrical components (e.g., electrical connectors), that are susceptible to boric acid corrosion.

# 19.2.4 BURIED PIPING AND TANKS INSPECTION PROGRAM

The Buried Piping and Tanks Inspection Program manages loss of material from the external surfaces of buried carbon steel, cast iron, and stainless steel components. The program includes both preventive measures and visual inspections. Preventive measures consist of coatings and wrappings which are required by design in accordance with industry standards. Buried components in the scope of license renewal will be inspected when they are excavated for maintenance or when exposed for any other reason.

Prior to entering the period of extended operation, a review will be performed to determine if at least one opportunistic or focused inspection of buried piping and tanks has been performed within the 10-year period prior to the period of extended operation. If an inspection did not occur, a focused inspection will be performed prior to the period of extended operation.

In addition, a focused inspection of buried piping and tanks will be performed within 10 years after entering the period of extended operation, unless an engineering evaluation concludes that sufficient opportunistic and focused inspections have occurred during this time to demonstrate the ability of the underground coatings to protect the underground piping and tanks from degradation.

The Buried Piping and Tanks Inspection Program will be implemented prior to the period of extended operation.

# 19.2.5 CASS RCS FITTING EVALUATION PROGRAM

The Cast Austenitic Stainless Steel (CASS) Reactor Coolant System (RCS) Fitting Evaluation Program manages the effects of loss of fracture toughness due to thermal aging for susceptible CASS components in the RCS. This program augments VEGP Inservice Inspection Program requirements.

This aging management program evaluates the susceptibility of CASS components to thermal aging embrittlement based on casting method, molybdenum content, and percent ferrite. Screening for susceptibility to thermal aging is not required for pump casings and valve bodies, based on the assessment documented in the letter dated May 19, 2000, from Christopher Grimes, Nuclear Regulatory Commission (NRC), to Douglas Walters, Nuclear Energy Institute (NEI), ADAMS Accession No. ML003717179. The existing ASME Section XI inspection requirements, including the alternative requirements of ASME Code Case N-481 for pump casings, are adequate for all pump casings and valve bodies.

The program provides aging management through either a flaw tolerance evaluation or enhanced volumetric examination. Additional inspection or evaluations to demonstrate that the material has adequate fracture toughness are not required for components that are not susceptible to thermal aging embrittlement.

Based on screening consistent with the process specified in NUREG-1801, Rev. 1, Section XI.M12, the VEGP components that require additional aging management under this program are the VEGP Unit 1 Loop 4 and Unit 2 Loop 1 RCP inlet elbows. For these two casings, loss of fracture toughness due to thermal aging will be managed by component-specific flaw tolerance evaluation, additional inspections, or a combination of these techniques.

The CASS RCS Fitting Evaluation Program will be implemented prior to the period of extended operation.

# 19.2.6 CLOSED COOLING WATER PROGRAM

The Closed Cooling Water (CCW) Program manages loss of material, cracking, and reduction of heat transfer in closed-cycle cooling water systems and the components cooled by these systems. The program is based on the EPRI CCW chemistry guidelines.

The program includes maintenance of corrosion inhibitor, pH buffering agent, and biocide concentrations. Concentrations of detrimental ionic species are monitored and reduced if necessary. Important diagnostic parameters are monitored and evaluated for significant trends. The program also uses corrosion-monitoring activities including trending of iron and copper concentrations and component inspections. Corrosion rate monitoring methods may also be used.

Prior to the period of extended operation, VEGP will enhance the CCW Program to indicate the components in each system that are most susceptible to various corrosion mechanisms and to ensure that corrosion monitoring is appropriately accomplished. This qualitative assessment will be based on an understanding of corrosion principles associated with CCW chemistries and on review of system, plant, and industry operating experience. Parameters considered in the review will include system flow parameters (focusing on identification of stagnant regions and on

intermittently operated systems), normal operating temperatures, and component geometries (e.g., creviced areas).

# 19.2.7 DIESEL FUEL OIL PROGRAM

The Diesel Fuel Oil Program is a plant-specific program that manages loss of material in the diesel fuel oil systems for the emergency diesel generators and the diesel engine-driven fire water pumps through monitoring and maintenance of diesel fuel oil quality. The program is based on VEGP Technical Specifications requirements and supplemental requirements. Draining, cleaning, and internal condition inspections of diesel fuel oil components are implemented under other VEGP aging management programs as noted below.

- Periodic cleaning and inspection of the interior of the EDG system's diesel fuel oil storage tanks is performed under the Periodic Surveillance and Preventive Maintenance Program.
- Visual inspection of the diesel engine-driven fire water pumps fuel supply lines for leakage during diesel operation is performed under the Fire Protection Program.
- The One-Time Inspection Program describes inspections to verify the effectiveness of the Diesel Fuel Oil Program. The inspections include thickness measurements of storage tank bottom surfaces to verify that significant degradation of the tank base material is not occurring.

# 19.2.8 EXTERNAL SURFACES MONITORING PROGRAM

The External Surfaces Monitoring Program inspects external surfaces of mechanical system components requiring aging management for license renewal in external air environments. Surfaces constructed from materials susceptible to aging in these environments are inspected at frequencies that assure the effects of aging are managed such that system components will perform their intended function during the period of extended operation.

The program detects corrosion, flange leakage, missing or damaged insulation, damaged coatings, and indications of fretting or wear. Inspections of insulated surfaces are performed on a sampling basis, targeting areas identified by baseline inspections and operating experience as most susceptible. Accessible polymers and elastomers are also inspected.

Systems and components which are normally inaccessible and therefore not readily available for inspection are inspected when they are made accessible during outages or routine maintenance or repair or they may be inspected by remote means.

The External Surfaces Monitoring Program will be implemented prior to the period of extended operation.

SNC will perform an inspection of an emergency diesel generator fuel oil day tank vent line. This inspection will determine whether a debris screen is installed on the open end of the vent line. If a screen is installed, the inspection will further determine the material of construction of the debris screen.

## 19.2.9 FIRE PROTECTION PROGRAM

The Fire Protection Program includes inspections, performance testing, and condition monitoring of water- and gas-based fire protection systems, fire barriers, and fire pump diesels and their fuel oil supply components. The program manages fire protection components relied upon for 10 CFR 50.48 compliance such that the intended functions will be maintained through the period of extended operation.

The water-based and gas-based fire suppression systems are tested and inspected in accordance with plant procedures based, in part, on the applicable National Fire Protection Association codes and standards. Periodic inspections, performance testing, and system monitoring provide an effective means to assure functionality of these components.

Diesel-driven fire pumps and fuel oil supply components are periodically inspected and tested to ensure that the diesels, pumps, and fuel oil supply components can perform their intended functions.

The fire barrier inspections include periodic visual inspection of structural fire barriers, including fire walls, floors, ceilings, fire penetration seals, and fire doors.

VEGP will implement the following enhancements to the Fire Protection Program:

- Wall thickness evaluations will be performed on water suppression piping systems using nonintrusive volumetric testing or visual inspections to ensure that wall thicknesses are within acceptable limits. Initial wall thickness evaluations will be performed before the end of the current operating term. Subsequent evaluations will be performed at plant-specific intervals during the period of extended operation. The plant-specific inspection intervals will be determined based on previous evaluations and site operating experience.
- A sample of sprinkler heads will be inspected using the guidance of NFPA 25 "Inspection, Testing and Maintenance of Water-Based Fire Protection Systems" (1998 Edition), Section 2-3.1.1 or NFPA 25 (2002 Edition), Section 5.3.1.1.1. Where sprinkler heads have been in service for 50 years, they will be replaced or representative samples from one or more sample areas will be submitted to a recognized testing laboratory for field service testing. This sampling will be performed every 10 years after the initial field service testing. The 50 years of time in service begins when the system was placed in service, not when the plant became operational.
- Prior to the period of extended operation, Fire Protection Program procedures will be revised to provide more detailed instructions for visual inspection of fire pump diesel fuel supply lines for leakage, corrosion, and general degradation while the engine is running during fire suppression system pump tests.

#### 19.2.10 FLOW-ACCELERATED CORROSION PROGRAM

The Flow-Accelerated Corrosion (FAC) Program manages loss of material (wall thinning) due to FAC in susceptible plant piping and other components. The FAC Program is based on the guidance of NSAC-202L-R2, "Recommendations for an Effective Flow-Accelerated Corrosion Program," including subsequent revisions. The program includes analysis to determine susceptible locations, predictive modeling techniques, baseline inspections of wall thickness, followup inspections, and repair or replacement of degraded components as necessary.

VEGP also uses the FAC Program and its inspection techniques to manage wall thinning that is occurring in piping components downstream of the steam generator blowdown demineralizers. The wall thinning has been attributed to the acidic conditions of the demineralizer effluent, not FAC.

# 19.2.11 FLUX THIMBLE TUBE INSPECTION PROGRAM

The Flux Thimble Tube Inspection Program manages loss of material due to fretting/wear of the incore flux detector thimble tubes. The program implements the VEGP response to NRC Bulletin No. 88-09, "Thimble Tube Thinning in Westinghouse Reactors." The program uses proven nondestructive examination techniques to monitor for wear of the flux thimble tubes.

Wear rate predictions determine the need for corrective actions such as repositioning, capping, or replacement of a flux thimble tube. The wear-rate predictions are also used to establish the interval to the next inspection.

Prior to the period of extended operation, a VEGP program procedure will be issued documenting the Flux Thimble Tube Inspection Program administration and implementing activities credited for license renewal.

# 19.2.12 GENERIC LETTER 89-13 PROGRAM

The Generic Letter 89-13 Program includes the activities which implement the VEGP response to the NRC recommended actions contained in Generic Letter (GL) 89-13, "Service Water System Problems Affecting Safety-Related Equipment." The Generic Letter 89-13 Program activities include mitigation, as well as performance and condition monitoring techniques, to ensure that the effects of aging on the Nuclear Service Cooling Water (NSCW) system, and on those components supplied by the NSCW system will be managed.

Prevention or mitigation of fouling and loss of material in the NSCW system and NSCW supplied components is accomplished, in part, by intermittent injection of appropriate water treatment chemicals. Other preventive and monitoring aspects of the VEGP Generic Letter 89-13 Program include periodic flushing of lines to mitigate or prevent fouling, periodic measurement of flow rates through selected components, periodic analysis of corrosion coupons, and cleaning of selected heat exchangers at regular intervals. Some components are visually inspected for fouling or loss of material. Volumetric examination may be used to detect degradation.

Prior to the period of extended operation, VEGP will implement the following enhancements to the Generic Letter 89-13 Program:

- An overall program procedure will be prepared which describes the various program activities that comprise the Generic Letter 89-13 Program and their implementing controls such as chemistry procedures, maintenance activities, scheduled surveillances, or other mechanisms.
- The VEGP Generic Letter 89-13 Program activities will include inspection of the NSCW transfer pumps' casings and bolting and NSCW cooling tower spray nozzles.

### 19.2.13 INSERVICE INSPECTION PROGRAM

The VEGP Inservice Inspection Program is a plant-specific program that mandates examinations, testing, and inspections of components and systems to detect deterioration and manage aging effects. The program uses periodic visual, surface, and volumetric examination and leakage tests of Class 1, 2, and 3 pressure-retaining components, their integral attachments, and supports to detect and characterize flaws.

The program is implemented in accordance with 10 CFR 50.55(a), which imposes the inservice inspection requirements of ASME Section XI for Class 1, 2, and 3 pressure-retaining components, their integral attachments, and supports. Inspection, repair, and replacement of these components are covered in Subsections IWB, IWC, IWD, and IWF, respectively.

In conformance with 10 CFR 50.55a(g)(4)(ii), and as based on ASME Inservice Inspection Program B (IWA-2432), the VEGP Inservice Inspection Program is updated at the end of each inspection interval to the latest edition and addenda of the Code specified in 10 CFR 50.55a, 12 months before the start of the inspection interval.

#### 19.2.14 NICKEL ALLOY MANAGEMENT PROGRAM FOR NONREACTOR VESSEL CLOSURE HEAD PENETRATION LOCATIONS

The Nickel Alloy Management Program for Nonreactor Vessel Closure Head Penetration Locations is a plant-specific program that manages cracking due to primary water stress corrosion cracking (PWSCC) for nonreactor vessel head nickel alloy component locations. The overall goal of the program is to maintain plant safety and minimize the impact of PWSCC on plant availability through assessment, inspection, mitigation, and repair or replacement of susceptible components. Program development is based on MRP-126, "Generic Guidance for Alloy 600 Management."

The program is based on the following set of implementation commitments:

- 1. SNC will continue to participate in industry initiatives directed at resolving PWSCC issues, such as owners' group programs and the EPRI Materials Reliability Program.
- 2. SNC will comply with applicable NRC orders.
- 3. SNC will submit a program inspection plan for VEGP that includes implementation of applicable NRC bulletins, generic letters, and staff-accepted industry guidance. The inspection plan will be submitted to the staff for review and approval not less than 24 months prior to entering the period of extended operation for VEGP Units 1 and 2. The inspection plan will include assessments of each of the 10 aging management program elements defined in Section A.1.2.3 of NUREG-1800, Revision 1.

Nickel Alloy Management Program for Nonreactor Vessel Closure Head Penetration Locations will be fully implemented prior to the period of extended operation.

#### 19.2.15 NICKEL ALLOY MANAGEMENT PROGRAM FOR REACTOR VESSEL CLOSURE HEAD PENETRATIONS

The Nickel Alloy Management Program for Reactor Vessel Closure Head Penetrations addresses industry concerns regarding the potential for PWSCC in nickel alloy components exposed to the reactor coolant environment. The program is based upon the requirements of NRC First Revised Order EA-03-009, which establishes requirements for susceptibility ranking

and inspections. Susceptibility ranking is based on calculated effective degradation years and the results of previous inspection findings. Inspection frequencies are determined by the susceptibility category. Inspections to detect cracking include bare metal visual examinations and nonvisual techniques.

The program implements commitments for reactor vessel closure head penetrations associated with nickel alloys from NRC orders, bulletins, and generic letters and staff-accepted industry guidelines.

# 19.2.16 OIL ANALYSIS PROGRAM

The VEGP Oil Analysis Program ensures that the lubricating oil and hydraulic fluid environments of in-scope mechanical systems are maintained to the required quality. The Oil Analysis Program maintains lubricating oil and hydraulic fluid system contaminants (primarily water and particulates) within acceptable limits, thereby preserving an environment that is not conducive to deleterious aging effects. Program activities include sampling and analysis of lubricating oil and hydraulic fluid for detrimental contaminants.

The One-Time Inspection Program includes inspections planned to verify the effectiveness of the Oil Analysis Program.

Prior to the period of extended operation, VEGP will implement the following enhancements to the Oil Analysis Program:

- An overall program procedure or guideline will be prepared to formalize the sampling and analysis activities performed.
- Viscosity, relative level of oxidation, and flashpoint of lubricating oil samples will be determined for components where the lubricating oil is changed based on its analyzed condition (instead of being changed on a regular schedule regardless of condition). The relative level of oxidation of the lubricating oil will be monitored by analysis of the neutralization number or other appropriate parameters(s). Flashpoint monitoring will be performed for those components which have the potential for contamination of the lubricating oil with a light hydrocarbon such as fuel oil.
- When a lubricating oil sample's wear metal content screening results exceed the limits established for the wear metal content screening, the lubricating oil from that component will be subjected to additional testing. The additional testing may include detailed particle counting, elemental analysis, or analytical ferrography as necessary to validate the initial screening results and to diagnose the source of the particulates.

#### 19.2.17 ONE-TIME INSPECTION PROGRAM

The VEGP One-Time Inspection Program provides objective evidence that an aging effect is not occurring, or that the aging effect is occurring slowly enough to not affect the component or structure intended function during the period of extended operation, and therefore will not require additional aging management.

The program uses one-time inspections of plant piping and components to verify the effectiveness of aging management programs or to confirm the insignificance of potential aging effects where:

- a. An aging effect is not expected to occur but there is insufficient data to rule it out with reasonable confidence,
- b. An aging effect is expected to progress very slowly in a specified environment, but localized conditions may be more adverse than specified, or
- c. The characteristics of the aging effect include a long incubation period relative to the operating life of the plant.

The inspections will be performed within a window of 10 years immediately preceding the period of extended operation.

The inspections will include a baseline and a followup inspection of the effectiveness of the Boral<sup>™</sup> neutron-absorbing panels credited in the criticality analysis for the Unit 1 spent fuel storage racks to provide reasonable assurance that the panels will continue to perform their reactivity control function during the period of extended operation. The baseline inspection will be performed within a window of 10 years immediately preceding the period of extended operation. The followup inspection will be performed at a date to be determined based on the results of the baseline inspection and relevant industry guidance, not to exceed 10 years after the baseline inspection.

# 19.2.18 ONE-TIME INSPECTION PROGRAM FOR ASME CLASS 1 SMALL BORE PIPING

The VEGP One-Time Inspection Program for ASME Class 1 Small Bore Piping addresses NRC concerns on the potential for cracking of Class 1 piping with a diameter less than NPS 4.

To address SCC concerns, volumetric examination of a sample population of ASME Class 1 Piping butt welds less than NPS 4 will be performed. Examination locations will be selected using a risk-based approach that will consider susceptibility, inspectability, dose, and operating experience.

To address unanticipated thermal fatigue cracking of ASME Class 1 piping less than NPS 4, VEGP will screen and evaluate pipe lines using MRP-146, "Management of Thermal Fatigue in Normally Stagnant Nonisolable Reactor Coolant System Branch Lines," or later updated guidance. Small bore piping inspections will be performed to detect thermal fatigue only at piping locations that fail screening and are not monitored for thermal cycling.

Examinations performed by the program may be incorporated into an NRC-approved Risk-Informed Inservice Inspection Program. The inspections will be performed within a window of 10 years immediately preceding the period of extended operation.

# 19.2.19 ONE-TIME INSPECTION PROGRAM FOR SELECTIVE LEACHING

The VEGP One-Time Inspection Program for Selective Leaching addresses selective leaching in susceptible cast iron and copper alloy components. The program includes a one-time examination of a sample population of components most likely to exhibit selective leaching. Initial examinations will be completed prior to entering the period of extended operation. If degradation due to selective leaching is identified, additional examinations will be performed.

Examination techniques may include hardness measurement (where feasible-based on form and configuration), visual examination, metallurgical evaluation, or other proven techniques determined to be effective in identifying and assessing the extent of selective leaching. The inspections will be performed within a window of 10 years immediately preceding the period of extended operation.

#### 19.2.20 OVERHEAD AND REFUELING CRANE INSPECTION PROGRAM

The VEGP Overhead and Refueling Crane Inspection Program manages the effects of general corrosion and wear of the crane bridge and trolley structural girders and beams and the crane rails and support girders in the scope of license renewal.

The Overhead and Refueling Crane Inspection Program is a condition monitoring program that includes the following nuclear safety-related and quality-related material handling systems: refueling machine, fuel handling machine bridge crane, spent fuel cask bridge crane, and the containment building (reactor) polar crane.

Prior to the period of extended operation, VEGP will enhance applicable plant procedures to explicitly identify inspection of crane rails and crane structural components for loss of material due to corrosion and wear and for indication of rail misalignment.

### 19.2.21 PERIODIC SURVEILLANCE AND PREVENTIVE MAINTENANCE ACTIVITIES

The Periodic Surveillance and Preventive Maintenance Activities is a plant-specific program that includes existing and new periodic inspections and tests that are relied on by license renewal to manage the aging effects applicable to the components included in the program. The Periodic Surveillance and Preventive Maintenance Activities Program is generally implemented through repetitive tasks and surveillances.

Inspection and testing intervals are dependent on the component, material, and environment and take into consideration industry and plant-specific operating experience and manufacturer's recommendations.

The extent and schedule of inspections and testing assure detection of component degradation prior to loss of intended functions. Established techniques such as visual inspections are used. The following existing surveillance and maintenance activities are credited for license renewal:

- Control building control room filter unit seal inspections.
- Emergency diesel generator (EDG) diesel fuel oil storage tank cleaning and inspections.
- Steam generator blowdown trim heat exchanger inspections.
- NSCW cooling tower fill and drift eliminator testing.
- Diaphragm inspections for the boric acid storage tank, condensate storage tank, and reactor makeup water storage tank.

Prior to the period of extended operation, VEGP will enhance the Periodic Surveillance and Preventive Maintenance Activities to include the following additional surveillance and maintenance activities:

• Steam generator blowdown secondary sample bath shell inspections.

- Steam generator blowdown corrosion product monitor cooler shell inspections.
- Potable water system water heater housing inspections (for the in-scope water heaters).

#### 19.2.22 PIPING AND DUCT INTERNAL INSPECTION PROGRAM

The VEGP Piping and Duct Internal Inspection Program manages corrosion of steel, stainless steel, and copper alloy components and degradation of elastomer components due to changes in material properties. Inspections are normally performed concurrent with scheduled preventive maintenance, surveillance testing, and corrective maintenance activities. Specific examinations not coordinated with existing work activities may also be performed at the discretion of the program owner. Inspection locations and intervals are dependent on assessments of the likelihood of significant degradation and on current industry and plant-specific operating experiences.

Examination techniques will be appropriate to detect and assess the aging mechanism of concern and may include visual examination, nonvisual NDE such as ultrasonic testing or radiography, physical manipulation of elastomers, etc.

The Piping and Duct Internal Inspection Program will be implemented prior to the period of extended operation.

### 19.2.23 REACTOR VESSEL CLOSURE HEAD STUD PROGRAM

The VEGP Reactor Vessel Closure Head Stud Program provides direction for loss of material and cracking in the reactor vessel closure head studs, nuts, and washers. Program aspects include preventive measures, as described in Regulatory Guide 1.65, and condition monitoring.

Preventive measures include material controls and the use of approved lubricants. The VEGP reactor vessel head studs are fabricated from modified SA-540 Grade B24 material as specified in ASME Boiler and Pressure Vessel Code case 1605. This Code case is not specified in Regulatory Guide 1.65 but has been approved by the NRC via Regulatory Guide 1.85. VEGP actual stud material properties have ultimate tensile strengths less than 170 ksi. Reactor vessel closure head studs and nuts are lubricated with an approved, stable lubricant at each reassembly.

Condition monitoring includes examination and leakage detection consistent with the VEGP Inservice Inspection Program.

# 19.2.24 REACTOR VESSEL INTERNALS PROGRAM

The Reactor Vessel Internals Program is a plant-specific program that addresses material degradation issues for the VEGP reactor vessel internals.

The program will be based on the following set of implementation commitments:

- a. SNC will participate in the industry program for investigating and managing aging effects on reactor vessel internals.
- b. SNC will evaluate and implement the results of the industry programs, such as the EPRI Material Reliability Program (MRP), as applicable to the VEGP reactor vessel internals.

c. SNC will submit an inspection plan for the VEGP reactor vessel internals to the NRC for review and approval not less than 24 months before entering the period of extended operation for VEGP Units 1 and 2. This inspection plan will address the bases, inspection methods, and acceptance criteria associated with aging management of the reactor vessel thermal sleeves and the core support lugs (along with the associated support pads and attachment welds).

The Reactor Vessel Internals Program will be implemented prior to the period of extended operation.

# 19.2.25 REACTOR VESSEL SURVEILLANCE PROGRAM

The Reactor Vessel Surveillance Program manages loss of fracture toughness due to neutron embrittlement in reactor vessel alloy steel materials exposed to neutron fluence exceeding 1 x  $10^{17}$  n/cm<sup>2</sup> (E > 1.0 MeV). The program is based on 10 CFR 50, Appendix H, "Reactor Vessel Material Surveillance Requirements," and ASTM E 185-82, "Standard Practice for Conducting Surveillance Tests for Light-Water Cooled Nuclear Power Reactor Vessels."

Capsules are periodically removed during the course of plant operating life. Neutron embrittlement is evaluated through surveillance capsule testing and evaluation, fluence calculations and benchmarking, and monitoring of effective full power years (EFPYs).

For both the VEGP Unit 1 and 2 reactor vessels, capsules with accumulated neutron fluence equivalent to 60 years of operation have already been pulled and tested. The remaining capsules (2 capsules in each unit) will be removed such that, at the time of removal, each of the remaining capsules will have accumulated neutron fluence that is not less than once, nor greater than twice, the peak end of life fluence expected for an additional 20-year license renewal term (80 years of operation).

The Reactor Vessel Surveillance Program will be enhanced as follows:

- 1. Prior to removal of the last surveillance capsule in each unit, program documents will be revised to require that tested and untested specimens from all capsules removed from the VEGP reactor vessels remain in storage.
- 2. Alternate dosimetry will be installed to monitor neutron fluence on the reactor vessel after removal of the last surveillance capsule in that unit. This enhancement will be implemented prior to removal of the last surveillance capsule in each unit.

#### 19.2.26 STEAM GENERATOR TUBING INTEGRITY PROGRAM

The Steam Generator Tubing Integrity Program is a subprogram of the Steam Generator Program, which is an integrated program for managing the condition of the VEGP steam generators. The program focuses on steam generator tube integrity, tube planning, and the management and repair of steam generator tubing. The Steam Generator Program is in compliance with the program described in NEI 97-06, Steam Generator Program Guidelines, and VEGP Technical Specifications, subsection 5.5.9. Program deviations from NEI 97-06 are prepared and approved in accordance with NEI 97-06 and EPRI steam generator management program guidance.

The program includes a balance of prevention, inspection, evaluation and repair, and leakage monitoring. Major program elements include degradation assessments, inspection, integrity assessments, leakage monitoring, and chemistry controls.

# 19.2.27 STEAM GENERATOR PROGRAM FOR UPPER INTERNALS

The Steam Generator Program for Upper Internals is a plant-specific subprogram of the VEGP Steam Generator Program, which is an integrated program for managing the condition of the steam generators. The Steam Generator Program is in compliance with the program described in NEI 97-06, Steam Generator Program Guidelines.

The Steam Generator Program for Upper Internals includes VEGP Steam Generator Program activities associated with aging management of the steam generator upper internals components determined to be within the scope of license renewal. The program implements inspection activities intended to detect degradation of secondary side internals needed to maintain tubing integrity and accomplish steam generator intended functions. An assessment based upon steam generator design, potential degradation mechanisms, and related VEGP and industry operating experience is performed to establish inspection requirements for secondary side internals components. The resulting inspection requirements are incorporated into the steam generator inspection plans.

# 19.2.28 WATER CHEMISTRY CONTROL PROGRAM

The VEGP Water Chemistry Control Program mitigates loss of material, cracking, and reduction of heat transfer in system components and structures through the control of water chemistry. The program includes control of detrimental chemical species and the addition of chemical agents.

The VEGP Water Chemistry Control Program is based on the EPRI water chemistry guidelines for primary and secondary water chemistry control.

The One-Time Inspection Program includes inspections to verify the effectiveness of the Water Chemistry Control Program.

VEGP will monitor spent fuel pool aluminum concentrations to ensure the Boral spent fuel racks will continue to perform their intended function during the period of extended operation. If adverse trends are identified, SNC will implement corrective actions. Additionally, SNC will monitor industry experience related to Boral and will take appropriate actions if significant degradation of Boral is identified.

# 19.2.29 10 CFR 50 APPENDIX J PROGRAM

The 10 CFR 50 Appendix J Program monitors leakage rates through the containment pressure boundary, including penetrations and access openings. Containment leak rate tests assure that leakage through the primary containment and systems and components penetrating primary containment does not exceed the allowable leakage limits specified within the VEGP Technical Specifications. Corrective actions are taken if leakage rates exceed established administrative limits for individual penetrations or the overall containment pressure boundary.

# 19.2.30 INSERVICE INSPECTION PROGRAM – IWE

The VEGP Inservice Inspection Program – IWE is a plant-specific program implemented in accordance with 10 CFR 50.55(a), which imposes the inservice inspection requirements of ASME Section XI, Subsection IWE. The program manages aging effects for the containment liners and its integral attachments including connecting penetrations and parts forming the

leaktight boundary. The primary inspection method for the program is periodic visual examination along with limited volumetric examinations utilizing ultrasonic thickness measurements as needed.

In conformance with 10 CFR 50.55a(g)(4)(ii) and as based on ASME Inservice Inspection Program B (IWA-2432), the VEGP Inservice Inspection Program – IWE is updated at the end of each 120-month inspection interval to the latest edition and addenda of the Code specified in 10 CFR 50.55a, 12 months before the start of the inspection interval.

Prior to the period of extended operation, the VEGP Inservice Inspection Program – IWE will be revised to provide more explicit direction to the registered professional engineer for trending and evaluating conditions identified during visual examinations.

### 19.2.31 INSERVICE INSPECTION PROGRAM – IWL

The VEGP Inservice Inspection Program – IWL is a plant-specific program implemented in accordance with 10 CFR 50.55(a), which imposes the inservice inspection requirements of ASME Section XI Subsection IWL, except where an exemption, relief, or an alternative has been authorized by the NRC, for Class CC components. The program manages the reinforced concrete and unbonded post-tensioning systems of the containment structures.

In conformance with 10 CFR 50.55a(g)(4)(ii) and as based on ASME Inservice Inspection Program B (IWA-2432), the VEGP Inservice Inspection Program – IWL is updated at the end of each 120-month inspection interval to the latest edition and addenda of the Code specified in 10 CFR 50.55a, 12 months before the start of the inspection interval.

Prior to the period of extended operation, the VEGP Inservice Inspection Program – IWL will be revised to provide more explicit direction to the registered professional engineer for trending and evaluating conditions identified during concrete visual examinations.

# 19.2.32 STRUCTURAL MONITORING PROGRAM

The VEGP Structural Monitoring Program is based on the requirements and guidance set forth in 10 CFR 50.65, "Requirements for Monitoring the Effectiveness of Maintenance at Nuclear Power Plants," and Regulatory Guide 1.160, Rev. 2, "Monitoring the Effectiveness of Maintenance at Nuclear Power Plants." VEGP uses the Structural Monitoring Program to monitor the condition of structures and structural components within the scope of the Maintenance Rule, thereby providing reasonable assurance that there is no loss of structure or structural component intended function.

Prior to the period of extended operation, VEGP will implement the following enhancements to the Structural Monitoring Program:

- The scope of the Structural Monitoring Program will be expanded to include the additional structures that require monitoring for license renewal.
- The scope of inspection for structures that require monitoring for license renewal will be clarified. An area-based inspection will be performed unless a detailed inspection scope is provided.
- The Structural Monitoring Program scope for hangers and supports will be clarified.

- Program requirements will be revised to include periodic groundwater monitoring to confirm that groundwater chemistry remains nonaggressive as defined in NUREG 1801.
- Underwater inspection of the NSCW cooling tower basins, including appropriate inspection and acceptance criteria, will be added to the Structural Monitoring Program.
- Guidance will be given regarding proper documentation of condition adverse to quality and its probable causes for any CR written against a finding during Structural Monitoring Program walkdown.
- For any finding (e.g., crack, leakage, etc.) guidance will be given for data to be collected and evaluated.
- More explicit direction will be given for trending of the problems.

### 19.2.33 STRUCTURAL MONITORING PROGRAM – MASONRY WALLS

The Structural Monitoring Program – Masonry Walls is part of the VEGP Structural Monitoring Program that implements structures monitoring requirements as specified by 10 CFR 50.65. The Masonry Wall Program manages aging of masonry walls, and structural steel restraint systems of the masonry walls, within scope of license renewal. The program includes the concrete masonry units and restraint systems used to seal and provide radiation shielding of some access openings in the Seismic Category I structures.

The program contains inspection guidelines and lists attributes that cause aging of masonry walls, which are to be monitored during structural monitoring inspections, as well as establishes examination criteria, evaluation requirements, and acceptance criteria.

The Structural Monitoring Program – Masonry Walls will be enhanced prior to the period of extended operation to include monitoring of masonry walls in the structures which are in scope for license renewal but are not currently monitored under the program.

#### 19.2.34 NON-EQ CABLES AND CONNECTIONS PROGRAM

The Non-EQ Cables and Connections Program will be used to maintain the function of electrical cables and connections, which are not subject to the environmental qualification requirements of 10 CFR 50.49, but are exposed to adverse localized environments caused by heat, radiation, or moisture. An adverse localized environment is an environment that is significantly more severe than the service condition for the insulated cable or connection.

A representative sample of accessible insulated cables and connections within the scope of license renewal will be visually inspected for cable and connection jacket surface anomalies such as embrittlement, discoloration, and cracking. The technical basis for the sample selections of cables and connections to be inspected is provided. The scope of this sampling program includes electrical cables and connections in adverse localized environments.

The Non-EQ Cables and Connections Program will be implemented and the first inspection completed prior to the period of extended operation.

#### 19.2.35 NON-EQ INACCESSIBLE MEDIUM-VOLTAGE CABLES PROGRAM

The Non-EQ Inaccessible Medium-Voltage Cables Program manages the aging effects for inaccessible medium-voltage cables (cables with operating voltage from 2 kV to 35 kV) in the scope of license renewal exposed to significant moisture and voltage. The aging effect of concern is "localized damage and breakdown of insulation." The program includes periodic inspection and removal of water accumulation in cable manholes and periodic cable testing.

Manholes which retain water and contain medium-voltage cables in the scope of license renewal are periodically inspected for water collection and the accumulated water removed, as needed. The frequency of inspection is based on actual plant-experience but at least once every 2 years.

In-scope medium-voltage cables exposed to significant moisture and voltage are tested at least once every 10 years to provide an indication of the condition of the conductor insulation. The specific test performed is a proven test for detecting deterioration of the insulation system due to wetting.

The Non-EQ Inaccessible Medium-Voltage Cables Program will be implemented and the first inspections completed prior to the period of extended operation.

#### 19.2.36 NON-EQ ELECTRICAL CABLE CONNECTIONS ONE-TIME INSPECTION PROGRAM

The Non-EQ Cable Connections One-Time Inspection Program is a plant-specific program that performs one-time inspections on a sample of bolted connections in the scope of license renewal to confirm that loosening of electrical connections is not an aging effect requiring additional aging management during the period of extended operation. The program inspects for loosening of bolted connections due to thermal cycling, ohmic heating, electrical transients, vibration, chemical contamination, corrosion, and oxidation.

The factors considered for sample selection are application (medium and low voltage, defined as < 35 kV), circuit loading (high loading), and location (high temperature, high humidity, vibration, etc.). The technical basis for the sample selections will be documented. Inspection methods may include thermography, contact resistance testing, or appropriate methods including visual inspection based on plant configuration and industry guidance.

The inspections will be performed within a window of 10 years immediately preceding the period of extended operation.

#### 19.2.37REFERENCES

- NUREG-1800, Standard Review Plan for Review of License Renewal Applications for Nuclear Power Plants, U. S. Nuclear Regulatory Commission, (Rev. 1), September 2005.
- 2. NUREG-1801, Generic Aging Lessons Learned (GALL) Report, U. S. Nuclear Regulatory Commission, (Rev. 1), September 2005.
- 3. Vogtle Electric Generating Plant Technical Specifications, Units 1 and 2.
# 19.3 AGING MANAGEMENT PROGRAMS - TIME LIMITED AGING ANALYSES (TLAA)

# 19.3.1 ENVIRONMENTAL QUALIFICATION PROGRAM

The Environmental Qualification (EQ) Program implements the requirements of 10 CFR 50.49. The EQ Program has been established to demonstrate that certain electrical components located in harsh plant environments are qualified to perform their safety functions in those harsh environments, consistent with 10 CFR 50.49 requirements. The EQ Program manages component thermal, radiation, and cyclical aging, as applicable, through the use of aging evaluations. The program requires action be taken before individual components in the scope of the program exceed their qualified life. Actions taken include replacement on a specified time interval of piece parts or complete components to maintain qualification and reanalysis.

As required by 10 CFR 50.49, EQ components not qualified for the current license term are to be refurbished, replaced, or have their qualification extended prior to reaching the aging limits established in the evaluation. Some aging evaluations for EQ components specify a qualification of at least 40 years and are considered TLAAs for license renewal. The EQ Program ensures that these EQ components are maintained within the bounds of their qualification bases.

# **19.3.2 FATIGUE MONITORING PROGRAM**

The VEGP Fatigue Monitoring Program consists of two existing programs, which are the Fatigue and Cycle Monitoring Program and Thermal Stratification Data Collection. The Fatigue and Cycle Monitoring Program, also known as the VEGP Component or Cyclic Transient Limit Program (CCTLP), is described in subsection 5.5.5 of the Technical Specifications. This program provides controls to track the transient cycles to ensure that components are maintained within the design limit. The component cyclic or transient limits are provided in VEGP UFSAR paragraph 3.9.N.1. The Thermal Stratification Data Collection Program monitors for adverse thermal stratification and cycling resulting from isolation valve leakage in the normally stagnant nonisolable reactor coolant system (RCS) branch lines identified in the VEGP response to IEB 88-08. The VEGP Fatigue Monitoring Program uses a combination of cycle counting, cycle-based fatigue monitoring, and stress-based fatigue monitoring to monitor and track fatigue usage.

At least 2 years prior to the period of extended operation, the Fatigue Monitoring Program will be enhanced as follows:

- 1. Implementing documents will be revised to address the effect of the full structural weld overlays applied to the pressurizer spray and surge nozzles on the stress-based module calculation of cumulative usage factor (CUF).
- 2. The VEGP UFSAR will be revised to require fatigue monitoring of the accumulator/reactor heat removal (RHR) nozzles and pressurizer heater penetrations.
- 3. Implementing documents will be revised to reduce acceptable CUF values to account for environmental fatigue effects for those NUREG-6260 locations monitored for fatigue.

- 4. Implementing documents will be revised to explicitly require that the corrective actions initiated for exceeding an acceptance criterion include a review to identify and assess any additional affected reactor coolant pressure boundary locations.
- 5. SNC will revise the FatiguePro software to calculate a minimum projected value of 1 for any events that may potentially occur.
- 6. SNC will revise the FatiguePro initial CUF values for the Unit 1 and Unit 2 hot leg surge nozzles, pressurizer surge nozzles, and pressurizer heater penetrations to double the current values and recalculate the current and projected CUFs.
- 7. SNC will implement a fatigue management software program that uses six stress components in the stress-based fatigue calculation. The software will be appropriately benchmarked against an ASME NB-3200 fatigue analysis, and the stress-based fatigue monitoring locations will be modeled with the as-built configuration. The new software will be used to reproject 60-year CUF values for the monitored locations. When those locations were evaluated for environmental effects on fatigue, the new software will also be used to demonstrate that the environmental effects on fatigue will be adequately managed for those locations during the period of extended operation.

# 19.3.3 REFERENCES

- 1. NUREG-1800, Standard Review Plan for Review of License Renewal Applications for Nuclear Power Plants, U. S. Nuclear Regulatory Commission (Rev. 1), September 2005.
- 2. NUREG-1801, Generic Aging Lessons Learned (GALL) Report, U. S. Nuclear Regulatory Commission (Rev. 1), September 2005.
- 3. Vogtle Electric Generating Plant Technical Specifications, Units 1 and 2.

# 19.4 EVALUATION OF TIME LIMITED AGING ANALYSES (TLAA)

In accordance with 10 CFR 54.21(c), an application for a renewed operating license must include evaluation of TLAAs for the period of extended operation. This section summarizes the TLAAs identified for VEGP license renewal.

# 19.4.1 REACTOR VESSEL NEUTRON EMBRITTLEMENT ANALYSES

Analyses associated with embrittlement of reactor vessel materials due to neutron irradiation are TLAAs. The end-of-life (EOL) bases for these analyses are selected to bound the projected effective full-power years (EFPY) for an operating term of 60 years.

The following VEGP analyses are TLAAs that address the effects of neutron embrittlement on the VEGP reactor vessels:

- Neutron fluence.
- Upper-Shelf Energy (USE).
- Pressurized Thermal Shock (PTS).
- Adjusted Reference Temperature (ART).
- Pressure-Temperature (P-T) limits.

## 19.4.1.1 <u>Neutron Fluence Calculation</u>

The VEGP reactor vessel neutron fluence calculations were projected out to EOL for the period of extended operation in accordance with 10 CFR 54.21(c)(1)(ii). The reactor vessel neutron fluences, including extended beltline materials, were calculated using a method satisfying the requirements set forth in Regulatory Guide 1.90, "Calculational and Dosimetry Methods for Determining Pressure Vessel Neutron Fluence," Revision 0 (March 2001). These projections are used in the USE, PTS, ART, and P-T analyses described in the sections that follow.

## 19.4.1.2 Upper-Shelf Energy (USE) Calculation

Charpy impact test upper-shelf absorbed energy (USE) of no less than 50 ft-lbs throughout the life of the reactor vessel, unless an approved analysis supports a lower value.

The VEGP analyses have been projected to the end of the period of extended operation for the reactor vessel materials (base materials and welds) with projected fluence exceeding 1 x  $10^{17}$  n/cm<sup>2</sup> (MeV > 1.0). All Unit 1 and Unit 2 base materials and welds have a USE value at EOL of greater than 50 ft-lbs, which meets the acceptance criteria of 10 CFR 50, Appendix G. Therefore, these TLAAs have been shown to be acceptable for the period of extended operation in accordance with 10 CFR 54.21(c)(1)(ii).

# 19.4.1.3 Pressurized Thermal Shock (PTS) Calculation

The requirements of 10 CFR 50.61 provide for protection against PTS events in pressurized water reactors. The screening criterion in 10 CFR 50.61 is  $270^{\circ}$ F for plates, forgings, and axial welds and  $300^{\circ}$ F for circumferential welds. According to this regulation, if the calculated RT<sub>PTS</sub> for the reactor beltline materials is less than the specified screening criterion, then the vessel is acceptable with regard to the risk of vessel failure during postulated pressurized thermal shock transients.

The RT<sub>PTS</sub> calculations for VEGP Units 1 and 2 have been projected to the end of the period of extended operation for all reactor vessel materials (base materials and welds) with projected fluence exceeding 1 x  $10^{17}$  n/cm<sup>2</sup> (MeV > 1.0). All Unit 1 and Unit 2 base materials and welds meet the screening criteria contained in 10 CFR 50.61 at EOL. Therefore, these TLAAs have been shown to be acceptable for the period of extended operation in accordance with 10 CFR 54.21(c)(1)(ii).

# 19.4.1.4 Adjusted Reference Temperature (ART) Calculation

The ART values are an input to the pressure-temperature (P-T) limit curves discussed in the following section. The calculations determining the ART for the critical locations of the reactor vessel meet the definition of the TLAA pursuant to the criteria of 10 CFR 54.3. These ART calculations have been projected through the end of the period of extended operation and the results demonstrate the beltline materials remain limiting, and the projected ART values permit adequate operating margins to P-T limits through the period of extended operation. Therefore, these TLAAs have been shown to be acceptable for the period of extended operation in accordance with 10 CFR 54.21(c)(1)(ii).

# 19.4.1.5 Pressure-Temperature (P-T) Limits Calculation

Appendix G of 10 CFR Part 50 requires heatup and cooldown of the reactor pressure vessel be accomplished within established pressure and temperature limits. Plant-specific calculations establish these limits. The calculations utilize materials and fluence data obtained through plant-specific reactor surveillance capsule programs. The calculations for VEGP Units 1 and 2 meet the definition of a TLAA.

As described in the Pressure Temperature Limits Report (PTLR), the Reactor Vessel Surveillance Program updates the P-T limit curves considering the data gained from examination of surveillance specimens from capsules that SNC pulls. The content and update of the PTLR is in accordance with the requirements of subsection 5.6.6 of the VEGP Technical Specifications. When the operating conditions of each unit merit the use of a difference curve, the PTLR for that unit is updated to include P-T limit curves that bound the current level of neutron embrittlement (i.e., EFPY) for the unit. Therefore, this TLAA demonstration is made in accordance with 10 CFR 54.21(c)(1)(ii) and (iii).

The VEGP PTLR (for each unit) will be updated to address neutron embrittlement for the 60year operating life prior to the unit entering the period of extended operation.

# 19.4.2METAL FATIGUE ANALYSIS

The thermal fatigue analyses of the VEGP mechanical components have been identified as TLAAs.

# 19.4.2.1 ASME Section III, Class 1 Component Fatigue Analysis

The VEGP design incorporates the requirements of Section III Class 1 of the ASME Code, which requires a discrete analysis of the thermal, mechanical, and dynamic stress cycles on components that make up the reactor coolant pressure boundary. Although original design specifications commonly state that the transient conditions are for a 40-year design life, the fatigue analyses themselves are based on specified numbers of design transients, rather than on a specific operating life. Operating experience at VEGP and similar units has demonstrated that the analyzed numbers of design basis transients are, in general, conservative for a 40-year life. The Fatigue Monitoring Program monitors and tracks the transient cycles.

To address the additional operating term, the VEGP design transient cycles were projected through the period of extended operation. For the feedwater cycling, loss of charging flow, and loss of letdown and return to service transients, VEGP relies on cumulative usage factor (CUF) monitoring of the limiting component locations in lieu of cycle counting. Therefore, the CUFs were projected for these limiting locations in lieu of projecting their transient cycles. These limiting component locations are the steam generator main and auxiliary feedwater nozzles and the normal and alternate charging nozzles. The results of the cycles and CUF projections show that the original transient cycles were conservative and that the design fatigue analyses for Class 1 components and piping remain valid for 60 years.

In addition to the original design transients, fatigue loading transients and issues have been subsequently identified that are not part of the original fatigue analyses. For the pressurizer lower head and surge line, thermal stratification and insurge/outsurge transients are evaluated (IEB 88-11). Also, the impact of the reactor coolant system environment on the fatigue life of piping and components (GSI-190) requires specific evaluation for license renewal.

To address NRC IEB 88-11, the impact of thermal stratification on the fatigue usage in the surge line was evaluated for VEGP. The original evaluation showed that the surge line fatigue usage was acceptable for 40 years of operation, including the effects of thermal stratification due to insurge and outsurges from the pressurizer. For license renewal, stress-based fatigue monitoring is credited for managing the CUF of the surge line, including the effects of pressurizer insurge/outsurge and thermal stratification in both the pressurizer lower head and both surge line nozzles.

Generic Safety Issue (GSI) 190 addresses fatigue life of metal components and was closed by the NRC in December 1999. In the closure letter, however, the NRC concluded that licensees should address the effects of reactor coolant environment on the fatigue life of selected components as aging management programs are formulated in support of license renewal.

The effects of reactor coolant environment on component fatigue life for locations equivalent to those in Section 5.4 of NUREG/CR-6260 for the newer vintage Westinghouse plants have been evaluated for VEGP using the formulas from NUREG/CR-5704 for stainless steel components and from NUREG/CR-6583 for carbon and low-alloy steel components.

For the following locations, the application of the appropriate environmental factors to the design CUF values that were calculated based on the VEGP set of original design transients yielded acceptable results (e.g., CUF < 1.0):

- Reactor vessel shell and lower head.
- Reactor vessel inlet and outlet nozzles.

For the following locations, the application of the appropriate environmental factors to the CUF values that were calculated based on the VEGP set of original design transients yielded

unacceptable results without additional management. VEGP manages the environmentally adjusted fatigue CUF values for these locations using fatigue monitoring implemented by the Fatigue Monitoring Program:

- Surge line hot leg nozzle.
- Pressurizer heater penetrations.
- Pressurizer surge line nozzles.
- Charging nozzles.
- Safety injection nozzles.
- Accumulator / RHR nozzles.

NRC Branch Technical Position MEB 3-1 is the basis for the VEGP criteria for the postulation of high-energy line breaks (HELBs) with the exception of lines that have eliminated postulated breaks based on leak-before-break analysis. One of the criteria in MEB 3-1 for Class 1 piping is postulating pipe breaks at any intermediate locations where the CUF exceeds 0.1. The NRC staff has determined that this analysis qualifies as a TLAA.

The existing VEGP HELB analyses have been shown to remain valid for the period of extended operation as long as the Fatigue Monitoring Program maintains the CUF of the charging nozzles less than or equal to 1.0. Therefore, this TLAA has been demonstrated to be acceptable for the period of extended operation in accordance with 10 CFR 54.21(c)(1)(i) and 10 CFR 54.21(c)(1)(ii).

Full structural weld overlays (FSWOL) have been installed on the pressurizer spray nozzles, pressurizer safety and relief nozzles, and the pressurizer surge nozzles. Fatigue crack growth analyses using ASME Code Section XI methodology were performed to demonstrate the fatigue qualification at the structural weld overlay regions. Reconciliation of the existing fatigue evaluation was performed for the limiting locations outside the FSWOL, and it was demonstrated that the pressurizer nozzles would still meet the applicable ASME Code Section III requirements. In summary, the reconciliation of the existing fatigue evaluation that was performed for the limiting locations outside the FSWOL is a TLAA that remains valid for the period of extended operation, because the cycles assumed will not be exceeded during 60 years of operation. Therefore, this TLAA has been demonstrated to be acceptable for the period of extended operation in accordance with 10 CFR 54.21(c)(1)(i).

In conclusion, the VEGP fatigue TLAAs for ASME Class 1 components have been evaluated and shown to remain valid or are adequately managed for the period of extended operation, in accordance with the demonstration methods of 10 CFR 54.21(c)(1)(i) and 10 CFR 54.21(c)(1)(iii). The Fatigue Monitoring Program monitors and tracks transient cycles and their severity and performs CUF monitoring of selected components to ensure that Class 1 components are maintained within their fatigue design limits.

# 19.4.2.2 ASME Section III, Non-Class 1 Component Fatigue Analysis

The design of ASME III Code Class 2 and 3 piping systems at VEGP incorporates stress reduction factors for determining the acceptability of the piping design with respect to thermal stresses. Those in-scope components that are designed in accordance with ANSI B31.1 requirements also incorporate stress-reduction factors based upon an assumed number of

thermal expansion cycles. In general, 7000 full-temperature thermal cycles are assumed in the calculation of the thermal expansion stress, leading to a stress-reduction factor of 1.0 in the stress analyses.

SNC evaluated the validity of this assumption of 7000 full-temperature thermal cycles for 60 years of plant operation. The results of this evaluation indicate that the 7000-thermal cycle assumption remains valid and bounding for 60 years of operation. Therefore, the existing pipe stress calculations are valid for the extended period of operation in accordance with 10 CFR 54.21(c)(1)(i).

There are non-Class 1 fatigue evaluations that use a different method of analysis than the 7000 cycles described above. In general, those evaluations use the same cycles, or a subset of the cycles, used for the Class 1 piping. These analyses include the letdown heat exchangers, containment cooler cooling coils, and the main stream isolation valves. In each case, the analysis was determined to remain valid for the period of extended operation in accordance with 10 CFR 54.21(c)(1)(i).

# 19.4.2.3 Reactor Coolant Pump Flywheel Fatigue

A calculation was performed for the VEGP reactor coolant pump flywheels which assumes that each pump will be subjected to 6000 start/stop cycles over a 60-year life. Current projections indicate that the 6000 start/stop cycles will remain bounding for 60 years of operation by a large margin. Therefore, fatigue of the reactor coolant pump flywheels is demonstrated in accordance with 10 CFR 54.21(c)(1)(i).

# 19.4.2.4 Fatigue of Reactor Vessel Supports

The Westinghouse Generic Technical Report WCAP 14422, Revision 2a, identifies fatigue of reactor vessel supports as a potential TLAA if the supports of the reactor vessel were constructed in accordance with the 1963 version of the AISC Code.

The reactor pressure vessel supports embedded within the primary shield wall are procured in accordance with ASME Code, Section III, Division 1, Subsection NF; however, since they are outside the ASME jurisdictional boundary, their design follows AISC specifications. Therefore, both the 1969 version of the AISC Code and ASME Code, Section III, Division 1, Subsection NF apply to the supports.

In the SER for WCAP 14422, the NRC has indicated that licensees must ensure that a version of the AISC Code later than 1963 was used. Since the design used the 1969 version of the AISC Code, the existing analysis is demonstrated to be valid for the extended term of operation in accordance with 10 CFR 54.21(c)(1)(i).

# 19.4.2.5 Fatigue of Steam Generator Secondary Manway and Handhole Bolts

Westinghouse performed a fatigue calculation for steam generator secondary manway and handhole bolts that assumed the same cycles used for Class 1 component fatigue evaluations. That calculation resulted in a qualified life for the manway bolts of only 20 years. In 1993, it was determined that after low-temperature rerate, the qualified life of the manway bolts would be reduced to 14.5 years. A new secondary side manway and handhole bolts fatigue evaluation was performed based on actual cycles to qualify the bolts for 40 years with rerating.

To ensure that the cycle limits for these bolts are not exceeded, SNC will replace both the secondary side manway bolts and the handhole bolts after 35 years of service, unless a less restrictive replacement schedule is developed and documented based on potential updated analyses initiated by the Bolting Integrity Program. SNC considers this fatigue evaluation a TLAA that is managed by the Bolting Integrity Program. Therefore, this TLAA is demonstrated in accordance with 10 CFR 54.21(c)(1)(iii).

# 19.4.2.6 Fatigue of Reactor Vessel Internals

A fatigue analysis of the reactor vessel internals was not required when VEGP was originally designed. However, as part of rerating, Westinghouse performed a fatigue calculation for reactor vessel internals that assumed the same cycles used for Class 1 component fatigue evaluations and resulted in CUFs less than 1.0 for all subcomponents evaluated.

VEGP evaluated this TLAA for the extended period of operation. Since the analysis utilized the same design transients as the Class 1 component evaluations, the evaluation of the ASME Class 1 piping and component design transient cycles is also applicable to the reactor vessel internals. The design cycles for the transients applicable to the reactor vessel internals are bounded by the RCS design cycles, therefore the reactor vessel internals fatigue analysis remains valid for the period of extended operation. This TLAA is demonstrated in accordance with 10 CFR 54.21(c)(1)(i).

# 19.4.3 ENVIRONMENTAL QUALIFICATION CALCULATIONS

The NRC has established environmental qualification (EQ) requirements in 10 CFR Part 50 Appendix A and in 10 CFR 50.49. The Environmental Qualification Program for VEGP has been established to demonstrate that certain electrical components are qualified to perform safety functions in the harsh environment following a DBA. Elements of the proof of qualification involve the original 40-year license period. Hence, the qualification reports and calculations that comprise the EQ Program meet the definition of a TLAA. Qualified lives for EQ components have already been determined, and these components are tracked to determine when they are nearing the end of their qualified lives. For those components that are nearing the end of their qualified lives, the EQ Program has provisions for the component to be reevaluated for longer service, refurbished, requalified, or replaced. The EQ Program will be continued through the period of extended operation. Therefore, this TLAA is demonstrated in accordance with 10 CFR 54.21(c)(1)(iii).

## 19.4.4 CONTAINMENT TENDON PRESTRESS ANALYSIS

To meet the requirements on 10 CFR 50.55a (b)(2)(ix)(B), SNC uses an analysis to predict the amount of residual prestress in the containment tendons for VEGP. This analysis meets the definition of a TLAA. SNC extended the analysis to estimate the amount of residual prestress on the tendons after 60 years of operation. The analysis results conclude that acceptable containment tendon prestress will be retained throughout the period of extended operation. Therefore, adequate containment prestress for the period of extended operation is demonstrated in accordance with 10 CFR 54.21(c)(1)(ii).

Results from containment tendon surveillances conducted under the Inservice Inspection Program – IWL periodically update the analysis and confirm prestresses remain above the minimum required values.

# 19.4.5 PENETRATION LOAD CYCLES

A fatigue analysis was required for some of the VEGP containment penetrations. Those analyses qualify as TLAAs. Review of the transient assumptions for those evaluations against the transient assumptions for Class 1 component fatigue determined that none of the cycles assumed in the penetration fatigue analyses will be exceeded within the period of extended operation. Therefore, fatigue analyses for containment penetrations are acceptable without revision, and the TLAAs are demonstrated in accordance with 10 CFR 54.21(c)(1)(i).

## 19.4.6 OTHER PLANT-SPECIFIC ANALYSIS

## 19.4.6.1 Leak-Before-Break Analysis

Plant-specific leak-before-break (LBB) analyses have been performed for both VEGP units. These analyses provide the technical justification for changes to the structural design basis involving protection against the effects of postulated pipe ruptures and are identified as TLAAs since they include assumptions regarding fatigue cycles and material fracture toughness properties.

VEGP LBB analyses exist for the Units 1 and 2 reactor coolant loop piping, the pressurizer surge line, and the Unit 2 accumulator injection and the RHR branch connection lines.

The LBB analyses for the pressurizer surge line and the Unit 2 RHR branch connection line were reviewed and determined to be acceptable without revision for the period of extended operation. Therefore, these LBB analyses are demonstrated in accordance with 10 CFR 54.21(c)(1)(i).

The analyses for the primary coolant loops and the Unit 2 accumulator line have been evaluated and updated to address operation through 60 years, including reductions in cast material fracture toughness properties due to thermal aging. Therefore, these LBB analyses are demonstrated in accordance with 10 CFR 54.21(c)(1)(ii).

WCAP-10551-P, Addendum 1 performed an LBB evaluation for the Units 1 and 2 primary loop piping that explicitly addressed the PWSCC concern for the Alloy 82/182 welds in this piping. However, the NRC has not yet accepted the process used as adequately addressing their concerns. Once the NRC has accepted a process for addressing PWSCC of Alloy 82/182 welds in LBB evaluations and at least 2 years prior to the period of extended operation, SNC will verify the LBB evaluation in WCAP-10551-P, Addendum 1 meets the conditions of that process or have it reperformed using the acceptable process.

## 19.4.6.2 Fuel Oil Storage Tank Corrosion Allowance

The VEGP diesel fuel oil storage tanks and associated piping are not provided with cathodic protection; therefore, a liberal corrosion allowance was included. A calculation performed to evaluate the corrosion allowance included a 40-year assumption and has been determined to be a TLAA.

The calculation determined the depth of penetration for a hole of approximately 1/32 in. diameter (0.001 in<sup>2</sup>) in the coating. The calculation was reviewed for license renewal, and it was determined that depth of penetration due to corrosion would not exceed the corrosion allowance during a 60-year operating life. Specifically, consideration of 60 years instead of

40 years in the calculation increases the depth of penetration due to corrosion from 25% to 51% of the corrosion allowance for the tanks and from 50% to 76% of the corrosion allowance for the pipes.

Therefore, demonstration is in accordance with 10 CFR 54.21(c)(1)(ii).

# 19.4.6.3 Steam Generator Tube, Loss of Material

VEGP UFSAR subsection 5.4.2 describes allowances for erosion and corrosion that are partially based upon a measured loss of material rate for 40 years. These allowances are used as inputs to demonstrate that stress limits established by Regulatory Guide 1.121 continue to be satisfied. Subsection 5.4.2 demonstrates that a large margin exists between the allowable tube wall degradation which satisfies Regulatory Guide 1.121 limits and the tube plug limits established by the VEGP Steam Generator Tubing Integrity Program. Increasing the expected corrosion allowance to address the period of extended operation has an insignificant effect on this margin. Further, steam generator tubing wall loss is managed by the Steam Generator Tubing Integrity Program and the requirements of Regulatory Guide 1.121 are considered within that program.

Therefore, this TLAA is managed by the Steam Generator Tubing Integrity Program and is demonstrated in accordance with 10 CFR 54.21(c)(1)(iii).

# 19.4.6.4 <u>Cold Overpressure Protection System</u>

As described in paragraph 5.2.2.10 of the VEGP UFSAR, VEGP has a cold overpressure mitigation system (COPS). A calculation has been performed to confirm that the setpoints will maintain the system pressure within the established limits when the pressure difference between the pressure transmitter and reactor midplane and maximum temperature/pressure instrument uncertainties are applied to the setpoints. This calculation meets the definition of a TLAA.

The P-T limit curves in the VEGP PTLR have been evaluated for 36 EFPY. When a revision to the PTLR is issued, the cold overpressure mitigation system setpoints will also be updated to reflect the period covered by the PTLR revision. Therefore, this cold overpressure mitigation setpoint calculation TLAA is demonstrated in accordance with 10 CFR 54.21(c)(1)(ii).

As described in the PTLR, the Reactor Vessel Surveillance Program updates the P-T limit curves considering the data gained from capsules SNC pulls, and the content and update of the PTLR is in accordance with VEGP Technical Specifications, subsection 5.6.6. The associated COPS setpoints are also updated as operational needs dictate to bound the current level of neutron embrittlement (i.e., EFPY) for the unit. Therefore, this TLAA demonstration is made in accordance with 10 CFR 54.21(c)(1)(ii) and (iii).

The VEGP PTLR (for each unit) will be updated to address neutron embrittlement for a 60-year operating life, including any changes to the COPS setpoints, prior to the unit entering the period of extended operation.

# 19.4.6.5 Underclad Cracking of the Reactor Pressure Vessel

There is no plant-specific evaluation of underclad cracking at VEGP, and no such cracks have been identified. Freedom from underclad cracking is ensured by special evaluation of the procedure qualification for cladding applied on low-alloy steel (SA-508, Class 2) in accordance

with Regulatory Guide 1.43. However, SNC conservatively includes underclad cracking as a TLAA. Analyses performed by Westinghouse in WCAP-15338 demonstrate that growth of underclad cracks in Westinghouse reactor pressure vessels (RPVs) does not represent a significant challenge to reactor vessel integrity for an operating term of 60 years. The assumptions used as inputs to WCAP-15338 are applicable to VEGP. The results of these analyses demonstrate that underclad cracking of reactor vessel components is not an aging effect requiring management for VEGP. TLAA disposition is in accordance with 10 CFR 54.21(c)(1)(i).

# 19.4.7 REFERENCES

- 1. NUREG-1800, Standard Review Plan for Review of License Renewal Applications for Nuclear Power Plants, U. S. Nuclear Regulatory Commission (Rev. 1), September 2005.
- 2. NUREG-1801, Generic Aging Lessons Learned (GALL) Report, U. S. Nuclear Regulatory Commission (Rev. 1), September 2005.
- 3. Vogtle Electric Generating Plant Unit 1 & Unit 2 Technical Specifications.

# TABLE OF CONTENTS

# 1.0 INTRODUCTION AND GENERAL DESCRIPTION OF PLANT

## 1.1 Introduction

- 1.1.1 Plant Units
- 1.1.2 Plant Location
- 1.1.3 Containment Type
- 1.1.4 Nuclear Steam Supply System
  - 1.1.4.1 Reactor Type and Supplier
  - 1.1.4.2 Power Output
- 1.1.5 Schedule for Completion and Commercial Operation
- 1.1.6 Format and Content
  - 1.1.6.1 Regulatory Guide 1.70
  - 1.1.6.2 Standard Review Plan
  - 1.1.6.3 Tables and Figures
  - 1.1.6.4 Numbering of Pages
  - 1.1.6.5 Amending the FSAR
- 1.2 General Plant Description
  - 1.2.1 Site Characteristics
    - 1.2.1.1 Location
    - 1.2.1.2 Site Ownership
    - 1.2.1.3 Access to the Site
    - 1.2.1.4 Site Environs
    - 1.2.1.5 Geology
    - 1.2.1.6 Seismology
    - 1.2.1.7 Hydrology
    - 1.2.1.8 Meteorology
    - 1.2.1.9 Environmental Radiation Monitoring
  - 1.2.2 Facility Arrangement
    - 1.2.2.1 Units 1 and 2 Shared Facilities
    - 1.2.2.2 Units 1 and 2 Shared Systems

- 1.2.3 Nuclear Steam Supply System
  - 1.2.3.1 Reactor Core
  - 1.2.3.2 Reactor Coolant System
- 1.2.4 Steam and Power Conversion System
- 1.2.5 Containment
- 1.2.6 Safety Features
  - 1.2.6.1 Emergency Core Cooling System
  - 1.2.6.2 Containment Spray System
  - 1.2.6.3 Containment Cooling System
  - 1.2.6.4 Penetration Room Filtration Systems
  - 1.2.6.5 Hydrogen Recombiners
- 1.2.7 Unit Control
- 1.2.8 Plant Electrical Power
- 1.2.9 Plant Instrumentation and Control System
- 1.2.10 Auxiliary Systems
  - 1.2.10.1 Chemical and Volume Control System
  - 1.2.10.2 Residual Heat Removal System
  - 1.2.10.3 Auxiliary Feedwater System
  - 1.2.10.4 Component Cooling Water System
  - 1.2.10.5 Auxiliary Component Cooling Water System
  - 1.2.10.6 Fuel Handling and Storage System
  - 1.2.10.7 Sampling Systems
  - 1.2.10.8 Cooling Water Systems
  - 1.2.10.9 Plant Ventilation Systems
  - 1.2.10.10 Plant Fire Protection System
  - 1.2.10.11 Compressed Air Systems
  - 1.2.10.12 Radioactive Waste Disposal System
- 1.2.11 Shared Facilities and Equipment
- 1.2.12 Principal Design Criteria
  - 1.2.12.1 General Design Criteria
  - 1.2.12.2 System Criteria
- 1.2.13 Particularly Difficult Engineering Problems
- 1.2.14 Extrapolation of Technology

- 1.3 .Comparison Tables
  - 1.3.1 Comparison with Similar Facility Designs
  - 1.3.2 Comparison of Final and Preliminary Information
- 1.4 Identification of Agents and Contractors
  - 1.4.1 Applicant Construction Manager and Operator
    - 1.4.1.1 Description of Business
    - 1.4.1.2 Description of Corporate Organization
    - 1.4.1.3 Technical Qualifications
  - 1.4.2 Southern Nuclear Operating Company, Inc.
  - 1.4.3 Bechtel Power Corporation
    - 1.4.3.1 Nuclear Experience
  - 1.4.4 Nuclear Steam Supply System Manufacturer
  - 1.4.5 Division of Responsibility
    - 1.4.5.1 Design Stage
    - 1.4.5.2 Utility Company
    - 1.4.5.3 Architect/Engineer
    - 1.4.5.4 Procurement of Safety-Related Equipment
    - 1.4.5.5 Construction
    - 1.4.5.6 Operation
- 1.5 Requirements for Further Technical Information
  - 1.5.1 Core Stability Evaluation Program
  - 1.5.2 Fuel Development Program for Operation at High-Power Densities
  - 1.5.3 Incore Detector Program
  - 1.5.4 Full-Length Emergency Core Cooling Heat Transfer Test
  - 1.5.5 Blowdown Forces Program
  - 1.5.6 Reactor Vessel Thermal Shock Program
  - 1.5.7 Environmental Testing of Engineered Safety Features Related Equipment
  - 1.5.8

## TABLE OF CONTENTS (CONTINUED)

1.5.91.5.10 Generic Programs of Bechtel

- 1.6 Material Incorporated by Reference
- 1.7 1.7 Drawings and Other Detailed Information
  - 1.7.1 Electrical, Instrumentation, and Control Drawings
  - 1.7.2 Piping and Instrumentation Diagrams
  - 1.7.3 Other Detailed Information
- 1.8 Differences from Standard Review Plan Acceptance Criteria
- 1.9 Conformance to NRC Regulatory Guides
  - 1.9.1 Regulatory Guide 1.1, November 1970, Net Positive Suction Head for Emergency Core Cooling and Containment Heat Removal System Pumps
    - 1.9.1.1 Regulatory Guide 1.1 Position
    - 1.9.1.2 VEGP Position
  - 1.9.2 Regulatory Guide 1.2, November 1970, Thermal Shock to Reactor Pressure Vessels
    - 1.9.2.1 Regulatory Guide 1.2 Position
    - 1.9.2.2 VEGP Position
  - 1.9.3 Regulatory Guide 1.3, Revision 2, June 1974, Assumptions Used for Evaluating the Potential Radiological Consequences of a Loss-of-Coolant Accident for Boiling Water Reactors
  - 1.9.4 Regulatory Guide 1.4, Revision 2, June 1974, Assumptions Used for Evaluating the Potential Radiological Consequences of a Loss-of-Coolant Accident for Pressurized Water Reactors
    - 1.9.4.1 Regulatory Guide 1.4 Position
    - 1.9.4.2 VEGP Position
  - 1.9.5 Regulatory Guide 1.5, March 1971, Assumptions Used for Evaluating the Potential Radiological Consequences of a Steam Line Break Accident for Boiling Water Reactors

- 1.9.6 Regulatory Guide 1.6, March 1971, Independence Between Redundant Standby (Onsite) Power Sources and Between Their Distribution Systems
  - 1.9.6.1 Regulatory Guide 1.6 Position
  - 1.9.6.2 VEGP Position
- 1.9.7 Regulatory Guide 1.7, Revision 2, November 1978, Control of Combustible Gas Concentrations in Containment Following a Loss-of-Coolant Accident
  - 1.9.7.1 Regulatory Guide 1.7 Position
  - 1.9.7.2 VEGP Position
- 1.9.8 Regulatory Guide 1.8, Revision 2, April 1987, Qualification and Training of Personnel for Nuclear Power Plants
  - 1.9.8.1 Regulatory Guide 1.8 Position
  - 1.9.8.2 VEGP Position
- 1.9.9 Regulatory Guide 1.9, Revision 2, December 1979, Selection, Design, and Qualification of Diesel Generator Units Used as Standby (Onsite) Electric Power Systems at Nuclear Power Plants
  - 1.9.9.1 Regulatory Guide 1.9 Position
  - 1.9.9.2 VEGP Position
- 1.9.10 Regulatory Guide 1.10, Revision 1, January 1973, Mechanical (Cadweld) Splices in Reinforcing Bars of Category I Concrete Containment Structures
  - 1.9.10.1 Regulatory Guide 1.10 Position
  - 1.9.10.2 VEGP Position
- 1.9.11 Regulatory Guide 1.11, March 1971, Instrument Lines Penetrating Primary Reactor Containment
  - 1.9.11.1 Regulatory Guide 1.11 Position
  - 1.9.11.2 VEGP Position
- 1.9.12 Regulatory Guide 1.12, Revision 2, March 1997, Instrumentation for Earthquakes
  - 1.9.12.1 Regulatory Guide 1.12 Position
  - 1.9.12.2 VEGP Position
- 1.9.13 Regulatory Guide 1.13, Revision 1, December 1975, Spent-Fuel Storage Facility Design Basis
  - 1.9.13.1 Regulatory Guide 1.13 Position
  - 1.9.13.2 VEGP Position

#### TABLE OF CONTENTS (CONTINUED)

- 1.9.14 Regulatory Guide 1.14, Revision 1, August 1975, Reactor Coolant Pump Flywheel Integrity
  - 1.9.14.1 Regulatory Guide 1.14 Position
  - 1.9.14.2 VEGP Position-Prelicense
  - 1.9.14.3 VEGP Position-Post License
- 1.9.15 Regulatory Guide 1.15, Revision 1, December 1972, Testing of Reinforcing Bars for Category Concrete Structures
  - 1.9.15.1 Regulatory Guide 1.15 Position
  - 1.9.15.2 VEGP Position
- 1.9.16 Regulatory Guide 1.16, Revision 4, August 1975, Reporting of Operating Information Appendix A, Technical Specifications
  - 1.9.16.1 Regulatory Guide 1.16 Position
  - 1.9.16.2 VEGP Position
- 1.9.17 Regulatory Guide 1.17, June 1973, Protection of Nuclear Power Plants Against Industrial Sabotage
  - 1.9.17.1 Regulatory Guide 1.17 Position
  - 1.9.17.2 VEGP Position
- 1.9.18 Regulatory Guide 1.18, Revision 1, December 1972, Structural Acceptance Test for Concrete Primary Reactor Containments
  - 1.9.18.1 Regulatory Guide 1.18 Position
  - 1.9.18.2 VEGP Position
- 1.9.19 Regulatory Guide 1.19, Revision 1, August 1972, Nondestructive Examination of Primary Containment Liner Welds
  - 1.9.19.1 Regulatory Guide 1.19 Position
  - 1.9.19.2 VEGP Position
- 1.9.20 Regulatory Guide 1.20, Revision 2, May 1976, Comprehensive Vibration Assessment Program for Reactor Internals During Preoperational and Initial Startup Testing

1.9.20.1 Regulatory Guide 1.20 Position 1.9.20.2 VEGP Position

- 1.9.21 Regulatory Guide 1.21, Revision 1, June 1974, Measuring, Evaluating, and Reporting Radioactivity in Solid Wastes and Releases of Radioactive Materials in Liquid and Gaseous Effluents from Light-Water-Cooled Nuclear Power Plants
  - 1.9.21.1 Regulatory Guide 1.21 Position

### TABLE OF CONTENTS (CONTINUED)

1.9.21.2 VEGP Position

- 1.9.22 Regulatory Guide 1.22, February 1972, Periodic Testing of Protection System Actuation Functions
  - 1.9.22.1 Regulatory Guide 1.22 Position
  - 1.9.22.2 VEGP Position
- 1.9.23 Regulatory Guide 1.23, February 1972, Onsite Meteorological Programs
  - 1.9.23.1 Regulatory Guide 1.23 Position
  - 1.9.23.2 VEGP Position
- 1.9.24 Regulatory Guide 1.24, March 1972, Assumptions Used for Evaluating the Potential Consequences of a Pressurized Water Reactor Radioactive Gas Storage Tank Failure
  - 1.9.24.1 Regulatory Guide 1.24 Position
  - 1.9.24.2 VEGP Position
- 1.9.25 Regulatory Guide 1.25, March 1972, Assumptions Used for Evaluating the Potential Radiological Consequences of a Fuel Handling Accident in the Fuel Handling and Storage Facility for Boiling and Pressurized Water Reactors
  - 1.9.25.1 Regulatory Guide 1.25 Position
  - 1.9.25.2 VEGP Position
- 1.9.26 Regulatory Guide 1.26, Revision 3, February 1976, Quality Group Classifications and Standards for Water-, Steam-, and Radioactive-Waste-Containing Components of Nuclear Power Plants
  - 1.9.26.1 Regulatory Guide 1.26 Position
  - 1.9.26.2 VEGP Position
- 1.9.27 Regulatory Guide 1.27, Revision 2, January 1976, Ultimate Heat Sink for Nuclear Power Plants
  - 1.9.27.1 Regulatory Guide 1.27 Position
  - 1.9.27.2 VEGP Position
- 1.9.28 Regulatory Guide 1.28, Revision 2, February 1979, Quality Assurance Program Requirements (Design and Construction)
  - 1.9.28.1 Regulatory Guide 1.28 Position
  - 1.9.28.2 VEGP Position
- 1.9.29 Regulatory Guide 1.29, Revision 3, September 1978, Seismic Design Classification

- 1.9.29.1 Regulatory Guide 1.29 Position
- 1.9.29.2 VEGP Position
- 1.9.30 Regulatory Guide 1.30, August 1972, Quality Assurance Requirements for the Installation, Inspection, and Testing of Instrumentation and Electric Equipment
  - 1.9.30.1 Regulatory Guide 1.30 Position
  - 1.9.30.2 VEGP Position
- 1.9.31 Regulatory Guide 1.31, Revision 3, April 1978, Control of Ferrite Content in Stainless Steel Weld Metal
  - 1.9.31.1 Regulatory Guide 1.31 Position
  - 1.9.31.2 VEGP Position
- 1.9.32 Regulatory Guide 1.32, Revision 2, February 1977, Criteria for Safety-Related Electric Power Systems for Nuclear Power Plants
  - 1.9.32.1 Regulatory Guide 1.32 Position
  - 1.9.32.2 VEGP Position
- 1.9.33 Regulatory Guide 1.33, Revision 2, February 1978, Quality Assurance Program Requirements (Operation)
  - 1.9.33.1 Regulatory Guide 1.33 Position
  - 1.9.33.2 VEGP Position
- 1.9.34 Regulatory Guide 1.34, December 1972, Control of Electroslag Weld Properties
  - 1.9.34.1 Regulatory Guide 1.34 Position
  - 1.9.34.2 VEGP Position
- 1.9.35 Regulatory Guide 1.35, Revision 2, January 1976, Inservice Inspection of Ungrouted Tendons in Prestressed Concrete Containment Structures
  - 1.9.35.1 Regulatory Guide 1.35 Position
  - 1.9.35.2 VEGP Position
- 1.9.36 Regulatory Guide 1.36, February 1973, Non-Metallic Thermal Insulation for Austenitic Stainless Steel
  - 1.9.36.1 Regulatory Guide 1.36 Position
  - 1.9.36.2 VEGP Position
- 1.9.37 Regulatory Guide 1.37, March 1973, Quality Assurance Requirements for Cleaning of Fluid Systems and Associated Components of Water-Cooled Nuclear Power Plants
  - 1.9.37.1 Regulatory Guide 1.37 Position

#### TABLE OF CONTENTS (CONTINUED)

1.9.37.2 VEGP Position

- 1.9.38 Regulatory Guide 1.38, Revision 2, May 1977, Quality Assurance Requirements for Packaging, Shipping, Receiving, Storage and Handling of Items for Water-Cooled Nuclear Power Plants
  - 1.9.38.1 Regulatory Guide 1.38 Position
  - 1.9.38.2 VEGP Position
- 1.9.39 Regulatory Guide 1.39, Revision 2, September 1977, Housekeeping Requirements for Water-Cooled Nuclear Power Plants
  - 1.9.39.1 Regulatory Guide 1.39 Position
  - 1.9.39.2 VEGP Position
- 1.9.40 Regulatory Guide 1.40, March 1973, Qualification Tests of Continuous-Duty Motors Installed Inside the Containment of Water- Cooled Nuclear Power Plants
  - 1.9.40.1 Regulatory Guide 1.40 Position
  - 1.9.40.2 VEGP Position
- 1.9.41 Regulatory Guide 1.41, March 1973, Preoperational Testing of Redundant Onsite Electric Power Systems To Verify Proper Load Group Assignments
  - 1.9.41.1 Regulatory Guide 1.41 Position 1.9.41.2 VEGP Position
- 1.9.42 Regulatory Guide 1.42
- 1.9.43 Regulatory Guide 1.43, May 1973, Control of Stainless Steel Weld Cladding of Low-Alloy Steel Components
  - 1.9.43.1 Regulatory Guide 1.43 Position1.9.43.2 VEGP Position
- 1.9.44 Regulatory Guide 1.44, May 1973, Control of the Use of Sensitized Stainless Steel
  - 1.9.44.1 Regulatory Guide 1.44 Position
  - 1.9.44.2 VEGP Position
- 1.9.45 Regulatory Guide 1.45, May 1973, Reactor Coolant Pressure Boundary Leakage Detection System
  - 1.9.45.1 Regulatory Guide 1.45 Position 1.9.45.2 VEGP Position
- 1.9.46 Regulatory Guide 1.46, May 1973, Protection Against Pipe Break Inside Containment

- 1.9.46.1 Regulatory Guide 1.46 Position
- 1.9.46.2 VEGP Position
- 1.9.47 Regulatory Guide 1.47, May 1973, Bypassed and Inoperable Status Indication for Nuclear Power Plant Safety Systems
  - 1.9.47.1 Regulatory Guide 1.47 Position
  - 1.9.47.2 VEGP Position
- 1.9.48 Regulatory Guide 1.48, May 1973, Design Limits and Loading Combinations for Seismic Category 1 Fluid System Components
  - 1.9.48.1 Regulatory Guide 1.48 Position
  - 1.9.48.2 VEGP Position
- 1.9.49 Regulatory Guide 1.49, Revision 1, December 1973, Power Levels of Nuclear Power Plants
  - 1.9.49.1 Regulatory Guide 1.49 Position
  - 1.9.49.2 VEGP Position
- 1.9.50 Regulatory Guide 1.50, May 1973, Control of Preheat Temperatures for Welding of Low-Alloy Steel
  - 1.9.50.1 Regulatory Guide 1.50 Position
  - 1.9.50.2 VEGP Position
- 1.9.51 Regulatory Guide 1.51
- 1.9.52 Regulatory Guide 1.52, Revision 2, March 1978, Design, Testing, and Maintenance Criteria for Post-Accident Engineered Safety Feature Atmosphere Cleanup System Air Filtration and Adsorption Units of Light-Water-Cooled Nuclear Power Plants
  - 1.9.52.1 Regulatory Guide 1.52 Position
  - 1.9.52.2 VEGP Position
- 1.9.53 Regulatory Guide 1.53, June 1973, Application of the Single-Failure Criterion to Nuclear Power Plant Protection Systems
  - 1.9.53.1 Regulatory Guide 1.53 Position
  - 1.9.53.2 VEGP Position
- 1.9.54 Regulatory Guide 1.54, June 1973, Quality Assurance Requirements for Protective Coatings Applied to Water-Cooled Nuclear Power Plants
  - 1.9.54.1 Regulatory Guide 1.54 Position
  - 1.9.54.2 VEGP Position

- 1.9.55 Regulatory Guide 1.55, June 1973, Concrete Placement in Category 1 Structures
  - 1.9.55.1 Regulatory Guide 1.55 Position1.9.55.2 VEGP Position
- 1.9.56 Regulatory Guide 1.56, Revision 1, July 1978, Maintenance of Water Purity in Boiling Water Reactors
- 1.9.57 Regulatory Guide 1.57, June 1973, Design Limits and Loading Combinations for Metal Primary Reactor Containment System Components
  - 1.9.57.1 Regulatory Guide 1.57 Position
  - 1.9.57.2 VEGP Position
- 1.9.58 Regulatory Guide 1.58, Revision 1, September 1980, Qualification of Nuclear Power Plant Inspection, Examination, and Testing Personnel
  - 1.9.58.1 Regulatory Guide 1.58 Position
  - 1.9.58.2 VEGP Position Prelicense
  - 1.9.58.3 VEGP Position Post License
- 1.9.59 Regulatory Guide 1.59, Revision 2, August 1977, Design Basis Floods for Nuclear Power Plants
  - 1.9.59.1 Regulatory Guide 1.59 Position
  - 1.9.59.2 VEGP Position
- 1.9.60 Regulatory Guide 1.60, Revision 1, December 1973, Design Response Spectra for Seismic Design of Nuclear Power Plants
  - 1.9.60.1 Regulatory Guide 1.60 Position
  - 1.9.60.2 VEGP Position
- 1.9.61 Regulatory Guide 1.61, October 1973, Damping Values for Seismic Design of Nuclear Power Plants
  - 1.9.61.1 Regulatory Guide 1.61 Position
  - 1.9.61.2 VEGP Position
- 1.9.62 Regulatory Guide 1.62, October 1973, Manual Initiation of Protective Actions
  - 1.9.62.1 Regulatory Guide 1.62 Position
  - 1.9.62.2 VEGP Position
- 1.9.63 Regulatory Guide 1.63, Revision 2, July 1978, Electric Penetration Assemblies in Containment Structures for Light-Water-Cooled Nuclear Power Plants
  - 1.9.63.1 Regulatory Guide 1.63 Position

- 1.9.63.2 VEGP Position
- 1.9.64 Regulatory Guide 1.64, Revision 2, June 1976, Quality Assurance for the Design of Nuclear Power Plants
  - 1.9.64.1 Regulatory Guide 1.64 Position
  - 1.9.64.2 VEGP Position
- 1.9.65 October 1973, Materials and Inspections for Reactor Vessel Closure Studs
  - 1.9.65.1 Regulatory Guide 1.65 Position 1.9.65.2 VEGP Position
- 1.9.66 Regulatory Guide 1.66
  - 1.9.66.1 Regulatory Guide 1.66 Position 1.9.66.2 VEGP Position
- 1.9.67 Regulatory Guide 1.67
  - 1.9.67.1 Regulatory Guide 1.67 Position
  - 1.9.67.2 VEGP Position
- 1.9.68 Regulatory Guide 1.68, Revision 2, August 1978, Initial Test Programs for Water-Cooled Nuclear Power Plants
  - 1.9.68.1 Regulatory Guide 1.68 Position
  - 1.9.68.2 VEGP Position
  - 1.9.68.3 Regulatory Guide 1.68.2, Revision 1, July 1978, Initial Startup Test Program To Demonstrate Remote Shutdown Capability for Water-Cooled Nuclear Power Plants
  - 1.9.68.4 Regulatory Guide 1.68.3, April 1982, Preoperational Testing of Instrument and Control Air Systems
- 1.9.69 Regulatory Guide 1.69, December 1973, Concrete Radiation Shields for Nuclear Power Plants
  - 1.9.69.1 Regulatory Guide 1.69 Position
  - 1.9.69.2 VEGP Position
- 1.9.70 Regulatory Guide 1.70, Revision 3, November 1978, Standard Format and Content of Safety Analysis Reports for Nuclear Power Plants
  - 1.9.70.1 Regulatory Guide 1.70 Position
  - 1.9.70.2 VEGP Position
- 1.9.71 Regulatory Guide 1.71, December 1973, Welder Qualification for Areas of Limited Accessibility

- 1.9.71.1 Regulatory Guide 1.71 Position
- 1.9.71.2 VEGP Position
- 1.9.72 Regulatory Guide 1.72, Revision 2, November 1978, Spray Pond Piping Made from Fiberglass-Reinforced Thermo-Setting Resin
  - 1.9.72.1 Regulatory Guide 1.72 Position
  - 1.9.72.2 VEGP Position
- 1.9.73 Regulatory Guide 1.73, January 1974, Qualification Tests of Electric Valve Operators Installed Inside the Containment of Nuclear Power Plants
  - 1.9.73.1 Regulatory Guide 1.73 Position
  - 1.9.73.2 VEGP Position
- 1.9.74 Regulatory Guide 1.74, February 1974, Quality Assurance Terms and Definitions
  - 1.9.74.1 Regulatory Guide 1.74 Position
  - 1.9.74.2 VEGP Position
- 1.9.75 Regulatory Guide 1.75, Revision 2, September 1978, Physical Independence of Electric Systems
  - 1.9.75.1 Regulatory Guide 1.75 Position1.9.75.2 VEGP Position
- 1.9.76 Regulatory Guide 1.76, April 1974, Design Basis Tornado for Nuclear Power Plants
  - 1.9.76.1 Regulatory Guide 1.76 Position
  - 1.9.76.2 VEGP Position
- 1.9.77 Regulatory Guide 1.77, May 1974, Assumptions Used for Evaluating a Control Rod Ejection Accident for Pressurized Water Reactors
  - 1.9.77.1 Regulatory Guide 1.77 Position
  - 1.9.77.2 VEGP Position
- 1.9.78 Regulatory Guide 1.78, Assumptions for Evaluating the Habitability of a Nuclear Power Plant Control Room During a Postulated Hazardous Chemical Release
  - 1.9.78.1 Regulatory Guide 1.78 Position 1.9.78.2 VEGP Position
- 1.9.79 Regulatory Guide 1.79, Revision 1, September 1975, Preoperational Testing of Emergency Core Cooling Systems for Pressurized Water Reactors
  - 1.9.79.1 Regulatory Guide 1.79 Position
  - 1.9.79.2 VEGP Position

- 1.9.80 Regulatory Guide 1.80, June 1974, Preoperational Testing of Instrument Air Systems
  - 1.9.80.1 Regulatory Guide 1.80 Position
- 1.9.81 Regulatory Guide 1.81, Revision 1, January 1975, Shared Emergency and Shutdown Electric Systems for Multi-Unit Nuclear Power Plants
  - 1.9.81.1 Regulatory Guide 1.81 Position
  - 1.9.81.2 VEGP Position
- 1.9.82 Regulatory Guide 1.82, June 1974, Sumps for Emergency Core Cooling and Containment Spray Systems
  - 1.9.82.1 Regulatory Guide 1.82 Position
  - 1.9.82.2 VEGP Position
- 1.9.83 Regulatory Guide 1.83, Revision 1, July 1975, Inservice Inspection of Pressurized Water Reactor Steam Generator Tubes
  - 1.9.83.1 Regulatory Guide 1.83 Position
  - 1.9.83.2 VEGP Position
- 1.9.84 Regulatory Guide 1.84, Revision 20, November 1982, Design and Fabrication Code Case Acceptability - ASME Section III, Division 1
  - 1.9.84.1 Regulatory Guide 1.84 Position
  - 1.9.84.2 VEGP Position
- 1.9.85 Regulatory Guide 1.85, Revision 20, November 1982, Materials Code Case Acceptability-ASME Section III, Division 1
  - 1.9.85.1 Regulatory Guide 1.85 Position 1.9.85.2 VEGP Position
- 1.9.86 Regulatory Guide 1.86, June 1974, Termination of Operating Licenses for Nuclear Reactors
  - 1.9.86.1 Regulatory Guide 1.86 Position
  - 1.9.86.2 VEGP Position
- 1.9.87 Regulatory Guide 1.87, Revision 1, June 1975, Guidance for Construction of Class 1 Components in Elevated Temperature Reactors
- 1.9.88 Regulatory Guide 1.88, Revision 2, October 1976, Collection, Storage, and Maintenance of Nuclear Power Plant Quality Assurance Records
  - 1.9.88.1 Regulatory Guide 1.88 Position

### TABLE OF CONTENTS (CONTINUED)

1.9.88.2 VEGP Position

- 1.9.89 Regulatory Guide 1.89, November 1974, Qualification of Class 1E Equipment for Nuclear Power Plants
  - 1.9.89.1 Regulatory Guide 1.89 Position
  - 1.9.89.2 VEGP Position
- 1.9.90 Regulatory Guide 1.90, Revision 1, August 1977, Inservice Inspection of Prestressed Concrete Containment Structures with Grouted Tendons

1.9.90.1 Regulatory Guide 1.90 Position

- 1.9.90.2 VEGP Position
- 1.9.91 Regulatory Guide 1.91, Revision 1, February 1978, Evaluations of Explosions Postulated To Occur on Transportation Routes near Nuclear Power Plants
  - 1.9.91.1 Regulatory Guide 1.91 Position
  - 1.9.91.2 VEGP Position
- 1.9.92 Regulatory Guide 1.92, Revision 1, February 1976, Combining Modal Responses and Spatial Components in Seismic Response Analysis
  - 1.9.92.1 Regulatory Guide 1.92 Position1.9.92.2 VEGP Position
- 1.9.93 Regulatory Guide 1.93, December 1974, Availability of Electric Power Supplies
  - 1.9.93.1 Regulatory Guide 1.93 Position
  - 1.9.93.2 VEGP Position
- 1.9.94 Regulatory Guide 1.94, Revision 1, April 1976, Quality Assurance Requirements for Installation, Inspection, and Testing of Structural Concrete and Structural Steel During the Construction Phase of Nuclear Power Plants
  - 1.9.94.1 Regulatory Guide 1.94 Position
  - 1.9.94.2 VEGP Position
- 1.9.95 Regulatory Guide 1.95, Revision 1, January 1977, Protection of Nuclear Power Plant Control Room Operators Against an Accidental Chlorine Release
  - 1.9.95.1 Regulatory Guide 1.95 Position
  - 1.9.95.2 VEGP Position
- 1.9.96 Regulatory Guide 1.96, Revision 1, June 1976, Design of Main Steam Isolation Valve Leakage Control Systems for Boiling Water Reactor Nuclear Power Plants

#### TABLE OF CONTENTS (CONTINUED)

- 1.9.97 Regulatory Guide 1.97, Revision 2, December 1980, Instrumentation for Light-Water-Cooled Nuclear Power Plants To Assess Plant Conditions During and Following an Accident
  - 1.9.97.1 Regulatory Guide 1.97 Position
  - 1.9.97.2 VEGP Position
- 1.9.98 .Regulatory Guide 1.98, March 1976, Assumptions Used for Evaluating the Potential Radiological Consequences of a Radioactive Offgas System Failure in a Boiling Water Reactor
- 1.9.99 Regulatory Guide 1.99, Revision 2, May 1988, Radiation Embrittlement of Reactor Vessel Materials
  - 1.9.99.1 Regulatory Guide 1.99 Position
  - 1.9.99.2 VEGP Position
- 1.9.100 Regulatory Guide 1.100, Revision 1, August 1977, Seismic Qualification of Electric Equipment for Nuclear Power Plants

1.9.100.1 Regulatory Guide 1.100 Position 1.9.100.2 VEGP Position

1.9.101 Regulatory Guide 1.101, Revision 3, August 1992, Emergency Planning for Nuclear Power Plants

1.9.101.1 Regulatory Guide 1.101 Position 1.9.101.2 VEGP Position

1.9.102 Regulatory Guide 1.102, September 1976, Flood Protection for Nuclear Power Plants

> 1.9.102.1 Regulatory Guide 1.102 Position 1.9.102.2 VEGP Position

1.9.103 Regulatory Guide 1.103, Revision 1, October 1976, Post-Tensioned Prestressing Systems for Concrete Reactor Vessels and Containments

> 1.9.103.1 Regulatory Guide 1.103 Position 1.9.103.2 VEGP Position

- 1.9.104 Regulatory Guide 1.104, Overhead Crane Handling for Nuclear Power Plants
- 1.9.105 Regulatory Guide 1.105, Revision 1, November 1976, Instrument Setpoints

1.9.105.1 Regulatory Guide 1.105 Position 1.9.105.2 VEGP Position

## TABLE OF CONTENTS (CONTINUED)

- 1.9.106 Regulatory Guide 1.106, Revision 1, March 1977, Thermal Overload Protection for Electric Motors on Motor-Operated Valves
  - 1.9.106.1 Regulatory Guide 1.106 Position
  - 1.9.106.2 VEGP Position
- 1.9.107 Regulatory Guide 1.107, Revision 1, February 1977, Qualifications for Cement Grouting for Prestressing Tendons in Containment Structures
  - 1.9.107.1 Regulatory Guide 1.107 Position 1.9.107.2 VEGP Position
- 1.9.108 Regulatory Guide 1.108, Revision 1, August 1977, Periodic Testing of Diesel Generator Units Used as Onsite Electric Power Systems at Nuclear Power Plants

1.9.108.1 Regulatory Guide 1.108 Position 1.9.108.2 VEGP Position

1.9.109 Regulatory Guide 1.109, Revision 1, October 1977, Calculation of Annual Doses to Man from Routine Releases of Reactor Effluents for the Purpose of Evaluating Compliance with 10 CFR 50, Appendix I

1.9.109.1 Regulatory Guide 1.109 Position 1.9.109.2 VEGP Position

1.9.110 Regulatory Guide 1.110, March 1976, Cost-Benefit Analysis for Radwaste Systems for Light-Water-Cooled Nuclear Power Reactors

> 1.9.110.1 Regulatory Guide 1.110 Position 1.9.110.2 VEGP Position

1.9.111 .Regulatory Guide 1.111, Revision 1, July 1977, Methods for Estimating Atmospheric Transport and Dispersion of Gaseous Effluents in Routine Releases from Light-Water-Cooled Reactors

> 1.9.111.1 Regulatory Guide 1.111 Position 1.9.111.2 VEGP Position

1.9.112 Regulatory Guide 1.112, Revision O-R, April 1976, Calculation of Releases of Radioactive Materials in Gaseous and Liquid Effluents from Light-Water-Cooled Power Reactors

1.9.112.1 Regulatory Guide 1.112 Position1.9.112.2 VEGP Position

1.9.113 Regulatory Guide 1.113, Revision 1, April 1977, Estimating Aquatic Dispersion of Effluents from Accidental and Routine Reactor Releases for the Purpose of Implementing Appendix I

### TABLE OF CONTENTS (CONTINUED)

1.9.113.1 Regulatory Guide 1.113 Position 1.9.113.2 VEGP Position

1.9.114 Regulatory Guide 1.114, Revision 1, November 1976, Guidance on Being Operator at the Controls of a Nuclear Power Plant

> 1.9.114.1 Regulatory Guide 1.114 Position 1.9.114.2 VEGP Position

1.9.115 Regulatory Guide 1.115, Revision 1, July 1977, Protection Against Low Trajectory Turbine Missiles

> 1.9.115.1 Regulatory Guide 1.115 Position 1.9.115.2 VEGP Position

1.9.116 Regulatory Guide 1.116, Revision O-R, June 1976, Quality Assurance Requirements for Installation, Inspection, and Testing of Mechanical Equipment and Systems

1.9.116.1 Regulatory Guide 1.116 Position 1.9.116.2 VEGP Position

1.9.117 Regulatory Guide 1.117, Revision 1, April 1978, Tornado Design Classification

1.9.117.1 Regulatory Guide 1.117 Position 1.9.117.2 VEGP Position

1.9.118 Regulatory Guide 1.118, Revision 2, June 1978, Periodic Testing of Electric Power and Protection Systems

1.9.118.1 Regulatory Guide 1.118 Position 1.9.118.2 VEGP Position

- 1.9.119 Regulatory Guide 1.119
- 1.9.120 Regulatory Guide 1.120, Revision 1, November 1977, Fire Protection Guidelines for Nuclear Power Plants

1.9.120.1 Regulatory Guide 1.120 Position 1.9.120.2 VEGP Position

1.9.121 Regulatory Guide 1.121, August 1976, Bases for Plugging Degraded PWR Steam Generator Tubes

> 1.9.121.1 Regulatory Guide 1.121 Position 1.9.121.2 VEGP Position

## TABLE OF CONTENTS (CONTINUED)

- 1.9.122 Regulatory Guide 1.122, Revision 1, February 1978, Development of Floor Design Response Spectra for Seismic Design of Floor-Supported Equipment or Components
  - 1.9.122.1 Regulatory Guide 1.122 Position
  - 1.9.122.2 VEGP Position
- 1.9.123 Regulatory Guide 1.123, Revision 1, July 1977, Quality Assurance Requirements for Control of Procurement of Items and Services for Nuclear Power Plants
  - 1.9.123.1 Regulatory Guide 1.123 Position 1.9.123.2 VEGP Position
- 1.9.124 Regulatory Guide 1.124, Revision 1, January 1978, Service Limits and Loading Combinations for Class 1 Linear-Type Component Supports

1.9.124.1 Regulatory Guide 1.124 Position1.9.124.2 VEGP Position

1.9.125 Regulatory Guide 1.125, Revision 1, October 1978, Physical Models for Design and Operation of Hydraulic Structures and Systems for Nuclear Power Plants

> 1.9.125.1 Regulatory Guide 1.125 Position 1.9.125.2 VEGP Position

1.9.126 Regulatory Guide 1.126, Revision 1, March 1978, An Acceptable Model and Related Statistical Methods for the Analysis of Fuel Densification

> 1.9.126.1 Regulatory Guide 1.126 Position 1.9.126.2 VEGP Position

1.9.127 Regulatory Guide 1.127, Revision 1, March 1978, Inspection of Water-Control Structures Associated with Nuclear Power Plants

> 1.9.127.1 Regulatory Guide 1.127 Position 1.9.127.2 VEGP Position

1.9.128 Regulatory Guide 1.128, Revision 1, October 1978, Installation Design and Installation of Large Lead Storage Batteries for Nuclear Power Plants

1.9.128.1 Regulatory Guide 1.128 Position1.9.128.2 VEGP Position

1.9.129 Regulatory Guide 1.129, Revision 1, February 1978, Maintenance, Testing, and Replacement of Large Lead Storage Batteries for Nuclear Power Plants

1.9.129.1 Regulatory Guide 1.129 Position1.9.129.2 VEGP Position

#### TABLE OF CONTENTS (CONTINUED)

- 1.9.130 Regulatory Guide 1.130, Revision 1, October 1978, Service Limits and Loading Combinations for Class 1 Plate- and Shell-Type Component Supports
  - 1.9.130.1 Regulatory Guide 1.130 Position 1.9.130.2 VEGP Position
- 1.9.131 Regulatory Guide 1.131, August 1977, Qualification Tests of Electrical Cables, Field Splices, and Connections for Light-Water-Cooled Nuclear Power Plants
  - 1.9.131.1 Regulatory Guide 1.131 Position 1.9.131.2 VEGP Position
- 1.9.132 Regulatory Guide 1.132, Revision 1, March 1979, Site Investigations for Foundations of Nuclear Power Plants

1.9.132.1 Regulatory Guide 1.132 Position 1.9.132.2 VEGP Position

1.9.133 Regulatory Guide 1.133, Revision 1, May 1981, Loose-Part Detection Program for the Primary System of Light-Water-Cooled Reactors

> 1.9.133.1 Regulatory Guide 1.133 Position 1.9.133.2 VEGP Position

1.9.134 Regulatory Guide 1.134, April 1987, Medical Certification and Monitoring of Personnel Requiring Operator Licenses

> 1.9.134.1 Regulatory Guide 1.134 Position 1.9.134.2 VEGP Position

1.9.135 Regulatory Guide 1.135, September 1977, Normal Water Level and Discharge at Nuclear Power Plants

1.9.135.1 Regulatory Guide 1.135 Position 1.9.135.2 VEGP Position

1.9.136 Regulatory Guide 1.136, Revision 2, June 1981, Materials, Construction, and Testing of Concrete Containments

1.9.136.1 Regulatory Guide 1.136 Position 1.9.136.2 VEGP Position

1.9.137 Regulatory Guide 1.137, Revision 1, October 1979, Fuel-Oil Systems for Standby Diesel Generators

> 1.9.137.1 Regulatory Guide 1.137 Position 1.9.137.2 VEGP Position

#### TABLE OF CONTENTS (CONTINUED)

1.9.138 Regulatory Guide 1.138, April 1978, Laboratory Investigations of Soils for Engineering Analysis and Design of Nuclear Power Plants

1.9.138.1 Regulatory Guide 1.138 Position

- 1.9.138.2 VEGP Position
- 1.9.139 Regulatory Guide 1.139, May 1978, Guidance for Residual Heat Removal

1.9.139.1 Regulatory Guide 1.139 Position

- 1.9.139.2 VEGP Position
- 1.9.140 Regulatory Guide 1.140, Revision 1, October 1979, Design, Testing, and Maintenance Criteria for Normal Ventilation Exhaust System Air Filtration and Adsorption Units of Light-Water-Cooled Nuclear Power Plants

1.9.140.1 Regulatory Guide 1.140 Position 1.9.140.2 VEGP Position

1.9.141 Regulatory Guide 1.141, April 1978, Containment Isolation Provisions for Fluid Systems

> 1.9.141.1 Regulatory Guide 1.141 Position 1.9.141.2 VEGP Position

1.9.142 Regulatory Guide 1.142, October 1981, Revision 1, Safety-Related Concrete Structures for Nuclear Power Plants (Other Than Reactor Vessels and Containments)

> 1.9.142.1 Regulatory Guide 1.142 Position 1.9.142.2 VEGP Position

1.9.143 Regulatory Guide 1.143, Revision 1, October 1979, Design Guidance for Radioactive Waste Management Systems, Structures, and Components Installed in Light-Water-Cooled Nuclear Power Plants

> 1.9.143.1 Regulatory Guide 1.143 Position 1.9.143.2 VEGP Position

1.9.144 Regulatory Guide 1.144, Revision 1, September 1980, Auditing of Quality Assurance Programs for Nuclear Power Plants

> 1.9.144.1 Regulatory Guide 1.144 Position 1.9.144.2 VEGP Position

1.9.145 Regulatory Guide 1.145, August 1979, Atmospheric Dispersion Models for Potential Accident Consequence Assessments at Nuclear Power Plants

> 1.9.145.1 Regulatory Guide 1.145 Position 1.9.145.2 VEGP Position

### TABLE OF CONTENTS (CONTINUED)

- 1.9.146 Regulatory Guide 1.146, August 1980, Qualification of Quality Assurance Program Audit Personnel for Nuclear Power Plants
  - 1.9.146.1 Regulatory Guide 1.146 Position
  - 1.9.146.2 VEGP Position
- 1.9.147 Regulatory Guide 1.147, Revision 12, May 1999, Inservice Inspection Code Case Acceptability, ASME Section XI, Division 1
  - 1.9.147.1 Regulatory Guide 1.147 Position 1.9.147.2 VEGP Position
- 1.9.148 Regulatory Guide 1.148, March 1981, Functional Specification for Active Valve Assemblies in Systems Important to Safety in Nuclear Power Plants

1.9.148.1 Regulatory Guide 1.148 Position1.9.148.2 VEGP Position

- 1.9.149 Regulatory Guide 1.149, April 1987, Nuclear Power Plant Simulation Facilities for Use in Operator License Examinations
  - 1.9.149.1 Regulatory Guide 1.149 Position 1.9.149.2 VEGP Position
- 1.9.150 Regulatory Guide 1.150, February 1983, Ultrasonic Testing of Reactor Vessel Welds During Preservice and Inservice Examinations

1.9.150.1 Regulatory Guide 1.150 Position 1.9.150.2 VEGP Position

1.9.155 Regulatory Guide 1.155, August 1988, Station Blackout

1.9.155.1 Regulatory Guide 1.155 Position 1.9.155.2 VEGP Position

- 1.9.163 Regulatory Guide 1.163, September 1995, Performance Based Containment Leak Testing Program
  - 1.9.163.1 Regulatory Guide 1.163 Position
  - 1.9.163.2 VEGP Position
- 1.9.166 Regulatory Guide 1.166, Revision 0, March 1997, Pre-Earthquake Planning and Immediate Nuclear Power Plant Operator Post-Earthquake Actions
  - 1.9.166.1 Regulatory Guide 1.166 Position
  - 1.9.166.2 VEGP Position

## TABLE OF CONTENTS (CONTINUED)

- 1.9.167 Regulatory Guide 1.167, Revision 0, March 1997, Restart of a Nuclear Power Plant Shutdown by a Seismic Event
  - 1.9.167.1Regulatory Guide 1.167 Position1.9.167.2VEGP Position
- 2.0 SITE CHARACTERISTICS
- 2.1 Geography and Demography
  - 2.1.1 Site Location and Description

2.1.1.1 Reactor Coordinates
-----------------------------

- 2.1.1.2 Site Area Map
- 2.1.1.3 Boundaries for Establishing Effluent Release Limits
- 2.1.2 Exclusion Area Authority and Control

2.1.2.1	Authority
	,

- 2.1.2.2 Control of Activities Unrelated to Plant Operation
- 2.1.2.3 Arrangements for Traffic Control
- 2.1.2.4 Abandonment or Relocation of Roads
- 2.1.3 Population Distribution

2.1.3.1	Population Within 10 Miles
2.1.3.2	Population Between 10 and 50 Miles
2.1.3.3	Transient Population
2.1.3.4	Low Population Zone
2.1.3.5	Population Center
2.1.3.6	Population Density
2.1.3.7	Methodology

- 2.2 Identification of Potential Hazards in Site Vicinity
  - 2.2.1 Location of Nearby Industrial, Military, and Transportation Facilities
  - 2.2.2 Description

2.2.2.1	Facilities
2.2.2.2	Products and Materials
2.2.2.3	Pipelines
2.2.2.4	Waterways
2.2.2.5	Airports
2.2.2.6	Projections of Industrial Growth

2.2.3 Evaluation of Potential Accidents

# TABLE OF CONTENTS (CONTINUED)

- 2.2.3.1 Determination of Design Basis Events
- 2.2.3.2 Effects of Design Basis Event

## 2.3 Meteorology

2.3.1 Regional Climatology

2.3.1.2 Regional Meteorological Conditions for Design and Operating Bases (Severe Weather)

# 2.3.2 Local Meteorology

2.3.2.1 Norma	and Extreme Values	of Meteorological Parameters
---------------	--------------------	------------------------------

- 2.3.2.2 Potential Influence of the Plant and Its Facilities on Local Meteorology
- 2.3.2.3 Local Meteorological Conditions for Design and Operating Bases
- 2.3.3 Onsite Meteorological Measurements Programs
- 2.3.4 Short Term Diffusion Estimates

2.3.4.1	Short Term Releases
2.3.4.2	Long Term Releases
2.3.4.3	Determination of Stability Classifications
2.3.4.4	Analytical Methods for Dispersion Computations

2.3.5 Long Term Diffusion Estimates

2.3.5.1	Objective
2.3.5.2	Calculations

- 2.4 Hydrologic Engineering
  - 2.4.1 Hydrologic Description
    - 2.4.1.1 Site and Facilities
    - 2.4.1.2 Hydrosphere
  - 2.4.2 Floods

2.4.2.1	Flood History
2.4.2.2	Flood Design Considerations
2.4.2.3	Effects of Local Intense Precipitation

# TABLE OF CONTENTS (CONTINUED)

2.4.3 Probable Maximum Flooding on Streams and Rivers

2.4.3.1	<b>Probable Maximum</b>	Precipitation

- 2.4.3.2 Precipitation Losses
- 2.4.3.3 Runoff and Stream Course Models
- 2.4.3.4 Probable Maximum Flood Flow
- 2.4.3.5 Water Level Determinations
- 2.4.3.6 Coincident Wind Wave Activity
- 2.4.4 Potential Dam Failures

2.4.4.1	Dam Failure Permutations
2.4.4.2	Unsteady Flow Analysis of Potential Dam Failures
0110	Water Level at Diant Site

- 2.4.4.3 Water Level at Plant Site
- 2.4.5 Probable Maximum Surge and Seiche Flooding
- 2.4.6 Probable Maximum Tsunami Flooding
- 2.4.7 Ice Effects
- 2.4.8 Cooling Water Canals and Reservoirs

2.4.8.1	Canals
2.4.8.2	Reservoirs

- 2.4.9 Channel Diversions
- 2.4.10 Flood Protection Requirements
- 2.4.11 Cooling Water Supply

2.4.11.1	Low Flow in Streams
2.4.11.2	Low Water Resulting from Surges
2.4.11.3	Historical Low Water
2.4.11.4	Future Control
2.4.11.5	Plant Requirements
2.4.11.6	Heat Sink Dependability Requirements

2.4.12 Ground Water

2.4.12.1	Description and Onsite Use
2.4.12.2	Sources
2.4.12.3	Monitoring of Safeguard Requirements
2.4.12.4	Design Basis for Subsurface Hydrostatic Loading

2.4.13 Accidental Releases of Liquid Effluents in Ground and Surface Water
### TABLE OF CONTENTS (CONTINUED)

Liquid Effluents in Surface Waters

2.4.13.1	Consideration of Accidental Spill of Radioactive Material in Ground Water
2.4.13.2	Dispersion, Dilution, and Travel Times of Accidental Releases of

- 2.4.14 Technical Specifications and Emergency Operating Requirements
- 2.5 Geology, Seismology, and Geotechnical Engineering
  - 2.5.1 Basic Geologic and Seismic Information
    - 2.5.1.1 Regional Geology
    - 2.5.1.2 Site Geology
  - 2.5.2 Vibratory Ground Motion

- 2.5.2.2 Geologic Structures and Tectonic Activity
- 2.5.2.3 Correlation of Earthquake Activity with Geologic Structures or Tectonic Provinces
- 2.5.2.4 Maximum Earthquake Potential
- 2.5.2.5 Seismic Wave Transmission Characteristics of the Site
- 2.5.2.6 Safe Shutdown Earthquake
- 2.5.2.7 Operating Basis Earthquake
- 2.5.3 Surface Faulting
  - 2.5.3.1 Geologic Conditions of the Site
  - 2.5.3.2 Evidence of Fault Offset
  - 2.5.3.3 Earthquakes Associated with Capable Faults
  - 2.5.3.4 Investigations of Capable Faults
  - 2.5.3.5 Correlation of Epicenters with Capable Faults
  - 2.5.3.6 Description of Capable Faults
  - 2.5.3.7 Zone Requiring Detailed Faulting Investigation
  - 2.5.3.8 Results of Faulting Investigation
- 2.5.4 Stability of Subsurface Materials and Foundations
  - 2.5.4.1 Geologic Features
  - 2.5.4.2 Properties of Subsurface Materials
  - 2.5.4.3 Exploration
  - 2.5.4.4 Geophysical Surveys
  - 2.5.4.5 Excavation and Backfill
  - 2.5.4.6 Site Ground Water Conditions
  - 2.5.4.7 Response of Soil and Rock to Dynamic Loading
  - 2.5.4.8 Liquefaction Potential
  - 2.5.4.9 Earthquake Design Basis
  - 2.5.4.10 Static Stability

### TABLE OF CONTENTS (CONTINUED)

2.5.4.11 Design Criteri
-------------------------

- 2.5.4.12 Techniques To Improve Subsurface Conditions
- 2.5.4.13 Subsurface Instrumentation
- 2.5.4.14 Construction Notes
- 2.5.4.15 Standard Review Plan Evaluation
- 2.5.5 Stability of Slopes
  - 2.5.5.1 Slope Characteristics
  - 2.5.5.2 Design Criteria and Analysis
  - 2.5.5.3 Log of Borings
  - 2.5.5.4 Compacted Backfill
- 2.5.6 Embankments and Dams

### Appendix 2A Population Distribution Methodology

- 2A.1 Introduction
- 2A.2 Definitions
- 2A.3 Assumptions
- 2A.4 Materials
- 2A.5 Methodology
- 2A.6 Procedures

### Appendix 2B Geology

- 2B.1 Introduction
- 2B.2 Field Investigations
- 2B.3 Report of Geology and Foundation Conditions
  - 2B.3.1 Introduction
  - 2B.3.2 Summary of Conclusions
  - 2B.3.3 Geologic Conditions
  - 2B.3.4 Excavation and Foundation Considerations
  - 2B.3.5 Conclusions
- 3.0 DESIGN OF STRUCTURES, COMPONENTS, EQUIPMENT, AND SYSTEMS
- 3.1 Conformance with Nuclear Regulatory Commission General Design Criteria

### TABLE OF CONTENTS (CONTINUED)

- 3.1.1 Overall Requirements
- 3.1.2 Protection by Multiple Fission Product Barriers
- 3.1.3 Protection and Reactivity Control Systems
- 3.1.4 Fluid Systems
- 3.1.5 Reactor Containment
- 3.1.6 Fuel and Reactivity Control

### 3.2 Classification of Structures, Components, Equipment, and Systems

3.2.1 Seismic Classification

3.2.1.1	Definitions
3.2.1.2	Classifications

3.2.2 VEGP Classification System

3.2.2.1	Nuclear Safety Classifications
3.2.2.2	Seismic Classification
3.2.2.3	Codes and Standards
3.2.2.4	Clarification to VEGP Classification System

#### 3.3 Wind and Tornado Loadings

3.3.1 Wind Loadings

3.3.1.1	Design Wind Velocity	/

- 3.3.1.2 Determination of Applied Forces
- 3.3.2 Tornado Loadings

3.3.2.1	Applicable Design Parameters
---------	------------------------------

- 3.3.2.2 Determination of Forces on Structures
- 3.3.2.3 Effect of Failure of Structures or Components Not Designed for Tornado Loads
- 3.4 Water Level (Flood) Design
  - 3.4.1 Flood Protection
    - 3.4.1.1 Flood Protection Measures for Seismic Category 1 Structures

### TABLE OF CONTENTS (CONTINUED)

### 3.4.1.2 Permanent Dewatering System

3.4.2 Analytical and Test Procedures

#### 3.5 Missile Protection

3.5.1 Missile Selection and Description

- 3.5.1.2 Internally Generated Missiles (Inside Containment)
- 3.5.1.3 Turbine Missiles
- 3.5.1.4 Missiles Generated by Natural Phenomena
- 3.5.1.5 Missiles Generated by Events near the Site
- 3.5.1.6 Aircraft Hazards
- 3.5.1.7 Gravity-Generated Missiles
- 3.5.1.8 Standard Review Plan Evaluation
- 3.5.2 Structures, Systems, and Components To Be Protected from Externally Generated Missiles

- 3.5.2.2 Missile Barriers Within Containment
- 3.5.2.3 Barriers for Missiles Generated Outside of Plant Structures
- 3.5.2.4 Missile Barriers Within Plant Structures Other Than Containment
- 3.5.3 Barrier Design Procedures
- 3.6 Protection Against the Dynamic Effects Associated with the Postulated Rupture of Pipings
  - 3.6.1 Postulated Piping Failures in Fluid Systems Inside and Outside Containment

3.6.1.1	Design Bases
---------	--------------

- 3.6.1.2 Description
- 3.6.1.3 Safety Evaluation
- 3.6.2 Determination of Break Locations and Dynamic Effects Associated with the Postulated Rupture of Piping
  - 3.6.2.1 Criteria Used To Define High/Moderate-Energy Break/Crack Locations and Configurations
  - 3.6.2.2 Analytical Methods To Define Forcing Functions and Response Models
  - 3.6.2.3 Dynamic Analysis Methods To Verify Integrity and Operability
  - 3.6.2.4 Protective Assembly Design Criteria
  - 3.6.2.5 Material To Be Submitted for the Operating License Review

## TABLE OF CONTENTS (CONTINUED)

- 3.7 Seismic Design
  - 3.7.B Seismic Design

### 3.7.B.1 Seismic Input

3.7.B.1.1	Design Response Spectra
3.7.B.1.2	Design Time-History
3.7.B.1.3	Critical Damping Values
3.7.B.1.4	Supporting Media for Seismic Category 1
	Structures
3.7.B.1.5	Standard Review Plan Evaluation

### 3.7.N Seismic Design

3.7.N.2

3.7.N.1 Seismic Input

3.7.N.1.1	Design Response Spectra
3.7.N.1.2	Design Time-History
3.7.N.1.3	Critical Damping Values
3.7.N.1.4	Supporting Media for Seismic Category 1
	Structures

### 3.7.B.2 Seismic System Analysis

3.7.B.2.1	Seismic Analysis Methods
3.7.B.2.2	Natural Frequencies and Response Loads
3.7.B.2.3	Procedures Used for Modeling
3.7.B.2.4	Soil-Structure Interaction
3.7.B.2.5	Development of Floor Response Spectra
3.7.B.2.6	Three Components of Earthquake Motion
3.7.B.2.7	Combination of Modal Responses
3.7.B.2.8	Interaction of Non-Category 1 Structures with
	Seismic Category 1 Structures
3.7.B.2.9	Effects of Parameter Variations on Floor Response
	Spectra
3.7.B.2.10	Use of Constant Vertical Static Factors
3.7.B.2.11	Method Used To Account for Torsional Effects
3.7.B.2.12	Comparison of Responses
3.7.B.2.13	Methods for Seismic Analysis of Dams
3.7.B.2.14	Determination of Seismic Category 1 Structure
	Overturning Moments
3.7.B.2.15	Analysis Procedure for Damping
3.7.B.2.16	Standard Review Plan Evaluation
Solemic Sve	stem Analysis
Seisinic Sys	Signi Analysis
37N21	Seismic Analysis Methods

3.7.N.2.2 Natural Frequencies and Response Loads

### TABLE OF CONTENTS (CONTINUED)

- 3.7.N.2.3 Procedures Used for Modeling
- 3.7.N.2.4 Soil-Structure Interaction
- 3.7.N.2.5 Development of Floor Response Spectra
- 3.7.N.2.6 Three Components of Earthquake Motion
- 3.7.N.2.7 Combination of Modal Responses
- 3.7.N.2.8 Interaction of Non-Category 1 Structures with Seismic Category 1 Structures
- 3.7.N.2.9 Effects of Parameter Variations on Floor Response Spectra
- 3.7.N.2.10 Use of Constant Vertical Static Factors
- 3.7.N.2.11 Method Used To Account for Torsional Effects
- 3.7.N.2.12 Comparison of Responses
- 3.7.N.2.13 Methods for Seismic Analysis of Dams
- 3.7.N.2.14 Determination of Seismic Category 1 Structure Overturning Moments
- 3.7.N.2.15 Analysis Procedure for Damping
- 3.7.N.2.16 Standard Review Plan Evaluation

#### 3.7.B.3 Seismic Subsystem Analysis

- 3.7.B.3.1 Seismic Analysis Methods
- 3.7.B.3.2 Determination of Number of Earthquake Cycles
- 3.7.B.3.3 Procedure Used for Modeling
- 3.7.B.3.4 Basis for Selection of Frequencies
- 3.7.B.3.5 Use of Equivalent Static Load Method of Analysis
- 3.7.B.3.6 Three Components of Earthquake Motion
- 3.7.B.3.7 Combination of Modal Responses
- 3.7.B.3.8 Analytical Procedures for Piping Systems
- 3.7.B.3.9 Multiple Supported Equipment and Components with Distinct Inputs
- 3.7.B.3.10 Use of Constant Vertical Static Factors
- 3.7.B.3.11 Torsional Effects of Eccentric Masses
- 3.7.B.3.12 Buried Seismic Category 1 Piping Systems and Tunnels
- 3.7.B.3.13 Interaction of Other Piping with Seismic Category 1 Piping
- 3.7.B.3.14 Seismic Analyses for Reactor Internals
- 3.7.B.3.15 Analysis Procedure for Damping

#### 3.7.N.3 Seismic Subsystem Analysis

- 3.7.N.3.1 Seismic Analysis Methods
- 3.7.N.3.2 Determination of Number of Earthquake Cycles
- 3.7.N.3.3 Procedure Used for Modeling
- 3.7.N.3.4 Basis for Selection of Frequencies
- 3.7.N.3.5 Use of Equivalent Static Load Method of Analysis
- 3.7.N.3.6 Three Components of Earthquake Motion
- 3.7.N.3.7 Combination of Modal Responses
- 3.7.N.3.8 Analytical Procedures for Piping

### TABLE OF CONTENTS (CONTINUED)

3.7.N.3.9	Multiple Supported Equipment Components with
	Distinct Inputs
3.7.N.3.10	Use of Constant Vertical Static Factors
3.7.N.3.11	Torsional Effects of Eccentric Masses
3.7.N.3.12	Buried Seismic Category 1 Piping Systems and
	Tunnels
3.7.N.3.13	Interaction of Other Piping with Seismic Category 1
	Piping
3.7.N.3.14	Seismic Analyses for Reactor Internals
3.7.N.3.15	Analysis Procedure for Damping

3.7.4 Seismic Instrumentation

3.7.4.1	Comparison with Regulatory Guide 1.12
---------	---------------------------------------

- 3.7.4.2 Location and Description of Instrumentation
- 3.7.4.3 Control Room Operator Notification
- 3.7.4.4 Comparison of Measured and Predicted Responses
- 3.8 Design of Category 1 Structures
  - 3.8.1 Concrete Containment
    - 3.8.1.1 Description of the Containment
    - 3.8.1.2 Applicable Codes, Standards, and Specifications
    - 3.8.1.3 Loads and Load Combinations
    - 3.8.1.4 Design and Analysis Procedures
    - 3.8.1.5 Structural Acceptance Criteria
    - 3.8.1.6 Materials, Quality Control, and Special Construction Techniques
    - 3.8.1.7 Testing and Inservice Inspection Requirements
    - 3.8.1.8 Standard Review Plan Evaluation
  - 3.8.2 Steel Containment
    - 3.8.2.1 Description of Penetration Sleeve Assemblies, Class MC
      - Components and Parts, and Class 2 Piping Penetrations
    - 3.8.2.2 Applicable Codes, Standards, and Specifications
    - 3.8.2.3 Loads and Load Combinations
    - 3.8.2.4 Design and Analysis Procedures
    - 3.8.2.5 Structural Acceptance Criteria
    - 3.8.2.6 Materials, Quality Control, and Special Construction Techniques
    - 3.8.2.7 Testing and Inservice Inspection Requirements
    - 3.8.2.8 Standard Review Plan Evaluation
- 3.8.3 Concrete and Steel Internal Structures of Concrete Containment
  - 3.8.3.1 Description of the Internal Structures
  - 3.8.3.2 Applicable Codes, Standards, and Specifications
  - 3.8.3.3 Loads and Load Combinations

- 3.8.3.4 Design and Analysis Procedures
- 3.8.3.5 Structural Acceptance Criteria
- 3.8.3.6 Materials, Quality Control, and Special Construction Techniques
- 3.8.3.7 Testing and Inservice Inspection Requirements
- 3.8.3.8 Standard Review Plan Evaluation
- 3.8.4 Other Seismic Category 1 Structures
  - 3.8.4.1 Description of the Structures
  - 3.8.4.2 Applicable Codes, Standards, and Specifications
  - 3.8.4.3 Loads and Load Combinations
  - 3.8.4.4 Design and Analysis Procedures
  - 3.8.4.5 Structural Acceptance Criteria
  - 3.8.4.6 Materials, Quality Control, and Special Construction Techniques
  - 3.8.4.7 Testing and Inservice Inspection Requirements
  - 3.8.4.8 Standard Review Plan Evaluation
- 3.8.5 Foundations

3.8.5.1	Description of the Foundations
---------	--------------------------------

- 3.8.5.2 Applicable Codes, Standards, and Specifications
- 3.8.5.3 Loads and Load Combinations
- 3.8.5.4 Design and Analysis Procedures
- 3.8.5.5 Structural Acceptance Criteria
- 3.8.5.6 Materials, Quality Control, and Special Construction Techniques
- 3.8.5.7 Testing and Inservice Inspection Requirements
- 3.8.5.8 Standard Review Plan Evaluation
- 3.9 Mechanical Systems and Components
  - 3.9.B Mechanical Systems and Components
    - 3.9.B.1 Special Topics for Mechanical Components
      - 3.9.B.1.1 Design Transients
      - 3.9.B.1.2 Computer Programs Used in Analysis
      - 3.9.B.1.3 Experimental Stress Analysis
      - 3.9.B.1.4 Considerations for the Evaluation of the Faulted Condition
      - 3.9.B.1.5 Weld Inspection Criteria
    - 3.9.N.1 Special Topics for Mechanical Components
      - 3.9.N.1.1 Design Transients
      - 3.9.N.1.2 Computer Programs Used in Analysis
      - 3.9.N.1.3 Experimental Stress Analysis
      - 3.9.N.1.4 Considerations for the Evaluation of the Faulted Condition

3.9.B.2 Dynamic Testing and Analysis		esting and Analysis
	3.9.B.2.1	Piping Vibration, Thermal Expansion, and Dynamic
	3.9.B.2.2	Seismic Qualification Testing of Safety-Related
	3.9.B.2.3	Dynamic Response Analysis of Reactor Internals Operational Flow Transients and Steady-State
	3.9.B.2.4	Preoperational Flow-Induced Vibration Testing of
	3.9.B.2.5	Dynamic System Analysis of the Reactor Internals
	3.9.B.2.6	Correlations of Reactor Internals Vibration Tests with the Analytical Results
3.9.N.2 Dynamic Testing and Analysis		esting and Analysis
	3.9.N.2.1	Piping Vibration, Thermal Expansion, and Dynamic Effects
	3.9.N.2.2	Seismic Qualification Testing of Safety-Related
	3.9.N.2.3	Dynamic Response Analysis of Reactor Internals Under Operational Flow Transients and Steady- State Conditions
	3.9.N.2.4	Preoperational Flow-Induced Vibration Testing of Reactor Internals
	3.9.N.2.5	Dynamic System Analysis of the Reactor Internals Under Faulted Condition
	3.9.N.2.6	Correlation of Reactor Internals Vibration Tests with the Analytical Results
3.9.B.3 American Society of Mechanica 3 Components, Component Su Structures		ociety of Mechanical Engineers Code Class 1, 2, and ents, Component Supports, and Core Support
	3.9.B.3.1	Loading Combinations, Design Transients, and Stress Limits
	3.9.B.3.2	Pump and Valve Operability Assurance
	3.9.B.3.3	Design and Installation Criteria, Pressure-Relieving Devices
	3.9.B.3.4 3.9.B.3.5	Component Supports Standard Review Plan Evaluation
3.9.N.3	American S 3 Compone Structures	ociety of Mechanical Engineers Code Class 1, 2, and ents, Component Supports, and Core Support

3.9.N.3.1	Loading Combinations, Design Transients, and Stress Limits (For ASME Code Class 2 and 3
	Components)
3.9.N.3.2	Pump and Valve Operability Assurance
3.9.N.3.3	Design and Installation Details in Mounting of
	Pressure Relief Devices
3.9.N.3.4	Component Supports
20N25	Standard Bayiow Blan Evaluation

- 3.9.N.3.5 Standard Review Plan Evaluation
- 3.9.4 Control Rod Drive System
  - 3.9.4.1 Descriptive Information of CRDS
  - 3.9.4.2 Applicable CRDS Design Specifications
  - 3.9.4.3 Design Loads, Stress Limits, and Allowable Deformations
  - 3.9.4.4 CRDM Performance Assurance Program
- 3.9.5 Reactor Pressure Vessel Internals
  - 3.9.5.1 Design Arrangements
  - 3.9.5.2 Design Loading Conditions
  - 3.9.5.3 Design Bases
- 3.9.6 Inservice Testing of Pumps and Valves
  - 3.9.6.1 Inservice Testing of Pumps
  - 3.9.6.2 Inservice Testing of Valves
  - 3.9.6.3 Relief Request
- 3.10 Seismic Qualification of Seismic Category 1 Mechanical and Electrical Equipment
  - 3.10.B.1 Seismic Qualification Criteria
  - 3.10.N Seismic and Dynamic Qualification of Seismic Category 1 Mechanical and Electrical Equipment
    - 3.10.N.1 Seismic and Dynamic Qualification Criteria
      - 3.10.N.1.1 Qualification Standards
      - 3.10.N.1.2 Performance Requirements for Seismic Qualification
      - 3.10.N.1.3 Acceptance Criteria
    - 3.10.B.2 Methods and Procedures for Qualifying Mechanical and Electrical Equipment
      - 3.10.B.2.1 Means of Qualification
      - 3.10.B.2.2 Method of Qualification

### TABLE OF CONTENTS (CONTINUED)

3.10.N.2	Methods and Procedures for Qualifying Electrical Equipment, Instrumentation, and Mechanical Components		
	3.10.N.2.1	Seismic Qualification of Instrumentation and Electrical Equipment	
	3.10.N.2.2	Seismic and Operability Qualification of Active	
	3.10.N.2.3	Seismic Qualification of Other Seismic Category 1 Mechanical Equipment	
3.10.B.3	Methods an Mechanical	d Procedures of Analysis or Testing of Supports of and Electrical Equipment and Instrumentation	
3.10.N.3	Method and Equipment,	Procedures for Qualifying Supports of Electrical Instrumentation, and Mechanical Components	
	3.10.N.3.1	Test Configurations	
3.10.B.4	Operating L	icense Review	
	3.10.B.4.1 3.10.B.4.2	Qualification and Documentation Procedures Standard Review Plan Evaluation	
3.10.N.4	Operating L	icense Review	
	3.10.N.4.1 3.10.N.4.2	Documentation Standard Review Plan Evaluation	

3.11 Environmental Design of Mechanical and Electrical Equipment

3.11.B.1	Equipment Identification and Environmental Conditions

3.11.B.1.1 Standard Review Plan Evaluation 3.11.B.1.2 Reference

#### 3.11.N Environmental Design of Mechanical and Electrical Equipment

3.11.N.1	Equipment Identification and Environmental Conditions		
	3.11.N.1.1	Equipment Identification	
	3.11.N.1.2	Definition of Environmental Conditions	
	3.11.N.1.3	Equipment Operability Times	
	3.11.N.1.4	Standard Review Plan Evaluation	
3.11.B.2	Qualification	n Tests and Analyses	

### TABLE OF CONTENTS (CONTINUED)

- 3.11.N.2 Qualification Tests and Analyses
  - 3.11.N.2.1 Environmental Qualification Criteria
  - 3.11.N.2.2 Environmental Design of Mechanical Equipment
  - 3.11.N.2.3 Environmental Design of Electrical Equipment
- 3.11.B.3 Qualification Test Results
- 3.11.N.3 Qualification Program Results
- 3.11.4 Loss of Ventilation
- 3.11.5 Estimated Chemical and Radiation Environment

3.11.5.1	Chemical Environment
3.11.5.2	Radiation Environment

Appendix 3A Conformance with Regulatory Guides

Appendix 3B Computer Programs Used for Structural, Seismic, and Geotechnical Analyses

- 3B.1 Computer Programs Used for Structural and Seismic Analyses by Bechtel Power Corporation 3B.1.1 Bechtel CE 201 (CE 217), Bechtel Structural Analysis Program, Post Processor (BSAP-POST) 3B.1.2 Bechtel CE 207, BSAP-DYNAM 3B.1.3 Bechtel CE 239, Hemispherical Dome Tendon Analysis (TENDON) 3B.1.4 Bechtel CE 251 3D, Composite Modal Damping (GEMD) 3B.1.5 Bechtel CE 450, Turbine Missile Probability (TURMIS) 3B.1.6 Bechtel CE 800, Bechtel Structural Analysis Program (BSAP) 3B.1.7 Bechtel CE 802, Response Spectra Analysis (SPECTRA) 3B.1.8 Bechtel CE 982, Continuum Linear Analysis for Soil Structure Interaction (CLASSI) 3B.1.9 Bechtel CE 915, A Computer Program for Earthquake Response Analysis of Horizontally Layered Sites (SHAKE)
  - 3B.1.10 Bechtel ME 351, Pipe Force and Whip Analysis (PRTHRUST/PIPERUP)

### TABLE OF CONTENTS (CONTINUED)

3B.1.11	Bechtel TE 301, Two-Dimensional Structural Analysis (MFRAME)
3B.1.12	FLUSH (Control Data Corporation Version)
3B.1.13	The Structural Design Language (ICES-STRUDL-II McDonnell-Douglas Automation Version)
3B.1.14	ICES-LEASE (McDonnell-Douglas Automation Version)
3B.1.15	ICES-SEPOL (McDonnell-Douglas Automation Version)
3B.1.16	Bechtel CE 212 BSAP-PRE
3B.1.17	Other Computer Programs Used in Structural Analyses

Appendix 3C Design of Structures for Tornado Missile Impact

### 3C.1 Introduction

- 3C.1.1 Procedures
- 3C.2 Local Effects
  - 3C.2.1Reinforced Concrete Elements3C.2.2Steel Elements
- 3C.3 Structural Response Due to Missile Impact Loading

3C.3.1	General
3C.3.2	Structural Assessment

Appendix 3D Seismic Response Spectra

- 3D.1 Seismic Response Spectra
- 3D.2 Seismic Analysis Methods
  - 3D.2.1 Confirmatory Study

3D.2.1.1	Introduction
3D.2.1.2	Soil-Structure Interaction Analyses Per S3 and S4 Method
3D.2.1.3	Soil-Structure Interaction Analysis Using the Impedance Method
3D.2.1.4	Comparison of Response Spectra from the S3 and S4 Method
	and the Impedance Method
3D.2.1.5	Conclusions

3D.2.2 Sensitivity Study

### TABLE OF CONTENTS (CONTINUED)

ntroduction

- 3D.2.2.2 Soil Profile
- 3D.2.2.3 Ground Motions
- 3D.2.2.4 Variation of Peak Ground Acceleration with Depth
- 3D.2.2.5 Comparison of Ground Surface Motions with those Developed at 76 ft Depth
- 3D.2.2.6 Conclusion
- 3D.2.3 Methodology to Account for Torsion Caused by the Seismic Wave Propagation Effects

3D.2.3.1	Introduction
3D.2.3.2	Safety-Related Structures
3D.2.3.3	Safety-Related Equipment, Systems, and Components

3D.3 Seismic Response Spectra Comparison

Appendix 3E Impedance Functions for an Arbitrarily Shaped Foundation on a Layered Medium

- Appendix 3F Hazards Analysis
- 3F.1 Introduction
- 3F.2 Analysis Assumptions
  - 3F.2.1 Earthquake Analysis Assumptions
  - 3F.2.2 Pipe Break Analysis Assumptions
  - 3F.2.3 Missiles Analysis Assumptions
  - 3F.2.4 Flooding Analysis Assumptions
- 3F.3 Protection Mechanisms
- 3F.4 Hazards Evaluations
  - 3F.4.1 Auxiliary Feedwater Pump Rooms 3F.4.2 Main Steam Isolation Valve and Main Feedwater Isolation Valve Compartment 3F.4.3 Evaluation of Reactor Coolant System Loop Branch Line Breaks 3F.4.4 Turbine Building Flooding Evaluation
- 4.0 REACTOR
- 4.1 Summary Description
- 4.2 Fuel System Design

#### TABLE OF CONTENTS (CONTINUED)

4.2.1 Design Bases

4.2.1.1	Cladding
---------	----------

- 4.2.1.2 Fuel Material
- 4.2.1.3 Fuel Rod Performance
- 4.2.1.4 Spacer Grids
- 4.2.1.5 Fuel Assembly Structural Design
- 4.2.1.6 Incore Control Components
- 4.2.1.7 Surveillance Program
- 4.2.2 Description and Design Drawings

4.2.2.1	Fuel Rods
	1 4011 (040

- 4.2.2.2 Fuel Assembly Structure
- 4.2.2.3 Incore Control Components
- 4.2.3 Design Evaluation

4.2.3.1	Cladding

- 4.2.3.2 Fuel Materials Considerations
- 4.2.3.3 Fuel Rod Performance
- 4.2.3.4 Spacer Grids
- 4.2.3.5 Fuel Assembly
- 4.2.3.6 Reactivity Control Assembly and Burnable Absorber Rods
- 4.2.4 Testing and Inspection Plan
  - 4.2.4.1 Quality Assurance Program
  - 4.2.4.2 Quality Control
  - 4.2.4.3 Online Fuel Failure Monitoring
  - 4.2.4.4 Incore Control Component Testing and Inspection
  - 4.2.4.5 Tests and Inspections by Others
  - 4.2.4.6 Inservice Surveillance
  - 4.2.4.7 Onsite Inspection
- 4.2.5 Westinghouse Creep/Growth Test in VEGP Unit 2
- 4.3 Nuclear Design
  - 4.3.1 Design Bases

4.3.1.1	Fuel Burnup
4.3.1.2	Negative Reactivity Feedbacks (Reactivity Coefficient)
4.3.1.3	Control of Power Distribution
4.3.1.4	Maximum Controlled Reactivity Insertion Rate
4.3.1.5	Shutdown Margins
4.3.1.6	Stability

### TABLE OF CONTENTS (CONTINUED)

4.3.1.7 Anticipated Transients Without SCRAM

#### 4.3.2 Description

4.3.2.1	Nuclear Design Description
4.3.2.2	Power Distribution
4.3.2.3	Reactivity Coefficients
4.3.2.4	Control Requirements
4.3.2.5	Control Rod Patterns and Reactivity Worths
4.3.2.6	Criticality of the Reactor During Refueling
4.3.2.7	Stability
4.3.2.8	Vessel Irradiation

4.3.3 Analytical Methods

4.3.3.1	Fuel Temperature (Doppler) Calculations
4.3.3.2	Macroscopic Group Constants
4.3.3.3	Spatial Few-Group Diffusion Calculations

- 4.4 Thermal and Hydraulic Design
  - 4.4.1 Design Bases
    - 4.4.1.1 Departure from Nucleate Boiling Design Basis
    - 4.4.1.2 Fuel Temperature Design Basis
    - 4.4.1.3 Core Flow Design Basis
    - 4.4.1.4 Hydrodynamic Stability Design Basis
    - 4.4.1.5 Other Considerations
  - 4.4.2 Description of Thermal and Hydraulic Design of the Reactor Core

4.4.2.1	Summary Comparison
4 4 9 9	

- 4.4.2.2 Critical Heat Flux Radio or DNBR and Mixing Technology
- 4.4.2.3 Linear Heat Generation Rate
- 4.4.2.4 Void Fraction Distribution
- 4.4.2.5 Core Coolant Flow Distribution
- 4.4.2.6 Core Pressure Drops and Hydraulic Loads
- 4.4.2.7 Correlation and Physical Data
- 4.4.2.8 Thermal Effects of Operational Transients
- 4.4.2.9 Uncertainties in Estimates
- 4.4.2.10 Flux Tilt Considerations
- 4.4.2.11 Fuel and Cladding Temperatures
- 4.4.3 Description of the Thermal and Hydraulic Design of the RCS

4.4.3.1	Plant Configuration Data
---------	--------------------------

- 4.4.3.2 Operating Restrictions on Pumps
- 4.4.3.3 Power-Flow Operating Map (Boiling Water Reactor)

### TABLE OF CONTENTS (CONTINUED)

- 4.4.3.4 Temperature-Power Operating Map (PWR)
- 4.4.3.5 Load Following Characteristics
- 4.4.3.6 Thermal and Hydraulic Characteristics Summary Table

### 4.4.4 Evaluation

4.4.4.1	Critical Heat Flux
	• · · · ·

- 4.4.4.2 Core Hydraulics
- 4.4.4.3 Influence of Power Distribution
- 4.4.4.4 Core Thermal Response
- 4.4.4.5 Analytical Methods
- 4.4.4.6 Hydrodynamic and Flow Power Coupled Instability
- 4.4.4.7 Fuel Rod Behavior Effects from Coolant Flow Blockage
- 4.4.5 Testing and Verification
  - 4.4.5.1 Tests Prior to Initial Criticality
  - 4.4.5.2 Initial Power and Plant Operation
  - 4.4.5.3 Component and Fuel Inspections
- 4.4.6 Instrumentation Requirements

4.4.6.1	Incore Instrumentation
4.4.6.2	Overtemperature and Overpower <i>Δ</i> T Instrumentation
4.4.6.3	Instrumentation To Limit Maximum Power Output
4.4.6.4	Digital Metal Impact Monitoring System

#### 4.5 Reactor Materials

- 4.5.1 Control Rod Drive System Structural Materials
  - 4.5.1.1 Materials Specifications
  - 4.5.1.2 Fabrication and Processing of Austenitic Stainless Steel Components
  - 4.5.1.3 Contamination Protection and Cleaning of Austenitic Stainless Steel
  - 4.5.1.4 Other Materials
- 4.5.2 Reactor Internals Materials
  - 4.5.2.1 Materials Specifications
  - 4.5.2.2 Controls on Welding
  - 4.5.2.3 Nondestructive Examination of Tubular Products and Fittings
  - 4.5.2.4 Fabrication and Processing of Austenitic Stainless Steel Components
    - 4.5.2.5 Contamination Protection and Cleaning of Austenitic Stainless Steel

- 4.6 Functional Design of Reactivity Control Systems
  - 4.6.1 Information for Control Rod Drive System
  - 4.6.2 Evaluations of the CRDS
  - 4.6.3 Testing and Verification of the CRDS
  - 4.6.4 Information for Combined Performance of Reactivity Systems
  - 4.6.5 Evaluation of Combined Performance
- 4A Response to NUREG-0737, II.F.2, Instrumentation for Detection of Inadequate Core Cooling
- 5.0 REACTOR COOLANT SYSTEM AND CONNECTED SYSTEMS
- 5.1 Summary Description
  - 5.1.1 Design Bases
  - 5.1.2 Design Description
  - 5.1.3 System Components
  - 5.1.4 System Performance Characteristics
- 5.2 Integrity of Reactor Coolant Pressure Boundary
  - 5.2.1 Compliance with Codes and Code Cases
    - 5.2.1.1 Compliance with 10 CFR 50.55a
    - 5.2.1.2 Applicable Code Cases
  - 5.2.2 Overpressure Protection
    - 5.2.2.1 Design Bases
    - 5.2.2.2 Design Evaluation
    - 5.2.2.3 Piping and Instrumentation Diagrams
    - 5.2.2.4 Equipment and Component Description
    - 5.2.2.5 Mounting of Pressure Relief Devices
    - 5.2.2.6 Applicable Codes and Classification
    - 5.2.2.7 Material Specifications
    - 5.2.2.8 Process Instrumentation
    - 5.2.2.9 System Reliability
    - 5.2.2.10 RCS Pressure Control During Low-Temperature Operation
    - 5.2.2.11 Consequences of a Postulated Loss of a dc Bus Coupled with a Single Failure Disabling a PORV Allowing a Cold Pressurization Event

### TABLE OF CONTENTS (CONTINUED)

5.2.2.12 Testing and Inspection

#### 5.2.3 Reactor Coolant Pressure Boundary Materials

- 5.2.3.1 Material Specifications
- 5.2.3.2 Compatibility with Reactor Coolant
- 5.2.3.3 Fabrication and Processing of Ferritic Materials
- 5.2.3.4 Fabrication and Processing of Austenitic Stainless Steel
- 5.2.4 Inservice Inspection and Testing of Reactor Coolant Pressure Boundary
  - 5.2.4.1 System Boundary Subject to Inspection
  - 5.2.4.2 Arrangement and Accessibility
  - 5.2.4.3 Examination Techniques and Procedures
  - 5.2.4.4 Inspection Intervals
  - 5.2.4.5 Examination Categories and Requirements
  - 5.2.4.6 Evaluation of Examination Results
  - 5.2.4.7 System Leakage and Hydrostatic Pressure Tests
- 5.2.5 Detection of Leakage Through Reactor Coolant Pressure Boundary
  - 5.2.5.1 Design Bases
  - 5.2.5.2 Identified Intersystem Leakage Detection
  - 5.2.5.3 Unidentified Leakage Detection
  - 5.2.5.4 Safety Evaluation
  - 5.2.5.5 Tests and Inspections
  - 5.2.5.6 Instrumentation Applications

#### 5.3 Reactor Vessel

- 5.3.1 Reactor Vessel Materials
  - 5.3.1.1 Material Specifications
  - 5.3.1.2 Special Processes Used for Manufacturing and Fabrication
  - 5.3.1.3 Special Methods for Nondestructive Examination
  - 5.3.1.4 Special Controls for Ferritic and Austenitic Stainless Steels
  - 5.3.1.5 Fracture Toughness
  - 5.3.1.6 Material Surveillance
  - 5.3.1.7 Reactor Vessel Fasteners
- 5.3.2 Pressure-Temperature Limits
  - 5.3.2.1 Limit Curves
  - 5.3.2.2 Operating Procedures
- 5.3.3 Reactor Vessel Integrity
  - 5.3.3.1 Design

5.3.3.2	Materials of	Construction

- 5.3.3.3 Fabrication Methods
- 5.3.3.4 Inspection Requirements
- 5.3.3.5 Shipment and Installation
- 5.3.3.6 Operating Conditions
- 5.3.3.7 Inservice Surveillance
- 5.4 Component and Subsystem Design
  - 5.4.1 Reactor Coolant Pump Assembly
    - 5.4.1.1 Design Bases
    - 5.4.1.2 Pump Assembly Description
    - 5.4.1.3 Design Evaluation
    - 5.4.1.4 Tests and Inspections
    - 5.4.1.5 Pump Flywheel
  - 5.4.2 Steam Generators
    - 5.4.2.1 Design Bases
    - 5.4.2.2 Design Description
    - 5.4.2.3 Design Evaluation
    - 5.4.2.4 Steam Generator Materials
    - 5.4.2.5 Steam Generator Inservice Inspection
    - 5.4.2.6 Quality Assurance
  - 5.4.3 Reactor Coolant Piping
    - 5.4.3.1 Design Bases
    - 5.4.3.2 Design Description
    - 5.4.3.3 Design Evaluation
    - 5.4.3.4 Tests and Inspections
  - 5.4.4 Main Steam Line Flow Restrictions

5.4.4.1	Design Bases
5.4.4.2	Design Description
5.4.4.3	Design Evaluation
5.4.4.4	Inspections

- 5.4.5 Materials and Inspections
- 5.4.6 Reactor Vessel Design Data
- 5.4.7 Residual Heat Removal System

5.4.7.1	Design Bases
5.4.7.2	System Design

### TABLE OF CONTENTS (CONTINUED)

5.4.7.3	Performance Evaluation
5.4.7.4	Preoperational Testing
<b>F 4 7 F</b>	Dellah lite Teste and have a

- 5.4.7.5 Reliability Tests and Inspections
- 5.4.8 Reactor Water Cleanup System
- 5.4.9 Main Steam Line and Feedwater Piping
- 5.4.10 Pressurizer

5.4.12

5.4.10.1	Design Bases
5.4.10.2	Design Description
5.4.10.3	Design Evaluation
5.4.10.4	Tests and Inspections

5.4.11 Pressurizer Relief Discharge System

5.4.11.1	Design Bases
5.4.11.2	System Description
5.4.11.3	Safety Evaluation
5.4.11.4	Instrumentation Requirements
5.4.11.5	Inspection and Testing Requirements
Valves	

5.4.12.1	Design Bases
5.4.12.2	Design Description
5.4.12.3	Design Evaluation
5.4.12.4	Tests and Inspections

5.4.13 Safety and Relief Valves

5.4.13.1	Design Bases
5.4.13.2	Design Description
5.4.13.3	Design Evaluation
5.4.13.4	Tests and Inspection

5.4.14 Component Supports

5.4.14.1	Design Bases
5.4.14.2	Description
5.4.14.3	Evaluation
<b>ГААА</b>	Tasta and Incoment

- 5.4.14.4 Tests and Inspections
- 5.4.15 Reactor Vessel Head Vent System

5.4.15.1	Design Bases
5.4.15.2	System Description
5.4.15.3	Safety Evaluation
E A A E A	Inon-otion and Testing De

5.4.15.4 Inspection and Testing Requirements

### TABLE OF CONTENTS (CONTINUED)

### 5.4.15.5 Instrumentation Requirements

### 6.0 ENGINEERED SAFETY FEATURES

#### 6.1 Engineered Safety Features Materials

6.1.1 Metallic Materials

- 6.1.1.2 Composition, Compatibility, and Stability of Containment and Core Spray Coolants
- 6.1.2 Organic Materials

6.1.2.1	Protective Coatings
6.1.2.2	Other Organic Materials

6.1.3 Post-Accident Chemistry

#### 6.2 Containment Systems

6.2.1 Containment Functional Design

6.2.1.1	Containment Structure
6.2.1.2	Containment Subcompartments
6.2.1.3	Mass and Energy Release Analysis for Postulated Loss-of-
	Coolant Accidents
6.2.1.4	Mass and Energy Release Analysis for Postulated Secondary
	System Pipe Ruptures Inside Containment
6.2.1.5	Minimum Containment Pressure Analysis for Performance
	Capability Studies on Emergency Core Cooling System
6.2.1.6	Instrumentation Requirements

6.2.2 Containment Heat Removal Systems

6.2.2.1	Containment (	Cooling System
---------	---------------	----------------

- 6.2.2.2 Containment Spray System
- 6.2.3 Secondary Containment Functional Design
- 6.2.4 Containment Isolation System

6.2.4.1	Design Bases
---------	--------------

- 6.2.4.2 System Description
- 6.2.4.3 Design Evaluation
- 6.2.4.4 Tests and Inspections
- 6.2.4.5 Instrumentation Application

### TABLE OF CONTENTS (CONTINUED)

#### 6.2.5 Combustible Gas Control in Containment

- 6.2.5.1 Design Bases
- 6.2.5.2 System Design
- 6.2.5.3 Design Evaluation
- 6.2.5.4 Tests and Inspections
- 6.2.5.5 Instrumentation Requirements
- 6.2.5.6 Materials
- 6.2.6 Containment Leakage Testing
  - 6.2.6.1 Containment Integrated Leakage Rate Test (Type A Test)
  - 6.2.6.2 Containment Penetration Leakage Rate Tests (Type B Tests)
  - 6.2.6.3 Containment Isolation Valve Leakage Rate Tests
  - 6.2.6.4 Scheduling and Reporting of Periodic Tests
  - 6.2.6.5 Special Testing Requirements

#### 6.2.7 Fracture Prevention of Containment Pressure Boundary

- 6.2.7.1 Design Bases
- 6.2.7.2 Specifications for Ferritic Materials
- 6.2.7.3 Documentation
- 6.2.7.4 Evaluation

#### 6.3 Emergency Core Cooling System

- 6.3.1 Design Bases
- 6.3.2 System Design
  - 6.3.2.1 Schematic Piping and Instrumentation Diagrams
  - 6.3.2.2 Equipment and Component Descriptions
  - 6.3.2.3 Applicable Codes and Standards
  - 6.3.2.4 Material Specifications and Compatibility
  - 6.3.2.5 System Reliability
  - 6.3.2.6 Protection Provisions
  - 6.3.2.7 Provisions for Performance Testing
  - 6.3.2.8 Manual Actions
- 6.3.3 Performance Evaluation
  - 6.3.3.1 Increase in Heat Removal by the Secondary System
  - 6.3.3.2 Decrease in Heat Removal by the Secondary System
  - 6.3.3.3 Decrease in RCS Inventory
  - 6.3.3.4 Use of Dual Function Components
  - 6.3.3.5 Limits on System Parameters

- 6.3.4 Tests and Inspections
  - 6.3.4.1 Emergency Core Cooling System Performance Tests
  - 6.3.4.2 Reliability Tests and Inspections
- 6.3.5 Instrumentation Requirements

6.3.5.1	Deleted
6.3.5.2	Pressure Indication
6.3.5.3	Flow Indication
6.3.5.4	Level Indication
6.3.5.5	<b>Operating Status Indication</b>

- 6.4 Habitability Systems
  - 6.4.1 Design Bases
  - 6.4.2 System Design
    - 6.4.2.1 Definition of the Control Room Envelope
    - 6.4.2.2 Ventilation System Design
    - 6.4.2.3 Leaktightness
    - 6.4.2.4 Interaction with Other Zones and Pressurized Equipment
    - 6.4.2.5 Shielding Design
    - 6.4.2.6 Reference
  - 6.4.3 System Operational Procedures
    - 6.4.3.1 Normal Mode
    - 6.4.3.2 Emergency Mode
    - 6.4.3.3 Smoke Removal Mode
  - 6.4.4 Design Evaluations

6.4.4.1	Radiological Protection
~	

- 6.4.4.2 Toxic Gas Protection
- 6.4.4.3 Implementation of Design Bases
- 6.4.5 Testing and Inspection

6.4.5.1	Preoperational Testing
6.4.5.2	Inservice Testing

- 6.4.6 Instrumentation Requirement
- 6.5 Fission Product Removal and Control Systems

- 6.5.1 Engineered Safety Features Filter Systems
  - 6.5.1.1 Design Bases
  - 6.5.1.2 System Design
  - 6.5.1.3 Design Evaluation
  - 6.5.1.4 Tests and Inspections
  - 6.5.1.5 Instrumentation Requirements
  - 6.5.1.6 Materials
  - 6.5.1.7 Standard Review Plan Evaluation
- 6.5.2 Containment Spray Systems (Fission Product Removal)
  - 6.5.2.1 Design Bases (For Fission Product Removal)
  - 6.5.2.2 System Design (For Fission Product Removal)
  - 6.5.2.3 Design Evaluation
  - 6.5.2.4 Tests and Inspections
  - 6.5.2.5 Instrumentation Requirements
  - 6.5.2.6 Materials
  - 6.5.2.7 Standard Review Plan Evaluation
- 6.5.3 Fission Product Control Systems

6.5.3.1	Primary Containment
6.5.3.2	Secondary Containment

- 6.6 Inservice Inspection of Class 2 and 3 Components
  - 6.6.1 Components Subject to Examination
  - 6.6.2 Accessibility
  - 6.6.3 Examination Techniques and Procedures
  - 6.6.4 Inspection Intervals
  - 6.6.5 Examination Categories and Requirements
  - 6.6.6 Evaluation of Examination Results
  - 6.6.7 System Pressure Tests
  - 6.6.8 Augmented Inservice Inspection To Protect Against Postulated Piping Failures
- Appendix 6A Resolution of NRC Generic Letter 2004-02
- 6A.1 Introduction and Risk-Informed Approach Summary
- 6A.2 Debris Generation

- 6A.3 Debris Transport to the Sump Strainers
- 6A.4 Chemical Effects
- 6A.5 Sump Strainer Evaluations
- 6A.6 Downstream Effects Components and Systems
- 6A.7 Downstream Effects Fuel and Vessel
- 6A.8 Analyzed Debris Limits
- 6A.9 References
- 7.0 INSTRUMENTATION AND CONTROLS
- 7.1 Introduction
  - 7.1.1 Identification of Safety-Related Systems
    - 7.1.1.1 Reactor Protection System
    - 7.1.1.2 Other Instrumentation Systems Required for Safety
    - 7.1.1.3 Systems Required for Safe Shutdown
    - 7.1.1.4 Control Systems Not Required for Safety
    - 7.1.1.5 Comparison with Other Plants
  - 7.1.2 Identification of Safety Criteria
    - 7.1.2.1 Design Bases
    - 7.1.2.2 Independence of Redundant Safety-Related Systems
    - 7.1.2.3 Physical Identification of Safety-Related Equipment
    - 7.1.2.4 Conformance to Criteria
    - 7.1.2.5 Conformance to Regulatory Guide 1.22
    - 7.1.2.6 Conformance to Regulatory Guide 1.53 and IEEE Standard 379-1972
    - 7.1.2.7 Conformance to IEEE Standard 338-1975
- 7.2 Reactor Trip System
  - 7.2.1 Description

7.2.1.1	System Description
7.2.1.2	Design Bases Information

- 7.2.1.3 Final System Drawings
- 7.2.2 Analyses

### TABLE OF CONTENTS (CONTINUED)

7.2.2.1	Failure	Modes ar	nd Effects	Analyses
				,

- 7.2.2.2 **Evaluation of Design Limits**
- Specific Control and Protection Interactions 7.2.2.3
- 7.2.2.4 **Additional Postulated Accidents**
- 7.2.2.5 **Tests and Inspections**

#### 7.3 **Engineered Safety Features Systems**

7.3.1 Nuclear Steam Supply System ESFAS

7.3.1.1	Introduction
7.3.1.2	Analysis

7.3.2 Emergency Core Cooling System

7.3.2.1	Description
7.3.2.2	Analysis

7.3.3 Containment Combustible Gas Control System

7.3.3.1	Description
7.3.3.2	Analysis

7.3.4 Containment Purge Isolation System

7.3.4.1	Description
7.3.4.2	Analvsis

7.3.5 Fuel Handling Building Ventilation Isolation

7.3.5.1	Description
	<b>D</b> ' <b>D</b>

- 7.3.5.2 7 3 5 3 **Design Bases**
- Drawings
- 7.3.5.4 Analysis
- 7.3.6 Control Room Ventilation Isolation

7.3.6.1	Description
7.3.6.2	Analysis

7.3.7 Auxiliary Feedwater System

7.3.7.1	Description
7.3.7.2	Analysis

- 7.3.8 Main Steam and Feedwater Isolation
  - 7.3.8.1 Description

#### TABLE OF CONTENTS (CONTINUED)

- 7.3.8.2 Analysis
- 7.3.9 Nuclear Service Cooling Water

7.3.9.1	Description
7.3.9.2	Analysis

7.3.10 Component Cooling Water System

7.3.10.1	Description
7.3.10.2	Analysis

7.3.11 Containment Heat Removal System

7.3.11.1	Description
7.3.11.2	Analysis
7.3.11.3	Summary
7.3.11.4	Loss-of-Coolant Protection
7.3.11.5	Drawings

- 7.3.12 Control Building Engineered Safety Features Heating, Ventilation, and Air-Conditioning System
  - 7.3.12.1 Description
  - 7.3.12.2 Analysis
  - 7.3.12.3 Summary
- 7.3.13 Auxiliary Building Engineered Safety Features Heating, Ventilation, and Air-Conditioning System
  - 7.3.13.1 Description
  - 7.3.13.2 Analysis
  - 7.3.13.3 Summary
- 7.3.14 Auxiliary Feedwater Pumphouse Engineered Safety Features Heating, Ventilation, and Air-Conditioning System

7.3.14.1	Description
----------	-------------

- 7.3.14.2 Analysis
- 7.3.14.3 Summary
- 7.3.15 Diesel Generator Building Engineered Safety Features Heating, Ventilation, and Air-Conditioning System
  - 7.3.15.1 Description
  - 7.3.15.2 Analysis
  - 7.3.15.3 Summary

### TABLE OF CONTENTS (CONTINUED)

7.3.16 Electrical Tunnel Engineered Safety Features Heating, Ventilation, and Air-Conditioning System

7.3.16.1	Description
----------	-------------

- 7.3.16.2 Analysis
- 7.3.16.3 Summary
- 7.3.17 Diesel Generator Fuel Oil System

7.3.17.1	Decription
7.3.17.2	Analysis
7.3.17.3	Summary

- 7.4 Systems Required for Safe Shutdown
  - 7.4.1 Hot Standby

7.4.1.1	Auxiliary Feedwater Control
7.4.1.2	Power-Operated Atmospheric Steam Relief Valves
7.4.1.3	Centrifugal Charging System Controls
7.4.1.4	Coolant Circulation
7.4.1.5	Other Systems Required for Hot Standby

- 7.4.2 Cold Shutdown
- 7.4.3 Safe Shutdown from Outside the Control Room

7.4.3.1	Description
7.4.3.2	Analysis
7.4.3.3	Alternate Shutdown Indication System

- 7.5 Information Systems Important to Safety
  - 7.5.1 Safety-Related Display Instrumentation Introduction
  - 7.5.2 Description of Information Systems

- 7.5.2.2 Variable Types
- 7.5.2.3 Variable Categories
- 7.5.2.4 Post Accident Monitoring Instrumentation Program
- 7.5.3 Description of Variables

7.5.3.1	Type A Variables
7.5.3.2	Type B Variables
7.5.3.3	Type C Variables

#### TABLE OF CONTENTS (CONTINUED)

7.5.3.4	Type D Variables
---------	------------------

- 7.5.3.5 Type E Variables
- 7.5.3.6 Plant Safety Monitoring System
- 7.5.4 Additional Information
- 7.5.5 Bypassed and Inoperable Status Indication for Engineered Safety Features Systems

7.5.5.1	Description
7.5.5.2	Conformance to Regulatory Guide 1.47
7.5.5.3	Conformance to Branch Technical Position ICSB-21
7.5.5.4	System Drawings

- 7.6 Interlock Systems Important to Safety
  - 7.6.1 Instrumentation and Control Power Supply System

7.6.1.1	Description
7.6.1.2	Analysis

7.6.2 Residual Heat Removal Isolation Valves

7.6.2.1	Description
7.6.2.2	Analysis

- 7.6.3 Refueling Interlocks
- 7.6.4 Accumulator Motor-Operated Valves
- 7.6.5 Switchover from Injection to Recirculation
  - 7.6.5.1 Description of Instrumentation Used for Switchover
  - 7.6.5.2 Initiation Circuit
  - 7.6.5.3 Logic
  - 7.6.5.4 Bypass
  - 7.6.5.5 Interlocks
  - 7.6.5.6 Sequence
  - 7.6.5.7 Redundancy
  - 7.6.5.8 Diversity
  - 7.6.5.9 Actuated Devices
- 7.6.6 Interlocks Isolating Safety Systems from Nonsafety Systems
  - 7.6.6.1 Deleted
  - 7.6.6.2 Refueling Water Storage Tank Isolation
  - 7.6.6.3 Deleted
  - 7.6.6.4 Isolation of Reactor Coolant Pump Thermal Barrier Cooling Water

- 7.6.6.5 Electric Steam Boiler Isolation
- 7.6.6.6 Steam Generator Blowdown Isolation
- 7.6.6.7 CVCS Letdown Line Isolation
- 7.6.6.8 Analysis
- 7.6.7 Interlocks for Reactor Coolant System Pressure Control During Low-Temperature Operations
  - 7.6.7.1 Analysis of Interlock
  - 7.6.7.2 Pressurizer Pressure Relief System
- 7.7. Control Systems Not Required for Safety
  - 7.7.1 Description of Control Systems Not Required for Safety
    - 7.7.1.1 Reactor Control System
    - 7.7.1.2 Rod Control System
    - 7.7.1.3 Plant Control Signals for Monitoring and Indicating
    - 7.7.1.4 Plant Control System Interlocks
    - 7.7.1.5 Pressurizer Pressure Control
    - 7.7.1.6 Pressurizer Water Level Control
    - 7.7.1.7 Steam Generator Water Level Control
    - 7.7.1.8 Steam Dump Control
    - 7.7.1.9 Incore Instrumentation
    - 7.7.1.10 Boron Concentration Measurement System
    - 7.7.1.11 ATWS Mitigation System Actuation Circuitry Description
  - 7.7.2 Analysis of Control Systems Not Required for Safety
    - 7.7.2.1 Separation of Protection and Control System
    - 7.7.2.2 Response Considerations of Reactivity
    - 7.7.2.3 Step-Load Changes Without Steam Dump
    - 7.7.2.4 Loading and Unloading
    - 7.7.2.5 Load Rejection Furnished by Steam Dump System
    - 7.7.2.6 Turbine-Generator Trip With Reactor Trip
    - 7.7.2.7 Core Cooling Monitor
    - 7.7.2.8 Reactor Vessel Level Instrumentation System
    - 7.7.2.9 Control Systems Failure Analysis
    - 7.7.2.10 ATWS Mitigation System Actuation Circuitry Analysis
- 8.0 ELECTRIC POWER
- 8.1 Introduction

- 8.1.1 Utility Grid Description
- 8.1.2 Onsite Power System Description
- 8.1.3 Safety-Related Loads
- 8.1.4 Design Bases
  - 8.1.4.1 Offsite Power System
  - 8.1.4.2 Onsite Power System
  - 8.1.4.3 Design Criteria, Regulatory Guides, and IEEE Standards
- 8.2 Offsite Power System
  - 8.2.1 System Description

8.2.1.1	Offsite Sources
0 2 1 2	Switchvord

- 8.2.1.2 Switchyard
- 8.2.2 Analysis
  - 8.2.2.1 Loss of VEGP Unit 1 or 2 or the Largest Unit
  - 8.2.2.2 VEGP Voltage Operating Range
  - 8.2.2.3 VEGP Transient Stability
  - 8.2.2.4 Conformance to Criteria
  - 8.2.2.5 Standards and Guides
- 8.3 Onsite Power Systems
  - 8.3.1 AC Power Systems
    - 8.3.1.1 Description
    - 8.3.1.2 Analysis
    - 8.3.1.3 Physical Identification of Safety-Related Equipment
    - 8.3.1.4 Independence of Redundant Systems
    - 8.3.1.5 Standard Review Plan Evaluation
  - 8.3.2 DC Power Systems
    - 8.3.2.1 Description
    - 8.3.2.2 Analysis
  - 8.3.3 Fire Protection for Cable Systems
- 8.4 Station Blackout (SBO)
  - 8.4.1 Introduction

### TABLE OF CONTENTS (CONTINUED)

8.4.1.1 SBO Coping Evaluation

### 9.0 AUXILIARY SYSTEMS

- 9.1 Fuel Storage and Handling
  - 9.1.1 New Fuel Storage

9.1.1.1	Design Bases
9.1.1.2	Facilities Description
9.1.1.3	Safety Evaluation
9.1.1.4	Test and Inspections
9.1.1.5	Instrumentation

9.1.2 Wet Spent Fuel Storage

9.1.2.1	Design Bases

- 9.1.2.2 Facilities Description
- 9.1.2.3 Safety Evaluation
- 9.1.3 Spent Fuel Pool Cooling and Purification System

9.1.3.1	Design Bases
---------	--------------

- 9.1.3.2 System Description
- 9.1.3.3 Component Description
- 9.1.3.4 System Operation
- 9.1.3.5 Safety Evaluation
- 9.1.3.6 Tests and Inspections
- 9.1.3.7 Standard Review Plan Evaluation
- 9.1.4 Light Load Handling System (Related to Refueling)
  - 9.1.4.1 Design Bases
  - 9.1.4.2 System Description
  - 9.1.4.3 Safety Evaluation
  - 9.1.4.4 Inspection and Testing Requirements
  - 9.1.4.5 Instrumentation Requirements
  - 9.1.4.6 Standard Review Plan Evaluation
- 9.1.5 Overhead Heavy Load Handling Systems

- 9.1.5.2 System Description
- 9.1.5.3 Safety Evaluation
- 9.1.5.4 Tests and Inspections
- 9.1.5.5 Instrumentation Applications
- 9.1.5.6 Load Handling Procedures

#### TABLE OF CONTENTS (CONTINUED)

#### 9.1.6 Dry Spent Fuel Storage

9.1.6.1 Facility Description

#### 9.2 Water Systems

9.2.1 Nuclear Service Cooling Water System

9.2.1.1	Design Bases
9.2.1.2	System Description

- 9.2.1.3 Safety Evaluation
- 9.2.1.4 Tests and Inspections
- 9.2.1.5 Instrumentation Applications
- 9.2.2 Component Cooling Water System

- 9.2.2.2 System Description
- 9.2.2.3 Safety Evaluation
- 9.2.2.4 Tests and Inspections
- 9.2.2.5 Instrumentation Applications
- 9.2.3 Demineralized Water Makeup System

9.2.3.1 Design Bas	es
--------------------	----

- 9.2.3.2 System Description
- 9.2.3.3 Safety Evaluation
- 9.2.3.4 Tests and Inspections
- 9.2.3.5 Instrumentation Applications
- 9.2.4 Potable and Sanitary Water System

- 9.2.4.2 System Description
- 9.2.4.3 Safety Evaluation
- 9.2.4.4 Tests and Inspections
- 9.2.4.5 Instrumentation Applications
- 9.2.5 Ultimate Heat Sink

- 9.2.5.2 System Description
- 9.2.5.3 Safety Evaluation
- 9.2.5.4 Tests and Inspections
- 9.2.5.5 Instrumentation Applications
- 9.2.5.6 Standard Review Plan Evaluation

#### TABLE OF CONTENTS (CONTINUED)

9.2.6 Condensate Storage Facility

- 9.2.6.2 System Description
- System Operation 9.2.6.3
- 9.2.6.4 Safety Evaluation
- **Tests and Inspections** 9.2.6.5
- Instrumentation Applications 9.2.6.6
- 9.2.7 Reactor Makeup Water Facility

9.2.7.1	Design Bases
9.2.7.2	System Description
9.2.7.3	Safety Evaluation

Tests and Inspections 9.2.7.4

- Instrumentation Applications 9.2.7.5
- 9.2.8 Auxiliary Component Cooling Water
  - 9.2.8.1 **Design Bases**
  - System Description 9.2.8.2
  - Safety Evaluation 9.2.8.3
  - **Tests and Inspections** 9.2.8.4
  - Instrumentation Applications 9.2.8.5
  - Standard Review Plan Evaluation 9.2.8.6
- 9.2.9 Chilled Water Systems

9.2.9.1	Essential Chilled Water System	l
---------	--------------------------------	---

- 9.2.9.2 Normal Chilled Water System
- 9.2.10 Turbine Plant Closed Cooling Water System

9.2.10.1	Design Bases
9.2.10.2	System Description
9.2.10.3	Safety Evaluation
9.2.10.4	Tests and Inspections
9.2.10.5	Instrument Applications

- 9.2.11 Turbine Plant Cooling Water System
  - 9.2.11.1 **Design Bases**
  - 9.2.11.2 System Description
  - 9.2.11.3 Safety Evaluation
  - 9.2.11.4 Tests and Inspections
  - 9.2.11.5 **Instrument Applications**
- 9.3 **Process Auxiliaries**

- 9.3.1 Compressed Air System
  - 9.3.1.1 Design Bases
  - 9.3.1.2 System Description
  - 9.3.1.3 Safety Evaluation
  - 9.3.1.4 Tests and Inspections
  - 9.3.1.5 Instrumentation Applications
- 9.3.2 Process Sampling Systems

9.3.2.1	Design Bases
9.3.2.2	System Description
9.3.2.3	Safety Evaluation

- 9.3.2.4 Tests and Inspections
- 9.3.2.5 Instrumentation Applications
- 9.3.3 Equipment and Floor Drainage Systems
  - 9.3.3.1 Design Bases
  - 9.3.3.2 System Description
  - 9.3.3.3 Safety Evaluation
  - 9.3.3.4 Tests and Inspections
  - 9.3.3.5 Instrumentation Applications
- 9.3.4 Chemical and Volume Control System (Including Boron Recycle System)
  - 9.3.4.1 Chemical and Volume Control System
  - 9.3.4.2 Boron Recycle System
- 9.3.5 Auxiliary Gas Systems

- 9.3.5.2 System Description
- 9.3.5.3 Safety Evaluation
- 9.3.5.4 Tests and Inspections
- 9.3.5.5 Instrumentation Requirements
- 9.3.6 Standby Liquid Control System (Boiling Water Reactor)
- 9.3.7 Chlorine Supply System

- 9.3.7.2 System Description
- 9.3.7.3 Safety Evaluation
- 9.3.7.4 Inspection and Testing Requirements
- 9.3.7.5 Instrumentation Requirements
- 9.4 Air-Conditioning, Heating, Cooling, and Ventilation Systems
  - 9.4.1 Control Building Ventilation Systems
    - 9.4.1.1 Design Bases
    - 9.4.1.2 System Description
    - 9.4.1.3 System Operation
    - 9.4.1.4 System Components
    - 9.4.1.5 Safety Evaluation
    - 9.4.1.6 Tests and Inspections
    - 9.4.1.7 Instrumentation Application
    - 9.4.1.8 Onsite Technical Support Center HVAC
  - 9.4.2 Fuel Handling Building Ventilation System
    - 9.4.2.1 Normal Operation of the Fuel Handling Building Ventilation System
    - 9.4.2.2 Post-Accident or Emergency Operation of the FHB Ventilation System
  - 9.4.3 Auxiliary and Radwaste Building Ventilation System
    - 9.4.3.1 Auxiliary Building Normal Ventilation System
    - 9.4.3.2 Auxiliary Building Emergency Ventilation System Radwaste Building Ventilation Systems Radwaste Processing Facility
  - 9.4.4 Turbine Building Ventilation System
    - 9.4.4.1 Design Bases
    - 9.4.4.2 System Description
    - 9.4.4.3 Safety Evaluation
    - 9.4.4.4 Tests and Inspections
    - 9.4.4.5 Instrumentation Applications
  - 9.4.5 Control Building Engineered Safety Features Ventilation Systems
    - 9.4.5.1 Design Bases
    - 9.4.5.2 System Description
    - 9.4.5.3 System Operation
    - 9.4.5.4 Safety Evaluations
    - 9.4.5.5 System Components
    - 9.4.5.6 Tests and Inspections
    - 9.4.5.7 Instrumentation Applications
  - 9.4.6 Containment Building Ventilation System

9.4.6.1	Design Base	s
---------	-------------	---

- 9.4.6.2 System Description
- 9.4.6.3 Safety Evaluation

- 9.4.6.4 Testing and Inspection
- 9.4.6.5 Instrumentation Applications
- 9.4.7 Diesel Generator Building Ventilation System
  - 9.4.7.1 Design Bases
  - 9.4.7.2 System Description
  - 9.4.7.3 Safety Evaluation
  - 9.4.7.4 Tests and Inspections
  - 9.4.7.5 Instrumentation Applications
  - 9.4.7.6 Standard Review Plan Evaluation
- 9.4.8 Auxiliary Feedwater Pumphouse Heating, Ventilation, and Air-Conditioning System
  - 9.4.8.1 Design Bases
  - 9.4.8.2 System Description
  - 9.4.8.3 Safety Evaluation
  - 9.4.8.4 Tests and Inspections
  - 9.4.8.5 Instrumentation Applications
- 9.4.9 Miscellaneous Heating, Ventilation, and Air-Conditioning Systems
  - 9.4.9.1 Equipment Building Ventilation System
  - 9.4.9.2 Electrical Tunnel Ventilation System
  - 9.4.9.3 Piping Penetration and MSIV Area Ventilation Systems
- 9.5 Other Auxiliary Systems
  - 9.5.1 Fire Protection Program
    - 9.5.1.1 Design Basis
    - 9.5.1.2 Systems Description
    - 9.5.1.3 Safety Evaluation
    - 9.5.1.4 Inspection and Testing Requirements
    - 9.5.1.5 Personnel Qualification and Training
    - 9.5.1.6 Standard Review Plan Evaluation
  - 9.5.2 Communication Systems
    - 9.5.2.1 Design Bases
    - 9.5.2.2 System Description
    - 9.5.2.3 Inspection and Testing Requirements
  - 9.5.3 Lighting Systems
    - 9.5.3.1 Design Bases
    - 9.5.3.2 System Description

- 9.5.3.3 Failure Analysis
- 9.5.3.4 Test and Inspections
- 9.5.4 Emergency Diesel Generator Fuel Oil Storage and Transfer System
  - 9.5.4.1 Design Bases
  - 9.5.4.2 System Description
  - 9.5.4.3 Safety Evaluation
  - 9.5.4.4 Tests and Inspections
  - 9.5.4.5 Instrumentation Applications
- 9.5.5 Diesel Generator Cooling Water System
  - 9.5.5.1 Design Bases
  - 9.5.5.2 System Description
  - 9.5.5.3 Safety Evaluation
  - 9.5.5.4 Tests and Inspections
  - 9.5.5.5 Instrumentation Applications
- 9.5.6 Diesel Generator Starting System
  - 9.5.6.1 Design Bases
  - 9.5.6.2 System Description
  - 9.5.6.3 Safety Evaluation
  - 9.5.6.4 Tests and Inspections
  - 9.5.6.5 Instrumentation Applications
- 9.5.7 Diesel Generator Lubrication System
  - 9.5.7.1 Design Bases
  - 9.5.7.2 System Description
  - 9.5.7.3 Safety Evaluation
  - 9.5.7.4 Tests and Inspections
  - 9.5.7.5 Instrumentation Applications
- 9.5.8 Diesel Generator Combustion Air Intake and Exhaust System
  - 9.5.8.1 Design Bases
  - 9.5.8.2 System Description
  - 9.5.8.3 Safety Evaluation
  - 9.5.8.4 Tests and Inspections
  - 9.5.8.5 Instrumentation Applications
- 9.5.9 Auxiliary Steam System
  - 9.5.9.1 Design Bases
  - 9.5.9.2 System Description
  - 9.5.9.3 Safety Evaluation
  - 9.5.9.4 Tests and Inspections

# TABLE OF CONTENTS (CONTINUED)

# 9.5.9.5 Instrumentation Applications

9.5.10 Deleted

# Appendix 9A Fire Hazards Analysis

### 9A.1 Introduction

9A.1.1	Fire Area 1-AB-LD-A
9A.1.2	Fire Area 1-AB-LD-B
9A.1.3	Fire Area 1-AB-LD-C
9A.1.4	Fire Area 1-AB-LD-D
9A.1.5	Fire Area 1-AB-LD-E
9A.1.6	Fire Area 1-AB-LD-F
9A.1.7	Fire Area 1-AB-LD-G
9A.1.8	Fire Area 1-AB-LD-H
9A.1.9	Fire Area 1-AB-LD-I
9A.1.10	Fire Area 1-AB-LD-J
9A.1.11	Deleted
9A.1.12	Fire Area 1-AB-LC-A
9A.1.13	Fire Area 1-AB-LC-B
9A.1.14	Fire Area 1-AB-LC-C
9A.1.15	Fire Area 1-AB-LC-D
9A.1.16	Fire Area 1-AB-LC-E
9A.1.17	Deleted
9A.1.18	Deleted
9A.1.19	Fire Area 1-AB-LB-A
9A.1.20	Fire Area 1-AB-LB-B
9A.1.21	Deleted
9A.1.22	Fire Area 1-AB-LA-A
9A.1.23	Fire Area 1-AB-LA-B
9A.1.24	Fire Area 1-AB-LA-C
9A.1.25	Fire Area 1-AB-LA-D
9A.1.26	Fire Area 1-AB-LA-E
9A.1.27	Fire Area 1-AB-L1-B
9A.1.28	Fire Area 1-AB-L1-C
9A.1.29	Deleted
9A.1.30	Deleted
9A.1.31	Fire Area 1-AB-L1-G
9A.1.32	Fire Area 1-AB-L1-H
9A.1.33	Fire Area 1-AB-L2-A
9A.1.34	Deleted
9A.1.35	Fire Area 1-AB-L2-C
9A.1.36	Deleted
9A.1.37	Fire Area 1-AB-L2-E
9A.1.38	Deleted
9A.1.39	Fire Area 1-CB-LC-A
9A.1.40	Fire Area 1-CB-LC-B

9A.1.41	Deleted
9A.1.42	Fire Area 1-CB-LB-A
9A.1.43	Fire Area 1-CB-LB-B
9A.1.44	Fire Area 1-CB-LB-C
9A.1.45	Fire Area 1-CB-LB-D
9A.1.46	Fire Area 1-CB-LB-E
9A.1.47	Fire Area 1-CB-LB-F
9A.1.48	Fire Area 1-CB-LB-G
9A.1.49	Fire Area 1-CB-LB-H
9A.1.50	Fire Area 1-CB-LB-I
9A.1.51	Fire Area 1-CB-LB-J
9A.1.52	Fire Area 1-CB-LB-K
9A.1.53	Fire Area 1-CB-LB-L
9A.1.54	Fire Area 1-CB-LB-M
9A.1.55	Fire Area 1-CB-LB-N
9A.1.56	Fire Area 1-CB-LB-O
9A.1.57	Fire Area 1-CB-LB-P
9A.1.58	Fire Area 1-CB-LB-Q
9A 1 59	Deleted
9A 1 60	Fire Area 1-CB-I B-S
9A 1 60A	Fire Area 1-CB-I B-T
9A 1 61	Fire Area 1-CB-I A-A
9A 1 62	Fire Area 1-CB-I A-B
9A 1 63	Fire Area 1-CB-I A-C
9A 1 64	Fire Area 1-CB-I A-D
9A 1 65	Fire Area 1-CB-I A-F
9A.1.66	Fire Area 1-CB-LA-F
9A.1.67	Fire Area 1-CB-LA-G
9A.1.68	Fire Area 1-CB-LA-H
9A.1.69	Fire Area 1-CB-LA-I
9A 1 70	Fire Area 1-CB-I A-J
9A.1.71	Fire Area 1-CB-LA-K
9A.1.72	Fire Area 1-CB-LA-L
9A.1.73	Fire Area 1-CB-LA-M
9A.1.74	Fire Area 1-CB-LA-N
9A.1.75	Fire Area 1-CB-LA-O
9A.1.76	Fire Area 1-CB-LA-P
9A.1.77	Fire Area 1-CB-LA-Q
9A.1.78	Fire Area 1-CB-LA-R
9A.1.79	Fire Area 1-CB-LA-S
9A 1 80	Fire Area 1-CB-I A-T
9A 1 80A	Fire Area 1-CB-LA-U
9A 1 81	Fire Area 1-CB-I 1-A
9A 1 82	Fire Area 1-CB-L1-R
9A 1 83	Fire Area 1-CB-L1-C
9A 1 84	Fire Area 1-CB-I 1-D
9A.1.85	Fire Area 1-CB-I 1-F
9A.1.86	Fire Area 1-CB-I 1-F
9A.1.87	Fire Area 1-CB-L1-G

9A.1.88	Fire Area 1-CB-L1-TSC
9A.1.89	Fire Area 1-CB-L2-A
9A.1.90	Fire Area 1-CB-L2-B
9A.1.91	Deleted
9A.1.92	Deleted
9A.1.93	Fire Area 1-CB-L2-E
9A.1.94	Fire Area 1-CB-L3-A
9A.1.95	Fire Area 1-CB-L3-B
9A.1.96	Fire Area 1-CB-L3-C
9A.1.97	Deleted
9A.1.98	Deleted
9A.1.99	Deleted
9A.1.100	Deleted
9A.1.101	Fire Area 1-CB-L3-H
9A.1.102	Deleted
9A.1.103	Fire Area 1-CB-L3-J
9A.1.104	Fire Area 1-CB-L3-K
9A.1.105	Fire Area 1-CB-L3-L
9A.1.105A	Fire Area 1-CB-L3-M
9A.1.106	Fire Area 1-CB-L4-A
9A.1.107	Fire Area 1-FB-LC-A
9A.1.108	Deleted
9A.1.109	Fire Area 1-FB-L3-A
9A.1.110	Fire Area 1-FB-L3-B
9A.1.111	Fire Area 1-CTB
9A.1.112	Deleted
9A.1.113	Fire Area 1-EB-B
9A.1.114	Fire Area 1-DB-L1-A
9A.1.115	Fire Area 1-DB-L1-B
9A.1.116	Fire Area 1-DB-L1-C
9A.1.117	Fire Area 1-DB-L1-D
9A.1.118	Fire Area 1-DPB-A
9A.1.119	FIFE Area 1-DPB-B
9A.1.120	Deleted
9A.1.121	FILE ALEA T-AFD-A
9A.1.122	File Alea I-AFD-D
9A.1.123	File Alea 1-AFB-C
9A.1.124 0A 1 124A	Plated
9A.1.124A	Deleted
9A.1.124D	Eiro Aroa 1 NSD I A A
9A.1.125 0A 1 126	Fire Area 1 NSP   A B
9Α.1.120 9Δ 1 126Δ	Fire Area 1-NSP-LA-D
QΔ 1 120A	
9A 1 128	Fire Area 1-HPR-I 1-A
9A 1 129	Fire Area 1-RTR-I $1_{-}\Delta$
9A 1 130	Fire Area 1-RPF-I 1-A
9A 1 131	Deleted
9A 1 132	Fire Area 1-0.SR
<u> </u>	

9A.2.1	Fire Area 2-AB-LD-A
9A.2.2	Fire Area 2-AB-LD-B
9A.2.3	Fire Area 2-AB-LD-C
9A.2.4	Fire Area 2-AB-LD-D
9A.2.5	Fire Area 2-AB-LD-E
9A.2.6	Fire Area 2-AB-LD-F
9A.2.7	Fire Area 2-AB-LD-G
9A.2.8	Fire Area 2-AB-LD-H
9A.2.9	Fire Area 2-AB-LD-I
9A.2.10	Fire Area 2-AB-LD-J
9A.2.11	Fire Area 2-AB-LC-A
9A.2.12	Fire Area 2-AB-LC-B
9A.2.13	Fire Area 2-AB-LC-C
9A.2.14	Fire Area 2-AB-LC-D
9A.2.15	Fire Area 2-AB-LC-E
9A.2.16	Fire Area 2-AB-LB-A
9A.2.17	Fire Area 2 AB LA A
9A.2.18	Fire Area 2 AB LA B
9A.Z.19	File Alea 2-AD-LA-D
9A.2.20	File Alea 2-AD-LA-C
QΔ 2 22	Fire $\Delta rea 2-\Delta B_1 \Delta_F$
9A 2 23	Fire Area 2-AB-LI-R
9A 2 24	Fire Area 2-AB-I 1-C
9A.2.25	Fire Area 2-AB-L2-A
9A.2.26	Fire Area 2-AB-L2-C
9A.2.27	Fire Area 2-AB-L2-E
9A.2.28	Fire Area 2-CB-LC-A
9A.2.29	Fire Area 2-CB-LC-B
9A.2.30	Fire Area 2-CB-LB-A
9A.2.31	Fire Area 2-CB-LB-B
9A.2.32	Fire Area 2-CB-LB-C
9A.2.33	Fire Area 2-CB-LB-D
9A.2.34	Fire Area 2-CB-LB-E
9A.2.35	Fire Area 2-CB-LB-F
9A.2.36	Fire Area 2-CB-LB-G
9A.2.37	Fire Area 2-CB-LB-H
9A.2.38	Fire Area 2-CB-LB-I
9A.2.39	Fire Area 2-CB-LB-J
9A.2.40	File Area 2-CB-LB-K
9A.2.41	File Alea 2-CD-LD-L Fire Area 2 CB LB M
9A.2.42 QΔ 2 /3	Fire Area 2-CB-LD-W
9Δ 2 <i>Δ</i> Δ	Fire Area 2-CB-LD-N
9A 2 45	Fire Area 2-CB-LB-O
9A.2.46	Fire Area 2-CB-LB-Q
9A.2.47	Fire Area 2-CB-LB-T
9A.2.48	Fire Area 2-CB-LB-X

# TABLE OF CONTENTS (CONTINUED)

9A.2.49	Fire Area 2-CB-LA-A
9A.2.50	Fire Area 2-CB-LA-B
9A.2.51	Fire Area 2-CB-LA-C
9A.2.52	Fire Area 2-CB-LA-D
9A.2.53	Fire Area 2-CB-LA-F
9A.2.54	Fire Area 2-CB-LA-G
9A.2.55	Fire Area 2-CB-LA-H
9A.2.56	Fire Area 2-CB-LA-I
9A.2.57	Fire Area 2-CB-LA-J
9A.2.58	Fire Area 2-CB-LA-K
9A.2.59	Fire Area 2-CB-LA-L
9A.2.60	Fire Area 2-CB-LA-M
9A.2.61	Fire Area 2-CB-LA-N
9A.2.62	Fire Area 2-CB-LA-O
9A.2.63	Fire Area 2-CB-LA-P
9A.2.64	Fire Area 2-CB-LA-Q
9A.2.65	Fire Area 2-CB-LA-R
9A.2.66	Fire Area 2-CB-LA-S
9A.2.67	Fire Area 2-CB-LA-T
9A.2.68	Fire Area 2-CB-LA-X
9A.2.69	Fire Area 2-CB-L1-E
9A.2.70	Fire Area 2-CB-LI-F
9A.2.71	Fire Area 2-CB-L2-A
9A.2.72	Fire Area 2-CB-L2-B
9A.2.73	Fire Area 2-CB-L3-B
9A.2.74	Fire Area 2-CB-L3-C
9A.2.75	Fire Area 2-FB-LC-A
9A.2.76	Fire Area 2-CTB
9A.2.77	Fire Area 2-EB-B
9A.2.78	Fire Area 2-DB-L1-A
9A.2.79	Fire Area 2-DB-L1-B
9A.2.80	Fire Area 2-DB-L1-C
9A.2.81	Fire Area 2-DB-L1-D
9A.2.82	Fire Area 2-DPB-A
9A.2.83	Fire Area 2-DPB-B
9A.2.84	Fire Area 2-AFB-A
9A.2.85	Fire Area 2-AFB-B
9A.2.86	Fire Area 2-AFB-C
9A.2.87	Fire Area 2-AFB-D
9A.2.88	Fire Area 2-NSP-LA-A
9A.2.89	Fire Area 2-NSP-LA-B

Appendix 9B Comparison of VEGP Units 1 and 2 with Requirements of the Branch Technical Position CMEB 9.5-1

10.0 STEAM AND POWER CONVERSION SYSTEM

# TABLE OF CONTENTS (CONTINUED)

- 10.1 Summary Description
  - 10.1.1 General Description
  - 10.1.2 Protective and Safety-Related Features
- 10.2 Turbine-Generator

10.2.1	Design Bases	
	10.2.1.1	Safety Design Bases
	10.2.1.2	Power Generation Design Bases

10.2.2 Description

10.2.2.1	Turbine-Generator Description
10.2.2.2	Turbine-Generator Cycle Description
10.2.2.3	Automatic Controls
10.2.2.4	Turbine Protective Trips
10.2.2.5	Protective System
10.2.2.6	Other Protective Systems
10.2.2.7	Plant Loading and Load Following
10.2.2.8	Inspection and Testing Requirements

10.2.3 Turbine Integrity (Unit 1)

10.2.3.1	Materials Selection
10.2.3.2	High-Temperature Properties
10.2.3.3	Turbine Disk Design
10.2.3.4	Preservice Tests and Inspections
10.2.3.5	Inservice Inspection

10.2.4 Turbine Integrity (Unit 2)

10.2.4.1.1	Materials Selection
10.2.4.1.2	Fracture Toughness
10.2.4.1.3	High-Temperature Properties
10.2.4.1.4	Turbine Disk Design
10.2.4.1.5	Preservice Tests and Inspections
10.2.4.1.6	Inservice Inspection

- 10.2.5 Evaluation
- 10.2.6 Instrumentation Applications
- 10.3 Main Steam Supply System
  - 10.3.1 Design Bases

# TABLE OF CONTENTS (CONTINUED)

10.3.1.1 Safety Design Bases	
------------------------------	--

10.3.1.2 Power Generation Design Basis

### 10.3.2 Description

10.3.2.1	General Description
10.3.2.2	Component Description
10.3.2.3	System Operation

10.3.3 Evaluation

10.3.4 Inspection and Testing Requirements

10.3.4.1	Preservice Valve Testing
10.3.4.2	Preservice System Testing
10.3.4.3	Preservice Pipe Testing
10.3.4.4	Inservice Testing

10.3.5 Water Chemistry

10.3.5.1	Chemistry Control Basis
10.3.5.2	Contaminant Ingress
10.3.5.3	Condensate Polishing
10.3.5.4	Chemical Addition
10.3.5.5	Action Levels for Abnormal Conditions
10.3.5.6	Layup and Heatup
10.3.5.7	Chemical Analysis Basis
10.3.5.8	Sampling
10.3.5.9	Condenser Inspection
10.3.5.10	Data Management and Corrective Actions for Out-of-Specification
	Condition
10.3.5.11	Conformance to Branch Technical Position MTEB 5-3

### 10.3.6 Steam and Feedwater System Materials

10.3.6.1	Fracture Toughness
10.3.6.2	Material Selection and Fabrication

### 10.4 Other Features of Steam and Power Conversion System

10.4.1 Main Condensers

10.4.1.1	Design Bases
10.4.1.2	System Description
10.4.1.3	Safety Evaluation
10.4.1.4	Tests and Inspections
10.4.1.5	Instrumentation Applications

- 10.4.2 Main Condenser Evacuation System
  - 10.4.2.1 Design Bases
  - 10.4.2.2 System Description
  - 10.4.2.3 Safety Evaluation
  - 10.4.2.4 Tests and Inspections
  - 10.4.2.5 Instrumentation Applications
- 10.4.3 Turbine Steam Sealing System

10.4.3.1	Design	Bases
----------	--------	-------

- 10.4.3.2 System Description
- 10.4.3.3 Safety Evaluation
- 10.4.3.4 Tests and Inspections
- 10.4.3.5 Instrumentation Applications
- 10.4.4 Turbine Bypass System

- 10.4.4.2 System Description
- 10.4.4.3 Safety Evaluation
- 10.4.4.4 Inspection and Testing Requirements
- 10.4.4.5 Instrumentation Applications
- 10.4.5 Circulating Water System

- 10.4.5.2 System Description
- 10.4.5.3 Safety Evaluation
- 10.4.5.4 Tests and Inspections
- 10.4.5.5 Instrumentation Applications
- 10.4.6 Condensate Cleanup System

10.4.6.1	Design Bases
----------	--------------

- 10.4.6.2 System Description
- 10.4.6.3 Safety Evaluation
- 10.4.6.4 Tests and Inspections
- 10.4.6.5 Instrumentation Applications
- 10.4.7 Condensate and Feedwater System

- 10.4.7.2 System Description
- 10.4.7.3 Safety Evaluation
- 10.4.7.4 Tests and Inspections
- 10.4.7.5 Instrumentation Applications
- 10.4.8 Steam Generator Blowdown Processing System

### TABLE OF CONTENTS (CONTINUED)

- 10.4.8.1 Design Bases
- 10.4.8.2 System Description
- 10.4.8.3 Design Evaluation
- 10.4.8.4 Tests and Inspections
- 10.4.9 Auxiliary Feedwater System

10 4 9 1	Design Bases
10.4.0.1	
10.4.9.2	System Description
10.4.9.3	Safety Evaluation
10.4.9.4	Inspection and Testing Requirements
10.4.9.5	Instrumentation Requirements

10.4.10 Condensate and Feedwater Chemical Injection System

10.4.10.1	Design Bases
10.4.10.2	System Description
10.4.10.3	Safety Evaluation
10.4.10.4	Tests and Inspections
10.4.10.5	Instrumentation Applications

Appendix 10A VEGP Auxiliary Feedwater System Reliability Analysis

#### 10A.1 Introduction

- 10A.1.1 Statement of Purpose
- 10A.1.2 Background and Objectives
- 10A.2 AFWS Description

10A.2.1 Safety-Related Function	on
---------------------------------	----

- 10A.2.2 System Configuration
- 10A.2.3 Instrumentation and Controls
- 10A.2.4 Support Systems
- 10A.2.5 Testing, Maintenance, and Plant Procedures
- 10A.2.6 Physical Separation

#### 10A.3 Reliability Evaluation

- 10A.3.1 Methodology
- 10A.3.2 System Success Criteria
- 10A.3.3 Component Failure Data
- 10A.3.4 Analysis Boundary Conditions and Assumptions
- 10A.4 Quantitative Findings
  - 10A.4.1 Conditional Unavailabilities

### TABLE OF CONTENTS (CONTINUED)

- 10A.4.2 Dominant Contributors
- 10A.4.3 Comparative Studies Results

### 11.0 RADIOACTIVE WASTE MANAGEMENT

#### 11.1 Source Terms

11.1.1 Reactor Coolant and Secondary Side Activity

11.1.1.1	Design Basis Model
11.1.1.2	Realistic Model

- 11.1.2 Tritium Production and Release to the Reactor Coolant
- 11.1.3 Spent Resin Volumes
- 11.2 Liquid Waste Management Systems
  - 11.2.1 Design Bases

- 11.2.1.2 Controlled Release of Radioactivity
- 11.2.1.3 Equipment Design

#### 11.2.2 System Descriptions

- 11.2.2.1 Reactor Coolant Drain Tank Subsystem
- 11.2.2.2 Drain Channel A Subsystem
- 11.2.2.3 Drain Channel B Subsystem
- 11.2.2.4 Spent Resin Handling Subsystem
- 11.2.2.5 Liquids from Sources Other Than the Liquid Waste Processing System
- 11.2.2.6 Equipment Description
- 11.2.2.7 Instrumentation Design
- 11.2.2.8 Operating Procedures
- 11.2.3 Radioactive Releases
  - 11.2.3.1 Criteria for Discharge, Recycle, or Further Treatment of Liquid Waste
  - 11.2.3.2 Estimated Releases
  - 11.2.3.3 Release Points
  - 11.2.3.4 Dilution Factors
  - 11.2.3.5 Estimated Doses
- 11.3 Gaseous Waste Management Systems

### TABLE OF CONTENTS (CONTINUED)

- 11.3.1 Design Bases
- 11.3.2 System Description

11.3.2.1	Component Design
----------	------------------

- 11.3.2.2 Instrumentation and Control Design
- 11.3.2.3 System Operation
- 11.3.3 Radioactive Releases

11.3.3.1	Discharge Requirements
11.3.3.2	Westinghouse Pressurized Water Reactor Experience Releases
11.3.3.3	Estimated Releases
11.3.3.4	Release Points for Determination of Dilution Factors
11.3.3.5	Effluent Concentration and Dilution Factors
11.3.3.6	Estimated Doses from Atmospheric Releases

### 11.4 Solid Waste Management System

11.4.1 Design Bases

- 11.4.1.2 **Power Generation Design Bases**
- 11.4.2 System Description

11.4.2.1	General Description
11.4.2.2	Component Description
11.4.2.3	System Operation
11.4.2.4	Portable Radwaste System
11.4.2.5	Dry Active Waste

- 11.4.3 Safety Evaluation
- 11.5 Process and Effluent Radiological Monitoring and Sampling Systems

1.5	.1	Design	Bases
-----	----	--------	-------

11.5.2 System Description

11.5.2.1	Digital Radiation	Monitoring System
	5	5,

- 11.5.2.2 11.5.2.3 Monitor Description
- Monitor Functions
- 11.5.2.4 Alarm Setpoints
- 11.5.2.5 Inservice Inspection, Calibration, and Maintenance
- 11.5.3 Effluent Monitoring and Sampling

### TABLE OF CONTENTS (CONTINUED)

11531	Plant Sampling	System
11.5.5.1	Flant Sampling	System

- 11.5.3.2 Sampling Requirements for Process and Effluent Radiation Monitoring System
- 11.5.4 Process Monitoring and Sampling
- 11.5.5 Post-Accident Radiation Monitoring
  - 11.5.5.1 Potential Gaseous Accident Release Pathways 11.5.5.2 Standard Review Plan Evaluation
- 12.0 RADIATION PROTECTION
- 12.1 Ensuring That Occupational Radiation Exposures Are as Low as Reasonably Achievable
  - 12.1.1 Policy Considerations

12.1.1.1	Design and Construction Policies
----------	----------------------------------

- 12.1.1.2 Operation Policies
- 12.1.1.3 Conformance with Regulatory Guide 8.8
- 12.1.1.4 Conformance with Regulatory Guide 8.10
- 12.1.2 Design Considerations
  - 12.1.2.1 General Design Considerations for ALARA Exposures
  - 12.1.2.2 Equipment General Design Considerations for ALARA
  - 12.1.2.3 Facility Layout General Design Considerations for ALARA
- 12.1.3 Operational Considerations
  - 12.1.3.1 Procedures
  - 12.1.3.2 General ALARA Techniques

### 12.2 Radiation Sources

12.2.1 Contained Sources

12.2.1.1	Sources for Full-Power Operation
12.2.1.2	Sources for Shutdown
12.2.1.3	Sources for Design Basis Accident
12.2.1.4	Stored Radioactivity

- 12.2.1.5 Field Run Pipe Routing
- 12.2.2 Airborne Radioactive Material Sources
  - 12.2.2.1 Model for Calculating Airborne Concentrations
  - 12.2.2.2 Sources Resulting from Design Basis Accidents

## TABLE OF CONTENTS (CONTINUED)

### 12.3 Radiation Protection Design Features

#### 12.3.1 Facility Design Features

12.3.1.1	Plant Design Description for ALARA
----------	------------------------------------

- 12.3.1.2 Radiation Zoning and Access Control
- 12.3.2 Shielding

12.3.2.1	Design Objectives
12.3.2.2	General Shielding Design
12.3.2.3	Shielding Calculational Methods

12.3.3 Ventilation

12.3.3.1	Design Objectives
12.3.3.2	Design Criteria
12.3.3.3	Design Guidelines
12.3.3.4	Design Description
12.3.3.5	Air Cleaning Units

12.3.4 Area Radiation and Airborne Radioactivity Monitoring Instrumentation

12.3.4.1	Area Radiation Monitoring
12.3.4.2	Standard Review Plan Evaluation

### 12.4 Dose Assessment

12.4.1 Exposures Within the Plant Structures

- 12.4.1.2 Airborne Radioactivity Dose Estimates
- 12.4.1.3 Standard Review Plan Evaluation

### 12.4.2 Radiation Exposure Outside the Plant Structures

- 12.4.2.1 Direct Radiation Dose Estimates
- 12.4.2.2 Exposures Due to Airborne Radioactivity
- 12.4.3 Exposures to Construction Workers

12.4.3.1 Direct Radia	ation Dose Estimates
-----------------------	----------------------

- 12.4.3.2 Exposures Due to Airborne Radioactivity
- 12.4.4 Decommissioning Dose Estimates

### TABLE OF CONTENTS (CONTINUED)

#### 12.5 Health Physics Program

12.5.1	Organization
--------	--------------

12.5.1.1	Program Objectives
12.5.1.2	Staff Organization
12.5.1.3	Program Organization

12.5.2 Equipment, Instrumentation, and Facilities

12.5.2.1	Facilities
12.5.2.2	Health Physics Instrumentation

#### 12.5.3 Procedures

12.5.3.1	Radiation and Contamination Surveys
12.5.3.2	Methods To Maintain Exposures ALARA
12.5.3.3	Controlling Access and Stay Time
12.5.3.4	Contamination Control
12.5.3.5	Respiratory Protection
12.5.3.6	Personnel Dosimetry
12.5.3.7	Radiation Work Permits
12.5.3.8	Radiation Protection Training
12.5.3.9	Radioactive Materials Safety Program

### 13.0 CONDUCT OF OPERATIONS

- 13.1 Organizational Structure of Applicant
  - 13.1.1 Management and Technical Support Organization
    - 13.1.1.1 Design and Operating Responsibilities
    - 13.1.1.2 Organizational Arrangement
    - 13.1.1.3 Qualifications
  - 13.1.2 Operating Organization

13.1.2.1	Plant Organization
----------	--------------------

- 13.1.2.2 Plant Personnel Responsibilities and Authorities
- 13.1.2.3 Operating Shift Crews
- 13.1.3 Qualifications of Personnel

13.1.3.1	Qualification Requirements
----------	----------------------------

- 13.1.3.2 Definitions
- 13.1.3.3 Qualification of Plant Personnel

### TABLE OF CONTENTS (CONTINUED)

#### 13.2 Training Programs

13.2.1 Accredited Training Programs

- 13.2.1.1 Training Program Records
- 13.2.1.2 Simulator
- 13.2.1.3 Instructor Qualification
- 13.2.2 Other Training Programs

13.2.2.1	General Employee Training Program
13.2.2.2	Fire Brigade Training Program
13.2.2.3	Quality Control Training Program

13.2.2.4 Mitigating Core Damage Training Program

#### 13.3 Emergency Planning

- 13.4 Operational Review
  - 13.4.1 Onsite Review
  - 13.4.2 Independent Review
- 13.5 Plant Procedures
  - 13.5.1 Administrative Procedures
    - 13.5.1.1 Administrative Procedure Descriptions
  - 13.5.2 Operating and Maintenance Procedures

13.5.2.1	Operating Procedures
13.5.2.2	Maintenance Procedures
13.5.2.3	Other Procedures and Program

- 13.6 Industrial Security
- 13.7 Plant Records
  - 13.7.1 Records Retention
- 13.8 Risk Informed Categorization and Treatment
  - 13.8.1 Introduction

- 13.8.2 SSC Categorization
- 13.8.3 SSC Treatment
- 13.8.3.1 Treatment of Component Categories
- 13.8.3.2 Enhanced Treatment of RISC-2 SSCs
- 13.8.4 References
- 14.0 INITIAL TEST PROGRAM
- 14.1 Specific Information To Be Included in Preliminary Safety Analysis Reports
- 14.2 Initial Test Program
  - 14.2.1 Summary of Test Program and Objectives

14.2.1.1	Preoperational	Test Program

- 14.2.1.2Startup Test Program
- 14.2.2 Test Procedures

14.2.2.1	Procedure Preparation
14.2.2.2	Procedure Review and Approval
14.2.2.3	Conduct of the Test Program
14.2.2.4	Review, Evaluation, and Approval of Test Results

- 14.2.3 Conformance of Test Program with Regulatory Guides
- 14.2.4 Utilization of Reactor Operating and Testing Experience in the Development of the Test Program
- 14.2.5 Trial Use of Plant Operating and Emergency Procedures
- 14.2.6 Initial Fuel Loading and Initial Criticality

14.2.6.1	Initial Fuel Loading
14.2.6.2	Postloading Tests
14.2.6.3	Low-Power Testing
14.2.6.4	Power Level Ascension

- 14.2.7 Test Program Schedule
- 14.2.8 Individual Test Descriptions
  - 14.2.8.1 Preoperational Test Procedures

### TABLE OF CONTENTS (CONTINUED)

### 14.2.8.2 Startup Test Procedures

#### 15.0 ACCIDENT ANALYSES

#### 15.0 Accident Analyses

15.0.1 Classification of Plant Conditions

15.0.1.1	Condition I: Normal Operation and Operational Transients
15.0.1.2	Condition II: Faults of Moderate Frequency
15.0.1.3	Condition III: Infrequent Faults
15.0.1.4	Condition IV: Limiting Faults
15.0.1.5	TMI Action Plan (NUREG-0737) Implementation

- 15.0.2 Optimization of Control Systems
- 15.0.3 Plant Characteristics and Initial Conditions Assumed in the Accident Analyses

15.0.3.1	Design Plant Conditions

- 15.0.3.2 Initial Conditions
- 15.0.3.3 Power Distribution

15.0.4 Reactivity Coefficients Assumed in the Accident Analyses

- 15.0.5 Rod Cluster Control Assembly Insertion Characteristics
- 15.0.6 Trip Points and Time Delays to Trip Assumed in Accident Analyses
- 15.0.7 Instrumentation Drift and Calorimetric Errors, Power Range Neutron Flux
- 15.0.8 Plant Systems and Components Available for Mitigation of Accident Effects
- 15.0.9 Fission Product Inventories

15.0.9.1	Inventory in the Core
15.0.9.2	Inventory in the Fuel Pellet Clad Gap

15.0.10 Residual Decay Heat

15.0.10.1	Total Residual Heat
15.0.10.2	Distribution of Decay Heat Following Loss-of-Coolant Accident

15.0.11 Computer Codes Utilized

15.0.11.1	FACTRAN Computer Code
15.0.11.2	LOFTRAN Computer Code
15.0.11.3	Cross-Section Generation Computer Codes
15.0.11.4	Spatial Two-Group Diffusion Calculation Codes

15.0.11.5	TWINKLE Computer Code
15.0.11.6	THINC Computer Code

- 15.0.12 Limiting Single Failures
- 15.0.13 Operator Actions
- 15.1 Increase in Heat Removal by the Secondary System
  - 15.1.1 Feedwater System Malfunctions That Result in a Decrease in Feedwater Temperature
    - 15.1.1.1 Identification of Causes and Accident Description
    - 15.1.1.2 Analysis of Effects and Consequences
    - 15.1.1.3 Conclusions
  - 15.1.2 Feedwater System Malfunctions That Result in an Increase in Feedwater Flow
    - 15.1.2.1 Identification of Causes and Accident Description
    - 15.1.2.2 Analysis of Effects and Consequences
    - 15.1.2.3 Conclusions
  - 15.1.3 Excessive Increase in Secondary Steam Flow
    - 15.1.3.1 Identification of Causes and Accident Description
    - 15.1.3.2 Analysis of Effects and Consequences
    - 15.1.3.3 Conclusions
  - 15.1.4 Inadvertent Opening of a Steam Generator Relief or Safety Valve
    - 15.1.4.1 Identification of Causes and Accident Description
    - 15.1.4.2 Analysis of Effects and Consequences
    - 15.1.4.3 Conclusions
  - 15.1.5 Steam System Piping Failure
    - 15.1.5.1 Identification of Causes and Accident Description
    - 15.1.5.2 Analysis of Effects and Consequences
    - 15.1.5.3 Radiological Consequences
- 15.2 Decrease in Heat Removal by the Secondary System
  - 15.2.1 Steam Pressure Regulator Malfunction or Failure That Results in Decreasing Steam Flow
  - 15.2.2 Loss of External Electrical Load

### TABLE OF CONTENTS (CONTINUED)

- 15.2.2.1 Identification of Causes and Accident Description
- 15.2.2.2 Analysis of Effects and Consequences
- 15.2.2.3 Conclusions

#### 15.2.3 Turbine Trip

- 15.2.3.1 Identification of Causes and Accident Description
- 15.2.3.2 Analysis of Effects and Consequences
- 15.2.3.3 Conclusions
- 15.2.4 Inadvertent Closure of Main Steam Isolation Valves
- 15.2.5 Loss of Condenser Vacuum and Other Events Resulting in Turbine Trip
- 15.2.6 Loss of Non-emergency ac Power to the Plant Auxiliaries
  - 15.2.6.1 Identification of Causes and Accident Description
  - 15.2.6.2 Analysis of Effects and Consequences
  - 15.2.6.3 Radiological Consequences
  - 15.2.6.4 Conclusions
- 15.2.7 Loss of Normal Feedwater Flow
  - 15.2.7.1 Identification of Causes and Accident Description
  - 15.2.7.2 Analysis of Effects and Consequences
  - 15.2.7.3 Conclusions
- 15.2.8 Feedwater System Pipe Break
  - 15.2.8.1 Identification of Causes and Accident Description
  - 15.2.8.2 Analysis of Effects and Consequences
  - 15.2.8.3 Conclusions
- 15.3 Decrease in Reactor Coolant System Flowrate
  - 15.3.1 Partial Loss of Forced Reactor Coolant Flow
    - 15.3.1.1 Identification of Causes and Accident Description
    - 15.3.1.2 Analysis of Effects and Consequences
    - 15.3.1.3 Conclusions
  - 15.3.2 Complete Loss of Forced Reactor Coolant Flow
    - 15.3.2.1 Identification of Causes and Accident Description
    - 15.3.2.2 Analysis of Effects and Consequences
    - 15.3.2.3 Conclusions
  - 15.3.3 Reactor Coolant Pump Shaft Seizure (Locked Rotor)

### TABLE OF CONTENTS (CONTINUED)

- 15.3.3.1 Identification of Causes and Accident Description
- 15.3.3.2 Analysis of Effects and Consequences
- 15.3.3.3 Radiological Consequences

#### 15.3.4 Reactor Coolant Pump Shaft Break

- 15.3.4.1 Identification of Causes and Accident Description
- 15.3.4.2 Conclusion
- 15.4 Reactivity and Power Distribution Anomalies
  - 15.4.1 Uncontrolled Rod Cluster Control Assembly Bank Withdrawal from a Subcritical or Low-Power Startup Condition
    - 15.4.1.1 Identification of Causes and Accident Description
    - 15.4.1.2 Analysis of Effects and Consequences
    - 15.4.1.3 Conclusions
  - 15.4.2 Uncontrolled Rod Cluster Control Assembly Bank Withdrawal at Power
    - 15.4.2.1 Identification of Causes and Accident Description
    - 15.4.2.2 Analysis of Effects and Consequences
    - 15.4.2.3 Conclusions
  - 15.4.3 Rod Cluster Control Assembly Misalignment (System Malfunction or Operator Error)
    - 15.4.3.1 Identification of Causes and Accident Description
    - 15.4.3.2 Analysis of Effects and Consequences
    - 15.4.3.3 Conclusions
  - 15.4.4 Startup of an Inactive Reactor Coolant Pump at an Incorrect Temperature
    - 15.4.4.1 Identification of Causes and Accident Description
    - 15.4.4.2 Analysis of Effects and Consequences
    - 15.4.4.3 Conclusions
  - 15.4.5 A Malfunction or Failure of the Flow Controller in a Boiling Water Reactor Loop That Results in an Increased Reactor Coolant Flowrate
  - 15.4.6 Chemical and Volume Control System Malfunction That Results in a Decrease in the Boron Concentration in the Reactor Coolant
    - 15.4.6.1 Identification of Causes and Accident Description
    - 15.4.6.2 Analysis of Effects and Consequences
    - 15.4.6.3 Conclusions

- 15.4.7 Inadvertent Loading and Operation of a Fuel Assembly in an Improper Position
  - 15.4.7.1 Identification of Causes and Accident Description
  - 15.4.7.2 Analysis of Effects and Consequences
  - 15.4.7.3 Conclusions
- 15.4.8 Spectrum of Rod Cluster Control Assembly Ejection Accidents
  - 15.4.8.1 Identification of Causes and Accident Description
  - 15.4.8.2 Analysis of Effects and Consequences
  - 15.4.8.3 Radiological Consequences
- 15.4.9 Steamline Break with Coincidental Rod Cluster Control Assembly Withdrawal at Power
  - 15.4.9.1 Identification of Causes and Accident Description
  - 15.4.9.2 Analysis of Effects and Consequences
  - 15.4.9.3 Conclusions
- 15.5 Increase in Reactor Coolant Inventory
  - 15.5.1 Inadvertent Operation of the Emergency Core Cooling System During Power Operation
    - 15.5.1.1 Identification of Causes and Accident Description
    - 15.5.1.2 Analysis of Effects and Consequences
    - 15.5.1.3 Conclusions
  - 15.5.2 Chemical and Volume Control System Malfunction That Increases Reactor Coolant Inventory
  - 15.5.3 A Number of Boiling Water Reactor Transients
- 15.6 Decrease in Reactor Coolant Inventory
  - 15.6.1 Inadvertent Opening of a Pressurizer Safety or Relief Valve
    - 15.6.1.1 Identification of Causes and Accident Description
    - 15.6.1.2 Analysis of Effects and Consequences
    - 15.6.1.3 Conclusion
  - 15.6.2 Break in Instrument Line or Other Lines from Reactor Coolant Pressure Boundary That Penetrate Containment
    - 15.6.2.1 Assumptions Instrument Line Break
    - 15.6.2.2 Assumptions Letdown Line Break
    - 15.6.2.3 Mathematical Models Used in the Analysis
    - 15.6.2.4 Identification of Leakage Pathways

- 15.6.2.5 Identification of Uncertainties and Conservatisms in the Analysis
- 15.6.2.6 Conclusions
- 15.6.3 Steam Generator Tube Failure
  - 15.6.3.1 Identification of Causes and Accident Description
  - 15.6.3.2 Analysis of Effects and Consequences
  - 15.6.3.3 Thermal and Hydraulic Analysis
  - 15.6.3.4 Offsite Radiation Dose Analysis
- 15.6.4 Spectrum of Boiling Water Reactor Steam System Piping Failures Outside of Containment
- 15.6.5 Loss-of-Coolant Accidents
  - 15.6.5.1 Identification of Causes and Frequency Classification
  - 15.6.5.2 Sequence of Events and Systems Operations
  - 15.6.5.3 Core and System Performance
  - 15.6.5.4 Radiological Consequences
- 15.7 Radioactive Release from a Subsystem or Component
  - 15.7.1 Radioactive Waste Gas Decay Tank Failure
    - 15.7.1.1 Identification of Causes
    - 15.7.1.2 Sequence of Events and System Operations
    - 15.7.1.3 Core and System Performance
    - 15.7.1.4 Barrier Performance
    - 15.7.1.5 Radiological Consequences
  - 15.7.2 Radioactive Liquid Waste System Leak or Failure
    - 15.7.2.1 Identification of Causes
    - 15.7.2.2 Sequence of Events and System Operation
    - 15.7.2.3 Core and System Performance
    - 15.7.2.4 Barrier Performance
    - 15.7.2.5 Radiological Consequences
  - 15.7.3 Postulated Radioactive Release Due to Liquid Tank Failure (Ground Release)
    - 15.7.3.1 Identification of Causes and Frequency Classification
    - 15.7.3.2 Sequence of Events and Systems Operation
    - 15.7.3.3 Modeling of Accident Sequence
    - 15.7.3.4 Radiological Consequences
  - 15.7.4 Fuel Handling Accidents
    - 15.7.4.1 Identification of Causes and Accident Description

### TABLE OF CONTENTS (CONTINUED)

- 15.7.4.2 Sequence of Events and Systems Operations
- 15.7.4.3 Core and System Performance
- 15.7.4.4 Barrier Performance
- 15.7.4.5 Radiological Consequences

15.7.5 Spent Fuel Cask Drop Accident

- 15.8 Anticipated Transients Without Trip
- Appendix 15A Accident Analysis Radiological Consequences Evaluation Models and Parameters
- 15A.1 General Accident Parameters
- 15A.2 Offsite Radiological Consequences Calculational Models

Pathways

- 15A.2.2 Single-Region Release Model
- 15A.2.3 Two-Region Spray Model in Containment LOCA
- 15A.2.4 Offsite Thyroid Dose Calculation Model
- 15A.2.5 Offsite Beta Skin Dose Calculational Model
- 15A.2.6 Offsite Gamma Body Dose Calculational Model
- 15A.3 Control Room Radiological Consequences Calculational Models
  - 15A.3.1 Integrated Activity in Control Room
  - 15A.3.2 Integrated Activity Concentration in Control Room from Single-Region System
  - 15A.3.3 Control Room Thyroid Dose Calculational Model
  - 15A.3.4 Control Room Beta Skin Dose Calculational Model
  - 15A.3.5 Control Room Gamma Body Dose Calculation
    - 15A.3.5.1 Model for Radiological Consequences Due to Radioactive Cloud External to the Control Room
- 15A-4 Effect of Extended Fuel Burnups on Accident Doses

Appendix 15B Deleted

- 16.0 TECHNICAL SPECIFICATIONS
- 16.1 Preliminary Technical Specifications
- 16.2 Proposed Final Technical Specifications

- 16.2.1 Foreword
- 16.2.2 Applicability
- 16.2.3 Format
- 16.2.4 Schedule
- 16.3 Technical Specification Improvement Program
  - 16.3.1 Requirement 1 Reactor Trip System Response Times
  - 16.3.2 Requirement 2 Engineered Safety Feature Actuation System Response Times
  - 16.3.3 Requirement 3 Loose Part Detection System
  - 16.3.4 Reactor Vessel Material Irradiation Specimens
  - 16.3.5 Requirement 5 Containment Isolation Valves
  - 16.3.6 Requirement 6 Containment Penetration Conductor Overcurrent Protection
  - 16.3.7 Requirement 7 Area Temperature Monitoring
  - 16.3.8 Requirement 8 Turbine Overspeed Protection
  - 16.3.9 Requirement 9 Technical Requirements Manual
  - 16.3.10 Requirement 10 Reactor Coolant System Pressure Isolation Valves and Leakage Limits
- 16.4 Risk Managed Technical Specifications
  - 16.4.1 PRA Functionality
  - 16.4.2 PRA Technical Adequacy
- 17.0 QUALITY ASSURANCE
- 17.1 Deleted
- 17.2 Operations Quality Assurance Program
- 18.0 HUMAN FACTORS ENGINEERING
- 18.1 Control Room Design
  - 18.1.1 Control Room Description
    - 18.1.1.1 Introduction
    - 18.1.1.2 Description
  - 18.1.2 Design Bases
    - 18.1.2.1 Anthropometrics18.1.2.2 Auditory Communications18.1.2.3 Plant Communications

- 18.1.2.4 Controls
- 18.1.2.5 Control and Display Integration
- 18.1.2.6 Design for Maintainability
- 18.1.2.7 Design for Personnel Requirements
- 18.1.2.8 Environment
- 18.1.2.9 Hazards/Safety
- 18.1.2.10 Information Utilization
- 18.1.2.11 Labeling (Grouping, Marking)
- 18.1.2.12 Visual Display (Meters, Recorders, Lights, Monitors, and Annunciators)
- 18.1.2.13 Workspace
- 18.2 Preliminary Review of the Control Room and Control Board Design
  - 18.2.1 Review Procedures

18211	Introduction
10.2.1.1	muouuum

- 18.2.1.2 Anthropometrics
- 18.2.1.3 System Evaluation
- 18.2.1.4 Component Evaluation
- 18.2.1.5 Scenario Evaluation
- 18.2.1.6 Reviewer Experience
- 18.2.2 Control Room and Control Board Evaluation Results
  - 18.2.2.1 Human Factors Deficiencies Categories
  - 18.2.2.2 Findings and the Resolutions Thereof
- 18.3 Detailed Control Room Design Review
  - 18.3.1 Background
  - 18.3.2 Objectives of the CRDR
  - 18.3.3 Control Room Design Review Process
    - 18.3.3.1 Preliminary CRDR Status Evaluation
    - 18.3.3.2 Operating Experience Review
    - 18.3.3.3 Problem Reports
    - 18.3.3.4 Control Room Survey
    - 18.3.3.5 Task Analysis
    - 18.3.3.6 Instrumentation and Control Characteristics Review
  - 18.3.4 Evaluation of Human Engineering Discrepancies
    - 18.3.4.1 Objectives of Evaluation Process
    - 18.3.4.2 Evaluation Criteria
    - 18.3.4.3 Resolution of Human Engineering Discrepancies

### TABLE OF CONTENTS (CONTINUED)

### 18.3.5 Coordination with Other Activities

- 18.3.5.1 Emergency Operating Procedures
- 18.3.5.2 Safety Parameter Display System (SPDS)
- 18.3.5.3 Regulatory Guide 1.97

# 19.0 LICENSE RENEWAL – AGING MANAGEMENT PROGRAMS AND ACTIVITIES

- 19.1 Introduction
  - 19.1.1 Background
    - 19.1.1.1 License Renewal Rule and Process
    - 19.1.1.2 Aging Management Programs
    - 19.1.1.3 Time-Limited Aging Analyses
  - 19.1.2 Aging Management Programs
  - 19.1.3 Aging Management Programs Time-Limited Aging Analyses
  - 19.1.4 TLAA Evaluations
- 19.2 Aging Management Program Descriptions
  - 19.2.1 ACCW System Carbon Steel Components Program
  - 19.2.2 Bolting Integrity Program
  - 19.2.3 Boric Acid Corrosion Control Program
  - 19.2.4 Buried Piping and Tanks Inspection Program
  - 19.2.5 CASS RCS Fitting Evaluation Program
  - 19.2.6 Closed Cooling Water Program
  - 19.2.7 Diesel Fuel Oil Program
  - 19.2.8 External Surfaces Monitoring Program
  - 19.2.9 Fire Protection Program
  - 19.2.10 Flow-Accelerated Corrosion Program
  - 19.2.11 Flux Thimble Tube Inspection Program
  - 19.2.12 Generic Letter 89-13 Program
  - 19.2.13 Inservice Inspection Program
  - 19.2.14 Nickel Alloy Management Program for Nonreactor Vessel Closure Head Penetration Locations
  - 19.2.15 Nickel Alloy Management Program for Reactor Vessel Closure Head Penetrations
  - 19.2.16 Oil Analysis Program
  - 19.2.17 One-Time Inspection Program
  - 19.2.18 One-Time Inspection Program for ASME Class 1 Small Bore Piping
  - 19.2.19 One-Time Inspection Program for Selective Leaching
  - 19.2.20 Overhead and Refueling Crane Inspection Program
  - 19.2.21 Periodic Surveillance and Preventive Maintenance Activities
  - 19.2.22 Piping and Duct Internal Inspection Program
  - 19.2.23 Reactor Vessel Closure Head Stud Program

- 19.2.24 Reactor Vessel Internals Program
- 19.2.25 Reactor Vessel Surveillance Program
- 19.2.26 Steam Generator Tubing Integrity Program
- 19.2.27 Steam Generator Program for Upper Internals
- 19.2.28 Water Chemistry Control Program
- 19.2.29 10 CFR 50 Appendix J Program
- 19.2.30 Inservice Inspection Program IWE
- 19.2.31 Inservice Inspection Program IWL
- 19.2.32 Structural Monitoring Program
- 19.2.33 Structural Monitoring Program Masonry Walls
- 19.2.34 Non-EQ Cables and Connections Program
- 19.2.35 Non-EQ Inaccessible Medium-Voltage Cables Program
- 19.2.36 Non-EQ Electrical Cable Connections One-Time Inspection Program
- 19.3 Aging Management Programs Time-Limited Aging Analyses (TLAA)
  - 19.3.1 Environmental Qualification Program
  - 19.3.2 Fatigue Monitoring Program
- 19.4 Evaluation of Time-Limited Aging Analyses (TLAA)
  - 19.4.1 Reactor Vessel Neutron Embrittlement Analyses
    - 19.4.1.1 Neutron Fluence Calculation
    - 19.4.1.2 Upper-Shelf Energy (USE) Calculation
    - 19.4.1.3 Pressurized Thermal Shock (PTS) Calculation
    - 19.4.1.4 Adjusted Reference Temperature (ART) Calculation
    - 19.4.1.5 Pressure-Temperature (P-T) Limits Calculation
  - 19.4.2 Metal Fatigue Analysis
    - 19.4.2.1 ASME Section III, Class 1 Component Fatigue Analylsis
    - 19.4.2.2 ASME Section III, Non-Class 1 Component Fatigue Analylsis
    - 19.4.2.3 Reactor Coolant Pump Flywheel Fatigue
    - 19.4.2.4 Fatigue of Reactor Vessel Supports
    - 19.4.2.5 Fatigue of Steam Generator Secondary Manyway and Handhole Bolts
    - 19.4.2.6 Fatigue of Reactor Vessel Internals
  - 19.4.3 Environmental Qualification Calculations
  - 19.4.4 Containment Tendon Prestress Analysis
  - 19.4.5 Penetration Load Cycles
  - 19.4.6 Other Plant Specific Analysis
    - 19.4.6.1 Leak-Before-Break Analysis
    - 19.4.6.2 Fuel Oil Storage Tank Corrosion Allowance
    - 19.4.6.3 Steam Generator Tube Loss of Material
    - 19.4.6.4 Cold Overpressure Protection System
    - 19.4.6.5 Underclad cracking of the Reactor Pressure Vessel

# TABLE OF CONTENTS (CONTINUED)

19.4.7 References

# LIST OF TABLES

1.3.1-1	Design Comparison
1.3.1-2	Design Comparison of Major Balance of Plant Design Features
1.3.2-1	Significant Design Changes from the PSAR
1.3.2-2	TMI Action Plan Tasks and Location of FSAR Description
1.6-1	Bechtel Topical Reports Incorporated by Reference
1.6-2	Westinghouse Topical Reports Incorporated by Reference
1.6-3	Other Topical Reports Incorporated by Reference
1.7.1-1	Cross Reference of Engineering Drawings to UFSAR Discussion Section
1.8-1	List of Differences with the Standard Review Plan
1.9-1	Design and Fabrication Code Cases
1.9-2	Materials Code Cases
1.9-3	NSSS ASME Code Cases
2.1.3-1	Population by Segment for North Sector (0 to 50 Miles)
2.1.3-2	Population by Segment for North-Northeast Sector (0 to 50 Miles)
2.1.3-3	Population by Segment for Northeast Sector (0 to 50 Miles)
2.1.3-4	Population by Segment for East-Northeast Sector (0 to 50 Miles)
2.1.3-5	Population by Segment for East Sector (0 to 50 Miles)
2.1.3-6	Population by Segment for East-Southeast Sector (0 to 50 Miles)
2.1.3-7	Population by Segment for Southeast Sector (0 to 50 Miles)
2.1.3-8	Population by Segment for South-Southeast Sector (0 to 50 Miles)
2.1.3-9	Population by Segment for South Sector (0 to 50 Miles)
2.1.3-10	Population by Segment for South-Southwest Sector (0 to 50 Miles)
2.1.3-11	Population by Segment for Southwest Sector (0 to 50 Miles)
2.1.3-12	Population by Segment for West-Southwest Sector (0 to 50 Miles)
2.1.3-13	Population by Segment for West Sector (0 to 50 Miles)

REV 24 10/22

- 2.1.3-14 Population by Segment for West-Northwest Sector (0 to 50 Miles)
- 2.1.3-15 Population by Segment for Northwest Sector (0 to 50 Miles)
- 2.1.3-16 Population by Segment for North-Northwest Sector (0 to 50 Miles)
- 2.1.3-17 Population by Sectors, 0- to 10-Mile Radius Total
- 2.1.3-18 Population by Annular Rings, 0- to 10-Mile Radius Total
- 2.1.3-19 Population by Sectors, 20- to 50-Mile Radius Total
- 2.1.3-20 Population by Annular Rings, 20- to 50-Mile Radius Totals
- 2.1.3-21 Communities of Greater Than 1000 Persons Within 30 Miles of VEGP
- 2.1.3-22 Transient Population (North Sector 0 to 10 Miles)
- 2.1.3-23 Transient Population (North-Northeast Sector 0 to 10 Miles)
- 2.1.3-24 Transient Population (Northeast Sector 0 to 10 Miles)
- 2.1.3-25 Transient Population (East-Northeast Sector 0 to 10 Miles)
- 2.1.3-26 Transient Population (East Sector 0 to 10 Miles)
- 2.1.3-27 Transient Population (East-Southeast Sector 0 to 10 Miles)
- 2.1.3-28 Transient Population (Southeast Sector 0 to 10 Miles)
- 2.1.3-29 Transient Population (South-Southeast Sector 0 to 10 Miles)
- 2.1.3-30 Transient Population (South Sector 0 to 10 Miles)
- 2.1.3-31 Transient Population (South-Southwest Sector 0 to 10 Miles)
- 2.1.3-32 Transient Population (Southwest Sector 0 to 10 Miles
- 2.1.3-33 Transient Population (West-Southwest Sector 0 to 10 Miles)
- 2.1.3-34 Transient Population (West Sector 0 to 10 Miles)
- 2.1.3-35 Transient Population (West-Northwest Sector 0 to 10 Miles)
- 2.1.3-36 Transient Population (Northwest Sector 0 to 10 Miles)
- 2.1.3-37 Transient Population (North-Northwest Sector 0 to 10 Miles)
- 2.1.3-38 Transient Population Annular Rings and Enclosed Population (0 to 10 Miles)

2.2.2-1	Description of Savannah River Plant Facilities
2.2.2-2	Description of Barnwell Nuclear Fuel Plant Facility
2.2.2-3	Description of Chem-Nuclear Systems, Inc., Facility
2.2.2-4	Major Industries (25 or More Employees) Within 25-Mile Radius of VEGP
2.2.2-5	Description of Products and Materials: Savannah River Plant (Facilities Within 5 Miles of VEGP)
2.2.2-6	Applicable Toxicity Limits
2.2.2-7	Burke County, Georgia, Transportation Accident Data, Within 5 Miles of the Site
2.2.2-8	Hazardous Rail Cargo Traffic Estimates
2.2.2-9	Hazardous Cargo Rail Accident Data
2.2.2-10	Description of Products and Materials: Barnwell Nuclear Fuel Plant Allied-General Nuclear Services
2.2.2-11	Description of Products and Materials: Chem-Nuclear Systems, Inc.
2.2.2-12	Terminal Area Forecast, Fiscal Years 1980-1990 Total Flights
2.2.3-1	Deleted
2.2.3-2	Deleted
2.2.3-3	Deleted
2.2.3-4	Deleted
2.2.3-5	Deleted
2.2.3-6	Chemicals Transported Along Major Transportation Routes Within 5 Miles of VEGP - Quantities and Distances
2.2.3-7	Deleted
2.2.3-8	Deleted
2.2.3-9	Deleted
2.2.3-10	Deleted
2.2.3-11	Volume Concentrations of Fuel-Air Mixtures Versus Flammability Limit

- 2.2.3-12 Regulatory Guide 1.91 Allowable Distance and Actual Distance of Hazardous Chemicals
- 2.2.3-13 Transportation Sources Toxic Gas Release Information
- 2.2.3-14 Chemicals Analyzed That Are at Offsite Storage Facilities Quantities and Distances
- 2.2.3-15 Offsite Sources Toxic Gas Release Information
- 2.2.3-16 Deleted
- 2.2.3-17 Oils and Solvents Stored at the Combustion Turbine Plant
- 2.2.3-18 Onsite Chemical Storage
- 2.2.3-19 Parameters Used in Toxic Gas Analysis
- 2.2.3-20 Toxic Gas Release Information
- 2.2.3-21 Deleted
- 2.2.3-22 Deleted
- 2.3.1-1 Recurrence Intervals of Rainfall Falling in Periods of 30 Min to 10 Days
- 2.3.1-2 Mean Number of Days with Precipitation  $\ge 0.01$ ,  $\ge 0.1$ ,  $\ge 0.5$  in. in Augusta, Georgia
- 2.3.1-3 Table of Fujita-Pearson Tornado Scale
- 2.3.1-4 Tornado Wind Groups
- 2.3.1-5 Seasonal and Annual Frequency of Lightning Strikes to the Reactor Containment Building for Bolts of Various Current Magnitude
- 2.3.2-1 Normal, Means, and Extremes for Augusta, Georgia
- 2.3.2-2 VEGP Onsite Weather Instruments
- 2.3.2-3 1972 to 1973 Wind Persistence VEGP Meteorological Tower 33-ft Level
- 2.3.2-4 1972 to 1973 Wind Persistence VEGP Meteorological Tower 150-ft Level
- 2.3.2-5 1977 to 1978 Wind Persistence VEGP Meteorological Tower 33-ft Level
- 2.3.2-6 1977 to 1978 Wind Persistence VEGP Meteorological Tower 150-ft Level

- 2.3.2-7 1978 to 1979 Wind Persistence VEGP Meteorological Tower 33-ft Level
- 2.3.2-8 1978 to 1979 Wind Persistence VEGP Meteorological Tower 150-ft Level
- 2.3.2-9 1980 to 1981 Wind Persistence VEGP Meteorological Tower 33-ft Level
- 2.3.2-10 1980 to 1981 Wind Persistence VEGP Meteorological Tower 150-ft Level
- 2.3.2-11 1959 to 1963 Wind Persistence Augusta, Georgia
- 2.3.2-12 Precipitation Data for Augusta Area 1902 to 1950
- 2.3.2-13 Precipitation Data for Augusta Area 1940 to 1950
- 2.3.2-14 Joint Frequency Tables of Windspeed and Direction 33-ft Level
- 2.3.2-15 Joint Frequency Tables of Windspeed and Direction 33-ft Level
- 2.3.2-16 Joint Frequency Tables of Windspeed and Direction 150-ft Level
- 2.3.2-17 Joint Frequency Tables of Windspeed and Direction 150-ft Level
- 2.3.2-18 Joint Frequency Tables of Windspeed and Direction 100-ft Level
- 2.3.2-19 Joint Frequency Tables of Windspeed and Direction 33-ft Level
- 2.3.2-20 Joint Frequency Tables of Windspeed and Direction 150-ft Level
- 2.3.2-21 Seasonal and Annual Morning and Afternoon Mean Mixing Height in the VEGP Site Region
- 2.3.3-1 VEGP Meteorological Tower Summary of Data Recovery
- 2.3.4-1 Summary of Accident X/Q Valves
- 2.3.4-2 Assumed Distance to Site Boundary in Each Direction
- 2.3.4-3 Temperature Difference Groups for Determining Pasquill Stability Categories
- 2.3.5-1 List of Computer Runs at VEGP Site
- 2.3.5-2 Gaseous Discharge Point at VEGP Site
- 2.3.5-3 Vent Design Information at VEGP Site
- 2.3.5-4 Tabulation of Input Assumptions for Calculations at VEGP Site
- 2.3.5-5 Terrain Elevation above Plant Grade
- 2.3.5-6 Atmospheric Dispersion Factors for Vogtle
- 2.3.5-7 Atmospheric Dispersion Factors for Vogtle
- 2.3.5-8 Atmospheric Dispersion Factors for Vogtle
- 2.3.5-9 Atmospheric Dispersion Factors for Vogtle
- 2.3.5-10 Diffusion and Deposition Estimates for All Receptor Locations
- 2.3.5-11 Diffusion and Deposition Estimates for All Receptor Locations
- 2.3.5-12 Atmospheric Dispersion Factors 1980-81
- 2.4.1-1 Savannah River Subbasins and Drainage Areas above VEGP
- 2.4.1-2 Safety-Related Structures and Access to Them
- 2.4.1-3 Water Control Structures, Savannah River Basin
- 2.4.2-1 Gauging Station Records, Savannah River Basin, Savannah River at Augusta, Ga., Annual Flood Peaks
- 2.4.2-2 Probable Maximum Precipitation
- 2.4.2-3 Yard Drainage, Recommended 100-Year Rainfall Criteria
- 2.4.3-1 Probable Maximum Precipitation
- 2.4.3-2 6-h Sequential Incremental Probable Maximum Precipitation
- 2.4.3-3 Spatial Isohyetal Adjustments for the Three Greatest 6-h Probable Maximum Precipitation
- 2.4.3-4 Spatially Adjusted Three Greatest 6-h Probable Maximum Precipitation
- 2.4.3-5 6-h Incremental Probable Maximum Precipitation Arranged in Critical Sequence
- 2.4.3-6 Cumulative Probable Maximum Precipitation for Various Storm Positions and Calculation Procedures
- 2.4.7-1 Minimum Water Temperature of the Savannah River
- 2.4.12-1 Water-Bearing Properties of Materials Underlying VEGP and Vicinity
- 2.4.12-2 Ground Water Use
- 2.4.12-3 Water Quality Analyses VEGP Wells

- 2.4.12-4 Water Quality Analyses Domestic Wells
- 2.4.12-5 Water Quality Analyses Springs
- 2.4.12-6 Water Quality Analyses Surface Water
- 2.4.12-7A Observation Wells
- 2.4.12-7B Observation Wells in Confined Aquifers
- 2.4.12-7C Observation Wells in Marl
- 2.4.12-7D Water Supply Wells
- 2.4.12-8 Summary of Aquifer Characteristics Calculations
- 2.4.12-9 Observation Wells Monitoring Program 1993-Present
- 2.4.12-10 Permeability Tests Blue Bluff Marl
- 2.4.12-11 Porosity
- 2.4.12-12 Permeability Test Results
- 2.4.12-13 Permeability Tests
- 2.4.12-14 Porosity
- 2.4.12-15 Permeability of Backfill
- 2.4.13-1 Equilibrium Distribution Coefficient Power Block Excavation Backfill
- 2.5.1-1 Stratigraphic Units in the Vicinity of VEGP
- 2.5.2-1 Significant Earthquakes Within 200 Miles of the Site
- 2.5.2-2 Modified Mercalli Intensity Scale of 1931 (Abridged)
- 2.5.4-1 Engineering Properties of Site Soils
- 2.5.4-2 Engineering Properties for Design
- 2.5.4-3 Design Values of Shear Modulus
- 2.5.4-4 In Situ Soils Basic Soil Properties for Dynamic Design
- 2.5.4-5 Compilation of Shear Wave Data

LIST OF TABLES (CONTINUED)

Summary of Data Compilation Based on Cross-Hole Data

2.5.4-6

2.5.4-7	Elastic Moduli Results		
2.5.4-8	Design Static Properties for Backfill Compacted to 97-Percent Rela Compaction (ASTM D 1557)	ative	
2.5.4-9	Design Dynamic Properties for Backfill Compacted to 97-Percent R Compaction (ASTM D 1557)	Relative	
2.5.4-10	Summary of Test Fill Results for Heavy Equipment Compaction		
2.5.4-11	Summary of Test Fill Results for Hand Compaction Equipment		
2.5.4-12	Summary of Results of Bearing Capacity Analysis		
2.5.4-13	Summary of Measured Heave in Power Block		
2.5.4-14	Deleted		
2.5.4-15	Deleted		
2B-1	Drill Hole Summary		
2B-2	Core Sample Inventory		
3.2.2-1	Classification of Structures, Components, and Systems		
3.2.2-2	Principal Codes and Standards		
3.2.2-3	Summary of Construction Codes and Standards for VEGP Compor Quality Classification System	nents by l	NRC
3.5.1-1	Internally Generated Missiles Outside Containment Rotating Comp	onent Fa	ilures
3.5.1-2	Internally Generated Missiles Inside Containment Rotating Compor	nent Failu	ures
3.5.1-3	Internally Generated Missiles Inside Containment Pressurized Com Failures	nponent	
3.5.1-4	Turbine Missile Strike and Damage Probabilities Per Missile Fragm Unit 2	ent from	
3.5.1-5	Tornado Missiles Postulated at the Construction Permit Stage		
3.5.1-6	Tornado Missiles Considered in the VEGP Design		
3.5.1-7	Protected System and Component Barriers Against Externally Gen Missiles	erated	
	C	REV 24	10/22

- 3.5.1-8 Internal Rotating Components Excluded as Missile Generation Sources
- 3.6.1-1 Essential, High-Energy, and Moderate-Energy Systems
- 3.6.1-2 Design Comparison to Positions of NRC Branch Technical Positions ASB 3-1
- 3.6.1-3 Design Comparison to NRC Branch Technical Position MEB 3-1
- 3.6.2-1 High-Energy Pipe Break Stress Analysis Results
- 3.6.2-2 Hazard Analysis Results
- 3.6.2-3 Postulated Break Locations for the LOCA Analysis of the Primary Coolant Loop
- 3.7.B.1-1 Damping Values for Fixed Base Structures and Components
- 3.7.B.1-2 Embedment Depths of Category 1 Structures
- 3.7.N.1-1 Damping Values Used for Seismic Systems Analysis
- 3.7.B.2-1 Classification of Seismic Systems
- 3.7.B.2-2 Accelerations of Major Seismic Category 1 Structures
- 3.7.B.2-3 The Frequencies for Floor Response Spectra Calculations
- 3.7.4-1 Seismic Monitoring Instrumentation Requirements
- 3.8.1-1 Containment Load Combinations and Load Factors
- 3.8.4-1 Other Category 1 Structures Steel Design Load Combinations Elastic Method
- 3.8.4-2 Other Category 1 Structures Steel Design Load Combinations Plastic Method
- 3.8.4-3 Other Category 1 Structures Concrete Design Load Combinations Strength Method
- 3.8.5-1 Minimum Design Factors of Safety
- 3.9.N.1-1 Summary of RCS Design Transients
- 3.9.N.1-2 Monitored Component Cyclic or Transient Limits
- 3.9.N.1-3 Monitored Component Fatigue Locations
- 3.9.B.3-1 Load Combinations for ASME Section III, Division 1, Code Class 1, 2, and 3 Component and Component Supports

- 3.9.B.3-2 Stress Criteria for ASME III Class 1, 2, and 3 Piping
- 3.9.B.3-3 Stress Criteria for ASME III Class 1, 2, and 3 Vessels
- 3.9.B.3-4 Stress Criteria for ASME III Class 1, 2, and 3 Valves
- 3.9.B.3-5 Stress Criteria for ASME Section III, Division 1, Code Class 1, 2, and 3 Pumps
- 3.9.B.3-6 Stress Limits for Class 1 Component Supports
- 3.9.B.3-7 Stress Limits for Class 2 and 3 Component Supports
- 3.9.B.3-8 BOP Active Pumps
- 3.9.B.3-9 BOP Active Valves
- 3.9.B.3-10 Evaluation of NRC Regulatory Guide 1.148 (March 1981)
- 3.9.N.3-1 Active Pumps
- 3.9.N.3-2 Active Valves
- 3.9.N.3-3 Load Combinations and Acceptance Criteria for Pressurizer Safety and Relief Valve Piping Upstream of Valves
- 3.9.N.3-4 Load Combinations and Acceptance Criteria for Pressurizer Safety and Relief Valve Piping Seismically Designed Downstream Portion
- 3.9.N.3-5 Definitions of Load Abbreviations
- 3.9.5-1 Maximum Deflections Allowed for Reactor Internal Support Structures
- 3.10.B.1-1 Deleted
- 3.11.B.1-1 Environmental Conditions
- 3.11.B.1-2 Deleted
- 3.11.N.1-1 Deleted
- 3.11.B.3-1 Conformance to the Requirements of NUREG-0588
- 3.11.N.3-1 Comparison of NSSS EQ Program to NUREG-0588
- 3C-1 Ductility Ratios
- 3D-1 Records Used for Deconvolution Analyses
- 3D-2 Records Used for Upward Wave Propagation Analyses

3D-3	Computed Variations of Peak Accelerations (g) with Depth from Deconvolution Analyses
3D-4	Computed Variations of Peak Accelerations (g) with Depth from Upward Wave Propagation Analyses
3F-1	Sample of Hazards Analysis Results for Auxiliary Building (Levels B, C, and D)
3F-2	AFW Pump House Design Parameters
3F-3A	Unit 1 MSIV/MFIV Room Flooding (Case 2) and Pressure/Temperature (Case 1) Analysis
3F-3B	Unit 1 Control Building MSIV/MFIV Vault Area MSLB Pressure/Temperature (Case 3) Analysis
3F-4	Evaluation of RCS Loop Branch Line Breaks
3F-5	Jet Impingement Barriers
4.1-1	Reactor Design Comparison Table
4.1-2	Analytical Techniques in Core Design
4.1-3	Design Loading Conditions for Reactor Core Components
4.2-1	Fuel Assembly Component Stresses (Percent of Allowable)
4.3-1	Reactor Core Description (First Cycle)
4.3-2	Nuclear Design Parameters (First Cycle)
4.3-3	Reactor Requirements for Rod Cluster Control Assemblies
4.3-4	Deleted
4.3-5	Axial Stability Index Pressurized Water Reactor Core with a 12-ft Height
4.3-6	Typical Neutron Flux Levels (n/cm2/s) at Full Power
4.3-7	Comparison of Measured and Calculated Doppler Effects
4.3-8	Saxton Core II Isotopics, Rod MY, Axial Zone 6
4.3-9	Critical Boron Concentrations (ppm) HZP, BOL
4.3-10	Comparison of Measured and Calculated Ag-In-Cd Rod Worth

- 4.3-11 Comparison of Measured and Calculated Moderator Coefficients at HZP, BOL
- 4.3-12 Benchmark Critical Experiments
- 4.4-1 Thermal and Hydraulic Comparison Table
- 4.4-2 Void Fractions at Nominal Reactor Conditions
- 5.1.2-1 System Design and Operating Parameters
- 5.2.1-1 Applicable Code Addenda for RCS Components
- 5.2.3-1 Primary and Auxiliary Components Material Specifications
- 5.2.3-2 Reactor Vessel Internals Material Specifications
- 5.2.3-3 Recommended Reactor Coolant Water Chemistry Specification
- 5.3.1-1 Reactor Vessel Quality Assurance Program
- 5.3.1-2 VEGP Unit 1 Reactor Vessel Fracture Toughness Properties
- 5.3.1-3 VEGP Unit 2 Reactor Vessel Fracture Toughness Properties
- 5.3.1-4 VEGP Unit 1 Reactor Vessel Closure Head Bolting Material Properties
- 5.3.1-5 VEGP Unit 2 Reactor Vessel Closure Head Bolting Material Properties
- 5.3.1-6 Deleted
- 5.3.1-7 Reactor Vessel Material Surveillance Program
- 5.3.1-8 Reactor Vessel Material Surveillance Program Withdrawal Schedule (Unit 1)
- 5.3.1-9 Reactor Vessel Material Surveillance Program Withdrawal Schedule (Unit 2)
- 5.3.2-1 Deleted
- 5.3.2-2 Vogtle Unit 1 Reactor Vessel Core Beltline Region Toughness Properties
- 5.3.2-3 VEGP Unit 1 Reactor Vessel Core Beltline Region Weld Metal Toughness Properties
- 5.3.2-4 VEGP Unit 2 Reactor Vessel Core Beltline Region Toughness Properties
- 5.3.2-5 VEGP Unit 2 Reactor Vessel Core Beltline Region Weld Metal Toughness Properties
- 5.3.3-1 Reactor Vessel Design Parameters

- 5.3.3-2 Unit 1 Reactor Vessel Values for Analysis of Potential Pressurized Thermal Shock Events
- 5.3.3-3 Unit 2 Reactor Vessel Values for Analysis of Potential Pressurized Thermal Shock Events
- 5.4.1-1 Reactor Coolant Pump Design Parameters
- 5.4.1-2 Reactor Coolant Pump Quality Assurance Program
- 5.4.2-1 Steam Generator Design Data
- 5.4.2-2 Steam Generator Quality Assurance Program
- 5.4.3-1 Reactor Coolant Piping Design Parameters
- 5.4.3-2 Reactor Coolant Piping Quality Assurance Program
- 5.4.7-1 Design Bases for RHRS Operation
- 5.4.7-2 RHRS Component Data
- 5.4.7-3 Failure Modes and Effects Analysis Residual Heat Removal System Active Components Normal Cooldown Operation
- 5.4.7-4 Residual Heat Removal System Safety Grade Cold Shutdown Operations -Failure Modes and Effects Analysis
- 5.4.10-1 Pressurizer Design Data
- 5.4.10-2 Reactor Coolant System Design Pressure Settings
- 5.4.10-3 Pressurizer Quality Assurance Program
- 5.4.11-1 Pressurizer Relief Tank Design Data
- 5.4.11-2 Discharges to the Pressurizer Relief Tank
- 5.4.11-3 Pressurizer Relief Discharge System Nondestructive Testing Program
- 5.4.12-1 Reactor Coolant System Valve Design Parameters
- 5.4.12-2 Reactor Coolant System Valves Nondestructive Examination Program
- 5.4.12-3 Reactor Coolant System Pressure Isolation Valves
- 5.4.13-1 Pressurizer Safety and Relief Valves Design Parameters

- 5.4.15-1 Reactor Vessel Head Vent System Equipment Design Parameters
- 6.1.1-1 Principal ESF Pressure-Retaining Materials
- 6.1.1-2 Principal ESF Materials Exposed to Reactor Coolant or Containment Spray
- 6.1.2-1 Regulatory Guide 1.54, Revision 0, June 1973, Quality Assurance Requirements for Protective Coatings Applied to Water-Cooled Nuclear Power Plants
- 6.1.2-2 Containment Components Coating Schedule
- 6.1.2-3 Protective Coatings on Westinghouse-Supplied Equipment Inside Containment
- 6.1.2-4 Other Organic Materials
- 6.2.1-1 Containment Design Limits and Calculated Containment Peak Pressure and Temperature
- 6.2.1-2 Assumptions for Containment Analysis Part 1
- 6.2.1-3 Assumptions for Containment Analysis Part 2
- 6.2.1-4 Containment Structural Heat Sinks
- 6.2.1-5 Thermophysical Properties of Containment Heat Sinks
- 6.2.1-6 Accident Sequence for Double-Ended Pump Suction Break, Minimum Safety Injection
- 6.2.1.7 Deleted
- 6.2.1-8 Postulated Subcompartment Pipe Breaks
- 6.2.1-9 Reactor Cavity Flow Model: Node Characteristics (Initial Conditions Remain the Same)
- 6.2.1-10 Reactor Cavity Flow Model: Flow Characteristics
- 6.2.1-11 Steam Generator Compartment Model (Pipe Break Below el 220 ft): Node Characteristics Loop Closure Weld (336-in.2 Break)
- 6.2.1-11A Steam Generator Compartment Model (Pipe Break Above el 220 ft): Node Characteristics
- 6.2.1-12 Steam Generator Compartment Model (Pipe Break Below el 220 ft): Node Characteristics Steam Generator Inlet Nozzle (306-in.2 Break)
- 6.2.1-13 Steam Generator Compartment Model (Pipe Break Below el 220 ft): Node Characteristics Steam Generator Outlet Nozzle (436-in.2 Break)

6.2.1-14	Steam Generator Compartment Model (Pipe Break Below el 220 ft): Node Characteristics Reactor Coolant Pump Outlet Nozzle (236-in.2 Break)	
6.2.1-15	Steam Generator Compartment Model (Pipe Break Below el 220 ft): Node Characteristics Reactor Coolant Pump Inlet Nozzle (366-in.2 Break)	
6.2.1-16	Steam Generator Compartment Model (Pipe Break below el 220 ft): Node Characteristics Steam Generator Inlet Elbow (736-in.2 Break)	
6.2.1-16A	Steam Generator Compartment Model (Pipe Break Above el 220 ft): Flow Characteristics	
6.2.1-17	Steam Generator Compartment Model (Pipe Break Below el 220 ft): Flow Characteristics Steam Generator Inlet Elbow (763-in.2 Break)	
6.2.1-18	Steam Generator Compartment Model (Pipe Break Below el 220 ft): Flow Characteristics Steam Generator Inlet Nozzle (306-in.2 Break)	
6.2.1-19	Steam Generator Compartment Model (Pipe Break Below el 220 ft): Flow Characteristics Steam Generator Outlet Nozzle (436-in.2 Break)	
6.2.1-20	Steam Generator Compartment Model (Pipe Break Below el 220 ft): Flow Characteristics Reactor Coolant Pump Outlet Nozzle (236-in.2 Break)	
6.2.1-21	Steam Generator Compartment Model (Pipe Break Below el 220 ft): Flow Characteristics Reactor Coolant Pump Inlet Nozzle (336-in.2 Break)	
6.2.1-22	Steam Generator Compartment Model (Pipe Break Below el 220 ft): Flow Characteristics Loop Closure Weld (336-in.2 Break)	
6.2.1-23	Pressurizer Compartment Model: Flow Characteristics (Spray Line) Loss Coefficient (K)	
6.2.1-24	Pressurizer Compartment Flow Characteristic Mode (Surge Line) Loss Coefficient (K)	
6.2.1-25	Pressurizer Compartment Model: Node Characteristics	
6.2.1-26	Reactor Cavity Time Dependent Flow Conditions	
6.2.1-26A	Steam Generator Compartment Time Dependent Flow Conditions (16 in. Feedwater Line Condition)	
6.2.1-27	Steam Generator Compartment Time Dependent Flow Conditions (236 in. <sup>2</sup> Break)	
6.2.1-28	Pressurizer Compartment Mass and Energy Release Pressurizer Surge Line Break	

# LIST OF TABLES (CONTINUED)

6.2.1-28A	Pressurizer Compartment Mass and Energy Release Pressurizer Spray Line Break
6.2.1-29	Blowdown Mass and Energy Releases Double-Ended Pump Suction
6.2.1-30	Blowdown Mass and Energy Releases Double-Ended Hot Leg Guillotine
6.2.1-31	Reflood Mass and Energy Releases Double-Ended Pump Suction - Minimum Safety Injection
6.2.1-32	Deleted
6.2.1-33	Principal Parameters During Reflood Double-Ended Pump Suction - Minimum Safety Injection
6.2.1-34	Deleted
6.2.1-35	Post-Reflood Mass and Energy Releases Double-Ended Pump Suction - Minimum Safety Injection
6.2.1-36	Deleted
6.2.1-37	Mass Balance Double-Ended Pump Suction - Minimum Safety Injection
6.2.1-38	Deleted
6.2.1-39	Mass Balance Double-Ended Hot Leg Guillotine
6.2.1-40	Energy Balance Double-Ended Pump Suction - Minimum Safety Injection
6.2.1-41	Deleted
6.2.1-42	Energy Balance Double-Ended Hot Leg Guillotine
6.2.1-43	System Parameters
6.2.1-44	Deleted
6.2.1-45	Deleted
6.2.1-46	Deleted
6.2.1-47	Deleted
6.2.1-48	Deleted
6.2.1-49	Deleted

- 6.2.1-50 Deleted
- 6.2.1-51 Deleted
- 6.2.1-52 Deleted
- 6.2.1-53 Deleted
- 6.2.1-54 Deleted
- 6.2.1-55 Deleted
- 6.2.1-56 Deleted
- 6.2.1-57 Deleted
- 6.2.1-58 Deleted
- 6.2.1-59 Deleted
- 6.2.1-60 Spectrum of Secondary System Pipe Ruptures Analyzed
- 6.2.1-61 Mass and Energy Release Data for Case 15 Peak Calculated Containment Temperature for MSLB
- 6.2.1-62 Mass and Energy Release Data for Case 6 Peak Calculated Containment Temperature for MSLB
- 6.2.1-63 Deleted
- 6.2.1-64 Specific Plant Design Input for MSLB Analysis
- 6.2.1-65 Summary of Results for MSLB Containment Pressure Temperature Analysis
- 6.2.1-66 Sequence of Events for Case 11 Peak Calculated Containment Pressure for MSLB
- 6.2.1-67 Sequence of Events for Case 13 Peak Calculated Containment Temperature for MSLB
- 6.2.1-68 Blowdown Mass and Energy Releases for Limiting DECLG (CD = 0.6, Low T<sub>AVG</sub>)
- 6.2.1-69 Reflood Mass and Energy Releases DECLG (CD = 0.6, Low T<sub>AVG</sub>, Min SI)
- 6.2.1-70 Broken Loop Injection Spill During Blowdown (CD = 0.6, Low  $T_{AVG}$ )
- 6.2.1-71 Active Heat Sinks for Minimum Containment Pressure Analysis
- 6.2.1-72 Passive Heat Sinks

6.2.1-73	Forces and Moments for Broken Cold Leg with Support
6.2.1-74	Uplift Forces and Moments for Broken Cold Leg with Support
6.2.2-1	Containment Fan Cooler Design Data
6.2.2-2	Containment Fan Cooling Heat Removal Capacity (Post-LOCA Modes)
6.2.2-3	Containment Cooling System Failure Mode and Effects Analysis
6.2.2-4	Containment Spray System Component Design Parameters
6.2.2-5	Failure Mode and Effects Analysis - Containment Spray System - Active Components
6.2.4-1	Containment/Penetration/Isolation Valve Information
6.2.4-2	Containment Isolation Valves
6.2.5-1	Design Data for Hydrogen Recombiners
6.2.5-2	Containment Building - Failure Modes and Effects Analysis, Post-LOCA Cavity Purge System
6.2.5-3	Design Data for Principal Components of Post-LOCA Containment Hydrogen Purge System
6.2.5-4	Design Data for Principal Components of Post-LOCA Cavity Purge System
6.2.5-5	Design Data for Principal Components of the Containment Hydrogen Monitoring System
6.2.5-6	Plant Parameters Used To Calculate Post-Accident Hydrogen Production
6.2.5-7	Deleted
6.2.5-8	Deleted
6.2.5-9	Post-Accident Containment Temperature Transient for Hydrogen Generation Analysis
6.2.5-10	Corrosion Rates Used in the Post-Accident Containment Hydrogen Generation Analysis
6.2.6-1	Penetrations Not Vented to Containment or Drained During Type A Testing
6.3.2-1	Emergency Core Cooling System Component Parameters

- 6.3.2-2 Emergency Core Cooling System Relief Valve Data
- 6.3.2-3 Motor-Operated Isolation Valves in the Emergency Core Cooling System
- 6.3.2-4 Materials Employed for Emergency Core Cooling System Components
- 6.3.2-5 Emergency Core Cooling System Safeguards Operations-Failure Modes and Effects Analysis
- 6.3.2-6 Emergency Core Cooling System Recirculation Piping Passive Failure Analysis Long-Term Phase
- 6.3.2-7 Sequence of Switchover Operations
- 6.3.2-8 Emergency Core Cooling System Air-Operated Valves
- 6.3.2-9 Failure Modes and Effects Analysis for Safety-Grade Cold Shutdown Operations
- 6.3.3-1 Emergency Core Cooling System Shared Functions Evaluation
- 6.3.3-2 Normal Operating Status of Emergency Core Cooling System Components for Core Cooling
- 6.4.2-1 Performance Characteristics of Major System Components
- 6.4.2-2 Leakage Data and Leakage Rates
- 6.4.4-1 Control Room Emergency HVAC System Failure Modes and Effects Analysis
- 6.4.6-1 Control Room HVAC Indications and Alarms
- 6.5.1-1 ESF Filter System Design Parameters (For Unit 1 or 2)
- 6.5.1-2 ESF Filter System Instrumentation (Typical for Each Filter Train)
- 6.5.1-3 ESF Filter System Materials
- 6.5.2-1 Containment Unsprayed Regions
- 6.5.2-2 Input Parameters and Results of Spray Iodine Removal Analysis
- 6.5.3-1 Primary Containment Operation Following a Design Basis Accident
- 7.1.1-1 Listing of Criteria as Applied to Instrumentation and Control Systems
- 7.2.1-1 List of Reactor Trips
- 7.2.1-2 Protection System Interlocks

- 7.2.1-3 Reactor Trip System Instrumentation
- 7.2.1-4 Reactor Trip System Instrumentation Response Times
- 7.2.1-5 Reactor Trip Allocation Times
- 7.2.2-1 Reactor Trip Correlation
- 7.3.1-1 Instrumentation Operating Condition for Engineered Safety Features
- 7.3.1-2 Instrumentation Operating Conditions for Isolation Functions
- 7.3.1-3 Interlocks for Engineered Safety Features Actuation System
- 7.3.1-4 Primary System Accidents and Required Instrumentation Ruptures in Small Pipes, Cracks in Large Pipes, Ruptures of Large Pipes, and Steam Generator Tube Rupture
- 7.3.1-5 Secondary System Accidents and Required Instrumentation, Minor Secondary System Pipe Break, and Major Secondary System Pipe Break
- 7.3.1-6 Engineered Safety Features Response Items
- 7.3.1-7 Engineered Safety Features Allocation Times
- 7.3.3-1 Containment Combustible Gas Control System Actuated Equipment List
- 7.3.4-1 Containment Purge Isolation System Actuated Devices
- 7.3.5-1 Fuel Handling Building Ventilation Post-Accident Actuated Equipment
- 7.3.6-1 Control Room Ventilation Isolation Control System Monitor Sensitivities and Response Times
- 7.3.6-2 Control Room Ventilation Isolation Control System Actuated Equipment List
- 7.3.8-1 Main Steam, Mainsteam Bypass, and Main Feedwater and Feedwater Bypass Isolation Actuated Devices
- 7.3.9-1 NSCW Actuated Devices
- 7.3.10-1 Component Cooling Water System Actuated Devices
- 7.3.11-1 Containment Heat Removal System Actuated Devices
- 7.3.12-1 Control Building ESF HVAC System Actuated Equipment List
- 7.3.13-1 Auxiliary Building ESF HVAC System Actuated Equipment List

- 7.3.14-1 Auxiliary Feedwater Pumphouse ESF HVAC System Actuated Equipment List
- 7.3.15-1 Diesel Generator Building ESF HVAC System Actuated Equipment List
- 7.3.16-1 Electrical Tunnel ESF HVAC System Actuated Equipment List
- 7.3.17-1 Diesel Generator Fuel Oil System Actuated Equipment List
- 7.4.1-1 Systems Available for Safe Shutdown
- 7.4.2-1 Instrumentation and Controls Available for Hot Standby and Hot or Cold Shutdown
- 7.5.2-1 Post-Accident Monitoring Instrumentation
- 7.5.2-2 Summary of Selection of Criteria
- 7.5.2-3 Summary of Design, Qualification, and Interface Requirements
- 7.5.3-1 Summary of Type A Variables
- 7.5.3-2 Summary of Type B Variables
- 7.5.3-3 Summary of Type C Variables
- 7.5.3-4 Summary of Type D Variables
- 7.5.3-5 Summary of Type E Variables
- 7.5.4-1 Summary of Variables and Categories
- 7.5.4-2 NUREG-0737 Conformance
- 7.5.5-1 ESF Systems Monitored on the System Status Monitoring Panel
- 7.6.2-1 Interlock Table for Outer/Inner Isolation Valves
- 7.7.1-1 Plant Control System Interlocks
- 8.1-1 Acceptance Criteria and Guidelines for Electric Power Systems
- 8.2.1-1 Summary of 230- and 500-kV Line Construction
- 8.2.1-2 The Assignment of 230-kV Circuit Breaker Supplies to Substation Batteries
- 8.2.1-3 The Assignment of 500-kV Circuit Breaker Supplies to Substation Batteries
- 8.2.2-1 46-, 69-, 115-, 230-, and 500-kV Line Interruptions Caused by Lightning Interruptions for 100 Miles for Year 1979

- 8.2.2-2 Summary of Transmission Line Failures 1979
- 8.3.1-1 Diesel Generator Annunciator Points
- 8.3.1-2 Diesel Generator Loading Profile for LOCA and Loss of Offsite Power
- 8.3.1-3 Onsite Power System Failure Modes and Effects Analysis
- 8.3.1-4 Circuits Analyzed for Separation Requirements
- 8.3.2-1 125-V dc Battery A Load Requirements (LOCA/LOSP)
- 8.3.2-2 125-V dc Battery B Load Requirements (LOCA/LOSP)
- 8.3.2-3 125-V dc Battery C Load Requirements (LOCA/LOSP)
- 8.3.2-4 125-V dc Battery D Load Requirements (LOCA/LOSP)
- 8.3.2-5 Class 1E 125-V dc and 120-V Vital ac System Failure Modes and Effects Analysis
- 8.3.2-6 125-V dc Battery A Load Requirements (SBO)
- 8.3.2-7 125-V dc Battery B Load Requirements (SBO)
- 8.3.2-8 125-V dc Battery C Load Requirements (SBO)
- 8.3.2-9 125-V dc Battery D Load Requirements (SBO)
- 9.1.2-1A Loads and Load Combinations Unit 1 Pool
- 9.1.2-1B Loads and Load Combinations Unit 2 Pool
- 9.1.3-1 Spent Fuel Pool Cooling and Purification System Component Design Parameters
- 9.1.3-2 Failure Mode and Effects Analysis for Cooling Portion of SFPCPS
- 9.1.5-1 Spent Fuel Cask Bridge Crane and Polar Crane Data
- 9.1.5-2 Evaluation of Overhead Heavy Load Handling Systems
- 9.1.5-3 Containment Building Overhead Load Handling Systems
- 9.1.5-4 Fuel Building Cask and Fuel Handling Systems
- 9.1.5-5 Single Failure Proof Crane Conformance to NUREG-0554
- 9.1.5-6 NUREG 0612 Guidelines and Location of FSAR Description

#### LIST OF TABLES (CONTINUED)

9.1.5-7 Integrated Head and Internals 9.2.1-1 Nuclear Service Cooling Water System Heat Loads and Flows 9.2.1-2 Nuclear Service Cooling Water Failure Modes and Effects Analysis 9.2.1-3 NSCW System Component Data 9.2.2-1 CCW System Component Data 9.2.2-2 CCW System Heat Loads and Flows 9.2.2-3 Component Cooling Water System Failure Modes and Effects Analysis 9.2.3-1 Demineralized Water Makeup System Component Data 9.2.4-1 Potable and Sanitary Water System Component Data 9.2.5-1 **Ultimate Heat Sink Process Parameters** 9.2.5-2 Maximum Ultimate Heat Sink Heat Loads 9.2.5-3 Ultimate Heat Sink Design LOCA Case 9.2.5-4 Ultimate Heat Sink Performance (Post-LOCA) 9.2.5-5 Ultimate Heat Sink Maximum Temperature Case 9.2.5-6 Ultimate Heat Sink Design MSLBA Case 9.2.5-7 Ultimate Heat Sink Performance (Post-MSLB) 9.2.5-8 Meteorological Data for Augusta, Georgia, Airport (Maximum NSCW Temperatures) 9.2.5-9 Meteorological Data for Augusta, Georgia, Airport (Maximum NSCW Tower Basin Water Usage) 9.2.5-10 Total NSCW Heat Loads for Shutdown with Loss of Offsite Power 9.2.7-1 Summary of Primary Makeup Water Requirements 9.2.8-1 Auxiliary Component Cooling Water System Heat Loads and Flows 9.2.8-2 Auxiliary Component Cooling Water System Component Data 9.2.9-1 Essential Chilled Water System Design Data

- 9.2.9-2 Normal Chilled Water System Design Data
- 9.2.9-3 Essential Chilled Water System Failure Modes and Effects Analysis
- 9.2.10-1 Turbine Plant Closed Cooling Water System During Normal Power Generation Operation
- 9.2.11-1 Turbine Plant Cooling Water System Flow Requirements Normal Power Generation Operation
- 9.3.1-1 Component Description Compressed Air System
- 9.3.1-2 Safety-Related Pneumatically Operated Valves
- 9.3.2-1 Nuclear Sampling System Liquid Sample Points
- 9.3.2-2 Nuclear Sampling System Gaseous Sample Points
- 9.3.2-3 Turbine Plant Sampling System
- 9.3.2-4 Process Grab Sample Points
- 9.3.2-5 Radwaste Grab Sample Points
- 9.3.2-6 Post-Accident Sampling Sample Points
- 9.3.2-7 Environmentally Qualified Post-Accident Sampling Valves
- 9.3.3-1 Equipment and Floor Drainage System Component Data
- 9.3.3-2 Sump Parameters
- 9.3.3-3 Equipment and Floor Drainage Systems Failure Mode and Effects Analysis
- 9.3.4-1 Chemical and Volume Control System Design Parameters
- 9.3.4-2 Chemical and Volume Control System Principal Component Data Summary
- 9.3.4-3 Boron Recycle System Principal Component Data Summary
- 9.3.4-4 Differential Pressures
- 9.3.5-1 Design Parameters for Major Components of the Auxiliary Gas System
- 9.3.7-1 Chlorination System Component Data
- 9.4.1-1 Outside Design Conditions
- 9.4.1-2 Normal Environmental Design Conditions

#### LIST OF TABLES (CONTINUED)

9.4.1-3 Heating, Ventilation, Air-Conditioning, and Cooling Systems 9.4.1-4 Design Parameters for Onsite Technical Support Center HVAC System Components 9.4.2-1 Fuel Handling Building Ventilation System Component Data 9.4.2-2 Fuel Handling Building Ventilation Failure Modes and Effects Analysis 9.4.3-1 Auxiliary Building Normal Ventilation System Design Data for Major Components 9.4.3-2 ESF Room Cooler Design Data for Major Components 9.4.3-3 Auxiliary Building Emergency Filtration System Design Data for Major Components (Per Unit) 9.4.3-4 Auxiliary Building - ESF Room Cooler Failure Modes and Effects Analysis 9.4.3-5 Piping Penetration Filter Exhaust System Failure Modes and Effects Analysis 9.4.3-6 Alternate Radwaste Building Design Data for Major Components 9.4.3-7 Radwaste Processing Facility Design Data for Major Components 9.4.4-1 Turbine Building HVAC Systems, Condenser Vacuum Exhaust Filter System, and Steam Packing Exhauster Filter System Component Data 9.4.5-1 Engineered Safety Feature Ventilation System Component Data (Per Unit) 9.4.5-2 Control Building Safety Feature Electrical Equipment Room System Failure Modes and Effects Analysis 9.4.6-1 Design Data for Major Components of Containment Normal, Auxiliary, and Lower Level Heat Removal Systems 9.4.6-2 Design Data for Major Components of Containment Purge and Preaccess Filter Systems 9.4.6-3 Design Data for Major Components of Reactor Cavity, Reactor Support, and **CRDM** Cooling Systems 9.4.6-4 Comparison of Minipurge System with Branch Technical Position CSB6-4 9.4.7-1 Diesel Generator Building Ventilation System Component Data (Per Train) 9.4.7-2 Diesel Generator Building Ventilation System Failure Modes and Effects Analysis 9.4.8-1 Auxiliary Feedwater Pumphouse HVAC System Component Data (Per Unit)

cxvii

- 9.4.8-2 Auxiliary Feedwater Pumphouse HVAC System Failure Modes and Effects Analysis
- 9.4.9-1 Equipment Building Ventilation System Component Data
- 9.4.9-2 Electrical Tunnel Ventilation System Component Data (Per Unit)
- 9.4.9-3 Electrical Tunnel Ventilation System Single Failure Analysis
- 9.4.9-4 Piping Restraint and MSIV Ventilation Systems Component Data (Per Unit)
- 9.5.1-1 Fire Event Safe Shutdown Systems/Components
- 9.5.1-2 Fire Protection System Component Data
- 9.5.1-3 Safe Shutdown Function Success Paths
- 9.5.1-4 Fire Event Safe Shutdown Separation Analysis Methodology
- 9.5.1-5 Deleted
- 9.5.1-6 Areas Designed for Storage of Radioactive Material
- 9.5.1-7 Components Designed to Retain and Collect Radioactivity
- 9.5.1-8 Major Process Collection Points in Radioactive Waste Systems
- 9.5.1-9 Exceptions to NFPA Codes
- 9.5.1-10 Fire Protection Functionality Requirements
- 9.5.1-10A Fire Detection Instruments
- 9.5.1-10B Sprinkler Systems
- 9.5.1-10C Hose Stations
- 9.5.1-10D Yard Fire Hydrants and Associated Hydrant Hose Houses
- 9.5.2-1 Communication Equipment and Locations Available for Safe Shutdown
- 9.5.3-1 Emergency and Essential Lighting System Failure Modes and Effects Analysis
- 9.5.3-2 Illumination Levels of Emergency Lighting for the Main Control Board and Remote Shutdown Panels
- 9.5.4-1 Standby Diesel Generator Fuel Oil Storage and Transfer System Component Data (Quantities Per Unit)

- 9.5.4-2 Failure Modes and Effects Analysis
- 9.5.4-3 Standby Diesel Generator Fuel Oil Storage and Transfer System Indicating and Alarm Devices
- 9.5.5-1 Emergency Diesel Engine Cooling Water System Component Data (Per Diesel Engine)
- 9.5.5-2 Diesel Generator Cooling Water System Failure Modes and Effects Analysis (Preliminary)
- 9.5.6-1 Standby Diesel Generator Starting System Component Data
- 9.5.6-2 Diesel Generator Starting System Failure Modes and Effects Analysis
- 9.5.7-1 Emergency Diesel Engine Lubrication System Component Data
- 9.5.7-2 Diesel Generator Lubrication System Failure Modes and Effects Analysis
- 9.5.7-3 Emergency Diesel Generator Lubrication System Indicating Devices
- 9.5.8-1 Emergency Diesel Engine Combustion Air Intake and Exhaust System Component Data
- 9.5.8-2 Diesel Generator Combustion Air Intake and Exhaust System Failure Modes and Effects Analysis
- 9.5.9-1 Equipment Supplied by Auxiliary Steam System
- 9.5.10-1 Deleted
- 9A-1 Fire Area Index
- 9A-2 Fire Zone to Fire Area Cross Reference List
- 10.1-1 Significant Design Features and Performance Characteristics for Major Steam and Power Conversion System Components
- 10.2.2-1 Turbine-Generator and Auxiliaries Design Parameters
- 10.2.2-2 Events Following Loss of Turbine Load with Postulated Equipment Failures
- 10.3.2-1 Main Steam Supply System Design Data
- 10.3.2-2 Design Data for Main Steam Safety Valves
- 10.3.2-3 Description of Main Steam Piping

- 10.3.3-1 Main Steam System Failure Modes and Effects Analysis
- 10.3.5-1 Secondary Side Water Chemistry Specifications During Power Operation
- 10.4.1-1 Main Condenser Design Data
- 10.4.2-1 Condenser Air Ejection System Design Data
- 10.4.5-1 Design Parameters for Major Components of Circulating Water System
- 10.4.6-1 Condensate Filter/Demineralizer System Major Components
- 10.4.6-2 Impurity Concentrations in Influent Condensate
- 10.4.6-3 Water Chemistry for Reactor Coolant System
- 10.4.6-4 Water Chemistry for Circulating Water System
- 10.4.6-5 Deleted
- 10.4.7-1 Condensate and Feedwater System Component Failure Analysis
- 10.4.7-2 Condensate and Feedwater System Design Data For Major Components
- 10.4.7-3 Condensate and Feedwater System Controls and Indication
- 10.4.8-1 Parameters Used To Calculate Design Basis SGBPS Sources
- 10.4.8-2 Steam Generator Secondary Side Water Specific Activity
- 10.4.8-3 Steam Generator Blowdown Processing System Parameters
- 10.4.8-4 Steam Generator Blowdown Processing System Major Component Parameters
- 10.4.9-1 Auxiliary Feedwater System Component Data
- 10.4.9-2 NSSS Steam Generator Makeup Requirements
- 10.4.9-3 Auxiliary Feedwater Flow Information for Various Postulated Events
- 10.4.9-4 Auxiliary Feedwater System Failure Modes and Effects Analysis
- 10.4.9-5 Auxiliary Feedwater System Summary of Indication and Controls
- 10.4.10-1 Condensate and Feedwater Chemical Injection System
- 10A-1 AFWS Control Capabilities

- 10A-2 Electric Power Source List for the AFWS
- 10A-3 Component Groupings in AFWS Block Diagram
- 10A-4 AFWS Component Failure Data
- 11.1-1 Parameters Used in the Calculation of Design Basis Fission and Corrosion Product Activities
- 11.1-2 Reactor Coolant Design Basis Fission and Corrosion Product Specific Activity
- 11.1-3 Pressurizer Activities
- 11.1-4 Volume Control Tank Activities
- 11.1-5 Design Basis Iodine Concentrations in Secondary Side Water at Technical Specification Limits
- 11.1-6 Parameters Used To Describe the Reactor System Realistic Basis
- 11.1-7 Specific Activities in Principal Fluid Streams Realistic Basis
- 11.1-8 Parameters Specified by Regulatory Guide 1.112 Appendix B (Input Parameters for the GALE Computer Code)
- 11.1-9 Tritium Production
- 11.1-10 Expected Spent Resin Volume
- 11.2.1-1 Parameters Used in the Calculation of Estimated Activity in Liquid Wastes
- 11.2.1-2 Equipment Design Parameters
- 11.2.1-3 Estimated Liquid Waste Processing System Isotopic Inventories and Concentrations
- 11.2.1-4 Summary of Tank Level Indication, Level Annunciators, and Overflows
- 11.2.2-1 Range of Measured Decontamination Factors for Selected Isotopes
- 11.2.3-1 Calculated Releases of Radioactive Materials in Liquid Effluents (Per Unit) Assuming Expected Fuel Leakage
- 11.2.3-2 Comparison of Calculated Concentrations in Effluent Water Discharge with 10 CFR 20 Limits Assuming Expected Fuel Leakage
- 11.2.3-3 Radioactive Liquid Releases from Westinghouse-Designed Pressurized Water Reactor Plants (Zircaloy Fuel Cladding)

- 11.2.3-4 Estimated Individual Doses from Liquid Effluents as Low as Reasonably Achievable
- 11.3.2-1 Design Basis Accumulated Radioactivity in the GWPS after 40 Years Operation of Two Units
- 11.3.2-2 Expected Accumulated Radioactivity in the GWPS after 40 Years Operation of Two Units
- 11.3.2-3 Reduction in RCS Gaseous Fission Products Resulting from Normal Operation of the GWPS
- 11.3.2-4 Process Parameters for GWPS, Indefinite Holdup, 1-Month Accumulation
- 11.3.2-5 Process Parameters for GWPS, Indefinite Holdup, 90-Day Accumulation
- 11.3.2-6 Process Parameters for GWPS, Indefinite Holdup, 90-Day Holdup and Release
- 11.3.2-7 GWPS Component Description
- 11.3.3-1 Gaseous Releases from Westinghouse-Designed Operating Reactors
- 11.3.3-2 Expected Annual Average Release of Airborne Radionuclides
- 11.3.3-3 Comparison of Calculated Maximum Offsite Airborne Concentration with 10 CFR 20 Assuming Expected Fuel Leakage
- 11.3.3-4 Estimated Annual Doses to an Individual from Gaseous and Particulate Effluents
- 11.4.2-1 Parameters Used in the Calculation of Estimated Activity To Be Solidified
- 11.4.2-2 Estimated Expected Annual Activities of the Influents to the Solid Radwaste System
- 11.4.2-3 Estimated Maximum Annual Activities of the Influents to the Solid Radwaste System
- 11.4.2-4 Estimated Annual Quantities of Radwaste
- 11.4.2-5 Estimated Expected Annual Activities in Shipped Radwaste
- 11.4.2-6 Estimated Maximum Annual Activities in Shipped Radwaste
- 11.4.2-7 Summary of Backflushable Filters
- 11.5.2-1 Design Parameters for Process and Effluent Radiation Monitors
- 11.5.2-2 Detector Requirements for Process and Effluent Radiation Monitors

- 11.5.2-3 Conditions of Service for Process and Effluent Radiation Monitors
- 11.5.2-4 Deleted
- 11.5.2-5 Flowrate Monitoring for Liquid and Gaseous Normal Release Pathways
- 11.5.3-1 Airborne Radiological Sampling Capability for PERMS
- 11.5.3-2 Liquid Radiological Sampling Capability for PERMS
- 11.5.5-1 Design Parameters for Post-Accident Radiation Monitors
- 11.5.5-2 Detector Requirements for Post-Accident Radiation Monitors
- 11.5.5-3 Conditions of Service for Post-Accident Radiation Monitors
- 11.5.5-4 Deleted
- 12.2.1-1 Core Average Gamma Ray Source Strengths at Various Times after Shutdown
- 12.2.1-2 Radiation Sources Reactor Coolant Nitrogen-16 Activity
- 12.2.1-3 Nitrogen-16 Radiation Sources Pressurizer
- 12.2.1-4 Radiation Sources Pressurizer
- 12.2.1-5 Isotopic Composition and Specific Activity of Typical Out-of-Core Crud Deposits
- 12.2.1-6 Radiation Sources - CVCS: Regenerative Heat Exchanger Source Strengths; Regenerative Heat Exchanger Specific Activity; Excess Letdown Heat Exchanger (Tube Side) Source Strengths: Excess Letdown Heat Exchanger (Tube Side) Specific Activity; Letdown Heat Exchanger and Seal Water Heat Exchanger (Tube Side) Source Strengths; Letdown Heat Exchanger and Seal Water Heat Exchanger (Tube Side) Specific Activity; Letdown Reheat Heat Exchanger Source Strengths; Letdown Reheat Heat Exchanger Specific Activity; Letdown Chiller Heat Exchanger (Tube Side) and Moderating Heat Exchanger Source Strengths; Mixed Bed Demineralizer Source Strengths for 30 ft<sup>3</sup> of Resin; Cation Bed Demineralizer Source Strengths for 20 ft<sup>3</sup> of Resin; Boron Thermal Regeneration Demineralizer Source Strengths for 150 ft<sup>3</sup> of Resin; Chemical and Volume Control System Filter Source Strengths Assumed To Establish Shielding Requirements; Volume Control Tank Liquid Phase Source Strengths for a 200-ft<sup>3</sup> Liquid Phase: Volume Control Tank Vapor Phase Source Strengths for a 200-ft<sup>3</sup> Vapor Phase
- 12.2.1-7 Steam Generator Blowdown Demineralizer Source Strengths for 75 ft<sup>3</sup> of Resin
- 12.2.1-8 Steam Generator Blowdown Demineralizer Specific Activity
- 12.2.1-9 Steam Generator Blowdown Spent Resin Storage Tank Source Strengths

- 12.2.1-10 Steam Generator Blowdown Spent Resin Storage Tank Specific Activity
- 12.2.1-11 Steam Generator Blowdown Prefilter Source Strengths
- 12.2.1-12 Steam Generator Blowdown Outlet and Spent Resin Sluice Filter Source Strengths
- 12.2.1-13 Dimensions and Composition of Steam Generator Blowdown Purification System Filters
- 12.2.1-14 Deleted
- 12.2.1-15 Recycle Holdup Tank Specific Activities
- 12.2.1-16 Deleted
- 12.2.1-17 Deleted
- 12.2.1-18 Spent Fuel Gamma Ray Source Strengths at Various Times after Shutdown
- 12.2.1-19 Spent Fuel Pool Specific Activities for Determining Shield Wall Thicknesses
- 12.2.1-20 Spent Fuel Pool Demineralizer Source Strengths for 30 ft<sup>3</sup> of Resin
- 12.2.1-21 Spent Fuel Pool Demineralizer Specific Activity
- 12.2.1-22 Spent Fuel Pool Filter Source Strengths
- 12.2.1-23 Spent Fuel Pool Filter Dimensions and Composition Assumed To Establish Shielding Requirements
- 12.2.1-24 Deleted
- 12.2.1-25 Waste Monitor Tank Demineralizer Specific Activity
- 12.2.1-26 Reactor Coolant Drain Tank Liquid Phase Specific Activity
- 12.2.1-27 Reactor Coolant Drain Tank Vapor Phase Specific Activity
- 12.2.1-28 Floor Drain Tank, Waste Monitor Tanks, and Waste Holdup Tank Specific Activity
- 12.2.1-29 Chemical Drain Tank Specific Activity
- 12.2.1-30 Spent Resin Storage Tank Specific Activity
- 12.2.1-31 Deleted
- 12.2.1-32 Deleted

- 12.2.1-33 Radioactive Gas Distribution Gaseous Radwaste System Operating
- 12.2.1-34 Volume Control Tank Noble Gas Stripping Fractions
- 12.2.1-35 Waste Gas Processing System Kr-85 Inventory
- 12.2.1-36 Hydrogen Recombiner, Waste Gas Compressor, and Gas Decay Tank Specific Activity
- 12.2.1-37 Radiation Sources Residual Heat Removal System
- 12.2.1-38 Residual Heat Removal Loop Specific Activity 4 Hours after Shutdown
- 12.2.1-39 Residual Heat Removal Loop Specific Activity 8 Hours after Shutdown
- 12.2.1-40 Core Average and Spent Fuel Neutron Source Strengths at Various Times after Shutdown
- 12.2.1-41 Irradiated Sb-Be Secondary Source Rod Source Strengths
- 12.2.1-42 Hafnium Control Rod Source Strengths
- 12.2.1-43 Irradiated Type 304 Stainless Steel (0.12 Weight Percent Cobalt) Cladding Source Strengths
- 12.2.1-44 Irradiated Incore Detector and Drive Cable Maximum Withdrawal Source Strengths
- 12.2.1-45 Irradiated Incore Detector Drive Cable Source Strengths
- 12.2.1-46 Irradiated Type 316 Stainless Steel (0.12 Weight Percent Cobalt) Flux Thimble Source Strengths
- 12.2.1-47 Integrated Gamma Ray and Beta Source Strengths at Various Times Following a Design Basis Accident (TID-14844 Release Fractions)
- 12.2.1-48 Source Strength in the Residual Heat Removal Loop at Various Times Following an Equivalent Full-Core Meltdown Accident
- 12.2.1-49 Post-Accident Source Terms Specific Activities for TID-14844 Release Fractions
- 12.2.1-50 Refueling Water Storage Tank Specific Activities
- 12.2.1-51 Reactor Makeup Water Storage Tank Specific Activities
- 12.2.1-52 Condensate Storage Tank Specific Activities

- 12.2.2-1 Parameters and Assumptions for Calculating Airborne Radioactive Concentrations
- 12.2.2-2 Airborne Radioactivity Concentrations
- 12.3.1-1 Radiation Zones
- 12.3.1-2 Equipment Specification Limits for Cobalt Impurity Levels
- 12.3.1-3 Approximate Reactor Coolant System Wetted Surface Areas
- 12.3.1-4 Approximate Reactor Coolant System Wetted Surface Areas of Stellite
- 12.3.1-5 Post-Accident Radiation Levels in Vital Areas (NUREG-0737, 11.B.2 Shielding Review)
- 12.3.4-1 Conditions of Service for Area Radiation Monitors
- 12.3.4-2 Range and Control Functions for Area Radiation Monitors
- 12.4.1-1 Average Number of Personnel Per PWR Unit for the Period 1969-1981
- 12.4.1-2 Average Occupational Radiation Exposure (Man-Rem Dose) Per PWR Unit for the Period 1969-1981
- 12.4.1-3 Average Occupational Radiation Exposure (Man-Rem Dose) Based on PWR Plant Age
- 12.4.1-4 Distribution of the Number of Personnel (>100 mrem/Year) According to Work Function
- 12.4.1-5 Distribution of Personnel (#100 mrem/Year) According to Employee Category
- 12.4.1-6 Percentages of Personnel Dose by Work Function
- 12.4.1-7 Annual Occupational Exposures for Various PWR Vendor's Units
- 12.4.1-8 Average Annual Occupational Exposure and Power for Individual PWR
- 12.4.1-9 Average Individual Exposure Based on PWR Plant Age
- 12.4.1-10 Cumulative Average of Annual Exposure by Years of Operation
- 12.4.1-11 Occupational Radiation Exposure Estimate for Inservice Inspection and Special Maintenance (For One Unit)
- 12.4.1-12 Plant Staff Classification Working Time and Zone Distribution (VEGP Units 1 and 2)

- 12.4.1-13 Estimated Annual Gamma Dose to Plant Personnel (Extracted Number of Manhours of Occupancy Per Year)
- 12.4.1-14 Doses to Plant Personnel Caused by Airborne Radioactivity
- 12.4.1-15 Total Body Dose Factors
- 12.4.1-16 Assumptions Used To Determine Doses to Plant Personnel Caused by Airborne Radioactivity
- 12.4.3-1 Construction Worker Doses
- 12.4.3-2 Individual Dose
- 12.4.3-3 Near Field Dispersion (X/Q) and Deposition (D/Q) Values
- 12.4.4-1 Estimated Occupational Radiation Exposure During Decommissioning
- 12.4.4-2 Dominant Radioactive Isotopes at Decommissioning
- 12.5.2-1 Fixed Laboratory Instrumentation
- 12.5.2-2 Portable Health Physics Survey Instruments
- 12.5.2-3 Personnel Monitoring Equipment
- 12.5.2-4 Health Physics Equipment
- 13.1.3-1 Acceptable Education and Experience for Selection and Training of VEGP Personnel
- 14.2.1-1 Preoperational Test Procedures
- 14.2.1-2 Startup Test Procedures
- 15.0.3-1 Nuclear Steam Supply System Power Ratings
- 15.0.3-2 Summary of Initial Conditions and Computer Codes Used
- 15.0.3-3 Nominal Values of Pertinent Plant Parameters Utilized in the Accident Analyses
- 15.0.6-1 Trip Points and Time Delays to Trip Assumed in Accident Analyses
- 15.0.7-1 Determination of Maximum Overpower Trip Point, Power Range Neutron Flux Channel, Based on Nominal Setpoint Considering Inherent Instrumentation Errors
- 15.0.8-1 Plant Systems and Equipment Available for Transient and Accident Conditions

- 15.0.8-2 Fission Product Barriers for Transient and Accident Analyses.
- 15.0.9-1 Fuel and Rod Gap Inventories, Core (Ci)
- 15.0.12-1 Single Failures Assumed in Accident Analyses
- 15.0.13-1 Operator Actions Required for Small and Large LOCAs
- 15.0.13-2 Short-Term Operator Actions Required for Steam Generator Tube Rupture
- 15.1.2-1 Time Sequence of Events for Incidents That Result in an Increase in Heat Removal by the Secondary System
- 15.1.5-1 Equipment Required Following a Rupture of a Main Steam Line
- 15.1.5-2 Parameters Used in Evaluating the Radiological Consequences of a Main Steam Line Break
- 15.1.5-3 Radiological Consequences of a Main Steam Line Break
- 15.2.3-1 Time Sequence of Events for Incidents Which Result in a Decrease in Heat Removal by the Secondary System
- 15.2.6-1 Parameters Used in Evaluating Radiological Consequences of Nonemergency ac Power
- 15.2.6-2 Radiological Consequences of Loss of Nonemergency ac Power
- 15.3.1-1 Time Sequence of Events for Incidents Which Result in a Decrease in Reactor Coolant System Flowrate
- 15.3.3-1 Summary of Results for Locked Rotor Transients (Four Loops Operating Initially)
- 15.3.3-2 Parameters Used in Evaluating the Radiological Consequences of a Locked Rotor Accident
- 15.3.3-3 Radiological Consequences of a Locked Rotor Accident
- 15.4.1-1 Time Sequence of Events for Incidents Which Result in Reactivity and Power Distribution Anomalies
- 15.4.6-1 Parameters
- 15.4.8-1 Parameters Used in the Analysis of the Rod Cluster Control Assembly Ejection Accident
- 15.4.8-2 Parameters Used in Evaluating the Radiological Consequences of a Control Rod Ejection Accident

- 15.4.8-3 Radiological Consequences of a Control Rod Ejection Accident
- 15.5.1-1 Time Sequence of Events for Incidents Which Result in an Increase in Reactor Coolant Inventory
- 15.6.1-1 Time Sequence of Events for Incidents Which Cause a Decrease in Reactor Coolant Inventory
- 15.6.2-1 Parameters Used To Evaluate the Consequences of a Break in an RCS Instrument Line Outside the Containment
- 15.6.2-2 Parameters Used To Evaluate the Consequences of a Letdown Line Break Outside the Containment
- 15.6.2-3 Doses Resulting from an RCS Instrument Line Break Outside Containment
- 15.6.2-4 Doses Resulting from a Letdown Line Break Outside Containment
- 15.6.3-1 Operator Action Times for Design Basis SGTR Analysis
- 15.6.3-2 Sequence of Events
- 15.6.3-3 Mass Release Results
- 15.6.3-4 Parameters Used in Evaluating Radiological Consequences
- 15.6.3-5 Iodine Specific Activities in the Secondary Coolant Based on 1, 60, and 0.1  $\mu$ Ci/gram of D.E. I-131
- 15.6.3-6 Deleted
- 15.6.3-7 Deleted
- 15.6.3-8 Breathing Rates
- 15.6.3-9 Deleted
- 15.6.3-10 Whole Body and Beta Skin Dose Conversion Factors for Noble Gasses
- 15.6.3-11 Offsite Radiation Doses
- 15.6.5-1 Summary of Large Break LOCA Analysis Assumptions
- 15.6.5-2 Large-Break LOCA Results
- 15.6.5-3 Large Break LOCA Time Sequence of Events
- 15.6.5-4 Peak Clad Temperature Changes for Large Break LOCA Analysis

#### LIST OF TABLES (CONTINUED)

15.6.5-5 Summary of Small Break LOCA Analysis Assumptions 15.6.5-6 Small Break LOCA Low TAVG NOTRUMP Transient Results 15.6.5-7 Small Break LOCA Low TAVG Beginning of Life (BOL) Rod Heatup Results 15.6.5-8 Peak Clad Temperature Changes for Small Break LOCA Analysis 15.6.5-9 Parameters Used in Evaluating the Radiological Consequences of a Loss-of-Coolant Accident 15.6.5-10 Design Comparison to the Regulatory Positions of Regulatory Guide 1.4, Assumptions Used for Evaluating the Potential Radiological Consequences of a Loss-of-Coolant Accident for Pressurized Water Reactors, Revision 2, June 1974 15.6.5-11 Doses Resulting from a Loss-of-Coolant Accident 15.6.5-12 Activity Releases to the Environment Due to a Loss-of-Coolant Accident 15.7.1-1 Parameters Used in Evaluating the Radiological Consequences of a Waste Gas Decay Tank Rupture 15.7.1-2 Design Comparison to the Regulatory Positions of Regulatory Guide 1.24, Assumptions Used for Evaluating the Potential Radiological Consequences of a Pressurized Water Reactor Radioactive Gas Storage Tank Failure, Revision 0, March 23, 1972 15.7.1-3 Radiological Consequences of a Waste Gas Decay Tank Rupture 15.7.2-1 Parameters Used in Evaluating the Radiological Consequences of a Liquid Radwaste Tank Failure 15.7.2-2 Radiological Consequences of a Liquid Radwaste Tank Failure 15.7.3-1 Recycle Holdup Tank Data for Failure Analysis 15.7.4-1 Parameters Used in Evaluating the Radiological Consequences of a Fuel Handling Accident 15.7.4-2 Regulatory Guide 1.25, Assumptions Used for Evaluating the Potential Radiological Consequences of a Fuel Handling Accident in the Fuel Handling and Storage Facility for Boiling and Pressurized Water Reactors, Revision 0, Dated March 23, 1972 15.7.4-3 Deleted 15.7.4-4 Radiological Consequences of a Fuel Handling Accident

15A-1	Parameters Used in Accident Analysis
15A-2	Limiting Short-Term Atmospheric Dispersion Factors for Accident Analysis
15A-3	Core Fission Product Inventory
15A-4	Primary Coolant Noble Gas Concentrations
15A-5	Dose Conversion Factors Used in Accident Analysis
15A-6	Reactor Coolant lodine Concentrations for 1 $\mu Ci/g$ and 60 $\mu Ci/g$ of Dose Equivalent I-131
15A-7	lodine Appearance Rates in the Reactor Coolant (Ci/s)

- 18.1-1 Main Control Room Annunciator Alarm Sequence, First-Out Sequence of Operation (Modified)
- 18.1-2 Main Control Room Annunciator Alarm Sequence, Ring-Back Sequence of Operation
- 18.1-3 Main Control Room Annunciator Alarm Sequence

#### LIST OF FIGURES

- 2.1.1-1 Population Centers
- 2.1.1-2 Area Map
- 2.1.1-3 Area Map
- 2.1.3-1 Population Distribution Within 10 Miles
- 2.1.3-2 Population Between 16.1 and 80.5 Kilometers (10 and 50 Miles)
- 2.1.3-3 1987 Population Projections (50-Mile Radius)
- 2.1.3-4 2028 Population Projections (50-Mile Radius)
- 2.1.3-5 Population Between 10 and 50 Miles
- 2.2.1-1 Location of Major Industrial Facilities
- 2.2.1-2 Major Industrial Facilities, Military Bases, and Pipelines Within 25 Miles
- 2.2.1-3 Highway Transportation Routes, Airports, Dock, and Railroad Facilities Within 25 Miles
- 2.2.2-1 Savannah River Plant Site Map
- 2.2.2-2 Chem-Nuclear Systems, Inc. Land Utilization Plan
- 2.3.1-1 Total Number of Hail Reports 3/4 in. and Greater by 1° Squares 1955 to 1967
- 2.3.1-2 Total Number of Hail Reports 3/4 in. and Greater by 2° Squares 1955 to 1967
- 2.3.1-3 Snow Load in lb/ft<sup>2</sup> on the Ground, 100-Year Mean Recurrence Interval
- 2.3.1-4 Total Number Days with Freezing Precipitation 1939 to 1948
- 2.3.1-5 Extreme Radial Thickness of Glaze on Utility Wires 1928-29 to 1936-37
- 2.3.1-6 Tropical Storm and Hurricane Probability
- 2.3.1-7 Total Tornadoes by 1° Squares 1955 to 1967
- 2.3.1-8 Total Tornadoes by 2° Squares 1955 to 1967
- 2.3.1-9 Total Windstorms 50 Knots and Greater by 2° Squares 1955 to 1967
- 2.3.1-10 Total Windstorms 50 Knots and Greater by 1° Squares 1955 to 1967
- 2.3.1-11 Tornado Wind Groups

# LIST OF FIGURES (CONTINUED)

2.3.2-1	Monthly Wind Roses from Augusta Airport (1959 to 1963)
2.3.2-2	Seasonal Wind Roses for Augusta Airport (1959 to 1963)
2.3.2-3	Annual Wind Rose for Augusta Airport (1959 to 1963)
2.3.2-4	33-ft Level Monthly Wind Rose 1972 to 1973 VEGP Site Data
2.3.2-5	33-ft Level Seasonal Wind Rose 1972 to 1973 VEGP Site Data
2.3.2-6	33-ft Level Annual Wind Rose 1972 to 1973 VEGP Site Data
2.3.2-7	150-ft Level Monthly Wind Rose 1972 to 1973 VEGP Site Data
2.3.2-8	150-ft Level Seasonal Wind Rose 1972 to 1973 VEGP Site Data
2.3.2-9	150-ft Level Annual Wind Rose 1972 to 1973 VEGP Site Data
2.3.2-10	33-ft Level Monthly Wind Rose 1977 to 1978 VEGP Site Data
2.3.2-11	33-ft Level Seasonal Wind Rose 1977 to 1978 VEGP Site Data
2.3.2-12	33-ft Level Annual Wind Rose 1977 to 1978 VEGP Site Data
2.3.2-13	150-ft Level Monthly Wind Rose 1977 to 1978 VEGP Site Data
2.3.2-14	150-ft Level Seasonal Wind Rose 1977 to 1978 VEGP Site Data
2.3.2-15	150-ft Level Annual Wind Rose 1978 to 1979 VEGP Site Data
2.3.2-16	33-ft Level Monthly Wind Rose 1978 to 1979 VEGP Site Data
2.3.2-17	33-ft Level Seasonal Wind Rose 1978 to 1979 VEGP Site Data
2.3.2-18	33-ft Level Annual Wind Rose 1978 to 1979 VEGP Site Data
2.3.2-19	150-ft Level Monthly Wind Rose 1978 to 1979 VEGP Site Data
2.3.2-20	150-ft Level Seasonal Wind Rose 1978 to 1979 VEGP Site Data
2.3.2-21	150-ft Level Annual Wind Rose 1978 to 1979 VEGP Site Data
2.3.2-22	33-ft Level Monthly Wind Rose 1980 to 1981 VEGP Site Data
2.3.2-23	33-ft Level Seasonal Wind Rose 1980 to 1981 VEGP Site Data
2.3.2-24	33-ft Level Annual Wind Rose 1980 to 1981 VEGP Site Data
2.3.2-25	150-ft Level Monthly Wind Rose 1980 to 1981 VEGP Site Data
2.3.2-26	150-ft Level Seasonal Wind Rose 1980 to 1981 VEGP Site Data
----------	---
2.3.2-27	150-ft Level Annual Wind Rose 1980 to 1981 VEGP Site Data
2.3.2-28	Monthly Average and Average Daily Extremes of Dry Bulb Temperature
2.3.2-29	Monthly Average and Average of Daily Extremes of Dry Bulb Temperature
2.3.2-30	Monthly Average and Average of Daily Extremes of Wet Bulb Temperature
2.3.2-31	Monthly Average and Average Daily Extremes of Dewpoint Temperature
2.3.2-32	Monthly Average and Average of Daily Extremes of Wet Bulb Temperature
2.3.2-33	Monthly Average and Average Daily Extremes of Dewpoint Temperature
2.3.2-34	Monthly Average and Average Daily Extremes of Dry Bulb Temperature
2.3.2-35	Monthly Average and Average Daily Extremes of Wet Bulb Temperature
2.3.2-36	Monthly Average and Average Daily Extremes of Dewpoint Temperature
2.3.2-37	Monthly Average and Averages of Daily Extremes of Relative Humidity
2.3.2-38	Monthly Average and Average of Daily Extremes of Relative Humidity
2.3.2-39	Monthly Average and Average Daily Extremes of Relative Humidity
2.3.2-40	Monthly Average and Averages of Daily Extremes of Absolute Humidity
2.3.2-41	Monthly Average and Average of Daily Extremes of Absolute Humidity
2.3.2-42	Monthly Average and Average Daily Extremes of Absolute Humidity
2.3.2-43	Monthly Average and 24-Hour Maximum Precipitation
2.3.2-44	Precipitation Wind Roses for Augusta Airport (1959 to 1963)
2.3.2-45	Precipitation Wind Roses for Augusta Airport (1959 to 1963)
2.3.2-46	33-ft Level Precipitation Wind Rose 1972 to 1973
2.3.2-47	33-ft Level Precipitation Wind Rose 1972 to 1973
2.3.2-48	33-ft Level Precipitation Wind Rose 1977 to 1978
2.3.2-49	33-ft Level Precipitation Wind Rose 1977 to 1978

- 2.3.2-50 33-ft Level Precipitation Wind Rose 1978 to 1979
- 2.3.2-51 33-ft Level Precipitation Wind Rose 1978 to 1979
- 2.3.2-52 Precipitation Wind Rose Seasonal 1980-81
- 2.3.2-53 Precipitation Wind Rose Annual 1980-81
- 2.3.2-54 Monthly Average and Averages of Daily Extremes of Visibility
- 2.3.2-55 Topographic Features 0- to 5-Mile Radius from Plant Site
- 2.3.2-56 Topographic Features 0- to 50-Mile Radius from Plant Site
- 2.3.2-57 Topographic Features
- 2.3.4-1 Cumulative Probability of Hourly X/Q Values
- 2.3.4-2 Cumulative Distribution of X/Q for Directions (N through SSE)
- 2.3.4-3 Cumulative Distribution of X/Q for Directions (N through SSE)
- 2.3.4-4 Cumulative Distribution of X/Q for Directions (N through SSE)
- 2.3.4-5 Plot of X/Q at the LPZ vs. Averaging Times for All Direction Sectors
- 2.3.4-6 Plot of X/Q at the LPZ vs. Averaging Times for All Direction Sectors
- 2.3.4-7 Plot of X/Q at the LPZ vs. Averaging Times for All Direction Sectors
- 2.3.4-8 Plot of X/Q vs Averaging Time Based on Vogtle Site Data (4-Year Composite)
- 2.3.5-1 Isopleth of Annual Average X/Q
- 2.3.5-2 Isopleth of Annual Average Depleted X/Q
- 2.3.5-3 Isopleth of Annual Average Deposition (D/Q)
- 2.3.5-4 Isopleth of Annual Average X/Q
- 2.3.5-5 Isopleth of Annual Average Depleted X/Q
- 2.3.5-6 Isopleth of Annual Average Deposition D/Q
- 2.4.1-1 Hydrologic Map, Savannah River Basin Above Site
- 2.4.1-2 Savannah River Basin
- 2.4.3-1 Depth-Area-Duration Envelopes

- 2.4.3-2 PMP Depth-Duration Curve for Drainage Area (8015 mi<sup>2</sup>) above the Plant
- 2.4.3-3 Savannah River Steady Flow Condition-Discharge Relation at River Mile 151.1
- 2.4.3-4 Probable Maximum Flood Hydrograph at River Mile 151.1 by Routing Corps of Engineers PMF Outflow Hydrograph (Valley Storage Taken Into Account) at Clark Hill Dam Through Savannah River
- 2.4.4-1 Location of Dams on the Savannah River
- 2.4.4-2 Savannah River Stream Profile
- 2.4.4-3 Clark Hill Dam Plan and Section
- 2.4.4-4 Area and Capacity Curves for Major Upstream Dams
- 2.4.4-5 Clark Hill Tailwater Rating Curve
- 2.4.4-6 Stability of Clark Hill Dam Spillway Section
- 2.4.4-7 Standard Project Flood at River Mile 151.1
- 2.4.4-8 Effect of Dam Failure at VEGP Site
- 2.4.11-1 Flow Duration Curve, Savannah River at Butler Creek River Mile 187.0
- 2.4.11-2 Savannah River Low Flow Rating Curves at Intake Structure River Mile 151.1
- 2.4.11-3 Savannah River Intermediate Flow Rating Curve at Intake Structure River Mile 151.1
- 2.4.11-4 Locations of Sections for Low Water Study
- 2.4.11-5 VEGP Low Water Study
- 2.4.11-6 Savannah River at Augusta, Gauge Readings for September 23, 24, and 25, 1939, Minimum Flow of Record
- 2.4.12-1 Site Area
- 2.5.4-1 Settlement of Units 1 and 2 Power Block Structures
- 2.5.4-2 Heave Points Location Plan
- 2.5.4-3 Young's Modulus vs. Depth
- 2B-1 Heave Point Locations

- 3.5.1-1 Missile Ejection Angles
- 3.5.1-2 Missile Trajectories
- 3.6.2-1 Loss of Reactor Coolant Accident Boundary Limits
- 3.6.2-2 Typical U-Bar Restraint
- 3.6.2-3 Typical EAM Restraint
- 3.7.B.1-1 Safe Shutdown Earthquake Horizontal Response Spectra
- 3.7.B.1-2 Safe Shutdown Earthquake Vertical Response Spectra
- 3.7.B.1-3 Operating Basis Earthquake Horizontal Response Spectra
- 3.7.B.1-4 Operating Basis Earthquake Vertical Response Spectra
- 3.7.B.1-5 Synthetic Time History Horizontal Direction
- 3.7.B.1-6 Synthetic Time History Vertical Direction
- 3.7.B.1-7 Damping Values for Cable Trays and Supports
- 3.7.B.1-8 Damping Ratios vs. Shear Strain for Compacted Sand Backfill
- 3.7.B.1-9 Damping Ratios vs. Shear Strain for Clay Marl Bearing Stratum
- 3.7.B.1-10 Damping Ratios vs. Shear Strain for Lower Sand Stratum
- 3.7.B.1-11 Alternate Damping Values for ASME Boiler and Pressure Vessel Code, Section III, Division 1, Code Class 1,2, and 3 Piping Systems
- 3.7.N.1-1 Alternate Damping Values Used for Seismic Analysis of Piping for Westinghouse-Analyzed Piping Systems\*
- 3.7.B.2-1 Lumped Parameter Model of Diesel Generator Building
- 3.7.B.2-2 Power Block Plan View Showing Sections for Finite Element Soil-Structure Interaction Flush Models
- 3.7.B.2-3 Flush Model Along Section 2 of Figure 3.7.B.2-2
- 3.7.B.2-4 Flush Model Along Section 4 of Figure 3.7.B.2-2
- 3.7.B.2-5 G/Gmax vs. Shear Strain for Compacted Sand Backfill
- 3.7.B.2-6 G/Gmax vs. Shear Strain for Saturated Clay Marl-Bearing Stratum

- 3.7.B.2-7 G/Gmax vs. Shear Strain for Lower Sand Stratum
- 3.7.B.2-8 Auxiliary Building el 109 ft 0 in., SSE Horizontal
- 3.7.B.2-9 Nuclear Service Cooling Water Tower el 128 ft 0 in., SSE Horizontal
- 3.7.B.2-10 Containment Building el 144 ft 0 in., SSE Horizontal
- 3.7.B.2-11 Fuel Handling Building el 152 ft 0 in., SSE Horizontal
- 3.7.B.2-12 Control Building el 173 ft 0 in., SSE Horizontal
- 3.7.B.2-13 Auxiliary Building el 109 ft 0 in., OBE Horizontal
- 3.7.B.2-14 Nuclear Service Cooling Water Tower el 128 ft 0 in., OBE Horizontal
- 3.7.B.2-15 Containment Building el 144 ft 0 in., OBE Horizontal
- 3.7.B.2-16 Fuel Handling Building el 152 ft 0 in., OBE Horizontal
- 3.7.B.2-17 Control Building el 173 ft 0 in., OBE Horizontal
- 3.7.B.2-18 Auxiliary Building el 109 ft 0 in., SSE Vertical
- 3.7.B.2-19 Nuclear Service Cooling Water Tower el 128 ft 0 in., SSE Vertical
- 3.7.B.2-20 Containment Building el 144 ft 0 in., SSE Vertical
- 3.7.B.2-21 Fuel Handling Building el 152 ft 0 in., SSE Vertical
- 3.7.B.2-22 Control Building el 173 ft 0 in., SSE Vertical
- 3.7.B.2-23 Auxiliary Building el 109 ft 0 in., OBE Vertical
- 3.7.B.2-24 Nuclear Service Cooling Water Tower el 128 ft 0 in., OBE Vertical
- 3.7.B.2-25 Containment Building el 144 ft 0 in., OBE Vertical
- 3.7.B.2-26 Fuel Handling Building el 152 ft 0 in., OBE Vertical
- 3.7.B.2-27 Control Building el 173 ft 0 in., OBE Vertical
- 3.7.B.2-28 Soil-Structure Interaction Model Common Nodes
- 3.7.N.2-1 Multidegree-of-Freedom System
- 3.7.4-1 Seismic Monitoring Instrument Locations
- 3.7.4-2 Deleted

- 3.8.1-1 Temperature Gradients Through Containment Wall for LOCA and for Main Steam Line Break
- 3.8.1-2 Deleted
- 3.9.N.1-1 Reactor Coolant Loop Supports System Dynamic Structural Model
- 3.9.N.1-2 RPV Shell Submodel
- 3.9.N.1-3 Core Barrel Submodel
- 3.9.N.1-4 Internals Submodel
- 3.9.B.3-1 Pipe Support Reference Sketches
- 3.9.B.3-2 Pump Components Showing ASME Section III Jurisdictional Boundary (Typical Sketch)
- 3.9.B.3-3 Components Such as Heat Exchangers and Tanks Showing ASME Section III NF Jurisdictional Boundary (Typical Sketch)
- 3.9.N.3-1 Class 1 Piping Layout Safety Lines
- 3.9.N.3-2 Class 1 Piping Layout Relief Lines
- 3.9.N.3-3 Downstream Piping Layout Safety and Relief Lines
- 3.9.4-1 Full-Length Control Rod Drive Mechanism
- 3.9.4-2 Full-Length Control Rod Drive Mechanism Schematic
- 3.9.4-3 Nominal Latch Clearance at Minimum and Maximum Temperatures
- 3.9.4-4 Control Rod Drive Mechanism Latch Clearance Thermal Effect
- 3.9.5-1 Lower Internals Assembly
- 3.9.5-2 Upper Core Support Assembly
- 3.9.5-3 Plan View of Upper Core Support Assembly
- 3.11.B.1-1 Temperature Conditions for Equipment Qualification in the Auxiliary Building; Pressure Conditions for Equipment Qualification in the Auxiliary Building; Temperature Conditions for Equipment Qualification Inside MSIV or MFIV Area; Pressure Conditions for Equipment Qualification Inside MSIV or MFIV Area; Temperature Conditions for Environmental Qualification Inside Containment (Vapor Region); Temperature Conditions for Equipment Qualification Inside Containment (Sump Region); Pressure Conditions for Equipment Qualification

	Inside Containment; Temperature Conditions for Equipment Qualification in the Auxiliary Building; LOCA Temperature Conditions for Equipment Qualification
3C-1	Maximum Allowable Ductility Ratio for Reinforced Concrete Section With Beam- Column Action
3D-1	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 169 ft 0 in.
3D-2	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 169 ft 0 in.
3D-3	Operating Basis Earthquake Horiz. Accel. Response Spectra el 169 ft 0 in.
3D-4	Operating Basis Earthquake Vertical Accel. Response Spectra el 169 ft 0 in.
3D-5	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 220 ft 0 in.
3D-6	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 220 ft 0 in.
3D-7	Operating Basis Earthquake Horiz. Accel. Response Spectra el 220 ft 0 in.
3D-8	Operating Basis Earthquake Vertical Accel. Response Spectra el 220 ft 0 in.
3D-9	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 261 ft 0 in.
3D-10	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 261 ft 0 in.
3D-11	Operating Basis Earthquake Horiz. Accel. Response Spectra el 261 ft 0 in.
3D-12	Operating Basis Earthquake Vertical Accel. Response Spectra el 261 ft 0 in.
3D-13	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 220 ft 0 in.
3D-14	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 220 ft 0 in.
3D-15	Operating Basis Earthquake Vertical Accel. Response Spectra el 220 ft 0 in.
3D-16	Operating Basis Earthquake Horiz. Accel. Response Spectra el 220 ft 0 in.
3D-17	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 258 ft 0 in.
3D-18	Operating Basis Earthquake Horiz. Accel. Response Spectra el 258 ft 0 in.
3D-19	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 323 ft 0 in.
3D-20	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 323 ft 0 in.
3D-21	Operating Basis Earthquake Horiz. Accel. Response Spectra el 323 ft 0 in.
3D-22	Operating Basis Earthquake Vertical Accel. Response Spectra el 323 ft 0 in.

3D-23	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 400 ft 0 in.
3D-24	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 400 ft 0 in.
3D-25	Operating Basis Earthquake Horiz. Accel. Response Spectra el 400 ft 0 in.
3D-26	Operating Basis Earthquake Vertical Accel. Response Spectra el 400 ft 0 in.
3D-27	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 180 ft 0 in.
3D-28	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 180 ft 0 in.
3D-29	Operating Basis Earthquake Horiz. Accel. Response Spectra el 180 ft 0 in.
3D-30	Operating Basis Earthquake Vertical Accel. Response Spectra el 180 ft 0 in.
3D-31	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 220 ft 0 in.
3D-32	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 220 ft 0 in.
3D-33	Operating Basis Earthquake Horiz. Accel. Response Spectra el 220 ft 0 in.
3D-34	Operating Basis Earthquake Vertical Accel. Response Spectra el 220 ft 0 in.
3D-35	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 260 ft 0 in.
3D-36	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 260 ft 0 in.
3D-37	Operating Basis Earthquake Horiz. Accel. Response Spectra 260 ft. 0 in.
3D-38	Operating Basis Earthquake Vertical Accel. Response Spectra 260 ft 0 in.
3D-39	Safe Shutdown Earthquake Horiz. Accel. Response Spectra 143 ft 6 in.
3D-40	Safe Shutdown Earthquake Vertical Accel. Response Spectra 143 ft 6 in.
3D-41	Operating Basis Earthquake Horiz. Accel. Response Spectra el 143 ft 6 in.
3D-42	Operating Basis Earthquake Vertical Accel. Response Spectra el 143 ft 6 in.
3D-43	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 220 ft 0 in.
3D-44	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 220 ft 0 in.
3D-45	Operating Basis Earthquake Horiz. Accel. Response Spectra el 220 ft 0 in.
3D-46	Operating Basis Earthquake Vertical Accel. Response Spectra el 220 ft 0 in.

3D-47	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 288 ft 3 in.
3D-48	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 288 ft 3 in.
3D-49	Operating Basis Earthquake Horiz. Accel. Response Spectra el 288 ft 3 in.
3D-50	Operating Basis Earthquake Vertical Accel. Response Spectra el 288 ft 3 in.
3D-51	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 160 ft 0 in.
3D-52	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 160 ft 0 in.
3D-53	Operating Basis Earthquake Horiz. Accel. Response Spectra el 160 ft 0 in.
3D-54	Operating Basis Earthquake Vertical Accel. Response Spectra el 160 ft 0 in.
3D-55	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 220 ft 0 in.
3D-56	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 220 ft 0 in.
3D-57	Operating Basis Earthquake Horiz. Accel. Response Spectra el 220 ft 0 in.
3D-58	Operating Basis Earthquake Vertical Accel. Response Spectra el 220 ft 0 in.
3D-59	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 288 ft 2 in.
3D-60	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 288 ft 2 in.
3D-61	Operating Basis Earthquake Horiz. Accel. Response Spectra el 288 ft 2 in.
3D-62	Operating Basis Earthquake Vertical Accel. Response Spectra el 288 ft 2 in.
3D-63	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 137 ft 0 in.
3D-64	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 137 ft 0 in.
3D-65	Operating Basis Earthquake Horiz. Accel. Response Spectra el 137 ft 0 in.
3D-66	Operating Basis Earthquake Vertical Accel. Response Spectra el 137 ft 0 in.
3D-67	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 218 ft 6 in.
3D-68	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 218 ft 6 in.
3D-69	Operating Basis Earthquake Horiz. Accel. Response Spectra el 218 ft 6 in.
3D-70	Operating Basis Earthquake Vertical Accel. Response Spectra el 218 ft 6 in.
3D-71	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 219 ft 0 in.

3D-72	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 219 ft 0 in.
3D-73	Operating Basis Earthquake Horiz. Accel. Response Spectra el 219 ft 0 in.
3D-74	Operating Basis Earthquake Vertical Accel. Response Spectra el 219 ft 0 in.
3D-75	Safe Shutdown Earthquake Horiz. Accel. Response Spectra el 254 ft 0 in.
3D-76	Safe Shutdown Earthquake Vertical Accel. Response Spectra el 254 ft 0 in. and el 274 ft 0 in.
3D-77	Operating Basis Earthquake Horiz. Accel. Response Spectra el 254 ft 0 in.
3D-78	Operating Basis Earthquake Vertical Accel. Response Spectra el 254 ft 0 in. and el 274 ft 0 in.
3D-79	Containment Building el 169 ft Basemat
3D-80	Containment Building el 169 ft Basemat
3D-81	Containment Building el 169 ft Basemat
3D-82	Containment Building el 195 ft Internal Str.
3D-83	Containment Building el 195 ft Internal Str.
3D-84	Containment Building el 195 ft Internal Str.
3D-85	Containment Building el 220 ft Operating LvI.
3D-86	Containment Building el 220 ft Operating LvI.
3D-87	Containment Building el 220 ft Operating Lvl.
3D-88	Containment Building el 323 ft Polar Crane
3D-89	Containment Building el 323 ft Polar Crane
3D-90	Containment Building el 323 ft Polar Crane
3D-91	Control Building el 180 ft Basemat
3D-92	Control Building el 180 ft Basemat
3D-93	Control Building el 180 ft Basemat
3D-94	Control Building el 220 ft

- 3D-95 Control Building el 220 ft
- 3D-96 Control Building el 220 ft
- 3D-97 Control Building el 260 ft
- 3D-98 Control Building el 260 ft
- 3D-99 Control Building el 260 ft
- 3D-100 Ground Motion Analyses
- 3D-101 Ground Motion Analyses
- 3D-102 Ground Motion Analyses
- 3D-103 Ground Motion Analyses
- 3D-104 Ground Motion Analyses
- 3D-105 Ground Motion Analyses
- 3D-106 Ground Motion Analyses
- 3D-107 Ground Motion Analyses
- 3D-108 Ground Motion Analyses
- 3D-109 Ground Motion Analyses
- 3D-110 Ground Motion Analyses
- 3D-111 Ground Motion Analyses
- 3D-112 Ground Motion Analyses
- 3D-113 Ground Motion Analyses
- 3D-114 Ground Motion Analyses
- 3D-115 Ground Motion Analyses
- 3D-116 Ground Motion Analyses
- 3D-117 Ground Motion Analyses
- 3D-118 Ground Motion Analyses
- 3D-119 Ground Motion Analyses

- 3D-120 Ground Motion Analyses
- 3D-121 Ground Motion Analyses
- 3D-122 Ground Motion Analyses
- 3D-123 Ground Motion Analyses
- 3D-124 Ground Motion Analyses
- 3D-125 Ground Motion Analyses
- 3D-126 Ground Motion Analyses
- 3D-127 Ground Motion Analyses
- 3D-128 Containment Building el 169 ft Basemat
- 3D-129 Containment Building el 169 ft Basemat
- 3D-130 Containment Building el 169 ft Basemat
- 3D-131 Containment Building el 195 ft Internal Str.
- 3D-132 Containment Building el 195 ft Internal Str.
- 3D-133 Containment Building el 195 ft Internal Str.
- 3D-134 Containment Building el 220 ft Operating Lvl.
- 3D-135 Containment Building el 220 ft Operating Lvl.
- 3D-136 Containment Building el 220 ft Operating Lvl.
- 3D-137 Containment Building el 323 ft Polar Crane
- 3D-138 Containment Building el 323 ft Polar Crane
- 3D-139 Containment Building el 323 ft Polar Crane
- 3D-140 Containment Building el 180 ft Basemat
- 3D-141 Containment Building el 180 ft Basemat
- 3D-142 Containment Building el 180 ft Basemat
- 3D-143 Control Building el 220 ft

- 3D-144 Control Building el 220 ft
- 3D-145 Control Building el 220 ft
- 3D-146 Control Building el 260 ft
- 3D-147 Control Building el 260 ft
- 3D-148 Control Building el 260 ft
- 3E-1 Model and Coordinate System
- 3E-2 Polar Coordinate System
- 3F-1 Nodal Boundary for Auxiliary Building MSIV/MFIV Rooms; Nodal Boundary for Control Building - MSIV/MFIV Rooms (Inside and Outside Restraint Walls)
- 3F-2 Nodal Diagram for MSIV/MFIV GOTHIC Analysis
- 3F-3 Time-History Plot for Maximum Compartment Composite Temperature for MSIV Area Break Node
- 4.2-1 Fuel Assembly Cross-Section, 17 x 17
- 4.2-2 Fuel Assembly Outline, 17 x 17
- 4.2-3 Fuel Rod Schematic (LOPAR and VANTAGE 5)
- 4.2-4 Plan View (LOPAR and VANTAGE 5)
- 4.2-5 Top Grid to Nozzle Attachment (LOPAR and VANTAGE 5)
- 4.2-6 Grid to Thimble Attachment (LOPAR and VANTAGE 5)
- 4.2-7 Guide Thimble to Bottom Nozzle Joint (LOPAR and VANTAGE 5)
- 4.2-8 RCC and Drive Rod Assembly with Interfacing Components
- 4.2-9 Full-Length RCCA Outline
- 4.2-10 Absorber Rod
- 4.2-11 Wet Annular Burnable Absorber (WABA) Assembly
- 4.2-12 BA Rod Cross Section (Wet Annular Burnable Absorber)
- 4.2-13 Primary Source Assembly
- 4.2-14 Secondary Source Assembly

4.2-15	Thimble Plug Assembly
4.2-16	Stainless Steel Rod Insert Assembly
4.3-1	Fuel Loading Arrangement
4.3-2	Production and Consumption of Higher Isotopes
4.3-3	Boron Concentration Versus First Cycle Burnup With and Without BA Rods
4.3-4	Rod Arrangement Within an Assembly
4.3-5	Loading Pattern
4.3-6	Normalized Power Density Distribution Near Beginning of Life, Unrodded Core, Hot Full Power, No Xenon
4.3-7	Normalized Power Density Distribution Near Beginning of Life, Unrodded Core, Hot Full Power, Equilibrium Xenon
4.3-8	Normalized Power Density Distribution Near Beginning of Life, Group D at Rod Insertion Limits, Hot Full Power, Equilibrium Xenon
4.3-9	Normalized Power Density Distribution Near Middle of Life, Unrodded Core, Hot Full Power, Equilibrium Xenon
4.3-10	Normalized Power Density Distribution Near End of Life, Unrodded Core, Hot Full Power, Equilibrium Xenon
4.3-11	Normalized Power Density Distribution Near End of Life, Group D at Rod Insertion Limits, Hot Full Power, Equilibrium Xenon
4.3-12	Rodwise Power Distribution in a Typical Assembly (G-10 Near Beginning of Life, Hot Full Power, Equilibrium Xenon, Unrodded Core
4.3-13	Rodwise Power Distribution in a Typical Assembly (G-10) Near End of Life, Hot Full Power, Equilibrium Xenon, Unrodded Core
4.3-14	Typical Axial Power Shapes Occurring at Beginning of Life
4.3-15	Typical Axial Power Shapes Occurring at Middle of Life
4.3-16	Typical Axial Power Shapes Occurring at End of Life
4.3-17	Deleted
4.3-18	Flow Chart for Determining Spike Model

- 4.3-19 Predicted Power Spike Due to Single Nonflattened Gap in the Adjacent Fuel
- 4.3-20 Power Spike Factor as a Function of Axial Position
- 4.3-21 Maximum FQ = Power Versus Axial Height During Normal Operation
- 4.3-22 Peak Linear Power During Control Rod Malfunction Overpower Transients
- 4.3-23 Peak Linear Power During Boration Dilution Transients
- 4.3-24 Comparison Between Calculated and Measured Relative Fuel Assembly Power Distribution
- 4.3-25 Comparison of Typical Calculated and Measured Axial Shapes
- 4.3-26 Measured Values of FQ for Full-Power Rod Configurations
- 4.3-27 Typical Doppler Temperature Coefficient at Beginning of Life and End of Life
- 4.3-28 Typical Doppler-Only Power Coefficient at Beginning of Life
- 4.3-29 Typical Doppler-Only Power Defect at Beginning of Life and End of Life
- 4.3-30 Typical Moderator Temperature Coefficient at Beginning of Life Unrodded
- 4.3-31 Typical Moderator Temperature Coefficient at End of Life
- 4.3-32 Typical Moderator Temperature Coefficient as a Function of Boron Concentration at Beginning of Life Unrodded
- 4.3-33 Typical Hot Full-Power Temperature Coefficient vs Cycle Burnup
- 4.3-34 Typical Total Power Coefficient at Beginning of Life and End of Life
- 4.3-35 Typical total Power Defect at Beginning of Life and End of Life
- 4.3-36 Rod Cluster Control Assembly Pattern
- 4.3-37 Typical Accidental Simultaneous Withdrawal of Two Control Banks at End of Life, Hot Zero Power, Bank D and B Moving in the Same Plane
- 4.3-38 Design Trip Curve
- 4.3-39 Typical Normalized Rod Worth Versus Percent Insertion, All Rods Out But One
- 4.3-40 Axial Offset Versus Time, PWR Core with a 12-ft Height and 121 Assemblies
- 4.3-41 X-Y Xenon Test Thermocouple Response Quadrant Tilt Difference Versus Time

- 4.3-42 Calculated and Measured Doppler Defect and Coefficients at Beginning of Life, 2-Loop Plant, 121 Assemblies, 12-ft Core
- 4.3-43 Comparison of Calculated and Measured Boron Concentration, 2-Loop Plant, 121 Assemblies, 12-ft Core
- 4.3-44 Comparison of Calculated and Measured CB, 2-Loop Plant, 121 Assemblies, 12ft Core
- 4.3-45 Comparison of Calculated and Measured CB, 3-Loop Plant, 157 Assemblies, 12ft Core
- 4.3-46 Burnup Credit Requirement for All Cell Storage
- 4.3-47 Burnup Credit Requirement for 3-Out-of-4 Checkerboard Storage
- 4.3-48 Minimum Burnup Requirements
- 4.3-49 Burnup Credit Requirement for 3x3 Checkerboard Storage
- 4.3-50 IFBA Requirement for Center Assembly in 3x3 Checkerboard
- 4.3-51 Minimum IFBA Requirement for the Center Assembly 3x3 Checkerboard Storage
- 4.3-52 Checkerboard Storage Configurations
- 4.3-53 Interface Requirements
- 4.4-1 Measured vs. Predicted Critical Heat Flux WRB-1 Correlation
- 4.4-2 TDC Versus Reynolds Number for 26-in. Grid Spacing
- 4.4-3 Normalized Radial Flow and Enthalpy Distribution at 4-ft Elevation
- 4.4-4 Normalized Radial Flow and Enthalpy Distribution at 8-ft Elevation
- 4.4-5 Normalized Radial Flow and Enthalpy Distribution at 12-ft Elevation, Core Exit
- 4.4-6 Thermal Conductivity of UO<sub>2</sub> (Data Corrected to 95% Theoretical Density)
- 4.4-7 Deleted
- 4.4-8 RCS Temperature-Percent Power Map
- 4.4-9 100% Power Shapes Evaluated at Conditions Representative of Loss of Flow; All Shapes Evaluated with  $\mathbf{F}_{\Delta H}^{N} = 1.55$
- 4.4-10 Distribution of Incore Instrumentation

- 4.4-11 Loose Parts Monitoring System Block Diagram
- 4A-1 Core Exit Thermocouple Monitoring System (Layout Sketch)
- 4A-2 Core Subcooling Margin System
- 4A-3 Reactor Vessel Level Instrumentation Fluid Layout Drawing
- 4A-4 Reactor Vessel Level Instrumentation System
- 4A-5 Vogtle PSMS Electrical Interconnection
- 5.1.2-1 RCS Process Flow Diagram
- 5.2.5-1 Primary Coolant Leak Detection Response Time
- 5.3.3-1 Reactor Vessel
- 5.4.1-1 Reactor Coolant Pump
- 5.4.1-2 Reactor Coolant Pump Estimated Performance Characteristic
- 5.4.2-1 Model F Steam Generator
- 5.4.2-2 Quatrefoil Tube Support Plates
- 5.4.7-1 RHRS Process Flow Diagram
- 5.4.7-2 RHRS Pump Performance Curve
- 5.4.7-3 RHRS Cooldown Curves
- 5.4.10-1 Pressurizer
- 5.4.11-1 Pressurizer Relief Tank
- 5.4.11-2 Pressurizer Safety and Relief Valve Piping and Support Arrangement
- 5.4.13-1 Pressurizer Safety Valves Relief Rate
- 5.4.14-1 Reactor Vessel Supports
- 5.4.14-2 Steam Generator Supports
- 5.4.14-3 Reactor Coolant Pumps Supports
- 5.4.14-4 Pressurizer Supports

- 6.1.2-1 Containment Temperature Profile for Post-LOCA Coating Test
- 6.2.1-1 Containment Pressure Transient LOCA Double-Ended Pump Suction -Minimum Safety Injection
- 6.2.1-2 Containment Pressure Transient LOCA Double-Ended Pump Suction -Maximum Safety Injection
- 6.2.1-3 Containment Pressure Transient LOCA Double-Ended Hot Leg Guillotine
- 6.2.1-4 Containment Temperature Transient LOCA Double-Ended Pump Suction -Minimum Safety Injection
- 6.2.1-5 Containment Temperature Transient LOCA Double-Ended Pump Suction -Maximum Safety Injection
- 6.2.1-6 Containment Temperature Transient LOCA Double-Ended Hot Leg Guillotine
- 6.2.1-7 Containment Fan Cooler Performance LOCA Double-Ended Pump Suction -Maximum Safety Injection
- 6.2.1-8 Deleted
- 6.2.1-9 Deleted
- 6.2.1-10 Deleted
- 6.2.1-11 Deleted
- 6.2.1-12 Deleted
- 6.2.1-13 Deleted
- 6.2.1-14 Deleted
- 6.2.1-15 Reactor Cavity Pressure Response Nodes E1 Through E65
- 6.2.1-16 Reactor Cavity Model
- 6.2.1-16A Steam Generator Compartment Model (Pipe Break Above el 220 ft)
- 6.2.1-17 Steam Generator Compartment Model (Pipe Break Below el 220 ft)
- 6.2.1-17A Steam Generator Compartments Differential Pressure Response Across Doghouse Wall (144.5 in.<sup>2</sup> Main Feedwater Line Break)
- 6.2.1-18 Steam Generator Compartment Pressure Response Nodes E1 Through E74 (236-in.<sup>2</sup> Break Area)

### LIST OF FIGURES (CONTINUED)

6.2.1-19 Steam Generator Inlet Nozzle - Nodes E1 Through E74 (306-in.<sup>2</sup> Break Area) 6.2.1-20 Reactor Coolant Pump Inlet Nozzle - Nodes E1 Through E74 (336-in.<sup>2</sup> Break Area) 6.2.1-21 Steam Generator Outlet Nozzle - Nodes E1 Through E74 (436-in.<sup>2</sup> Break Area) 6.2.1-22 Steam Generator Inlet Elbow - Nodes E1 Through E74 (763-in.<sup>2</sup> Break Area) 6.2.1-23 Loop Closure Weld - Nodes E1 Through E74 (336-in.<sup>2</sup> Break Area) 6.2.1-24 Pressurizer Model 6.2.1-25 Pressurizer Compartment Pressure Response - Nodes 1 Through 28 (308-in.<sup>2</sup> Break Area) 6.2.1-25A Pressurizer Compartment Pressure Response - Nodes 1 through 28 Spray Line Break at Top of Pressurizer 6.2.1-26 Containment Pressure Transient MSLB - 0.22 ft<sup>2</sup> Double-Ended Rupture Without Entrainment - 30% Power Case 11 6.2.1-27 Containment Temperature Transient MSLB - 0.22 ft<sup>2</sup> Double-Ended Rupture Without Entrainment - 30% Power Case 11 6.2.1-28 Containment Pressure Transient MSLB - 0.86 ft<sup>2</sup> Split Rupture - 102% Power Case 13 6.2.1-29 Containment Temperature Transient MSLB - 0.86 ft<sup>2</sup> Split Rupture - 102% Power Case 13 6.2.1-30 Containment Pressure DECLG (CD = 0.6, Low  $T_{AVG}$ , Min SI Cosine) 6.2.1-31 Containment Temperature DECLG (CD = 0.6, Low  $T_{AVG}$ , Min SI Cosine) 6.2.1-32 Containment Wall Condensation Heat Transfer Coefficient DECLG (CD = 0.6, Low T<sub>AVG</sub>, Min SI Cosine) 6.2.4-1 Valve Arrangement 6.2.5-1 Electric Hydrogen Recombiner (Typical) 6.2.5-2 Electric Hydrogen Recombiner System Schematic 6.2.5-3 Deleted 6.2.5-4 Deleted 6.2.5-5 Containment Hydrogen Concentration

625-6	Deleted
0.2.0-0	Deleteu

- 6.2.5-7 Hydrogen Accumulation from All Sources
- 6.3.2-1 Emergency Core Cooling System Process Flow Diagram
- 6.3.2-2 Performance Curves Residual Heat Removal Pumps
- 6.3.2-3 Performance Curves Centrifugal Charging Pumps
- 6.3.2-4 Performance Curves Safety Injection Pumps
- 6.3.2-5 RWST Sizing
- 6.5.2-1 Spatial Drop-Size Distribution
- 6.5.2-2 Spray Envelope Reduction Factor
- 6.5.2-3 Capacity Curve Spraco 1713A Nozzle
- 7.1.1-1 Protection System Block Diagram
- 7.2.1-1 Setpoint Reduction Function for Overpower and Overtemperature  $\Delta T$  Trips
- 7.2.1-2 Reactor Trip/Engineered Safety Features Actuation Mechanical Linkage
- 7.3.1-1 Typical Engineered Safety Features Test Circuits
- 7.3.1-2 Engineered Safeguards Test Cabinet (Index, Notes, and Legend)
- 7.6.2-1 Logic Diagram for the RHRS Isolation Valves
- 7.6.4-1 Functional Block Diagram of Accumulator Isolation Valve
- 7.6.5-1 Safety Injection System Recirculation Sump and RHR Suction Isolation Valves
- 7.7.1-1 Simplified Block Diagram of Reactor Control System
- 7.7.1-2 Control Bank Rod Insertion Monitor
- 7.7.1-3 Rod Deviation Comparator
- 7.7.1-4 Block Diagram of Pressurizer Pressure Control System
- 7.7.1-5 Block Diagram of Pressurizer Level Control System
- 7.7.1-6 Block Diagram of Steam Generator Water Level Control System Unit 2

7.7.1-7	Block Diagram of Main Feedwater Pump Speed Control System Unit 2
7.7.1-8	Block Diagram of Steam Dump Control System
7.7.1-9	Basic Flux-Mapping System
7.7.1-10	Actuation Logic System Architecture
7.7.2-1	Simplified Block Diagram Rod Control System
7.7.2-2	Control Bank D Partial Simplified Schematic Diagram Power Cabinets 1BD and 2BD
8.3.1-1	Penetration Overcurrent Protection Coordination Curves
9.1.1-1	New Fuel Rack
9.1.2-1	Spent Fuel Pool Storage Rack Array, Side View (Unit 1)
9.1.2-1	Spent Fuel Pool Storage Rack Array, Isometric View (Unit 2)
9.1.2-2	Spent Fuel Pool Storage Cell Nominal Dimensions (Unit 1)
9.1.2-3	Spent Fuel Pool Storage Layout (Unit 1)
9.1.2-4	Spent Fuel Pool Rack's, Schematic Model for Dynrack (Unit 1)
9.1.2-4	Spent Fuel Pool Rack's, Two-Dimensional View of the Spring-Mass Simulation (Unit 2)
9.1.2-4	Spent Fuel Pool Rack's, Impact Spring Arrangement at Node 1 (Unit 2)
9.1.2-4	Spent Fuel Pool Rack's, Schematic Model for Dynrack (Unit 2)
9.1.2-5	Spent Fuel Pool Storage Layout (Unit 2)
9.1.4-1	Refueling Machine
9.1.4-2	Fuel Handling Machine
9.1.4-3	New Fuel Elevator
9.1.4-4	Fuel Transfer System
9.1.4-5	Fuel Transfer System

- 9.1.4-6 Spent Fuel Handling Tool
- 9.1.4-7 New Fuel Handling Tool
- 9.1.4-8 Reactor Internals Lifting Device
- 9.1.4-9 Deleted
- 9.1.5-1 Arrangement Drawing Spent Fuel Cask Bridge Crane
- 9.1.5-2 Arrangement Drawing Containment Building Polar Crane
- 9.1.5-3 Maximum Allowable Lift Heights and Loads
- 9.1.5-4 Circuit Arrangement for Local and Remote Display of Detection and Suppression Systems
- 10.2.2-1 Turbine Overspeed Protection Schematic
- 10.2.2-2 Protective Systems Signal Flow Diagram
- 10.4.8-1 Steam Generator Blowdown Processing System Process Flow Diagram
- 10A-1 AFWS
- 10A-2 Unit 1 Auxiliary Feedwater/Steam Generators Intake
- 10A-3 AFWS Simplified Piping Layout
- 10A-4 AFWS Reliability Evaluation Methodology Flow Chart
- 10A-5 Unit 1 AFWS Block Diagram
- 10A-6 AFWS Expanded Block Diagram
- 10A-7 Unit 1 AFWS Fault Tree Model
- 11.2.2-1 LWPS Process Flow Diagram
- 11.2.3-1 Process Flow Model (for CVCS Letdown & Equipment Drains; for Clean Waste; for Dirty Waste & Laundry; for Radwaste Processing Facility; and for Steam Generator Blowdown)
- 11.3.2-1 GWPS Fission Gas Accumulation Based On Continuous Core Operation of 2 Units at 3565 MWt with 1% Fuel Defects
- 11.3.2-2 Expected GWPS Fission Gas Accumulation Based On Table and Full-Power Operation of 2 Units at 3565 MWt

- 11.3.2-3 Process Diagram During Operation With GDT Pressure < 20 psig
- 11.3.2-4 Process Diagram During Operation With GDT Pressure Between 20 and 100 psig
- 11.3.2-5 Waste Gas Compressor Package
- 11.3.2-6 Catalytic Hydrogen Recombiner Package
- 11.5.2-1 Radiation Monitoring System Block Diagram
- 11.5.2-2a Safety Subsystem Block Diagram
- 11.5.2-2b Safety Subsystem Block Diagram
- 11.5.2-3 PERMS Communication Console
- 11.5.2-4a Three-Channel Airborne Monitoring Assembly
- 11.5.2-4b Three-Channel Airborne Monitoring Assembly
- 11.5.2-5 Monitoring Locations for Liquid Release Pathways
- 11.5.2-6 PERMS Normal Gaseous Effluent Release Pathway
- 12.2.1-1 Azimuthal Distribution of Neutron Flux Incident on the Primary Shield at the Reactor Core Midplane
- 12.2.1-2 Azimuthal Distribution of Gamma Ray Energy Flux Incident on the Primary Shield at the Reactor Core Midplane
- 12.2.1-3 Azimuthal Distribution of Gamma Ray Dose Rate Incident on the Primary Shield at the Reactor Core Midplane
- 13.1.1-1 Nuclear Operations Organization VEGP Units 1 and 2 (Deleted)
- 13.1.1-2 SNC Technical Support Organization VEGP Units 1 and 2 (Deleted)
- 13.1.2-1 Nuclear Operations Organization VEGP Units 1 and 2 (Deleted)
- 13.5.1-1 Limited Access Area
- 14.2.7-1 Initial Test Program Milestone Schedule
- 15.0.1-1 Excessive Heat Removal Due to Feedwater Systems Malfunction
- 15.0.1-2 Excessive Load Increase

- 15.0.1-3 Depressurization of Main Steam System
- 15.0.1-4 Loss of External Load
- 15.0.1-5 Loss of Offsite Power to Station Auxiliaries (Blackout)
- 15.0.1-6 Loss of Normal Feedwater
- 15.0.1-7 Major Rupture of a Main Feedwater Line
- 15.0.1-8 Loss of Forced Reactor Coolant Flow
- 15.0.1-9 Uncontrolled Rod Cluster Control Assembly Bank Withdrawal
- 15.0.1-10 Dropped Rod Cluster Control Assembly
- 15.0.1-11 Single Rod Cluster Control Assembly Withdrawal at Full Power
- 15.0.1-12 Startup of an Inactive Reactor Coolant Loop
- 15.0.1-13 Boron Dilution
- 15.0.1-14 Rupture on Control Rod Drive Mechanism Housing
- 15.0.1-15 Inadvertent ECCS Operation at Power
- 15.0.1-16 Accidental Depressurization of Reactor Coolant System
- 15.0.1-17 Steam Generator Tube Rupture
- 15.0.1-18 Loss-of-Coolant Accident
- 15.0.1-19 CVCS Letdown Line Rupture
- 15.0.1-20 GWPS Gas Decay Tank Rupture
- 15.0.1-21 Recycle Holdup Tank Failure
- 15.0.1-22 Fuel Handling Accident in Fuel Building
- 15.0.1-23 Fuel Handling Accident Inside Containment
- 15.0.1-24 Spent Fuel Cask Drop Accident
- 15.0.1-25 Abbreviations and Symbols Used in Sequence Diagrams
- 15.0.4-1 Doppler Power Coefficient Used in Accident Analysis
- 15.0.4-2 Moderator Density Coefficient Used in Accident Analysis

- 15.0.5-1 Rod Position vs. Time on Reactor Trip
- 15.0.5-2 RCCA Reactivity Worth vs. Fraction Full Insertion
- 15.0.5-3 Normalized RCCA Bank Worth vs. Time After Trip
- 15.0.6-1 Overtemperature and Overpower  $\Delta T$  Protection
- 15.0.6-2  $F(\Delta I)$  Penalty
- 15.1.2-1 Transients for Feedwater Control Valve Malfunction
- 15.1.2-2 Transients for Feedwater Control Valve Malfunction
- 15.1.2-3 Transients for Feedwater Control Valve Malfunction
- 15.1.3-1 10 Percent Step Load Increase, Minimum Moderator Feedback, Manual Rod Control
- 15.1.3-2 10 Percent Step Load Increase, Minimum Moderator Feedback, Manual Rod Control
- 15.1.3-3 10 Percent Step Load Increase, Maximum Moderator Feedback, Manual Rod Control
- 15.1.3-4 10 Percent Step Load Increase, Maximum Moderator Feedback, Manual Rod Control
- 15.1.3-5 10 Percent Step Load Increase, Minimum Moderator Feedback, Automatic Rod Control
- 15.1.3-6 10 Percent Step Load Increase, Minimum Moderator Feedback, Automatic Rod Control
- 15.1.3-7 10 Percent Step Load Increase, Maximum Moderator Feedback, Automatic Rod Control
- 15.1.3-8 10 Percent Step Load Increase, Maximum Moderator Feedback, Automatic Rod Control
- 15.1.4-1 K<sub>eff</sub> vs. Coolant Average Temperature
- 15.1.4-2 Failure of a Steam Generator Safety or Dump Valve Heat Flux vs. Time, Steam Flow vs. Time, Average Temp. vs. Time, Reactivity vs. Time
- 15.1.4-3 Failure of a Steam Generator Safety or Dump Valve RCS Pressure vs. Time, Pressurizer Water Vol. vs. Time, Boron Concentration vs. Time

- 15.1.4-4 Failure of a Steam Generator Safety or Dump Valve (RCS Pressure vs. Time, Boron Concentration vs. Time)
- 15.1.4-5 Failure of a Steam Generator Safety or Dump Valve (Pressurizer Water Volume vs. Time)
- 15.1.5-0 1.4 ft<sup>2</sup> Steam Line Rupture at Hot Zero Power K<sub>eff</sub> vs. Coolant Average Temperature
- 15.1.5-1 1.4 ft<sup>2</sup> Steam Line Rupture at Hot Zero Power With Offsite Power Available. Nuclear Power, Core Heat Flux, and Pressurizer Pressure Transients
- 15.1.5-2 1.4 ft<sup>2</sup> Steam Line Rupture at Hot Zero Power With Offsite Power Available. Feed Flow, Core Flow, and Steam Flow Transients
- 15.1.5-3 1.4 ft<sup>2</sup> Steam Line Rupture at Hot Zero Power With Offsite Power Available. SG Pressure, Average Core Moderator Temperature, and Vessel Inlet Temperature Transients
- 15.1.5-4 1.4 ft<sup>2</sup> Steam Line Rupture at Hot Zero Power With Offsite Power Available. Pressurizer Water Volume, Reactivity, and Core Boron Transients
- 15.1.5-5 1.4 ft<sup>2</sup> Steam Line Rupture at Hot Zero Power Without Offsite Power Available. Nuclear Power, Core Heat Flux, and Pressurizer Pressure Transients
- 15.1.5-6 1.4 ft<sup>2</sup> Steam Line Rupture at Hot Zero Power Without Offsite Power Available. Feed Flow, Core Flow, and Steam Flow Transients
- 15.1.5-7 1.4 ft<sup>2</sup> Steam Line Rupture at Hot Zero Power Without Offsite Power Available. SG Pressure, Average Core Moderator Temperature, and Vessel Inlet Temperature Transients
- 15.1.5-8 1.4 ft<sup>2</sup> Steam Line Rupture at Hot Zero Power Without Offsite Power Available. Pressurizer Water Volume, Reactivity, and Core Boron Transients
- 15.1.5-9 Nuclear Power, Pressurizer Pressure and Pressurizer Water Volume Transients for 0.8 ft<sup>2</sup> Steam Line Break at Full Power
- 15.1.5-10 Core Average Temperature, Core Inlet Temperature, and Core Heat Flux Transients for 0.8 ft<sup>2</sup> Steam Line Break at Full Power
- 15.1.5-11 Total Steam Flow and Steam Generator Pressure Transients for 0.8 ft<sup>2</sup> Steam Line Break at Full Power
- 15.2.3-1 Turbine Trip Accident with Pressurizer Spray and Power-Operated Relief Valves, Minimum Moderator Feedback

- 15.2.3-2 Turbine Trip Accident with Pressurizer Spray and Power-Operated Relief Valves, Minimum Moderator Feedback
- 15.2.3-3 Turbine Trip Accident with Pressurizer Spray and Power-Operated Relief Valves, Max. Moderator Feedback
- 15.2.3-4 Turbine Trip Accident with Pressurizer Spray and Power-Operated Relief Valves, Maximum Moderator Feedback
- 15.2.3-5 Turbine Trip Accident Without Pressurizer Spray and Power-Operated Relief Valves, Minimum Moderator Feedback
- 15.2.3-6 Turbine Trip Accident Without Pressurizer Spray and Power-Operated Relief Valves, Minimum Moderator Feedback
- 15.2.3-7 Turbine Trip Accident Without Pressurizer Spray and Power-Operated Relief Valves, Maximum Moderator Feedback
- 15.2.3-8 Turbine Trip Accident Without Pressurizer Spray and Power-Operated Relief Valves, Maximum Moderator Feedback
- 15.2.6-1 Pressurizer Pressure and Water Volume Transients for Loss of Offsite Power
- 15.2.6-2 Loop Temperature and Steam Generator Pressure Transients for Loss of Offsite Power
- 15.2.7-1 Pressurizer Pressure and Water Volume Transients for Loss of Normal Feedwater
- 15.2.7-2 Loop Temperature and Steam Generator Pressure Transients for Loss of Normal Feedwater
- 15.2.8-1 Nuclear Power Transient and Core Heat Flux for Main Feedline Rupture with Offsite Power Available
- 15.2.8-2 Pressurizer Pressure and Water Volume Transients for Main Feedline Rupture with Offsite Power Available
- 15.2.8-3 Reactor Coolant Temperature Transients for the Faulted and Intact Loops for Main Feedline Rupture with Offsite Power Available
- 15.2.8-4 Steam Generator Pressure and Water Mass Transients for Main Feedline Rupture with Offsite Power Available
- 15.2.8-5 Nuclear Power and Core Heat Flux Transients for Main Feedline Rupture Without Offsite Power Available
- 15.2.8-6 Pressurizer Pressure and Water Volume Transients for Main Feedline Rupture Without Offsite Power Available

- 15.2.8-7 Reactor Coolant Temperature Transients for the Faulted and Intact Loops for Main Feedline Rupture Without Offsite Power Available
- 15.2.8-8 Steam Generator Pressure and Steam Generator Water Mass Transients for Main Feedline Rupture Without Offsite Power Available
- 15.3.1-1 Flow Transients for Four Loops in Operation, Two Pumps Coasting Down
- 15.3.1-2 Nuclear Power and Pressurizer Pressure Transients for Four Loops in Operation, Two Pumps Coasting Down
- 15.3.1-3 Average and Hot Channel Heat Flux Transients for Four Loops in Operation, Two Pumps Coasting Down
- 15.3.1-4 DNBR Versus Time for Four Loops in Operation, Two Pumps Coasting Down
- 15.3.2-1 Flow Transients for Four Loops in Operation, Four Pumps Coasting Down
- 15.3.2-2 Nuclear Power and Pressurizer Pressure Transients for Four Loops in Operation, Four Pumps Coasting Down
- 15.3.2-3 Average and Hot Channel Heat Flux Transients for Four Loops in Operation, Four Pumps Coasting Down
- 15.3.2-4 DNBR Versus Time for Four Loops in Operation, Four Pumps Coasting Down
- 15.3.3-1 Flow Transients for Four Loops in Operation
- 15.3.3-2 Peak Reactor Coolant Pressure for Four Loops in Operation
- 15.3.3-3 Average and Hot Channel Heat Flux Transients for Four Loops in Operation
- 15.3.3-4 Nuclear Power and Maximum Clad Temperature at Hot Spot Transients for Four Loops in Operation
- 15.4.1-1 Neutron Flux Transient for Uncontrolled Rod Withdrawal from a Subcritical Condition
- 15.4.1-2 Thermal Flux Transient for Uncontrolled Rod Withdrawal from a Subcritical Condition
- 15.4.1-3 Fuel and Clad Temperature for Uncontrolled Rod Withdrawal from a Subcritical Condition
- 15.4.2-1 Uncontrolled RCCA Bank Withdrawal from Full Power with Minimum Reactivity Feedback (80 pcm/s Withdrawal Rate)
- 15.4.2-2 Uncontrolled RCCA Bank Withdrawal from Full Power with Minimum Reactivity Feedback (80 pcm/s Withdrawal Rate)

- 15.4.2-3 Uncontrolled RCCA Bank Withdrawal from Full Power with Minimum Reactivity Feedback (80 pcm/s Withdrawal Rate)
- 15.4.2-4 Uncontrolled RCCA Bank Withdrawal from Full Power with Minimum Reactivity Feedback (2 pcm/s Withdrawal Rate)
- 15.4.2-5 Uncontrolled RCCA Bank Withdrawal from Full Power with Minimum Reactivity Feedback (2 pcm/s Withdrawal Rate)
- 15.4.2-6 Uncontrolled RCCA Bank Withdrawal from Full Power with Minimum Reactivity Feedback (2 pcm/s Withdrawal Rate)
- 15.4.2-7 Minimum DNBR vs. Reactivity Insertion Rate for Rod Withdrawal from 100 Percent Power
- 15.4.2-8 Minimum DNBR vs. Reactivity Insertion Rate for Rod Withdrawal from 60 Percent Power
- 15.4.2-9 Minimum DNBR vs. Reactivity Insertion Rate for Rod Withdrawal from 10 Percent Power
- 15.4.3-1 Nuclear Power Transient and Core Heat Flux Transient for Dropped RCCA
- 15.4.3-2 Pressurizer Pressure Transient and Core Average Temperature Transient for Dropped RCCA
- 15.4.4-1 Improper Startup of an Inactive Reactor Coolant Pump
- 15.4.4-2 Improper Startup of an Inactive Reactor Coolant Pump
- 15.4.4-3 Improper Startup of an Inactive Reactor Coolant Loop
- 15.4.4-4 Improper Startup of an Inactive Reactor Coolant Pump
- 15.4.4-5 Improper Startup of an Inactive Reactor Coolant Pump
- 15.4.7-1 Representative % Change in Local Assy. Avg. Power for Interchange Between Region 1 and Region 3 Assy.
- 15.4.7-2 Representative % Change in Local Assy. Avg. Power for Interchange Between Region 1 and Region 2 Assy. with BP Rods Retained by the Region 2 Assy.
- 15.4.7-3 Representative % Change in Local Assy. Avg. Power for Interchange Between Region 1 and Region 2 Assy. with the BP Rods Transferred to Region 1 Assy.
- 15.4.7-4 Representative % Change in Local Assy. Avg. Power for Enrichment Error (Region 2 Assy. Loaded into Core Central Position)

- 15.4.7-5 Representative % Change in Local Assy. Avg. Power for Loading Region 2 Assy. into Region 1 Position Near Core Periphery
- 15.4.8-1 Nuclear Power Transient BOL Full Power
- 15.4.8-2 Hot Spot Fuel and Clad Temperature vs. Time BOL Full Power
- 15.4.8-3 Nuclear Power Transient EOL Zero Power
- 15.4.8-4 Hot Spot Fuel and Clad Temperature vs. Time EOL Zero Power
- 15.4.9-1 Steam Line Break Coincident with Control Rod Withdrawal: Steam Flow and Steam Pressure
- 15.4.9-2 Steam Line Break Coincident with Control Rod Withdrawal: RCS Pressure and Core Average Temperature
- 15.4.9-3 Steam Line Break Coincident with Control Rod Withdrawal: Core Heat Flux and Reactivity
- 15.5.1-1 Inadvertent Operation of ECCS at Power
- 15.5.1-2 Inadvertent Operation of ECCS During Power Operation
- 15.5.1-3 Inadvertent Operation of ECCS During Power Operation
- 15.6.1-1 Nuclear Power and DNBR Transients for Inadvertent Opening of a Pressurizer Safety Valve
- 15.6.1-2 Pressurizer Pressure Transients and Core Average Temperature Transient for Inadvertent Opening of a Pressurizer Safety Valve
- 15.6.3-1 Pressurizer Level
- 15.6.3-2 RCS Pressure
- 15.6.3-3 Secondary Pressure
- 15.6.3-4 Intact Loop T<sub>hot</sub> and T<sub>cold</sub>
- 15.6.3-5 Ruptured Loop T<sub>hot</sub> and T<sub>cold</sub>
- 15.6.3-6 Differential Pressure
- 15.6.3-7 Primary to Secondary Break Flow
- 15.6.3-8 Ruptured SG Water

- 15.6.3-9 Ruptured SG Water Mass
- 15.6.3-10 Ruptured SG Atmospheric Mass Releases
- 15.6.3-11 Intact SG Atmospheric Mass Release
- 15.6.3-12 Iodine Transport Model
- 15.6.3-13 Break Flow Flashing Fraction
- 15.6.5-1 Sequence of Events for Large Break LOCA Analysis
- 15.6.5-2 Cladding Temperature at PCT and Burst Elevations
- 15.6.5-3 Core Pressure During Blowdown
- 15.6.5-4 Vessel Liquid Levels During Reflood
- 15.6.5-5 Core Inlet Flooding Rate During Reflood
- 15.6.5-6 Normalized Core Power During Blowdown
- 15.6.5-7 Containment Pressure
- 15.6.5-8 Core Inlet and Outlet Mass Flow Rate During Blowdown
- 15.6.5-4 Cladding Surface Heat Transfer Coefficient at PCT and Burst Elevations
- 15.6.5-5 Vapor Temperature at PCT and Burst Elevations
- 15.6.5-11 Break Mass Flow Rate During Blowdown
- 15.6.5-12 Break Energy Release Rate During Blowdown
- 15.6.5-13 Fluid Quality at PCT and Burst Elevations
- 15.6.5-14 Fluid Mass Velocity at PCT and Burst Elevations
- 15.6.5-15 Intact Loop Accumulator Mass Flow Rate During Blowdown
- 15.6.5-16 Intact Leg Accumulator and SI Mass Flow Rate During Reflood
- 15.6.5-17 Safety Injection Flow (Minimum Safety Injection)
- 15.6.5-18 Safety Injection Flow (Maximum Safety Injection)
- 15.6.5-19 Small Break LOCA Intact Loop Safety Injection Flow
- 15.6.5-20 Small Break LOCA Broken Loop Safety Injection Flow

- 15.6.5-21 Small Break LOCA Power Shape
- 15.6.5-22 Reactor Coolant System Pressurizer Pressure
- 15.6.5-23 Core Mixture Level 2 in., Low T<sub>AVG</sub>
- 15.6.5-24 Reactor Coolant System Pressurizer Pressure 3 in., Low TAVG
- 15.6.5-25 Core Mixture Level 3 in., Low T<sub>AVG</sub>
- 15.6.5-26 Core Steam Flow Rate 3 in., Low TAVG
- 15.6.5-27 Pumped Safety Injection 3 in., Low TAVG
- 15.6.5-28 Peak Clad Temperature at 11.25 ft 3 in., Low T<sub>AVG</sub>
- 15.6.5-29 Hot Spot Fluid Temperature at 11.25 ft 3 in., Low TAVG
- 15.6.5-30 Hot Rod Heat Transfer Coefficient at 11.25 ft 3 in., Low TAVG
- 15.6.5-31 Reactor Coolant System Pressurizer Pressure 4 in., Low TAVG
- 15.6.5-32 Core Mixture Level 4 in., Low T<sub>AVG</sub>
- 15.6.5-33 Peak Clad Temperature at 10.75 ft 4 in., Low T<sub>AVG</sub>
- 15A-1 Release Pathways
- 17.2.1-1 Deleted
- 18.1-1 Control Room Layout (Unit 1)
- 18.3-1 Human Engineering Discrepancy Evaluation Algorithm

Page/Section	Rev
Table of Contents	24
Section 1.1	16
Section 1.2	22
Section 1.3 t 1.3.1-1 (5 sheets) t 1.3.1-2 (5 sheets) t 1.3.2-1 (6 sheets) t 1.3.2-2 (3 sheets)	16 16 16 14 21
Section 1.4	14
Section 1.5	16
Section 1.6 t 1.6-1 (2 sheets) t 1.6-2 (15 sheets) t 1.6-3	14 14 14 14
Section 1.7 t 1.7.1-1 (43 sheets)	14 17
Section 1.8 t 1.8-1 (5 sheets)	14 14
Section 1.9 t 1.9-1 (8 sheets) t 1.9-2 (7 sheets) t 1.9-3 (4 sheets)	24 13 13 13
Section 2.1	15   15

# Page/Section

Page/Section
--------------

## Rev

t 2.2.3-14 t 2.2.3-15 t 2.2.3-16 deleted (Rev. 0) t 2.2.3-17 t 2.2.3-18 (7 sheets) t 2.2.3-19 t 2.2.3-20 (2 sheets) f 2.2.1-1 f 2.2.1-2. f 2.2.1-3 f 2.2.2-1 f 2.2.2-1 f 2.2.2-2	14 14 14 22 14 14 14 14 14
Section 2.3 t 2.3.1-1 t 2.3.1-2 t 2.3.1-3 (2 sheets) t 2.3.1-4 t 2.3.1-5 t 2.3.2-1 (6 sheets) t 2.3.2-2	20 14 14 14 14 14 22
t 2.3.2-3 (4 sheets) t 2.3.2-3 (4 sheets) t 2.3.2-4 (4 sheets) t 2.3.2-5 (4 sheets) t 2.3.2-6 (4 sheets) t 2.3.2-7 (4 sheets) t 2.3.2-8 (4 sheets)	22 14 14 14 14 14
t 2.3.2-9 (4 sneets) t 2.3.2-10 (4 sheets) t 2.3.2-11 (4 sheets) t 2.3.2-12 t 2.3.2-13 t 2.3.2-14 (48 sheets) t 2.3.2-15 (4 sheets)	14 14 14 14 14 14
t 2.3.2-16 (48 sheets) t 2.3.2-17 (4 sheets) t 2.3.2-18 (4 sheets) t 2.3.2-19 (4 sheets) t 2.3.2-20 (4 sheets) t 2.3.2-21 t 2.3.2-21 t 2.3.2-21	14 14 14 14 14 14
t 2.3.3-1 t 2.3.4-1 t 2.3.4-2 t 2.3.4-3 t 2.3.5-1 t 2.3.5-2 t 2.3.5-3	14 14 14 14 14 14
t 2.3.5-4 t 2.3.5-5 t 2.3.5-6 (2 sheets) t 2.3.5-7 (2 sheets) t 2.3.5-8 (2 sheets) t 2.3.5-8 (2 sheets) t 2.3.5-9 (2 sheets)	14 14 14 14 14 14

t 2.3.5-10 (2 sheets)
t 2 3 5 11 (2  shoots) 14
1 2.3.3-11 (2 3)
t 2.3.5-12 (2 sheets)
$f_{2,3,1_{-1}}$ 1/
† 2.3.1-2
f 2 3 1_3 1/
12.0.1-0
† 2.3.1-4
f 2 3 1 5 1/
12.5.1-5
† 2.3.1-6
f 2 3 1 7 1/
1 2.3. 1-7
f 2.3.1-8
f 2 2 1 0 1/
1 2.3. 1-9
f 2.3.1-10
f 0 0 1 11 14
12.3.1-11
f 2.3.2-1 (3 sheets) 14
f 2 2 2 2 1 (c c c c c c )
1 2.3.2-2
f 2.3.2-3 14
$f_{2} = 0.2 + 0.000 $ (2 about a) 14
1 Z.3.Z-4 (3 Sheets)
f 2.3.2-5 14
f0006 14
1 2.3.2-0
f 2.3.2-7 (3 sheets) 14
$f_{0,2,2,0}$
1 2.3.2-0
f 2.3.2-9 14
$f_{2} = 0.2 + 0.0 (2 \text{ shoots})$ 14
1 2.3.2-10 (3 Sheets)
f 2.3.2-11
f 2 2 2 1 2 1/
1 2.3.2-12
t 2.3.2-13 (3 sheets) 14
$f_{2} = 2 - 14$ 14
t 2.3.2-15
f 2 3 2-16 (3 sheets) 14
$f_{0,0,0,0,1,7}$
12.3.2-17
f 2.3.2-18
$f_{2,2,2,10}(2 \text{ shoots})$ 14
1 2.3.2-19 (3 Sheets)
f 2.3.2-20 14
f 2 3 2 21 1/
t 2.3.2-22 (3 sheets) 14
f 2 3 2-23 14
† 2.3.2-24 14
f 2 3 2-25 (3 sheets) 14
† 2.3.2-26 14
f 2 3 2_27 14
† 2.3.2-28 14
f 2 3 2_29 14
f 2.3.2-30 14
f 2 3 2-31 14
I Z.J.Z-JZ14
f 2.3.2-33 14
f0 0 0 0 4 4
I Z.J.Z-J414
f 2.3.2-35
f0 0 0 06 44
12.3.2-30
f 2.3.2-37
f 2 3 2_38 1/
1 2.0.2-00
1/ 1/

# Page/Section

## Rev

Page/Section

## Rev

f 2.3.2-401	4
f 2.3.2-411	4
f 2.3.2-421	4
f 2.3.2-431	4
f 2.3.2-441	4
f 2.3.2-451	4
f 2.3.2-461	4
f 2.3.2-471	4
f 2.3.2-481	4
f 2.3.2-491	4
f 2.3.2-501	4
f 2.3.2-511	4
f 2.3.2-521	4
f 2.3.2-531	4
f 2.3.2-541	4
f 2.3.2-551	4
f 2.3.2-561	4
f 2.3.2-57 (4 sheets)1	4
f 2.3.4-1	4
f 2.3.4-2 (2 sheets)1	4
f 2.3.4-3 (2 sheets)1	4
f 2.3.4-4 (2 sheets)1	4
f 2.3.4-5	4
f 2.3.4-61	4
f 2.3.4-71	4
f 2.3.4-81	4
f 2.3.5-11	4
f 2.3.5-21	4
f 2.3.5-31	4
f 2.3.5-41	4
f 2.3.5-51	4
f 2.3.5-61	4
Section 2.41	6
t 2.4.1-1	4
t 2.4.1-2 (5 sheets)1	4
t 2.4.1-3 (2 sheets)1	4
t 2.4.2-1 (4 sheets)1	4
t 2.4.2-2	4
t 2.4.2-3	4
t 2.4.3-11	4
t 2.4.3-21	4
t 2.4.3-31	4
t 2.4.3-41	4
t 2.4.3-51	4
t 2.4.3-61	4
t 2.4.7-11	4
t 2.4.12-11	4
t 2.4.12-21	4
t 2.4.12-3 (2 sheets)1	4
t 2.4.12-41	4
t 2.4.12-51	4
t 2.4.12-61	4

t 2.4.12-7A (2 sheets)	21
t 2.4.12-7B (2 sheets)	14
t 2.4.12-7C	14
t 2.4.12-7D	14
t 2.4.12-8	14
t 2 4 12-9	14
t 2.4 12.10 (4 sheets)	14
t 2 / 12-10 (+ 510013)	1/
t 2.4.12 - 11	1/
$t 2.4.12 - 12 (2 sheets) \dots$	14
1 2.4.12-13 (2 sneets)	14
t 2.4.12-14	14
t 2.4.12-15	14
t 2.4.13-1	14
f 2.4.1-1	14
f 2.4.1-2	14
f 2.4.3-1	14
f 2.4.3-2	14
f 2.4.3-3	14
f 2 4 3-4	14
f 2 4 4-1	14
f 2 / / 2	1/
f 2 4 4 2 (2  shoots)	14
f 2.4.4-3 (2.5)	14
1 2.4.4-4 (5 Sheets)	14
1 2.4.4-5	14
1 2.4.4-6	14
f 2.4.4-7	14
f 2.4.4-8 (2 sheets)	14
f 2.4.11-1	14
f 2.4.11-2	14
f 2.4.11-3	14
f 2.4.11-4	14
f 2 4 11-5	14
f 2 4 11-6	14
f 2 / 12_1	1/
1 2.4.12-1	14
Section 2.5	ດາ
	24 40
L 2.5. I - I	13
t 2.5.2-1 (9 sneets)	13
t 2.5.2-2 (2 sheets)	13
t 2.5.4-1	13
t 2.5.4-2	13
t 2.5.4-3	13
t 2.5.4-4	13
t 2.5.4-5	13
t 2.5.4-6	13
t 2 5 4-7	13
t 2 5 4-8	13
t 2 5 4-9	12
t 2 5 <i>1</i> _10	12
+ 2 5 <i>A</i> 44	10
1 2.3.4-11	13
ι ∠.ο.4-1∠ (∠ sneeιs)	13
t 2.5.4-13	13
t 2.5.4-14 deleted	13

# Page/Section

Page/Section
0

## Rev

t 2.5.4-15 deleted
Section 2A15
Section 2B
Section 3.119
Section 3.2 1   (t 3.2.2-1 contents) 13   t 3.2.2-1 (98 sheets) 24   t 3.2.2-2 (2 sheets) 14   t 3.2.2-3 14
Section 3.314
Section 3.414
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Section 3.6 17   t 3.6.1-1 (2 sheets) 14   t 3.6.1-2 (2 sheets) 14   t 3.6.1-3 (3 sheets) 14   t 3.6.2-1 14   t 3.6.2-2 deleted (Rev. 0) 14   t 3.6.2-3 14   f 3.6.2-1 14   f 3.6.2-3 14   f 3.6.2-3 14   f 3.6.2-3 14
Section 3.7 24   t 3.7.B.1-1 14   t 3.7.B.1-2 14   t 3.7.N.1-1 14   t 3.7.B.2-1 14

t 3.7.B.2-2	4 4 4 4 4 4 4
f 3.7.B.1-5	4 4 4 4 4
f 3.7.B.1-11	4 4 4 4 4
f 3.7.B.2-5	4 4 4 4 4
f 3.7.B.2-11	44444
f 3.7.B.2-17	444444
f 3.7.B.2-23 14 f 3.7.B.2-24 14 f 3.7.B.2-25 14 f 3.7.B.2-26 14 f 3.7.B.2-27 14 f 3.7.B.2-28 14 f 3.7.N.2-1 14 f 3.7.N.2-1 14	444444444
1 3.7.4-1 14   Section 3.8 19   t 3.8.1-1 (2 sheets) 14   t 3.8.4-1 14   t 3.8.4-2 14   t 3.8.4-3 14   t 3.8.5-1 14   f 3.8.1-1 (2 sheets) 14	4 9444444
# Page/Section

f 3.8.1-2 (Deleted)	15
f 3.8.1-2 (Deleted) Section 3.9 t 3.9.N.1-1 (3 sheets) t 3.9.N.1-2 (3 sheets) t 3.9.B.3-1 (5 sheets) t 3.9.B.3-2 t 3.9.B.3-3 t 3.9.B.3-4 t 3.9.B.3-5 t 3.9.B.3-6 t 3.9.B.3-7 (3 sheets) t 3.9.B.3-8 (2 sheets) t 3.9.B.3-9 (7 sheets) t 3.9.B.3-9 (7 sheets) t 3.9.B.3-10 (4 sheets) t 3.9.N.3-1 t 3.9.N.3-2 (8 sheets) t 3.9.N.3-2 (8 sheets) t 3.9.N.3-5 t 3.9.N.3-5 t 3.9.N.3-5 t 3.9.N.1-1 f 3.9.N.1-2 f 3.9.N.1-2 f 3.9.N.1-2 f 3.9.N.1-3 f 3.9.N.3-1 f 3.9.N.3-2 (3 sheets) t 3.9.N.3-5 t 3.9.N.3-5 t 3.9.N.1-1 f 3.9.N.1-2 f 3.9.N.1-2 f 3.9.N.1-3 f 3.9.N.3-1 f 3.9.N.3-2 f 3.9.N.3-1 f 3.9.N.3-2 f 3.9.N.3-2 f 3.9.N.3-3 f 3.9.N.3-1 f 3.9.N.3-2 f 3.9.N.3-2 f 3.9.N.3-3 f 3.9.N.3-1 f 3.9.N.3-2 f 3.9.N.3-3 f 3.9.N.3-2 f 3.9.N.3-3 f 3.9.N.3-1 f 3.9.N.3-2 f 3.9.N.3-3 f 3.9.N.3-1 f 3.9.N.3-2 f 3.9.N.3-2 f 3.9.N.3-3 f 3.9.N.3-1 f 3.9.N.3-2 f 3.9.N.3-2 f 3.9.N.3-3 f 3.9.N.3-3 f 3.9.N.3-3 f 3.9.N.3-1 f 3.9.N.3-2 f 3.9.N.3-3 f 3.9.N.3-1 f 3.9.A.4 f 3.9.A.4 f 3.9.5.1 f 3.9.5 f 3.9.5 f 3.9.5 f 3.9.	15 20 13 14 15 13
f 3.9.5-3	13 13
Section 3.10	14
Section 3.11 t 3.11.B.1-1 (92 sheets) t 3.11.B.3-1 (14 sheets) t 3.11.N.3-1 (16 sheets) f 3.11.B.1-1 (18 sheets)	16 21 14 14 14
Section 3A	14
Section 3B	14
Section 3C	22

-
t 3C-1 (2 sheets)14
f 3C-1
Section 3D 14
t 3D-1 14
t 3D-2 14
t 3D_3 14
t 3D-4 14
f 3D_1 1/
f 2D 2 14
f 2D 2 14
1 3D-3
1 3D-4
1 3D-5
1 3D-0
f 3D-714
f 3D-814
f 3D-914
f 3D-1014
f 3D-1114
f 3D-1214
f 3D-1314
f 3D-1414
f 3D-1514
f 3D-1614
f 3D-1714
f 3D-18
f 3D-19 14
f 3D-20 14
f 3D-21 14
f 3D-22
f 3D_23 1/
f 3D 24 14
f 2D 25
1 3D-23
1 3D-20
1 3D-27
1 3D-28
f 3D-29
f 3D-3014
f 3D-3114
f 3D-3214
f 3D-3314
f 3D-34 14
f 3D-3514
f 3D-3614
f 3D-3714
f 3D-3814
f 3D-3914
f 3D-40
f 3D-41
f 3D-42 14
f 3D-43 14
f 3D-44 14
f 3D-45

# Page/Section

## Rev

Page/Section

f 3D-46		14
f 2D 47		1 T 1 A
1 JD-47		14
t 3D-48		14
f 3D-49	······	14
f 3D-50	· · · · · · · · · · · · · · · · · · ·	14
f 3D-51		14
f 3D-52		14
f 3D-53		14
f 3D-54		14
f 3D_55		1 <i>1</i>
f 3D 56		17 17
f 2D 57		14
f 3D-57		14 4 4
1 3D-58		14
f 3D-59	······	14
f 3D-60	•••••••••••••••••••••••••••••••••••••••	14
f 3D-61	······	14
f 3D-62	· · · · · · · · · · · · · · · · · · ·	14
f 3D-63		14
f 3D-64		14
f 3D-65		14
f 3D_66		 14
f 3D_67		17 17
f 2D 60		14
f 3D-00		14 11
1 3D-09		14
t 3D-70	······	14
f 3D-71	······	14
f 3D-72	······	14
f 3D-73	· · · · · · · · · · · · · · · · · · ·	14
f 3D-74		14
f 3D-75		14
f 3D-76		14
f 3D-77		14
f 3D_78		 14
f 3D-70		17 17
f 2D 90		14
f 2D 04		14 11
1 3D-81		14
f 3D-82	······	14
f 3D-83	•••••••••••••••••••••••••••••••••••••••	14
f 3D-84	······	14
f 3D-85	· · · · · · · · · · · · · · · · · · ·	14
f 3D-86		14
f 3D-87		14
f 3D-88		14
f 3D-89		14
f 3D-90		 14
f 3D_01		1/
f 2D 02		14
f 3D-92		14 11
1 3D-93		14
1 3D-94	······································	14
t 3D-95	······	14
t 3D-96	······································	14
f 3D-97	······································	14
f 3D-98	······································	14

f 3D-9914
f 3D-100 14
f 3D-101 14
f 3D-102 14
f 3D-103 14
f 3D-104 14
f 3D-105
f 3D-106
f 3D 107 14
f 2D 109 14
f 2D 100
f 2D 110
1 3D-1 10
1 3D-111
1 3D-112
f 3D-113
f 3D-114
f 3D-11514
f 3D-11614
f 3D-11714
f 3D-11814
f 3D-11914
f 3D-12014
f 3D-12114
f 3D-12214
f 3D-12314
f 3D-12414
f 3D-12514
f 3D-12614
f 3D-12714
f 3D-128
f 3D-129 14
f 3D-130
f 3D-131 14
f 3D-132 14
f 3D-133 14
f 3D-134 14
f 3D-135 14
f 3D-136
f 3D-137 1/
f 3D 138 1/
f 2D 120
f 2D 140
f 2D 1/1 1/
1 3D-141
1 3D-142
1 3D-143
1 JU-144
1 3D-145
1 3D-146
13D-147
T 3D-148
Section 3E
t 3E-1

# Page/Section

## Rev

Page/Section
--------------

f 3E-21	4
Appendix 3F       2         t 3F-1 (84 sheets)       1         t 3F-2       1         t 3F-3A (4 sheets)       1         t 3F-3B (2 sheets)       1         t 3F-4 (3 sheets)       2         t 3F-5       1         f 3F-1 (4 sheets)       1         f 3F-3       1	23  5  4  5 22  4  4  4
Section 4.1	24 23 19 14
Section 4.2.       2         t 4.2-1       1         f 4.2-1 (2 sheets)       1         f 4.2-2 (7 sheets)       2         f 4.2-3 (4 sheets)       1         f 4.2-4 (2 sheets)       1         f 4.2-5 (4 sheets)       1         f 4.2-6 (2 sheets)       1         f 4.2-7 (2 sheets)       1         f 4.2-8       1         f 4.2-9       2         f 4.2-10       2         f 4.2-12 (2 sheets)       1         f 4.2-13       1         f 4.2-14 (2 sheets)       1         f 4.2-15 (2 sheets)       1	244444444444444444444444444444444444444
Section 4.3	21 24 13 13 13 13 13 13 13 13 13 13 13 13 13

$f \downarrow 3 \downarrow (2 \text{ shoots})$ 1	2
14.3-4 (2 sileets)	5
1 4.3-5 (2 sneets)1	3
f 4.3-6 1	3
f 4.3-71	3
f 4.3-81	3
f 4 3-9 1	ŝ
f A = 2 + 10	2
14.0-10 I	5
14.3-11	3
t 4.3-12 1	3
f 4.3-13 1	3
f 4.3-141	3
f 4 3-15 1	3
f A = 2 + 16	2
f 4 0 47 delete d (Dev. 0)	5
14.3-17 deleted (Rev. 3)	3
1 4.3-18 deleted (Rev. 6) 1	3
f 4.3-19 deleted (Rev. 6)1	3
f 4.3-20 deleted (Rev. 6)1	3
f 4 3-21	3
f A = 2 + 2 + 2 + 2 + 2 + 2 + 2 + 2 + 2 + 2	2
14.0-22	5
14.3-231	3
t 4.3-24 1	3
f 4.3-25 1	3
f 4.3-261	3
f 4 3-27 1	3
f / 3_28	ž
f 4 2 20	2
14.3-29	3
t 4.3-30 1	3
f 4.3-31 1	3
f 4.3-321	3
f 4.3-33	3
f 4 3-34 1	ž
f / 2 25	2
14.0-00	5
14.3-301	3
t 4.3-37 1	3
f 4.3-38 1	3
f 4.3-391	3
f 4.3-40 1	3
$f_{A} \xrightarrow{3} 41$	ž
f A 2 A 2	2
14.0-42	5
T 4.3-431	3
t 4.3-441	3
f 4.3-451	3
f 4.3-461	3
f 4 3-47	3
f A 2 A Q  1	2
14.3-40	5
14.3-49	3
1 4.3-50 1	3
f 4.3-52 1	3
f 4.3-511	3
f 4.3-53 1	3
f <u>4</u> 3-54	ž
г т т I f / ЭББ //	2
14.3-331	3
1 4.3-56	3

#### Page/Section

Page/Section	Rev
f 4.3-57	13
Section 4.4 t 4.4-1 (3 sheets) t 4.4-2 f 4.4-1 (2 sheets) f 4.4-2. f 4.4-3. f 4.4-3. f 4.4-4. f 4.4-5. f 4.4-6. f 4.4-6. f 4.4-7 (deleted) f 4.4-8. f 4.4-9. f 4.4-10. f 4.4-11	20 24 15 15 14 14 14 14 14 15 14 14 14
Section 4.5	16
Section 4.6	14
Section 4A f 4A-1 f 4A-2 f 4A-3 f 4A-4 f 4A-5	14 13 13 13 13 13
Section 5.1 t 5.1.2-1 (2 sheets) f 5.1.2-1 (3 sheets)	19 20 14
Section 5.2 t 5.2.1-1 t 5.2.3-1 (5 sheets) t 5.2.3-2 t 5.2.3-3 (2 sheets) f 5.2.5-1	19 13 15 13 20 13
Section 5.3 t 5.3.1-1 (2 sheets) t 5.3.1-2 t 5.3.1-3 t 5.3.1-4 t 5.3.1-5 t 5.3.1-6 deleted (Rev. 4) t 5.3.1-7 t 5.3.1-8 t 5.3.1-8 t 5.3.2-1 deleted (Rev. 4) t 5.3.2-2 (2 sheets) t 5.3.2-3 t 5.3.2-4 (2 sheets)	24 13 13 13 13 13 13 17 13 17 13 13 13 13 13

Page/Section	Rev
t 5.3.2-5 (2 sheets) t 5.3.3-1 t 5.3.3-2 t 5.3.3-3 f 5.3.3-1	
Section 5.4	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Section 6.1 t 6.1.1-1 (2 sheets) t 6.1.1-2 (2 sheets) t 6.1.2-1 (3 sheets) t 6.1.2-2 (10 sheets) t 6.1.2-3 t 6.1.2-4 f 6.1.2-1	15 14 15 14 20 14 14

# Page/Section

Section 6.2	24
t 6 2 1_1	23
t 6 2 1_2	.20
t 6 2 1 3	.20
t = 0.2.1 - 0	.20 22
t 6.2.1-4 (2 Sileets)	.20 12
10.2.1-0	. IS 12
LO.2. I-O	. 13
t 6.2.1-7 deleted (Rev. 0)	.13
10.2.1-8	.13
t 6.2.1-9 (6 sheets)	.13
t 6.2.1-10 (3 sheets)	.13
t 6.2.1-11 (7 sheets)	.13
t 6.2.1-11A	.13
t 6.2.1-12 (7 sheets)	.13
t 6.2.1-13 (7 sheets)	.13
t 6.2.1-14 (7 sheets)	.13
t 6.2.1-15 (7 sheets)	.13
t 6.2.1-16 (7 sheets)	.13
t 6.2.1-16A	.13
t 6.2.1-17 (5 sheets)	.13
t 6.2.1-18 (5 sheets)	.13
t 6 2 1-19 (5 sheets)	13
t 6 2 1-20 (5 sheets)	13
$t 6 2 1_{-}21 (5 sheets)$	13
t 6 2 1-22 (5 sheets)	13
t 6 2 1 23 (2 sheets)	13
t = 0.2.1 - 2.0 (2  sileets)	10
t 6.2.1-24 (2 Sheets)	10
t 0.2.1-25 (4 sheets)	.13
t 6.2.1-26 (7 sneets)	.13
t 6.2.1-26A	.13
t 6.2.1-27 (36 sheets)	.13
t 6.2.1-28 (7 sheets)	.13
t 6.2.1-28A (7 sheets)	.13
t 6.2.1-29	.23
t 6.2.1-30	.23
t 6.2.1-31	.23
t 6.2.1-32	.23
t 6.2.1-33	.23
t 6.2.1-34	.23
t 6.2.1-35	.23
t 6.2.1-36	.23
t 6.2.1-37	.23
t 6.2.1-38	.23
t 6 2 1-39	23
t 6 2 1-40	23
t 6 2 1-41	23
t 6 2 1-47	.20
t 6 2 1_/3	20. 22
t = 0.2, $1-40$	د2. 12
10.2.1-44 intrough 09 deteted (Rev. 4)	10. 15
10.2.1-00	CI.
10.2.1-01 (9 SILEUS)	10
$t 0.2.1-02$ ( $\delta$ Sileets)	.13
ι ο.∠. ι-ος ueleteu (κεν. /)	.13

Page/Section	Rev
t 6.2.1-64 (3 sheets)	
t 6.2.1-65	
t 6.2.1-66	
t 6.2.1-67	
t 6.2.1-68	
t 6.2.1-69	
t 6 2 1-70	13

t 6.2.1-65	8
t 6.2.1-66	5
t 6.2.1-67	3
t 6.2.1-68	3
t 6 2 1-69 1	3
t 6 2 1-70 1	3
t 6 2 1-71 1	3
t 6 2 1-72 (2 sheets)	3
t 6 2 1-73	3 2
1621-701	3
t 6 2 1_75	3
16221-75	2
10.2.2-1	2
10.2.2-2	ა ი
t 6.2.2-3 (5 Sheets)	ა ი
10.2.2-4	3
t 6.2.2-5 (2 sneets)	3
t 6.2.4-1 (17 sheets)2	4
t 6.2.4-2 (15 sheets)2	4
t 6.2.5-1 1	3
t 6.2.5-2 (2 sheets)1	3
t 6.2.5-3 1	3
t 6.2.5-41	3
t 6.2.5-51	3
t 6.2.5-6 (4 sheets)1	5
t 6.2.5-7 deleted (Rev. 0)1	3
t 6.2.5-8 deleted (Rev. 0)1	3
t 6.2.5-9	3
t 6 2 5-10 1	3
t 6 2 6-1 (2 sheets)	3
f 6 2 1-1 2	3
f 6 2 1-2 2	3
f 6 2 1_3	2
$f = 6 2 1_{-1}$	2
f 6 2 1 5	2
10.2.1-52 f6.2.1.6	0
10.2.1-02 fc 0.1.7	ວ ວ
10.2.1-7	.ວ ດ
10.2.1-8 infough - 14 deleted (Rev. 4) 1	3
1 6.2.1-15 (65 sneets)1	3
f 6.2.1-16 (3 sneets)1	3
f 6.2.1-16A (3 sheets)1	3
f 6.2.1-17 (6 sheets)1	3
t 6.2.1-1/A1	3
f 6.2.1-18 (74 sheets)1	3
f 6.2.1-19 (74 sheets)1	3
f 6.2.1-20 (74 sheets)1	3
f 6.2.1-21 (74 sheets)1	3
f 6.2.1-22 (74 sheets)1	3
f 6.2.1-23 (74 sheets)1	3
f 6.2.1-24 1	3
f 6.2.1-25 (8 sheets)1	3
f 6.2.1-25A (8 sheets)1	3
f 6.2.1-26	5

Page/	Section
-------	---------

t 6.3.2-1 (3 sheets)......24 t 6.3.2-3 (3 sheets).....23 t 6.3.2-4 (2 sheets)......14 t 6.3.2-5 (9 sheets).....14 t 6.3.2-7 (4 sheets)......23 t 6.3.2-9 (5 sheets).....14 t 6.3.3-1......14 t 6.3.3-2.....14 f 6.3.2-1 (22 sheets).....15 f 6.3.2-2.....14 f 6.3.2-3.....14 f 6.3.2-4......14 f 6.3.2-5......23

 Section 6.5
 21

 t 6.5.1-1 (3 sheets)
 13

 t 6.5.1-2 (2 sheets)
 13

 t 6.5.1-3 (3 sheets)
 13

 t 6.5.2-1
 13

 t 6.5.2-2 (2 sheets)
 21

 t 6.5.3-1
 13

 f 6.5.2-2 (2 sheets)
 21

 t 6.5.2-1
 13

 f 6.5.2-2 (2 sheets)
 21

 t 6.5.2-3
 13

Section 6A.....24

Rev
-----

Page/Section
--------------

Section 7.1
Section 7.2.       22         t 7.2.1-1 (2 sheets)       14         t 7.2.1-2 (2 sheets)       19         t 7.2.1-3 (3 sheets)       19         t 7.2.1-4 (3 sheets)       19         t 7.2.1-5 (2 sheets)       19         t 7.2.2-1 (5 sheets)       17         f 7.2.1-1       14         f 7.2.1-2       14
Section 7.3.24t 7.3.1-114t 7.3.1-2 (2 sheets)14t 7.3.1-2 (2 sheets)14t 7.3.1-3 (2 sheets)14t 7.3.1-414t 7.3.1-514t 7.3.1-6 (6 sheets)24t 7.3.1-7 (2 sheets)19t 7.3.3-1 (2 sheets)19t 7.3.5-114t 7.3.6-114t 7.3.6-214t 7.3.8-1 (2 sheets)14t 7.3.10-114t 7.3.10-114t 7.3.12-1 (2 sheets)14t 7.3.13-1 (2 sheets)14t 7.3.15-1 (2 sheets)14t 7.3.16-114t 7.3.17-114f 7.3.1-114f 7.3.1-114f 7.3.1-214
Section 7.4
Section 7.5.       14         t 7.5.2-1 (14 sheets)       24         t 7.5.2-2       14         t 7.5.2-3       14         t 7.5.3-1       14         t 7.5.3-2 (2 sheets)       14         t 7.5.3-3       14         t 7.5.3-4 (4 sheets)       14

Pane	150	ctic	۱n
i ayu	100	ouc	

t 7.5.3-5	1 1 1
Section 7.6	)       
Section 7.7       22         t 7.7.1-1       17         f 7.7.1-1       13         f 7.7.1-2       17         f 7.7.1-3       13         f 7.7.1-4       13         f 7.7.1-5       13         f 7.7.1-6       18         f 7.7.1-8       13         f 7.7.1-9       13         f 7.7.1-10       14         f 7.7.2-1       13	2737333333433
Section 8.1	3 1
Section 8.2       22         t 8.2.1-1 (2 sheets)       22         t 8.2.1-2       14         t 8.2.1-3       22         t 8.2.2-1       14         t 8.2.2-2       14	22424
Section 8.3       24         t 8.3.1-1 (3 sheets)       14         t 8.3.1-2       16         t 8.3.1-3 (12 sheets)       14         t 8.3.1-4 (6 sheets)       14         t 8.3.2-1       24         t 8.3.2-2       23         t 8.3.2-3       13         t 8.3.2-4       13         t 8.3.2-5 (10 sheets)       13         t 8.3.2-7       23         t 8.3.2-8       13         t 8.3.2-9       13         f 8.3.1-1 (35 sheets)       22	44344333343332
Section 8.4	1

Section 9.1	3333)))))))
f 9.1.2-7 (2 sheets)	333
f 9.1.2-5	333
f 9.1.4-4	3333
f 9.1.4-8	323
f 9.1.5-2	3
5 ection 9.2	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$
t 9.2.2-3 (13 sheets)	· · · · · · · · · · · · · · · · · · ·
t 9.2.5-3 (2 sheets)	5555
t 9.2.5-7 (2 sheets)	, , , , , , , , , , , , , , , , , , , ,
t 9.2.8-1	3

Page/Section
--------------

t 9.2.9-1 t 9.2.9-2. t 9.2.9-3 (2 sheets) t 9.2.10-1 t 9.2.11-1	13 13 13 23 22
Section 9.3. t 9.3.1-1 (2 sheets)	22 13 13 13 13 13 13 13 13 13 13 13 13 13
Section 9.4 t 9.4.1-1 t 9.4.1-2 (2 sheets) t 9.4.1-3 (5 sheets) t 9.4.1-4 (2 sheets) t 9.4.2-1 (2 sheets) t 9.4.2-2 (4 sheets) t 9.4.3-1 t 9.4.3-2 (3 sheets) t 9.4.3-3 t 9.4.3-4 (11 sheets) t 9.4.3-5 (6 sheets) t 9.4.3-6. t 9.4.3-7. t 9.4.3-7. t 9.4.4-1 (7 sheets) t 9.4.5-1. t 9.4.5-2 (10 sheets) t 9.4.6-1 t 9.4.6-3. t 9.4.6-3. t 9.4.6-4 (4 sheets)	23 $14$ $14$ $14$ $14$ $14$ $14$ $14$ $14$
t 9.4.7-2 (3 sheets) t 9.4.8-1 t 9.4.8-2 (2 sheets) t 9.4.9-1 t 9.4.9-2	14 14 14 14 14 14

t 9.4.9-3 (4 sheets)
Section $9.5.$ 22 $t 9.5.1-1$ ( $81$ sheets)21 $t 9.5.1-2$ 13 $t 9.5.1-2$ 13 $t 9.5.1-3$ 13 $t 9.5.1-3$ 13 $t 9.5.1-4$ 13 $t 9.5.1-5$ deleted (Rev. 0)13 $t 9.5.1-6$ 16 $t 9.5.1-7$ (2 sheets)19 $t 9.5.1-8$ 13 $t 9.5.1-9$ (12 sheets)23 $t 9.5.1-9$ (12 sheets)24 $t 9.5.1-10$ (11 sheets)24 $t 9.5.1-10A$ (12 sheets)24 $t 9.5.1-10B$ (5 sheets)13 $t 9.5.1-10C$ (10 sheets)13 $t 9.5.1-10D$ 13 $t 9.5.2-1$ 13 $t 9.5.3-2$ (2 sheets)13 $t 9.5.4-1$ (2 sheets)13 $t 9.5.4-1$ (2 sheets)13 $t 9.5.4-2$ (6 sheets)13 $t 9.5.5-2$ (4 sheets)13 $t 9.5.6-2$ (5 sheets)13 $t 9.5.7-3$ 13 $t 9.5.7-2$ (6 sheets)13 $t 9.5.8-1$ 13 $t 9.5.9-1$ 13 $t 9.5.1-1$ 21
Section 9A
9A.1.1       14         9A.1.2       19         9A.1.3       14         9A.1.4       14         9A.1.5       14         9A.1.6       14         9A.1.7       15         9A.1.8       14         9A.1.9       14         9A.1.10       14         9A.1.11       (deleted)         14       9A.1.12         14       9A.1.13

# Page/Section

Page/Section
--------------

9A.1.14	15
9A.1.15	14
9A 1 16	14
$0\Delta 1 17 (deleted)$	14
$0^{1}$ 1 18 (deleted)	11
	···· 14
9A.1.19	14
9A.1.20	14
9A.1.21	14
9A.1.22	14
9A.1.23	14
9A.1.24	14
9A.1.25	14
9A 1.26	14
9A 1 27	14
QΔ 1 28	14
0A 1 20 (deleted)	11
9A. 1.29 (deleted)	14
9A. 1.30 (deleted)	14
9A.1.31	14
9A.1.32	14
9A.1.33	24
9A.1.34 deleted (Rev. 0)	14
9A.1.35	14
9A.1.36 (deleted)	14
9A 1 37	14
9A 1 38 (deleted)	14
	20
0. 1. 1. 0.	20
9A   40	
$0 \wedge 1 \wedge 1 = 0$	
9A.1.41 deleted (Rev. 0)	14
9A.1.41 deleted (Rev. 0) 9A.1.42	14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43	14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44	14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45	14 14 14 20 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.46	14 14 14 20 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44. 9A.1.45 9A.1.46 9A.1.47.	14 14 20 14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42. 9A.1.43. 9A.1.44. 9A.1.45. 9A.1.45. 9A.1.46. 9A.1.47. 9A.1.48.	14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42. 9A.1.43. 9A.1.44. 9A.1.45. 9A.1.45. 9A.1.46. 9A.1.47. 9A.1.48. 9A.1.48.	14 14 20 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.48 9A.1.49 9A.1.49 9A.1.50	14 14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51	14 14 14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51 9A.1.51	14 14 14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51 9A.1.52 0A.1.52	14 14 14 14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51 9A.1.51 9A.1.53 9A.1.53	14 14 14 14 14 14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51 9A.1.51 9A.1.52 9A.1.53 9A.1.54	14 14 14 14 14 14 14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51 9A.1.51 9A.1.52 9A.1.53 9A.1.54 9A.1.55	14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51 9A.1.51 9A.1.52 9A.1.53 9A.1.54 9A.1.55 9A.1.56	14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 120 120
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51 9A.1.51 9A.1.52 9A.1.53 9A.1.54 9A.1.55 9A.1.55 9A.1.56 9A.1.57	14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.44 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51 9A.1.51 9A.1.52 9A.1.52 9A.1.54 9A.1.55 9A.1.55 9A.1.56 9A.1.57 9A.1.58	14 14 14 14 14 14 14 14 14 14 14 14 14 20 20 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.43 9A.1.45 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51 9A.1.51 9A.1.52 9A.1.52 9A.1.55 9A.1.55 9A.1.55 9A.1.56 9A.1.57 9A.1.58 9 A.1.58 9 A.1.59 deleted (Rev. 0)	14 14 14 14 14 14 14 14 14 14 14 20 14 20 14 21 14
9A.1.41 deleted (Rev. 0)	14 14 14 14 14 14 14 14 14 14 14 20 14 20 14 21 14 14
9A.1.41 deleted (Rev. 0) 9A.1.42 9A.1.43 9A.1.43 9A.1.45 9A.1.45 9A.1.45 9A.1.46 9A.1.47 9A.1.48 9A.1.49 9A.1.50 9A.1.51 9A.1.51 9A.1.52 9A.1.52 9A.1.55 9A.1.55 9A.1.55 9A.1.56 9A.1.57 9A.1.58 p 9A.1.59 deleted (Rev. 0) p 9A.1.60 9A.1.60	14 14 14 14 14 14 14 14 14 14 14 20 14 20 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0)	14 14 14 14 14 14 14 14 14 14 14 20 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0)	14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0)	14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0)	14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14
9A.1.41 deleted (Rev. 0)	14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14

9A.1	.66
9A.1	.67
9A.1	.68
9A.1	.69
9A.1	.70
9A.1	.71
9A 1	72 14
9A 1	73 17
9A 1	74 14
QΔ 1	75 14
QΔ 1	76 14
0Δ 1	77 1/
0/ 1	78 1/
0/ 1	70 11
9A.1	.79
9A.1	.00
9A. I	.01
9A. I	.02
9A.1	.83
9A.1	.84
9A.1	.85
9A.1	.86
9A.1	.87
9A.1	.88
9A.1	.89
9A.1	.90
9A.1	.91 deleted (Rev. 0)14
9A.1	.92 deleted (Rev. 0)14
9A.1	.93
9A.1	.94
9A.1	.95
9A.1	.96
9A.1	.97 deleted (Rev. 4)14
9A.1	.98 deleted (Rev. 4)14
9A.1	.99 deleted (Rev. 4)14
9A.1	.100 deleted (Rev. 0)14
9A.1	.101
9A.1	.102 deleted (Rev. 0)14
9A.1	.103
9A.1	.104
9A.1	.105
9A.1	.105A
9A.1	.106
9A.1	.107
9A 1	108  deleted (Rev 0) 14
9A.1	.109
9A 1	.110
9A.1	.111
t 9A	1.111-1 (2 sheets)
9A 1	.112-1 deleted (Rev. 0) 14
9A 1	113 14
9A 1	114 14
<u>9</u> Δ 1	115 14
<u>9</u> Δ 1	116 14
5, 1, 1	TT

# Page/Section

9A.1.117	14
9A 1 118	14
9A 1 119	14
9A 1 120 deleted (Rev. 0)	1/
0A 1 121	14
9A.1.121	14
9A.1.122	14
9A.1.123	.14
9A.1.124	.14
9A.1.124A deleted	16
9A.1.124B deleted	16
9A 1 125	14
9A 1 126	14
ΩΛ 1 126Λ	11
9A. 1. 120A	14
9A.1.127 (deleted)	14
9A.1.128 (deleted)	14
9A.1.129	.14
9A.1.130	.14
9A.1.131 deleted (Rev. 0)	.14
9A.1.132	16
	-
QA 2 1	20
ΟΛ 2 2	15
	10
9A.2.3	14
9A.2.4	.14
9A.2.5	.14
9A.2.6	.14
9A.2.7	15
9A.2.8	15
9A 2 9	14
9A 2 10	14
ΩΔ 2 11	1/
0Λ 2 12	20
9A.2.12	20
9A.2.13	1D
9A.2.14	14
9A.2.15	14
9A.2.16	15
9A.2.17	15
9A.2.18	14
9A.2.19	14
9A 2 20	14
9A 2 21	14
0Λ 2 22	11
	14
98.2.23	14
9A.2.24	15
9A.2.25	24
9A.2.26	.14
9A.2.27	14
9A.2.28	14
9A.2.29	20
9A 2 30	14
QΔ 2 31	15
ΟΛ 2 22	20
JT. L. J. J. L. J.	20
YA.Z.JJ	15

9A.2.34	14
9A.2.35	14
9A.2.36	14
9A.2.37	15
9A.2.38	14
9A.2.39	14
9A.2.40	14
9A.2.41	20
9A.2.42	14
9A.2.43	20
9A.2.44	20
9A.2.45	14
9A.2.46	14
9A.2.47	15
9A.2.48	IƏ
9A.2.49	14
9A.2.50	14
9A.2.51	14
9A.2.52 QΔ 2 53	14
QΔ 2 5 <i>1</i>	14
9A 2 55	10
9A 2 56	10
9A 2 57	
9A.2.58	
9A.2.59	14
9A.2.60	22
9A.2.61	14
9A.2.62	14
9A.2.63	14
9A.2.64	14
9A.2.65	15
9A.2.66	14
9A.2.67	14
9A.2.68	15
9A.2.69	14
9A.2.70	20
9A.2./1	15
9A.2.72	14
9A.2.73	14
9A.2.74	14
9A.2.75	14
9A.2.70	24
0A 2 77	14
9A.2.77	14
9A.2.70 0Δ 2 70	14
9A 2 80	1 <del>4</del>
9A 2 81	14
9A 2 82	14
9A.2.83	
9A.2.84	
9A.2.85	14

Page/Section	Rev
9A.2.86	14
9A.2.87	14
9A.2.88	15
9A.2.89	15
Section 9B	24
Section 10.1	22
t 10.1-1 (3 sheets)	17
Section 10.2 t 10.2.2-1 t 10.2.2-2 f 10.2.2-1 f 10.2.2-2 f 10.2.2-2	24 14 22 22 22
Section 10.3	24
t 10.3.2-1 (2 sheets)	14
t 10.3.2-2	14
t 10.3.2-3	14
t 10.3.3-1 (13 sheets)	24
t 10.3.5-1 (4 sheets)	20
Section 10.4 t 10.4.1-1 (2 sheets) t 10.4.2-1 t 10.4.5-1 t 10.4.6-2 (2 sheets) t 10.4.6-2 (2 sheets) t 10.4.6-3 t 10.4.6-4 (2 sheets) t 10.4.7-1 t 10.4.7-2 (2 sheets) t 10.4.7-3 t 10.4.8-1 t 10.4.8-2 t 10.4.8-3 t 10.4.8-3 t 10.4.8-4 (3 sheets) t 10.4.9-1 (2 sheets) t 10.4.9-3 (4 sheets) t 10.4.9-5 (2 sheets) t 10.4.10-1 f 10.4.8-1	23 14 14 14 14 14 14 14 14 14 14 14 15 14 15 14
Section 10A	14
t 10A-1 (2 sheets)	14
t 10A-2	14
t 10A-3	14
t 10A-4 (3 sheets)	14
t 10A-5	14
f 10A-1	14

. -

Page/Section Rev	
f 10A-2	•
Section 11.1	
Section 11.2.       22         t 11.2.1-1 (2 sheets)       13         t 11.2.1-2 (8 sheets)       22         t 11.2.1-3 (10 sheets)       13         t 11.2.1-4       13         t 11.2.2-1       13         t 11.2.3-1       15         t 11.2.3-2 (2 sheets)       13         t 11.2.3-3 (2 sheets)       13         t 11.2.3-4       13         f 11.2.3-1 (7 sheets)       13	
Section 11.3.       16         t 11.3.2-1       16         t 11.3.2-2       16         t 11.3.2-3       16         t 11.3.2-4 (2 sheets)       16         t 11.3.2-5 (2 sheets)       16         t 11.3.2-6 (2 sheets)       16         t 11.3.2-7 (2 sheets)       16         t 11.3.3-1       14         t 11.3.3-2       15         t 11.3.3-3 (2 sheets)       14         t 11.3.3-4 (6 sheets)       14         f 11.3.2-3       16         f 11.3.2-4       14         f 11.3.2-5       14         f 11.3.2-6       14         f 11.3.2-6       14	
Section 11.4	

Page/Section
--------------

Rev
-----

t 11.4.2-2 (2 sheets)	14
t 11.4.2-3 (2 sheets)	14
t 11.4.2-4	14
t 11.4.2-5	14
t 11.4.2-6	14
t 11.4.2-7	19
Section 11.5	24
t 11.5.2-1 (4 sheets)	19
t 11.5.2-2 (3 sheets)	16
t 11.5.2-3 (2 sheets)	16
t 11.5.2-4 deleted (Rev. 0)	14
t 11.5.2-5 (2 sheets)	16
t 11.5.3-1 (2 sheets)	24
t 11.5.3-2.	14
t 11.5.5-1 (2 sheets)	24
t 11.5.5-2 (2 sheets)	24
t 11.5.5-3.	24
f 11.5.2-1	20
f 11.5.2-2a	19
f 11.5.2-2b	19
f 11.5.2-3	14
f 11.5.2-4a	19
f 11.5.2-4b	19
f 11.5.2-5	14
f 11.5.2-6	14
Section 12.1	23

Section 12.2	19
t 12.2.1-1	14
t 12.2.1-2	14
t 12.2.1-3	14
t 12.2.1-4	14
t 12.2.1-5	14
t 12.2.1-6 (15 sheets)	14
t 12.2.1-7	14
t 12.2.1-8	14
t 12.2.1-9	14
t 12.2.1-10	14
t 12.2.1-11	14
t 12.2.1-12	14
t 12.2.1-13	14
t 12.2.1-14 deleted (Rev. 4)	14
t 12.2.1-15	14
t 12.2.1-16 deleted (Rev. 4)	14
t 12.2.1-17 deleted (Rev. 4)	14
t 12.2.1-18	14
t 12.2.1-19 (2 sheets)	14
t 12.2.1-20	14
t 12.2.1-21	14
t 12.2.1-22	14

$\begin{array}{c} t \ 12.2.1-23 \ \dots \ 14 \\ t \ 12.2.1-24 \ deleted \ (Rev. \ 4) \ \dots \ 14 \\ t \ 12.2.1-25 \ \dots \ 14 \\ t \ 12.2.1-26 \ \dots \ 14 \\ t \ 12.2.1-27 \ \dots \ 14 \\ t \ 12.2.1-28 \ \dots \ 14 \\ t \ 12.2.1-29 \ \dots \ 14 \\ t \ 12.2.1-30 \ \dots \ 14 \\ t \ 12.2.1-31 \ deleted \ (Rev. \ 4) \ \dots \ 14 \\ t \ 12.2.1-32 \ deleted \ (Rev. \ 4) \ \dots \ 14 \\ t \ 12.2.1-32 \ deleted \ (Rev. \ 4) \ \dots \ 14 \\ t \ 12.2.1-32 \ deleted \ (Rev. \ 4) \ \dots \ 14 \\ t \ 12.2.1-33 \ \dots \ 14 \\ t \ 12.2.1-36 \ \dots \ 14 \\ t \ 12.2.1-36 \ \dots \ 14 \\ t \ 12.2.1-37 \ \dots \ 14 \\$
112.2.1-39       14         t       12.2.1-40       14         t       12.2.1-41       14         t       12.2.1-42       (2 sheets)       14         t       12.2.1-42       (2 sheets)       14         t       12.2.1-43       14       14         t       12.2.1-43       14       14         t       12.2.1-45       14       14         t       12.2.1-45       14       14         t       12.2.1-47       (2 sheets)       14         t       12.2.1-47       (2 sheets)       14         t       12.2.1-47       (2 sheets)       14         t       12.2.1-50       14       14         t       12.2.1-51       14       14         t       12.2.1-52       14       14         t       12.2.2-2       (2 sheets)       14         t       12.2.2-2       (2 sheets)       14         t       12.2.1-1       14         f       12.2.1-2       14         f       12.2.1-2       14
Section 12.3
Section 12.4

Page/Section

# Page/Section

t 12.4.1-8 (2 sheets) t 12.4.1-9 t 12.4.1-10 (2 sheets) t 12.4.1-11 t 12.4.1-12 (2 sheets) t 12.4.1-13 t 12.4.1-14 t 12.4.1-15 t 12.4.1-16 (2 sheets) t 12.4.3-1 t 12.4.3-2 t 12.4.3-3 t 12.4.4-1 t 12.4.4-2	13 13 13 13 13 13 13 13
Section 12.5 t 12.5.2-1 t 12.5.2-2 t 12.5.2-3 t 12.5.2-4	23 14 14 14 17
Section 13.1 t 13.1.3-1 (3 sheets) f 13.1.1-1 f 13.1.1-2. f 13.1.2-1	24 24 22 19 18
Section 13.2	24
Section 13.3	14
Section 13.4	15
Section 13.5 f 13.5.1-1	16 14
Section 13.6	14
Section 13.7	22
Section 13.8	22
Section 14.1	14
Section 14.2 t 14.2.1-1 (7 sheets) t 14.2.1-2 (4 sheets) f 14.2.7-1	15 14 14 14
Section 15.0 t 15.0.3-1 t 15.0.3-2 (3 sheets) t 15.0.3-3	23 15 23 15

$\begin{array}{cccccccccccccccccccccccccccccccccccc$
f 15.0.1-11
f 15.0.1-12
f 15.0.1-14
f 15.0.1-15
f 15.0.1-16
f 15.0.1-18
f 15.0.1-19
f 15.0.1-20
f 15.0.1-22
f 15.0.1-23
f 15.0.1-25
f 15.0.4-1
f 15.0.5-1
f 15.0.5-2 14
f 15.0.5-3
f 15.0.6-2
Section 15.1 24
t 15.1.2-1 (3 sheets)23
t 15.1.5-1 (2 sheets)14
t 15.1.5-3
f 15.1.2-1
t 15.1.2-2
f 15.1.3-1
f 15.1.3-2
f 15.1.3-3
1 10.1.0-4

Rev	Page/Section

f 15.1.3-514
f 15.1.3-614
f 15.1.3-714
f 15.1.3-814
1 15.1.4 - 114
f 15.1.4-3
f 15.1.4-414
f 15.1.4-514
f 15.1.5-023
f 15.1.5-123
1 10.1.0-223 f 15 1 5_3
f 15.1.5-4
f 15.1.5-523
f 15.1.5-623
f 15.1.5-723
f 15.1.5-823
f 15.1.5-923 f 15.1.5.10
f 15 1 5-11 23
10110 11 11 120
Section 15.221
t 15.2.3-1 (5 sheets)21
t 15.2.6-1 (2 sheets)15
[ 15.2.6-2
f 15 2 3-2 14
f 15.2.3-3
f 15.2.3-414
f 15.2.3-514
f 15.2.3-614
f 15.2.3-714
f 15.2.3-814
f 15 2 6-2 21
f 15 2 7-1 21
f 15.2.7-221
f 15.2.8-114
f 15.2.8-214
f 15.2.8-314
† 15.2.8-4
1 15.2.8-5
f 15 2 8-7 14
f 15.2.8-8
Section 15.321
t 15.3.1-1 (2 sheets)14
t 15.3.3-1
t 15.3.3-2 (2 Sileets)
f 15 3 1-1

$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
Section 15.419t 15.4.1-1 (4 sheets)15t 15.4.6-114t 15.4.8-114t 15.4.8-2 (2 sheets)15t 15.4.8-315f 15.4.1-114f 15.4.1-214f 15.4.1-314f 15.4.2-114f 15.4.2-214f 15.4.2-314f 15.4.2-414f 15.4.2-514f 15.4.2-614f 15.4.2-714f 15.4.2-814f 15.4.2-914f 15.4.3-114f 15.4.3-214f 15.4.4-314f 15.4.4-314f 15.4.4-314f 15.4.7-214f 15.4.7-314f 15.4.7-314f 15.4.7-514f 15.4.8-114f 15.4.8-314f 15.4.8-314f 15.4.8-314f 15.4.8-314f 15.4.9-314
Section 15.5

# Page/Section

Page/Section
--------------

$\Omega_{\rm continue} = 4 \Gamma_{\rm c} \Omega_{\rm c}$
Section 15 b 23
t 15.6.1-114
t 15.6.2-1
+ 15 6 2 2 15
1 15.0.2-2
t 15.6.2-315
t 15 6 2-4 15
l 15.0.3-1
t 15.6.3-224
t 15 6 3-3 23
(10.0.0-020)
t 15.6.3-4 (2 sneets)23
t 15.6.3-5
t 15 6 2 6 (deleted) 15
(15.0.3-0 (deleted)
t 15.6.3-7 (deleted)15
t 15 6 3-8 15
t 15.0.0 0 (deleted)
t 15.6.3-9 (deleted)15
t 15.6.3-10
t 15 6 3-11 23
(15.0.5-11
t 15.6.5-116
t 15 6 5-2 14
+ 15 6 5 2 14
l 15.0.5-5
t 15.6.5-4
t 15 6 5-5 16
l 15.0.5-014
t 15.6.5-714
t 15 6 5-8 14
(10.0.0-01+
t 15.6.5-9 (4 sneets)15
t 15.6.5-10 (7 sheets)
t 15.6.5 11 (2  sheets) 16
(15.0.5-11 (2 Sileets)
t 15.6.5-1215
f 15 6 1-1 14
f 15 6 1 2 1/
1 15.0.1-2
f 15.6.3-123
f 15 6 3-2 23
f 1E C 2 2 22
1 15.0.3-3
f 15.6.3-4
f 15 6 3-5 23
f 15.6.3-5
f 15.6.3-523 f 15.6.3-623
f 15.6.3-5
f 15.6.3-5       23         f 15.6.3-6       23         f 15.6.3-6       23         f 15.6.3-7       23         f 15.6.3-8       23         f 15.6.3-9       23         f 15.6.3-10       23         f 15.6.3-11       23         f 15.6.3-12       14         f 15.6.3-13       23         f 15.6.5-1       14         f 15.6.5-2 (9 sheets)       14         f 15.6.5-3 (9 sheets)       14         f 15.6.5-4 (9 sheets)       14         f 15.6.5-5 (9 sheets)       14         f 15.6.5-6 (9 sheets)       14         f 15.6.5-7 (9 sheets)       14         f 15.6.5-7 (9 sheets)       14         f 15.6.5-7 (9 sheets)       14         f 15.6.5-8 (9 sheets)       14

Section 15.7
Section 15.8
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Section 15B14
Section 16.114

Page/Section	
--------------	--

Page/Section

Section 16.2	14
Section 16.3 t 16.3-1 (3 sheets) (deleted) t 16.3-2 (6 sheets) (deleted) t 16.3-3A (2 sheets) (deleted) t 16.3-3B (2 sheets) (deleted) t 16.3-4 (15 sheets) (deleted) t 16.3-5 (30 sheets) (deleted) t 16.3-6 (2 sheets) (deleted) t 16.3-7 (2 sheets) (deleted)	19 17 14 14 14 14 17 15 14
Section 16.4	21
Section 17.1 deleted (Rev. 0)	13
Section 17.2 f 17.2.1-1 (deleted)	15 15
Section 18.1 t 18.1-1 t 18.1-2 t 18.1-3 (2 sheets) f 18.1-1	22 14 14 14 14
Section 18.2	14
Section 18.3 f 18.3-1 (2 sheets)	14 14
Section 19.1	16
Section 19.2	23
Section 19.3	16
Section 19.4	22